



# < c`\\]a `fl5i ghfU`]U'L'Dhm@hX` Ö` } || ^ÂJæ) åÂÛ`æ\^/ Ò} çã[ } { ^} œ\$AT æ) æ\*^{ ^} ôÂJdæe^\*^/

R″|^ÁG€GFÁ

WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION

Á

V@#Á^][¦dxÁ@ze-Áa^^}Á,¦^] æ'^åÁa^ÁÕPÖÁ{¦¦Á?[|&ã[ ÁÇÁCE •dæ)áæÁDÁÚc´ÁŠca/Áæ)åÁ{ æ`Á;}|^Áa^Á •^åÁ æ)åÁ^|āѷåÁ;}Áa^Á?[|&ã[ ÁÇÁCE •dæ)áæÁDÁÚc´ÁŠca'Á{[¦Ás@∘Á,`¦][•^Áæt'¦^^åÁa^ç, ^^}ÁÕPÖÁæ)åÁs@Á P[|&ã[ ÁÇÁCE •dæ)áæÁDÁÚc´ÁŠca/Áæ•Á ^oA;`ó4\$)Á ^ &cā[}ÁFÈFÁ;-Ás@á\*Á^][¦dÉÁ

ÕPÖÁ(c@o¦,ã^A&aã&akaaā(•Á^•][}•ãaājāčÁ(Áa)^^\•[}Á(c@oká@a)ÁP[|&ã(ÁQAE•daajaaAÚCAÁCAÁ æarāj\*ÁşiÁ&[}}^&cā)+Á,ão@áo@aÁ^][¦džŐPÖÁadp+[Á^¢&\`å^•Áa[]|&råÁ,æoka)cā\•Áa)åÁ&[}åãaā[}•É&(Á c@ Á¢cc}oÁ(^\*æ)^Á,^¦{ã•āa}^EÅ

V@:Á^¦ça&^•Á`}å^¦cæa`^}Á\$i^ÁÕPÖÁ\$jÁ\${[}}^&caj;}Å āc@á,¦^]ædāj\*Ás@aiA^][¦cá,^¦^Áaj ãz^åÁt[Ás@[•^Á •]^&ãã&ed|^Á\$i^cæaj^åÁ\$jÁs@:Á^][¦cáse)åÁse'^Á`àb^&cát[Ás@:Á8[]^Áaj ãzæaja]•Á<?cátjÁs@:Á^][¦dÉÁ

V@A[] ājā[}•Ê\$&[}&|`•ā[}•Ásej å Ásej ^Á^&[{ { ^} å aceati[}•Ásj Ás@arÁ^][¦dése'^Ásiæe ^å A[}Á&[}åãati[}•Á ^}&[`}c^\^å Ásej å Ásj -{[¦{aceati[}}Á^çã], ^å Ásec Ás@arÁ(A, A, \^];]æbaceati[}A[, As@A^][¦dÉktÕPÖÁ@ee A, [Á !^•][}•ãa ājāĉ A[¦A[à]ā\*aceati[}Át[Á]å\*ace^Ás@arÁ^][¦dÁt[Áse&&[č]}dÁt[¦Á?ç^}oA[¦Á&@ee]\*^•A[&&č]|ia]\*Á •čà•^č`^}oAt[Ás@Asace^Ás@eenÁs@A^][¦dÁ;æA][¦dÁ;æAs[č]

V@A[]∄jā[}•É&[}&|`•ā[}•A⇔)åA⇔)^Á^&[{{ ^} åæaā[}•AsjAs@aāA^][¦oÁ⇔A&aæa^AaA;}Aæ•\*{]cā[}•Á {æå^Aa^Ao^AOPÖAa^•&¦ãa^åAsjAs@aA^][¦dĚÁKÕPÖA&ä&a}aaā[•Áãæaā]ãô Á⇔ãa]\*Á;[{Á⇔)^A[-Ás@-Á æ•\*{]cā[}•Áa^ā]\*Asj&[¦|^&dĚA

ÕPÖÁ@æe Á, ¦^] æi^åÁs@æiÁ^][¦qá, }Ás@ Ásæe æi Á, Ásj-{|;{ ææā;}}Á, ![çãa^åAsî^AP[|&ã; ÁqÃOE • dæjãæADÁÚĉ Á Šcå Áæj å Á; c@ ¦•Á, @;Á, ![çãa^åÁsj-{|;{ ææā;}}Át; ÁÕPÖÁçāj&]čáāj\*ÁÕ[ç^!}{ ^}oÁsĕ c@; !ãa?•DÉÅ, @3&@Á ÕPÖÁ@æe Á;[o45jå^]^}å^}d^Áç^!ãa?åÁ;!Á&@ &\^åAsi^[}åÁs@ Ásé !\^^åAs[]^Á;-Á, [!\ÈÕPÖÁs[[^•Á }[oÁse&&^]oÁãæàājãĉ ÁsjÁ&[}}^&aaj; Á;ã@Áč&@Á`}ç^!ãa?åÁsj-{!{ ææā;}ÊÉsj&]čáj\*Á'!![!•Ásè)åÁ;{ ã•ã;}•Á ã;Ás@Á^][!dÁ;@3&@Á, ^!^Á&eĕ • ^åÁsî^Á!![!•Á;!Á;{ ã•ã;}\*Ásj & Ko®emásj-{];{ ææā;}ÈÉsj&]čáj

А	
Á	

Á

# HUV`Y`cZVt2bhYbhg`

FÈ	_Qd[ å`&đį} }
	FÈG` ÒT ÙÁ, àb &cãç^• A <del>ll III III III III III III III III III </del>
GÈ	Ú¦[ ២&&&^& ] @
	GÈÈ S^^Áãc∕Á§ -{¦{ æaā]}Á
	ŒÈG` Ú¦[b^&cA∱ç^¦çæ], <b>Á⊞ШШШШШШШШШШШШШШШШШШШШШШШШШШШШШШШШШШШ</b> Ã`
	ŒÌHĨ Ùã∿Á&@⇔æ&ơ\¦ã cãe Â <u>IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII</u>
	GÈÈ Ò¢dæ&dā;}Á;∧d@;å[∥*^Á <del>àààààààààààààààààààààààààààààààààààà</del>
	ŒĔ Ú¦[ &∿••ậ * A <del>ddinininininininininininininininininini</del>
	ŒË Pæĭ∣æ*∧Á[čơ•Áæ)åÁ;¦[&∿åč¦^•Á <del>addadadadadadadadadadadadadadadadadada</del>
	QĒĽ Ú æ) dæ) åÁvč ǧ { ^} då
	GÈÈ P[覕Áį,-Áį]^¦æaāį}ÉÁş[¦\-[¦&^Áæ)åÁd'& & Áį[ç^{ ^}σ.A
	ĊĖ Ùã Ázzajān Altininininininininininininininininininin
	GÈ€` W¢äãã• Á
ΗÈ	Š^*ā/æāç^Á¦æ{ ^, [ ¦\ A
	HÈĖĖ Ô[}åãąį}•Ą́į ≁&[}•^}oÁ <del>diliiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiii</del>
	HÈG`Š^*ã  æąį̃} Å
	HÈH U]^¦æðaðað J¦[çæ‡• Áð 11 11 11 11 11 11 11 11 11 11 11 11 11
	HÈ Ò}çã[}{ ^}œ4Á[ && A
	HĚ Č}çã[}{ ^}œ¢Áã\Áœ•^••{ ^}oÅ
ΙÈ	Q] ^{ ^} cæð } Å
	IÈȨ Ò}çã[}{ ^}œ4Á(æ)æ*^{ ^}œÁ^•ơ{ Åå[&`{ ^}œaã(}
	IÈG` U¦*a}ãæaã{}Á&@edo <mark>Autoronononononononononononononononononono</mark>
	IÈH'Ü[ ^∙Áæ)åÁ^•][}•ãaậãa?•A <del>000000000000000000000000000000000000</del>
ĺÈ	Ò}çã[}{ ^}œ¢/dæğiği*Áegiaé^}^•●Á <del>000000000000000000000000000000000000</del>
	ÍÈF. Qå &að }Á <del>ddi di </del>
	ÍÈG V[[ à[¢Áæ‡)•Á <del>ð HÍIHHHHHHHHHHHHHHHHHHHHHHHHHHHHHHHHHHH</del>
	ĺÈH′ Væ'*^c^åÁ*}çã[}{ ^}cæÁtæãjãj*Á <del>àDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDD</del>
	ÍÈĖ V¦æ‡jāj*Á∧&{¦å∙Æ <del>bbbbbbbbbbbbbbbbbbbbbbbbbbbbbbbbbbbb</del>
ÎÈ	Q&ãå^}♂Áæ)åÁ{^¦*^}&ã•A <del>ùIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII</del>
	ÎÈF` Ò{ ^¦*^}&´Æ{ }: œ&œ Â <del>□□□□□□□□□□□□□□□□□□□□□□□□□□□□□□□□□□□□</del>
	ÊÊG` Q&&ãA^}ơ\$şç^∙cãtæaậ}Â <del>ù!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!!</del>
	ÎÈH Þ[cãæcea]}A
ΪÈ	
	ΪĖĖ - Φ, Φ, ¦} æļ&[{ { `}} æææ‡] Å
	ΪĖG Öā^&oÁ&[{{``}āc Á&]}•` œeāį}Å
	ΪĖΗ΄ CE^}&`Á&[{{`}}ä8ææa‡}Å
	ΪÈĖ Ô[{{`}ãc Á&[}•ĭ œæãç∧Á&[{ ãcc^∧Â <u>dthhhhhhhhhhhhhhhhhhhhhhhhhhhhhhhhhhhh</u>

Á

- Á
- ΪĚ́  $\hat{O}[\ ] | z_{2} \circ A$
- ΪĒĖ́
- ÌÈ  $\mathbb{Q} \cdot [ ^{8} (\mathbf{\hat{a}} + \mathbf{\hat{b}}) + \mathbf{\hat{b}} (\mathbf{\hat{b}} + \mathbf{\hat{b}}) + \mathbf{\hat{b}} (\mathbf{\hat{b} + \mathbf{\hat{b}}) + \mathbf{\hat{b}} (\mathbf{\hat{b} + \mathbf{\hat{b}}) +$ ÌÈ
  - Ò}çã[ } { ^} æþ\se åãa] \* A ÌÈG
  - ÌÈH
  - ÌÈʻ
  - Ü^][¦cā]\*Áə)åÁ^&|¦å•A
- JÈ JÈ
  - JÈG ÒT ÙÁ] åæ∿• Æ

## HUV<sup>°</sup>Y<sup>°</sup>]bXYI

Á

```
ÒT ÙÁ àb 8cã: ^• Ásì å Ásət * ^ @ Áthinin hinnin hinni hinnin hinnin hinnin hinnin hinnin hinnin hinnin hinnin hin
Væà|^ÁFËF`
                                                               U] \wedge |aea{\bar{a}}| \hat{A}a{\bar{a}} \wedge \hat{E}_{A} \wedge |\hat{a}| a^{*} \hat{A}a{\bar{a}} a^{*} \hat{A}a{\bar{a}} a^{*} \hat{A}{\bar{a}} |c \wedge \{ \land \} \circ \hat{A} = \hat{E}_{A} \wedge |a| \hat{A}a{\bar{a}} a^{*} \hat{A}a{\bar{a}} a^{*}
Væà|^ÁGËF`
Væà|^ÁHËF`
                                                               Š^*ã|æã]}Á`{{ @^Â
Væà|^ÁHËG
                                                              Üã\Á æiðe
Væà∣^ÁnHËH
                                                               Üã\Áœ•^••{ ^} o
Væà∣^ÁHËË
Væà∣^ÁLËF
                                                               Væà∣^ÁlËG
                                                               Væà∣^ÂÎËF
                                                              Ò{^¦*^}&^Á&j}@26606å^@2#
                                                             Ò} çã[ } { ^} æŧÁ^] [ ¦æ] * Á
Væà∣^ÂİËF
Á
```

### : ][ i fY ]bXYI "

```
Øðir í hÁr 🗄 í 🖞 Úær Ár Bæðar í Ásað á Ásað í í í dað skull sku
\mathcal{Q}
Ø∄č¦^Á,ËG`Ò}çã[}{ ^}œ4Á, æ}æ*^{ ^}œÁdč&č¦^Á
```

Á Á

# 5ddYbX]Wg

- OŢ[]^}åãcÁOŹÁ.ÁÒ}çã[[}{ ^}cæ†Á,[|ã&`Á
- CE[]^}åãcÁÓÁÁÚãc^Áj|æ}∙Á
- OĘ]]^}åã¢ÁÔÁÁOE^^}&°Á&[¦¦^∙][}å^}&^Á
- OĘ]]^}åã¢ÁÖÁÁÜ[ã¼Áæ)åÁ,æe∧¦Á(æ)æ≛^{ ^}ơ∮,|æ)Á
- OĘ]]^}åã¢ÁÒÁ.ÁŠæ);å∙&æ];^Á(æ);æ\*^{ ^}o∱,|æ);Á
- OŢ]]^}åã¢ÁØÁ.Á⊅[ãi^A,íaa)æ\*\*^{ ^}o∱,∣aa)Á
- OĘ]]^}åã¢ÁÕÁÁOBãÁ ǎæjãĉÁ(æ);æ≛^{ ^}ơÁ,|æ);Á
- OŢ]]^}åã¢Á₽ÁÁ/¦æ→ã&Á(æ)æ\*^{ ^}ơ∱|æ)Á
- OE[]^}åãcÁQÁÁOEā[¦ãtājæ¢Á&čic覿¢Á@?¦ãæet^Á(æ)æt^{^}o∱,|æ)Á
- OEJ]^}åã¢ÁRÁ.ÁYærc∿Á(æ)æ\*^{ ^}oÁj|æ)Á
- C0[]^}åã¢ÁSÁÁÚ[||˘qã[}Á§J&ãå^}ơÁ^•][}•^Á(aa)æ≛^{ ^}ơÁ|aa)Á
- OE[]^}åã¢ÁŠÁÁÔ}çã[]{ ^}œ4Á([}ãt[¦ā)\*Á,¦[\*¦æ4(Á
- Á
- Á

## % HÁ = bhfcXi VMjcb

P[|&ā[ÁÇCE•dæ)aāebÁÚc´ÁŠca:ÁÇP[|&ā[DÁ&`¦¦^}d^Á;]^¦æe>•ÁÖ`}|[^ÁÜæ)åÁÛ`æ¦^Áç@A´āc^DÉA;@B&@ÁsA´ |[&æe\*åÁæeÁÚ[@•çā]|^ET[[àæ)|ÁÜ[æåÉæe}]¦[¢ā[æe\*|^Ás@^^Áā[{ ^d^•ÁÇ{ DÁ[`c@A;AÚ[@•çā]|^Á Ç2āī`¦^ÁFË=DĚZ/@:Áãc^Á&[{ ]¦ã^•ÁœÂ{ [[];ā]\*Á[@•KÁA

- ●Á Š[o•ÁFÁæ)åÁGÁ§IÁÖÚÁÌÌ€FJJÁÁ
- ●Á Š[ơÁGÁÖÚÁiÌÍÌJÍÁÁ
- •Á Š[ơÆÄÖÚÄÌÌ€G€€ÆÄ
- ●Á Š[o=Á IÊÂ FÊÆFÎ GÊÆFÌ GÆÐ) å ÁFFÌ HÁÖÚÁ ÍÍÍ GFÁÁ
- ●Á Š[ơÆÄÖÚÁG€ÌGIJÁ

Ú¦[b/&cAOE]] ¦[çæ4ÁEδ €€H€Á, æ3 Á[`\* @oÁe) å Áţ à cæaj ^ å Áş ÁG€EÌ Áţ ¦Ác@ Á&[}•d` &caţ } Áe) å Áţ ] ^ ¦æaţi } Á [-Ás@ ÁÖ` } |[^ÂÜæ) å ÂÛ` æ}¦^ÁeeAk@ Á ãt^Á } å ^ ¦ÁÚæi Ó HOEA, Ác@ Á Environmental Planning and Assessment Act 1979ÁÇÒÚBOEACBACDĚÁUæi Ó HOEA, Ác@ ÁÒÚBOEACBAÓœe Á āj & ^Áa ^ } Án] ^ æ† å ÉÉQ \_ ^ ç^ ¦Á Ù&@ å` |^ OEEÔ |æ\*•^Á+Áţ -Ác@ ÁÒÚBOEACBAÓ, ½ [çãa ^ •Á[¦Ác@ Á&[}cāj` ^ å Á • ^Áţ -ÂÙ ^ &caţ } Á í Y Áţ Á { [åã° ÁeÁ, ½ [b & cóaġ]] ½ çæ4Á ¦æ) c\* å Á } å ^ ¦ÁUæi o HOEA, Ác@ ÁOBcAÓÇ } [, } Áee Á/¦æ) •ãaţ } æÁÚæi o HOEÁ Ú |[b & co DEÁ

V@ÁÚ¦[b\&cÁOE]]¦[çæ‡Á, æ=Áæ&&[{]æ]ðªåÁa^Áæ)ÁO}çã[]}{^}cæ‡ÁOE●•^●●{^}ó∱ÇOEDÁÇÚŠOEÞQVÁ Ô[}●`|cā]\*ÊÃOEEEIDEÁV@ÁÚ¦[b\&cÁOE]]¦[çæ‡Áæ‡|[, ^åÁ{[¦Á●^A[-Á©@Á\*ãe^Á[¦Ás@Á×¢dæ&cā]}Á[-Á\*æ]åÁa\*^Á { ^æ]●Á[-Áæ](@Á\*`&cā]}Áa¦^å\*^ÁÇ\_^ÓA\*¢dæ&cā]}DÁæ)åÁa'/Á\*¢dæ&cā]}ÁQ3]ãããæ‡Á[ç^¦à`¦å^}Áæ)åÁa¦ã&ã •A |[æ[Á\*¢dæ&cā]}DÉÁUæ)åÁ\*¢dæ&cā]}ÁæóA©Á\*ãe^Á&æ}Áa^Á&æ‡!ðªåÁ[\*ŏÁ\*}cā]ÁFÁRæ]`æ†ÁOEHÍÉÅ\_ão@ÁæÁ { æçã] `{Á\*¢dæ&cā]}AÁæe^Á[-Á+EEÉEEEÁ[}}^●Á]^\*●Á]^¦Áæ}}`{EÁ

Q ÁQE€J Ébeok[[åãaBæena]}Át[Áo@ Á×¢ãrca]\*ÁÚ¦[b×8oA0E]]¦[çæakÁy€ඞ´€€H€ÁT U ÖÁFDÁ, æráka]]¦[ç^åÁba^Ás@ Á Ö^]ædq(^}oA[-ÁÚ|æa}}ā]\*Áea}åÁÖ}çãl[}{{^}oAçÖÚÒDÁCJ[\_, ÁÖ^]ædq(^}oA[-ÁU|æa}}ā]\*É4Q å\*•d^Áea}åÁ Ò}çãl[}{^}oÁçÖÚOB ÒDÁ}å^¦Á^8ca]}ÁÍÍYÁ[-Ás@ ÁÒÚBOEADBOÉADBOÉAV@Át[[åãaBæena]}Á+[`\*@AAt[Á&L]•d`&AA æ4t(æ&@a]^¦^Á@ åÁdt[ÁQ2`•^Át[[àā]^Á]|æa)óA``ā]{^}oÁ,ão@a)Ác@ Át`æa¦^Áb[`}åæa^ÈÁ

U}ÁFFÁR' |^ÁGEFÏÉBaay ÁÒCEÁ^][¦cÁ, æ∘Á`à{ ãoc°åÁţ ÁÖÚÒÁţÕPÖÉBGEFÏDÁţ Á`]][¦cÁasÁù^&cāţ}ÂíÍYÁ æ]]|83ææāţ}ÁţÁ[Á;[åã°ÁÙ&@å`|^ÁGÉAÔ[}åãaāţ}ÂÁţÁú'|[b^&cÁCEJ]¦[çæ)ÁÊ´<del>CEHE</del>ÁţÁ3;&¦^æ^Á {æçãţ `{Á@æç^Áç^@38\^Á;[ç^{ ^}@ ÁæsÁ©Áãc^Á;[{ ÂÁţÁQ`¦ÈÁ'@å Á; æÁs}]'[ç°åÁş { açcãţ `{Á@eç`Áç^@38\^Á;[ç^{ ^}@ ÁæsÁ©Á; ác^Á;[{ ÂÁţÁQ`¦ÈÁ'@å Á; æÁs}]'[ç°åÁş { accãţ `{ acc3;}á3;\*|`ÈÁ

V@:Á;ãc^Áadp•[Á;]^¦aac^•Á;}å^¦Áa); ÁÖ}çã[]{ ^}cadÁÚ¦[c^&cã;]}ÁŠã&^}&^ÁÇĎÚŠDÆT€HIÏĖÁ

 $\begin{array}{l} & \bigvee \left\{ A \right\} \left\{$ 

V@ða ÁÐT ÙÁ@æe Á&^^} Á,¦^]æ hå, ão@Á∧-∧¦^} &^ Át[Ás@∘Á{[||[¸ã] \* Áta[& `{ ^} cæeā[} } KÁ

•Á Ò}çã[}{ ^}cæ¢ÁŒe•^••{ ^}c£šš[ œ/ÁFÁæ) å/ÁGÁ§ ÁÖÚÏÌ€FJJÁæ) å/ÁŠ[c/AFÁ§, ÁÖÚÏÍÍGFÉÄÖ`} |[ ^Á Úæ\É4ÇÚŠŒEQ(ÁÔ[}•`|cā]\*ÉÂÚ^] c^{ à^¦ÁGE€ÏEDĂ

Á

- •Á Ö`} |[^ÂÜæ) å•ÂÛ`æ¦^ ÁT [åãa38ææā] } Ág ÁÚ¦[b^8cóAQ]] ¦[çæ¢ÆÎ ´€€H€ÂQ} çã[] { ^} œ¢Á QE•^••{ ^} óÁQÕ PÖÉAQ€FÌ DĚÁ
- •Á Ö^ç^|[]{ ^}ơÂÔ[}•^}ơÂ₽[ÈĚÊ´€€H€ĖĂ
- Á ÒÚŠÁFHEÏ Ï ÈÁ
- •Á Š[&æ‡Áæ)åÁrœæ^Á\*[ç^¦}{^}œk±\*^}&ã^•Á`à{ã•ã[}•ĚÁ
- •Á QÙUFI€€FKG€FÍÁÁÔ}çã[}{^}œ‡ÁTæ}æ\*^{^}oÂÛ^•c^{•ÈÁ
- •Á Ö^] ælq(^}of(-ÁQ, +æ, d`&c` |^ÉÚ|æ)} ā) \* Áæ) å ÁÞæč | æ4Ü/•[` | &^• ÁÇÖQÚÞÜDÁÇƏ€E DÆÕ`ãå^|ā) ^• Á -{ | Ás@ Á, !^] ælæaāi } Á(-ÁÔ) çã[ } { ^} œ4ÁT æ) æ\* ^{ ^} ÁÚ|æ) • ÉÁ

#### %% 9A G<sup>•</sup>di fdcgY<sup>•</sup>

V@ ÁÒT ÙÁ; čdj ^• Á@ Á}çã[}{ ^} œdÁ, æj æ\* ^{ ^} ơ∫, læ&a&x • Áæj å Á, l[ & ^å ' /• Áţ Áa ^Át [ ||[ , ^å Á å ' laj \* Áv@ Át] ^ lææjt] Át -Áv@ Á `æ l ^ ÉÁ /@ Át lat æ^ Át ` l] [ • ^ Át -Áv@ ÁOT ÙÆs Ág Ás[ & `{ ^} oÁv@ Á ^} çã[]{ ^} œdÁ ^ ccaj \* ÉÁ `æ l ^ Át] ^ lææjt] • ÉÁ } çã[]{ ^} cædÁã \ • ÉA } çã[]{ ^} cædÁ æ \ ^ OT ÙÆs Ág Ás [] { ^} cædÁ ~ ccaj \* ÉÁ `æ l ^ Át] ^ lææjt] • ÉA } çã[] { ^} cædÁã \ • ÉA } çã[] { ^} cædÁ æ \ ^ OT ÙÆs Ág Ás [] { ^} cædÁ ~ ccaj \* ÉÁ `æ l ^ Át] | lææjt] • ÉA } çã[] { ^} cædÁã \ • ÉA } çã[] { ^} cædÁ [] | ^{ ^} cædÁ [] ag \* Áæj å Á^] [ lcaj \* ÉA /@s Ás Ág Át Á\* ãa ^ Áv@ Á `æ l ^ e Á\*} çã[] { ^} cædÁ [] ^ l -{ ^} cædá ] ÉÁ [] ag \* Áæj å Á^] [ lcaj \* ÉA /@s Ás Át Á\* ča ^ Av@ Á `æ l ^ e Á\*} çã[] { ^} cædÁ [] ^ l -{ ^} cædá ] éA / \* ` læet | ^ Áæj å Á, [ la& Á^ ` ă ^{ ^} e Ás ÁscÁ ^ • c^ { æsãAát æ} } ^ l As à -æstátãsæ Ás[] caj `ædát ] l [ co { ^} cÁ

#### %%Å 9A G<sup>·</sup>cV<sup>·</sup>YWj Yg<sup>·</sup>

V@^Á^^Á^}çã[}{ ^}cæaþÁ^\-{-{|{ æ} &^Á,`àb^&cãç^•Á{ ¦Ás@ã;ÁÔT ÙÁse/kÁ

- •Á Ô[{]|ãæ) &^Á, ãc@Á^|^çæ) oÁ} çã[] { ^} cæ)Á^\*ã |æaã, } Áæ) åÁ^\* ĭ |æaã, } ÈÁ
- •Á Ô[{]|ãa) &^Á, ão@k@ /&[}åãã] •Á, 4Ö^ç^|[]{^} o^Ô[} •^} o/Ê BÉEEH€/a) åÂÒÚŠÈÁ
- •Á Tājā[ārāj\*Á,[||ǐcāj}ÊĂ, æic^Á\*^}^¦æaāj}ÊŹaa)åÁ\*}çã[]{ ^}caa|Áaj]æ&orÈĂ

S^^Á\*}çã[}{ ^}œ4¼,àb\*&cãç^•Áæ)åÁæe\*^o•Á{¦Ás@·Á \* æ¦^Áæ4^Á@{,}}Á§,Á/æè|^Ár笹ÉÅ,@&&@éæ+•[Á ∄åã&æe\*•Á,@~¦^Ás@·Á^´Á\*}çã[]{ ^}œ4¼,àb\*&cãç^•Áæ4Åæåå¦^••^åŧ,Ás@·ÁÒTÙĚÁ

Á

Á



#### HUV`Y`%% 9A G`cV^YVMjj Yg`UbX`HUf[Yhg`

O≣]^&oÁ	Uàb⁄scãç^Á	Væ*^cÁ	Ù^&cãį}Á
Ù[āµÁæ)åÁ Yæe∿¦Á	<ul> <li>V[Á} • `  ^Ás@eedái ] æ∨ Ái } Ái [ājÁæ) å Á; æz^\ Á `æjác` å` lāj * Ái ] ^ lææai } • Áæd^Ái [ājāi ãi ^å ^åÁæ) å Á; ãs@aj Ás@ • &amp;[]^Ái^¦{ ãsc^å /åi^Ás@ /åi^ç^ []{ ^} c/&amp;[} • ^} dÈ</li> </ul>	<ul> <li>Ò} • ` ¦^Á¥   Á&amp;[ { ]  ãab) &amp;^Á; ãu@kb@ Á^ ^çab) oÁ/*ã  aeaãç^Á^` čã^{ ^} o ab) å ÁÔ[ OÈ</li> <li>T ^^oÁÒ} çã[ } { ^} oÁÚ![ c^&amp;aãi } ÁŠã&amp;^} &amp;^ÁQÔÚŠDÁ; aea^! Á č adaôc åã &amp;@ab*^Á; adaat ^c^!•Á{ ! Áda Á;  ab} }^åÅåã &amp;@ab*^•È</li> <li>Ò} • č  ^Ákbaãajāj *Á{ } Á[āAbab) åÁ; aea^!Á{ ab} aet ^{ ^} bÈ</li> </ul>	OŢ[]^}åã¢ÁÖÁ
Šæ)å∙&æ}^Á	<ul> <li>Ú¦[çãà^Áç^*^œaaā] Àà ~~^\+Át Át@ Át Át@ Át ^\alpha &amp; At /li></ul>	<ul> <li>Ù&amp;\^^} Ás@ Ás^ç^ [] { ^} cÁ\[ { Á``  [`} åāj*Áj\[]^\cave È</li> <li>Ú\[çãa^Áscá ãuāj } adÁ, accaç^Á@acca ãuacÁ{[\Á\[ \arksi accavA[ \Á\[ \arksi accavA[ \Arksi acc</li></ul>	OŢ[]^}åã¢ÁÒÁ
Þ[ ã ^Á	<ul> <li>Câ^} cã- Á^} • ãtă; ^Á^&amp; A^â; i - Ási à A^} • `  ^ Ási ] ![] ! ãte: ^ ^ }; cã[] { ^} cahÁs[] d[] • Ási à A^i ![ &amp;^ à `  ^ • Ási ^ ã] ![] ! ãte: ^ ^ }; cã[] { ^} cahÁs[] d[] • Ási à A^i ![ &amp;^ à `  ^ • Ási ^ ã] a []  ^{{ ^}} c à Ási ` ] a * Á] ] ^! ateta ] ath Assa ãt; ãta • È</li> <li>T āj āj ã a ^ Á; [ c^} căthÁsta c^^! • ^ A [] a * Ási ] atso Åsi • È</li> <li>T āj āj ã a ^ Á; [ c^} căthÁsta c^^! • ^ A [] a * Ási ] atso Åsi • È</li> <li>T āj āj ã a ^ Á; [ c^] căthÁsta c^^! • ^ A [] a * Ási ] ath Assa a</li></ul>	<ul> <li>Ò}•`¦^Áč   Á&amp;[{] ãæ} &amp;^Á ão@Ác@Á^ ^çæ) oÁ^*ã  ææãç^Á^č`ã^{ ^&gt; o æ) å ÁÔ[OÈ</li> <li>Þ[Áv¢&amp;^^åæ) &amp;^Át -Ác@Á;]^!ææã; a¢Á,[ã*^Áã;ã*È</li> <li>Þ[Áv •cãæ}åÁ&amp;[{] ææ] o Á'[{ ásæå bæ&amp;^} oÁ^•ãå^} o Áÿ Á^ ææã; át }[ã*^Á*^}^!ææã; È</li> <li>Þ[Áč oÁ; -ÁQč`!•Á;[!\È</li> </ul>	OĘI]^}åã¢ÁØÁ
OBEAÚřæ¢ác Á	<ul> <li>Ò} • `  ^ Áq3]   []   ãỡæ ʿ Á§[ } d[  • Áq3 å Á,  [ &amp; ^å `  ^ • Áq4 ^ ã]</li> <li>ã[ ]  ^{ { ^} da Ås `   ã] * Ás@ Á, ] ^   æaāi } Á Ás@ Á ` æs ^ Ág</li> <li>ãq [ ãð Á,   Á, ã] ā] ã ^ Ás • of ^ } ^   æaāi } É Ås@ Á ` æsi ^ Åg</li> <li>ãq [ ãð Á,   Á, ā] ā] ã ^ Ás • of ^ } ^   æaāi } É Ås@ Á ` æsi ^ Åg</li> <li>ãg [ ãð Á,   Á, ā] ā] ã ^ Ás • of ^ } ^   æaāi } É Ås@ Á ` æsi ^ Åg</li> <li>ãg [ ãð Á,   Á, ā] ā] ã ^ Ás • of ^ } ^   æaāi } É Ås@ Á ` æsi ^ Åg</li> <li>ãg [ ãð Á,   A ^ &gt; ã ãi, [ c ^ ] æad / Åsi * of ^ / ^ / æaāi } E Ås * Ås * Åsi ^ Åg</li> <li>ãs • Ás * Ås * Ås * of ^ / ^ / asi ^ / * Ås * Åsi ^ / /li></ul>	<ul> <li>Tājāţā ñā ^ Áxaj å Áţ æj æt ^ Áş [ c^ } cãn plÁxná Á čæjáč Đǎ • chất ] æstor Át [ { Ás@ å^c^  [ ] { ^} chất Áxas &amp; [ å æj &amp; ^ Å ã @Á / ^cæj ch/* ã  æstaç ^ !^ č ã ^ { ^} o Áxaj å Á0 [ CE</li> <li>Ô[ } d[   Átá • o Áxaj å Át¢ @et • ch { ã • ã } • Áş - Áş  æj o Áxaj å Á č č ā } { a } {</li></ul>	OŢ[]^}åã¢ÁÕÁ

O≣]^&cÁ	Uàb⁄&aãç^Á	Væ*^Á	Ù^&cãį}Á
V¦æ <b>-æ</b> Á	<ul> <li>Ò}•`¦^Áæ]] []¦ãæe*Á&amp;[}d[ •Áæ)åÁ,![&amp;^å`¦^•Áæ}^</li> <li>ãį] ^{ ^} c*åÁ§,Á;!å^!Áţ</li> <li>Áţ</li> <li>ã*^Áx@:Áξ</li> <li>ā&amp;o*Áξ</li> <li>Áξ</li> <li>ã*^Áx@:Áξ</li> <li>a&amp;o*Áξ</li> <li>Áξ</li> <li>á* A</li> <li>A</li> <li< td=""><td><ul> <li>Ò}•`¦^Á¥   Á&amp;[{] ãæ} &amp;^Á ã@Á@ Á^ ^çæ) ơÁ^*ã  æãç^Á^``ã^{ ^} o æ} å ÁÔ[CE</li> <li>Þ[ÁŠ•oãæ*å Å&amp;[{] æðj o Á^ æe^å Á&amp;[Áã*Ásæ-3&amp;</li> <li>Þ[Á[æå Ååæ{æ* ^Á+[{ Á `æ+ ^Áç^@3&amp; ^Á[[ç^{ ^} o Áa^^[} å }[¦{æ\$ Å\$ aæ Áæ} å Á* æ;</li> </ul></td><td>OŢ]]^}åã¢Á₽Á</td></li<></ul>	<ul> <li>Ò}•`¦^Á¥   Á&amp;[{] ãæ} &amp;^Á ã@Á@ Á^ ^çæ) ơÁ^*ã  æãç^Á^``ã^{ ^} o æ} å ÁÔ[CE</li> <li>Þ[ÁŠ•oãæ*å Å&amp;[{] æðj o Á^ æe^å Á&amp;[Áã*Ásæ-3&amp;</li> <li>Þ[Á[æå Ååæ{æ* ^Á+[{ Á `æ+ ^Áç^@3&amp; ^Á[[ç^{ ^} o Áa^^[} å }[¦{æ\$ Å\$ aæ Áæ} å Á* æ;</li> </ul>	OŢ]]^}åã¢Á₽Á
P^¦ãæé^Á	<ul> <li>Ò} • ` ¦^Áæ] ]  [] ¦ãæe* Á&amp;[ } d[  •Áæ) å Á;  [&amp; ^å*  ^• Áæ)^</li> <li>ã[ ]  ^{ ^} c^å Á§ Á; lå^! Á[ Áæ;[ãå Åæ; æ* ^ Á; l</li> <li>åã č ¦àæ) &amp; ^ Á; -Á@; lãæ* ^ Áær { • È</li> </ul>	<ul> <li>Ò}•`¦^Áĭ   Á&amp;[{] ãæ) &amp;^Á ã@Á@Á^ ^çæ) ơÁ^*ã  æãç^Á^`čã^{ ^} œ æ) åÁÔ[CE</li> <li>Þ[Ásaæ{æ*^Á([Á@];ãæ**^Ásc^{ •</li> <li>OE[ Ásãe^Acœ-Áse) åÁ&amp;[}cæ&amp;q[ !•Áslæã] ^åÁ[}Á`}^¢] ^&amp;c^åÁã]å• ] ![q[&amp;[]</li> </ul>	OE[]^}åã¢ÁQÁ
Yærc^Á	<ul> <li>Ò}•`¦^Áæ]] []¦ãæe^Á&amp;[}d[ •Áæ)åÁ,![&amp;^å`¦^•Áæ}^</li> <li>ã[] ^{ ^}c^åÁ§,Á[¦å^!Á§,Á[á]ã[ã ^^Áx@Á§[]æ&amp;o•Á§[Áœ</li> <li>[[&amp;æ¢Á*}çã[]{ ^}oÁæ)åÁ&amp;[{ { `} ãc Á![{ Á æ c^È</li> </ul>	<ul> <li>Ò} • ` !^Á¥   Á&amp;[ { ]  ãæ) &amp;^Á, ãu@Á@ Á^ ^çæ) oÁ^*ã  æãç^Á^` ă^{ ^} o æ) å ÁÔ[ OĒ</li> <li>Y æ c'Á*^} ^ !æãi] &gt; Á; ∄ ãi ã ^ å Áû@[ ` * @Á@ Á@ I æ &amp;@ Á, -Á, æ c' { æ) æ * ^{ ^} o∱, !ãi !ãã • È</li> <li>O∏[Á, æ c'Ási Á; æ) æ * ^ å Ási Áæ&amp;&amp;[ ! åæ) &amp;^Á, ãu@Á@ ÁY æ c' Ô æ • ãã&amp;æãi] &gt; ÁÕ *ãa^ ã] • • ÁçDÚOEÂGEFI D</li> </ul>	OŢ[]^}åã¢ÁRÁ

# &" Dfc/YVMXYgVVJdhjcb

#### & '% ? Ymg]hY ]bZcfa Uh]cb

Ø^æč¦^Á	Ö^œ∰•Á
OE[]¦[ç^åÁ∿¢dæ&caa[}}Áæer^Á	H <del>⊂⊂΀⊂⊂</del> Á[}}^•Á2Áa;}`{ Á
Ò•cāį æe^åÁæçæajæà ^Á^•[č¦&^Á	Î <del>ÎECCÎECC</del> Á <sup>H</sup> Á
Ò∙cãį æe∿å/ų́]^¦æeāj*/ų́^¦āįåÁ	GÎÁ^æl•ÁÁ}œl‡ÁFÁRæ)ĭæl^ÁG€HÍÁ
Ò¢clæ&cāį}Áå∧]c@Á	FGÁ(Áa^ [,Á)æciˈaaþÁt¦[`}åÁ(^ç^ Á
Ùãt^Áæh^æÁ	ÎGJÊİÁ@^&æak^•Á
Ûĭæ¦^Áæl^æÁ	ÍÎËİÁ@∿&cæa⇔^∙AÁ
ŠÕŒÁ	V, ^^åÁŠ[&æ¢#Ő[ç^¦}{ ^}♂ÁŒ!^æÁ

#### &"&Á Dfc/YVWicj Yfj ]Yk <sup>·</sup>

P[|&ā[ Á[]^¦æơ∧ÃÖ`}|[^ÁÜæ)å•ÁÛ`æ¦^ÁæơŠ[ ÁFRŐÚG€È GIJLÁŠ[ ÁFÌGÁÖÚÏÍÍÍÏ GFLÁŠ[ ÁFÌHÁ ÖÚÏÍÍÍÏ GFLÁŠ[ ÁII ÁÖÚÏÍÍÍÏ GFLÁŠ[ ÁIFAÖÚÏÍÍÍÏ GFLÁŠ[ ÁFÎGŐÚÏÍÍÍÏ GFLÁŠ[ ÁGÁÖÚÄÌÌ€FJJLÁŠ[ ÁFÁ ÖÚÏÌ€FJJLÁŠ[ ÁFRŐÚÏÌ€G€€LÁŠ[ ÁGAÖÚÏÌÍÍÌJÍÉÁÚ[ œrçā∦^ET[[àæ)AŰ[ æåÉÁÚ[ œrçā∦^ÈÁ

 $V @ \dot{A}^* ad |^{\dot{A}} |[ a^* &^{\bullet} \dot{A} ad \dot{A} d \dot{A}$ 

#### &" Á G]HY VX UFUVHY f]gh]Vg'

V@Á \* æl ^ Áæí [ \* of&[ { ] | ã ^ • Áç [ Á ¢ d æ&dā] } Á,[ } å • Á&[ } • ã dā \* Á ÁGÍ Á@ &ææ ^ • Áæj å ÁrFEİ Á@ &ææ ^ • Á |^•] ^ &dãç^| ÈÈÙæ t ^ Ár Ás Ás \* | ! ^ } d ^ Á,] ^ ! æaā] > Áæj å Ási Á[ &ææ \* å Áæj ] ! [ ¢ā[ ææ ^ | ^ Ár ÈFÁ { Á,[ ¦ coáj, ~Á∞ Á • [ \* co2 ¦ } Ás[ \* ] å æð ÊÊ, @ð • oÆ æ \* ÁCÁ, āļ Ási ^ Á[ &ææ \* å Ási Ási@ Á [ \* coák æ oÆs] ] ! [ ¢ā[ ææ / ° Ár €€Æ Á ~ [ { Ás@ Á [ \* co2 ¦ } Ás[ \* ] å æð ÈÉÓ[ coákæ ^ æ Áæð ^ Á[ &ææ \* å Ásiæ åæå ææ \* å Ásiæ åææ oÆs[ \ } oÆ [ \* ] \* å æð îÊ, ãæ \* Å ~ [ { Ás@ Á [ \* co2 ¦ } Ási[ \* ] å æð ÈÉÓ[ coákæ ^ æ Áse ^ Á æð Å ásiæ å ææ \* å Ásiæ åææ å Åsiæ å ææ \* oÆ Å T [ ] à æ∮ÁÔ! ^ ^ \ Ásæ Æ ã æð & • Áæj \* ā] \* Á' [ { Ár €€Æ Á Æ Æ

OB&&^●●ÁţÁv@^Áãr^ÁarÁ¦áĮ æsiậîÁj¦[çãa^åÁv¦[{ÁÚ[corçã||^ÁT[[àæa|ÁÜ[ænàÈĂv@~Áæ&&A•●A,ãr@3yÁx@A ]¦[]^¦c`Á@æe/Áa^^}ÁA\*æap^åÈĂOEAá^åã&æer\*åÆj\*¦^●●Áæe}åÁ\*\*¦^●●Áæ&ãjãc´ÁārÁj¦[çãa^åAţ}ÁÚ[corçã||^Á T[[àæa|ÁÜ[ænàÈĂ

Ó`}åāj\*Ájāļ/Ás∧Áj¦[çãā^åÁÇājÁ,ædDÁk[Ási[c@Áv¢dæ&cā[}Á,[]å•ÁsejåÁt,]^¦æaā[}æk/Áseh⁄ærÁk[Áseh/(æ¢ā[č{Á @?ā\*@A∱-ÁrH∈∈Á({Á;¦ÁGBEA′(ÁCBPÖÁ[Áse-Ás[Áā[ãoÁ][[å/ās[]æ&orEÁ

OZáç^\*^œec^å/áçãa čæçÁsič~~^¦ÆsiÁ[&æec^å/áed[č}å/ási[c@Ár¢dæ&cāt]}Áj[}å•ÈÁ/@/Ásič~~^¦Á[}^ÆsiÁ &[{]|ãt\_^}c^åÁjãc@Á/Ëç^\*^œec^å/áæ}å•&æ}^å/áset^æe/Áset[}\*/ás@/Áræec^¦}Ási[č}åæt^Át\_Áset|Á[ce/Át[Á T[[àæ‡lÁÔ¦^^\ÁÇFÍÁ@eetDÉA

#### &" '% FYgci fVY XYgV/jdh]cb

#### Concrete sand

V@AÖ`}|[^AÛ`æ¦^Áræ)åÉasec^¦Á;æe@aj\*ÉasrÁ`ãaæa|^ÁserÁseAs[}&¦^c^Áræ)åÁseååãaãç^ÉA;@a&@ásrÁs@Ár@Á {æ4p;¦Á•^Á;∞Áræ)åÉSS[;Ár¢dæ3cdī}}A&[•orÁ;æ}^Ás@Áræ)åA&[{]^cãaãç^A5jÁs@Ár>[¦c@;¦}Ár>,Á Ù[`c@ÁYæ}^•ÁsejåÁÚ[`c@ÁOæecAÛ`^^}•|æ}åÁ{æ}\^orÉA

#### Loam

OB;ÁB;ç^•cā\*æaā;}ÁB;d[Á[æ;(Á^•[č¦&^•Á;}Ás@Á\*ãz^Á;æ•Á&æ;là\*åA;čó4B;Á;ãa:EƏƏƏÏÁÇÕ[~^^Á Õ^[•&a\*}&^•ÉAƏƏƏEÜDÉA;ão@Áaa;]¦[¢ã;æz^\^ÁJ€EE€EÁ;}}^•A;A\*\*ãaæà|^A;[æ;(Ása^^)cãa\*àÈEÁ

#### Fill material

Tæc^¦ãæþÁ∿¢&æçææ^åÁ√[{ ÁÖ`}|[^ÁÛæ)åÁÛ`æ¦^ÁãrÁ憕[Á•^åÁæe Á¼[₀, Á\*¦æå^+Áã‡|Á; æe^¦ãæþÁá

#### Plastering and rendering sand

V@ Á æ) å Á ¢ dæ& c^ å Á ¦[{ Á @ ÁÖ`}|[^ Á Ùæ) å Á Û `æ ¦^Ê Á, @ } Á, æ•@ å É Én Á `ãæ æ) |^ Á [¦Á [, ^¦Á ¦æ å^Á ]|æ• c^¦ā] \* Áæ) å Á^} å ^¦ā] \* Á æ) å È Á

#### Other uses

U c@ ¦Áho]^&ãædaã c+Á, ¦[å`&o•Á, @3&@Áãá&k|[•^|^Át[Ás@ Á';|æåā]\*Át, Ás@ ÁÖ`} |[^ÁÛ`æ¦^Á;æ}å•Á5, &|`å^kÁ

- •Á Õ[|~Á&[覕^Á;æ}å•ÈÁ
- ●Á Õ¦[˘ơÁ;æ);å●ĚÁ
- ●Á Øãj^Áãjc^¦Áræ);å●ÈÁ

#### &"(Á 91 hfUVMjcb a Yh\cXc`c[m

Ò¢dæ&cāį}ÁšarÁsej]¦[ç^åÁt[Ása^Á'}å^\cæsta^}ÁšjÁc;[ÁÇCDÁrcæsta•ÈŹV@Á;[\c@\}Á\¢dæ&cāį}Áset^æÁqQÙcæstaÁ FD≸arÁ&č\\^}d^Á;]^\æcāį}æ¢EŹ;ãc@AÜcæstaÁGAsa^āj\*Ás@AA[čc@\}Á\¢dæ&cāj}Áset^æásej]¦[ç^åÁt[lÁčč\^Á ^¢dæ&cāj}ĚÁ

V[]Á[āļÁārÁdā]]^åÁsaæ&\Ás@[`\*@ás¦^Ár¢&æçææā]}Áse)åÁ[|å/AşiÁ±`àÁcæ\*^•o(AţiÁr}•`¦^Ár¦[•ā]}Áse)åÁ å`•oÁ\*^}^!ææā]}ÆsiÁ^]oAţiÁsa4(ājā]`{ÈAÙ[{^A;ç^!à`!å^}Ase}ÁsēA`ca‡ā^åAţi}ÁárA`cæ\*Aç?ÈÈAşiÁs@A´ ]!^]æ}ææāj}ÁjÁ^^@æàājānææāj}Áse?~æDÉQ\_^^ç^!Ê4(`&@ájÁ-ás@árÁ[āj/AsiÁ`ãnæà|^Áţi¦Á •^Áse-Ási`ājå^!•Áţ!Á à!ā&\ā\*•Á[æ;EÁ

OB;ÁB;ãnãadaÁ± čàÁ cæt ^on[(-Á, ^o%s¦^å\* ^åÁ\*¢&æçæaaā]; Áñs Ác@}Á'; å^¦cæs ^}ÊÁ;@38&@4ás Aš\*}å^åA'[{Ác@A |æt\*^Ás[å^Á, Ác@ Ár¢dæa3caā]; Ásd-∞abÈV@3a ÁB;ãnãada/ásd-∞aAf, Ár¢&æçæaā]; Áñs Ácāpā ^åÁse ÁsdAj, ^•Á^č ¦}Á,[}åÈÁ U}&^Ás@ Ápã^•]æ)Á[, -Ár¢dæa3caā]; ÁÙcæt ^ÁFÁ@ee Ár¢]ā\*^åÊasdAjä[&\*••Á, ä∥Af,&&č;¦Á{¦Ár¢dæa3caā]}Á Ùcæt ^ÁGÈÁÚ[æ)•Á[, -Ás@ Ásd]]¦[ç^åÁrcæt ^•Á; -Ás@ ÁÖč}}∥[^ÁDæ)åÁÛ`æs¦^Ásd-Ajk][çaã^åAjā ÁOE]]^}åãaAÓÈÁ

#### &') Á DfcWgg]b[ `

Ú¦[&^••ā]\*ÁārÁ}å^¦œah^}Át[Á^]æ}æe^Á;æ}očåÁæ}åÁ^•[`¦&^Áæ}åÁā}åÁā]^•Ác@[`\*@Á@妿i|&&Á •^]æ}ææat]}Á;ão@ajÁæ}Át;}•ãe^Á;æ•@Á;|æ}dĚAUæh^æà|^Át;æe\*¦ãæ‡+Át;[{Ác@ãrÁj;|[&^••Áæt^Árd[&\]ā/åÁæ}åÁ •[|åÁt[Ác@Át;æ\^dEĂ

W}, æ) c^åÁæ) åÁæ&aãa&A, æ c^Áaj^• Áæ^Á dæc^\*a&æ|^Á^à`¦aðaÅÇ^Äaj c^¦|^åDÁ, ãc@ajÁs@ Áaj^• Á^č¦}Á ][}åÁæeÁæA/ç^|Áa^|[, Ás@ Á, æc^¦Áæà|^Á{[Á|¦^ç^}c^}oA, ¢ãaæaā]}Á{, Ás@ Á&[}cæaj^åA, ^¦ãc^ÈÁ

Þ[Á&@{ 3&aaþÁ}¦[&^••ā]\*ÁārÁ}å^¦cæ\^}ÊÉæ¢c@[`\*@éæååãããç^•Át[Á; æāj cæājA,PÁ^ç^|•Á,ā|Á¦[{ Ácā[ ^Át[Á cā[ ^Áb^Á •^åÈĂ

#### &"\*Á <Ui `U[Y`fci hYg`UbX`dfcV\YXi fYg`

#### &"+Á D`Ubh'UbX'Yei ]da Ybh

Ú|æ); okæ); å Á\*``ā]; { ^}; oks@eenked^Á; ^&^••æ'`Á[; kk@^Á[;] ^;æeā]; } Á; -kk@^Á \*æ; '`Á\$j&|\* å^ká

- Ö¦^å\*^Á
- H€Á{[}}^Á\¢&æçæ[[¦Á
- Ùæ)åÂÛ^]ælæ[¦Á
- Ù&\^^} ā \* ÁÚ|æ) cÁ
- Øl[} dÔ}åÁŠ[æå^¦Á¢ÁGÁ
- GÍÁ/[}}^ÁÖ`{]Á/¦`&\Á
- U} Ë ã^ Áæ&ãããã Á ÁUæ-Áæ ^} ãã ÉA ~ã& Áæ&ãããã Áæ} å Áæ&ã

#### &", Á < ci fg`cZcdYfUhjcbžk cf\_ZcfVY`UbX`hfi VV`a cj Ya Ybhg`

P[`¦•ÉÁ,^¦āįå•Áæ)åÁæį ^•Á, Á,]^¦ææaj}Åæd^Áæe Á,`daj^åÆjÁ/ææi|^ÁŒËÆe)åÁæA^Áæ Á, ¦^•&¦aa^åÆiÅæA Ö^ç^|[]{ ^}ơÔ[}•^}oČ2Ú]^¦ææaj\*Áæaj\*ÁæajåÁœOAş[|`{ ^A, Á, æe^¦ãædA, ãœA, ÁœA^•[`¦&^A, ajA^^A c@Á^``ã^{ ^}oÁ[¦A]Át[ÁTÌÁ; ||Æati^A\*{]|[^^^•A, Å]`•Áæååãæti}≥æ4Æ{[}dæ&oÁ, æajc^}æa)&^A, i•[}}^|Á æA, æáÆs^Á^``ã^åÉU]^¦ææti}•Á, ajÆs^Æ{] |[^^^•A\*A] \* &c°åAT[}åæA&[ÂUæč¦åæÊEP[Át]^¦ææti}•Áæ^Át[Ás^Á `}å^¦œa^}A, ÁtjÂU`}åæA, ÁtjÁ; ajæA@{]ãaæê•ÈA

#### HUV`Y`&!% CdYfUhjcb`hja YgždYfjcXg`UbX`hfi VV`a cj Ya Ybhg`

Ÿ^æ∜^ÁJ]^¦æaāį}Á	Öæ̂∙Á Ú^¦Á Ƴ ^^\ Á	P[	Öæaậ^ÁU]^¦æaa}*Á/ãį^∙Á	V¦`&∖ÁT[ç^{^}orÁ Ú^¦ÁP[č¦Á
ÍGÁ,^^\●ÁæÁ^æłÁ Ção@Á©(Á¢&^]dāi}Á [-Ájčà]ãkÁ@(lãåæê●Á æ)åA[co@¦Á][c^}cãæ‡Á &[{]æ}^Á@(cå[,}Á ]^¦ājå●DÁ	ÍĚÁ	ÍÍÁ	T[}Á Á213ká K €€Áse; Át;Á K €€Á;{Á Ùækká K €€Áse; Át;Á T G K €€Á;{Á Þ[Á;]^¦ææð;} •Áse^A át;Á; & & `¦Å;}Á Ù`}åæ°•Á;¦Á,`à 33 Á@; ããæ°•Á Ç} ^••Æstæ Á;æð;c^}æ; & A & A & & A	GIÁ Þ[ơk[Á*¢&^^åÁ {[¦^Áo@ea)ÁGIÁ @raaç^Áç^@a& ^Á {[ç^{ ^} orÁgā)Á æ)åÁ[`004]^^¦Á@[`¦Á

#### &"- Á **G]HY ZJV[Y]h]Yg**

V@:Ááã^Á&覦^}q^Á&[}cæaậ,●Ác@:Á{[||[¸ã],\*KÁ

- Ò¢ã cậ \* Ás, ^||ậ \* Á@; `•^• ÈĂ
- Tænāj cænāj^å Ánāj å Ánāj cæn& cÁnāj ča ánāj čænā ánāj ča ánāj ča ánāj ča ánāj ča ánāj ča ánāj ča ánāj ča ánā
- OE ¦&& |c | æ|Á\$s|æ#j æ\* ^ Áj ^• Á` } } j \* Áj |^å[ { j æ) d` Á æ oÁ, ^• dĚA
- Qic^\} æ\$4\$e&&^••A\$læ&\•A{[\Aæ}{ A{ æ&@}^\^A{ [ç^{ ^} dĂ

- •Á Šæ\*^Áæ‡ ¦&&` |č ¦æ‡Á @ å•ÁçQ@•^Á @ å•Á, āļlÁà^Á dājā ^ åÁ{ ¦Ás@ Á;` ¦] [•^Á, Á d[ ¦æ\* ^Áæ) åÁ æ) &ājlæ^Á( æji c^}æ) &^Á, Á æ&@j ^\^Áæ) åÁ |æ) c⁄A``ā] { ^} c⁄æ••[ &ãæc^åÁ, ão@k@ Á`æ¦^ DĚÁ
- ●Á Ò¢dæ&dāj}Áæ)åÅ,¦[&^••ā]\*Áæ^æÈÁ
- A  $CE_{A}$   $ac_{A}$   $ac_{A}$   $bc_{A}$   $bc_{$

V@^Áæj]¦[ç^åÁíãc^Ájæĉ[čóÆiÁí@[,}Á§iÁOEj]^}åãcÁÓĚÁ

#### &'%\$Å I hj`]hjYg`

#### &'%\$'%'KUHYf'

Yæc^¦Áxe∙•[&ãæcc^åÁ,ão@Áx@A´`æs¦^Á,ā∥Ásà^Á@eseç∧∙c^åA(;}Á:ãc^ÈÁOE;Á(;}Ë∘ãc^Áæa)\Á&[}cæa3;ā)\*Á:d[¦æ≛^Á &æa]æ&ãc´Á(-Á<del>T⊂ÉEC€</del>ÁŠÁña:Á[&æac^åÁæábæ&c^}cAs[Ác@Áxe{^}ãã>ábä]\*ÉA

#### &'%\$''&' GYk U[ Y

Ù^,æ\*^Á\*^}^¦æe^åÁa^Áx@Á\*`æ¦^ÁarÁ(ājā[æ¢Ééee[&ãaee\*åÁ,ão@Ácæ-Á[&æe^åÁ(}}Áx@Áãe^ÉéV@Ácæ-Á æ{^}ãa?•ÁæjåÁ(~ã&^áka\*ā¦åā)\*Áæ}Áxe&&[{]æ)ð\*åÁa^ÁæjÁ(}Eēãe^A~~j\*^}o%aã][•æ‡Á^•c\*{ĚÁ

#### &'%\$''` 9`YV#f]V**y**hmi

Ò|^&cla&aĉÁasÁ`]]|a∿åÁ{[{Áx@^Áv¢ārcā}\*Á`]]|^Á{[{ÁYæd, a&\ÁÚæd\ÁÜ[æaåÉkçãaækkæaà[ç^\*¦[`}åÁ {æa})•ĚÁ

&'%\$''(`:iY``

Öa∿•^|ÆinÁq[¦^åA[;}ÁarÁbjÁazÁbjÁazÁv^|-Ábà`}å^åAiq[¦az\*^Ásaa)∖Ása)åAb[, •^¦ÈÁ

Á

Á

# " @Y[]g`Uhjj Y`ZfUa Yk cf\_

#### ''% 7 cbX]h]cbg cZVt/cbgYbh

V@AÖ^ç^|[]{^}ơÔ[}•^}ơ&[}åãaā[}•Á^|æaā]\*Áq[Ás@AÔTÙÁse¦^Á,`dā]^åAşlÁU&@å`|^Á.Á,As@Á Ö^ç^|[]{^}ơÔ[}•^}ơÁse]åA&[{]¦ãi^Ás@Á{[|[,ā]\*KÁ

1. The Proponent must prepare an Environmental Management Strategy for the project to the satisfaction of the Secretary. This strategy must:

(a) be submitted to the Secretary prior to starting quarrying operations on the site;

(b) be prepared in consultation with the relevant agencies;

(c) provide the strategic context for environmental management of the project;

(d) identify the statutory approvals that apply to the project;

(e) describe in general how the environmental performance of the project would be monitored and managed;

(f) describe the procedures that would be implemented to:

• keep the local community and relevant agencies informed about the construction, operation and environmental performance of the project;

- receive, handle, respond to, and record complaints;
- resolve any disputes that may arise during the life of the project;
- respond to any non-compliance;
- manage cumulative impacts; and
- respond to emergencies;

(g) describe the role, responsibility, authority, and accountability of the key personnel involved in the environmental management of the project; and

(h) include:

• reference to any strategies, plans and programs approved under the conditions of this consent; and

• a clear plan depicting all the monitoring to be carried out under the conditions of this consent.

The Proponent must implement the strategy as approved by the Secretary.

U c@ ¦Á&[} å ããā[} • ÊÅ^ |^çæ); cÁ{[Ás@ ÁÔT Ù ÊŠæ}^ Á,`` dā], ^å Áæ); å Áæå; å /•••^å Á§J Ás@ Á^ |^çæ); cÁ{[æ]; æ]; æ]; eá ] |æ); • Á§J &{``å^å Á§J Ás@ á ÁĎT Ù ÈÁ

#### ''&Á @Y[]g`Uh]cb'

#### HUV`Y'' !% @Y[]g`Uh]cb`gi a a Ufm

Š^*ã æaaį}Á	Ü^ ^çæ <del>)</del> &^Á	Ü^˘˘ã^{^}œÁ
Environmental Protection and Biodiversity Conservation Act 1999	V@ ÁÒÚÓÔÁŒko458a^} cãa?•ÁT ææc^!•Áţ.Á Þææti] }ædÁÖ}çã[] { ^} cædÁÚðt }ãaBæð &^Á QT ÞÒÙDÉA @8k@45A ãc^Át] ^!ææti] >•Á @æç^Ás@ Át[c*} cãædÁt fát] ]æ&cÁ][] }Á !^~ ˜ ã^-•Áæt]] ![çædÁt[[{ /sc@ ÁØ^å^!ædÁ Ò}çã[] { ^} cÁT ĝiã c*!EÁ	C # 2 [   a = b + a + 2 + a + 2 + a + 2 + a + 2 + a + 2 + a + 2 + a + 2 + a + a
Environmental Planning and Assessment Act 1979	CЩÁå^ç^ []{^}ơ与 Áş Á>ÙYÁsiÁ æ•^••^åÁşiÁs&&{¦åæ}&^Á,ão@ás@Á ]¦[çãaţi}•Á,Ás@ÁÒÚBOEX0&dĂ	$\begin{split} \tilde{U}_{ac}^{A}(A_{a}^{*}) & A_{ac}^{A}(A_{a}^{*}) & A_{ac}^{A}(A_{a}^{*}) & A_{ac}^{A}(A_{a}^{*}) & A_{a}^{A}(A_{a}^{*}) & A_{a}^{*}(A_{a}^{*}) & A_{a}^{*}(A$
Statement Environmental Planning Policies (SEPP)	ÙÒÚÚ∙Áæl^Á; æ)}ðj*Áðj•d`{^}orÁ å^ç^ []^åÁ[¦Á]^&ãã&Á; æ)}ðj*Á ã•`^•ÁðjÁ≂ÙYÉÁ	Ü^˘˘ã^{ ^}♂Ą Â, Á^ ^çæ)ơÁ ÙÒÚÚ∙Á, ^¦^Á&[}•ãa^¦^åÁ å˘¦āj*Áx@?ÁÒOĐĂ
Protection of the Environment Operations Act 1997	V@ÁÚUÒUÁO£8cÁ^**  æe^•Á,[  čqā}Á æ}åÁ, æ:c^Á(æ)æ*^{ ^}o45, ÁÞÙYÁ c@[**@\$a*i}á <sup>*</sup> á <sup>0</sup> }çā[}{ ^}œ4Á Ú¦[c^&cā]}ÁŠ382^}•^•ÁĊDÚŠDÁ{¦Á •&@å* ^åÁæ&cāçãa?•ÈÁ	Ö`}    ^ Á`ælî Á@ee Áæj Á æj ]   [ ç^å Á ¢dæ&cāj } Áæe^ Áj - Á HEETÊEEE Áj } } ^ • ÁzĂ ^ æb EĂ Ò¢dæ&cã; ^ Áj å` • dā • Á @a&@Á ^ ¢&^^å Á HETÊEEE Áj } } ^ • ÁzĂ ^ æb Á æ^ Ås^-aj ^ å Áæe Á &@ å`   ^ å Á æ&cã; ãzð • Áæj å Á^``ā^ Áæj Á ÒÚŠEĂ Ùãe^ Áj ] ^   æzāj } • Áæ† Á` à b &o Á ç Á@ Æ [ } å ãzāj } • Áj - ÁDÚŠÁ FHEI I EĂ
National Parks and Wildlife Act 1974	V@A⊳ÚYÁ088xo4j¦[çãâ∧∙Ás@Á ]¦[c^&cāj}}Áj,-Á02â[¦ãrājæap/(j,àb∿∨Áæ)jåÁ ] æ8k∧•ĚÁ	Þ[Á,^!{ ã e Ást Á^~ ă^å Á } å^!Á@ i ÁOBdÁQ ,^ç^!ÉB Á æ& [åæ] &^A á ã@ i @ i AOBdÉQ ,^ ^{]][^^^ & Ás à & i @ i AOBdÉst A æ st Á; Ás à A i & i @ i A i & i A æ st Á; Ás à A i A i & i A i & i A æ st Á; Ás à A i A i & i A i & i A æ st Á; Ás i A i A i & i A i & i A æ st Á; Ás i A i A i & i A i & i A i & i A æ st Á; Ás i A i A i & i A i & i A i & i A æ st Á; Ás i A i A i & i A i & i A i & i A i & i A æ st Á; Ás i A i A i A i & i &

Š^*ã æaāį}Á	Ü^ ^çæ <del>)</del> &^Á	Ü^˘˘ã^{^}œÁ
Roads Act 1993	V@ÁÜ[ zæå•Á0BkoÁ^or Á[`cÁ ¦^``ã^{ ^}o Á[¦Áæ]]¦[çæ‡Á[ Á&æ¦^Á [`cÁ&^¦cæä] Á [¦\•Á} å^¦cæh^} Á āc@aj Á c@ Áçã&aj ãĉ Á[-ÁxeÁ[ zæå ĔÁ	Wi å^i ko@ ÁOBcÉsej ^ Á [ !\ • Áţ Á æáj ` à lä&Á[ æåÁ^`` ã^• k@Á &[ } • ^} ơá Á@Á/, ^^åÁU@A Å[ } • ^} ơá Á@Á/, ^^åÁU@A Å[ } • ^} ơá Á@Á/, ^^åÁU@A Å[ } • Å oá A@A Ô[ ` } & alæÓ[ } åäāt } ÁHHÁ -Á Ù&@å`  ^ÁHÁ -Á@A Ö^ç^ [ ] { ^} ơÔ[ } • ^} ơÁ  ^` ă^• Á@exáxá ^ad^åÁ @e [æ* ^Á] æåÁa^Á&[ } • d` & a å A @e [æ* ^Á] æáÁa ^&[ } • d` & a å A @e áa^} Å[ æáÁa ^&[ } ^ & @e áa^} Å[ Å@A VÙÔÁ } å^¦ÂU^&at } ÅE [ -Á[ æåÁæ^c Ás] ^A [ -Á[ æåÁæ^c Ás] ^A @e áa^} Å] ] [ < a Å A @e áa^} Å] Å [ -Á[ æåÁæ^c Ás] ^A A ] ![ < a Åf a å ^!ÂU A A ] ![ < a Åf a å ^!ÂU A A A A A A A A A A A A A A A A A A A
Water Act 1912	V@ÁYæ&\Á038xc48[}d[]•Á@A ^¢dæ&c4]}Áæ}åÁ`•^Á{-Á æ&\A&}åA &æ\?3]*Á`Ó&&&c4]^â&*A •[`¦&^•E¾ @\^Á&A æ^\A@&3]*Á ] æ}A&A[O&AAA æ&\EA	CĘIÁ':[`}å, æc^¦Aå[¦^•Á ∄•œd ^åÁæAv@A`ãrAæA^Á [ã2^}•^åÁà`Áx@AÖ^]æd{^}oÁ [-Áræ?¦Áæ}åÁO}^!*^Á¢DY ÒDĚÁ CĘIÁ*¢dæ&cąĨ}Aà['^•Á]^\æc^Á ∄ Áæ&&[¦åæ}&^A ã@A©A æ]]¦[ç^åÁ&A*}•^Á&[}åãaį}•ĔÁ
Tweed Local Environmental Plan 2014	V ^^åÁŠÒÚÁQEFI Áå^-aj^•Á@Á ∶[}aj*Á{¦ÁœA*ãr^Å @&@∮¦^•&¦ãa^•Á  æjåÁ*•^Á{àb∿&ãç^•Á	Ü^``ā^{ ^} o Á Á Á@ Á/, ^^åÁ ŠÒÚÁGEFI Á ^!^Á&[}•ãå^!^åÁ ĝ Á@ ÁOCHŽÁ Y ã@Á^*ælåÁ[Á@ Áæî[` ơÁ Á c@ Á `æl' ÁbóAæe Áa^} Á å^{ [}•dæe åA@æk@ Á ]'[][•^åÁ ¢dæ&cã]} Á][}å•Á æ^A[&æe åA @ #]^A]}a•Á æ^A[&æe ÅUWCA]} Å/] å•Á ŠÒÚÁGEFI EÁ Y ã@ Á@ ÁUWCA[}^Á ±¢dæ&cã;^Á§ å`•dã•oÁæ^Á ]^!{ ã•ãa ^Á,`!•`æ) oÁ[ Á Ö^ç^[] { ^} ÓQ[}•^} dĔ

### ' " Á CdYfUhjcbU` Uddfcj U`g`

CE;] ¦[çæ+•ÉÁ&A^} &^• Á; ¦Á;^¦{ هَ• Á^˘ ă^åAà^Á& Áx@ Á^\* ã |ææā;} Áã c^åAa; Á/æà|^Á+ËFÁæ}^Á&^cæaā/^åAa; Á Væà|^Á+ËCÁæ) åÁ∞K&[]^Á; حÁc@ Áæ]] ¦[çæ+•Áæ) åÁ&A^ &^• Áæ^Aæçæaāæà|^Á;} Á ão^ÈÁ

#### HUV`Y`' !& 5ddfcj U``gi a a Ufm

<b>Œ]]¦[çæ‡Á</b>	Ü^*	Q∙`^Áaæe^Á
Ò}çã[}{ ^}œaÁÚ¦[c^&cā‡}Á Šã&^}&^ÁÇÒÚŠÁ>[bĚ⊼H∈ÏÏDÁ	Ò}çã[}{ ^}oÁÚ¦[ơ^&qãi}Á CE cq2;¦ãc ÁÇÔÚOEDÁ	F€ÁR(`}^Á
Ù^&cāį}ÁFHÌÁsa]¦[çæ‡Á	V, ^^åÁÙ@ã^ÁÔ[`}&ājÁ ÇÔ[`}&ājDÁ	ËÁ
Ó[¦^Ája&^}●^Á	ÖÚQÁÁU~a&∧Áį́A∕rær∿¦Á	FÎÁÛ^]c^{à^¦ÁG€€IÁ

#### ' "(Á 9bj ]fcba YbhU`dc`]Whi

 $CE \hat{k} [ ] \hat{k} + \hat{k} [ | & \hat{a} + \hat{k} \\ ce \hat{a} + \hat{a} + \hat{b} + \hat{c} + \hat{$ 

#### '') Á 9bj ]fcba YbhU`f]g\_ UggYgga Ybh

#### HUV`Y'' !' `F]g\_`a Uhf]l `

Šã^ ã@[[ åÁ	Ô[ }•^~`^} &^Á		
Á	Š[ , Á	T^åã{ Á	Pã @Á
Y ã∥Á 88°¦Á	ТÁ	PÁ	PÁ
Šãi^ ^Á[[Á][&&`¦Á	ŠET Á	ТÁ	PÁ
W} ã^ ^Á{[Á[&&č¦Á	ŠÁ	ŠET Á	ТÁ
Yã∥Á,[ơ4(,&&`¦Á	ŠÁ	ŠÁ	ŠET Á

#### HUV`Y`' !( `F]g\_`UggYgga Ybh`

Ce≊]^&oÁ	Ôl^æłā * <del>kaj</del> åÁ Õl` ààađ * Á	Ò¢&æ;æi }	Ү æ @ *Á	Tæ∿¦ãtt∳Á/¦æ}•][¦ó∱	Ü^ @क्के वैवेंत्स्वर्ज्त } Á
Ùĭ¦~æ&∧Á∕æe∿¦Á	ТÁ	PÁ	PÁ	ŠÁ	ТÁ
Õ¦[`}å,æe∧¦Á	ŠÁ	РÁ	РÁ	ŠÁ	ТÁ
Ô[} cæ{ ã; æcã; }Á	ŠET Á	ŠET Á	ŠETÁ	ŠÁ	ŠÁ
Ò&[  [ *^Á	ТÁ	ŠET Á	ŠETÁ	ŠET Á	ŠÁ
Þ[ãr^Áæ);åÁXãa¦æaqā[}Á	ТÁ	ТÁ	ТÁ	РÁ	ŠÁ
OBAÁ	ТÁ	ŠET Á	ŠÁ	ТÁ	ŠÁ
Šæ);åÁ∿•^Áæ);åÁ/¦æ);•][¦oÁ	ŠÁ	ТÁ	ŠÁ	ТÁ	ŠÁ
P^¦ãæ≛^Á	ТÁ	ŠÁ	ŠÁ	ŠÁ	ŠÁ
Yæc^Á	ТÁ	ТÁ	ТÁ	ŠÁ	ŠÁ

# (" =a d`Ya Ybhuhjcb



#### : ][ i fY`(!%? Ym9AG`Y`Ya Ybhg`

#### ('%Á 9bj]fcba YbhU`a UbU[ Ya Ybh'gnghYa 'XcVi/a Ybh'Uh]cb'

V@A^^A^}çā[}{ ^}œa‡A;æ}æ\*^{ ^}oA^\*o\*C^{ Aå[&`{ ^}o\*Aæ}åAs@®āA5jo\*\¦^|æeāj}•@3j•Aæ\*^A å^•&¦āa^åA5jAs@A[||[,ā]\*A`à•^&cāj}•EA

#### ('%% 9bj]fcba YbHJ`a UbU[ Ya Ybh'gnghYa `

 $\begin{array}{l} & \label{eq:constraints} \forall (\mathbf{a}, \mathbf{b}, \mathbf{a}) = \mathbf{b}^{1} \wedge \mathbf{b} = \mathbf{b}^{1} \wedge \mathbf{b} = \mathbf{b}^{1} \wedge \mathbf{b}^{1} + \mathbf{b}^{1$ 

#### ('%'&' 9bj ]fcba YbHJ`A UbU[ Ya Ybh'Gi V!d`Ubg`

OZÁ, ´{à^¦Á, Á^}çã[}{ ^} ca‡Á, aa) æt^{{ ^} cÂu`àĒÚ|aa) • Á`]][¦cÁc@ ÁÒT ÙĒV@ • ^Ás[&`{ ^} or Á@aæç^Á à^^}Á, ¦^]æt^åÁs[ÁsāA}cã^ÁA´`ã^{{ ^} or Ása) åÁj ¦[&^••^•Ása]]|a3aæà|^Ás[Á]^&ããa&Ás[]æsor Á;¦Ásae]^&or Á [-Ác@ Ásascāçānā\*•Ás^•& ¦ãa^åÁsjÁU^&cāt]}ÁTEÀO}çã[]{ ^} ca‡Á dæz^\*ã\*•Á; æi Ást‡•[Ása^Ása^ç^|[]^åÁsae Á !^``ã^åÁsQ[`\*@L`cÁc@ ÁÚ¦[b\*&cEÀV@ • ^Á;ā]/Ást‡•[Á`ãa^Á\*}çã[]{ ^} ca‡Á; æi Ast‡•[Ása^Ás^A; c] a, astar Á;} ãz EÁOZÁãro4, -ÁÚ[|a3a3\*•ÉÂU`àÉÚ|æ) • Ása) å Á dæz^\*ã\*•Á; ¦Ás@ Á`æ¦^É&sa+^Á; ¦[çãa^åÁsjÁA Væà|^Á ÉEÁ

#### HUV`Y`(!% 9bj ]fcba YbHJ``a UbU[ Ya Ybh`gi V!d`Ubg`

Ö[ & { ^} ơÁ/ãţ^ Á	Ö[& { ^} o Š[& eeaā] } Á
Ù[āļÁæ)åÁYæe∧¦ÁTæ)æ≛^{ ^}ơÁÚ æ)Á	CE[]^}åã¢ÁÖÁ
Šæ)å•&æ}^ÁTæ)æ*^{ ^}αÁ∪ æ)Á	O[]]^}åã¢ÁÔÁ
Þ[ãi^ÁTæ);æ*^{ ^}oÁÚ æ);Á	OĘI]^}åãcÁØÁ
OBãÁÛ ઁæ‡ãĉÁTæ);æ≛^{ ^} ơÁÚ æ);Á	OĘ[]^}åãcÁÕÁ
V¦æ⊶a&ÁTaa)a≛^{^}oÁÚ∥aa)Á	OĘ[]^}åã¢Á₽Á
OEa[¦ātājæ¢kÔč 覿¢k₽^¦ãaæt^ÁTæ)æt^{ ^}oÁÚ æ)Á	CE[]^}åãcÁQÁ
Yæ∙c^ÁTæ)æ*^{ ^}oÁÚ∣æ)Á	OĘ[]^}åãcÁRÁ

#### ('&A Cf[Ub]gUh]cb'VXUfh

V@^Á, æ);æ\*^{ ^};oÁ;d`&c`¦^Á;[¦Ás;]]^{ ^};œæaj;}Á;Á®@ã;ÁÔTÙÁs;Ás)]`•dæe^åÁs;Á28t`¦^Á;ÉGÈÁ;



: ][ i fY (!& 9bj ]fcba YbHJ `a UbU[ Ya Ybhghfi Wi fY `

### ("Á Fc`Yg'UbX'fYgdcbg]V]`]h]Yg'

 $S^{\hat{A}} = S^{\hat{A}} + [] \\ S^{\hat{A}} = S^{\hat{A} + [] \\ S^{\hat{A} + [] } \\ S^{\hat{A} + [] } \\ S^{\hat{A} + []$ 

HUV`Y`(!&`Fc`Yg`UbX`fYgdcbg]V]`]h]Yg`

Ü[  ^Á	Ü^•][}•ãàậããt•Á
Þ[¦c@}¦}ÁÞÙYÁ O≛*¦^*æ&∿Á Tæ}æ*^¦Á	Ú¦[çãā^Áæå^˘`æe^Á^&[覕^•ÁQ;^!•[}}^\ÉĂgiæ}&ãæ¢Áæ)åÁc^&@;[ [*ã&a¢DÁt;Á ^}•č¦^Á~~^&cãç^Áå^ç^ []{ ^}dĚati] ^{ ^}cæaāti}Áæ)åÁtiæabjc^}æ)&^Ati-Ác@éA ÒTÙÈĂ
Û`æ¦^ÁTæ)æ*^¦Á	<ul> <li>V@ A } çā[ } { ^ } œdd^ -] [ } • āi ājāž • A A@ A ) * æl ^ A a æl ^ [A a æl ^ [A a æl ^ [A a a æl ^ [A a a æl ^ [A a a æl ^ [A a a æl ^ [A a a æl ^ [A a a æl ^ [A a a æl ^ [A a a æl ^ [A</li></ul>
U] ^¦æ[[ ¦∙ Á	Ùãr^Á [  \^¦• Áng) å Ái ] ^ ¦æng[ !• Ánd Ái^•] [ }• ãna  ^ Ái ¦ Ángi ]  ^{ { } agi * Á ^} çã [ } { ^} caph Ka [ } d [  Ái ^ ængi '  ^• Ánd • [ & ãna * à Ă ão @ Ái@ ã Ana ængi ^ Ái [ ; \ EÁ / @ si Á ā & j * â • hÁ • Tā jā āi * á ð * Án @ ã Ángi ] æsko Ai } Án @ Ai > çã [ } { ^} o Ái @ ana * Ái   ændi āj * Ái [ ; \ EÉ • Qi ]  ^{ { ^} cā * Ándi à Ái æð æð jā * Ái > çã [ } { ^} cá @ ana * Ái   ændi āj * Ái [ ; \ EÉ • Qi ]  ^{ { ^} cā * Ándi à Ái æð æð jā * Ái > çã [ } { ^} ca #Ái } án   ændi āj * Ái [ ; \ EÉ • Qi ]  ^{ { ^} cā * Ándi à Ái æði æð æð jā * Ái > çã [ } { ^} ca #Ái } án   ændi āj * Ái [ ; \ E • Ü^] [   cā * Ándi à Ái æði æð æð a fi æð æð ^ i Ái @ } Ándi & ái & að a án æð ja * { ^ ændi i ^ i ^ i ^ i & a fi æð æð a Ái [ ] æð a ái & a a a a a a a a a a a a a a a a a a
ÞÙY ÁÚ æ)}āj*Á æ)åÁÖ}çã[}{^}oÁ Tæ)æ*^¦Á	<ul> <li>V@Á}çã[]{ ^}œdÁ^•][} •ãa ājādā •Á, Á@ÁÚ æj}ā, *Áej åÁÖ}çã[]{ ^} óA</li> <li>Tæjæ*^!Ág &amp;[`a^kÁ</li> <li>OE •ã ofáj Áå^ç^ []{ ^} dÉÅ [] ãt !å * Áej åÁ] åædā, *Á Áœ ÁÖT ÙÁej å</li> <li>œ•[&amp;ãæe*àÅ]aæ] •E</li> <li>Ü^][!oÁt ÁÕ^}^!ædAT æjæ*^!Å/Å Áœ Á^!-{!{ aj &amp; Aej à Ád ] åædā, *Á Áœ ÁÖT ÙÁej å</li> <li>@^][!oÁt ÁÕ^} ^!ædAT æjæ*^!Å/Å Áœ Á^!-{!{ aj &amp; Aej à Ád ] ] ^{ ^ } 6</li> <li>O} •`!^ÁA æjæ*^{ ^ } oÁ^çã, •Á Áœ ÁÖT ÙÁeA Å à^!œa ^} éa] *É</li> <li>O} •`!^ÁA æjæ*^{ ^ } oÁ^çã, •Á Áœ ÁÖT ÙÁeA Å à^!œa ^} éa] *É</li> <li>QA^ cã Á @ !^Á} çã[] { ^} œdA Å</li> <li>@ A œ A Aej à Aeg A Å Aeg A Å Aeg A Å A Å Aeg A Å A Å Å A Å Å Å Å Å Å Å Å Å Å Å Å Å</li></ul>

Ü[  ^Á	Ü^•][}•ãa ą̃ããv•Á
	<ul> <li>Uà catậi Átáy å Á ] å att Átáy Átáy Çã[ } { ^} cat Átát Át A ?</li> <li>Uà catậi Átáy à Á ] å att Átát Át A ?</li> <li>T at at A * à A ?</li> <li>T at at A * à A ?</li> <li>T at at A * à A ?</li> <li>T at at A * à A ?</li> <li>T at at A * à A ?</li> <li>T at at A * à A ?</li> <li>T at at A * à A ?</li> <li>A ?</li> <li>A ?</li> <li>A ?</li> <li>A * A * A ?</li> <li>A ?</li> <li>A ?</li> <li>A * A * A * A ?</li> <li>A ?</li> <li>A * A * A * A * A * A * A * A * A * A *</li></ul>
ÞÙY ÁÚ æ}}āj*Á æ}åÁÒ}çã[}{ ^}oÁ Ô[[¦åājæe[¦Á	<ul> <li>V@Á}çā[}{^}œ4Á^•][}*äāţāāā*Á, ÁœÁÚ]æ}jā*ÁæjåÁÖ}çā[}{^}œ4Á</li> <li>OE•ā cáŋ Á!^] æðj*ÁœÁÓT ÙÁţã &amp;[`áāj*Áæj^Áčč!^Á^çã ðj*IÈ</li> <li>Ö^ç^[] ÅÛ]eðj*ÁœÁÓT ÙÁţã &amp;[`áāj*Áæj^Áčč!^Á^çã ðj*IÈ</li> <li>Ö^ç^[] ÅÛ]eðj*ÁœÁÓT ÙÁţã &amp;[`áāj*Áæj^Áč]*[]</li> <li>Ö^ç^[] ÅÛ;[*ðj}ÁæjåÁÚ^áð] ÅÓ[}d[]ÅU]æj*ÁŋÁ&amp;]*`jææðj*Á</li> <li>D`*i^Á[] ÄÜ]^içã[!ÁæjåÁû@ÅA[^içæ]A^j¢Ő]}d[]ÅU]æj*ÁŋÁ&amp;]*`jææðj*Å</li> <li>O`*i^Á[] äť la*Á^&amp;[iå*Áæ]Å(@AA[^içæ]A^j*]!]]iãæ*^A^j*[]}^A æðjæðj*Å</li> <li>O`*i^Á[] ÄÜ]^içã[!ÁæjåÁû@A[iÅ^[^içæ]A^j*]!]]iãæ*^A^j*A[]}</li> <li>O`*i^Á[] äť la*Á^&amp;[iå*Áæ]Å[]![]iãæ*^A^j.*[]}]</li> <li>D`*i^Á[] ÄÜ</li> <li>A[] ÄÜA</li> <li>O`*i^Á[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] ÄÜA</li> <li>A[] Ä</li> <li>A[] A[] Ä</li> <li>A[] A[] A[] A[] A[] A[] A[] A[] A[] A[]</li></ul>

#### ("''% Gi Wt/2bhfUW/cfg

# ) " 9bj ]fcba YbHJ`HfU]b]b[ 'UbX'Uk UfYbYgg

OĘĮĄ́(^{à^\+ Á; Áx@ Á ઁ æ \^Á; [ \\-{ \&^Á; āļÁ^ &^āç^Á\*^}^\æ¢A} çā[ } { ^} œ ek kiæanjāj \* Á^\* æ kiāj \* Áx@ Á ā[ ] |^{ ^} œaeā] Ă; Áx} çā[ } { ^} œebÁ; æ) æ t^{ ^} o Á dæ \* \* ã + Ánj Áxes&[ \åæ) &^A, ã @ Áx@ Á^~ čā^{ ^} o Á [ Áx@ á ÁÒT ÙÁæ) å Á\* ] ] [ \cāj \* Áka; [ & `{ ^} œæā] } ĚÀÒ} çã[ } { ^} o Ákanj å Áka; { { `} ã ĉ Á kiæanjāj \* Á; āļÁb; &|` å^ÊÁ à č Á; æ Á; [ o Áka^ Áā] ã c^å Ák; Éb@ Á; ||[ ] āj \* KÁ

- Ò}çã[ } { ^} œa‡Á§jå \* &cãj } Áslæãjãj \* Á
- V[[|à[¢Áæaa]\•Á
- U c@\¦Á+]^&ãã&Átænjāj\*Áse Á^ĭĭā^åÁÇÈĖÁA}çã[]{{^}cÁ]ā|Á&[}d[|ÁsejåÁtænjæ\*^{^}dDÁ

V[Á}•`¦^Á→& & cāç^Áāt] |^{ } & cæaāt} Åt Áo@ Á'}çāt[ } { ^} cæ4Át æ}æt^{ ^} oftà lât æaāt} • É & @ ÁU`æ¦^Á Tæ}æt^¦Áās Á^•] [ }•ãa |^Át ¦Á\*}•`¦āt\*Áo@æxÁad|Árãc^Át^!•[ } }^|Áæb^Áæç æb^Át Áo@ Á^``ã^{ ^}orát Áo@áA` ÒT Ù Áæ}åÁ`] ] [ ¦cāt\*Áåt[ & `{ ^} cæaāt} } É Å

#### ) '%Á =bXi W¶cb<sup>·</sup>

Ùãr\Ár}çā[]{ ^}cæk/(æ)æ\*^{ ^}oÁ^~~ă^{ ^}oÁœ\*^&[{ { `}a8æe\*å/54,Á©A;ãe\*A;a\*a\*&a;a}Á;a\*&a;a}A; •cæ-ÉA`à&[}cæsq[}cæsq[:+A;a}aA;ããq[:+EX/@A5;a\*&a;a]A5;A5;A2]\*å^•A;@A[[[,ā]\*A5;-{];{æqa;}A4;

- O}çã[}{ ^}cadÁ^\* adÁ8[}c^¢c45]8(`åð]\*Ásiða \*Ásiða \*Asiða  Ùãc^Ár}çã[} { ^} cæþÁ; àb^&cãç^• Áæ}åÁæe\* ^o• Á
- Úãc^Ár}çã[] { ^} cæþÁ^ccãj \* ÁæjåÁ^} ãããç^Ár}çã[] { ^} cæþÁæe]^&c•Á
- Ò}çã[}{ ^}cæk/5g &ãã^}c/4^•][}•^/Áæ)å/4(æ)æ\*^{ ^}c/4\![&^å`¦^Á
- Uç^¦çã`, Á, Á`}çã[}{ ^} cæ ÁS[}d[|Á, ^æ č |^• Á Ásč ofæ) å Áscá Áč žejáč ÉÅ, [ã ^ Á, æ) æ \* { ^} dÉÅ , æe^\¦Á čæjáč ÉS;^\* ^ cæeáj } Á, ![c\*Scã] Ě Éæ) å Á, æ• c\* Á, æ) æ \* { ^} oÁ

 $Uc@!\dot{A};ca[] { ^} caet/as \bullet `^ \bullet Ase^ As[ { { `} as a ce^ a As[ A a ce- As@[ ` * @As[ [ |a[ cAset] \bullet Ase) a A[ c@ !A { ^^ca} * \bullet EAA }$ 

#### ) "& Á Hcc`Vcl `HJ`\_g`

V[[|à[¢Ávæ‡\●Á]ā|Aà∧Á;}^Á(^o@2,åÁ;Ávæārāj\*Áse;æc^}^●●Áse;āA^åč\*åč\*&ææāj\*Á,^\●[}}^|Á^\*æåäj\*Á ãr●č^●Á^|ææ^åÁt[Á\*}çã[]{{^}oÁse}åÁ&[{{``}ãc`ÈĂÜ^|^çæ}oÁ\*}çã[[}{ ^}œa‡Áse`^●Ásj&|čå^ÁQãčoÁ;ã||Á }[oÁs^Áqãiãc∿åÁt[DeA

- Þ[ãr^Á(ãuãtæaā) Á
- Oʻl[•ā]; Áæ); åÁ^åã], ^} cæaã]; Á&[; d[ |Á
- P[˘¦•Ą́,-Ą́,[¦∖Á
- V¦æ-a&Á
- Q &ãã^} oÁ^•] [ } •^Á
- P[˘•^\^^]ậੈ\*Á
- OEa[¦ātā]æaþÁæa)åÁ,[}ËOEa[¦ātā]æaþÁ@?¦ãæet^Á
- X^\*^œaaā]}Á&|/~æda]\*Á&[}d[|•Áæ);åÁ|.¦[c^&aaā]}Á
- Ö`•oÁæ)åÁ(å[`¦Á&[}d[|Á

#### ) " Á HUf[YhYX'Ybj]fcba YbhU`'hfU]b]b[ '

 $\begin{array}{l} \forall \mathcal{A} =$ 

- OBSããÁ\* |-æe^Á[ãļÁd:^æe{ ^}oÁ
- Ü^ç^\*^œeaña[}Á
- Q &ãã^} oÁ^•][}•^Á

#### ) "(Á HfU]b]b[ fYVtzfXg

V¦ænājāj\*Á^&[¦å•Áæb,^Áįænājœnāj^å/ĀgjÁæÁ∳ãč^Ás!ænājāj\*Á^\*ãic^¦Á;@38;@45gj&]ĭå^•KÁ

- Ö^cæa‡•Áţ~Ás@eÁţ^¦•[}Ási^āj\*Áslæāj^åÁQ;æ{^ÊA[|^Ê&{[{]æ}^DÁ
- V¦ænjāj\*Ásaæ^Á
- V^]^A[Addaaajāj\*Á

# \* "Á = bVJXYbhg UbX Ya Yf[YbVJYg

Q)&aã^}o•Áa)åÁ,^aelÁ(ã••^•Á,æʿÁ,&&č¦Á,ão@Á,[ơ·}oãaelÁ,¦Áa&&čaelÁ\*}çã[]{ ^}oaelÁ@eel{ÈAQ,&ãå^}o•Á āj&|čå^kÁ

- •Á Ò¢&^^åæ) &^Áį ~Á@Áįã ão Đ)^¦-{ |{ æ) &^Á&iã^}iád@Áæ] |[ çæ|Á
- •Á Ù] ậ|•Á; ¦Ár æ)•Á; Á@ee æ¦a [č•Á\*à•æ) &^•Á; ậ•ÊA; ^|•Ê&@{ &&æ]•DÁ
- Á Ô |^ælāj \* Á¦ ¦ Á\¢&æçæaāj } Áį čorāå^Ás@? Áæj ] ¦[ç^å Áà[č] } åæl^Á
- •Á Ò¢&^^åðj \*Ás@^Á,`{ à^¦Áj. Ás' & •Ásðj ] ¦[ç^åÁj.^¦Ásiæê Á
- •Á Ò¢&^^åāj\*Áx@^Áæj]¦[ç^åÁv¢dæ&caji}}Ájãj ãaÁ

V@Ánão×qnÁú[||˘qā;}ÁQ3&äāo^}cÁÜ^•][}•^ÁTae)ae\*^{^}cÁú|ae)ÁÇÚÖÜTÚDÁarÁæÁ^\*\*|aea[¦^Áo^\*\*āo^{^}cÁ ,@38@Áaj&|ĭå^•Áo@Án{{^¦\*^}&^A^•][}•^Á,¦[&^åĭ¦^•Áq[Áao^Áqā;]|^{ ^}c^åÈĂv@ÁÚÖÜTÚÁarÁ,¦[çãa^åÁ ājÁC0;]^}åãcÁSÈĂ

#### \* '% 9a Yf[ YbV/mV/zbHJV//g

Ö^cæa‡+Á[¦ʎː¦\*æ)ā\*æa‡i}+Ás@æa∱(æʿÁs\^Ás(}cæ&c^åÁt[Ár)=][}åÁt[Áæ)Á\*}çā[}{ ^}cæ4x{{ ^!\*^}& & Ásd^Á ]¦[çãå^åÁs\^[, ÁsjÁ/æà|^ÂiËEÈ/@•^Ás^cæa‡+Ásd^Ásqcæa‡æà|^Át[Áæ|Áiãr⁄Á,^¦+[}}^|A;}Aiãr∕Á;ā}æ\*Á æ}åÁsjå`&cā;}Á;ær¦ãædEÁ

#### HUV Y'\* !% 9a Yf[ YbVm/VtbHJVM XYHJ] g

O≇ ^} & ÁÁ	Ô[ } æ&oÁ	
Ú[ 38x/Ábá/ðā/Ábá/OE; àč a3;&/Á	€€€Á	
P[∙]ãaa¢Á	T č ¦,ā∥č { àæ9øÄÖãrda&oAP[•]ãæ‡Á Ò,ā) * ÁÙd^^dǼT č ¦,ā∥č { àæ99Á €GÂÎÎÏCÁFÌGGÁ	
ÞÙY ÁP^æko@Á	€GÂÍÌÌÁGÏÍ€Á FH€€ÃÍÍÁÍÍÁ	
ÞÙY ÁY [ ¦\ Ô[ ç^¦Á	FHFÁÐÍ€Á	
ÒÙŒÁ	FHFÁÍÍÍÁ	
V, ^^åÁÙ@ã^ÁÔ[ັ} &ãjÁ	€GÂÎÏ€ÁGI€€Á FHE€ÁGJGÂÏGÁ	
Υ ѾÒÙÁY ą̃å ã^ÁÜ^∙&`^Á	FH∈€Á€JIÁIHÏÁ	

#### \* "& A = bV]XYbh`]bj Ygh][ Uh]cb`

CE[|Á5], & āā^} or Ásel^Ás[[&`{ ^} c^åÉ59; ç^•cāt asenā] •Á&[} å`& c^å/se); å Áse&caā] }Á, |aa) •Á\*•caaai|āt @ å Á59; Á{ ¦å^¦Á co2een Á50; Á53; & āā^} c%s[[^•A,í] cÁ, & &` ¦Áset asend ÈÀY @ ¦^Á?••[} •Ásel^Á?ad} cÁ+[{ Ás@ Á59; ç^•cāt asenā] }Á; ¦Á &` ¦!^} cÁ, ¦[& ^å` ¦^•Ásel^Ásel^} cāað à Áser Ási^ā] \*Á59, ^~~^ & caāç^Ê5ko@ ÁÖT ÙÁ; ā||Ási^Á^çāt^àÁq[Á59; &|`å^Ác@ Á ā] ] ¦[ç^åÁ, ¦[& ^å` ¦^•Á; IÁ^``ā^{ ^} cĐĂ

O5;Á5)&ãå^}oÁ5jç^∙cãtaeãj}Á5j&|ĭå^∙Ás@?Á{[||[¸ã]\*Ásaea&AA\*|^{{ ^}@~hÁ

- Á V@: Á&æĕ• ^ ÉÁcāį ^ Áse) å Áåč ¦æsājį } Áj, Ás@: Á ç^} o Á
- A OBScall } Ászek ^ } Ás Á^ | ascall } Át Ás@ Á\* ç^ } oÁ, ãt@Á^• ] ^ &oÁt Á&t } cast  $\hat{A}$  [  $\hat{A}$  ascall }  $\hat{B}$ , [  $\hat{C}$  as  $\hat{A}$  ascall }  $\hat{E}$  by [  $\hat{C}$  as  $\hat{A}$  ascall }  $\hat{E}$  by a  $\hat{A}$  + ^  $\hat{A}$  ascall  $\hat{A}$  as  $\hat{A}$  + ^  $\hat{A}$  as  $\hat{A}$  ascall  $\hat{A}$  ascall  $\hat{A}$  as
- •Á Ö^czzáp•Áţ-Áze)^Áţ ^ ze\* \^Ázze ^} Áţ \´A; [] [ ^ åÁξ Ázze ^} Áξ Áş \^ç^} oÁzzá ^& `\^} zézé ^} Áş `A` & @źze ^} Á

Á

CЩ Á,^¦•[}}^|Áæc^Á^˘ă^âÁξ Á^][¦ơÁæd|Ás)&ãã^}orÉáee ÁsaÁse ÁsaÁse ÁsaÁşæa≱æà|^Á; ^co@;åÁ;-Á æåå¦^••āj\*Á;@2;lc&[{āj\*•ÁsjÁ,¦[&^åč¦^•Éác!æäjāj\*Á;¦Á^ččāj{^}cÉáce)åÁse ÁsaÁşa4;]][¦č}ãĉÁ[¦Á ãį]¦[ç^{^}ôÉá

#### \* " Á Bchjzjvvhjcb

V@ÁÜ`æ¦^ÁTæ)æ\*^¦Á¸ä|Áãį { ^åãæe^|^ÁÇ ão@3jÁGIÁQ; '⊧•DÁj[cã-ÁÖÚOBOÒÁea)åÁea)^Á;c@¦Á^|^çæ)oÁ æ\*^}&°Áį-Áa)^Á5j&ãå^}oÁc@een%&æi•^•Á;¦Áį æ°Á&æi•^Á;æe^¦ãæ4Á@ee{{ Át[Ác@Á\*}çã[]}{ ^}cDÁYão@3jÂÁ åæî•Á;-Á;[cã-ã]\*ÁÖÚOBOÒÁea)åÁ^|^çæ)oÁet^}&ã\*•Éáen&å^cæa‡^åÁ;¦ãec^}Á^][¦oÆiÁt[Áa^Á\*`à{ãec^åÁt[Á ÖÚOBOÒÁea)åÁea)^Á;c@¦Á^|^çæ)oÁet^}&?Ác@eenA

- æÈ Ö^∙&¦ãa^•Áx@/Ásæe^Ékaąĩ^Ékaąĩ^Ékaąĩ^Ékaąĩ^Ékaqĩaé ¦^Á;i-Áx@/Ár¢&^^åæa}&^Eaj&ããa^}c
- àÈ Qì^} cãa Ác@ Á&aĕ ^ ÁQ; ¦ Áã ^ | ^ Á&aĕ ^ ÁD, Ác@ Ár¢&^^ åaa) &^ E3, &ãã^ } c
- &È Ö^•&'aa^•Á, @eenÁee&caji } Á@eenÁa^^} Ásae\^} Ási // Ás
- åÈ Ö^•&¦ãa^•Áx@^Á;¦[][•^åÁ;^æ\*`¦^•Áq[Áseåå¦^••Áx@^Á\*¢&^^åæ)&^E3j&ãa^}c

V@AÛ≚æ¦^ÁTæ}æt^¦Á,ā|A\$ą { ^åãæer^|^ÁÇ,ãc@3,ÁGIÁ@{≚¦∙DA}[cã≏Ác@AÒÚOE4∖-Á}[||≚cā‡}Á5,&ãå^}o•Á;}Á [¦Áæ{[č}àÁs@A`ãrA,@3&@AeçrA{&&&`¦\^åA\$jÁc@A\${[č+•^A{-Áse&cãçãa?eA\$jÁc@A{[|[,ā]\*Ásaå&`{ •cæ}&^•hA

- Qás@Áæ&čæļá; \Áj[c\*}cãeļÁ@ed{ Áţ Ás@Á@edqc@át; \Áæ^č Át Á@{æ} Ás^āj\*•Át \Á\*&[•^•c\*{•Áār Á}}[c4;tā;ãeļÁ
- Qxfxx8cč aq4, l/A, [ c^ } cãaq4, [ • A, l/A, l[ ] ^ l ĉ Aŝaq ač ^ Áçã & č åã \* Á& 2 ač ] Á& o D fæ [ & ãæ \* å Å ão @ fac Á
   ] [ ||č qã } Á§ & ãa^ > o A ¢ & ^ å ÅÅF € Ê € € € Á

 $Y \mid \tilde{a}c^{A} \wedge$ 

# +''Á 7 ca a i b]**\\Uh**]cb<sup>·</sup>

Á

#### +'%Á =bhYfbU``Wta a i b]WUh]cb`

Ô|^æáÁj}^•Á; Á&[{{`} 38æaā;}Á@[`\*@;`Áæ¢|Á^ç^|•Áæ}åÁ`}&aā;}A&Z[{ a) æ\*^{{ }} add ave faithter and ave faithter and ave faithter and ave faithter and ave faithter and average faithter average faithter average faithter average faithter average faithter average fai

#### +"& A 8]fYVViVtea a i b]hmVtebgi `HJh]cb'

P[|&ā[ Á,ā|Aá^Á,![æ&cāç^Á5j Á^\*æåÁt[Á5j-{;{ ā}\*Ác@A[8æ4A8[{ {`}āĉ Áej åA]æcā&`|æ|^Ác@Á •`;![`}åā]\*Á,![]^;cā•Á,Ác@Á,^;-{;{ a} & &A(Á,Ác@A[]^;æaā]}•Áej åÁej ^Á;ā] ãa&eaj o4,:[][•^åÁ &@eaj\*^•Át[Ác@A[]^;æaā]}•Á;Ác@Á`æ;;^At[;Áão A[]^;æaā]}ækái`;æaā]}ÈAQ { ^åãæe^|^Á`;![`}åā]\*Á [[&ea4Á^•ãå^}orÁ;ā]Aéa^A5j-{;{ ^åA[,Ác@Á8[{ { ^} & ^} & ^} ]^;æaā]} akái`;æaā]}ÈAQ { ^åãæe^|^Â`;![`}åā]\*Á @ed4Á^•ãå^}orÁ;ā]Aéa^A5j-{;{ ^åA[,Ác@Á8[{ { ^} & ^} & ^} ] @ed4Á^•ãå^\*orÁ]æaā]\*Át[Á\*ão@;kC@Á]cæ\*^ÁrÁ;!ÁCA\*¢dæ&cā]}Áea^a EÁ

#### +" Á 5[YbWmWta a i b]WUhjcb

V@ÁÛ`æ¦^ÁTæ)æt^¦Ája|Áa^Ác@Á(æanjÁ)[ājoÁ(-Á&[}cæ&oÁjāc@Á^\*`|æa[¦^Áet^}&að•Á^\*æ¦åāj\*Á ^}çã[}{ ^}oÁenjåÁ&[{ { `}ãc Áas•`^•ĚV@ÁÛ`æ¦^ÁTæ)æt^¦Ája|Áa^A^•][}•ãa|^Á{¦ÁA^][¦cāj\*Á[}Á c@Á[}\*[āj\*Á\*}çã[]{ ^}cæAj^\-¦-{¦{ æ)&^Á[-Ác@Á]¦[b^&oÁ[Á^\*`|æa[¦^Áet^}&að•Á`&@Áæe ÁÒÚOEÉÁ ÖÜÒÉÃŎÚOBOÓAenjåÁÔ[`}&ā)EÁ

P[|&ā[Á @ed|Á;æā]æā]Áexá&[{{`}}ā&æaā]}Ásiā^&c[¦^Á;Áed|Á^|^çæ)ơ∱,~ā&^|•Áe)åÁ&[}œa&orÁ;ãc@a]Á\*æ&@Á |^|^çæ)ơŐÕ[ç^\}{{^}ofset^}&îÁg Á;\å^\Ác@exáe)Áse&cã;ãc Á;\Á^ç^}ơfs}ďs æ`As@exá,æ}æd;ædorÁsiā^&oÁ }[cãā&æaā]}Á;ā]Asi^Á;å^\cæ}^}ÈV@o•^Áse&cã;ãã\*oÁg &|`å^Ásičoáse^A;[cáAā]æd^åA[kÁ

- •Á OE;^Á; æb;¦Á&[}œæ; ā;ææā;}Á^ç^}oÁ
- •Á W}^æbc@aj\*Áæj^Áæb&@æ^[|[\*a&aaÁæb&^~a&orÁj¦Á^{ aaāj•Á
- •Á OĘ ^ Á ãt } ãaBaa) oÁzaātǐ |^ Áşi Ás@ Át } çã[ } { ^} cæþÁ, ^ |-{ | { æi & A A @ A ] ^ | æiāt } Ás@æeÁ, æi Ás ^ Á āj å ãaBæe^à Ásǐ |āj \* Á cæ) å æi å Á; [ } ãt |āj \* Á; | & ^å` | ^ • Á

#### +"(Á 7 ca a i b]hmWtbgi `HJhjj Y Wta a ]hhYY

Q\Áxe&&{|åæ}&^Á,ãc@ÁÔ[}åãaā]}ÁJÁ;~ÁÛ&@°å`|^Á.Á;~Áx@^ÁÖ^ç^|[]{^}o^}OÁÔ[}•^}oÉ≦xeÆÔ[{{`}}ãĉÁ Ô[}•`|cæeãç^ÁÔ[{{ ãoc^^ÁçÔÔÔDý, æ•Á•cæèlãe@°åÁ;!ā[¦Át[Á`zæ¦îā]\*Á;]^¦æeãį}•Ás\^\*ā]}ā]\*Á;}Ë≊ã^ÈÁ

V@~ÁÔÔÔÁ;@ee‡|Áa^Á&[{]¦ã^^åĄ[,Ác@:Á[[[[, ã] \* Á[ ^{ à^¦• kÁ

- Á OE; Á5; å^] ^} å^} oÁ&@eaái] ^¦•[} Á
- •Á V, [Á; ^{ à^\+ Á; Á;@^Á[&aa‡Á&[{ { `}}ãĉ Á;aa} å Ábá;\Á caa`^@; |å^\+ Á
- •Á U}^Á^]¦^•^}œæãç^Áţ~ÁœA/,^^åÁÛ@ã^ÁÔ[`}&ãjÁ
- •Á V, [Á^] ¦^•^} cæaãç^•Á(-ÁP[|&ã[Á
- ●Á V@~ÁŠæ))å@2||å^¦Á

V@Á&@æaā]^!•[}Áaā Áaēj][āj c\*åÅaî Áx@ÁÛ^&¦^cæeî Áţ Áx@ÁÖÚ0BÒÈÁ/@Á&@æaā]^!•[}Áj āļÁ^][!cÁ æ}}čæļî Áţ Áx@ÁÙ^&\^cæeî Áţ}Áx@Áţ]^!æaāţ}Áţ Áx@Á&[{ ãxc^^ÈÁ/@Á[ &æqÁ&[{ { `}} ã: Ásē}åÁţ c@!Á •cæe\^@[å^!Á^]!^•^}cæaãx^• Ásē}Åaēj][āj c\*åÅaî Áx@ÁÛ/&\~&\^cæeî Áţ ||[ ¸ā]\*Áseåç^!cæi^{ ^}cábaj åÁţ c@!Á { ^åãædĚÔ[{ ]æ}î Ásē}åÁÔ[`}&ājÁ^]!^•^}cæaãx^• Ásē^Aédj][āj c\*åÅaî Ár?[|&āţ Ásēj åÁx@ÁÔ[`}&ājÁ !^•]^&caãx^[] ĚĂ

#### V@∕ÁÔÔÔkÁ

- Ô[}ç^}^^ 4, {\ A^ca, -{\ A^ca, -{\ A^ca, -{\ A^ca, --ca, ^çã , Áæ) å Á, : [çã å Áæåçã & Á; } Áœ Á; } Áœ Á; çã [ } { ^} œ Á/, ^ : -{ :{ a) & Á; -Áœ Áå ^ ç^ [] { ^} dŹ
   ã & à à \* Áæ) ^ Á?; çã [ } { ^} œ Á
   ã [ :a] \* Á^• ` |œ É á ã Á^] [ :œ É á ã Á^] [ :œ É á
   a à a \* Áæ) ^ Á?; çã [ ] { ^} œ Á
   8 [ { ] |æ a œ Á

 $\mathsf{P}[|\&\tilde{a}[A\hat{a}ec/\delta for A[, ]A^{\circ}c]^{\circ} + \mathsf{K}A$ 

- Ú¦[çãå^•Á; [ÁQĐÁ; ~3&ãæþÁ^] ¦^•^} cæsã;^•Á; ÁP[|&ã; Á;[Ásec/} åÁseþ/ÁÔÔÔÁ; ^^c3; \*•Á
- Ú![çãā^•Á@AÔÔÔÁ,ã@Á^\*č|æbÁ§j-{; { æeā[} Å]; Ás@Á?;çã[] { ^} æebÁ], ^; -{ ; { æb & Abe} åA { @ ba ^c^[] { ^} œbA, ^b @ ba ^c^[] { ^} œbA, ^b @ ba ^c^[] { ^} œbA, ^b @ ba ^c^[] { ^} @ ba ^cA, ^b @ ba ^c
  - . V@^Á,\[b^&oÁse]]\[çæ‡Áse)åÁ^}çã[}{ ^}oÁ,\[c^&cã[}Áã&^}&^
  - . V@\Á`æ\¦^Ą[]^\æaāį}•Ą́|æ), Áæjå, Áæjå, Áæjå, Áæjå, Aæjæ\*^{ ^}, o∮, |æ), •
  - . Ü^• ઁ |œ Ą́į -Á́t} çã[ } { ^} cæļÁ́i [ } ãť[ ¦ãj \*
  - . O5;} ă¢4Â);çã[}{ ^} co+Á; A} co+A; A
  - . OEľåãoÁ^][¦o•
  - . Ü^][¦o•Áţ}Á&[{{`}ãĉÁ&[}&^\}•Áţ\Á&[{]|aaāj,o•Áaa}åÁ?[|&ãį qeÁ^•][}•^•
  - . OF ^ Á c@ ¦ Á j { | { accaj } Á s@ec Á ac Á a ^ Á ] ^ & & a á a ^ Á s@ Á ) ^ & ' ^ ce ^
- Ú¦[çãå^•Á(^^cāj \* Ázssājāãð)•Á[¦Ás@ ÁÔÔÔÁ
- CE:| æ) \* ^ Á æ^ Æj ] ^ &œj } Á! ; Ás@ ÁÔÔÔÊÆÁ, ^ &^ • æ' Á
- Yão@a)ÁGÌÁ\$uæê•ÁįÁo@A&[{{ãoc^^qA;^^caj\*ÉÁ;[},ae¦å•Á;[Á\*æ&@&&[{{ãoc^^Á;^{ a^{h}A}
  - . OEÁ&[]^Áţ-Ás@≥Áţājĭc^∙
  - . V@\Á&[{]aa}^q\Á^•][}•^Á{[Áaa}^Á`^•a];}•Á;lÁaaa, Á`^•a];
  - . OE; ^ Á5; -{ |{ accaj: } Á^~ `` ^ c^ å ÁsceÁs@ Á; ^ ^ ca; \* Ás ^ Ás@ Á&@easi ] ^ | [ }
- Tæ\^• Áx@•^Á(ājčc/• Áxaçæaajææ)/Át[Áx@/Ájča)æXáçãædáar Á ^à]æt^Á
- Ü^•][}å•ÁţÁæ)^Áæåçã&^Áţ¦Á^&[{ { ^} åææāţ}•Ás@ÁÔÔÔÁţæÂ@æç;^ÁşJÁ^|ææãţ}ÁţÁs@A
   ^}çã[]{ ^}œ4Áţæ3; æ\*^{ ^}o4şLÁş^!-{ [ { æ} 8^Aţ Ás@Áå^ç^|[] { ^}oÁ

#### +') Á 7 ca d`U]bhg<sup>·</sup>

D2\$&[{]|ænnjorÁ^\*ãrd^Ánjo^Á@zerÁso^}Á\*oczea:|ãr@°åÁsh^ÁP[|&ãi[Áti[Áti]^¦æanii]>'æanii]>•Ásh^\*āj}āj\*Áti]ËãrÁ •[Ác@zecÁc@Af,`à|ãbÁ@zeç^Ásca4,[ājcAti\_Á&[}cæsbcAtiÁr¢]¦^•••Ásaj^Á&[{]|ænijorÁtijAtiæanii]>•Ásh[Ac@Ati]^¦æanii]>•Á [-Ás@Af`æs¦^ÈÁDEÁt^\*ãa:|^ÁA^&[¦åÁti`•oÁsh^ÁA]cAti\_Ásce|/&[{]|ænijorÁtiæash^ÁtiÁs@Ati]^¦æanii]>'æanii]>•Á {]|[^^^•A, Ati]/Áset^}cAti ^{]][^^^•A, Ati]/Áset^}cAti CE;^Á&[{]|æanj,cÁ^&^āç^å Á<@eeqlÁå^Á{[¦, ædå^å Á<br/>tá (k@ ÁÛ`æd;^ÁTæ);æd\*^¦Á, @tá allÁaj;ç^•cātæe^Ás@ Á<br/>&[{]|æanj,ce Á^&^āç^å Á, allÁai^Å<br/>A<br/>&[å^å å A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>A<br/>

- V@\Á\$uæe^Aæ)åAæã(^A[, -Ás@A&[{]|æãjoÁ
- V@A, ^c@, åAa^A, @3&@4x@ A&[{]|aaa, cA, aaeA, aaa^A
- OE; ^ Á; ^ !• [ } æd/å ^ cæaije Á; Ás@ Á&[ { ] |æaij æ) ofs@æek, ^ !^ Á; ![ çãa ^ å Áà ^ Ás@ Á&[ { ] |æaij æ) ofs, ! É&ai, [ Á
   \* &@/å ^ cæaije Á; ^ !^ A; ![ çãa ^ å É&ai, [ c\* Ás[ Ás@æek/ --^ &ocá
- V@Áæ&cāj } Áæà^} ÁàˆÁœÁ´ æl¦^Áj æj æ\*^¦Á§ Á^|æaāj } Á[ ÁœÁ&[ { ] |æāj cÁ
- O5; ^ Át [ [[ , Á ] Á&[ } cæ&oá, ãc@k@ Á&[ { ] |æāj, æ); o Áæ); å Áæ); ^ Át [ [[ , Ë ] Áæ&cāt } Á } å ^ ¦ cæ\ ^ } Á
- QÁ, [Áæ&qā; }•Á, ^\^Áze\^} É& [& { ^} cæaā]; Af, -Ác@Á/2e [}•Á, @Á, [Áæ&qā; }Q DÁ, ^\^Áze\^} Á

CE[|Á&[{]|æanjoró Á^&^ãç^å Áj āļ|Áa^Á^&[¦å^å Ánjó Árog Ái}\*[ā]\*Á\*}çā[] { ^} cæahÁ^çað, •Á } å^!cæah^ } án Á P[|&ā] Ánaj å Áno [^-[¦^Á, ¦^+^} c\*å Ánj Áno Án] ^& cãg^ÁOEÒTÜÈV@ Á&[{]|æanjoró Á^\* ār c\*¦Áj āļ|Ána+p[ Ása^Á { æna^Ánac; cæanjæan |^Át[ Áneč å ãa[!•Ási` ¦ā]\*Áno Ánj å^] ^} å^} óA } cA] { æna ^ Ánge æna and [} { and [} Ange æna and [] } a

#### +'\* Á 8]gdi hY fYgc`i h]cb

Q Ás@ Árç^}cAs@eenAeenAsiā] čolÁestār^e Ási^ç ^^}ÁP[|&ā[ Áse) å Áse) [c@ ¦Áj æstĉ Ásj Á^|æazī[}Át[Ás@ Át] ^ ¦æazī[}Á [-Ás@ Á ĭæst^i Át¦Á^ ĭā^{ ^} œ Áse]]|&Bæazi|^Á }å^¦ÁÖ^ç^|[]{ ^}cASi @ e^Asi ^ (] }e^A; !^-^¦Ás@ Át æsec\*¦Át[Ás@ Át)^& '^æst Át Ás@ ÁÖÚOBSÒÉAj @ e^Asi ^ cr¦{ āj æazī[}Át Ás@ Ásiāa] ĭc^Át @eet|Ási^ÁajædÁ æ) å Ásiāj å āj \* Át]}Áset At Asec Æt

Ct≣|Áŝaã] č c^Á^•[|č cā[} Ásc&caçãaã∿• Á; ã||Ás^Ása]] ¦[] ¦ãæer∧|^ Á^&[ ¦å^åĔÁ

# , " = bgdYVMjcbgža cb]hcf]b[ 'UbX'Ui X]hjb[

#### , '%Á 9bj ]fcba YbhJ` ]bgdYVMjcbg UbX a cb]hcf]b[ `

Ò}çā[}{ ^} cæl/āj•]^&cāţ}•/Ásej å/á, [}ãt[¦āj\*Á, āļļÁsh^Á'}å^\cæl^} Át[Áşcælāãæe^Aseg/Ást]] æser Á; !^å äsec^å Á -[¦Áseg/ÁL][b%cdÉxt[Át, ^æe`|^Áseg/Å]^\*••Á; -Á^;çã[]{ ^} cæl/Ást[]cal/Åt] ^ cæsta[]At ^A co@a ÁOT ÙÉxej å/át[Áseå å\^•• Áse]] ![çæl/Å^` ă^{ ^} o ÉÁL@ • āseal/Å} çã[] { ^} cæl/át[] ãt[ ¦āj\*Á, '![ & ^å `!^• Á æ^/Ås^cæat/å/ásta Ásec Án |^çæ) cAOT Ù/Á` à Ëj |æ) • ÉÁ, ão@ ÁseA` `{ { æ^At - Áseg/At [] ãt[ ¦āj\*Á, '` ã^{ ^} o ÉÁL å^cæat/å/ásta Ásec Án |^çæ] / [] ãt[ ¦āj\*ÁL][ \* [æ] Ást ÁsEA

V@ÁÛ čæ¦^ÁTæ) æť^¦Á, āļļÁa^Áæáçãr^åÁ; -Áæ) ^Á,[}Ë8[}-{;{æ) &^•Á'[{ Á; [}ãī[¦ā]\*Áv) å Áå^cæā‡•Á, āļļÁ à^Áa[&č { ^} c\*åÈÁy @;!^Áæá,[}Ë8[}-{;{ æ) &^Æa Áa^c&eáA; !Á; [}ãī[¦ā]\*Á^^\* [e Áæ^A, []čæ] Åá^ÁœA ^c]^&c\*åÁæ)\*^Áæ) å Áæ⁄Aáa a Aæc/Áæidãa čææi [^Á4; Áw@ÁÚ![b &cAg2È È&æ/Aaj +ĭ^} &^aáAá Áæ&d; !•Á}å^!Á c@Áåã^&cÆ{[}d[|Á;-Ás@ÁÚ![b &cA2È ÈA;[ã\*A4;[{ Á×ĩã]{ ^}dDÉx@Á;![&^••Áa^a & &aAa; ÁD^&cā]} ÈHÁ ;ā]Áa^Áa[]|^{ ^} c\*åÈÁUc]•Áa; Ác@Á;![&^••Á;ā]Ác?]ā&æa]^Áa; &]čaAá

- OE; Áse) æf ã Á; Ás@ Á^• č | o Ás; Á; [ ¦^ Ás^ cæsiÁ; Ása^} cæŝ Á; [ • ãa |^ Ásæč ^ Á; ¦ Ás@ Á; [ } Ës[ } -{ ; { æ} s^ Á
- OEÁãe^Á§;•]^&cāj} Á
- Þ[ cā-ˆ ð] \* Á^|^çæ) ơ∮ ^¦•[ } } ^|Á , Á@ /á •`^Á
- Gâ^} cã^ ā,\* Ása) å Áset |^^ā,\* Át, } Áse8 cāt, } Át; Á^•[ |ç^ Át, | Át, ãcāt æe^ Ás@^ Át, [ } Ё8[ } -{ | { ab; 8^ A
- Q, ] |^{ ^} cāj \* Áse8cāj } Ág Á^ &cã~ Áj ¦ Á, ãcã æe^ Ás@^ Áj [ } Ë8[ } -{ ¦{ a} 8^ A

V@ Áxā[ā]\*Á{[¦Áxe)^Áxē[]¦[ç^{ ^} ơý,ā||Ása^Áxee'¦^^å/Asa^c,^^} Áx@ Á^|^çæa)ơ∱,æalca?•Áyāzè ÈÉAÛčæal¦^Á Tæ)æ\*^¦ÉÁ^•ãa^}dÉxeč o@;¦ãuā?•DÁsæe^å/q{}Áo@ Á/^ç^|Áj-Áxā\ÁçèÈÈxeaAîā\*}ãa3&æa)ơÁā:\ý,ā||Á^čĭā^Á ā[{ ^åãaxez^Áxa8ca[i]}DĚÁ

#### , "&Á 9bj ]fcba YbHJ`Ui X]h]b[ \*

Ùãr∿kheĕåãāā}\*/kārko@/kā∿•oA,æĉkā[Á,^æe\*¦^Ár}çã][}{ ^}cæd,^!-[¦{æ}}&^ÊA,^çã},Å[]^¦æeā}\*Á ^~^&&aãç^}^••Á,-Ár}çã]]{ { ^}cæd,Á':[cr&cā]}Á, ^æ\*'!^•Áæ}åAc@/kā^•oA,æĉAg Áac&@?ç^Áræeār-æ&d;!^Á ^}çã]]{ ^}cæd,č&{{ ^•Áx@[`\*@Áx{]}cā}`æd,Á[]![ç^{ ^}cā)č

#### , ''&'% =bhYfbU``Ui X]h`

Quơ\}æ¢keĕåãā}\*Á,ā|Aba^Á}å^\cæa^}Aba^Ac@AÛĭæ4\^ÁTæ)æ4\^ÁTæ)æ4\AQ;\Aba^|^\*æerDE#\^}^\æ|^Á,ãc@3,Ác@Á -ā•oÁācÁ{[}c@A{,-A{]^\æaa]}Aba}åAs@}A{}AbaAç,^|ç^Á{[}c@?Abæ•ãEA/@A{+^ĭ^}&&A{,Ab@Ab}& æčåãxA&[ĭ|åAba^ÁA|æac^åE5bA{[/Æ\*•ĭ^•Abd^AbbA}cãa?åAbJÁc@Áā•oÁ^, Ábĕåã•EA/@A{,ĭ\][•^A{,AA`&@A æčåãā3\*Á,ā|Aba^Á4[Áç^\ã^Ab{[{]]ãa}&^A,ác@Ab

- V@ā ÁÖT ÙÁa) åÁÜ àËÚ |a) Á
- OĘ]¦[çæ‡Á^˘˘ã^{ ^}œÁ
- OE; ^Á^|^çæ); of{^\* æ¢Áæ); å/(to@'; Á^``ā^{ ^} o Áç; È ÈÓÚŠÊÁ, ^; { ão ÉÁ^\*` |ææā; } ÊÁ&[ } d æ&oÁ å[ & { ^} cææā; } DÁ

#### , "&"& 91 HYfbU`Ui X]h

Q Áæ&&[¦åæ) &^Á ãu@k@ ÁÖ^ç^|[] { ^} xố[}•^} dÉæ) Á§ å^] ^} á^} xá} á^} xá} zá] &æ#¦æ} ák; č xá, ãu@g) Áç [ ÁÇEDÁ ^æ+ Á; Ak@ Á æ#o4; -Á `æ+l^ ā; \*Á;] ^¦ææā; }•Á; } Ë ãr É&æ) å Árç^¦^ Áãç^ ÁÇ DÁ ^^æ+ Áx@ ¦^æer'ŀÈÆr[ |&ā; Á; ā||Á; [ { ā; ær Áx@ Á; i=] [ •^ å Áæč å ãú; ¦Á[ ¦Áx@ Á§ ãiãæ+Áæč å ãdĂ; ãu@3; Árì Á; [ } c@ Á [ -Áx@ Á\*¦æ) cāj \* Á; -Á&[ }•^} dÉV @ ÁU^ &¦^æ+ Á; -Áx@ ÁÖÚOBÒÁ; ā||Áæ]; ] ¦[ ç^ Áx@ Áæ]; ] [ ā; c( ^} xá, Áx@ Á ] ¦[ ] [ •^ å Áæč å ãxÁræ; Æ Áş Áæ&&[ ¦åæ; & A, ãu@AÔ[ } å ãiã; } ÇæDÁ; -ÁU&@ å ` |^Â; Á; -Áx@ ÁÖ^ç^ |[ ] { ^} xÁ Ô[ }•^} xó, !ā; !Áç Áx@ Á&[ { { ã; •ã; }ā; \*Á; -Áx@ Áæč å ãiEÁ

V@āA,¦[&∧å`¦^Á,ā||A,&&`¦Á[¦Áee||Á`à•^``^}o^A}çā[]{{ ^}cea|Aĕĕåão•A5jÁee&&[¦åæ)&^Á,ãc@As@eA &[}åãaā]}ÈŘÖcæ&@ÁeĕåãaÁ@ee||A5j&|`å^Ác@·Á[||[,ā]\*Ásæeã&A\*|^{{ ^}o•A5jÁee&&[¦åæ)&^Á,ãc@A©)UÁ FJ€FFKG€FÌKÁ

Q20XÔ}d^ÁT^^caj\*Á.Á/@∘Á,`dāj^Á,-Á;@ ÁsĕåãaÁ,àb/&cãç^∙Á,ā∥Ásà^Ásãã&`••^åÁ,ãc@ks@ Á&[{]as}^Á {a3}æ\*^{^}oké^æ{EĂ

ÇāDÄÖ [&`{ ^} cÁÜ^çā', Á ÁOEļÁ`æs!^Ási [&`{ ^} cæeāj} \*Áţ Á^ |æeā] \*Áţ Áţ [}ãt[!ā] \*É∮, ![&^a` ¦^•É∮a&^} &^•ÉÁ &[} •^} o\*Ásej åÁţ c@ ¦Á'}çãt[]{ ^} cæþÁsi [&`{ ^} cæeāj} /Ási Ásee •^••^å/kset æðj • cÁst[{]|ãte} &^ /ksi ác\* ¦ãteÉÅsEÈÁ &[}åãtāj} •Áţ ~Á@ ÁÖ^ç^|[]{ ^} cÁÔ[} •^} dÊÔ} çãt[]{ ^} cæþÁú![c\*&cāj] /ÁSã&^} &^ /ksej åÁţ ãtë æsāj} Á &[ { ãt( ^} orÁj !/••^} c\*åÅ ãt@j Ác@ ÁÔOEEÁ

ÇāaDÁÙãc^ÁQv•]^&cāį}ÁÁJà•^¦çæaāį}•Áį.-Á\*ãc^Áj.^¦-{¦{ æ);&c^ésēj;•ó%s[}•^}oésēj;\*ó%s[}\*^}oésēj;\*óés

Çaç DÁÖ ¢ãaÁT ^^caj \* Á Á/@^Á,¦aj &aj æ aþÁ, à∙^¦çæ aðj } • Áæ b^Á^ça∿, ^å ÈÁ

Q Áse&&[¦åæ) &^Á, ãc@Ó[}åãã[}ÂiÁ[×ÁU&@åĭ|^ÁiÁ[×Ás@ÁÖ^ç^|[]{^}o^}dÓ[}•^}dÉs@ÁsĕåãA{``•dA

æĐà^Á&[}å`&c^åÁ\$a^Áxāæà|^Á`ææà|^Á`æ‡ãã?åÊ4x¢]^¦ã?}&^åÊ5æ}åÁ\$jå^]^}å^}o∱j^!•[}ÇDÁ,@ æ]][ājq(^}o^}a4s]^![ç^åÁ\$a^Áx@ÁÙ^&¦^cæ+ÊĂ

àDQ\&|`å^Á&[}•`|cæaā[}Á,ãc@kb@^Á^|^çæ);o/&et^}&ã^•È

&DOE•^••Áx@Á\*}çã[}{ ^}ca¢Á,^!-{!{ a) &^Á, ~k@Á,![b^&dÉaa) åÁãorÁ~~^&orÁ,}Áx@Á`!![`}åã;\* ^}çã[}{ ^}dĚ

åDOE•^••Á,@c@:¦Á;@A;¦[b^&c/áarÁ&[{]|^āj\*Á,ãc@A;@A^|^çaa);cA;caa);åae¦å•É4j,^¦-{;{aa);&^A; ^ae`;^• aa);åÁ;caaečd[;^Á/~`šā/{^}orĚ4

^DÜ^ç矞`, Ás@ Ásaå^˘˘æ&î Á(-Ása) ^ Áid ææ^\* ^ Ð) |æ) Ð) ¦[\*¦æ( Á^˘˘ā^åÁ´}å^¦Ás@ ÁÖ^ç^|[] { ^} c Ô[}•^} dĚÁ

-DÜ^&[{ { ^}åʎ; ^æ\*`¦^•ʎ; /ᡬᡂáā; }•Á; /ấų] ![ç^/á@/Á\*}çã[}{ ^}œ4⁄; ^¦-{ !{ æ} &^/á; /íœ/Á; ![b\*&ê æ)å⊕; /Ádæc\*\*^尹)æ; 尹):[\*:æ; Á^´`ã^åÁ / å^', Á@ ÁÖ^ç^|[] { ^} A^Ô[}•^} dĚÁ

\*DÓ^Á&[}å`&c^åÁæ)åÁ^][¦c^åÁ{[Ác@^Á?ææã?~æ&cā[}}Á{!~Ác@^ÁU^&¦^cæ^^È

Q Ázesze[¦åæ) &^Á, ãc@ÁÔ[}åãā]}ÁľA[, ÁÙ&@ å`|^Á[A[, ÁQ&AÔ^ç^|[]{ ^}ơÅÔ[}•^}ơÅ?[|&a] Á @eee|Á\*`à{ ãrÁ æ4&[]^Á[, Ác@ Ázeš åãdÁ^][¦ơÁ[Ác@ ÁÙ^&¦^cæ+ˆÁze) åÁ^|^çæ) ơÁze\*^}&ã> 4Å åc@^^Á[]{o@ Á[, Ác@ Á &[{]|^cā]}Á[, Ác@ Ázeš åãdĚODÁ^•][}•^Á[Aze)^Á[, Ác@ Á^&[{ { ^} åzesā]}•Á3JAc@ Ázeš åãoÁ^][¦ơÁ @eee|Áze+[Á à^Á\*`à{ ãcc\*àDÉÁ

Yãc@3),Á{}^ÁÇFDÁ{[}c@4{xÁc@Á&[{]|^ca[}Á;Áæ9,Á§;å^]^}å^}cÁ}çã[[}{^}ca4keĕåãaÁÇ3;Áæ&&[¦åæ9,&^Á jãc@4Ô[}åãa[}Ár∈A[x-AÙ&@å`|^AíA[xÁc@ÁÖ^ç^|[]{^}cÁ]}cÁ0[}•^}cDÉP[|&ã[Á]a[A[x];[çãa^Áæ&&[]^Á[xÁc@Á ¦^][¦cÁ[ÁÔ[`}&ã4kæ}åÁc@Áæà[ç^{ ^}cã]}^åÁ^|^çæ9;ckæ\*^}&ã•ÈEP[|&ã[Á]ā[k憕[Á\*}•`¦^Ás@æxÁæÁ &[]^Á[xÁc@Á^][¦c5a;Á{xæå^Á;`à|ã&|^Áæçæãjæà|^Á[;]ë=ãcA[¦Á[;]]ā^ÈA

#### , "Á Bcb!WebZcfa UbWYgžWeffYWYjjYUbX'dfYjYbHUhjjY'UVWjcbg'

- V@ ÁÖÚŒBÒÁæ) åÁs@ ÁÒÚŒÁ, āļ/ఓ^Á; [cāāð åÁ, @ \^ÁæA, [c\*} cāædÁ\*}çā[}{ ^} cædÁ@æd{ (Åç^} c/æ Á &æč • ^åÈÀÙ & @Áç^} or Áæd^Áæde [Á^] [ \c\* å/āj Ác@ ÁŒÈT ÜÊ£å, cæaājā) \* Áæj ^ Á, [ } Ë&[ { ] |ãæ) & ^Áæj åÁ c@ Áţ ãcāt ææāj } Áţ ^æe ` \^• Á } å^\cæd ^} ÊÆj & {` åāj \* Ás@ Áæc^• c/áţ [ } ãu[ \a] \* Á^• ` |or Áu[ Áå^] ã&cás@æcÁ c@ Áţ ãcāt ææāj } Áa^āj \* Á } å^\cæd ^} Ád < ^åãæeāj \* Ás@ Áæc • č/át [ } ãu[ \a] \* Á^• ` |or Áu[ Áå^] ã&cás@æcÁ</p>
- OE Á, ^ ¦ Á@ Á&[ } åãāţ } Áţ Á Á@ ÁÒÚŠÉÉB Á Á@ Á ç^} óĄ ÁzóA, [ } Ë&[ { ] |ãe} & A ÁB & ãa ^ } ó& zeč ð \* Áţ ¦ Á o@^zec^} ð \* Áţ zec^¦ ãe Á@ek{ Át[ Ás@ Á\*} ç ð [] { ^ } dÉBÈ ÉÅJ [ || č dţ } ÁB & ãa ^ } o É xeč AÒÚCEÁeg à Đ ¦ ÁÖÚCÁ Y zec^¦ ÁB Át[ Áb ^ Á, [ cãa? à ÁB[ { ^ à ãa zec^| ~ È Y | ã cc^} Åb ^ cza‡ • Áţ Á xe@ Á, [ cãa8zezaţ } Ás A Át[ Áb ^ Á, l [ çãa ^ à Át[ Á OÙCEÁeg à Đ ¦ ÁÖÚCÁ zec^¦ Á ã coga Á Áb zeĉ • Á'[ { Ác@ Áb zec^ A; } Á @B&@Ás@ ÁB & ãa ^ } of & & ` ' | ^ à ÉA

#### , "(Á FYdcfh]b[ UbX fYWcfXg

Ò}çā[}{ ^}œaḥ^^][¦cāj\*Á^´`ã^{ ^}œÁ[¦Áĩac^Á[]^¦ææā[}●Áseb^Á]¦ã[æeāîÁsb^¦ãç^åÁ\;[{Ás@A Ö^ç^|[]{ ^}œÁÔ[}●^^}œÁsejäÁse]|a3eæai|^Áa3e^}●^●Á[¦Á]^¦{ãerÈAO[¦Áse||Á^][¦cāj\*Á^č`ã^åÊ5s@AÛčæs¦^Á Tæjæ\*^¦ÁsiÁ^●][}●ãa|^Á{¦Á;æjæ\*ä]\*Ás@Á;¦^]æseæãa[}ÁsejäÁ\*ča{ã•ã[}Á;∞Á∞Á^][¦dĚA

V^]^Á	Ø^~~^} & Á	Ü^&ājā*}oÁ
O5;}迆ÁÜ^č¦}Á	OĘ}čæ∥^Á	ÒÚŒÁ
OĘ}}ĭæļÁÜ^çāt∖ġÁ	O5;}čæ≬^Á	O≌^}&að∙Áãac^åÁ5jÁù/^&ca‡}Á ÌÈÈÉÁ
Ò}çã[}{ ^}cæ¢Á[[}ãã[¦ã]*Á^•č o•Á [}Áso@·Á∫^à•ãc^Á	Û`æơ¦ſÁ	Ú"à∣ <b>&amp;</b> Á
Þ[cãa&æcaā[}Á[-Á][  čaā[}Á9]&äā^}oÁ ,@¦^Á[æe*¦ãæd-Á@æd{Áv[Áo@-Á ^}çãi[}{ ^}o≸arÁsæĕ•^åA[¦Á o@u^æe*}^åÁ	Yão@ajÁGIÁ@(,°¦•Á(,-Áaa)Á āj&aãa^}oÁ	ÒÚ00£ÄÖÚ08òÁa)åA(ico@¦Á ¦^ ^çaa)oÁaet*^}&&?∙Á
Ø´  Á,¦ãac^}Á^][¦OÁ,ÁacA,[  čaā,}Á 3)&ãã^}OÁ,ão@Á,[c^}cāadaÁ,¦Áac&ciadaÁ [⊶ãc^Áã,]æ&ociÁ	Yão@a)AìAsiæê∙A(,-A),[cã-ã)*Á c@~Á5j&ãã-^}oÁ	ÒÚCHÉÄÖÚGBÒÁa)åA(io@∘¦Á ¦^ ^çaa)oÁaet^}&&∿Á

#### HUV Y', !% 9bj ]fcba YbHJ fYdcfh]b[ '

V@^Á&[}c^}o4{;~Ás@^ÁOE;}ča¢ÄÜ^č¦}Áã;Áse•Á`]^&ããð\*åÁ§;Ás@^ÁÒÚŠĚÁ

Ü^][¦cāj\*Á^˘˘ã^{ ^}orÁşiÁ^|æeāj}}Áq[Ásj&ãå^}orÁsel^Ájčdaj^åÁsjÁÛ^&caj}}ÂiÈÁ

#### , "('% 5bbi U`Ybj ]fcba YbhU`a UbU[ Ya YbhfYdcfh

- ÖÚŒBÒÁ
- •Á ÒÚŒÁ
- ●Á ÖÚQÁYæe^\¦Á
- •Á ÜT ÙÁ
- •Á V,^^åÁÙ@ã^ÁÔ[`}&ã¦ÁÁ
- •Á V@^Á^] ¦^•^} œç^•Á; } Ás@ ÁÔÔÔÁ

V@•Á^][¦ơÁ(`∙ơkÁ

- Ö^•&¦âà^Á@^Á,[¦\Á&æ;¦ð\åÅ(`óAşiÁs@A/æeoAFGA([}c@A
- Ö^•&¦ãa^Áx@Ą[¦\•Áx@æ¢Ąā]|Áà^Á&æe¦ã^åÁį`óAşÁx@Ą^¢oÆCÁį[}c@Á
- Q&jǎ^ÁæÁ`{{ æ^Â, Á;œÁ; [}ãt[¦ā]\*Á^•`|œÁt[¦Á;@A;![b^&oAsi`¦ā]\*Á;@A;æ oA^æA
- Q84, \* å^Áæ) Áæ) æf ã Á, Áœ ^ Á; [} ãt ¦ã \* Á^• \* |œ Áæ\* æð Ó \$c@ Á^|^çæ) d Á
  - . Q.] æ\$oÁæ•^••{ ^} oÁ&¦ãe^¦ãæ∰ãįã•
  - . T[}ãt[¦ā]\*Á^•č|o•Á¦[{Áj¦^çã[č•Á^æ•
  - . Ú¦^åã&cāį} Á§iÁs@Áå[&`{ ^} œÁãa c^åÁ§iÁÔ[}åããāį}ÁGÁį ~ÂÙ&@å` |^ÁGÁį ~ÁS@ÁÖ^ç^|[]{ ^} c Ô[} •^} c
- Cā^}cã^ Ázð^ Át^} å•Á§ Áz@A, [}ãť; ¦ã \*Á^•`|o•Á;ç^¦Áz@Aã^Á; Áz@A, ¦[b^&cA
- (â^} cā^ Ác)^ Á,[} Ё8,[{]|ãe) &^ Ás`¦ā) \* Ác@^ Á,\^çā,`• Á^æA
- Ö^•&¦ãa^Á, @eeeÁee&cāį}•Á,^!^ÉĄ́,!Áse^Áà^āj\*Ékeèa^}Å{(A^}•`!^Á&[{]|ãe})&^Á

Y ão @aj Át, }^ÁÇFDÁt, [} co@át, Ás@ Á&[{]|^cat, }Át, Ása) ÁOEÒT ÜÁÇaj Ásce88[¦åæ) 88^Á, ão @ÁÔ[}åãata, }ÁF€Át, Á Ù&@ å`|^Át, Át, Ás@ ÁÖ^ç^|[]{ ^} cAÔ[}•^} cDÊÆP[|8at, Á, ä|Aj, ![çãå^Ásce48[]^ Át, Ás@ Á^][¦cAt, Ás@ ÁÔ[`}8atjÁ æ) åÁs@ Ásceai[ç^{ ^} cat, }^ cat, } 8at•ÈÆP[|8at, Á, ä|Ásc+[Á?}•`!^Ás@exAsce48[]^ Át, Ás@ Á^][¦cAt, ás@ Á ]`à|a&l^Ásaçasatjacaai|^Át, }Ëē ãs Ása) åÁt, }Ác@ Á&[{]æ}^Á, ^a •ãs ÈÁ

#### ,"("& FYWcfXg

Ò}çã[}{ ^} cæþÁ^&[¦å●Ásch^Á[æaē] cæaē] ^åÁs@eecÁsk^{{ [}●d æecÁ&[{ ]|ãee) &^Á, ãc@Ás@ ÁÒT ÙÆj ÁscÁ å^åã8æec^åÁājā] \*Á^●c^{ ÁzÁsiã ãezeþÁ[&æezā]}ĚÜ/&8[¦å●Ág[Á[æaē] æaē] æaēj cæaēj Ás]&|ĭå^KÁ

- V@ā ÁÒT ÙÁa) åÁae•[&ãaae^åÁ`àË;|æ)•Á
- OE[|Á&&^} ^ Áse) å Áj ^¦{ ão Ásh^caeáj^ å Ásj ÁÙ^ &caj } Á+È HÁ
- Qå šæqī} Áse) å Áslæqājāj \* Á^\* ãr c^¦ Á
- Ò}çã[}{ ^}œaþÁ§•]^&cãį}Áœ)åÁį[}ã[¦ã)\*Á^][¦orÁÓÁ^•`|orÁ

- •Á Q,8ãå^}ơÁ^][¦o•Áæ)åÁ^\*ã:ơ\¦Á
- •Á Ô[{]|ænĝ:o•Á^\*ã:o^¦Á
- •Á Yæ c^Áslæ) ~^¦Á^&^] o Áse) åÁ æ c^Á([ç^{ ^} A^\*ã c^¦Á
- •Á OĘĮÁ^\* č |æĘ[ ¦ Â&[ ¦ !^•] [ } å^} & A

O∰Á^&[¦å•Á^˘˘ã^å/å^ˆÁ@^ÁÒT ÙÁ&^KÁ

- Tæājcæāj^å/\$jÁæá/{^\*ãa|^Á[;¦{Á
- S^] cÁ[¦Áæá/\æ•ó/\Á^æ•ó/\Á^æ•Á
- Ú¦[å`&^åÁţÁæ}^Áæčc@;¦ã^åÁţ~æX^¦ÁţÁœAÖÚOB;ÒÁţ¦ÁÒÚOEÁ][}Á^``^•cÁ

# - " F Yj ]Yk <sup>°</sup>UbX<sup>°</sup>]a dfcj Ya Ybh

#### - '%Á 9A G`fYj ]Yk '

V@ ÁÛ ǎæ ¦^ ÁT æ); æ\* ^ ¦ Á; ā| Á^çā\*, Ás@ ÁÒT Ù Áæ); å Áão• Áã; ] |^{ { ^} cæaā; } Á; ãc@a; Á+Á; [ } c@; Á; Á

- OĘÁ§48ãå^}ơÁ^][¦ơÁ
- OĘÁOĘ ) čą4Ü^çã (Á
- OĘ Ásě åãoÁ^] [ ¦oÁ
- OB;^Á;[åãã&ææā;}•Á;[Ác@^Áæ];]¦[çæ‡Á

Ó^ç ^^}Áv@Á&@å`|^åÁ^çā`,•Éaxá^\*ã~c\/á,~⁄ã••`^•Á,ā|lÁa^Á;æaj;æaj;^åÁ;[Á\*}•`|^Áv@æx4x)^Áã•`^Á ¦æā^å/à^Á§;~{};}æ4xyå/Å;e;}}#4/,^!•[}}^|Áæ••[&ãæx\*åÅ;ã@Áv@Á`æ¦^Á®iÁ^&[¦å^åÈÁ

V@Á^çã^, Á, āļÁ&[}•ãa^¦ÁÇ, @\¦^Áæçæājæà|^Á,¦Áæj]]a&æà|^DxÁ

- Ô@ee)\*^•Át[Át@eA[[]^¦æea[[}}Á
- ●Á Ùãc^Á,^¦•[}}^|Á&[{{^}œÁ
- ●Á CE\*^}&`Á&[{{^}œA
- ●Á CEĽåãaÁãjåãj\*●Á
- •Á Ò}çã[}{ ^}cæ‡Á{ [}ãã[¦ậ]\*Á^&[¦å•Á
- ●Á Ô[{]|æn∄]o•Á
- Á Ö^ cæāj• Á ( Áse) : ( | | ^ & cãç^ Áse) : à Á | ^ ç^ } cæãç^ Áse8 cã; } Áse8 ^ } Á
- •Á Ò}çã[}{ ^}cæ\$4^{[}Ë8[}-{;{æ}&^•Ê4^}çã[}}^{ ^}cæ\$4^§•]^&cā{i}}^{ ^}cā&^\*E8\_j•]^&cā{i}}A^{[}cã&^•É8\_j•]^&cā{i}}A^{[}cã&^\*E8\_j•]^&cā{i}}A^{[}cã&^\*E8\_j•]^&ca{i}}A^{[}cã&^\*E8\_j•]^&ca{i}}A^{[}cã&^\*E8\_j•]^&ca{i}}A^{[}cã&^\*E8\_j•]^&ca{i}}A^{[}cã&^\*E8\_j•]^&ca{i}}A^{[}cã&^\*E8\_j•]^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{[}ci&^\*C8\_j\bullet^&ca{i}}A^{
- ●Á Q2&ãã^}ơÁ^][¦œÁ
- •Á Ô@æ)\*^•Á§jÁ(¦\*æ)æ7ææ§j}Ád`&č¦^•Áæ)åÁ^•][}•ãaðjãæ?•Á
- •Á V@:Á\*¢c\*}cA[{]|ãæ)&^Á,ão@4,àb\*&ãç^•Áæ)åÁæe\*^o•Á
- Á V@ Á ~ ^ & cá + & @ + ^ Á j Á ca) å a a a a a A \* ã | a a j } Á
- •Á Ô[Ë;¦åājæaāj}Án;Án;}çā[]{ ^}cæk/(æ);æ\*^{ ^}o/(Án;Åi`àË&[}dæ&d;!•Á

ÖÚOBÒÁ( ˘∙oÁ,[cāa?åÁsjÁ,¦ãaa)\*Á(-Áaa)^Á\*&@Á^ça?, Áa^a)\*Á'}å^¦caa\^}ÈÁY@ ¦^Ác@ Á^ça?, Á^aaá•Át[Á ¦^çara[}•Êác@}}Á;ãc@a)Á.Á,^^\•Á(-Ác@ Á^ça?, Ác@ Á^ça?^å/å[&`{ ^}oá(`\*oóáa^Á\*`à{ãcc^åÁt[¦Ác@ Á aa]]¦[çaaþÁ(-Ác@ ÁÙ^&l^caa)^ÈÁ

#### - "&Á 9A G<sup>°</sup>i dXUhYg<sup>°</sup>

V@Á,č c&[{ ^•Á, -Ác@ Áseà[ç^Á^çã, •Á, æ Áş &]č å^Áse{ ^}å{ ^}c Á[ Ác@a ÁÔT Ù/Áse}åÁ`]][¦cā]\*Á å[&č { ^}cæaā]}ÊĂ]åæe^•Á[Ác@ÁÚ|[b/&c/Áse]^&c Áse}å/Åā[]æ&orÁ^\*ã c/¦ÊÅ^Ë·çæţ æaā]}Á[ -Ác@ÁÚ|[b/&c/Áse [àb/&cãç^•Áse}å/Ásea\* ^orÊā, ¦Á^æa][&æaā]}Á[-ÁÚ|[b/&c/Á\*•[č ¦&^•ÈČe/Áse]^Áse{ ^}å{ ^}orÁse^Ase[}\*ãa^\!^åÁ ] ^&^\*•æ^Ê&c@•^Á, ^^å/A[ Ása^Ase]] ¦[ç^å/ás^ÁÖÚOBÒÈĂ

CB;^Áæ{ ^}å{ ^}œÁ( æå^Át[ÁÚ¦[b%&oÁ;^•c^{ •Á¦¦Áj¦[&^••^•Á]Åå^Á&[{ { `}}ä&æe^åAt[Á^|^çæ})oÁ ]^¦•[}}^\Áœ Á^``ã^àÁt[||[, ā]\*Á&[{ ]|^cā[}Á[×á@^Á^çã\,ÈÁ



; <8 Á2Ü^][¦ơÁ[¦Á?[|&ã; ÁQÍCE•dæ)ãæÁDÁÚc ÁŠcå ÁÉZÖ`}|[^ÁÛæ)åÁÛĭæ¦^ÉÁOGG€€ÍÎÁ
# 5ddYbX]I <sup>·</sup>5Á ÁÒ} çã[ } { ^} œ‡Á [ |ã& Á



# 9bj]fcba YbłU`Dc`]Wii

#### P[|&ã; ÁÇCE•dæ‡ãæÐÁÚc ÁŠcåÉÆË ED€F€Á

P[|&ãi ÁOE • dæjānækå^|ā²ç^• Ás@æeýi|[ c\*&cāj \* Ás@ Á\*}çāi[} { ^} ó¥i Áāj c\* ¦æ‡Ái Á\* • cæāj æài/Á å^ç^|[] { ^} dĚ V@ Á; !āj &āj |^• Á; -Á\* • cæāj æài/Á&^ç^|[] { ^} oÁ Áçæ‡ ^ Á&i ^ æaāj ?Ê\* • cæāj æài/Á ^} çāi[} { ^} cæ‡Á,^!-i-! { æj &^ Ásġ å Á&[ !] [ !æe^ Á [ &ãa‡Á\* • caāj ãà i]^ Á Åsi ^ e cæāj æài/Á OE • dæ‡äæaţ Ás` • ā ^•• Á dæe^\* \* ĚÁV@ !^ Ásé ^ Á [ ` ! Á ājæi• Á; -Á; ` ! ÁÒ} çãi[} { ^} cæ‡Á [ & A cæ‡Á [ ! Á @B&@ A ^ Á@æç^ Áse• • ã } ^ å Ásæ3cāj } • Át Áse• ã cát, ` ! Á i] æi • Á ( , æå • Á\* • cæāj æài/Á å^ç^|[] { ^} dĚÁ

#### %"AUbU[Ya YbhiGmghYa g`

˝Ô[{]|^Á,ão@ÁzdeļÁzde]]|&Baanai|^Át}çãi[}{ ^}cadeļÁaç;•ÊÁt^\*`|aazati}•ÊÁs[å^•Át,-Át¦aa8ua8t^Ásde)åÁ ç[|`}cade^Ásdet¦^^{{ }}@L Á

˝OBå[]oksa)åÁv}-[¦&^Á9jcv¦}æqkÁ(cæa)åæslå●Ása)åÁv^●cv{●Áv@æseÁv}●č¦^Ása[}cē)č[`●Á ^}çā[]{ ^}cæakÆqi]¦[ç^{ ^}ddÁ\ Á

‴Ù^ơ&[¦][¦æ¢/Ąiàb∿&ãą;^∙Áse)åÁse\*\*^o•Áse)åÁ;[}ãq[¦Áj¦[\*¦^••LÁ

Á ˝Ú¦[{[cʰÁʰ}çã[]{ ^}cæ‡Á&[{ ã¤c{ ^}cÁc@[ǐ\*@Ádæājā]\*Áæ)åÁājc^\*¦æaāj}}Áājd[Áàĭ•āj^••Á ]¦[&^••^•EĂ

#### &"FYgcifW/Thj]gUhjcb

‴Ò}•`¦^Á;]cā; `{Á•^Á;-Áæ;Á;ææ^¦ãæ†+Éãs;&|`åã;\*Á\*}^¦\*^Á\*~a&a?}&îɧ;ææ^¦Á\*~a&a?}&îÁ c@Á^å`&cā;}Á;-Á;æ•c^Ás;Áæa‡Á;]^¦ææā;}•LÁÁ Á

˝Ö^ç^|[]{^}ơ[ҳ-ʎᢒ₃}[çææãç^ʎsə)åʎ⁴ੱ∙œæãjæà|^ʎj.¦[åĭ&or ʎsə)åʎj.¦[&^••^•ÈĂ

#### Á '"9bj]fcbaYbhU`UbX`GcW]U`=adUWfg`

″U]^¦æe^Áæe Áæ) Á?}çã[}{ ^}cæ|^Áæ) åÁ[8ãæ4|^Á^•][}•ãà|^Áà`•ã}^••LÁÁ Á

´´Tậā[ã=^Ác@-Á\*}çã[}{ ^}cæ‡Á\$[]æ&o•Á[-Á[č|A[]^¦ææā[}●LÁÁ Á

˝Ü^@eeaàājāaæe^ÁjeeayaåÁs@eee/5ae Ášą[]æs8ec^åÁsà^Áţĭ¦Áses8eaãçãaã?●Ásj8aj\*Ásãà,\*Ásã^}caã-ãj\*Áş[][¦c`}ãaã?●Ás[Á ];[c^&eo/ásajåÁ?}@eej&?Ásč;|c`¦æojÁ\*ãe?●ÁsejåÁsājååãç^¦●ãc`LÁÁ

A

Á

ŰÜ^•][}åÁ§[Ás@eA&@ed|^}\*^•Á;¦^•^}c^åÁsˆÁ&|ã[æe∿Á&@ea)\*^ÁsàˆÁså^}cã÷ã;\*Á;]][¦č}ãa?•Á§[Á ¦^å`&^Á;`¦Ásæsià[}Á[[d]¦ã;dÈÁ

## A

**("GłU\_1\ c`XYf`FY`Uh]cbg`** ˝Ò~^&cãç^|^Á\}\*æt\*^Áæ)åÁ&[{{`}}ã&ææ^Á,ão@Árcæ\^@2|å^¦•Á§JÁ^|æaã[}Á§IÁ\;çã[]{ ^}cæþÁ {ææ^\•È

Tæl\ÁÖæ{]à^||Á ÔÒUÉAP[|&ã[Á0CE∙dæ)ãaĐÁÚcŠcåÁ

# 5ddYbX]I `6Á ÁJã^Á |æ)•Á



 

 REVISED SITE + LOCALITY PLAN
 CLIENT: RAMTECH PTY LTD REF: 0535 DATE: AUGUST 2007
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535
 Image: Client: Ramtech PTY LTD Ref: 0535













 Remove hornern rake perimeter bund
 Remove topsoil and create southern lake perimeter bund

140 Robins Town Centre Drive, Robins, Old. 4230 Phone 55789944 Mobile 0418 750919 Fax 55789945

12. Rivenestk One

DECKE

DATE 2810/0

GJ0554.1.6



FIGURED DMENSIONS TO BE READ IN PREFERENCE DATE 301000 DECKED

GJ0554.1.7









# 5ddYbX]I `7ÁÁŒ^}&^Á&[\\^•][}å^}&^Á

5ddYbX]I `8Á Á)[ ¾ & å Á æ \* ¦ Á æ \* \* { ^} o Á |æ Á

Á Á



# <c`Vja `f5i ghfU`]UL Dhm@hX`

Ä^¦a, ÅÛæ)åÁÛ æ¦^/ ]] { Ö`} [[^ÂÛæ) åÁÛ æ¦^/ Ä (æ)ÅÝ æ?\ÅT æ)æ \*A {^} ÅÚ|æ) Å

U&a{à^¦*Á*G€G€Á

Á

V@#Á^][¦dxÁ@ze-Áa^^}Á,¦^]æ\^åÁa^ÁÕPÖÁ[¦Á?P[|&ã[ÁÇE•dæ#aneDÁÚĉÁŠœåÁæ)åÁ(æ∂Á;}|^Áa^Á•^åÁ æ)åÁ^|ā\åÁ;}Áa^Á?[|&ã[ÁÇE•dæ#aneDÁÚĉÁŠœåÁ[¦Ás@:Á]`¦][•^Áæ\*¦^^åÁa^c;^^}ÅÕPÖÁæ)åÁs@A P[|&ã[ÁÇCE•dæ#aneDÁÚĉÁŠœåÁxe-Á\^óA[`óA§AÚ^&ca[}ÁFEFA[-Ás@eiA^][¦dEĂ

ÕPÖÁ;c@;¦,ãr^&aã&akaaaai,•Á^•][}•ãaājaã`Át[Áaa)^A;•[}Ajro;e[}Af;c@;!Ác@aa)A?[|&aaiAáQE•daajaaaDAÚc`ÁŠcaa'Á æiārāj\*ÁsjÁ&[}}^&cā];Á,ão@Ác@áA^][;dĚAÕPÖÁadp+[Ár¢&]ča^•Asi[]|&råÁ,æ;!æ)ca?•Áæ)åÁ&[}åãaaaj}•ÉAt[Á c@ Ár¢cr}oAfr\*æa|^Á,^;{ãr•ãaa|^ÈĂ

V@Á^¦ça&^•Á`}å^¦cæà^}Á\$i^ÁÕPÖÁ\$jÁ\${[}}^&caj;}Å ãc@Á;¦^]ædj;\*Ás@aiÁ^][¦cÁ;^¦^Áaj;ãc^åÁt[Ás@(•^Á •]^&ãã&ed|^Á\$i^cæaj^åÁ\$jÁs@Á^][¦cÁse)åÁse^Á\*àb^&cáj;Ás@Á&{[]^Áaj;ãæeaji}•Á<cój;Ás@Á^][¦dÉÁ

V@A[] ā]ā[}•Ê\$&[}&|`•ā[}•A\$a) å A\$a) ^ Á^&[{ { ^} å aceā[}•A\$j A\$@acāA^][¦oÁset^A\$ace^^åA[}A\$&[} å acāā]}•A ^}&[`}c^\^åA\$a) á A\$j -{[¦{ aceā]}A^çā?, ^å A\$ceA\$c@a^A\$ace^A[,A]|^] æbacaā]}A[,A\$c@A^][¦oĚAÕPÖA@ee A[[Á !^•][}•āa ājāc A[,!A[,à]ā\*aceā]}A{[A]] å ace^A\$c@a A^][!oÁs[A\$ce&&[`}oA[;!A^;c^}oA[,!A\$@ee) \*^•A[,&&`;!ā]\*A •`à•^``^}oÁs[A\$c@A\$ace^A\$c@eeA\$c@A^][!oA; ace A[,!^] æb^åĚA

V@A[]∄jā[}•Ê&[}&|`•ã]}•Áæ)åÁæ)^Á^&[{{ ^} åæaā]}•ÁÿÁœãA^][¦ơÁæ^Áàæe^åA[}Aœ\*`{]cā]}•Á {æå^Áa`ÁÕPÖÁå^•&¦ãa^åÁ§ÁœãA^][¦dĚÁÕPÖÁåã&&|æã]•ÁãæàãJãĉÁæãã]\*Á¦[{Áæ}^ÁçAœA æ•`{]cā]}•Áa^ā]\*Á§&[¦¦^&dĚĂ

ÕPÖÁ@æe Á, |^]æ\^åÁx@æ`Á^][¦ớң}Áx@ Ásæ æi Ą, -Áşi-ქ¦{æa‡i}Á, ![çãā^åÁs^ÁP[|&ā;ÁQE • dæjāæbÁúc'ÁŠcáÁ æ)åÁ(c@ !•Á, @, Á, ![çãā^åÁşi-ქ|{æa‡i}ÁţiÁÕPÖÁĢj&|ĭå∄i\*ÁÕ[ç^!}{^}ơ⁄sečc@;!ãæ?•DÉÅ,@&&@ŐPÖÁ @æe Á, [ớkşiå^]^}å^}d^Áç^!ãæ?åÁţiÁ&@ &\^åÁs^^[}åÁs@ Áseč !^^åÁ\*&[]^Á;-Á, [!\ÈŐPÖÁs[^•Á,[cÁ æ&&^]ơÁãæàãjãc ÁşiÁ&[}}^&cāt}Á,ão@Á`&@Á`}ç^!ãæ?åÁşi-ქ!{æa‡i}ÊÆsi&[ĭå∄i\*Á?!![!•Áæ}iåÁ{{ã•āt}>•ÁşiÁ c@Á^][¦ơÁ,@&&@Á,^!^⁄&&e^\*åÁs^ÁY!![!•ÁţiÁ{{ã•āt}>Á§i-át}>AşiÁs@æeÆsi}\_{{}}aæati}EÆsi&[`A

# HUV Y cZ Vtbhybhg

Á

FÈ Qd[å\*&d]} FÈF FÈG  $\vee_{a}^{*} \wedge_{\mathbf{P}} \mathbf{A}^{*}$ FÈH ŒÈ GÈ ŒĠ Õ`ãå∧lā∧•A Ô[}åãā;}•Á;Ád;]:[çæ GÌH GÈ GŤ Y æ^¦Á08cÁ8c/}•^ A  $\hat{O}_{\varphi}\hat{a}_{\varphi}\hat{a}^{*}\hat{A} \left\{ \hat{A} \right\} \left\{$ ΗÈ HÈÈ Ù[ā/áæ)å•&æ]^A HÈG HÈH HÈÈ Õ¦[`}å, æ^\A ΗĚĖ ΙÈ ΙÈ. Qd[ å` &d] } A IÈG Ùã~Á æ^¦Á•^A ĺÈ Ò; çã[ } { ^} æ‡⁄&[ } d[ |⁄4 ^æ` ¦^• ⁄A ÎÈ Ò}çã[ } { ^} æ�/\$j•] ^&cãį }• A ÎÈ ÎÈH Ô[ } @; \* ^ } & Á |æ ΪÈ Á Á Á

# HUV`Y`]bXYI <sup>·</sup>

Á Á

Væà ^ÁGËF	Ô[}åããą]}•Á;~Áæ]]¦[çæ)Á^ ^çæ)óÆ[Á@2#ÁĴYTÚ
Væà ^Á <del>HË</del> Ë	$\tilde{O} [~~\} a_{j} \approx  A_{j}^{\wedge} c_{j}  = \frac{1}{2} \left[ A_{j}^{\wedge} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j} + A_{j}^{\vee} c_{j}$
Væà∣^Á <del>HËG</del>	Õ¦[`}å, æv\¦Á æv\¦Á * ækî <b>Autoministri in the set of t</b>
Væà ^A <del>́HË</del> H	Ùʿ¦æ&^Á,æ^¦Áʿæ‡ãĉ A <del>ddidididididididididididididididididid</del>
Væà ^ÁËË	Þ^ớ{ [} ớ@́ Á æ^¦Ása¢a) &^Á ඎã ເໝ A <del>ddiddiddiddiddiddiddiddiddiddiddiddiddi</del>
Væà ^Á.ËF	Ò}çã[}{ ^}cæ¢Æ{[}d[ Á{, ^æ `¦^• Å
Væà∣^Á́EG	Ó ľ^ / Ŕ ¦^^} Á¢* æ Á¢/ ¦ á/ ç^  • ∰₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩
Væà∣^ÂËË	Ù[ā/Áæ)åÁ,ææ^¦Á;[}ã{;¦ā}*Á;¦[*¦æ; Á
Væà∣^ÂİËG	Ô[ } @ *^} & Á   @
Á	

# : ][ i fY<sup>·</sup>]bXYI "

 $\mathcal{Q}\tilde{a}^{*} \wedge \hat{A} \stackrel{\text{\tiny (IF}}{=} T[] \tilde{a}_{l} * \hat{A}_{l} & \mathcal{Q}_{\text{\tiny (IF}} + \hat{A}_{l} & \mathcal{Q}_{l} & \mathcal{Q}_{\text{\tiny (IF}} + \hat{A}_{l} & \mathcal{Q}_{l}  5ddYbX]Wg

```
CE,]^}åã¢ÁŒÁÁQE<sup>*</sup>^}&<sup>*</sup>ÁÔ[}•<sup>×</sup>|cææāį}Á
CE,]^}åã¢ÁÓÁÁÔ<sup>!</sup>[•ã;}Áæ)åÁ<sup>*</sup>^åã;^}ơ&S[}d[|Á;|æ)•Á
CE,]^}åã¢ÁÔÁÁÔ<sup>!</sup>[•ã;}Áœ)ÅÁ<sup>*</sup>Aů[ã¦ÁTæ)æ*^{{ }}ơÁÚ|æ)Á
Á
Á
```

# %<sup>A</sup> =bhfcXi VM\_cb<sup>-</sup>

## %% CV<sup>4</sup>YWj Yg

 $V@\dot{A}^{\dot{A}} \dot{A} ab \& ac / \dot{A} Ac @ \dot{A} T U Ac \dot{A} \dot{A} Ac @ \dot{A} + \dot{A} @ \dot{A} + \dot{A} @ \dot{A} + \dot{A} @ \dot{A} + \dot{A} @ \dot{A} + \dot{A} @ \dot{A} + \dot{A} @ \dot{A} + \dot{A} @ \dot{A} + \dot{A} @ \dot{A} + \dot{A} & \dot{A}$ 

## %'&Á HUf[Yhg

V@Á{I||[, ā]\*Áæa\*^orÁ@æç^Áa\^}Á\*•œaei|ãr@åÁ{I¦Ás@Á{a}æ\*^{ ^}of{Aá]āÁæ}åÁ;æe\*¦Áã[]æ&orÁ å`¦ā]\*Ás@Á[]^¦æeā[}æ4Áã^cã[^Á[-ÁÖ`}|[^ÁÜæ)åÁÛ`æ4¦^KÁ

- •Á Ò}•˘¦^Á˘||Á&[{]|ãa;) &^Á, ão@ko@ Á^|^ça;) o4/^\*ãr|aacãç^Á^˘˘ã^{ ^}o Ác;) a4Ô[OEÁ
- $\bullet \dot{A} = T \wedge \alpha \dot{O} \dot{U} \dot{S} \dot{A} = \alpha \dot{A} \dot{A} = \alpha \dot{A} \dot{A} = \alpha \dot{A} \dot{A} = \alpha \dot{A} \dot{A} = \alpha \dot{A} \dot{A} = \dot{A} + \dot{A} = \dot{A} + \dot{A} = \dot{A} + \dot{A} = \dot{A} + \dot{A} = \dot{A} + \dot{A} = \dot{A} + \dot{A} = \dot{A} + \dot{A} = \dot{A} + \dot{A} = \dot{A} + \dot{A} + \dot{A} = \dot{A} + \dot{A} + \dot{A} = \dot{A} + \dot{A} + \dot{A}$
- •Á Ò} ` \^ Ast and a \* A; } A [ a/set) å A and A

## %"Á 7 cbgi `HJh]cb

$$\begin{split} \dot{O}_{CC} &= \tilde{a}_{c}^{A} \& [ \} \bullet^{-1} [ ] &= A^{+}_{a}^{A} = A^{+}_{a}^{$$

Á

# &"Á FY`Yj Ubh``Y[]g`Uh]cb'

# &'%Á @Y[]g`Uh]cb`

Á

Š^\*ã=|æaā[}Á^|^çæa)-cÁ{[Á[āļÁæa)-åÁ;æa^\¦Á(æa)-æ\*^{ ^}cÁa)-&(^ \*

- •Á Protection of the Environment Operations Act 1997 (POEO Act)
- •Á Water Management Act 2000 (WM Act)
- •Á Fisheries Management Act 1994 (FM Act)
- •Á Water Act 1912 (Water Act)

Ø ¦c@¦Áŝã&\*••ā[}Á[,Ás@Ásaà[ç^Á^\*ã|æaā]}ÁārÁ];[çãå^åÁ§JÁs@ÁÒTÙÊseeÁ]^||ÁseÁs@ÁÒQÌÁse]åÁTUÖGÈĂ

# &'&Á ; i ]XY`]bYg<sup>·</sup>

V@:Á{[ ||[ , ]];\*Á\*`ãå^|]];^•Á@æç^Áà^^}Á&[ }•ãå^¦^åÁå`¦]];\*Áå^ç^|[ ] { ^} ơ{{, Á∞@éAÛY TÚKÁ

- Á OESaãÁÙ (~ ae^ ÁU [ 4ª ÁT a) ča þÁÇDEÙÙ T OEÔ ÁFJJÌ DÁ
- •Á Ù[ậ+Áa) åÂÔ[}•d č & cā; } ÉĂ a) æt ã; \*ÁV; à ca) ÁÙ[t]; { ^ÁCÒÁT ã; ^•Áa) åÂÛ č ad; lãv•ÉĂ l c@ÂÒåãā; }ÁÇÓ|č ^ÁÔ[[\DÁÇŠa) å&[{ÁOEEIDÁ
- •Á ÞÙY ÁCE  $\tilde{a}^{A}$  ÁQ  $c^{A}$   $a^{A}$  & AÚ[ | $a^{A}$  ÁQÖÚQÁ ÁJ  $a^{A}$   $a^{A}$   $a^{A}$   $a^{A}$
- •Á Õĭãå^|ã,^•Á{;¦ÁTaa) aztāj\*ÁÜã\•Á§jÁÜ^&\^æaã;}aa,ÁYaaz^¦ÁQ⊃PTÜÔÁG€€ÌDÁ
- •Á Ô[å^Á;ÁÚ¦æ&ca&A[¦ÂÚ[ã/Ása)åÁYæe^¦ÁTæ)æ\*^{^}o{;}ÁÔ[}•d`&cā;}ÂÚãe^•Á
- •Á Ö^ç^[[] { ^} dÖ^• ā } ÂU] ^&ãã&æaā } AÖÏ Á ÂUq ¦{ ] æe^¦ÂU ` æpác Áæ} åÁ/ ] ^^åÁWaæ) Á Ùq ¦{ ] æe^¦ÂU ` æpác ÁT æ} æt ^{ ^} dÚ[æ) Á

## &" Á 7 cbX]h]cbg cZ Uddfcj U

V@ÁÔ[OEÁ\^|^çæ)ók[Ác@aÁÛYTÚÁse^Ápã:c\*åÁsa^|[,ÁşiÁ/æa)^ÁGËEÈÁDEÁsi[••Á^-^¦^}&^ÁsiÁsep•[Áşi&]\*å^åÁ q[Áşiåã&ææ^Á;@¦^Ác@Á&[}åãoā]}ÁsiÁseåå¦^••^åÁşiÁc@aiÁÛYTÚÁ;¦Á[c@¦Á\*}çã[]{ ^}œ4Á;æ)æ\*^{ ^}cÁ å[& { ^}orÈÁ

Ô[}åããąi}Á¤[ÈĂ	Ü^˘˘ã^{ ^}ơÁ	Ü^-^^¦^} &^Á		
Ù&@@åĭ ^Á+ÐÃ Ô[}åããą[}ÂÍÁ	Ô¢&^] chee Á, æâ Ás Ú¦[][}^} chi ~ é d [¦Áe) &ā]æ^ Á] ^¦a cœede@ A ¢c æ8a æ} å Á[]^¦æe* å Á[ Á -¦[{Ác@ • ^ Á][} á	Væà∣^ÁÉËÁ		
Ù&@åĭ ∧ÁnĐÃ Ô[}åããậ[}ÁJÁ	V@ÁÚ¦[][}^}ơ\ [àb∿&cãç^•ÁşiÁ/æà *'[[`}å, æ¢\¦Áæåba æ]]¦[ç^å,Åà`Ás@/ Væà ^Á\KÁY æ¢\¦ÁÛ	Væà∣^ÂËÉÁ		
	Ú[ ∥` æ) ớ́́́́́́́	W}ão4(,~ÁT^æ`¦^Á	Yæe^¦ÁÛ过ãĉÁ Uàb∿&cãç^∙Á	
	Vĭ¦àãåãĉÁ	ÞVWÁ	ÍËG€Á	
	] PÁ	] PÁ	ÎĔËĂ	
	U <b>ậ∕a⇔</b> åÁ Õ¦^æ•^Á	{ * £ŠÁ	F€Á	
	Ùæþājāc Á	µÙÐ&{Á	ŁH€€€Á	
	Öã∙[  ç^åÁ U¢^*^}Á	{ * £ŠÁ	NÎÁ	

## HUV'Y'&!% 7 cbX]h]cbg'cZUddfcj U`'fY'Yj Ubh'hc'h\]g'GK A D'

Ô[}åããą;}ÁÞ[ĐÁ	Ü^˘˘ãă^{^}oÁ			Ü^-^^¦^} &^Á
	Ô@[¦[]@  ËæÁ	µ*£ŠÁ	G <del>ÏT</del> €Á	
	Øæ≏&æ‡Á &[ ã{[¦{ ● Á	T^åãaa)Á Þ[⊞B7€€€{ŠÁ	ŁF <del>€€€</del> Á	
	Ò} ơ∿¦[ 8[ 888ãÁ	T^åãæ);Á Þ[ÈBB∓€€{ŠÁ	ŁGH€Á	
	O≣*æ°Áæ)åÁ à ĭ^Á*¦^^}Á æ†æ°Á	Þ[È3&∧∥∙Ð(ŠÁÁ ÇTÈ5æ≎¦`*ậ][•æÐÁ	ŁÍ <del>€€€€</del> Á	
	Á	{{ <sup>H</sup> EŠÁÇ[cæ)Á àã[ç[ ǐ{^DÁ	ŁIÁ	
	Ù[ åã { Á	{ * <b>EŠ</b> Á	Í€€Á	
	Ú[cæe∙ã{Á ãį}Á	{ * £ŠÁ	I€Á	
	Tæ*}^∙ã{ Á	{ * £ŠÁ	F€€Á	
	Ô@{[¦ãâ^Á\$[}Á	{ * £ŠÁ	F€€€Á	
	Ùٽ∥]@eee∿Á§ą}Á	{ * <b>EŠ</b> Á	Ì€€Á	
	Óã&æelà[}æe∿Á ãį}Á	{ * £ŠÁ	I€€Á	
	Ù[ ĭà ^ÁCC[}Á ã[}Á	{ * EŠÁ	G€Á	
	Ù[ `à ^Á æ∤ั{ājã{Á āj}}Á	{*EŠÁ	€ĽÍÁ	
	OĘ[{[}ã{Á ã[}Ă	{ * <b>EŠ</b> Á	G€Á	
	Notes:			
	• The obiectives	for dissolved oxva	en. turbiditv and algae	
	are relevant to s	urface water only.	,, <b>,</b>	
	· The Departme	nt acknowledges th	at short term	
	exceedances of events such as f	these objectives ma looding.	ay occur during natural	
	· The Departme			
	quality may not r	meet the objectives	for some analytes,	
	water quality obj	ectives through imp	lementation of the Soil	
	and Water Mana	gement Plan (see o	condition 18 below), as	
	far as is reasona Proponent's com	ble and feasible an trol, to the satisfacti	d within the on of the Secretary.Á	
Ù&@åĭ ^ÁrHÉÂ Ô[}åãa‡j}Ár∓€Á Á	V@ÁÚ[]]{}^}ók æ&ãāÁ` -æ¢Á[ĨÅ c@Á 梦cæà ^Áæ Þ[Á][¢}cãa\ás&ā •ã¢ĒX} ^••Áæå^` { ^c@å*Áæ}]¦[ç⁄ Ú æ)ĒA	× 6⁄4) • ` ¦^Áx@eeÁse / āj ^• Á; ææ^¦ãædy≦si Á^č Á^[]} Ásee Aj[••āa]^A Á*  -æe^Á[ā/4` • 64a Á*  -æe^Á[ā/4` • 64a `æe^ ^Â,^` daapä^åÁ ^åÁ}å^¦Ác@ÁU[ā/ka)	Ýv¢&æçæz^åÅ[(c^}cāæ)Á ¦}^åÁaæ&\Á[Áa^ [_Á [Á¦^ç^}o∱¢ãāæaā]}ÈÁ ^Á^{[ç^âÁ-[{Ás©A ĴÁæ&&[¦åæ)&^Á,ão©Á åÁvæs∿¦ÁTæ)æ*^{ ^}oÁ	Væà ^Á ËÁ
Ù&@åĭ ∧Á+ÐÃ Ô[}åããa[}ÁFFÁ Á	V@ÁÚ![][}^}ơł, •[ā¦Áāj^•/⁄a æc*¦ãa }[Á^••/kā@ajÁ+/k -āj^•/kād^/kā^][•ãa -{[{ Áx@Ajæc*!/Á aa]]¦[ç^å/kārÁÖ[(	≚•o4\}•`¦^Ax@ee4ke#/ ¢Asia &@ee*^â45id[ ^d^•A{:[{ Ax@A;ae^ ?åA{[AzzA;ae4sia]ae4sia^]c@A ?¦~aa&^Ê4;} ^••Aa3}Aad Aa3}åAx@AL/^&¦^caa?Ê Aa3}åAx@AL/^&¦^caa?Ê	ἡ[c^} Œ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩ ₩	Væà ^Á ĒTÁ Ö^][•āā] * Ás@Á -ā, ^• ÁsæAstá^] c@Á -Å, Ás Á, [cÁ ] 'æsca3catĒA;[Á:@Á &[, ] c] /43 Á/æà ^Á i ĒTÁ@æ•Åa^} Å i ĒTÁ@æ•Åa^} Å at -} å^å á k[Á  ^-, %cks@Aā] • á à^ā] * Åa^][•ār å ÁsæÁ æ4(ājā] ~ { Åa^] c@Á [-Á+K Á

Ô[}åãaặ[}Á⇔[ÈÁ	Ü^~~ă^{ ^} ơÁ	Ü^-^\} &^Á
Ù&@°åĭ ^ÁnÉÃ Ô[}åããậ]}ÁFGÁ Á	V@ÁIJ[][}^}ơ₄ ` •ơ₄ æ) ﷺ ٨[ أَفَ عَدْمُمُ عَدْمُهُمُ اللَّهِ اللَّهِ عَدْمُهُمُ اللَّهُ عَدْمُهُمُ اللَّهُ عَدْمُ اللَّهُ عَدْمُهُمُ اللَّهُ عَدْمُهُمُ اللَّهُ عَدْمُهُمُ اللَّهُ عَدْمُهُمُ اللَّهُ فَ مَعْتَ عَدْمُ اللَّهُ عَدْمُهُمُ اللَّهُ مَعْتَ عَدْمُ اللَّهُ مَعْتَ عَدْمُ اللَّهُ مُعَامًا عَمْمُ اللَّهُ فَ مُعَامًا عَمْمُ اللَّهُ فَ مَعْتَ عَدْمُ اللَّهُ مُعَامُ اللَّهُ فَ مَعْتَ عَدْمُ عَدْمُ عَدَى اللَّهُ عَدْمُ اللَّهُ عَدْمُ اللَّهُ عَدْمُ اللَّهُ عَدْمُ اللَّهُ عَدْمُ اللَّهُ عَدْمُ اللَّهُ Protection Guidelines – On-site Sewage Management for Single Households (1998)ĔĂ	Væi¦^Á ËÁ
Ù&@°åĭ  ^ÁnĐÃ Ô[}åããậ]}ÁFHÁ	V@ÁÚ¦[][}^}ơ4(`•ơ4`}•`¦^Áo@aeA+[[[å/ài`}åā)*Áae[`}å/à@A Ùcæt^ÁrÁaa)åAUcæt^ÁCA [¦\•Áa[^•A,[of4¢&^^å/A+E€A;{ /ág Á @ã@áaai[ç^A;æc`¦æ4A`¦-æ&^Á^ç^ ĔA;/áaa*i; { Á@ã@á [-ÁCHÈEA;ÁOEPÖEA`} ^••Á;c@¦;ã^/áaa]]¦[ç^å/ài^Ás@Á Ù^&¦^cæ5`EA	Væà∣^ÁiËFÁ
Ù&@°åĭ  ^ÁnÉÃ Ô[}åããậ}}ÁFIÁ	V@ÁÚ[][}^}o4(`•o4^}•`¦^Áo@ee4(^¦ã(^c+'¦å(æaa)æ*^A(`•oÁ à^Áaj•cæaa¦^åÁaa)åA(]^¦æaaa[}æ4A;lã(Áo@A&[}•d`&cã[}A(-Á à`}åã)*Á(;Ás@A) æ&^{ ^}o4(-Áaa A()}Aíãc+ÉA	Væà∣∿ÁiÈEÁ
Ù&@°åĭ ^Á+ĐÃ Ô[}åãa∄}Á≂ÍÁ	O⊞Á^æało@,[¦\●É&aj&& ĭåāj*Áļ[[åÁ&anjåÁxæ& ĭ●ca&Aáaĭ}åäj*Á ,[¦\●ÉA(ĭ●cAáa^Á&[}cæaj^åÅ, @{  ´Å;ãc@ajÁc@Aŕãa^ÉA	Væà∣^ÁľËÁ
Ù&@°åĭ ^ÁnÉÃ Ô[}åããậ[}ÁrÎÁ	V@ÁÚ¦[][}^}ơĄ`*oó&xæ~Áå¦^å*āj*ÁæjåÁ¦[&^••āj*Á æ&cã;ããð•Á[cn{^••Áœ}ÁGIÁQ`¦•Á;lā¦ÁgiÁ@Á &[{{^}&x{^}c^}A;c^!+[,Á!{{Aæj^Aå!^å*^Á[}åĔÞ[Á å!^å*āj*Á;!Á;![&^••āj*Á;`•oÁ;&&`¦Á;@}Ás@Aå!^å*^Á ][}å•Áæc^Á;ç^¦+[];āj*ĚĂ	Væà∣^ÁÉËÁ
Ù&@°åĭ  ^ÁnĐÃ Ô[}åããậ[}Árī Á	V@ÁÚ[][}^}ớ, *ớA}* * /Áa@æÁ@Á[[åÁ:d[æ*Á &æ] æ&āî Á; Á@Á ã: /á: Á [Á ** Á@æ Á @ Á] / Ё ¢ã cij * Á[[åÁ •d[æ* A⁄&æ] æ&ãî Áœ/æ[Á cæ* ^ Á; Á@A ][b & dĚÖ^ cæ‡ Á, Á@Á æçæ‡æå ^Á[[åÁ:d] æ* A⁄&æ] æ&ãî Á; * oÆA^A][lc åÆj Á@Á OE} * æÁÜ^çã, ĚÁ	Væà∣^ÁiËFÁ
Ù&@@åĭ ^A∱HÂ Ô[}áãã‡}ÁrÌÁ	V@ÁU![][}^}of, `•of,!^] ab^/kekU[ \$\u00e9ba ah are ab in the above for the above fo	V@ni,∱i æ)iÁ
Ù&@@åĭ ^ÁnÉÂ Ô[}åããa[}ÁrĒJÁ	V@ÁYæc*¦ÁÓæ¢æa)&^Á; `•Ó\$9,& `å^kÁ GeĐÁå^œæ‡*Á;-ÁseljÁ;æc*¦Á^¢ctæ&c*åÊÁtæ}•-^¦!^åÊÄ:•^åÁse)åЦÁ åã&@æt*^åÅs^Ás@Á`æt!^LÁ GaDÁ©EÁ[`!&^Á;-ÁseljÁ;æc*¦Á&[  ^&c*åÁ;!Á:d[!^åÅ;}Á©EÁ;ã*ÊÁ 3;& `åä;*Áæaj;-æt]ÊA:d[:{; æc*¦Áse}åÁ'![`}å;æc*¦LÁse)åÁ G&DÁ:^Aæ*'!^•Át[Á;3;ä;ã*A; æc*¦Á:•^Ásî^Ás©A:![b*&dĚÁ	Ù^&cāį}ÁrtĒÁ
Ù&@ åĭ  ^Á+ÉÂ Ô[}åããậ}}ÁG€Á	V@ÁÔ![•য়] Ása) åÂ\/^åয়AØ[] d[  ÁU æ) Á; *•dÁGata/ÅAæ) åÂ\/^åয়AGata/ÅAæ) åÂ\/^åয়AGata/ÅAæ) åÂAæUrban Stormwater: Soils and Construction, Volume 1, 4thEdition, 2004 Gab å&[ Dab à ÂÔ[ ` ] & and Construction, Volume 1, 4thEdition, 2004 Gab å&[ Dab à ÂÔ[ ` ] & and Construction, Volume 1, 4thEdition, 2004 Gab å&[ Dab à ÂÔ[ ` ] & and Construction, Volume 1, 4thEdition, 2004 Gab à & and Construction, Volume 1, 4thEdition, 2004 Gab à & and Construction, Volume 1, 4thEdition, 2004 Gab à & and Construction, Volume 1, 4thEdition, 2004 Gab à & and Construction, Volume 1, 4thEdition, 2004 Gab à & and Construction, Volume 1, 4thEdition, 2004 Gab à & and Construction, Volume 1, 4thEdition, 2004 Gab à & and Construction, Volume 1, 4thEdition, 2004 Gab à & and Construction, Volume 1, 4thEdition, 2004 Gab à & and Construction Nor- Stormwater Quality and Tweed Urban StormwaterQuality Management PlanLÁGab & A A à a A & daGab & A A à a A & daGab & A A à a A & daGab & A A à a A & daGab & A A à a A & daGab & A A à a A & a A & a A & a A & a A & a A & a A & a A & a A & a A & a A & a A & a A & a A & a A & a A & a A & A &	Væaà ^ÁiÈFÁag)åÁ Œ[]^}åã¢ÁÓÁ

Ô[}åããąi}ÁÞ[ÈÁ	Ü^~~ă^{ ^}ơÁ	Ü^-^^¦^} &^Á
	Ç>DÁå∧∙&¦ãa∧Á,@æexÁ,^æe`¦∧∙Á,[č åÁà∧Áã,] ^{^}&^åÁų[Á {æãjææãjÁc@••^Árdč&c`¦∧∙Á;ç^¦Ácã,^ĚÁ	
Ù&@@åĭ ∧ÁHEÁ Ô[}åãaáj}ÁOEFÁ	V@ÁQBaāÁŮ  -æc^ÁÙ[ ᢤ紅 a)	Væaì ^ÁÉËrÁæ)åÁ CEJ]^}åã∉ÁÔÁ
Ù&@a° ∧Á+EÂ Ô[}åãã[}ÁGGÁ	V@ÁOj 、 ÉD : ハ > ÁOE* æð AT æ) æð 、	Væà∣^ÁiËFÁ
Ù&@@åĭ ∧ÁHÊÁ Ô[}åããa[}ÁGHÁ	V@ÁÜ <sup>*</sup>  -æ&A´A′æA <sup>*</sup>  ÁT []ã[  ã]*ÁÚ [*  æξ Á, *• dã] & * å^ ká GeDÁ <sup>®</sup> ^ cæã <sup>A</sup> å Ásæ ^  ã] ^ Ásæææá[} À *  -æ&A´, æA <sup>*</sup>  Á * æaš LÁ GeDÁ <sup>®</sup>  -æ&A´, æA <sup>*</sup>  Á§[ ]æ&OÁæ • ^•• { ^} d& ãA <sup>*</sup> ]äætÁ G&DÁæá],  [*  æξ Á[ Á, []ã[  Á*  -æ&A´, æA <sup>*</sup>  Á[], •Áæ) å Á * ædãî LÁ G&DÁæá],  [*  æξ Á[ Á, []ã[  Á*  -æ&A´, æA <sup>*</sup>  Á[ { Ás@A * ãA <sup>*</sup> LÁ GDÁæá],  [*  æξ Á[ Á, [] ã[  Áæ] \Áæ] à Ásh å Å cæã ããî LÁæ] å Á GDÁæá],  [*  æξ Á[ Á, [] ã[  Áæ] \Áæ] à Ásh å Å cæã ããî LÁæ] å Á GDÁæá],  [*  æξ Á[ Á, [] ã[  Áæ] \Áæ] à Ásh å Å cæã ããî LÁæ] å Á GDÁæá],  [*  æξ Á[ Á, [] ã[  Áæ] \Áæ] à Ásh å Å cæã ããî LÁæ] å Á GDÁæá],  [*  æξ Á[ Á, [] ã[  Áæ] \Áæ] à Ásh å Å cæã ããî LÁæ] å Á GDÁæá],  [*  æξ Á[ Á, [] ã[  Áæ] \Áæ] à Ásh å Å cæâ ããî LÁæ] å Á GDÁæá],  [*] & A[ Á[ A´æ] \Áæ] \Áæ] à Ásh å å cæâ ããî LÁæ] å Á GDÁæá],  [*] & A[ A´æ] \Áæ] \Áæ] \Áæ] à Ásh à Å cæâ ããî LÁæ] Å GDÁæá],  [*] & A[ A´æ] \Áæ] \Áæ] \Áæ] \Áæ] \Áæ] \Áæ] & A GDÁæá]  [*] & A[ A´æ] \Áæ] \Áæ] \Áæ] \Áæ] \Áæ] \A Gæã ææã] } Á; -ÁæA \Gæã à Ás \A <sup>*</sup> A <sup>*</sup> \æÅ] \Æ ã] ]æ&Oæ• • • • { ^} OÁ& ã <sup>*</sup> \æÆ]	Væà∥^ÂÎËËÁ
Ù&@@å* ^Á+EÂ Ô[}åãaậ}}ÁGIÁ	V@ÁŐ![`}åÁ'æ&\ÁT[}ãt[!ā;*ÁÚ![*!æţ Áţ`•ó45)& `å^ká GetJå^cæathåkæ*]ā^Átææath}Å![`}å,æe\A(\$\`e^ •Á50}åA ``ataî Étsiæ*åth}Å Acæzā cikadata)æti*äthá GaDAt![`}å,æe\At[]ækofse****{}åE GaDAti[`}å,æe\At[]ækofse****{}ôklæthá GaDAti[`}å,æe\At[]ækofse****{}ôklæthá GaDAti[`}å,æe\At[]æt[At][`}åAjæe\At(\$\`extrm{o}h^c<\extrm{o}h^c At(\$\`extrm{o}h^c GaDAti]}Êtsi At(`}åAt[`}åAjæe\At(\$\`extrm{o}h^c At(\$\`extrm{o}h^c Catal})Êtsi At(`}åAt[`}åAjæe\At(\$\`extrm{o}h^c At(\$\`extr	Væà∣^ÂÈËÁ

Ô[}åããąį}Á¤[ĔÁ	Ü^``ã^{ ^} ơÁ	Ü^-^\^} &^Á
Ù&@@åĭ ^ÁLÉÀ Ô[}åããa∯}ÁFOEÁ	V@AÚ¦[][}^}ơ4(``•ơ4`}•`¦^Ab@eeaAo@A(:aa)az*^{ ^}ơ4, aa)•A ¦^``ã^åÅ}å^!Ao@eaAog]![çaa4Aod^A;!^]az^åAsjAoz&&[¦åaa)&^A ¸ão@aba)^Á^ ^çaa)o4``ãå^ ãj^•Êaba)&\ša &{iša}& ĭá	Á
	ÇæÐÁæÁ* {{ æ¦^Á^ ^çæ}) óÁsæ&\*¦[*}åÁ{¦Ásæ^ ā], ^ÁsæædaÁ	Ù^&cãį}}Á∺Á
	ĢaDabeoda ^ e & lậi cặi } Áị - HÁ •Á c@ Á^ ^çæ) có ki cæ č (; l ^ Á^ č ă^ { ^} o ÁĢa & č åã * Áe) ^ Á ;^/°çæ) có be] ] [ çæ HŽABX } & ^ Á; l Á^ æ ^ Á&[ } åãā] } • DLÁ •Á æ) ^ Á^  ^çæ) có kaj ão A[ l Á] ^ l - [ ¦ { æ) &^ Á[ ^ æ č  ^ • E& ão \} ão dA æ) åÁ	Ù^&ca[}ÁGÊÂÙ^&ca[}Á FÈGÁse)åÁ⁄æà ^ÂÎËFÁ
	c@\Á]^&ãã&Aj^¦-{¦{ a} & & Aş åã&aat[\•As@ad%a&^A;\[][•^åAş Á à^Á•^åAş[Ás å*^As@Aj^¦-{¦{ a} & Af,-Æş,¦A* ăå^As@A ā;] ^{ ^} caata;}Aj-Æs@Aj. [b^&Aj,iAaa)^Aj, a) aet^{ ^} oÁ { ^ae `¦^•LÁ	
	Ç&DÁeceÁs^+&klājcāj}Áj~Ás@∿Áj^2æ*`¦^+Ás@ezeÁs[Ása^Ás[] ^{^}&^}&^åÁ q[Ás2[{] ^Á,ãc@abs@Á^ ^c;æ);cAicæečq[¦^Á^``ã^{{ ^}}@EÁajīãe=EÁ [¦Áj^¦-{]{ æ);&^∕Áj^2æ*`¦^+E&klãe×lãaabÁ	Væà∥∧ÁÍËEÁ
	ÇåDkæok,   [*   æçi Áqi Áqi [}ãqi   Áægi å Á^ ] [ ¦oAqi } Ás@ HA ●Á ãj ] æ∨ Áægi å Á^ } çã[ } { ^} œagi Aj, ^¦ -{   { ægi &^ Aj, ~Ás@ Á ]   [b% &du Áægi å Á ●Á ^ ~^ & &ãç,^ } ^•• Áqi – Áægi ^ Áqi ægi ægi ^{ ^} ðqi ^ ægi ` ¦^• Áçi ^ ^ æài [ ç^ DuÁ	Væà∣^ <b>Â</b> ËÉÁ
	Ç DÁwa&[}cā) * ^}&î A;  aa) Á[Á; aa) az ^Áanj ^Á }] ¦^åa33c^åÁ ã[] az Sor Áanj å Ás@ã Ás[}•^``^}&^•Áanj å Ás[Á?}•`¦^Ás@az Á [}*[ã] * Ás[] az Sor Á^å * &^Ás[Á^ç^ •Áa^ [, Á^ ^çanj ó Ss[] az SoÁ az ● ^ • • { ^} ó Ss! ão ¦ãa áz ár í * az Á] [ • • ãn] ^ [ • • ãn]^LÁ	Væà∥^ÂÎËÁ
	ÇDÁnazÁ,¦[*¦aq(Áng(Ang)ç∧∙cataaarAnag)å/Ang[] ^{ ^}oÁ,aê•Áng(Á ang[]¦[ç^Áno@Án}}çang[}{ ^}caa‡Aj,^¦-{¦{ ang}8∧Aj,-Ano@Aj,¦[b/8oA [ç^¦Ánagi^Á	Væà∥∧ÂiËEÁ
	Ç Dáezá, ![ ([ & [  Á[ !Á[ æ] æ# ā] * Áæ] åÁ^] [ :cā] * Áæ] ^ IÁ • Á ā] & äā^ } o LÁ • Á & [ { ]  æā] o LÁ • Á & [ { ]  æā] o LÁ • Á } [ ] ËE[ { ]  ãæ] & ^• Á ão @ Á cæeč ([ !^ Á^~ č ã^{ } * á^{ } * a^{ }	Ù^&ca≱}ÂiÈIÁ
	Q@D\$xxx4,\[d[&] Áf \Á,^\\$1,å&3&A^ça?_,Ár~Ax@A, aa)EA	Ù^&cãį}Ä̈́Á

# &"(Á 9bj]fcba YbhDfchYVMjcb`@JVMbVM`

V@ ÁÒÚŠÁ&[}åãaāį}•Áįčdā,^Áx@ Á[&ææāį}Á;Á, Á[]ãa[¦ā]\*Éx@ Á&[}&^}dæaā;}•Á;Á,[||čæa)•A;^¦{ãac^åA d[Áà^Ásiā &@ed\*^åÉxæt]|ā]\*Á¦^č`^}& Éxæt]|ā]\*Á;^c@tåAajåAx@ Áæaā;æd;Ax[]\*æa;åA[&ææā]}\*Á \_@}Ás@ Á&[}&^}dæaā;}Áājār Á&æað;Áa^Ár¢&^^å^åÉA/@ Á^č`ã^{ ^} •A; ^} •Á; ā]&]čå^åA\$jÁs@árÁ;|æ)ÉÅ;@'¦^Á/|^çæð;dĚA

## &') Á K UHYf 5VM [VYbgY]

V@ Á⊃ÙY ÁÖÚQÁ ÁJ~38∧Á; ÁY æc^¦Áã;•`^åÁ@ Á[ ||[ ¸ ð] \* Á; [ }ã[ ¦ð] \* Ái[ ¦^Á38∧}•^•Á[ ¦Á;@ ÁÖ` } ||[^Á Ùa) åÁĴčak¦^Á;} Á;@ ÁFÎ ÁÙ^] c^{ à^¦ÁG€€I KÁ

- ●Á H€ÓŠFÌH€ÏÎÁ
- ●Á H€ÓŠFÌH€ÏÏÁ
- ●Á H€ÓŠFÌH€ÏÌÁ
- ●Á H€ÓŠFÌH€ÏJÁ
- •Á H€ÓŠFÌ H€Ì€Á
- ●Á H€ÓŠFÌ H€Ì FÁ
- ●Á H€ÓŠFÌ H€Ì GÁ
- ●Á H€ÓŠFÌH€ÌIÁ
- ●Á H€ÓŠFÌH€ÌÎÁÁ

# **' 'Á 91 ]gh]b[ 'Ybj ]fcba Ybh'UbX ]a dUW#g**'

 $V @ \acute{A}[ ||[ ] \ddot{a} * \acute{A} ^ & a \ddot{a} ) \bullet \acute{A} ^{*} \{ \{ a \pm \ddot{a} ^ / \acute{A} @ \acute{A} \phi \ddot{a} a \ddot{a} * \acute{A} \} c \ddot{a}[ \} \{ ^ \} d \dot{E} \dot{a} a = ^ a \acute{A} ] [ \} \acute{A} @ \acute{A} j + [ \{ a \pm \ddot{a} \} \acute{A} ] ] | [ c \ddot{a} ^ a \dot{A} j \acute{A} ] \acute{A} @ \acute{A} ] ] \dot{A} = 0$ 

# ' '%Á FY[]cbU`[Yc`c[m

Á

V@ÁV, ^^åÁFKGÍ €É€€€ÃÕ^[|[\* 38æ¢ÁÛ^¦ã•ÁÛ@^x5ãa^}cãa?•Á@Á{[][\*¦aª}@38æ¢|î Á'|^çæe\*åÁãa\*^|āj^Á q[Ác@Á, ^•x5æ)åÁ,[¦c@á; Ác@Á;ã\*Áœ;áţ`c&¦[]•á; Á'¦^^; æ&\^ÊA|æe\*ÊA,@||ãe\*Áæ)åÁ`æc ãe\*ÁQ; æd¢i, Á c@Á^\*āj}æ¢i Á\*¢c\*}•ãç^Áp~¦æ}|^ã; @á2\*\}çæ¢AŐ[[`]DĚV@Á[[Ë;ā]\*Á{[[å]|æājÁ^&cāj}•Á; Ác@Á •ãe\*Á\*^}^¦æţî^Á&[}•ã; c\*åÁ; Á';&[}•[|ãaæe\*åÁå^][•ãe Á; Áãç^¦Á'¦æç^|•É&oqičçã { ÉA;æjåÁæ}åÁ&]æÂ{;á P[[[&^}^&]\*Áz

## ''&Á Gc]```UbXgVV/dY`

OE Á&|æ••ã&\åÁ}å^¦Á@ ÁÕ¦^æAÛ[ā/ÁÕ¦[č] ÁÔ|æ••ã&Bæa‡i}•ÁÇÕÙÕÔDE&@ Á[ā/Áæ)å•&æ‡]^•Á§iÁ@ Á[&æ4Á æ^æ&æ\*^&&^-āj^åA&e ÁSāj\*•&|ã-ÁÇæa\*ãæ)OD&e)åÁÚ[@•çā|^^A[ā/Áæ)å•&æ‡]^•EÖ^•&&¦ājcāt}•Á;\_ÁA[ā/Á |æ)å•&æ‡]^•Á§iÁs@ Á[&æ‡/Áæ^æ4,^¦^Á&[}åč &cvåA&îAT[¦æ)åÁÇFJJÎDÊ&@ Á[||[]ā]\*Á\*^}^\æ4Á å^•&&¦ājcāt}•Á;@¦^Á,[c\*åEÁ

Ùãr^Áşiç^•cãtæaāt}•Áşiåa8æær^Ás@Áãr^ÁārÁ\*A}^¦æa¦^Á}å^¦æaŝiÁsî^ÁæÁ'}ãt[¦{Á[āļA,¦[-āħ^Ás[}•ãcā;\*Á;-ÁæÁ •{æa¦Áset[`}ofa;-Át[]•[āļÁçæÁācîÁæ)åA;-Áh\*•Ás@eðiÁEÈHA; Ási^]c@DÉA;ç^¦|æêji\*Ás@Á^•[`¦&rAj:^Ási;Á {^åã { Át¦æaši^åÁæ)åA;-Áseç^¦æt^Ási^]c@AFGA; ÈATæðji^ÁslæêA;-Áşæðaæaà|^Ás@a&si}}^••AýEEĚA; Ás[Á;Á;DÁ \*^}^!æaļ^Á}å^!|að•Ás@Áæ)åA(\*•[`¦&rAf;&ææråAsæzÁæjÁseç^!æt^Ási^]c@A;-ÆHA; ÈA

Õ^}^¦æ|^Át@^Át[]•[āļÁt, ææ^¦āæd+Átsk^Át¦^ˆā:@Ási[], }Át[Ási[], }ā:@Ási|æ&\Átāc Áræ)å•Á, @B&@At¦æå^Á \*¦æå`æ|^Át[Áç^\î^Ásæ\Át¦^^Át[Á^||[],ā:@Át¦^ÊAj}^Át[Át, ^åã { Áræ)å•ÈA/[]•[āļÁt, ææ^¦ãæh+Á [&&ææt];}æ|^Át[}cæj;^åÁtiæ&^Ásiæk\Áz;^A\*atdEACE^ætA[-Ási[], }Ea|æ&\Ás;å`¦æe\*åÁræ)å•Á, ^¦^Á ^}&[`}c^¦^åÁsæAs^]c@/Ás^c, ^^}Áæ]]¦[¢ā;æe\*|^ÂiÁt, Át[ÁFÍÁt, Ás^|[], Á;æeš¦æ4A\*i[]`}åÁ^cç^|ÁsjÁ •][¦æå&Af[&ææt];}ÈÁ

#### ''&'% 5V][X`gi`ZUHY`gc]`g``

Á

CB;ÁszSáā Á ઁ |-æc^Á [āp-Áse•^••{ ^} oÁ} & [č] o^\^åÁ; A' [ðiāa|^Á([Á[, ÁszSáā Á, [č] åč & ði, \* Á, [c^} čáæ Á, ác@) Á c@ Á æ) åˆÁ;æc^i ãeop-Á, [][•^åÁ[¦Á¢&æçææti]}ÈÁV+ði \* Ás@ Áçæ) ÁÓ^^¦•Á; ^c@ åÊÉCCi Á, Ás@ Á FHÁ •æt] |^•Ác^•c^åÁ¢@ãa ãc^åÁszÁçãi |^} oÁ^æ&cati} Å ãr@ði Ás@ Á,æbæti ^c'¦•Á; -Á[c^} cãeopÁszBáā Á č |-æc^Á[āp-ÈÁ T[•oA, Ás@ Á^{ æðijā] \* Áæti] |^•Á¢@ãa ãc^åÁ,ãpÁti Á |ði @Á^æ&cati} • ÈÉÁ Õ^}^¦æ¦îÊ\$x@?Á,[ơ?}cãæ‡Áæ&ããÁ`|ææ^Á[ḁ̄•Á,^¦^Áæ••[&ãææ^åÁ,ãc@óx@A`¦^^A.Á&æa\A`¦^^Áāj^Á{[Á { ^åã { A`¦æaj}^åA`Aæ}å•A`}&[`}ơ\¦^åA∞Açæãæà|^Áå^] c@Ax@[`\*@[`dw@A;l[-a]^ÁÇa`o´A^}^¦æ|^Á æà[ç^Á.ĚĂ,Áw^|[, A`¦[`}åA/^ç^|DÁæ)åAx@Aàæeæ‡Á,āj^¦æ‡Á&|æôA;æe^¦ãæ‡ÉV@A&|æÂs[æ´A;[ơA;l[][•^åÁ -{¦Á^¢dæ&cāj}ÈĂ

# ' " Á G]hY WebhUa ]bUh]cb

Þ[Á,¦^çãį`•Á•^•Á,@B&@Á,æÂÁ@æçç^Á&[}ææt,åjæætåÁ@AÁã&Á@æçç^Áå^^}ÆåA\*}œæt,åÅæ}åÁæ}åÁ;[Áçãa`æ¢Á •ãt}•Á,~Á&[}ææt,äjææt,jÁ,^¦^Á/~&[¦å^åÈÁ/@AÚ!^|ã,äjæt^ÁÔ[}ææt,äjæet,åÅÔ[}ææt,äjæet,åÅ

- •Á Q,Á^|ææāj }Å{ [Åæ}å å [Åæ; å ææ; ā] ææ; ā] ææ; ā] ææ; [Å] ææ; ā] ææ; [Å] æ; [Å] æ; [

### '''(Á ; fci bXk UhYf<sup>·</sup>

#### '''('%` < mXfc[Yc`c[]WU``WcbhYIh`

OEcāābāadyÁ\*¦~~~&A\*¢]¦^••āį}•Á&[{]¦ā\*^Á~að]]¦[¢ā[æ~¢^[×]ÁÇDÁ{æ^{i]\*}, 288^Á\*¢];^>å\_æ~\A{[æ+A Ç\*|[`}å\_æ~¦Ë^å/å&æ{•DÁ[&æ~\*å/Áæ&|[••Á∞A\*ā\*A{[Aæ&ājāææ^Á∞A{[&æ\*\a]\*A[~A&ææd^ÈQA&[}b`}&aā]}Á jāc@Á©ā\*ÊÅjæč¦æ#\\*¦~æ&^Á\*¢]¦^••āį}•A{[~A\*\[`}å\_æ\*\A&&A{[&æ\*\*å/A\*[&æ\*\*å/A\*]\*A]\*A[&ã©3)A{©A d[][\*¦æ];@Ba#|^Á[, Á[]];0\*A{[Ac@A]['c@A[~Ac@A\*ã\*Aæ••[&ãæ\*\*åÅjão@ADOUUÁFIÁ,^dæ)åÁ &[{{č}}ā3\*•ÈÅ

ŒÁţœa¢Ą,-Á∓€Ás[¦^•/Ásd^Á{[&æz^åÅ;ão@3;ÁŒĚÁ{{Ą,-Ás@A`ãz^ÈAU-Ás@•^Ás^}ÁÇF€DÁs[¦^@;|^•Êą,}^ÁÇFDÁsA |[&æz\*åÁ[čœgÁ;-Ás@Á;ãz^ÈACE[Á^{æ3;j}\*As[¦^•Ásd^Á[&æz\*åÁ{[Ás@Å;^•ŒÊ\*^}^\ae|^Å;ão@3;Ás@Á[; ¦ãå\*^|j];^Ás@æzÁs[¦å^¦•Ás@Á;^•c^¦}Ás[č}åzsîĄ;-ÁT[[àæa|ÁÔ¦^^\Á?[[å]|æ3j)ÈÂU^ç^¦æqÁs[¦^•Á;^Á 憕[Á[à•^¦ç^åÁsaj]]¦[¢ã;æz^|`ÁHÁ{Á[¦œ34;-Ás@Á;ãz^Ásashæa&^}oÁ{[Ás@Á{[;}•@3)A;-ÁU[œ;çã|^ÈÁ

V@¦^Áa;Áä^|^ÁţÁţ}|^Áa;Áţājāţæ¢4@妿`|3&A&[}}^&cāţāć Áa^ç,^^}Áx@Á`}&[}•[|āāæe^åÁæjåÁ æ`ã^¦ÁæjåÁţ[c^}c⿢4\*¦[`}å,æe^¦Áeč`ã^¦•ÁşiÁc@Áãå\*^|āj^ÈAP[,^c,'ÈÅ;¦^c,āţ`•Ási¦ājāj\*ÁşiÁc@Á [[&ææāt]}Áæ{[`}åÁc@Át], }Áţ-ÁÚ[co•çāļ|^Á\*\*\*^•ók@Áã[¦^•Át&æe^åÁæt]]¦[¢āţæe^å/Ået]]¦[¢āţ @a^|^Át[Ása^Áşi•cæt]^åÁ;ãc@siÁc@Át@ætt[],Á}&[}~āj^åÁ%[æecætA\*ætjåAæč`ã^¦Áx@æcÅ[``|åÁsa^Áða^|^Át[Á ^¢c^}åÁt[`c@;ætåÁt[Ási&]\*å^Áx@Át[[[àæt]AÔ;^^\ÁQ[[[å]]æsiÁ\*A&cāt]}Á;Áx@Áã\*ÈĂ

#### '"("& G]HY [fci bXk UHYf"

Á

Ô[}åãdā[}•Á,ãc@a,Ác@~Ácĕčă^¦Á,^¦^Á`}&[}-āj^åÁse&¦[••Ác@A^ãc^ÊĂ,ãc@á,[Åsã\*&^¦}āa|^Asaā\_^¦^}&^AşÁ \*'[č]å,æc^¦Á@ æaåÁ,[c^åAsa^ç\_^^}Ác@A^@edpl[,ÁsejåAsa^^]Á,[}ãi[']ä\*Á,1a?:[{ ^c^\•ÈAÕ';[č]å,æc^\A |^ç^|•Á,ãc@a,Ác@a\*Áečăa^káe^Á\*^}~\adpliceAseA(!A,^ædÁc@A([][\*']a3;@a&edpá\*`|-æ&^Á,ãc@á;æaäneaā]}•ÁsjÁ \*'[č]å,æc^\Á,ãc@a,Ác@^Áãc^Ác4/Å&;[][,3]\*Á^æe[}æ¢Áæað,-æ¢lÊÅ,@a&e&saae^•Ác@acA\*`|-æ&^Ás3ae \*'[č]å,æc^\Á,ãc@a,Ác@^Áãc^Ác4^Á&](•^|^Áse•[}&ãece^åÈÁ

V@Áţ}|^Áãţ}ãã&æa, dá &&`!!^} &^Áţ-Át![`}å, æc^!Á,[c^åÁå`!āj\*Ác@Á@å![\*^[|[\*ãæa¢&e^++{ ^} dÁ , æe Áæe • [&ãæec\*åÁ, ãc@Ác@Áāj^Ëţ ^åã { Át |æāj ^åÁ`}&[} • [|ããæc\*åÁ`ædc [•^Á;æ)åÁd æč { ĚQE Á &[}-ã{ ^åÁå`!āj\*Ác@Át'![`}å, æc^!Áæe • • • { ^} dÉc@Áājå`!æc\*åÁ;æd å é ÁæjåÁt ædā ^ Akjæč { ÉQE Á &[}-ã{ ^åÁå`!āj\*Ác@Át'![`}å, æc\*!Áæe • • • { ^} dÉc@Áājå`!æc\*åÁ;æd å é ÁæjåÁt ædā ^ Akjæč Å &[}-ã4[^Åt[!{ ^åÁc@Á\*•ãã`ædÁaæe ^{ ^} dÁæt ^ Akjæč Åkjæč Åá ][c^}cãædÁa` ^Át Ác@Á@t @Áx[}c\*} dÁ ácÆa àÁkjæč Åá

Õãç^} Áv@enakó@ Á\*¦[`}å, æe^¦Á^ç^|Áa Á^]æaãç^|^Á }ã[¦{ Áne&i[••Á@ Á ã×Êžán Á\*}|ã ^|^Áv@enaké)^Á •ã}ãaBaa) ofaæ^iæhá'i[`}å, æe^iÁ([ç^{ ^}o%sči!^}d^A, &ssi]+ d^A, &ssi] Áv@AT[[àæhÁÔ!^^\ÁØ][[å]]æaj Á ][!cāţ}Á, -Ás@ Á ã×ÈÅUã}ãaBaa) ofaj &i^æ ^•ÁsjÁc@ Árç^|Á, -Á\*i[`}å, æe^iÁ{II[, ā]\*Áæaj-æhÁ?c^}o Á •\*\*\*^•o Ás@enakó@ Á\*![`}å, æe^iÁ^&@eh\*^Á^•][}•^Áá Á&I[•^|^Á^]æe^åÁs[Á]!^&aj åæaaj }Ása) åÁ\*i-æ&AÁ •\*\*\*^•o Ás@enakó@ Á\*![`}å, æe^iÁ^&@eh\*^Á^•][}•^Áá Á&I[•^|^Á^]æe^åÁs[Á][A^i^&aj ãæaaj }Ása) åÁ\*i-æ&AÁ •\*\*\*^•o Ás@enakó@ Á\*![`}å, æe^iÁ^&@eh\*^Á^•][}•^Áá Á&I[•^|^Á^]æe^åÁs[Á]!^&aj åæaaj }Ása) åÁ •\*\*\* •o Ás@enakó@ Á\*![`}å, æe^iÁ^&@eh\*^Á^•][}•^Áásá A&I[•^]^ Á^]æe^åÁsaj ãæaaj }Ása) åÁ •\*\*/\*\* • æe^iÁ\*}BjÁ![{ Á]•d^æ¢ Ásæas&@ ^}ofasd^æ BĚÕ![`}å, æe^iÁ^?ofasd} ·\*asa\]^Á{[ II[, ā]\*Ás@Ása^••æaaj }A\_i-Áæaj ~æhÁj ã@ás@AA(i [•cA ã] ãaæaa) of@ å![I[\* ã&æhÁ\*][`}å, æe^iÁ [`d]`o Áã ^|`Ás[Ása^Aæ•[&ãæe\*aÅ,ão@ásã^&oA?çæj[Cæ]•]ãæaaj }Á'[{ Á;æe`!^Ása+^æ Aæj åÁ \*\*I[`}å, æe^iÁsæe^-4[, Ás] c Ás@A ã\*o As@A^A(aab) +ÁsIæj æt ^Áj ~e Aæj ÅA[[`]`}à

Ùã⁄ Á	heđeì đogeci á Çiáđepödá	€ÎBEJEDEEELÁ ÇIÁCEPÖDÁ	Fheejideeeiá Çiádepödá	FïbFGED€€EIÁ ÇIÁCEPÖDÁ
ÖŠÚFÁ	€ÌH€Á	€ÈÉÍÁ	€ÈÈÌÁ	€ÌÈHÁ
ÖŠÚFŒÁ	€ÈĜÁ	€ÈGÍÁ	€ÈGHÁ	€ÈJÁ
ÖŠÚGÁ	€ÈCHÁ	€È€Á	€ÈÈÌÁ	FÈGÍÁ
ÖŠÚHÁ	€ÌÈFÁ	€ÈÉÍÁ	€ÈHÁ	€ÌÍÁ
ÖŠÚHDEÁ	€ÈFÁ	€ÌH€Á	€ÌH€Á	€ËGÁ
ÖŠÚI Á	€ÈGJÁ	€ÈGJÁ	ÆÈÈÌÁ	FHÈËÁ
ÖŠÚÍ Á	€ÌHÁ	€ÈGJÁ	€ÈFÁ	€ËÍÁ
ÖŠÚÎ Á	€ÌHÁ	€ÌHÍ	€ÌHÁ	FÈFJÁ
ÖŠÚÏ Á	€ÈGJÁ	À ĽŒ	€ÈGÍÁ	FÈ€JÁ
ÖŠÚÏ ŒÁ	€ÈCHÁ	€ÈGHÁ	€ÈFÁ	€ËJÁ
ÖŠÚÌ Á	€ÌÈHÁ	€ÌÈGÁ	€ÌHÌÁ	FÈFÎ Á
ÖŠÚÌ ŒÁ	€È FÁ	€ÌE€Á	€ÌHÏÁ	FÈCÌÁ
ÖŠÚJÁ	€ÈFÁ	€ÈGJÁ	ËÁ	€ĽĚHÁ
ÖŠÚF€Á	€ÌLGÁ	€ÌHÌÁ	€ÌḦ́Á	FÈFÁ
ÖŠÚF€ŒÁ	€ÈG Á	€ÈÍÁ	À ĐĐ	FÈHÎÁ
ÖŠÚFFÁ	€ÈGIÁ	ÈÐHÁ	€ÈFÁ	€ÌE€Á

#### HUV`Y'' !%; fci bXk UHYf``Yj Y```

#### '"("'; fcibXkUhYf`kUhYf`eiU`]hm

V@Á^•ĭ|œĄ(-Ác@Áæ;]|ā)\*Áæ)åÁæ)æf°ārÁj¦[\*¦æ;(Á}å^¦œa\^}Á`\*\*^•óAœæA\*¦[ĭ}å,æe^\Á´æa}Å ,ãc@)Ác@ÁT[[àæ||ÁÔ¦^^\ÁØ|[[å]|æäjÁQ;^æÁc@Árãe^DÁarÁs]ã&æ||^Á¦^•@áæ)åÁr^}^¦æ||^Á`ãææà|^Á[¦Á ã¦ãræaā;}Áæ)åÁå[{ ^•ca&Árq[&\Á\$;Áā]^Á;ãc@ÁCEÞZÒÔÔÁG<del>EEE</del>Á^˘ĭã^{ ^}œA

Õ¦[`}å, æe^\¦Á`æ¢aĉ´Á, æo·Á\*^}^¦æ¢|^Á&[}•ã:c^}cÁsæå[••Ás@A´ãz^É4@], ^ç^¦ÉasaÁ|â\*@o/\$aj&l^æ;^Á§iA;PÉA ^|^&c1a&æ¢/&[}å`&cãçãĉÁÇDÔDÉ&@[¦ãâ^Ė=`|~æe^Áæaãjaa)åÁ;æ4bj¦Á&æaãj}A&[}&\}dæaãj}•Á,^¦^Á,[c^åÁt[Á [&&`¦Á,ãc@ábj&\^æ^å/å^]c@ábjÁs@Áseັšã^¦ÉA

P[,^ç^¦Ác@•^Áçæäææa‡}•Á,^!^Á\*^}^!ælļ^Á |ællf^Á |ællfæbben å Áæbh Á'} |ælf^Á |ælf\* / Åt[Á^•`|c4sj Áæben Á'; |ælf\* / Åt] Å ^~^&or Á;} Á':[`}å, æen ¦Á`æljæc\*Á @;`|åÁ; ä¢z]\*Á; Á\*@ede|[, Áæb}åÅsh^] Á; æen ¦•Á; &&` ¦ÈkQA%ar Á\*¢] ^ &&råA co@edeAæah^``æen Á,^`dæljæz Å; Áæb}^Á |ællf\*@; Át [l'^Áse&ää a&A\*`|-æ&r Ásb}åĐ; Af ![`}å, æen !•Á; [`|åÁ; &&` ¦Á a\* ¦z]\*Ár@(A;]^!æeat]}æbA; @een ^A; -Á; [:\+Ásb`^Át[Áseat`}åæb} &A A; -Á @;||Ár@[`\*@]`cAso@A[zA;![-an Ásb}åA c@A/]ææat^Ar & e&n •Á; -Ást] æbA; @een ^A; -Á; [:\+Ásb`^Át[Áseat`}åæb} &A A; -Á @;||Ár@[`\*@]`cAso@A[zA;![-an Ásb}åA

Væà|^Á+ECA{`dā},^•Á@`Áā}åāj\*•Á{,~Á@`Á'¦[`}å, æe^¦Á`æ¢áĉ Á, @38.@Á{¦{ ^åÁ∞@ Áaæ•^|ã}^Á, æe^¦Á ``æ¢áĉ Á8[}åãaāĮ}•Á{¦Á∞@ Áã~ĚÁ Á

U-Á,[c^Á;[{ Ác@:Á\*;[`}å, æc^¦Á`æ¢aĉ Áa)åā)\*●ÁseÁo@:Á&[}&^}dæaā]}Á,-Ásaã•[|ç^åAs[]ĚÅ, @a&@áseÁ ^|^çæe^åÁ^|æaãç^Áq[Á;[•oA,æc\*¦æ4Á,æe^¦ĚAOE-ÁseÁ^•ĭ|dÊ4;[{ ^Áss[}Á+[&•Áse)åÁræaä)a]\*Á;æâA,&&\*¦Á æ+[ĭ}åÁs@:Á,^¦ã;^cv¦Á;~Ás@:Ár¢&æçæaã)}Ásc4~æ4sič¦ä]\*Á;]^!æaā]}æ4Á,[¦\•ÁseaAseAiA) A

# '') Á GifZUVVY'k UHYf''

#### '')'% 7 UHVN(a Ybh'UfYU'

V@Ááã^ÁãaÁ[&æes^åÁjão@át[[[àæe|ÁÔ¦^^\Á&æes&@]^}oÁea}åÁÙ@^}•ÁÔ¦^^\Á`àÁ&æes&@]^}oÁea†àÁ

V@Á`āč^ÁārÁ[&æet^åÁæ}]¦[¢ā[æet^|^Á\Á{Á]•d^æ{Á'[{Á@A',[`c@4, 4T[[àæ||AÔ|^^\ÈV@Á`ãc^Á@æ•Á æ4,`{à^¦A(, 4&æe&@&i|æaj)•Á, @a&@4(, ^!\*^Á5)q[Ás@A', æaj)Áæt'i&&`|c`¦æ4%i|æaj)•Á;}•ãc^ÈÁ

V@Á^|ææãç^|^Á|ææÁti][\*¦æj@Ái,∻&@Áiæ^Á`\*\*^•@Áoœæáki[•oÁ`¦ææAA`}Á;~Æsi,adaæ\*•Æsi,á&@Á •[ā≉Á;¦Á+[, •ÁtiÁT[[àæ4|ÁÔ¦^^\ÈÁŬ`¦æ&A^Á;[}åäj\*Æsi[^•Á;&&`¦Áææ^¦Áæaj;æ4|Å;@}Ás@Æsi;A ¦æ\*Áti¦Ás@Á;æ)åÁ@æ•Ási^^}Ár¢&^^å^åAti¦Ás@Á;@e4|[、Át¦[`}å,æe^¦Ásæai/^Á/æ&@•Ás@Á`¦æ&AÈÁ

VãaæpÁ{[, •Á^} ♂\Á@ Áse" ¦&X`|c` ¦æpÁsi¦æaj •Áslæç^\•ã,\*Á@ Á ãe^Áse}åÁslæç^|Áçæ? ã,\*Ásã œa) &^•Á] Á c^æ( Á å^] ^} åæ) o{{\} Ásj åãçãa `æpÁsãa^Á@ ã @ÈÉ/@^^Á{[[åÁ\*æe^•Áse}^Á,¦^•^} o{{\} Ás@ Á ãe^Áse}åÁse}^Á •^åÁ[Á ¦^\*` ]æe∿Á@ Á{[, Á; ÁsãaæpÁ, æe^¦•Ás[Áse\*¦&X`|c` ¦æpÁæ) åÁs[Ás@ Á, ^•o4\Ás@ Á ãe^ÈÁ

V@ ÁÞÙY ÁÖ^]æd(^}of(-ÁÚ¦ā[æ\*ÁQå\*•dā\*•522ā@ ¦ā\*•Á@æç^Á^\*\*^•c\*åÁs@••^Á\*æe\*•Ás\*Á(\*-af(]^}Á -{¦Áã=@á/æ••æ\*^ÈĂ

#### '') '& GifZUWY'k UhYfeiU`]hm

Yæc^¦Á`æ¢ãčĄ́ç^¦Ás@A´ãc^ÉA`{{ æċã^åAşi,Á/æài/ÁHÜHÉX^}^¦æ¢î^A`@A\$@AÂ/¢çæà;cÁ/,^^åAÛ@3AA Ô[`}&ãqA[¦ÁCEÞZÒÔÔA\*`ãa^|ãj^•Ąí¦Á^&[{ { ^}åæaāj} •A[¦ÁA&&@Aj;æbæt;^c^¦A[^&æc^¦Á[^&æc^]A ``æ¢ãčÁ^•`|o=Áşi,Á/æài/AHÜHA´čdāj^Aô@A3jåãj\*•A[,A\$@A\$aæo^|ãj^Á;æc^¦Á`æ¢ã襴,[}ãq[¦ãj\*Á;@3&@Á -{¦{ ^åÁs@A\$aæo^|ãj^Á;æc^¦Á`æ¢ãčÁ&[}åãaj} •A{[¦Ás@A´ãc^ÈA

 $V@\dot{A}|^{8}d = 3d = 4 \\ V@\dot{A}|^{8}d = 3d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V@\dot{A}|^{8}d = 4 \\ V$ 

Öã•[|ç^åAf₄¢^\*^}A, æ Á^&[¦å^åAæ Áå^ã,\*Á|ã @cî^Á[, É4,ão@ka@ Á+|ã @c%å^]¦^••ā;}Af, [•c4jã^|îÁ ¦^-{^&cāj\*Áaæ ^Á{[, Á+[{ Á+[č}å, æc^\;Á, @a&@bá Ác]ã&ædjî Áç^\:^Á[, Áş,Ásã •[|ç^åAf₄¢^\*^}ÈÁ

Þ`dār}orÁ,^!^Aá|ðā@lî^Ár|^çææ^å/&jÁ[{ ^Á[&ææā]}+Áæe Ár¢] ^&c^å/&i`^Á[Á@^Á`¦![`}}åāj\*Á æ\*¦ā&`|c`¦æ‡Áæ&cāçãað+ÈAQÁ\*^}^¦æ†É&@^Áãc+A`¦~æ&^Á;æc^¦Á`æ#ãc´Á;æe Á^|ææãç^|î Át[[å/æðjåÁ^-∤^&orÁ &[}åãaã]}+Á&@ebæ&c^¦ãa ca&A,-Áæ\*¦a&`|c`¦æ‡Á%¦æãj+Á%jÁ[, Árãj\*Á+[[å]|æãjÁ^}çã[]{ { ^}orÈÁ

### ' ')" <sup>·</sup> : `ccX]b[ <sup>·</sup>

OEÁ [[åÁse)æ^•ã Á&[}&|`å^åÁs@een Áse Ás[][][•^åÁså^ç^|[]{ ^}o^j [`|åÁ^•`|o4sj Ásensa ^&j Á[[åÁ |^ç^|•Á[-Á]]Á[ÁGï∈Á[{ Áseen Áse Áse Áse Ás[`}åæ^ Á[-ÁseA^A[-ÁseA^É],@B&@ý][`|åÁ@eeç^Á,[Á∄]ã&Bæ)oÁ æseç^¦•^Ás[]æ&dÉA

# HUV`Y'' !&`; fci bXk UhYf`k UhYf`ei U`]hm

Á	ÖŠÚFÁ	ÖŠÚF ŒÁ	ÖŠÚGÁ	ÖŠÚHÁ	ÖŠÚH ŒÁ	ÖŠÚI Á	ÖŠÚÍ Á	ÖŠÚÎ Á	ÖŠÚÏ Á	ÖŠÚÏ ŒÁ	ÖŠÚÌ Á	ÖŠÚÌ ŒÁ	ÖŠÚJÁ	ÖŠÚF €Á	ÖŠÚF €ŒÁ	ÖŠÚFFÁ
] PÁ	ÍÈFGÁ	ÏËÌÁ	ÎÈÈGÁ	ÍÈFIÁ	ÎÈHÎÁ	ÍÈÍÁ	ÍËIÁ	IËÍÁ	ÍÈGÍÁ	ΪÈH€Á	ÍËIÁ	ÌĚ€Á	ÍÈÈHÁ	ÍĚÏÁ	ΪĖ̈́JÁ	ÎÈG€Á
ÒÔÁÇ; ÙB&{ DÁ	€ÈÈJÌÁ	GÈG΀Á	€ÈGÍHÁ	IËÏ€Á	ÎĚÍ€Á	€ÌHIJÁ	€ÈÌ JÁ	€ÈFGGÁ	€ÈÍ€Á	IÈG€Á	€ÈHÍÁ	€ĽĚHÌÁ	€È€JJÁ	€ÈEGGÁ	FË FÍ Á	€ËIÏÁ
ö∪áÇ;*EŠDÁ	HÈLÁ	GÈJÁ	GÈ€FÁ	HÈH€Á	GĚÏÁ	GÌÈ€Á	hèciá	HÈ€Á	GÈÌHÁ	HÈHÌÁ	ÎÈFÎÁ	ΪÈΗÁ	IÈFIÁ	IÈ€Á	HÈGHÁ	GĚÍJÁ
∨^{] <b>ÈặçÔ</b> <sup>[</sup> <b>D</b> Á	FJÈ€Á	<b>GFÈGÁ</b>	FJĚÁ	GÍ È∓Á	GĚÁ	FJÈÁ	G€ÈGÁ	FÌÈÁ	G€ÈHÁ	FJĒÁ	FÌĒÁ	GGĚÍÁ	GFÈGÁ	FÌĚÁ	FÌÈ€Á	FJËÁ
O≣∖æpäjaĉÁÇ;*EŠŠÁæeÁ ÔæÔU <sup>H</sup> DÁ	FÁ	Í G€Á	I GÁ	HÁ	fî gá	HÁ	ΪÁ	ŁFÁ	HÁ	HÌÎÁ	HÁ	gí há	FGÁ	HÁ	I F€Á	ΙΙÁ
Ô@[¦ãå^ÁÙĭ ~æe^Á Üæa‡iÁÇĴŠKÙU⊧DÁ	€ÈË FÁ	FÈGFÁ	HÈHÎ Á	ÍÈÌJÁ	FŒÏ€Á	ÌÈ€€Á	FÈHÁ	GÌHÁ	FÈFI Á	HÈËÏÁ	GÈ€€Á	HÈÈÌÁ	€ÈIÁ	€ËJÁ	FÈ€HÁ	FÈ€HÁ
Öãi•[ ç^åÁ OĘ~{ā)ā{ÁÇ;*EŠSDÁ	€ÈÎÁ	Ł€È€FÁ	€È€JÁ	FÈÈÌ Á	€È€ÎÁ	€ÈÆÁ	€ÈJÁ	FÈHUÁ	€ÈŒ€Á	€ÈCCÁ	€ÈFHÁ	€È€ÎÁ	€ÈÈÌÁ	€ÈFIÁ	Ł€È€FÁ	Ł€È€FÁ
Öãr•[ ç^åÁQ[}Á Ç;*ESDÁ	ΪÈΗÁ	€È€FÁ	€ÈHÁ	ΪÈŦΪÁ	F€ÌÈ€Á	FÈÈÈÁ	GÈ€HÁ	IIĒ€Á	ÍÈGÁ	FÈHÏÁ	FÈGÁ	€ÈGÁ	ÌÈÌHÁ	HEÈÏÁ	Ł€È€FÁ	G€ËE€Á
Öãi•[ ç^åÁ Taa}*aa}^∙^AÇ;*EŠDÁ	€ÈÈÈÌIÁ	€ÈEI€ ÍÁ	€ÈEHIÁ	€ÈÉÍGÁ	€ĚÌGÁ	€ÈE€HÁ	€È€EÌÁ	€ÌH€€Á	€ÈEÏÎÁ	€È€JGÁ	€È€FFÁ	€È€FIÁ	€È€€Á	€ÈEHFÁ	€ÈÍ GÁ	€ÈÈÌIÁ
Öãr•[ ç^åÁse∳•^}a&Á Ç;*EŠDÁ	Ł€È€€ FÁ	Ł€È€€ FÁ	Ł€È€€ FÁ	€È€€ÎÁ	Ł€È€€ FÁ	€È€€GÁ	Ł€È€€ FÁ	Ł€È€€ FÁ	€È€€FÁ	Ł€È€€ FÁ	€È€€FÁ	€È€€GÁ	Ł€È€€ FÁ	€È€€GÁ	Ł€È€€ FÁ	€È€€JÁ
Ôæ†&ã{ ÁÇ; * EŠDÁ	ÍÁ	F€JÁ	FÁ	ЫÁ	ÎÎÁ	HÁ	ŁFÁ	FGÁ	ÎÁ	HÎÁ	GÁ	FHÁ	F€Á	HÁ	F€ÏÁ	FΪÁ
Tæ*}^∙ã{ÁÇ;*ĐŠDÁ	GÁ	ÍÍÁ	ΪÁ	ÌÎÁ	F€ÏÁ	١Á	ŁFÁ	GHÁ	١Á	Ì€Á	FÁ	GÁ	ΪÁ	ŁFÁ	I GÁ	FÌ Á
Ù[åã{ÁÇ;*ĐŠDÁ	G€Á	H€FÁ	FJÁ	Ì€FÁ	FFJ€Á	IÎÁ	HHÁ	FÎ Í Á	G€Á	Ì FÍ Á	G€Á	FÍ €Á	FÏ Á	F€Á	FÎ GÁ	ΪJÁ
Ú[cæ∙ã{ÁÇ;*ĐŠDÁ	GÁ	FÌÁ	GÁ	Η̈́Á	ÍIÁ	ŁFÁ	GÁ	F€Á	FÁ	ΙÍÁ	FÁ	١Á	GÁ	ΙÁ	FÍ Á	F€Á

Š[&æaā)}Á	Öæe^Á	Vãa∧Á	Yæe^¦Á  ^ç^ ÁQ;Á 00₽ÖDÁ	Ü^å[¢Á Ç≵XDÁ	Ò ÈÔ ÈÁ Ç: Ù B&(DÁ	V"¦àãããcÂ Ç⊃VWDÁ	] PÁ	∨^{]ÈÁ ÇÔ <sup>I</sup> DÁ	ÖUÁ Ç;*EŠDÁ	VÙÙÁ Ç:*EŠDÁ	ÞÁÇ,*ЊDÁ
ÙY FÁ	FÏÐFGB€EIÁ	Ql&[{ ãj*ÁTããÁxãã^Á	ÞŒÁ	ÞŒÁ	GÈÌÍÁ	ÞŒÁ	ÞŒÁ	GÈÉÁ	IĚÍÁ	ÞŒÁ	ÞŒÁ
ÙY GÁ	FÏÐFGBÆEIÁ	Qu&[{ ãj,*ÁTããÁxãã^Á	ÞŒÁ	ÞŒÁ	IÈEJÁ	ÞŒÁ	ÞŒÁ	GHÈ€Á	ÎÈJÁ	ÞŒÁ	ÞŒÁ
ùy há	FÏÐFGBÆEIÁ	Qu&[{ ãj,*ÁTããÁxãã^Á	ÞŒÁ	ÞŒÁ	FËHÁ	ÞŒÁ	ÞŒÁ	GÈHÁ	ÍËJÁ	ÞŒÁ	ÞŒÁ
ÙY I Á	FÏÐFGBÆEIÁ	Q1&[{ ãj*ÁTããÁxãã^Á	ÞŒÁ	ÞŒÁ	ÍÈÍÁ	ÞŒÁ	ÞŒÁ	GHÈÁ	JÈGÁ	ÞŒÁ	ÞŒÁ
ÙY Í Á	gi <del>befbe</del> í á	Uĭc*[ậj*ÁPãt@Á cããa^Á	€ĚÎÎÁ	GIÁ	HJĚÁ	ÍÁ	ÌÈEÏÁ	À Ġ ĐÁ	î kəhá	GÁ	€ÈHÁ
ÙY Î Á	GI⊞967FB9€ÍÁ	Uĭc*[ậ),*ÁPã®(Á cãã^Á	€ĒÌHÎÁ	Ġ I Á	g Èhá	FJÁ	ΪĚΗÁ	H€ÈGÁ	ÎÈEÎÁ	ÞŒÁ	ÞŒÁ
ÙYÏÁ	gi <del>befbe</del> í á	Uĭc*[ậj*ÁPã®Á cãã^Á	€ĽĚ€ÏÁ	GGFÁ	JÈHGÁ	FFÁ	ÏÈIÁ	GJËÁ	ΙËΪÁ	ÞŒÁ	ÞŒÁ
ÙY Ì Á	GI⊞967FB9€ÍÁ	Uĭc*[ậ),*ÁPã®(Á cãã^Á	€È€ÏÎÁ	FJI Á	€ËHÁ	F€Á	ÏÈH€Á	HÈÁ	ÎÈ€Á	ÌÁ	FĚÁ
ÙY JÁ	GI⊞967FB9€ÍÁ	Uĭc*[ậj*ÁPã®Á cãã^Á	€ÈHÍIÁ	G€Á	FÌÈÁ	FÍ Á	ÏĚ€Á	HFË Á	ÍÈÏÁ	GGÁ	€ÌÉÁ
ÙY F€Á	GI⊞967FB9€ÍÁ	Uĭc*[ậ),*ÁPã®(Á cãã^^Á	€ÈHHÁ	G€ÏÁ	HÈEJÁ	ÌÁ	ΪÈŦΪÁ	HQÈLÁ	ÍÈFGÁ	GÁ	€ÌÌÁ
ÙY FFÁ	gi <del>befbe</del> í á	Uĭc*[ậj*ÁPãt@Á cãã^Á	ÞŒÁ	<b>GFGÁ</b>	ĠĖÁ	FÁ	Ϊ È FÁ	H€ÈÌÁ	ÎÈEIÁ	ÞŒÁ	ÞŒÁ
ùy fgá	GI⊞967FB9€ÍÁ	Uĭc*[ậj*ÁPã®Á cãã^Á	€È JÏ Á	FÌ JÁ	ÌĔ€Á	F€Á	ÏÈGÁ	HFËÁ	ÍÈEÁ	ÞŒÁ	ÞŒÁ

# HUV`Y`' !' `Gi fZJVY'k UhYf`ei U`]hm

ÚÁÇ; \*EŠDÁ

ÞŒÁ

ÞŒÁ

ÞŒÁ

ÞŒÁ

€ÈFFÁ

ÞŒÁ

ÞŒÁ

€È€ÍÁ

€È€IÁ

€È€GÁ

ÞŒÁ

ÞŒÁ

# ''\*Á **≕adUW#g**`

Ú[c^}cãa‡Á,^\*æãā,^Áā;]æ&orÁ;}Á{[a]ÉX;[č}å,;æc^;kéa;åÁ\*;+æ&^Á;æc^;Á+[{Ác@A;h;[][•æ‡Áa;&;čå^kA

''\*\*'% Gc]`g`

- •Á Ù@¿\okav\{ Á\[•ā]}Á[.Áq[&\]]ā/\okav\{ Á\[•ā]}Á[.Áq[&\]]ā/\okav}åÁ\¢][•^åÁ[ā@A\|æeā]\*Át[Áç^@&Q\^Á[[ç^{ \} œ ÊA •d[&\]]ā/\\alpha\á[æev\;ãætp-Áæ)\alpha\á[x^\*^cæeā]}Á\{[çætÊA
- •Á Ô[{] æ&cāj} ¼ Ác@Ás^^] ^¦Á[ ã;Ád \* &č ¦^Á¦[{ Ác@Á; [ç^{ ^} o, 4; Á;æ\* ^Á; æ&@3; ^¦^ÈÁ
- •Á V¦æ)•][¦œeaā[}Á,~Á[ā¼,}d[Á,^æà^Á^æ^åA[æå•Á^•č|cã)\*Á¦[{Á,~•ãc^Áç^@a&\^Á, [ç^{ ^}œÈA
- •Á Ö`•oÁ\*^}^¦æaāį}Á¦[{Ár¢&æçæaāį}Á,[¦\•Áæ)åÁç^@a&|^Á,[ç^{ ^}oÁ;ç^¦Á\*¢][•^åÁ[ā+ÈÁ
- •Á Ô[}œa; ājæaāj}Á;'[{ÁjājÁj;¦Á&@{ã&aa;Á;]āj|•ĚÁ

#### ' ''\* ''&'; fci bXk UHYf

- •Á Ö¦æ;å[ ` }Áį Ás@ Á` ¦ ! [ ` }åãj \* Á; ær \ Áæà | ^ĚA
- $\bullet \dot{A} = \dot{O}c \left[ \left[ \bullet^{*}\right]^{A} \dot{A}_{t} \dot{A}_{t} = \dot{A}_{t} = \dot{A}_{t} \dot{A}_{t} = \dot{A}_{t} \dot{A}_{t} = \dot{A}_{t} = \dot{A}_{t} \dot{A}_{t} = \dot{A}_{t}$
- $\hat{O}$   $\hat{O}$   $\hat{O}$   $\hat{O}$   $\hat{A}$   ### ''\*'' GifZUWY'k UHYf'

- •Á V@Á; [] [ ^åÁ`æ; ^Á [ ` |åÁæ¢<; ÁœÁ[ &æáÅ[ ] [ \* ¦æ] @ Áæá@ Á ã</p>
  @áæá@ Á
  a' añ ai æj æ' ^Á; -Á` ¦ -æ&^ Á æ
  EV @ A& A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
  A & A
- •Á Ùǐ¦-æ&^Á, æe^\lÁsãa &@ed\*āj\*Á\[{ Ás@ Á`æb\^æ&k{``ab\^æ&k{``ab\^æ&k{``ab\^å&q`^àå, ae^\lá`æb\^å&q`c'là^åÁ æb^æ Ásæ`•āj\*Ás^&\^æ•^åÁ, æe^\lÁ`æ¢ačÁşjÁ, æe^\l¸ æê•ÈÁ
- •Á Ô[}cæţājæaāţ]Áçãæk&@{ã&eq+Ê2@å¦[&ædà[}•Áţ¦Áţc@¦Á&[}cæţājæe^+Á+]{Ásæ&&ãa^}cædÁ
  •]ājæt^ÈÁ
- •Á Ø[[åā]\*Á; -Á@A´āč^Á; @B&@A´ač Á; @B&@A´ač Á^•č |øÁ§ Á\[[åÁā`\Á{[Á cæ-ÉÅ^•č |øÁ§ Á; -Ë=ãč^Á\[[åÁ§]] æSee ÉÁ
  ^\[å^Áq (&)] ā/•Áæ) åÁ? dæ§ Á^àã (^} oÁæ) åÁåæ (æ\* ^Áčč ā] { ^} oÉA

# (''Á K UHY f`VU`UbVY'''

Á

V@Á{I||[, ā] \* Á^&cā[} • Á` { { æbā ^ Ác@ Á, æc^¦ Ábaabaa) &^Êbbaæ ^ å Á[} Ác@ Á5], -{ ¦{ æbā[} } Á, ![çãb ^ å Á5] Ác@ Á ÒT ÙÁQÚ|aa) ão/GEFÌ DDÁ

## ('%Á =bhfcXi VM\_cb'

V@AÖ`}|[^ÁÙæ)åÁÛ ĭæ¦^ÆsA´ãčæe^åÅ,ãc@3,Ác@AÙ@^}•ÁÔ¦^^\Á&æe&@(^}dÂ,@3&@ÁsAæ)Á ^]@{^¦æ‡ÉAd^æ{ Ác@æeA`ç^}迆^Ásã&&@ee\*^•Ásjd[AT[[àæ‡|AÔ¦^^\ÈAYæe^¦ÆsA^ĭĭã^åÅæeA@A ĭ`æ¦^Á{[¦Á •^ÁsjÁc@Á;æe@4]|æ}oÁæ)åÁ{[¦Æsĭ•oA`]]¦^••ã[}Ê5c@Á[æ‡u]iãĉA[-Á;@3&@Á;ã|Aŝa^Æsi]æ;}Á ⊰[{Ác@Á¢cdæacā[}Ásc4~æÆsē•^|-ÈÁ

Yæc^¦Æsákap•[Á[•okko@[č\*@Áçæa][¦ææa]}Á¦[{ Ás@A^¢dæ&aa]}Ákad^æakaa)åÁ,ãr@a]Ás@Aj¦[&^••ā]\*Áj-Á ^¢dæ&oc^åÁæa)å•ÈÄ

V@¦^Áæ'^Ác@^^^Á;æajiÁ\*[ĭ¦&^•Á;æc^¦Á{¦k&@Á´æk¦^Éð;æ{^|^K\*t¦[ĭ}å,æc^\Á\*^]æ\*^Á§iq[Ác@Á;ãuÁ ¦æaji-æ‡|Áæ‡|a]\*Á§iā^&q^Á§iq[Ác@Á;ãuÁæ)åÁæeq^Áæ)åÁ;Á^?æed%a[][¦œa)&^É4;~Ëiãc^Á`}[~Ác@æeÁ%}c∿¦•Á c@ÁÛĭ榦^ÁUãc^ÉA

### ("&Á G]HY k UHYf ï gY

#### ("&"% GUbX k Ug\]b[ 'UbX fYhYbh]cb cZk UhYf ]b a UhYf]U g

OEJ]¦[¢ā[æec^|îÂÊEEE≪ŐSÁ[-Á],¦[&^••Á]æec^¦ÁarÁ^˘˘ā^åÁq[Á],¦[å˘&^Áq[}}^Áq[}}^Áq[A];[åĭ&OHÁ] {æ¢ā[ ັ{A[-Á+HEEEÊEEEÁq[}}^•A[-Á], ær@°åÁnæ)åÅjā[Aŝa^A],¦[åĭ&^åÁqa}}ĭæ||^Á§JÁæ&&Q[¦åæ)&^Á]ão@Áo@Á å^ç^|[]{^}ơ%&[}•^}dĚ

Yão@Ás@Ás¢&^]cā[}Áį-Ás@Á[||[¸ā]\*Ébs@Á(æb);lãc´Á;-Ás@ãrÁ;l[&^••Á;æes\Á,ā|Abs^Á&[}cā)ĭ[ĭ•|^Á ¦^&c&|^åÁæg)åÁæçæanajæba|^Á{[¦Á^ĭ•^KÁ

- •Á V@Á, æ @åÁa; ajáA; |[å č &oÁ, ajÁ@æç^Áæá ([ã č ¦^Á&[] c'} oÁ, Áæ]] |[¢ā; æc'|^ Ã ÈÓE ÁæÁ &[] •^``^} &^ÊA] Á[ÁG ÁT ŠĐ^æÁ, ajÁ¢ ãaÁ@AÛ迦^ÂÛãc/Áæ Á; æc'\Á^æaj, ^åAş Á@Áæ) åÁ ] |[å č &orÈÁ
- •Á Yæc^¦Á•^å/\$j, ká@Aj, |æ&^{ ^} oAj, -Á3j, ^•Áj, ão@j, ká@A^Ë3j, c^¦} { ^} oAj, [}å•Áj, āµÁ^{ æ3j, Áj, ão@j, ká@A , æc^¦Á& & & Aj/ĚÅ

#### ("&"& 8i ghigi ddfYgg]cb

Á

Yær\'ÁārÁ•^åÁ{[¦Áå`•oÁ`]]¦^••ā[}Á{}Áv@A[æå•ÊAq[&\]ā/•Aæ}åA[c@¦Á¢][•^åA`¦~æ&^•Áæ]`}åÁ c@ÁÛ`æ¦^ÂUãr\ÈKÕãç^}Áv@æeAæ|A[æå•Á,ā]A&^Ar^æ/^åÊ{,ær\'Á`•^åA§JAå`•oA`]]¦^••ā[}ÁārÁ &[}•ãå^¦^åA[á`^|^Áq[Áa^A|æãç^|^Â[],ÈQQÁv@áÁ^\*æåA[ázAárÁaAárAko@æeAv@A[æ¢ā[`{ Áær^A[~Á ;ær\'Áæ]]|&Bæaā[}}Á,ā]Aár^Áæ}Áæç^!æ\*^A[~ÁHÁSŠE3)æêĚA

Óæe ^åÁt,}Áæ),Áæ)]|a8ææa‡i}Áaā; ^Át,-Án kÁQ; ¦e Ba æêÁQ;¦ÁF ÁSŠÁ, ^¦Áa æê DÁy;¦ÁF €€Áa æê e Ð ^æh Eko@ar Á\* čæe\*e Á d[[Áæ}]]¦[¢ā‡,æe\*|^ÁF ÁT ŠÐ ^æh EkoAfa; A ^¦cā}^}œk[Á;[c\*Ás@ææA5j Ás@ Á5j ānāæh Ár cæt ^e Át,-Ás@ Á\*¢ chæ&cā‡}ÁGà¦^Á ^¢&æçææa‡i}DDÉåš \* o A\*`]]¦^ • • a‡i}Áa^{ ag å • Áæ^ Áā ^|^Át[Áa^At,[¦^Át]}^¦[` • Ebæ¢cQ` \* @ÁadAar Á^æe[}æaa|^Á d[Áæe • č{ ^Ás@æd5j Á&/¦cæa‡iÁ&[} \* áāa‡} • Á;[Á\*`]]¦^ • • a‡i}Á; á‡IÁa^Á^č čá ^ á\*e Át

OE; Árçæ] [¦ææā] } Áræ• ^•••{ ^} oý, æe Á`} å^¦œah^} Ár @a; \* Áræāj, æaļÁræj å Árçæ] [¦ææā] } Áāt `¦^• Á[¦Ár〕] &BæaļÁra] ^ Á (FJÌ ODÁræj å Á, ^ cÁ(FJÏÎ DÁræāj, æaļÁ ^ æs• ÈÁ/@ Áræāj, æaļÁ, [`|å Ára ^ Áçæa ãanæi]^ Ár@[`\* @` cÁraj ^ Ár &[¦å ā] \* Á ]^¦ā[ å Á, ãr@Á[ [} c@f Ár[ cæaļ• Áræa][ ç^ Ár[ kár / [], Ár@ Ár, ^ å ãæn) Árçæt, ^ Árç^} Á, ãr@āj Áræj }` æaļÁ, ^ kar} cāt ]^¦ā[ å Á, ãr@Á[ [} c@f Ár[ cæaļ• Áræa][ ç^ Ár] (Åa ~ [], Ár@ Ár, ^ å ãæn) Árçæt, ^ Árç^} Á, ãr@āj Áræj }` æaļÁ, ^ kar} cāt ]^¦ā[ å Á, ãr@Ár[ [} c@f Ár][ kár][ ; Ár@ Ár, ^ å År • ^ å Áræj Árçæt, ^ Árçr} Á, åræðaj Åræðaj }` æaþÁ, ^ kar} cāt àæð; å• ÈÁ/@ Ár[ cæaþÁ[ ¦Ár@a; A] ^ kār å A' kār å Ár • ^ å Áræ Áræ Áræða Åræða] ðræðaj }` æaþá; æðaj }` æaþá; æða aæða å• ÈÁ/@ Ár[ cæáÁ[ ¦Ár@a; A] [ kār A] ~ kār å Ár • ^ å Áræ Aræða Áræða Åræða] ðræðaj }` æþá; æðaš ^ • DÁræ Ár@ Árai ãræða Á -æ&a[ ¦ÈŹ/@ða Áræj æð; • ār Ásj Á/æà]^A Æ ÆrÅa { [} • dæar • Ár@ærárc] ðræðaj ˆ Á, !^ &ða] ãrææða] } Ár¢&^ ^ å • Árçæð] [ kæðaj] } Á [ ç^¦ÁræÁFCAA, [] c@ðh, ^ kā] á ÈÁA

T[}c@Á	Ö¦^ÁÇ;{DÁ	T^åãæa)ÁÇ;{DÁ	Y^cÁÇ;{DÁ
Ræ),≚æl^Á	ÉÌIÈCÁ	ÉÌÏÈÁ	ÉGÈÁ
Ø^àlĭæĥÁ	ÉHÎÈEÁ	ËGJÈGÁ	ÉGÍ GÌFÁ
Tæ¦&@Á	ÉÏĚÁ	ÉFÌÈÁ	É΀ÎĖÁ
OEJ¦ãµÁ	ÉGJÈHÁ	ÉGJHÈLÁ	ÉIÍÈ€Á
TæÁ	ÉHJÈÁ	ÉGGÈÌÁ	ÉFÍ Ì ÈFÁ
R*}^Á	ËHÈ€Á	ÉÌGÈÌÁ	ÉÍÍÈÁ
R"  ^ Á	ËÈÁ	ÉÏJÈ€Á	ÉHIËÉÁ
Œ**•oÁ	ËH€ÈHÁ	ÉÏFÈFÁ	ËÏĚÁ
Ù^] ơ^{ à^¦Á	ÉFI Í È Á	ÉFÏ ÈÁ	ËFÈ€Á
U&d[à^¦Á	ÉÌÈÁ	ËÏËÁ	ËGIÈIÁ
Þ[ ç^{ à^¦Á	ËF€ÌÈÉÁ	ÉFÌ È€Á	ÉĨÏÈEÁ
Ö^&^{ à^¦Á	ËHÈDÁ	JFÈÁ	ËFĒÁ
Óæþæð) &^ Á	ÉÌÏÈÉÁ	ÉÍÍIÈÁ	ÉF€ÏIÈHÁ

#### HUV'Y'(!% BYha cbh\`mk UhYf'VU'UbWY'gHUhjghjVg'

 $Vaaa|^{A} \stackrel{()}{\boxplus} A \stackrel{()}{\oplus} A$ 

Őãç^}ÁœA\*`¦~æ&^ÁœA^æAţ~ÁtÎĖİÁ@æE&@ãA\*``æe∿•Át[ÁœA\*`¦]|ĭ•Áţ~Á+FIÈFÌÁTŠÈÄ

Õãç^}Áv@eeÁæãj-æļlÁsā:dānčdātšdātšdātšdātšAæš\*^|^Áçæbāæài|^Áeo)åÁj:|^&ājāææāj}Á\*^}^:æļl^Ár@eA\*;æļl^Áæ&^^å•Árçæ][:ææāj}Á -{:|Á, [•ơ4, [}ơ@:Á,-Á@:Á^æbÉ&arÁa\*,Á]a\*|^Ás@eeÁr¢&æçææāj}Á;ā]A^•č|ơ5ā,Ás@:Á{;:{ ææāj}Á; •ðt}ãa&æ)ơ&[}^Á;-Ás^]:|^••āj}Áeb[č]}åÁs@:Ár¢dæ&ætāj}Áeb^æebÉP[]\_^ç^:\É&arÆa\*,Á[••ãa|^Ásčiā;\*Á ^¢c^}å^åÁs:|^Á,^:ājå•ÉA:[{ ^Á[&æ‡ãr^åÁs:]æ;å[]}Á;æáák&&č;!Á;}Áææk\*{ ][:æfÁsæ:ã;EÁ

#### ("&"(`HchU`k UhYfigY`

Óæ•^å/{{}}Á@~Á&æ¢&`|æaā[}•Áæà[ç^Êk@~Á•cã[æer\åÁ;æer\¦Á•^Á{¦Åø@~Á`æ¦^ÁārÁæe Á{[|[,•kÁ

•Á 0.54, æ¢āį `{ Áį ÁGI ÁT ŠĐ^æAĮ Á, æv\ÁĮ •Áœ?[`\*@k@@Á^c?} œį] Áį Á, æv\Áj Áv¢][¦c\*åÁæ)åÈÁ

•Á FÁT ŠĐ^ælá[ -Á æz^lÁ •^åÁ[ láů \* • ơ4 \* ]] l^•• •ã] ÈÁ

Ú[cæà|^Á,æơ\¦Áæ)åÁ,æơ\¦Á cāpã ^åÁ§,Á(}Á ãơ Áæ( ^}ãæ?•Á,ā|/Áa^Á@ecç^•ơ\åÁ(}Á ãơ Áa^Á, Â ¦æāj,æơ\¦Áæ)\ÈÁ

#### ("&") HchU`k UhYf`Uj U]`UV]`]hm# gU[Y`

Őãç^}Áog Á∿¢dæ&cāţ}Áo⇔^æáţ–ÁîÎËİÁ@exáog ákag ákag ákag ákag advázág -æd|A[,-ÁFÏGEEFEA[{ Ébenát[cæd-Á],¦^&6]aīæeaā[}Á ãj]čo4[-ÁJÏÍÉEIÁTŠÁ&æg áka^Ása^{[}+odæe^åÈEÖçæg][¦æeāt]}ávæ&&[č}orÁ[¦ÂîÎFEEÎÁTŠÉEÅ~æçā]\*ávæÁ •č¦]|č•Á[-ÁHFIÈEÌÁTŠÉÄ

Õãç^}Á, æe^¦Á•æ\*^Áæ••[&ãæee^åÁ,ão@Á;æ)åÁ,¦[å\*&ãå]}Åæ)åÁ\*¢][¦ơ4/[••ÁÇEIÁTŠDÁæ)åÁå\*•óÁ •\*]]¦^•••ã[}ÁQFÁTŠDÉ&{[œ4Á,^ơ4\*`¦]|\*•Á,-Á,æe^¦ÆsiÁ§JÁs@/Á,¦å^¦Á,-ÁCÌJÈEÌÁTŠÈÄ

CE[Á\*`¦-æ&^Á, æe^\Á`}[~-Á, ä|/Áz^Ásáāç^\c^å/át[Á\*¢ã;cā]\*Ási¦æäj•ÁsjÁs@Áç3&ajãc´Áse)å/Ás@{\^-{ \^Á,[Á\*`|-æ&^Á ; æe^\/Ásj-{[,•/Áse^^Áã^\^Á.4x^\*+e=se^`ÉÁ
Á

Q, Áx@ Áx;ç^} chs@eenka@ Áxe{[`} ch\_4;} E=ãx^Á; een ¦ Á;d[¦æ\*^/&a^&|ā]^• A;ç^¦ Ázā; ^ÉAP[|&ā; Á;ā|A,^^åA(;Á cæ}^Áxe&cā;} Á;f Á§;&\^æ\*^Áx@ Áxe;æa#æà;|^Á; æ\*\'A;l Á^å`&^Áx@ Á; æ\*\'A&[}•`{]cā;} ĚA/@ Á;[||[;ā]\*Á []cā;} •Á;[`|å/áx^/&{[}•ãa^\^åKA

- •Á Ù[`|&āj\*Á, æe^|Á¦[{ Á¢ā;cāj\*Áå;æt, •Át}}Ásæåbţājāj\*Áæiå, êĂ; @BR@Áse Á, [|{ æth^Á&|[•^Át] Ásæá &ætjæstañ: ÉÁ/@áA, [`|åAtj&]`å^Ás@AT æsetāj` { ÁP ætç^•cæia|^ÁÜ ät @AÖætjæstañ: ÁQ;æseÉAF€Ã Á |`}[~-DÁt[¦Ás@A´ãe^ÉÁ
- •Á T[åã÷Ác@Á&|^æ)Á;æc^¦Áåãç^¦•ãį}Á&@ea)}^|Át[Áåã &@eat\*^ÁF€€ÃÁ;Ác@atÁ[,Ábg d[Ác@Á\*æl¦^ĚÁ
- •Á Q.&:\^æ^As@A;\[][\cai,}A; -A; a) åAs@eee/&; A[|åA;]; æ @ åAs[A^å` & As@Ase([`}oks[]•`{ ^åAs;A o@ A; æ @; \*A;\[&^••EA
- •Á Ö^&¦^æ^^Ás@·Áæ\*^Áæ}åA; |[å`&a]; Å[Á^å`&^Ás@·Ás@[`}of&[}•`{ ^å/§; Ás@·Á; æ @]; \*Á ] |[&^••EÅ

Á

## ) "Á 9bj ]fcba YbHJ` Wtbhfc` a YUgi fYg'

Ò}çā[}{ ^} œaþÁ^˘ ǎa^{ ^} @ Áæ) åÁ&[}d[|Á( ^æ č ¦^• Áæ^ Áæ^ Áæ^ Áæ^ áÆ) áÆ) áÆ@ ÁÔ[} åãā] • Á( ÁOE] ] [çæþÁ æ) åÁœ ÁÔQ)ÈÀU] ^&ãã&Á( ^æ č ¦^• Áæ) åÁ^˘ ã^{ ^} @ Á{ Áæå å¦^•• Á[ ǎfÁæ) åÁ æc^ ¦Á č æfãc ÁÆ] ] æ&œ Áæ^ Á [č dậ ^ åÆ) Á/æà |^Á, ÉE Áæ^ |[ ¸ ÈÁ

## HUV'Y') !% 9bj ]fcba YbHJ' Wtbhfc``a YUgi fYg''

Ü^-Á	Ò}çã[}{ ^}œa∳ÁTæ}æ*^{ ^}œ^*¦^Á	Vą̃ą*Á	Ü^•][}•āàājãc^Á
Gc]``			
ÙY €FÁ	Q å šo Astel Á ão Á coo - Á [ Á coo - Á ta - Á coo - Á ta - Á coo - Á ta - Á coo - Á ta - Á coo - Á ta - Á coo - Á ta - Á coo - Á ta - Á coo - Á ta - Á coo - Á ta - Á coo - Á ta - Á coo - Á ta - Á ta - Á ta - Á ta - Á ta - Á ta - Á ta - Á ta - Á coo - Á coo - Á ta - Á coo - Á coo - Á coo - Á ta - Á coo - Á ta - Á coo - A coo - A	U]^¦æa‡i}Á	Ûĭæ¦ſÁTæ)æ*^¦Á
ÙY €GÁ	Ò¦[•ã[}Áæ) åÁ^åã[^}¢K[}d[ Áå^çã&^•Á ã Áà^Á ã)•œa ^åÁ¦lã[¦Áű[Áx@/Á8[{{^}&~{^}¢f_A^¢dæ&dã]}Á ,[¦\•Á8]Áæ&&[¦åæ)&^A ãr@ÖC¦æ ã]*Á¤[•Á ÕRe≦ÍÍIIÈDÙÔÚÈEÁæ) åÁÕRe≦ÍIIÈDÙÔÚÈGÁæ ]¦^•^}♂åÁ§IÁOE]]^}åã¢ÁÓÈÁ	Ú¦∧ĘĮ]^¦æaaįį́}Á́	Û*æk¦^Áraa)æt^¦Á
ÙY €HÁ	V@Aj,^¦ãį,^ơː¦Ás`}å£32ææ8,@Asia#jÁ,ãj,Åá^A^^å^åÁ Q&[}•ã:c^}cÁ,ãc@Açãčă;ad+Ás`~^¦ãj*DÁ][}ASI{] ^dã]}Á [-ÁserAS[}•d`&dã]}ÈÁ/@ãÁ,ãjÅA?•`¦^Ás@Ag,c*¦ãčAj,Á c@Ásææ8,@Asia#jÁsiÁ;and;adaj^åAsia}åÁ;ajÅæç[ãåÁ ][c^}cãad+Á ã]•ÁsiàåÁ^åã[^}cA;Aj[•cÁ &[}•d`&dã]}Á;As@Aj,^¦ã[^c/kás`}åÈÁ	Ú¦^ĘĮ]^¦æaąį́}Á	Û*æ¦^ATæ)æ*^¦Á
ÙY €I Á	CEååãāj} æļÁ <sup>+</sup> [•ā]}ÁæjåÁ <sup>A</sup> ^åã; ^}óKej}d[ Á‰ <sup>+</sup> çã& <sup>+</sup> •Á āj& čåāj*Á <sup>†</sup> ãjoÁ <sup>+</sup> }& <sup>+</sup> EŠ&ææ&@åi¦æāj•ÉŽ}^¦ãj^c?¦Áàæj\•Á æjå'nãç^!•ãj}Á&@eaj}^ •Á;ãjÁ‰A <sup>†</sup> áj•cæaj/åÁ¦}ÁæjÁæe ¦^čă^åqábææãÉŽUč&@4ţ^æč¦^•Á;ãjA‰A <sup>†</sup> áj•cæaj/åÁ æ&&[¦åæj&^Á;ão@b@Ábbl[ã+ÁæjåÅÕ[]•dč&cãi}}Á *čãå^jã,^•ÁÁTæjætāj*ÁViàæjÁÙd[¦{_;æe^\;+Á ÇŠæjå&{{ÉCEEEDEĂ	U]^¦æa‡[}Á	Û`æ¦^ATæ)æ*^¦Á
ÙY €Í Á	Ò¦[•ā[}Áæ)åÁ^åã[^}ó&[}d[ Áŝ^çã&^•Á,ā Áŝ^Á ãj•œa  ^åÁ[}•ã¢Á]¦ã[¦Ág[Á^¢clæ&cã[}/ásÁ][ơ^}cãa‡Á[¦Á •@^∽óæ)åÐ[¦Á*ĭ  ^Á¦[•ã[}Á¢cão•ÈĂ	Ú¦^E‡]^¦æaãį}Åæ}åÁ []^¦æaãį}Á	Û*æ¦^ÁTæ)æ*^¦Á
ÙY €Î Á	Ùd[¦{ , æe^¦Áĭ }[~-Á,ā  Áà^Áåā^Asa^àAæ;æ∂Á√[{ Á åãrcĭ¦à^åAsachæebĂ	Ú¦^Ë[]^¦æaã[}Áæ)åÁ []^¦æaã[}Á	Ûĭæ¦^ÁTæ)æ*^¦Á
ÙY €Ï Á	Öãr&@æd∗*^Áç^ [&ãã`Áæ);åÁ&[}~ãt`¦ææā[}Á,ā A&a^Á &[}d[  ^åAq[Á*}•`¦^Á*¦[•ãç^Á4][,•Á&[Å,[ơ∱,&&`¦Á	U]^¦æaāį}Á	Û*æ¦^ÁTæ)æ*^¦Á
ÙY €Ì Á	Õ¦æ∙Á&[ç^¦Á;ā∥Áa∿Á;æājæaāj^åÁæ{[`}}åÁo@Á ]^¦ā[^c^¦Á[Áo@Á*¢dæ&cā]}Åæ^?æÁæ}åÁæåbæ&^}oÁ[Á c@Á@eĕ Á[æå•ÈÁ	U]^¦æaa‡i}Á	Û°æ¦^ÁTæ)æ*^¦Á
ÙY €JÁ	Y@:¦^Aj;¦æ&ca&caæaak)^EA;';-æ&^Á;æơ:\●Á;[{Á `}åãc`¦à^åÁed;^æeÁ;ā∥Áa;^Ásãç^¦c^åÁeç;æêÁ;[{Á ^¢dæ&cāį}}Ð;[¦\Áed^æeÈÁ	U]^¦æaa‡i}Á	Û°æ¦^ÁTæ)æ*^¦Á
5 WJX gi 🕻	ZJhY`gc]`g`		
ÙY F€Á	V@ Á∿¢clæ&cāj}Áech^æe,ÁjāļÁec^Áece•^*a,Áţ¦Áo@ Á ]¦^•^}&^Á;Áezeñá`¦-æe*Á[ā+ÉA9jÁeze&[¦åæ)&^Ájão@Á c@:ÁOE&ãa ÁŬ` -æe*ÁÛ[ā+ÁTæ)æ*^{^}áÚ æ)Á8jÁ C0[]]^}åã¢ÁÔEĂ	Ú¦^ĖĘ]^¦æeą̃į}Áæ)åÁ []^¦æeą̃į}Á	Ûĭæll^ÁTæ)æt^lÁ
ÙY FFÁ	Q¥oe&3ãa Ái` -æe^ Ái[āp Ásb:^Aj¦^oo^} dÉo@Á∿¢&æçæe^åÁ {æer}lãad-ÁjājÁsi^Ád:^æe^åÉ&BjÁse&&[¦åæ}&^Ájão@ó&Á OB&ãa ÁÙĭ -æe^ ÁÙ[āp ÁTæ}æt^{^} a^} dÚ æ)ÆBjÁOE[]^}åãcÁ ÔÉÁ	U]^¦æa‡į}Á	Û*æ¦^ÁTæ)æ*^¦Á

Ü^Á	Ò}çã[}{ ^}œa¥́Tæ}æ*{ ^}ơ∱T^æe*¦^Á	Vą̃ą*Á	Ü^∙][}∙ãaặãcÂ
ÙY FGÁ	OEIÁ k c d a star c a k k a ser l an t a k k k k k k k k k k k k k k k k k k	U]^¦æa‡i}Á	Û`ælî^ATæ)æt^¦Á
UY FHA	CE[A],} Erāč^A*`¦-æ&^A, æ&'¦A'[{ Ašāčč¦à^àAæ4^æe,A, ä A 妿aj,Ásjq[Ás@Aj,[}åÁ[Ás@æsÁæ3^Ás&ää&A^æ&qā]}•Á,ä Á à^Ás^c^&c*åÁæ3;åÁs@Ár¢dæ&qā]}Aj,¦[&\••A[[åãã*åÁs[Á æç[āaÁæÁs[}c3];ča];ča}&^Á;~Ás@Aj,¦[à ^{EÁ	U]^¦æaaį́}A	U°ælîATæ);æ*^¦A
ÙY FI Á	OEļÁs¦æajæ*^Á+[{ Ás@Á&:&{[}^Ð æ=@/s, æ)óA, áļÁs^Á }^*æaāç^ ^Á*!æah^åAsaæ&ká[Ás@Ağ c*!}{ ^}óA,[}åAş[Á ^}•`!^Ás@eenAaj^ÁA;a&&@eenA4[{ Ás@A;a)åA;a[}åAş[Á !^č'}•Á,æen\á;lÁa)^Á@å!æĕ]a&aaj^Á*`•]^}å^åÁ ]^!ãa&Aa[æen\á;lÁaaj•Ás[,æå•Ás@Aj}`Á*`•]^}å^åÁ ājcn:}{ ^}of,[}åEÁ	U]^¦æa‡i}Á	Û*æ¦^ATæ)æ*^¦Á
ÙY FÍ Á	GÁv@ Á^č¦}• Á æɛʰ¦Áਙí Á [ cku æ)•] [ !cʰå/ឆj Áæá/ āj ^]ā) ^ EÁ c@ Áç^ [ &āɛʿ Á āļ/Áa^Á ~ -3&ēa} cku [ Á } • ` !^ Ás@æa/, [ Á ] !^&āj āææāj } Á; -Á; ^!āā&/Áā] ^• Á; &&č !• Áač !āj * Á { [ ç^{ ^} chÁ; @• ^ Á æɛʰ!• Áu ], æ!å• Ás@ A; } ⋿āɛ^Á -āj ^• Æj cʰ! { ^} ch_ [ } àEÁ	U]^¦æaa‡i}Á	Ûĭæl¦^ATæ)æt^¦Á
ÙY FÎ Á	V@ Á^č¦}•Á æc∿¦Á[[ Á æc@Á ặļÁa^Ásā^&c^åÁse;æ Â ↓[ (Ása ^ Ás d' & ] ∄∿å ása å Å	U]^¦æaāį}Á	Ûĭæl¦^ÁTæ);æ*^lÁ
6`i Y; fY	Yb'5`[UY		
ÙY FÏ Á	Ö`¦āj*Áx@Aå¦^å*āj*Á;@æ•^Áţ-Áx@?Áţ]^¦ææāt})•Áx@A -{¦{ ^åAjæà^•Á,ãtJAå^Áaã}^åAt[Á})•`¦^Aax5a A& ^æAt[Á æaþlÁx0eæaA@A,æe^¦•Áxe^A,[cA{{¦Aå¦ã}\āj*Áţ¦Á,ãt{aj*ÈÅ	U]^¦æaāį}Á	Ûĭæk¦^ÁTæ);æ≛^¦Á
ÙY FÌ Á	Yæ}ðj*Áðð}•Áðj& `åðj*Á&`!!^}o%æp^!o4^ç^ •ÁÇ^^!Á qÍÁ Væà ^Á ËÐDA, ðjAà-Á^]oÁ, ão@ðjÁc@Á`æ!^A[,&AAbe)åÁ ^æ&@4ða^]^Áæ&&^••Á,[ðjoÁfÁc@Áæà^Áðjå^-ðjãr/]ÊÁ	U]^¦æa‡į}Á	Û≚æ¦ʿÁTæ)æ*^¦Á
ÙY FJÁ	X^*^ cæc*å Áçã čæ Á§ů ~~^\+ Áæc^ Áq[Á§ı^ A] [æ] c*å Á • čii[č] å ∄ * Ás[ c@ Áv¢ clæ&cī] } Áæc^ æ ÁÇ^^\ Ág[Á Šæ) å • &æ] ^ ÁT æ) æ * { ^} cÅ] æ Dæb) å Ác@ • ^ Á ∄lÁ æ • ã cásj Å ^ å &∄ * Á, člæ} c/{ _ 3 cáu] æ - Á @&@ A Å [^ å * &∄ * Ás@ A][c*} cæc4[ i Ás] * ^ Ë i ^ } Áæt* æ Á [`cài^æ • ĚÅ	U]^¦æaaį́}Á́	Û≚æ¦ʿÁTæ)æ*^¦Á
ÙY G€Á	2[   [, 3] * ٨ @ ٨ ɛ'، { 3] عطيمًا } ٨ ﴿ ـــــــــــــــــــــــــــــــــ	U]^¦æa‡i}Á	Û`æ¦^ÁTæ)æ*^¦Á
ùy gfá	CE; Áçãrānā; ×Á&[} dæ&d[!•Á; Á c@: Áçãrãnā[!•Áā^]^Á[Á &[{ ^Ánjd[XB]} cæ&d, ãr@kb@ Áæ}^Á æc*!Á; IÁa^A ^c][•^åA[Aee*![•]]•Ê, älAa^Aee\^âA @ c@: IÁ@^A @acç^Áed@ard[!^Á; Aee# ^!*^} abAa^Aee\^âA @ c@: IÁ@^A @acç^Áed@ard[!^Á; Aee# ^!*^} abAa^Aee\^âA @ c@: IÁ@^A @acç^Áed@ard[!^Á; Aee# ^!*^} abAa^Aee\^âA @acç^ÁeA @ac@aee#] á%aA[[Á]:[çãa^âA]; [!{ æea[} AA* æda]; A c@Aã`\EAOE; Ang -{!{ æea[} AA* @^A(A]; [`jâAa^A; ]; [çãa^âA å^cæaa]; Ac@A8[]:[^&A(A*)]] } •^ÁaAaAæ @A; IAee c@; æA æccæa&\Áa[^^A&&`IEA	U]^¦æaa‡i}Á	Û`æ¦^ATæ)æ*^¦Á
ùy qgá	QÁce)^Á,^\•[}}^ Áce^Á^´ă^ă^åAţfÁsājÁc@ãÁcee)å•ÁşiqfÁ c@Áceà^Áţ¦Aţ]^!æa‡i}æa‡i!Áaeţi]jä*Á,`!][•^•Áceô^Á jäjÁceeç^ÁceÁ`ãceeù ^Ár\}*coAî`àà^!At∥[ç^ÁqiÁceç[ãaÁ •\ājÁr¢][•`¦^ÈĂ	U]^¦æaāį}Á	Ûĭæ¦îÁTæ);æ*^¦Á

Ü^-Á	Ò}çã[}{ ^}œa¢ÁTæ)æt^{ ^}αÁT^æe`¦^Á	Vã; ã;*Á	Ü^•][}•āaājāc`Á
ùy ghá	Q[¦Á^[] ^Á^˘ă^åÁţÁ`à{^{*`à{^{*`à{^{*}}A}şid[Áv@A]aex^¦ÁæA aà Âdā ^A@aā;ā]*Á{¦Á`čā}{ ^}ơÁ^\;ca&a]*Á^&B ăã;ā]*Á`ão Ade^Á[ơÁ^&{{ { ^} å^àAee Âo@^Ác}àAe daa]ÁaajåAsãi`]]ơ&^  •Á@!^à`Á*aasjā]*Á'!^aev\!Á ^¢][•`!^Áţ[Át[¢ā]ÈĂ Q•ơ;aaâÉĂ[[!•^Áãadā*Á,ã], ^aeA@` åAa^Á[!}ÊĂ	U]^¦æaţi}Á	Û`æ¦^ÁTæ)æ*^¦Á
	a) äAad, ^A&[} ca&cA, ão@ko@ A aa ^A, aac^lAr@[` äAba^A -{I  [`, ^å/ba`Áaako@;  [`*@Á@[`, ^¦Áv[{ Áaa4& ^aa) Á, aac^lÁ •`]] ^ÉÁ		
ÙY G Á	U}Ëā&^Á^,æt*^Á,ā Áå^^Á,æ)æt*^åÅ{[Á&]{] ^Á,ão@Á c@~Á^``ã^{ ^}@~Á{,4&@~ÁO}çã[]{ ^}oÅæ)åÁP^æ)c@Á Ú¦[c^&cā]}ÁÕ`ãa^ ā]^•A.ÁU}Ëāz^ÁÙ^,æt*^Á Tæ)æt*{ ^}oÁ{[¦ÁUā]* ^ÁP[`•^@ å•ÁÇFJJÌDĂ	U]^¦æa‡į}Á	Û`æ¦^ÁTæ)æ*^¦Á
ùy ɗ á	Ú[c^}cãæ‡Á`dâ^}c∱a[`¦&^•Á;ā]AávÁ(æ)æ≛^åAá[Á {ājā[ā^^Ác@Áã^ ã@[[åÁ[-Áca‡*æ‡Áa][[{ •ÁsjÁc@Á [}•ãc^Áca^•ÈÁ	U]^¦æaaį́}Á	Û * æ¦ ^ ÁT æ); æ* ^¦ Á
Gif ZJWY	<b>«Uhyf</b> Á		
ÙY GÎ Á	Ù"¦-æ&∧Á,æe∿¦Á;[}ãq[¦ãj*Á,āq Áà∧Á'}å^¦œaè^}Á§jÁ æ&&[¦åæ)&∧Á,ãc@ÁÚ^&ca‡}ÅÂÈEEĂ	U]^¦æaāį}Á	Û*æ¦^ÁTæ)æ*^¦Á
ùy gi á	<ul> <li>Ùæţ ] [ā] * Á[ &amp;ææā] } • ÁÔÚŒÁÚ[ ā] ÁF Áæj å ÁÔÚŒÁÚ[ ā] ÁGÁ æh Â, k] [çãã ^ å Åæj å Á[ æð] æð] æð] ^ å Åāj Áæj Åæj ] { [ ] i ææ ^ Á</li> <li>&amp;[ } å ãaā] } Áa[ Á, ^k { að á</li> <li>&amp; V@ Á&amp;[ ^ a æh â</li> <li>&amp; V@ Á&amp;[ ^ a æh â</li> <li>&amp; V@ Á&amp;[ ^ a æh â</li> <li>&amp; V@ Á&amp;[ ^ a æh â</li> <li>&amp; V@ Á&amp;[ ^ a æh â</li> <li>&amp; V@ Á&amp;[ ] ^ a æh â</li> <li>&amp; V@ Á&amp;[ ] ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á&amp;[ ] [ ^ a æh â</li> <li>&amp; A V@ Á@[ ] [ ^ a æh â</li> <li>&amp; A V@ Á@[ ] [ ^ a æh â</li> <li>&amp; A W â</li> <li>&amp; A V@ A@[ ] [ ^ a æh â</li> <li>&amp; A @ A @ A @ A @ A @ A @ A @ A @ A @ A</li></ul>	U]^¦æaặį}Á	Û*æ¦^ÁTæ)æ*^¦Á
ùy gì á	Č¢ārcāj×Á`¦-æ&∧Ájæe∿¦Á&[}åãātj}•Ájáj/&s∧Á {ænjcænj∧å/μ`orãå∧Áv@vA∿¢&æçænatj}Á&ch~æbĂ	U]^¦æaāį}Á	Ûĭæ¦^ÁTæ)æ*^¦Á
ùy gjá	Y@\!^Án^åã[^}ơÁ,![à ^{ • Ánd^Aña^}cã&?aÊA^^cda;*Á§jÁ c@^Ás!^å*^Á,[}åÅ,ã Ás^Ánaña^åAs^Ás[•ð];*Á ãc@And - [&&` að;oAi`&@And^I]•`{ Aj;!Ánd}Ánd':^åÅ andor,}aæãn,^ÉÁ	U]^¦æa‡į}Á	Ûĭæll^ÁTæ)æ*^lÁ
ÙY H€Á	Ö`¦āj*Á& ^æðj*Á&)åÁ&[]•[ā/Ádā]]āj*Á,[]^¦ææ‡i}•ÉÅ •`¦-æ&^Á;æe^¦Á,[[, •Á;ā Á&^Á&ā^&c^åA&[, æbå•Á∞Á,}Ë •ãe^ÁæA^•EÅU^åã[^};ó&=àÅA*[[•ā]}Á&[}d[ Á, ^æ*`¦^•Á ;ā Áa^Á\$j•cæ4/^åÅä`¦āj*Áx©áÁ;cæ*^ÁæA,^¦Áx@A, æ)•Á ā)ÁQE]]^}åãaÁÓÉÅ	U]^¦æa‡į}Á	Û*æ¦^ÁTæ)æ*^¦Á
ùy h <b>f</b> á	Ö^,æe^¦ā) * Á¦[{ Á¦}Ë ãe^Á,æe^¦Áa[åã∿ĒA¢& čåā) * Á c@:Á¢cdæ&ca[}A,[}å • ĒÁ,ā Á,[ơ5a^Á}å^¦cæa\^}Á ,ãc@[čof,¦ā[¦Æa]]¦[çæaÅ{[{ ÁÖÚOBOĐĂ	U]^¦æaāį́}Á́	Û*æ¦^ÁTæ)æ*^¦Á
ùy hgá	Þ[Ásā&@æs*^Á;āļļÁ;&&`¦Á+[{Ás@As¦^å*^Á;[}å•Á `} ^•∙Asā&&@es*^Á&iãe^¦ãæé@eç¢As^^}Á;^dÉA	U]^¦æaāį}Á	Û*æ¦^ÁTæ)æ*^¦Á
ÙY HHÁ	Öãr&@eet*^Aç∿ [&ãcî Áea) å Á&[}-ãt`¦aeaā[}Á,ā Ába^Á &[}d[  ^å Áx[Á*}•`¦^Á*¦[•ãç^Á¦[,•Åba[Å,[o4[&&x`¦ÈÁ	U]^¦æaāį}Á	Û*æ¦^ÁTæ)æ*^¦Á
ùy hi á	Ò¢ã cā;*Á`¦ç^^^åá@?ă @A∱^*•Á§jÁ ¸æc^¦&[`¦•^•Bà¦æa9j•Á;ā Áà^Á§j•]^&c^åÁæ9jåÁ^] æ&∧åÁ æ•Á^``ã^åEÁ	U]^¦æa‡į}Á	Û*æ¦^ÁTæ)æ*^¦Á
ùy hí á	Ùc!^æ∢i àæ)∖Áæ)åÁa^åÁ¦[~ã}^Áæ)åÁ&[}åããã]}Á§[Áa^Á ¦^&[¦å^åÁæcÁæ]]¦[ç^åÁ[&ææã]}●ÈÁ	U]^¦æaāį}Á	Û*æ¦^ÁTæ)æ*^¦Á
ùy hỉ Á	Ö^∙at}aac^å£54(]^\çat`•£64`}å^åÁæa&a†aaa*•Á,ãc@4,a†Á æ)åÁ,æe^¦Á^]æbæe[¦Á^`•c?{Á,a‡ Áa^A,f,l[çãa^åA[¦Á & ^æ)aj*Áæ)å£D;lÁ(æanjc?}æ)&^A[,-Áç^@34 ^•£4, æ)oA[¦Á ^``a]{^}d	U]^¦æa‡į}Á	Û*æ¦ʿÁTæ)æ*^¦Á
ÙY HÌ Á	CE[Á&@ { 38ad+ ÉĂ× ^   • Áad) å Á; ã+ Át di   ^ å Áacók @ Á;   ^ { ã ^ • Á { ` • OÁs ^ Á&[ } cænaj ^ å Á; ão @ j, Áad] ] ! [ ] ! ãane^   ^ Ás ^ å å à ` à å áad+ æ Ác@ean { ^ ^ oKo@ A[   [ ] ∄ ane^   ^ Ás ^ å] ^ å Á  ^`` ã^ { ^} o hÁ aDÁ&[ { ]   ^ Á ão ®ady ^ Á^   ^ çæng o Ó E • clænamin Á Ucæng å æd å • Á - [ ! Ás@ Áña ` ãs • Ás ^ ä * Át di ! ^ å Á à DÁ@eqe^ Áad ` ãs • Ás ^ ä * Át di ! ^ å Á & DÁ@eqe^ Áad ` a * Ás ^ ä * Át di ! ^ å Ás å å æd + Á [ - Ás@ Áræt * 3 ãi ~ { Á&ed a æda û f . ÁF F € Ã Á; -Ás@ Áç [ ] ` { ^ Á [ - Ás@ Áræt * • o K&] } cæna ^ ! At di ! ^ å Á ão @ a Ás@ Ás` } å ÉÁ	U]^¦æáį́}Á	Û`æ¦^ÁTæ)æ*^¦Á

Ü^-Á	Ò}çã[}{ ^}cæ‡ÁTæ)æ≛^{ ^}oÁT^æe`¦^Á	Vą̃ą̃*Á	Ü^∙][}∙ãàã;ãã Á
ùy hì á	Ù]āļÁār•Á,āļÁa^A,¦[çãā^åÁæoÁqe Á&@{ a&aqA`d;¦æ≛^Á ~æ&a∄ãa?•£&[{][`}}åÁãa^•Áxe}áxe}åÁœe~Atiæaj^åÁyiÁc@ãÁ `•^Á	U]^¦æaaį́}Á	Û *æl ^ATæ)æt^lÁ
ÙY HJÁ	<ul> <li>Y @: !^Á^~~    ā * Á; } Á ār Ás Á^~ ă ^ ăr ă Ês @ A{   [, ā * Á { ab æt ^ { n } of, ] æs æs x A^~ ă al Ås ^ Ás ]  ^{ { n } { n } ob al Å.</li> <li>A Ü^~  ] ā * Á ā] / ás ^ Á { ab a * / ab a *</li></ul>	U]^¦æa‡i}Á	Û * æ¦ ^ ÁT æ) æt ^¦ Á
ÙY I €Á	Ü^*` æ\Á&@&\•Aţ+&ş^@&K ^•A,[!\]ā;*ÁæeAs@A``æ\^A ,ā Aਙ^A&[}å`&c^åAţA?•`\^Às@æAy[Aţā+Aţ\A`~ •Á&^^A  ^æ}āj*EX#	U]^¦æaāį}Á	Û*æk¦^ÁTæ);æ*^¦Á
; fci bXk	UhmfÁ		
ÙY I FÁ	OĘ[Á*¦[`}åË;æer\¦Á^ æer\åÁ*ãer\Áns&cāçãa?+•Á;āļ/Aa^Á `}å^¦cæa\}Á§jÁns&&[¦åæa}&A^Á;ão@AÖÜQá,ÁU~a&r\á;~Á Yæer\¦Á*¦[`}å;æer\¦Á&7}&?ÉÁ	U]^¦æaāį}Á	Û*æ¦^ÁTæ)æ*^¦Á
ÙY I GÁ	Õ¦[`}å, æe^\¦Á([}ãq[¦ã)*Á,ā∥Áa>Á'}å^¦cæà^}Á§)Á æ&&{[¦åæ}&^Á,ãc@AU^&ca]i}ÅiEeĐÁ	U]^¦æaāį}Á	Û*æ¦^ÁTæ)æ*^¦Á
:`ccX]b[ Á			
ÙY I HÁ	V@ÁārAārAárAárAár}å^árAár}å^åAárAí¦[çãa^Aí¦[c*&dār}Åi[{Aár}{A};[c*&dār}Åi[{Á -{[[å]*Á]ÅárÁ <del>t EE</del> Aí{{ÁşhEEAí{{ ÁşhA®ă î @Ásaa[ç^Á;aez`¦aabÁ *'[`}åÁnç^ ÊAirÁsaAíaezãi`{ Á@ã@AírÁcEEAíAOEPÖEĂ `} ^••Aíc@{;ã*Aás}]![ç^åÁsa^ÁsaAîv&i^cæa`ÈĂ V@Ása`}åãj*ÁsaÁarAár[}cæasj^åÁ;@{  ^Á;ão@3;Ás@Á •ãrÈĂ	U]^¦æa‡[}Á	Û * æk¦^ÁT æ) æ*^¦Á
ÙY I I Á	P[ &ā; Á; `•oA;¦[çãa∧A;∧¦ā; ^c∿¦A妿anjæ≛^A;¦ā;¦Áq[Á c@:Áx[}•d`&ca‡}A;Áaă`}åāj*A;¦Ác@©A; æa&∧{^}oA;Áān Á [}⊡ãe∿DĂ	U]^¦æa‡į}Á	Û*æ¦^ÁTæ)æ*^¦Á
ÙY I Í Á	Þ[Á\$¦^å*ā)*Á;¦A;¦[&^••ā)*Á;ā Á;&&`¦Á;@}Áo@Á å¦^å*^Á;[}å•Áae^Á;ç^¦-{[;ā)*Á;¦Á;ão@3)ÁGIÁ@[`¦•Á ]¦ā;¦Ág[Áo@{ Á;ç^¦-{[;ā}*ĚÁ	U]^¦æaāį}Á	Û*æk¦^ÁTæ);æ*^¦Á
ÙY I Î Á	P[ &ā; Á; āļÁ^} •`¦^Á@æok@æok@@A[[[åÁ;d[¦æ*^Á &æag}æ&ãč Á; -Á@Á;ãr^Ás; Á; [Á/•••Á;@æaj, Á;@Á;]^E°çā; cā]*Á •d;  æ*^Á&æg}æ&ãč ÁsæÁseļÁ;cæ*^•Á; -Á@A;]^¦æaā;} •ÉÅ Ö^cæaap Á; -Á@Áse;æaajæaa]^Á[[[åA;d[¦æ*^Á&æg]æ&ãč Á; āļÁ à^Á^][¦cråAşi,Ás@ÁOEDT ÜÉÅ	U]^¦æaa‡}Á	Û*æ¦hÁTæ)æ*^¦Á
ÙY I Ï Á	Ú æ)dŹA``ð]{ ^}óÁz)åÁ;æ&@)^\^Áx@ee/5a ÁæcÁã;\Á\[{Á -{[[åÁ;æe^\+oÁx[Áa>Áx[[ç^åÁx[Á@3a@\Á':[ĭ}åĔA	U]^¦æaāį}Á	Û*æ¦^ÁTæ)æ*^¦Á
ÙY I Ì Á	Òçæ&ĭæe^Áræe-ÁæeÁ^ĭĭāl^åÈÁ	U]^¦æqāį}Á	Û * æl¦^ ÁT æ) æ* ^¦Á
ÙY I JÁ	Y@\¦∧Áң[[åÁ,æc^\⊧Áţç^\ æ],Á@Aà`}åÉ&ed]]¦[]¦ãæec^Á ,æc^\Á¢•c3j*Áq[Áà^Á&ed-¦ã*åÁţ`óÁq[Áţ^æe`\^Ác@eeÁ ,æc^\Á `æ‡ãĉÁ^č¦}•Áq[Áàæe^Á^ç^ ●ÈĂ	U]^¦æaāį}Á	Û*æ¦^ÁTæ)æ*^¦Á
	Á		

## HUV`Y`) !&` 6`i Y`[ fYYb`U`[ UY`U`Yfh``Yj Y`g`

•

Õ≚ãã^ ậ,^Á	Ü^&[{{ ^}å^åÁæ&cãį}●Á
;fYYb`@YjY`GifjY]`UbW¥'AcXY V@ Á{caa∲áa[Áç[ ັ{^Á <sub>t</sub> -Áqd Á,[c*}cãad `Á{cã&ÁC}} ^DÁ &`aə][àa&c*¦ãa∞áa[^•Á,[c4×¢&^^å/aFÈEA({ EŠÁa)åÁ àā[ç[ ັ{^Á, Áqd Á&cãa)[àa&c*¦ãa∞áLlÁ;{ EŠÈÁ	●ÁT[}o@;Áæ{] ^●Á¦[{Áæè^Á
5 a VYf`@VjY`5`YfhAcXY V@Á{cæ4Áàã[ç[]ັ{^Á;Áœ4Á][c^}cãæ4î^Á{c&ãAQc}}îDÁ &`æ3][àæ&c^¦ãæ4æ3)åÅàã[ç[]č{^Á;Áœ4Á&cæ3][àæ&cc¦ãæÁ NFÈEÁ({ EŠÁLFÈEÁ;{ EŠEÄ	<ul> <li>Á Q[¦C; ät @lîÁ:æţ ]  ^•Á![{ Á;eà^Á</li> <li>Á Ù&amp;:/^}Á:æţ ]  ^•Á![!Á*/^}^@Exkag:æsæî ÁţiÁ</li> <li>] ![ å* &amp; Áţi ¢ā] ÁţÔî æ; [ Öc/&amp;Á*•dDÁ</li> <li>Á</li> <li>GÁ*/^}^@Exka*•cā] *Á\$^c/; { ãj ^•Á; [ oÁţi ¢ã8LÁ^ç^!oÁaæ8. Á</li> <li>[ Á [ } @@îÁ* ¦ç^ā]æ; &amp; Á( [ } ãţi ¦ā; *ÉÁ</li> </ul>
5 Whijch ``YjY`FYX`AcXY` V@Á{cad+%aä{ç[ ´{^Ár-Áse Áse&c`ad+Á{c&&A &`að][àaa&c^läadANFHEÁ;{ EŠÁQ&[}-ä{^áA^}^c&aad]^DÁ [¦Ásaā[ç[ ´{^Ár-Áse Á&`að][àaa&c^läadANIÁ;{ EŠEÁ	<ul> <li>A Uāł}æt^Ásqh'lcāj*Ák[A][ch]ažk*aġ*^\!Á[A]acæ-EÂ</li> <li>B Zaškoj !+ EŹ&amp;jä? co Asġ å Aşã ãi !+ LÁ</li> <li>A UA^Á, ÁUÚÒÁ[:A][/A][/A]A</li> <li>A W+AÁ, ÁÚÚÒÁ[:A][/A][/A]A</li> <li>A W+AÁ, ÁQ</li> <li>A W+AÁ, ÁQ</li> <li>A W+AÁ, ÁQ</li> <li>A A A A A A A A A A A A A A A A A A A</li></ul>

.

## \* "Á A cb]hcf]b[ `UbX`fYdcfh]b[ `

## \* '%Á 9bj ]fcba YbHJ` ]bgdYVMjcbg``

Á

Ü[čāj^Á, ^^\|^Áş,•]^&aţi}•Ásô@ÁÛčæl;^ÁTæ)æ\*^k4Q;k4s^|^\*æ\*DÁ, ä|A, &&`k6@[č\*Q;č4@A []^¦æ±aţi}æA/ãa^āţ ^A, Ás@Áčæl;^ÁtjÁs@^Å;ãa;Ása, čæ;ÅA)ãa\*ÉQ &A[āA, k4, æ\*\Á^|æ\*\å/ä;•č^•Á\*&@ & ^\[•āti}ÊA, āA]ä|ÊA, [[¦Á, æ\*\'ÁčætačÊ&=t\*æ\*A, čæ;ha\* ĉa;hæ ÊX •ã;\*Ás@AEnvironmental Inspection Checklistā;Ás@Ò;çāt[}{^}cætA[š];ša;ka;àATæ;æ\*A, ^}câ;hæ;ha\*A;

## \* ''& Á A cb]hcf]b[ `

Q•]^&cāt}•Ē4、[}āt[¦ā)\*Áa)åÁ^][¦cā)\*Án]^&ãa38Át[Ás@A(æ)æ\*^{ A};æ}æ\*^{ A};ā/ææ}åÁ,æ\*¦Ás@æ¢,á]As@æ¢,á]As^Á āt]|^{ ^} c^åAsčiā)\*Át]^¦ææat}}Át,-Ás@Aččæk¦^Áæ\*c\*åAs^|[,Áb;Á/ææi|^AîËEE2æd[}\*Á;ãc@A;@;ÆsiÁ ¦^•][}•āa|^ÈÁ

V@A\$j&\{ā[ Ádiāf\*^¦Açaaĕ^•A\$jÁ/aaai|^ÂEEAseA^Asiae=^åA[}€c@AsejåЦÁO€c@Aj,^¦&^}a#Açaaĕ^•A[A\*ā1@A '[`}å•Aj,-Aj,[}c@ÇÁ;[}ãā[']ā;\*Á}å^¦cæa\*^}AşjÁs@Aj,^¦ājåÂÙ^]c^{{ à^¦ÁO€c€ÎËCE\*\*•oAO€€ĨÈÂU[{ ^Á çaeč^•Á@æç^Asi^^}Áçaeā®åAq[Á^-4^&oAÔ[}åãaā]}ÁJAj,4Â&@@å`|^Á+DÉÁ

Á

## HUV`Y`\*!% Gc]`UbX`k UhYf`a cb]hcf]b[`dfc[fUa`

O≣]^&oÁ	Ø^~~^}&^Á	Ô^œŧ•À	O5jæ†^∙ãÁæ)åÁ5jc^¦ãįÁslāt*^¦Áçæ)ĭ^•Á	Ü^•][}•ãaậãc Á
Y^anc@¦Á	Öæn Á	Y ^æc@¦Á[¦^&æe or Á, āļļÁs^Á; [}ãt[¦^âÁ; Áş,-{¦{ Á`æ}¦^Á []^¦æzā]} • ĒÁ{¦Á¢æa; ] ^KÁ •Á QÁæāj Ási Ái; !^&æe dÆ ^åã; ^}ofæej åÁ';[•ā]} Á&[}d[ •Á, āļlÁs^Á &@ &\ ^åÆej åÁ; æāj cæāj ^åĔÁ •Á QÆsi¦^Á, ^æc@¦Ásej åÁ; āj å•Ásc*Á{{¦^&æe dÆsi`•of&[}d[ •Á; ā]lÁ à^Ási; ] ^{ ^}c*åĔÁ	ËĂ	Ûૻæ¦ʿÁTæ)æ*^¦Á
Üæği,≁æ¢ļÁ	Öæn Á	Üænāj-æn‡ÁaazÁv@Aj.¦^{ãr^●Aj.č●O%a^Aj.^æeč¦^åAsajåÁ^&[¦å^åAjājÁ {ā jāj^d^•Aj.צÁGIÁ@jč¦Åj.^¦ājåÊsaazkā@Aiæ;^Aásāj.^Ai>æ&@AiæêÉA	ËÁ	Û˘æ¦^∕Ææ}æ*^¦Á
Ò¦[•ā[}Áæ)åÁ∧åã[^}ơÁ &[}d[ ●Á	Y^^\ îÁaa)åÁ{[  [ ā];*Á ¦æaājÁQNF€Á({Á§jÁCIÁ@DÁ	Ò¦[•ā[}Áæ)åÁi^åã[^}ơ&{[}d[ •Áæć^Áq[Áà^Á([}ã[¦^åÁæ)åÁ {æã);œã),^åÉ&æ Á^č ă^åÈA	ĔĂ	Û*æ¦^∕Ææ}æ*^¦Á
Ó `^ÁÕ!^^}ÁQE*æ÷Á {[}ãt(¦ã)*Á,ãc@3,Ác@A ^¢clæ&cãt}}Áec^æÁ	U&4[à^¦Á[[ÁDE]¦ã ÁÁ -{¦ò}ã@4^Á TæÂ4[ÁÛ^]c^{à^¦ÁEĂ {[}cO4^Á	CEÁ æţ ]  ^Á ậl,Áa/Á&[   ^&c^åÅ4[ { Ác@ Ár¢d æ&cậ } Á,[ } å•Áæ) åÁ æ) æţ°•^åÁ[ ¦Áa] ´^Á ¦^^} Áæ‡ æ EĂ GÁ/• č  or Áā å æær Á&^  Á ¦[ , coátj ÁæÁrç^ Ár¢&^^åā] * Á €€Á &^  •Ð; ŠEÁ ^^\  ^Á æţ ]  ā * Á āl,Áa/Áā[ ]  ^{ { ^} c^åA } æðA ` &@Á æį ^ÁærÁc@ Ár•č  or Á; Ac@ Ác*•cā] * Áāj å ææer Ácí €€∕&^  •Ð; ŠÁ[ ¦ÁæÁ ] ^ ¦āj å Áj -Át ¦^æer ¦Ác@a) Ár4j [ } c@ Á	ŁÍ <del>€É€€Á&amp;</del> ∧∥•ÐĮŠÁÇTEàæ°¦**ð][•æÐÁ ŁIÁ({Á-HEŠÁÇE[cæ4/àsā]ç[ ĭ{^DÁ	Ûૻæ¦ʿÁTæ);æť^¦Á
	Ûĭæld^¦ ^Á	Ô@[¦[]@  ÁæÁ	GËF€Áµ*BŠÁ	Ûĭæ¦îÁTæ);æ*^¦Á
Ù*¦~æ&∧Á,æe^¦Á {[}ãt[¦ã)*Á,ãc@3)Áo@∘Á ^¢clæ&ca[}}Áee,^æÁ	Τ[}c@βÂ	Yæe^\¦Áĭæ¢ãcîÁ([}ãq[¦ā)*ÁæeÁÚ[}åÁFÁæ)åÁÚ[}åÁGA[[&æeā[}●Á[}Á Øätĭ¦^ÁûËFÐÁ	] PÁRAÍ È EÁ ÁÍ ÉÍ Á Ò ^&cla8æa‡Á&[}å" & cár, ãr ÁRAÍÍ ÉÍ €Á, Ù B&(Á Öã•[ ç^å Á[¢^*^} ÁRAÍNI È E€Á, *BŠÁ V″¦à ãã ãr ÁRAí CEÁÞ V WÁ U ã Ása) å Á″¦^æ•^ÁRAÍF€Á, *BŠÁ	Û*æ¦^ATæ}æ*^¦Á
Á	Û~ æơ\¦ſ Á	Yæe^\¦Á∵æ¢ãĉÁ;[}ãq(¦ā)*ÁæeAÚ[}åÆkæ)åÁÚ[}åÁGAA[&ææā;}∙Á;}Á Øãtč¦^ÂîËFEĂ	$\begin{array}{c} Ce\check{A}\!$	Ûĭæk¦îÁTæ);æt^¦Á

O≣]^&oÁ	Ø^~~^} & Á	Ö^æ	C5;æ¢î•ãrÁæ)åÆi;c∿¦ãį_Átå*^¦Áçæqĭ^•Á	Ü^•][}•ãaąããc Á
Á	Û`æơ¦ſÁ	X^¦ca8aapAj¦[-aậ^Á;[}ãq[¦āj*ÁaeeAj}^É;^d^Aşi@^¦çaap+ÁjāµAsi^Á `}å^¦caa}^}ÁşiÁs@Aasacāç^Á?¢d:asacāj}Ásed^adeAi	] PÁEÁ ÈEÁ Â Ě Á Ò ^&da&a¢Á&[}å čaāçāĉ ÁËÁLÍĚ €Á, ÙE&(Á Öã•[ ç^åÁ;¢^*^}ÁEŃ ÈE€Á, *EŠÁ V°¦àãbāĉ ÁËÁLG€ÁPVWÁ UāÁ&a)åÁ;¦^æ•ÁÉÁLF€Á, *EŠÁ	Û`æ¦^ÁTæ}æ*^¦Á
Á	Y @} & \$ a & & & & & & & & & & & & & & & & &	Úæ{] āְ*ÁæÓÚÚCEÚ[ā]dFÁæ}åÁÓÚŒÁ[ā]dGEÁæA@]}Á§IÁ Øā`!^ÂĒFEAÚæ{] ā]*ÁæÁ{Á @ā`!^ÂĒFEAÚæ{] ā]*ÁæÁ{Á @ā`!AÊFEAÚæ{] ā]*Áæ @áb@Aæ ^ç^}d6æ£aā]*Á{[{Áæā}-æ‡{Á;-Á¥gg'h UbACEĂA{{A§}A{{coh};c^!A æa∱^{ā}:aa‡A`!ā{Aā;AA} @Á@Á&{}& A QÁ@Á&{}& }dæaā{}Áā }A a A QÁ@Á&{}& A A A A A A A A A A A A A A A A A A A	UậÁæ) å/t"¦^æe ^Á Á≂ậAýãããa ^Á ] PÁÂÈËËËĂ VÙÙÁÁe€Ą(*ESSÁ	Û*æk¦^ÁTæ)}æ*^¦Á
Ù″¦-æ&∧Ájaer∿¦Á	Üæaj-æn∦Ðç^}o%aæe-^åAæ)åÁ ĭæko∿¦ ^Á	Yæ&^!Á * æ¢ač Á [ ] ãi [ ¦ā * Áseekb@ Á [ &ææi] • ÁÙY HĐÙY I ĐÙY I Á æ) å ÁÙY F€Á } Á2ā * !^A ĔFÁ [ *  å Åa ^ &[ ] å * &c^â Á [ ] o@ f Áeg â Á å * ¦ā * Áæi* ^ Áæij -æ  Á ç^ } o ĐÁ Á ÖÚOB ÒÁse& } [ ,  ^ å * ^ • Ás@æe Á @ !oko*.!{ Á ¢ & ^ åæi &	] PÁÁĚEĔĂ ÒÔÁÁÈÌ €€ËGI €€ÁµÙES4{Á Ù`•]^}å^åÁ[jãã•ÆÄkGÍÁ{*EŠÁ Öã•[lç^åÁ{¢^*^}ÄÄÅIÁ{*EŠÁ V[cæÁjáãt[*^}ÆÄkFÁ{*EŠÁ V[cæÁjá@[•]@[¦`•ÆÄkEEÈÈÁ{*EŠÁ	Û`æ¦^ÁTæ}æ*^¦Á
Ùd^æ( àæ)∖Áæ)åÁa^åÁ ]¦[-a]^Á	Üænāj-æa∦Đç^}ơ‰sæer^å/æajåÁ čăed∾¦¦^Á	Ùcl^æ; àæ)\Áæ)åÁa^åAj¦[~ā¦^Áæ)åÁ&[}åããā]}•Áæ6k@A[[&ææā]}•Á ÙYHĒÂÙYIĒÂÙYJÁæ)åÁÙYF€A[}A28ā`¦^ÂĒEÁ_ã Aâ^Açãa`æ ^Á ãj•]^&c^åAå`¦ãj*Á`¦-æ&^Á,æc'¦Á;æ;] ãj*Ávç^}orÈÁ	CE;^Á&@æ);*^●Ásĩ^Áq[Árãz^Á;]^¦ææā;}●Á æl^Ásã^}cāæ?åÁse);åÁ^]æãa^åÈÁ	Û ǎ æ¦ ʿ ÁT æ) æ* ^¦ Á
Ù] āļÁ āzÁ	T[}c@(îÁæ))åÁ{[∥[,ậ),*Á č∙∧Á	V@^Á1]ā‡lÁādáārÁt[Aà^Á&@~&\^åAæ)åAæ)^Át[ā•ā]*Át[æe^¦äæd+Át[Aà^Á ¦^] æ&^åÈÁ	Á	Û*æ¦^ÁTæ}æ*^¦Á
Ō¦[`}å,aner∿¦Á	T[}o@¢Á	V@Á*¦[`}å, æe^¦Á;[}ã#ţ¦ậ*Á,^  Á[&ææ‡]}•Áæċ^Á;@}Á;}Á Øat`¦^ÂEHEKÖŠÚFEKÖŠÚFEKÖŠÚÍEKÖŠÚÍAæ}åÁÖŠÚÍÁæ^ÁţÁa^Á {[}ã#ţ¦^åAa`¦ậ*ÁUœet^ÁFA;Áo@Ár¢dæ&caţ}EKÖŠÚÍEKÖŠÚÍÁ ÖŠŰÍEKÖŠÚÌE&æ}åÁÖŠÚF€Æek^ÁţÁa^Áţ[}ã#ţ¦^åAa`¦ā;ÁUœet^ÁGÁ [-Áo@Ár¢dæ&caţ}EÁ	]PÁÁBĚÁÄËEÁ ÒÔÁÁLOEEÁ(ÙE&(Á Öã•[ ç^å/ţ¢^*^}ÁËÄNFĔEÁ(*EŠÁ Š^ç^ ÆÄLOEÄÁ&@ee)*^Á¦[{Á@áq[¦&Bæe‡Á  ^ç^ ●Á	Û`æ¦^⁄nTæ)æ*^¦Á

C€]^&oÁ	Ø1^˘˘^}&^Á	Ö^æ	O5jæ†^∙ãrÁæ)åÁ5jc^¦ã;Ád'ã*^¦Áçæ†ĭ^•Á	Ü^•][}•ãaãããc Á
	Ûĭæ¢r\ ^Á	ÜÚCBOÁssak}[,  ^å*^•Ás@eseÁ@;lókr\{Ár¢&^^åæ)&^•Á, Ás@•^Á [àb%&äp,*•Á, æŝ Á, &&`lÁi`iā]*Á, æč`lædÁrç^} orÁ`&@ése Á{[[åā]*ĚA ÖÚCBOÁssak}[,  ^å*^•Ás@eseÁ,!^Èr¢ä cā]*Á, æc*lÁ`ætjáč Á, æŝ Á, [oÁ { ^^óko@A, àb%&ãç^•Á; IÁ[{ ^Ks}; æt] c*ÖžBj &{`åā]*A ætjäj ãč ĔA P[[&ā] A, `•oA dãç^A; A; A, ^^ók@A; æc*lÁ`ætjäč A, àb%&ãç^•Á c@[`*@5ii]]^{ ^} cætji}Á; ^^ó@AU[äjAsej åA'æc*lÁT æt) &cãr,*A c@[`*@5ii]]^{ ^} cætji}Á; ^ác@AU[äjAsej åA'æci]A æt} A c@[`*@5ii]]^{ ^} cætji}Á; As@AU[äjAsej åA'æci]A æt c@[`*@5ii]]^{ ^} cætji}Á; As@AU[äjAsej åA'æci]A æt c@[`*@5ii]]^{ ^} cætji}Á; As@AU[äjAsej åA'æci]A æt c@[`*@5ii]]^{ ^} cætji}Á; As@AU[äjAse cæt]] A as at a constant of the set	$\begin{array}{c} Ce\check{A}[]cQ\check{A}[]\check{at} \check{a}*\check{E}\check{A} \check{A}\\ Cf\{[\}\check{at}\check{A}\check{A}\check{D}\check{A}\\ \hat{O}\check{at}\check{A}\check{S}\check{A}\\ \hat{O}\check{at}\check{A}\check{S}\check{A}\\ \hat{O}\check{at}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\\ \hat{O}\check{at}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\\ \hat{O}\check{at}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\\ \hat{O}\check{at}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\\ \hat{O}\check{at}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\check{A}\\ \hat{O}\check{at}\check{A}$	Û≚æ¦^ÁTæ)æt^¦Á



Î AGEJEY @e-ok-çv:/&advA@e-da-v).katv.k@e-da-v).katv.k@e-da aj 80000/gaj al/kat (aj e1850000/ az vA[A]/.v-v).comaj) • A i A sitoj obe-kat (ad ban kat

## \* " Á 7 cbh]b[ Yb\\/md`Ub'

QÁvo@Áxæà[ç^Á;[}ãī[¦ā]\*Áå^c^&orÁæ}Áā;]æ&dÉazeÁ&[}cā]\*^}&`Á;|æ}Á;¦Ad:ā\*^¦Áæ}åÁ^•][}•^Á;|æ}Áā;Á{ à^Áā;]|^{{ ^} c^å Éaze Á;@\_;}Áā;Á/æà|^Â ËCEĂ

## HUV`Y`\*!&`7cbh]b[YbWnd`Ub`

V¦ð**^¦Á	Ü^•][}•^Á
Gc]`g	
Ò¦[•āį}ĐÃi∧åãį ^}caeaáį}ĐÃi åæ{æ*^åÁ&[}d[ •Áį¦Ácĭ¦àãaÁ,æe∿¦Á [à•^¦ç^åÁ	•Á Qa^} cã^ Ác@Á[`¦&^Á; Ác@Á]¦[à ^{ /kæ}å Ákæà^Ác@Á,^&^••æ^Ác^]•Á ¦^``ã^åÁk[Á;¦^ç^} cÁæÁ^&`;!^} &^ÈÁ
5 WJX gi `ZJhY gc]`g	
Ùði}•Áæ&ððáĂ` æ&A[ā/Aæ]åÁ;æ&A[ { æ}æ*A{ ^}œA[a/Aæ]åÁ;æ&A ^~^&@a;^KÁ •Á Ÿ^  [, Á*~+[[^^ & &^} & &^/Å]Å[ā/A •Ă Ÿ^  [, Á*~+[]^^ & &^} & &^/Å]Å[ā/A •Ă ÇI]}Å;æ&A •Á ÇI] •Á ÇI] •Á ČI] Å Cæðjā]*Á;Å[ā+Á]¦Å;æ&A •Á ČI], Å PÁ[ā+Á;!Å;æ&A	<ul> <li>• Á Ô[} cæði Ás@ Á; æc^¦ãæd-keð å Á`}[~-ÈA</li> <li>• Á Ùæ{]  ^Ásði å Ás'^ææÅ@ Á; æc^¦ãæd-kði Ásæ&amp;&amp;[¦åæði &amp;^Á, ão@k@ Á;¦[&amp;^å`¦^Á ä Á0£]]^} å ãr/ÁÔÈÅ</li> <li>• Á Ú^çãr, Á;¦[&amp;^å`¦^• Át[Ár}•`¦^Á; æði æt'^{ ^} ofár Ár~^&amp;cãr,^ÈÅ</li> </ul>
Øãr@Áå]•Áæ••[&ãæer∿åÅ;ão@kæ&ãåÅ •č -æer∖Á[āµ%ä[]æ∨Á	●Á Ùd[]Á,[¦\●Á\$\$;{^åãæe*\^Áæ)å/\$\$;]/\{^}oóko@ÁÔT ÙÁ\$\$;&ãå^}oÁ ]¦[&^å`¦^ÈÁ
6`i Y'; fYYb'5`[ UY	
ŒĘ*æa∲áa∥[[{●Áţà•^¦ç^åÁÁ	<ul> <li>Á Q.&amp;\^æ~^/Aj.e]^&amp;cqi.}+ÁiÁs@.Ár¢clæ&amp;cqi.}Aj.[}å/Ai[Å; ä&amp;/Åiæaji*ÁæjåÁ æe^\!Ái[}äqi.!ä*Åqi Á; ^^\]ÉA</li> <li>Á Č)•`!^Á,[Å,`à]ä&amp;/Ås&amp;&amp;^• Aqi Á; æe^\!Ás[å^EÅ</li> <li>Á Č)•`!^Á,[Å,`à]ä&amp;/Ås&amp;&amp;^• Aqi Á; æe^\!Ás[å^EÅ</li> <li>Á Ù@``àA, @.e]@.![`•Ár¢c^!e /År^A/8i[`å^*Å/@#`@.!Ás@ej Ás@.Ás.!är/äæEÅ å[e]*Á, As@.Aj.[}å.e/As[`]å/Ai^A &amp; [`å^*A^*AC]:å^*Å/@#`@.!Ás@ej Ás@.As.!är/äæEÅ å[e]*Á, As@.Aj.[}å.e/As[`]å/As^A &amp; As[`à^*A^*AC]:å^*A*As&amp;@j.äj/Ai, Aset { &amp; ag) å/A^*]•`{A @ ']å/As^A &amp; -åEV@A` ag) cãt Á; A*As&amp;@.Astai Aset { &amp; ag) å/A^*]•`{A @ ']å/As^A &amp; -åEV@A` ag) cãt Á; A*As&amp;@j.äj/As A å^cr:{} aj^å/As^As@.As[`]`{A Ase^*}AsA &amp; a*AsA &amp; a*Astai AseAsA &amp; AseAs { [!^/Asej]![]!ässer/Asej å/A` &amp; &amp; &amp; ** - `]Ássak@.Astai AseAsAstai AseAsAstai } Å &amp; A`D}*at^As[`]•`[cag) c/Aj ÁssacAstai AsiAs@.Ase]]![]!ässer/Aj äst aseaj }È</li> </ul>
GifZUWY'k UhYf	
Yæ&^¦Á`æ‡äĉÁ^∙` orÁ∿¢&^^åÁs@A ¦^ ^çæ}oÁtiã*^¦Áçæţĭ^Á	<ul> <li>Á W} å^\cæa ^ Áæáa^cæāp^å/ðşi] ^ &amp; cāi } Ái - Áædi/Á&amp;i } d[   • Áædi å kæði å k••• Áædi ^ Á ã • č ^ Áædi ^ ¿aði à leði</li> <li>Á ÖÚCB Ò Q-Áæái à j ã aði aði diçad äæeði } Áði Áædi ^ Á, æe^\: / Á P Á: / ÁÖÔÁi &amp;&amp; č \• Áædi Áædi A å æði ^ í aði aði aði aði aði aði aði aði aði aði</li></ul>
Ø″^ 63&@?{ &8æ¢(Á*)]ð∥A∱}Áæð)åÁ	•Á Qa^}cã^ÁæjåÁq[]Ás@Á[ĭ¦&^Á,Ás@Á^æà,Á;¦Á]ā)IÈÁ •Á CEJ][^Ás@Á]ā]Áãók[Á& ^æ)Á]Ás@Á]ā]IÈÁ •Á Öã][•^Á,Ás@Á*•^åÁ;æe*¦ãæd∲&∞ÁsÆjås?}•^åÁæ)å-ā]IÈÁ •Á Ü^][æ&^Ás@Á;æe*¦ã懕Á*•^åÁ§As@Á]ā]IÁãaÈÁ
Ø″^ 62&@{&8æ¢A*]ā¦lĄ[}Á;æe∿¦Á	<ul> <li>Á Ö!^å* ā) * Át Á&amp; æ^ Áa) å Áad Á* `ā] { ^} ófa Ásaðaj át Á* - ÉÅ</li> <li>Á O[[Át] à át] } Á[ `   &amp; • Át Áa Á* Á* { [ c^åÁ' [ { Á:@ Ásaðaj át Á* - Á@ Á] āl ÉÅ</li> <li>Á Ø * A[D] ā/Asaðo [   àā] * Á,   [ å ` &amp; o Åt Á* • ^ åÁt Á&amp; ] čæt áj æð á Å [ æ Å ] ál ÉÅ</li> <li>Á Ø * A[D] ā/Asaðo [   àā] * Á,   [ å ` &amp; o Åt Á* • ^ åÁt Á&amp; ] čæt áj æð á Å [ æ Å ] á</li> <li>Á Ø * A[D] ā/Asaðo [   àā] * Á,   [ å ` &amp; o Åt Á* • ^ åÁt Á* ] čæt áj æð á Å [ æ Å ] á</li> <li>Á Ø * A[D] ā/Asaðo [   àā] * Á,   [ å ` &amp; o Åt Á* • ^ åÁ* [ Å &amp; o Å å Åt Á* ] čæt áj æð á Å [ æ Å ] Á</li> <li>A Ø * A[D] ā/A [ Å &amp; o Åt Á* • Å &amp; o Åt Á* • Å &amp; o Åt Á* • Å &amp; o Åt á* • Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Å &amp; o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Å &amp; o Åt á* o Å* o Åt á* o Åt á* o Åt á* o Åt á* o Å* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Åt á* o Å* o Å* o Å* o Å* o Å* o Å* o Å* o</li></ul>

V¦ã*^¦Á	Ü^•][}•^Á
;fcibXkUhYf	
T[}ã[¦ã]*Á^•` o•Á¢&^^åÁs@Á ¦^ ^çæ)o√stã*^¦Áçæ,ĭ^Á	<ul> <li>Á Wy å^\cæa ^ Áædá^cæað ^ å/ðg • ] ^ &amp;caí } Ár - Áæd / &amp; [   • Áæd å Áædå å · • • Áæd ^ Á ã • ` ^ Áãa ^ cæað å ÉÅ</li> <li>• Á QÁæÁ ã } ãã&amp;æd o kjæd ãæañ } Ág Át   [ ` } å , æe ^ l Á P É Ó Ô Ái   Ár o c^ l Ár &amp; &amp; ` ! • Á æ Áæá ^ `   o Ái - Ái } E ãr Áædcañ ãã • É ái o c^ • cã ær Ás@ Á&amp;æč • ^ Ái - Ás@ Á ] ! [ à   ^{ { Aeg à Åæa ^ } cæa £ / A @ ed co Øed à Aeg i   o c c či ær Ás@ Á&amp;æč • ^ Ái - Ás@ Á ] ! [ à   ^{ { Aeg à Åæa ^ } cæa £ / A @ ed co Øed à Aeg i   o c c či ær Ás@ Á &amp; ` ] ]   ^ Ái } Á e A OE • • • • A A A @ ed co Øed à Aeg i   a&amp;o A - A * ! [ ` } å , æe ^ ! A * ] ]   ^ Ái } Á e A OE • • • • A A * cæañ } A @ ed co Øed à Aeg i ] a&amp;o A - A * ! [ ` } å , æe ^ ! A * ] ]   ^ Ái } Á æå b f ã ã * Á ! [ ] ^ cã • E Å</li> <li>• A U^ cã , A ! [ &amp; ^ å ' ! • A &amp; [ ásec [ ãa Ás@ Aeg • ` ^ / Ag A * c ' ! ^ E Å</li> <li>• A QÁ ! [ à   ~ { Á : • a o É A } * æ* ^ Áæd&amp;[ ] • *   cæ) o Ag [ Áæå cã ^ A [ } A &amp; @ A eg ] ! [ ] i ãæe ^ Ái ( ãa e æñ ] È</li> </ul>
:`ccX]b[	
Ø[[[åậ]*Áåæ; æ*^•Á~˘≩{ ^}ơÁ	●Á Ü^çã^, Á&[}d[ ●Áq[Áæç[ãåÁ&æq[æ*^Á&`¦ã]*Áң[[å●ÁşIÁo@∘Á*č¦^ÈÅ
Ø[[åậ,*Á∿¢&^^å∙Á,^¦ą́^ơ\Á à`}å•Á	<ul> <li>A Ü^çã, Ác@ Á@ ã @Aţ As`}å•Áţ Á`}•`¦^Ác@^Âsd^Ásd^ás^•ã}A @ ã @A\$</li> <li>A Qç^•cã æc^Áş &amp; ^æ ã * Ác@ Á@ ã @Aţ Ac@ Ás`}åÊ\$ş Á&amp;{ }•` cæeāt }Á ã @ADUBOÉA</li> </ul>

## \*''(Á FYdcfh]b[

V@ A\*^}^!a¢A^][:aţ \* A^~ ă^{ ^} a^{ ^} o Asd^As • & añ à Aş ÂÙ^&aţ } Â È Aţ Aso AÔT Ù ĔQ A^|æaţ } Áţ Á ![`cġ ^Aţ [}ãt !a] \* Áso A^ a|Aso A^ 8t !a^aAţ } Áso AEnvironmental Inspection Checklist āj Áso Ô} çã[ } { ^} cæAT [}ãt !a] \* Áso à AT a) a# ^{ ^} o Ad()a) ĚA

QxÁ^|æaāj}Áq[Á, æa^\A^æq[]|^•Êab@^Áq[||[, ā]\*Á, [č|åAba^Á^&q[¦å^åÊabe Áaz44, ā]aījič[KÁ

- •Á V@Aåæe^ÇDA[}Á,@a&@Aó@Aíæ{]|^Á,æaÁæà^}Á
- •Á V@Áaã,^ĢDáaæÁ,@3&@áv@Á;aaé]|^Á,æá&[||^&c^åÁ
- •Á V@^Á,[ā,cÁæcÁ,@3&@k@cA^æ{]|^Á,æe/Áæà^}Á
- •Á V@Á,æ{^Á(Á;@Á,^!•[}Á,@{Á&[||^&c^åÁ@Á;æ{]|^Á
- •Á V@\Áæ&cāçāa?t•Á;&&`¦¦ā;\*Áå`¦ā;\*Ác@\Á;[}ã[¦ā;\*Á
- Á OEÁ&[{]ælãa[}Á(Ác@\Á^•`|o•Á,ãc@kó@Aseå[]c^åA&lãcAlãæÁ

OEÁ^][¦oÁ,ā|lÁa\^Á,¦^]æ}^åÁa`Ás@AÛĭæ¦`ÁTæ}æ\*^¦Á{[||[¸ā]\*Á\ç^¦^ÁFGÁ([}o@A(,-Á([}ã[¦ā]\*Áæ)åÁæÁ •`{{æ^^Á,-Ás@A([}ā[¦ā]\*Á^•`|o•Á,ā|lÁa\^Á,¦^•^}&\*åA§A(xB;}`æA(Ü)][¦óA(Ç^-~\¦Á[Ás@AOTÙDĎÁ

QÁxay Á^¢&^^åæy &^Á(; Áx@exÁkiãe^¦ãæxÁzi Á^&[¦å^åÉAs@Áxee-^&c^åÁæy å[, }^!•Áxay åÄÖÚOBÒÁ, ä|Aki^Á,[cãæ?åÁ ā) Á, ¦ãxā) \* ÉExee Áki^•&¦ãæ^åÁ9, ÁÙ^&ca‡; }ÂÁ; Áx@ÁÒT ÙÉExay åÁ;¦[çãæ^åÁ;ãx@Á`æec+¦|^Á;[}ãt[¦ā) \* Á^•ĭ|o•Á č}cājÁx@Á^•ĭ|o•Á;@; Áx@exeÁx@Á;¦[b^&cÆsiÁ&[{]|^ā] \*Á;ãx@As@Á^|^çæy c∕&¦ãe^}ÄÁ

OĘ∥Á^&[¦å∙Á,ã∥Áà^kÁ

- ●Á Tænäjcænäj^åÁköjÁxæ4Å^\*äå|^Á{[¦{Á
- ●Á S^] cÁ{¦ÁæeÁ^æ•cÁ Á^æ•Á
- •Á Ú¦[å`&^åÁq[Áæ)^Áæ`c@;¦ã^^åÁq;~ã&^¦Áq;Áœ?ÁÔÚŒ&æ)åÐ;¦ÄÖÚŒ©ÒÁ][}Á^``^•cÁ

Á

Á

## +"Á FYj ]Yk <sup>·</sup>UbX<sup>·</sup>]a dfcj Ya Ybh<sup>·</sup>

Ô[}cāj`[`•Áā[]¦[ç^{ ^}o^{A}case AÛY T ÚÁ, āļlÁs^Áse&@a>ç^åÁājÁse&&[¦åæ)&^Á, ãc@ác@ AÔT ÙÉ&@[`\*@Á c@ Á;}\*[ā]\*Á^çæ;≊æāi}Á;~Á}çā[]{ ^}cæ‡Á ][|a&a>•É4;àb^&cā;^•Áse}åÁseæ\*^œÉÁ

V@?Á&[}cājĭ[ĭ•Á&[]¦[ç^{ ^}c^{ ^}c^{ \*}i[&^•\*Á&^^at}}^åÁ&[kÁÁ

- •Á Ga^}cã~Áse^æ Á[-Á]][|č}ãĉÁ[|Áã[]|[ç^{ ^}oÁ[-Á?]çã[]{ ^}cæ Á[æ]æ \* { ^}oÁse)åÁ ]^!-[!{æ)&^Á
- •Á Ö^ç^|[] Ázej å Áži ] |^{ ^} o Ázej a Áj /a Áj /a Ázej a Áj /a Ázej a Áj /a Ázej a Áj /a Ázej a Áj /a Ázej a
- $\bullet \dot{A} = X^{\dagger} \ddot{a} \dot{a} \otimes \dot{A} \sim \delta \tilde{a} \wedge \delta \tilde$
- •Á Ö[&`{ ^} cheap ^ A&@eap \* ^ A\$ A\$ |[ & ^a` |^ A^ ` |eap \* A' [{ A\$ ![ & ^ • A\$ ] ![ ç^{ ^ } dheap å A\$ eat ^ A &[ { ] æba [ } • A\$ ab & cap ^ • A\$ ab & cap ^ • A\$ ab a A\$ eat \* ^ o A

Á

Á Á Á Á

5ddYbX]Wg

; < 8 ÁQÜ^][¦ơÁ{¦Á?[|&ã[ ÁQÍOE•dæ)ãæÁDÍÚc ÁŠcåÁÉÏÖ`}|[^ÁÛæ)åÁŰ`æ\'^ÉÉGCGGEEÍÎÁ

## $5ddYbX]I ~~5\acute{A}\acute{A} \stackrel{(2)}{=} ^{\circ} \& \acute{A} \stackrel{(2)}{=} ^{\circ} |caaa_{i}| \& \acute{A} \stackrel{(2)}{=} ^{\circ} |caaa_{i}| \& \acute{A} \stackrel{(2)}{=} ^{\circ} \\$

Á Á

## **Ben Luffman**

From:	Ben Luffman
Sent:	Thursday, 8 August 2019 12:50 PM
То:	'landuse.enquiries@dpi.nsw.gov.au'
Cc:	'Victoria Musgrove'
Subject:	Dunloe Quarry Management Plan consultation
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
RepoType:	Jop

Hi,

We have updated the management plans for Dunloe Quarry follo g the nt ap roval of MOD2. T e conditions of the Project Approval – SSD 06\_0030 require a number of the lans to be ared in with the Dol. We have therefore provided a link below to the r vant plans for review.

### https://ghd.sendthisfile.com/M3RFj9HigPcjj1ATu8LUMpEj

The updates have mainly a reformattin to remove du on and inclusion of additional information to address the w requ ents of the

We would appreciate your c by 23 August 2019.

Please contac me if you ha e any q ons.

### Regards

### Ben Luffman | A GHD Associate

B.App.Sc. (Hons) | Grad.Dip. Urban and Regional Planning | Environmental Auditor **Technical Director - Environment** 

### GHD

#### **Proudly employee owned** T: +61 2 6650 5613 | M: +61 415 271 319 | E: ben.luffman@ghd.com 230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.ghd.com</u>



WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION

GHD acknowledges the Traditional Owners of Country throughout Australia. We pay respect to their continuing culture and Elders past, present and emerging. <u>Click here</u> to learn about our Reconciliation Action Plan.

D`YUgY'Wcbg]XYf'h Y'Ybj ]fcba YbhiVYZcfY'df]bhjb[ 'h ]g'Ya U]` 1 ream of paper = 6% of a tree / 5 CO2 in the at osphere | 3 heets of A4 paper = 1 li e of water

## **Ben Luffman**

From: Sent: To: Subject:	Geff Cramb <geff.cramb@epa.nsw.gov.au> Wednesday, 21 August 2019 11:20 AM Ben Luffman RE: Dunloe Quarry Management Plan consultation</geff.cramb@epa.nsw.gov.au>
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
RepoType:	Job

Dear Ben

The EPA do not intend to reviewdetmnt plans. The EPA are cont with thescope. Howe er, it is undethat EPA will uake complince reviews against the rnts othe-tProtection Lissued and the impltion of the management plan at our discretion.

k

8

### **Geff Cramb**

Operations Officer – Environment Management Unit North Coast, NSW Environment Protection Authority

+61 2 6640 2510

<u>geff.cramb@epa.nsw.gov.au</u><sup>''''</sup><u>www.epa.nsw.gov.au</u><sup>''''</sup><u>@EPA\_NSW</u> Report pollution and environmental incidents 131 555 (NSW only) or +61 2 9995 5555



I work flexibly. I'm sending this message now because it's a good time for me, but I don't expect that you will read, respond to or action it outside of your own regular hours.

From: Ben Luffman < <u>0</u>	>	
Sent: Thursday, 8 August 2019 12:51 PM		
To: Pe er Ly ch <u>h</u> O	>	
Cc: Janelle Ba croft K "		>; Victoria Musgrove
>		

Subject: Dunloe Quarry Management Plan consultation

Hi Peter,

Not sure if you are theperson to conta t but we have uated the mant plans for Dunloe Quarryfollowing the recent approval of U \)The conditions of the Project Approval – SSD 060030require a number ofthe plans to be preparein constion with the -h°We haverefore attached the re evant plans for r view.

The updates have mainly a reformattin to remove du on and inclusion of additional information to address the w requ ents of the

We would appreciate your c by 23 August 2019.

Please contac me if you ha e any q ons.

### Regards

## Ben Luffman | A GHD Associate

B.App.Sc. (Hons) | Grad.Dip. Urban and Regional Planning | Environmental Auditor **Technical Director - Environment** 

### GHD

### Proudly employee owned

T: +61 2 6650 5613 | M: +61 415 271 319 | E: <u>ben.luffman@qhd.com</u> 230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.qhd.com</u>



WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION

GHD acknowledges the Traditional Owners of Country throughout Australia. We pay respect to their continuing culture and Elders past, present and emerging. <u>Click here</u> to learn about our Reconciliation Action Plan.

D'YUgY'Wcbg]XYf'il Y'Ybj jfcba YbhVYZcfY'df]bhjb['il ]g'Ya U]' 1 ream of paper = 6% of a tree / 5 CO2 in the at osphere | 3 heets of A4 paper = 1 li e of water

#\V7@-Vu@LITY NOTICE: This email, i any attach s, is confi l and may be privileg If you are not the i ded re please notify the sender i , and please d you s ould no copy it or use it for any purpose or discl se its con nts to any ot er person. GHD and i s affiliates reserve the right to monitor and modify all email communi s through their n ks.

## 

Vj ku'go cki'ku'kpygpf gf 'hqt''y g''cf f tguugg\*u+'pco gf ''cpf ''o c{ ''eqpyckp''eqphkf gpykci'cpf lqt''r tkxkigi gf '' kphqto cykqp0''

Ku"{qw'ctg''pqv'y g'kpvgpf gf 'tgekr kgpv.'r ngcug''pqvkh{ 'y g''ugpf gt 'cpf 'y gp''f gngvg''kv'ko o gf kcvgn{0' Cp{ 'xkgy u'gzr tguugf 'kp''y ku''go ckrictg''y qug''qh'y g'kpf kxkf vcn'ugpf gt ''gzegr v'y j gtg''y g''ugpf gt ''gzr tguun{ ''cpf '' y ky ''cwy qtkv{ ''uvcvgu'y go ''q''dg''y g''xkgy u''qh'y g''Gpxktqpo gpv'Rtqvgevkqp''Cwy qtkv{0'

## RNGCUG'E QP UKF GT "VJ G'GP X KT QP O GP V'DGHQT G'RT KP V KP I "VJ KU'GO C KN"

aaaaaaaaaaaaaaaaaaaaaaa"" Vj ku'g/o cki'j cu'dggp'uecppgf 'hqt'xktwugu''

# 5ddYbX]I `6Á ÁÒ¦[•ą] } Áæ) å Á^åą̃[ ^} ơ &[ }d[ |Á] |æ)•Á Á Á



Ground surfaces will be kept damp (not wet). An on-site water cart will be available at all times. Surfaces shall be left in a rough cloddy condition to increase roughness and slow surface wind speed. Temporary access roads and parking areas shall be sealed with a gravel layer.

#### **Erosion Controls**

Works programme shall be scheduled to minimise the potential for soil loss. Works programme shall be scheduled to minimise the potential for soil loss. Sediment and erosion controls shall be installed prior to clearing and include: \* Diversion of stormwater around disturbed areas. \* Sediment control fences at the downslope perimeter of cleared and/or disturbed areas. These controls shall be functional prior to commencing upslope work. \* A negative grade towards the dredge ponds should be maintained for the area within the perimeter bund.

the perimeter bund. Stormwater numoff shall be directed away from construction entry/exit points. Temporary erosion measures (sill fences, catch dreins, perimeter banks and diversion channels) are to be employed onsite where reasonably deemed necessary. Access is to be provided for maintenance and sediment removal works. Perimeter bund is to be constructed prior to the commencement of resource extraction. Upon its completion the perimeter bund is to be vegetated/seeded.

#### Surface water monitoring

All surface water tested to conform with the following criteria:

Parameter	Release Criteria	Criteria Type
pH	6. 0 - 8.5	Range
Dissolved Oxygen (field measured)	>6.5 mg/L	Minimum
Oil and Grease	rease No visible film, No detectable odour	

Sampling and analysis of dredge pond surface water should be undertaken monthly

#### Acid sulfate soil identification

Soils recovered using dry excavation are to be tested in accordance with the ASSMAC guidelines. A minimum of 10 samples per quarter are to be collected for CRS/TAA analysis from the resource recovered by dredging.

#### Acid sulfate soil treatment

Soils requiring lime treatment will be treated to neutralise their equivalent TPA or equivalent validable sulfur, incorporating a mixing factor of safety of 1.5. Materials used to construct the bunds will be free from acid sulfate soils or suitably

treated. Exposed sides of open drains are to be treated with lime immediately after excavatio Pollution control

Petroleum and other chemical products shall be prevented from containing surface

Peroteum and other onemical products shall be prevented from containing surace water and soil. Any onsite fuel storage areas shall comply with Australian Standards. Adequate trade waste and litter bins shall be provided onsite and service d regularly. Concrete wastes and washouts shall not be deposited in any location where the wastes or washings can flow, or can be washed into any areas of retained vegetation or receiving waters.

#### Rehabilitation and landscaping

Progressive stabilisation of areas where construction delays occur and revegetation of completed areas. All landscaping and rehabilitation shall be completed so that a duration of less than 60 working days will elapse from final land shaping to permanent rehabilitation. All temporary erosion and sediment control works are to be removed once works are complete and revegetation is successfully established in for merk disturbed areas. Drainage channels are to be rehabilitated immediately after completion.

#### **Contractor Management**

Review of the ESCP and the works contracts by the proponent. Periodic checks to be made by an independent Environmental Consultant. Training for construction staff in implementation of ESCP provisions. Staff to be trained to implement dust minimisation measures.











Site Layout/Clearing Limits

Image source: N.C. White & Ass

Not to Scale

6

SILT FENCE SECTION 



 $5ddYbX]I \quad 7 \acute{A} \acute{A}Baa{} \acute{A}J^{*} | -aae^{A} \acute{A}J[a] \acute{A}T a a at ^{{}} at ^{{}} ( ^{{}} a \acute{A}J | a a \acute{A}J | a a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{A}J | a c ^{{}} a \acute{$ 

## 5VIX Gi ZUHY Gc]` =XYbhjZJVUhjcb UbX HfYUha Ybh

V@:ÁOBBaãÁUĭ|~aet^ÁÜ[āļÁTaa)aet^{^}oÁÚ|aa)Á@aetÁa^^}Á\^]aetaet^åÁ§iq[Áo@:Á{||[,ā]\*Ájaetorká

- Á OESaã ÁÙ (-æe^ ÁÙ[ ã¦Á/¦^æe{ ^} oÁ

## 5VIX Gi ZUHY Gc] HXYbhjZJVUhjcb

### 8fmiYI WUj UhYX'a UhYf]U`g`

OĘIÁ\*`¦~æ&^Á(; æe^¦ãedp•Á(iÁs^Ás¦^Ár¢&æçæe^åÁ; ā||Ás^Á≈a; ]|^åÁse&&[¦åāj\*Á(iÁs@Á(i||[;āj\*Á);[d[&[|Á;}Á æÁ cæt^åÁsæ•ã Á,¦āj¦Á(iÁs[{{ ^}&^{ ^}of,-Ás@Ár¢&æçæaāj}}Á,[¦\•EÁ

Sampling – Ù[āļÁæ;]|/•Áæ]]¦[¢ā;æ\*/`ÁEÈLÁ\*Á>æ&@k[Áa^Á&[||^&cvåÁ¦[{Á≈&@Á[āµ́@;¦ã[}Á;¦ÁæáÁ |^æ•cÁEĚLÁ;Áşcv¦çæ‡+Áå[,}Ác@Á[āµ́,¦[-ā]^ĚV@Á[āµí,¦[-ā]^Áå^•&&iā]cā;}Áæ)åÅå^]c@á;Á^æ&@á;æ;]|^Á ārÁt[Áa^Á/&[¦å^åĔĂ

Ù[āļÁ;æ;[]|^•Á;[Áa^Á\$[||^&c^åÁ\$jÁ^æ†^åÁ\$[}cæāj^¦•Á;¦Á\*^[|[\*38æ‡Á;æ;]|āj\*Áaæ\*•Ás@æeÁ\*¢&|čå^ÁæāiÈĂ Ùæ;[]|^•Áæb^Á;[Áa^Á^}cá;[Á@?Áæà[¦æ;[¦~á;¦Á;[:^}Á;ã@3jÁGIÁ@;`¦•ÈĂ

Analysis. ÁÙæ;]|^•Ásch^Á;[Ásu^Á&¦^^}^åÁ;[¦Ásc&ãa Á`|-æe^Á[ã;Á;[c^}cãe;Á;)åÁ}å^!\*[ÁÔ@[{ã{ Á Ü^å`&ãa|^ÁÙ`|]@¦Á;ĈÜÙDÁse)åÁ/[cæ;ÁQB&čæ;ÁQB&ãã;ấ,ÁçVOEDEÁse)æ;\*^•A\*ÁscaÁse)Áse]]¦[]¦ãæe^Áæà;[¦æ;[¦`ÈÁ

 $\begin{aligned} & (ACE) UT CEO ACFJJ Delescal Action Act$ 

### <mXfUi`]VVJ`mYI VVJjUhYX`a UhYf]U``

Tæc^¦ãa‡Á^¢dæ&c^åÁs^Á@妿ĕ|ã&Ás¦^å\*^Á;ā∥Ás^Á;æ;]|^åÁæ&&[¦åãj\*Á;[Ác@Á;[||[,ãj\*Áj¦[q[&[||KÁ

:fYeiYbWmÁÁ/^}Á;æ{]|^•Á;^¦Á ˘æ¦ơ\¦ĚÁ

GUa d`]b[ÁÂÛæ;] |^•Á, āļlÁà^Á&; ||^&c^åÁ; ||[,ā,\*Á@^Á,æ@; @; \*Á;¦[&^••ÈĂ

Ù[ā]Á æ{]|^•Á;[Á\$a^Á\${[||^&c^åÁ\$;jÁ^æ|^åÁ\${[}cæ];^\•Á;¦Á\*^[|[\*38æ4Áæ{]]|ð]\*Á\$aæ\*•Ás@æeÁ¢&3|čå^Áæ3iÈÁ Ùæ{]|^•Áæ4^Á;[Á\$a^Á^}oÁ;[Ás@Áæae`[¦æe[¦^Á;¦Á+[:^}Á;ão@3jÁGIÁ@{`¦•ÈÁ

5 b **U ng]g**描ට 2aදi] |^• Ásc^ Át { Áta^ Á & '^^ } ^ å Át ¦ Ásc8aã Á` |-æc^ Á [ 毳Á, [ ơ^ } cãc‡Áse) å Á` } å^ !\* [ ÁÔÜÙ Áse) å Á VCED Źsej æf` • ^ • ÁsceÁsej Ásej ] ¦ [ ] ¦ãæc^ Áæia [ ¦æt [ ¦ˆ ĚÁ

 $\begin{aligned} & (AOEUUT OEOA(FJJ) Dévescaļ ) Asiār \ & archi + AEEHA UA^* * arcad^ ) oA* * |] @ \A_i + AFI A_i [|AFNAD_i ] ) ^ A_i + Arc@ A \\ & ^* * arcad^ ) oAvesta = Archi +$ 

Á

## 5V[X Gi ZUHY Gc] HfYUha Ybh

### 8fmYI WUj UHYX'a UHYf]U`g`

Ö¦^Á\¢&æçæaāį}Á,,-Á[ā+Á\¢@ãaãā3;\*Áæ&ãaÁ`|-æe^Á,[c\}cãa¢E£aæe^åÁ,}Ác@Áæà[¦æq[¦^Á^•`|o•E5aáA(Á×A d^æe\åÁ§Áæ&&[¦åæ}&\Á,ão@á@Á[|[],ā]\*Á,¦[&^å`¦^•hÁ

- •Á Ù[ā•Á^č ăā]\*Ád^æe(^}oÁj ǎ]Áà^Ád^æe\*åÁj ão@Áaj ^Á; ¦ÁæÁš ãæà|^Á; ^č dæpā ā]\*Áet\*}oÁ; Á
  •Á Ù[ā•Á^č ăā]\*Ád^æe(^)oÁ; Áa]Áà^Ád^æe\*åÁj ão@Áaj ^Á; ¦ÁæÁš ãæà|^Á; '~; ÈÁQ,Á&æAš |æaā]\*Áo@Áet([č]oÁ
  >^č dæpā ^Áx@ ãAčč ãçæh }oÁ/ÚOEA; !Áčš ãçæh }oÁ; cãa ãæà|^Á; '~; ÈÁQ,Á&æAš |æaā]\*Áo@Áet([č]oÁ
  (Á] ^Á; !Á, ^č dæpā ā]\*Áet ^}oÁ; Áb
  Aét /Áa ^áæá\*à\*âÊéæá(; ãçā]\*Áæ&(; !Á; -Á æAcč Á; -ÁEĚ Á; āļlÁà^Áš, o\*àÈÁQÁãA
  &æ) /áa^Áå^{([])}•dæe\*àÁxô@æaÁæá\*à£éæá(; !Áã; Á][oÁ, ^\*à\*àÊÉæÁF; KFÁæaā; Á; ā]lÁà^Áš; ]|^{ { ^} c\*àÈA
- •Á V@Á&æ4&č |æe^åÁæ4 [č } oÁ; -Áã; ^Á; !Á,^č dæ4ã ã; \*Áæ\* ^} oÆi Át; Áå^Á] !^æåÁ; ç^!Ás@Áv¢dæ&cã; } Á ] !ã; !Át; Á&[ { ^} &^{ } &^{ } oÁ; -Áv¢&æçææã; } ÈÉT ã;cã; \*Á; -Ás@Á; æe^!ãæ4ª Á; ã|A; &&č !Áæ Ás@Á[㪠Áse^A Á ^¢&æçæe\*åĚÁ
- •Á OE‡ Átl^ææ^åÁţ ææ^¦ãæ‡+ÁşāţAŝa^Áşiæ&^åÆşiÁ] ææãæ‡ţîÁtlæ&& ^åÁæ4^æe Áşãe@aşiÁ@eAşi^¦ãţ ^cv¦Áa`}åÁ æ)åÁ}å^!\*[Áç^¦ãæ&æætậ}Åx'•cāj\*ÁœxAsæAşiÆxAsæAşi ãşãţ č{ÁşiÆ€Aætş]|^•AşiÆ@eAşi![&^•e-^åÁ •æ)åÁşiç^¦ÁæAščæcv¦ţîÁşi^lãţåÁt[¦Áæsjæ‡î•ãrÁsîÁs@AÔÜÙED/OEDE4şi ^c@gåÈAU}^Ásætş]|^Á;@gč|åÁsa^Á &[||^&cvåÁ¦[{Áræ&@aşi}E`ãxAşi![&^•e-āj\*Á}ãtĚA
- •Á QÁç^¦ãaBæeaāti} Áz cāj \* Ášti åaBæez Ášti } cāj čæp Áæaāti |^ Áti [Åti ^^ cáti a Bæez Ášti } cāti a Bæez Ášti } cāti a Bæez Ášti } cāti a Bæez Ášti } cāti a Bæez Ášti } cāti a Bæez Ášti } cāti a Bæez Ášti } cāti a Bæez Ášti } cāti a Bæez Ášti } cāti a Bæez Ášti a Bæez Ášti a Bæez Ášti a Bæez Ášti a Bæez Ásti a Bæez Ásti a Bæez Ásti a Bæez Ásti a Bæez Ásti a Bæez Ásti a Bæez Ášti a Bæez Ášti } cati a Bæez Ás

### <mXfUi`]VV/`mYI VV/j UHYX'a UHYf]U``

P^妿ĕ|a&æq|^Á^¢&æçæaā[}Á, æc^¦ãæqÁ^¢@ãaãa3]\*Áse&ãaÁ\*`|-æc^Á,[c^}cãaqÉ&aæ-^åA;}Ás@-Áæà[¦æq[¦^Á ¦^•`|œ É&á{Ás:^Ás!^æc^å/&j.Áse&&{[¦åæ]}&^Á;ã@Ás@A{[|[、ã]\*Á,¦[&^å'¦^KÁ

- ●Á V@^Áąĩ^Áæ)åÁ,æe^¦Á,ã¢č¦^ÁãaÁ[Áà^Áa]¦æî^åÁ,}d[Ás@A,æe@åÁæ)åÈÁ
- •Á OE[[Át]^æer\åÁ, æer\äæt+Á, ä] Ásh^Á, |æ&r\åÆj Á] æaæt+[`Át]ææt+Å Åtæ&t ^æ Á, ãr@aj Ás@Aj,^¦ā, ^cr\Ás`}åA æ) åÁ`}å^\\*[Ásr\äæ&æta] År\•cā] \* ÁserAstÁætAjær\Á, Ásæk(jā) ä, `{Át, Ár€Áæt, ] |^•Á, Ás@Aj,![&r••^åA •æ) åÁt, ç^\{ÁsetĂ`æcr\![`Ásr\ät åÁt[¦Áse) ætî•ã Ásî Ás@AÔÜÙED/OEDEAt, ^c@t, åÉtU}^Átæt, ] |^Á;@t` |åÁshÁ &[]|^&cråAt\[{ Áræ&@at,}E` ērAt,![&r••ā] \* Á; ã ÉtĂ
- •Á V@Áæ)åÁjāļÁj[oÁa^Áclæ)•][¦c^åÁj~~Áãc^Á'}cājÁç^¦ããBææāj}Ác•cāj\*ÁšjåãBææ?•Áæ&&A] cæà|^Á [¢ãããæàà|^Á`]]@¦Á&[}&^}dæāj}•ÁšjÁæ&&[¦åæ)&^Ájão@Á@ÁDÈÙÙT OEÔÁ`ãa^|ãj~•ÁšÈÈÉc@Á ^``ãçæ}^}oÁ`]]@¦ÁšiÁr••Ás@e)ÁEÈHÃÙÁe)åÁs@Á``ãçæ}^}oÁs&ãããc ÁšiÁr••Ás@e)ÁFÌÁ;[|Á PMEQ[}}^ĚÁ
- •Á QÁç^¦ãaBaeeaāt} Áz cāj \* Ástj å aBaeez Ást[} cāj \* adÁæaāt` ¦^Át[Át ^^cát] ^ &ãaBaeez Ást[} cāj \* adÁæaāt` ¦^Át[Át ^^cát] ^ &ãaBaeez Ást[} cāj \* At[ At] \* At]
- Á

ÕPÖÁ

GHEÁPæà[˘¦ÁÖ¦ãç^Á Ô[~•ÁPæà[˘¦ÁPÙYÁGIÍ€Á VH FÁGÂÎÍ€ÁÎ΀EÁÁØKÂ;FÁGÂÎÍ€ÃÁÎÊ€FÁÁÔKÁ&•{æã;O\*@aÈb2[{Á

î ÁÕPÖÁG€G€Á

V@arÁå[&č{^}ofarÁa)åÁ @eeplÁ^{æajÁœA,'[]^\c´Á, AÕPÖĖŹV@Aå[&`{^}of4; æáÁ;}|^Áa^Á + ^åÁ; ]`|][•^Á{[¦Á, @a&@áaÁ, æaÁ&[{{ã•ą}}^åÁa)åÁa)åÁa)åÁa)åÆa&{[¦åæ)&^Á;ãa@k@Á/^\{•Á;ÁÔ}\*æ\*^{^}oÁ[¦Áa@A &[{{ã•q}}ÈÁV}æčo@;¦ã^åÁ•^Á;ÁœãrÁå[&č{^}ofajÁa)åÁa)^Á{[¦{Á}@æep[^c,^\ÁárÁ];[@aaãc\*åÈÁ COEÜUEÜIÎÎÌIIÎ]GÍË ÌĐ@c]•KBD);[b%&orÈ® @abB{{Đ}&PD^,&æed^HĐ@;&ã]åč}][^•æ)åč`æBÖ/ac^\?BÖ[&č{^}or 50200€€ÍδÜÚ

ÌĐôởg, • Haop) ¦[b/& or È ôà È&[{ Đ} & Do^, & and d^ HĐôų |& a[ å` } |[^ • a) å``ab HÖ^ |ãç^ ¦^ HÖ[ & `{ ^} or HDGGOEEEÍ Î´ÜÚ V´Ö` } |[^ ÁU[ā] Aba) å ÁY and ^! AT an) and ^{{ ^}} of U[an) Èã[ & cÁ

Ü^çãið[}Á	OEco@;¦Á	Ü^çã∿, ^¦Á		OĘ]¦[ç^åÁ{¦¦Á	<b>@</b> ∙`^Á	
		Þæ{^Á	Ùãt}æcč¦^Á	Þæ{^Á	Ùãt}æcč¦∧Á	Öæe∿Á
€Á	ÓÁŠĭ~{a∌Á	ÙÁŠæç,∧¦Á	0	ÙÁŠæç^¦Á		G€B€J£D€FJÁ
FÁ	ÓÁŠੱ~{æ}Á	ÙÁŠæç,∧¦Á	tan	ÙÁŠæç,∧¦Á	fan	GF₽F€ED€€G€Á

Ö[& { ^} oÁÛcæeč • Á





Á

Á

5ddYbX]I '9Á ÁŠæ) å• &æ] ^Á( æ) æ\* ^{ ^} ( ^) ø) k



## <c`V**]**a `f**5**i ghfU`]UŁ`Dhm@hX`

Ö`} [[^ÁÛæ) åÁÛ \* æ¦^/
Šæ) å• &æ] ^ÁT æ) æ\* ^{ { ^} @£ A^ @£ A^ @£

R″|^*Á*G€GFÁ

Á

V@āÁ^][¦dxÁ@æe,Áa^^}Á,¦^]æi^åÁa^ÁÕPÖÁ[¦Á?[|&ã[ÁÇE•dæpäæDÁÚĉÁŠcå,Áæ)åÁ(æê,Á;}|^Áa^Á•^åÁ æ)åÁ^|ā?åÁ;}Áa^Á?[|&ã[ÁÇE•dæpäæDÁÚĉÁŠcå,Á{¦Ás@;Á,`¦][•^Áæ\*¦^^å,Áa^ç,^^}ÁÕPÖ,Áæ)åÁs@;Á P[|&ã[ÁÇCE•dæpäæDÁÚĉÁŠcå,Áæe,Á^oA;`óAşÁ^&cā[}Á,FECA;,Ás@ä;Á^][¦dEĂ

ÕPÖÁ(c@o¦,ã^A&aã&akaaā(•Á^•][}•ãaājāčÁ(Áa)^^\•[}Á(c@oká@a)ÁP[|&ã(ÁQCE•da‡ãaeDÁÚcŠcåÁ ælãāj\*ÁşiÁ&[}}^&cā);Á,ão@áo@áÁ^][¦dĚŐPÖÁad;•[Áv¢&|čå^•Áa[]|ðråÁ,æl¦æ)cðt•Áæ)åÁ&[}åãaāj}•É&[Á c@ Áv¢cr}oÁr\*æ|^Á,^¦{ã•ãa|^ÈĂ

V@:Á^¦ça&^•Á`}å^¦cæà^}Á\$i^ÁÕPÖA\$jÁ\${[}}^&caįi}Åjãc@4j¦^]ædāj\*Ás@aiA^][¦c4j^¦^ÁqiįãevåA\$[Ás@[•^Á •]^&ãã&ed|^Á\$i^cæaj^åA\$jÁs@:Á^][¦c4se)åÁse'^Á`àb^&c4j[Ás@:Á\*&[]^Áqiįãæeaji}•Á^c4jÁs@:Á^][¦dEÁ

V@A[]ājā[}•Ê&[}&|`•ā] •Áa)åÁa)^Á^&[{ { ^} åæaā]}•ÁajÁx@aa Á^][¦oÁad^Áaæa^àÁ][¦oÁad^Áaæ ^åÁ]}Áæ •`{ ]cā]}•Á { æå^Áa`ÁÔPÖÅa^•&¦ãa^åÁajÁx@aA`][¦dĚÁÕPÖÅaã&a|æaā[•ÁãæaàājāĉÁadāā]\*Á¦[{ Áad}^Á\_AœA æ •`{ ]cā]}•Áa^ā]\*Áaj&[¦|^&dEĂ

ÕPÖÁ@ee Á,¦^]æ\^åÁx@ex Á^][¦ớң}Áx@ Ásæe æi Ąi, -Áşi,-{¦{æaāţ}Á,¦[çãā^åÁs^ÁP[|&āţ ÁçCE • dæjæađÁuć ÁšcáÁ æ)åÁ(c@:¦•Á,@(Á,¦[çãā^åÁşi,-{¦{æaāţ}}ÁqiÁÕPÖÁgāj&|`åāj\*ÁÕ[ç^\}{^}ơ⁄sečc@[¦ãæ?•DÉ3,@a&@AÕPÖÁ @ee Á,[ơ⁄sξiå^]^}å^}d^Áç^¦ãæ?åÁqiÁ&@ &\^åÁs^^[}åÁs@ Áse\*!^^^åÁ\*&[]^Áqi-Á,[¦\ÈÕPÖÁs[[^•Á,[cÁ æ&&^]ơÁāæàājãč ÁşiÁ&[}}^&cāţ}Áşiã@Á`&@Á`}ç^¦ãæ?åÁşi,-{¦{æaāţ}Ê5kşi&|`åāj\*Á\*¦![¦•Áse}åÁqi{ã•āţ}•ÁşiÁ c@Á^][¦ơÁ,@a&@Á,^!^⁄&æe\*^åÁs`Á\*;![[+4,i]á{ã•āţ}•ÁşiÁœæAşi,-{¦{æaāţ}È6kşi&|`àāj\*Á\*;![

# HUV`Y`cZWzbHybhg`

FÈ	Q;d[å	لَمُنْسَنَّتُونَ اللَّهُ عَلَيْنَ المُنْتَقَاتِ { المُعَانِ المُنْسَنَّقَاتِ المُنْقَاتِ المُنافَ المُعَانِ المُ
	FÈ	Cą A
	FÈG	Uàb &ã;^• Á
	FÈH	
	FÈ '	
GÈ	Ò} çã[	}{ ^}œ¢Á^~~~ã^{ ^} @ Â
	GÊ ,	$\dot{S}^* \tilde{a} \mid a a \tilde{a} \mid \dot{a} a \dot{A}^* \tilde{a} a \mid \dot{a} \wedge a \dot{A} = A \dot{A} = $
	GÈG	Ô[}åãāā[}•Á[·Áæ]]¦[çæp <mark>ÆDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDDD</mark>
HÈ	Ò¢ã cậ	)*Á*}çã[}{ ^}cÁæjåÆą[]æ∨ A <del>uliiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiii</del>
	HÈE	Ò¢ã cậ * Á } çã[ } { ^} d
	HÈG	Q]æ&øÁ
ΙÈ	Ò} çã[	}{ ^}œŧ/&[}d[ Áţ ^æ`¦^• Á
ÍÈ	T[}ão	[¦ð] * Ázið á Á^] [ ¦æ] * A <del>ð 1111 1111 1111 1111 1111 1111 1111 </del>
	ĺÈ	Ò}çã[ }{ ^}œ¢∕ã•] ^&œį́ }• <b>Å</b>
	ÍÈG	T [ } ất ¦ậ * Á
	ĺÈH	Ô[ } @; * ^ } & Á   æ) Â
	ĺÈ	Ü^] [ ¦@] * Á

## HUV`Y`]bXYI '

Væà ^Á⊊ËF`	Ü^•][}•^Áţ Áæ*^}&@*•Á&[{ { ^}@ A
Væà∣^ <i>Á</i> ŒË <sup>™</sup>	Ô[}•^}ơ&{[}åãã]}•Á^ ^çæ)ơ&{[Ác@/ĞTÚÅ
Væà∣^ <b>Á⊣Ë</b> F	V@^æ^}^åÁ]^&&●Á
Væà ^Á.ËF`	Ò}çã[}{ ^}cæ∲&[}d[ •Áæ)åÁ{ãa∄æaậ}}Á{ ^æč¦^• Å <del>2000000000000000000000000000000000000</del>

## : ][ i fY ]bXYI "

Øðať`¦∧Á+HEF	$X^* ^{case} \} \& [ \{ \{ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \$
Øãtč¦∧ÁHËG	V@^æc\}^åÁ{[¦æ{{ &æa[}}●Â <del>11111111111111111111111111111111111</del>
Øãtč¦∧Á+HËH	V@^æe^}^åÁæĕ}æák^&{¦åðj*●á <del>kinninninninninninninninninninninninninn</del>
Øãtč¦∧ÁÍËF	Ú^¦{ æ}^}ơ∮,@;q;Á;[ā];o=Áæ)åÁ;[}ãq;+ä];*Á[&ææā]}●Á <del>ùÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌÌ</del>
•	
Á	
Á	
Á	

# 5ddYbX**]V¥**g<sup>.</sup>

```
CE[]^}åã¢ÁCEÁ→ÁÜ^@zeàiãjãzzetã[}Áse)åÁÜ^ç^*^czeztã[}ÁT ze)zet^{^}sÁÚ|ze)Á
```

```
O[[]]^}åã¢ÁÓÁ-ÁOE*^}&`Á&[}•`|cæeā[}Á
```

```
C0[] ^ } å㢠ÁÔÁ-ÁŠ[ } * Á/^ ¦ { ÁT æ} æ* ^ { ^ } cÁÙd æe^* ^ Á
```

```
O[[]^}åã¢ÁÖÁ-ÁS[æ†æá∕Tæ)æ*^{ ^}αÁÚ|æ)Á
```

```
OŢ;]^}åã¢ÁÒÁ-ÁT[}ãq[¦ā]*Á[¦{•Á
```

```
Á
```

```
Á
```

## 1. Introduction

This Landscape Management Plan (LMP) forms part of the Environmental Management Strategy (EMS) for Dunloe Sand Quarry. This LMP has been prepared to meet the requirements of the Minister's Conditions of Approval (CoA) outlined in Development Consent No. 06\_0030, the mitigation measures outlined in MOD2 (GHD 2017), the Environmental Impact Statement (EIS) (Planit 2007), the Environment Protection Licence 13077 (EPL) and relevant legislation.

## 1.1 Aim

The aim of this LMP is to describe the rehabilitation and biodiversity management strategies, procedures, controls and monitoring programs to be implemented to prevent or minimise impacts and facilitate effective rehabilitation of Dunloe Sand Quarry during operational and post-operational phases.

## 1.2 **Objectives**

To achieve this aim, Holcim will:

- Rehabilitate areas identified in the Rehabilitation and Revegetation Management Plan (RRMP).
- Protect and manage land outside the approved disturbance areas.

## **1.3 Targets**

The following targets have been established for the management of biodiversity and rehabilitation of Dunloe Sand Quarry:

- Screen the development from surrounding properties.
- Provide additional native habitat for flora and fauna through the creation of wildlife corridors.
- Protection and management of areas outside the disturbance areas.

Additional targets / completion criteria are included in the RRMP in Appendix A.

## **1.4 Consultation**

Extensive consultation was undertaken with the local community during preparation of the EIS and MOD2. Any concerns identified by relevant stakeholders were addressed in the EIS and MOD2 mitigation measures which have been incorporated into this LMP.

As per CoA 27(a), Schedule 3, the Department of Industry (DoI), Office of Environment and Heritage (OEH) and Department of Primary Industries - Fisheries (DPI Fisheries) now all part of the Department of Planning, Industry and Environment (DPI&E) and Tweed Shire Council were consulted in relation to this LMP. The Community Consultative Committee (CCC) will be updated about the revised management plans at the next meeting.

A summary of the agencies' comments and the response is provided in Table 1-1 and evidence of the consultation is provided in Appendix B. DPI had no objection to the LMP and no response was received from DoI.

## Table 1-1 Response to agencies comments

Agency comment	Response
Biodiversity and Conservation Division (previou	isly OEH)
Revises the Landscape Management Plan to	<b>)</b> :
Include the relevant qualifications and experience of the contributors to demonstrate compliance with Project Approval 27(a)	A letter dated 19/06/2009 from DPE approved the specialist who prepared the original Landscape Management Plan. As the current version is primarily a revision of the format, specialist input was not considered necessary.
Indicate in Table 2.1 that Project Approval Conditions 28 clauses (h) and (i) are addressed in Appendix C	The comment refers to the Koala Management Plan which is Appendix D. Conditions 28 clauses (h) and (i) in Table 2.1 updated to refer to Appendix D.
Revises the Rehabilitation and Revegetation Management Plan to include Project Condition 28 in the list of conditions addressed	The Rehabilitation and Revegetation Management Plan was prepared by a previous consultant and is not able to be updated.
Revises the Koala Management Plan to:	
Replace references to sections and tables of other management plans with the relevant content being referred to in order to minimise potential errors resulting from subsequent management plan revisions or amendments	References to relevant sections of other plans is considered appropriate because it avoids duplication and possible inconsistencies when one section is updated but not the other. Reference to Section 5.4.1 updated to Section 5.4 Form A in Appendix E includes koala observations, it is also included in the Environmental Inspection Checklist
Include mapping of koala habitat, koala records and potential koala movement corridors (i.e. habitat links) within and adjacent to the subject land and along the haul road between the quarry site and the Pottsville Road intersection	Table 3-1 and Figure 3-3 provide a description and map of koala sightings (noting the Flying Fox symbols in the north west of Figure 3-3 should be koalas)
Acknowledge the possibility of infrequent koala movement during hours of quarry operation	The Koala Management Plan has been updated to indicate the possibility of koala movements during quarry operations
Identify the most likely areas of interaction between koalas and quarry vehicles (e.g. koala habitat links)	Section D1.2.1 of the Koala Management Plan identifies the most likely areas of interaction – along the haul road adjacent to koala habitat
Include a proposed amendment to the quarry induction process to include an explanation of the legal consequences of unauthorised clearing of native vegetation on the quarry site	A dot point added to control B1 in Table 4-1 of the Landscape Management Plan
Include provision of compensatory koala food tree plantings as a contingency measure in the event of unauthorised clearing taking place	Compensatory koala food tree planting included in Table 1 of the Koala Management Plan
Ensure the proposed monitoring methodology focuses on identifying areas of koala activity susceptible to road strike rather than attempting to identify temporal changes in koala densities	Koala monitoring revised to focus on identifying areas of koala activity rather than temporal changes in koala densities

Agency comment	Response
Reduce the proposed koala road-strike threshold for management action from three koalas for the year to any koala at any time	Updated
Tweed Shire Council	
General comment:	
The bulk of the overall management plan comprises material from between 2006 and 2016 and it is difficult to interpret which previous study or work the published content refer to	Initial ecological investigations were undertaken and reported in the approved 2006 RRMP. This document was then updated in the 2016 RRMP, and attached to this LMP, to reflect the change to revegetation around the ponds. The RRMP is included to provide the historical context of the ecology and rehabilitation work undertaken for the site. Further amendments to the LMP are not considered necessary.
Hours of operation are proposed as the key threat mitigating factor in relation to koalas and vehicle strike. This does not satisfactorily account for winter, when the acknowledged high risk times of dawn and dusk occur during these hours of operation	Section 3.2.2 of the LMP and Section D1.2.1 of the Koala Management Plan have been updated.
The plan should note that koalas can be on the ground at any time of day or night	Section D1.2.1 of the Koala Management Plan has been updated.
The plan should also identify the times of the year when risk is higher due to seasonal movements of young males, presence of females with back young and the concurrence of haulage times with dawn and dusk during winter as per above	Section D1.2.1 of the Koala Management Plan has been updated.
All actions in relation to koala sightings and quarry related vehicle strike are assigned responsibility to the 'Planning and Environment Manager – NSW'. The document needs specify what organisation this relates to	Unless otherwise stated, roles / positions detailed throughout the EMS and sub-plans, including the LMP and Koala Management Plan, are internal Holcim roles.
No specific actions are proposed in response to quarry related vehicle koala strikes	The LMP/ Koala Management Plan proposes an adaptive management approach to ensure any management measures implemented are targeted and will be effective in managing site-specific issues. There have been no records of koala vehicle strike during the site's operation. The adaptive management approach outlined in the LMP/Koala Management Plan will be informed by ongoing monitoring and reporting in relation to koala impacts. A new Section D1.6 has been included in the Koala Management Plan outlining potential management measures which could be implemented where vehicle strike impacts are experienced. Any measures to be implemented will be selected in response to the findings of the investigation into the incident (e.g. location, contributing factors such as non-compliance with site rules, etc.) and in consultation with local koala experts

Agency comment	Response
There is no obligation or accountability for quarry staff to record koala sightings	This requirement is included in Section D1.3 of the Koala Management Plan. This has also been included in an update to Table 1 in Section D1.5 of the Koala Management Plan ensuring these requirements are included in the site's induction material.
3 koala strikes per year is too many for the endangered population to sustain	<ul> <li>Table 1 in Section D1.5 of the Koala Management Plan has been updated to include:</li> <li>Quarry-related vehicle koala strikes reach or exceed three for the year recorded for any two (2) years in a rolling five (5) year period.</li> </ul>
'Increased presence of koalas on haulage routes' is intimated as the likely cause for vehicle strikes. This is considered to be a premature and unfounded assumption	<ul> <li>The wording referred to in Table 1 in Section D1.5 of the Koala Management Plan has been amended to reiterate that adaptive management will be required in response to both:</li> <li>Increased occurrence of koala sightings on or nearby haul routes; and</li> <li>Increased incidence of koala vehicle atrikes as a result of guerry operations.</li> </ul>
## 2. Environmental requirements

#### 2.1 Legislation and guidelines

Legislation relevant to biodiversity and rehabilitation management includes:

- Environmental Planning and Assessment Act 1979 (EP&A Act)
- National Parks and Wildlife Act 1974 (NPW Act)
- Biodiversity Conservation Act 2016 (BC Act)
- Fisheries Management Act 1994 (FM Act)
- Biosecurity Act 2015
- Pesticides Act 1999
- Environmental Protection and Biodiversity Conservation Act 1999 (Commonwealth) (EPBC Act)

Further discussion of the above legislation is provided in the EMS, as well as the EIS and MOD2.

Guidelines referred to in the preparation of the LMP include:

- The constructed Wetlands Manual, Volumes 1 and 2 (DLWC 1998)
- Policy and Guidelines: Aquatic Habitat Management (DPI 1999)

#### **2.2 Conditions of approval**

The Development Consent conditions relevant to this LMP are listed in Table 2-1. A cross reference is also included to indicate where the condition is addressed in this LMP or other environmental management documents.

#### Table 2-1 Consent conditions relevant to the LMP

Condition No.	Requirement	Reference
Schedule 3, Condition 1	Within 1 month of the date of approval of the Landscape Management Plan (see condition 27 below), the Proponent must:	Table 4-1
	<ul> <li>(a) Engage a registered surveyor to mark out the boundaries of the approved limits of extraction;</li> </ul>	
	(b) Submit a survey plan of these boundaries to the Secretary; and	
	(c) ensure that these boundaries are clearly marked at all times in a permanent manner that allows operating staff and inspecting officers to clearly identify those limits.	

Condition No.	Requirement		Reference
Schedule 3, Condition 25	The Proponent must rehabilitate the site to the satisfaction of the Secretary. This rehabilitation must be generally consistent with the proposed rehabilitation activities described in the documents listed in condition 2 of Schedule 2 and comply with the objectives in Table 5.		Appendix A
	Feature	Objective	
	All areas of the site affected by the project	<ul> <li>Safe</li> <li>Hydraulically and geotechnically stable</li> <li>Non-polluting</li> <li>Fit for the intended post-quarrying land use/s</li> <li>Final landform integrated with surrounding natural landforms as far as is reasonable and feasible, and minimising visual impacts when viewed from surrounding land</li> </ul>	
	Surface infrastructure	Decommissioned and removed, unless otherwise agreed by the Planning Secretary	
	Void lake	<ul> <li>Water retained on the site maintains long-term water quality objectives fit for the intended post-mining purpose</li> <li>Water discharged from the site is suitable for receiving waters, aquatic ecology and riparian vegetation</li> </ul>	
Schedule 3, Condition 26	The Proponent must: (a) rehabilitate and revegetate the EA (see the revegetation plan in <i>b</i>	e 15 ha hectares of land identified in the Appendix 2); and	Appendix A
	(b) within 12 months of the comm make suitable arrangements to pu for the revegetation area to ensur purposes, to the satisfaction of th	encement of quarrying operations, rovide appropriate long term security re it is managed for conservation e Secretary.	Table 4-1
Schedule 3, Condition 27	<ul> <li>The Proponent must prepare a Landscape Management Plan for the project to the satisfaction of the Secretary. This plan must:</li> <li>(a) be prepared:</li> <li>by suitably qualified consultants, including a specialist hydrologist, coastal engineer, wetlands ecologist and landscape architect;</li> <li>in consultation with Council, Dol, OEH, DPI Fisheries and the CCC; and</li> <li>in accordance with extant guidelines including the Dol's Constructed Wetlands Manual, Volumes 1 and 2 and the DPI's Policy and Guidelines: Aguatic Habitat Management, 1999;</li> </ul>		This plan was updated by Ben Luffman (B.App.Sc. Hons). Refer to Table 1-1 for details regarding specialists.
	(b) be submitted to the Secretary prior to starting quarrying operations on the site; and		Completed circa 2009
	<ul><li>(c) include a:</li><li>Rehabilitation and Revegetation</li></ul>	on Management Plan; and	Appendix A
	• Long Term Management Strat The Proponent must implement the Note: The Department accepts the Plan may not include the detailed However, a conceptual strategy in along with a timetable for augment subsequent review of the plan.	tegy. The plan as approved by the Secretary. At the initial Landscape Management I Long Term Management Strategy. Inust be included in the initial plan, Intation of the strategy with each	Appendix C

Condition No.	Requirement	Reference
Schedule 3, Condition 28	The Rehabilitation and Revegetation Management Plan must include:	Appendix A
	(a) the rehabilitation objectives for the site and revegetation areas;	and Appendix D
	(b) a description of the short, medium, and long term measures that would be implemented to:	
	<ul> <li>rehabilitate and stabilise the site;</li> <li>implement the revegetation strategy; and</li> <li>manage the remnant vegetation and habitat on the site and in the revegetation areas;</li> </ul>	
	(c) detailed performance and completion criteria for the rehabilitation and stabilisation of the site and implementation of the revegetation strategy;	
	(d) a detailed description of how the performance of the rehabilitation of the quarry and the revegetation areas would be monitored over time to achieve the stated objectives;	
	(e) a detailed description of what measures would be implemented over the next 5 years to rehabilitate and manage the landscape of the site and revegetation areas including the procedures to be implemented for:	
	<ul> <li>progressively rehabilitating and stabilising areas disturbed by quarrying;</li> <li>implementing revegetation and regeneration within the disturbance</li> </ul>	
	areas and revegetation areas;	
	<ul> <li>protecting areas outside the disturbance areas, including SEPP 14 wetlands and SEPP 26 littoral rainforests;</li> </ul>	
	vegetation clearing protocols;	
	<ul> <li>managing impacts on fauna;</li> <li>controlling terrestrial and aquatic weeds and pests;</li> </ul>	
	controlling access;     bushfire menagement; and	
	<ul> <li>reducing the visual impacts of the project;</li> </ul>	
	(f) a description of the potential risks to successful rehabilitation and/or revegetation, and a description of the contingency measures that would be implemented to mitigate these risks;	
	(g) details of who is responsible for monitoring, reviewing, and implementing the plan; and	
	(h) a monitoring and reporting program of the project's impacts on Koalas, including road strike, to the satisfaction of the Secretary; and	0
	(i) adaptive management options for managing impacts on Koalas, including specific impact triggers, developed in consultation with Council.	
Schedule 3, Condition 29	The Long Term Management Strategy must:	Appendix C
	(a) define the objectives and criteria for quarry closure and post- extraction management;	
	(b) investigate options for the future use of the site;	
	(c) describe the measures that would be implemented to minimise or manage the ongoing environmental effects of the project; and	
	(d) describe how the performance of these measures would be monitored over time.	

Condition No.	Requirement	Reference
Schedule 3, Condition 30	Prior to starting quarrying operations on the site, the Proponent must lodge a rehabilitation bond for the project with the Secretary. The sum of the bond must be calculated at: (a) \$2.50/m2 for the total area to be disturbed and/or revegetated in each 5 year review period (see condition 31 below); and (b) \$1.50/m2 for the total area of land previously disturbed and/or rehabilitated by the project, to the satisfaction of the Secretary. <i>Notes:</i> <i>If the rehabilitation and revegetation works are completed to the</i> <i>satisfaction of the Secretary, the Secretary will release the rehabilitation</i> <i>bond.</i> <i>If the rehabilitation and revegetation works are not completed to the</i> <i>satisfaction of the Secretary, the Secretary will call in all or part of the</i> <i>rehabilitation bond, and arrange for the satisfactory completion of the</i> <i>relevant works.</i>	Table 4-1
Schedule 3, Condition 31	<ul> <li>Within 6 months of each Independent Environmental Audit (see condition 6 of schedule 5) excluding the inaugural audit, unless the Secretary directs otherwise, the Proponent must review, and if necessary revise, the sum of the rehabilitation bond to the satisfaction of the Secretary. This review must consider:</li> <li>(c) the effects of inflation;</li> <li>(d) any changes to the total area of disturbance; and</li> <li>(e) the performance of the rehabilitation and revegetation to date.</li> </ul>	Table 4-1
Schedule 3, Condition 38	The Proponent must establish and subsequently maintain the vegetated buffer around the extraction area within 12 months of the date of this approval. Note: The vegetation buffer must be detailed in the Landscape Management Plan.	Appendix A
Schedule 5, Condition 1A	The Proponent must ensure that the management plans required under this approval are prepared in accordance with any relevant guidelines, and include:	
	(a) a summary relevant background or baseline data;	Section 3
	<ul> <li>(b) a description of:</li> <li>the relevant statutory requirements (including any relevant approval, licence or lease conditions);</li> <li>any relevant limits or performance measures/criteria; and the specific performance indicators that are proposed to be used to judge the performance of, or guide the implementation of, the project or any management measures;</li> </ul>	Section 1.3 and Section 2.1
	(c) a description of the measures that to be implemented to comply with the relevant statutory requirements, limits, or performance measures/criteria;	Section 4
	<ul> <li>(d) a program to monitor and report on the:</li> <li>impacts and environmental performance of the project; and effectiveness of any management measures (see (c) above);</li> </ul>	Section 5.2
	(e) a contingency plan to manage any unpredicted impacts and their consequences and to ensure that ongoing impacts reduce to levels below relevant impact assessment criteria as quickly as possible;	Section 5.3
	(f) a program to investigate and implement ways to improve the environmental performance of the project over time	Section 6
	<ul> <li>(g) a protocol for managing and reporting any:</li> <li>incidents;</li> <li>complaints;</li> <li>non-compliances with statutory requirements; and exceedances of the impact assessment criteria and/or performance criteria; and</li> </ul>	Refer to the EMS Incidents will be reported in accordance with conditions 3 and 4 of Schedule 5

Condition No.	Requirement	Reference
	(h) a protocol for periodic review of the plan.	Section 6
Schedule 5, Condition 1B	<ul> <li>Within 3 months of the submission of:</li> <li>(a) an incident report under condition 4 below;</li> <li>(b) an Annual Review under condition 5 below;</li> <li>(c) an audit report under condition 6 below; and</li> <li>(d) any modifications to this approval,</li> <li>the Proponent must review the strategies, plans and programs required under this approval, to the satisfaction of the Secretary. The Proponent must notify the Department in writing of any such review being undertaken. Where this review leads to revisions in any such document, then within 6 weeks of the review the revised document must be submitted for the approval of the Secretary.</li> <li>Note: The purpose of this condition is to ensure that strategies, plans and programs are regularly updated to incorporate any measures recommended to improve environmental performance of the project.</li> </ul>	Refer to the EMS

#### 3.1 Existing environment

A Flora and Fauna Assessment completed in 2006 identified, eleven (11) vegetation communities on the property, as listed below and shown on Figure 3-1:

- Community 1: Tall Closed Forest (Littoral Rainforest)
- Community 2: Very Tall Open Forest/Woodland (Blackbutt)
- Community 3: Tall/Very Tall Closed Forest (Brushbox)
- Community 4: Tall Closed Forest (Coastal Swamp Box with Littoral Rainforest understorey)
- Community 5: Mid-High Woodland (Banksia)
- Community 6: Tall/Very Tall Closed Forest (Paperbark)
- Community 7: Mid-high/Tall Closed/Open Forest (Paperbark/Swamp Oak)
- Community 8: Mid-high/Tall Closed/Open Forest (Swamp Oak)
- Community 9: Low/Tall Open Forest (Mangrove)]
- Community 10: Low/Mid-high Closed Grassland (Pasture)/Open Paddock with scattered trees
- Community 11: Very Tall Closed Grassland (Sugarcane)

The fauna survey of the site (and immediately adjacent areas) resulted in the recording of 71 species of bird, 8 reptiles, 7 amphibians and 17 mammals (or evidence of their presence).

#### 3.2 Impacts

#### 3.2.1 Flora

The flora survey of the site identified 152 species on site. The development of the quarry will result in the removal of vegetation contained within Community 10 (Grassland/pasture with scattered trees) within the approved sand extraction areas. It is considered that the development of these areas will not have a significant environmental impact to these flora species.

#### 3.2.2 Fauna

The proposed development will result in a very minor loss (isolated trees / shrubs within paddocks) of fauna habitat.

The proposed vegetation removal / modification works are not considered to significantly impact upon the endemic fauna assemblage of the site or local / sub-regional populations.

A relatively low diversity of fauna was recorded or predicted to occur within the areas to be disturbed.

The quarry access road runs parallel with the northern Brushbox Forest (Community 3), which provides koala habitat. Potential impacts to koalas include unauthorised clearing of koala habitat and vehicle strike as a result of trucks and light vehicles accessing the Quarry.



#### **Figure 3-1 Vegetation communities**

#### 3.2.3 Threatened Species

Of the species identified on the property, six (6) species in total are identified as endangered under the *Threatened Species Conservation Act* 1995, as listed in Table 3-1 and shown on Figure 3-2 and Figure 3-3. It is considered that the proposal will not result in a significant adverse impact to these species.

#### Table 3-1 Threatened species

Species	Location	Status			
Fauna	Fauna				
Koala	NW corner of the site in association with Community 3 Brushbox Forest on Bedrock. Two individuals recorded via spotlighting.	Vulnerable			
Grey Headed Flying- fox	All areas where melaleuca, Eucalypts and Banksias. Predominately eastern forests adjacent Mooball Creek.	Vulnerable			
Little Bentwing Bat	Eastern forests on sand (predominately Littoral Rainforest, Swampbox Forest on sand, Paperbark Forest [Communities 1, 4, 6] and also NW Brushbox Forest on bedrock [Community 3]	Vulnerable			
Flora					
Cryptocarya foetida	Littoral rainforest, usually on sandy soils, but mature trees are also known on basalt soils.	Vulnerable			
Syzygium moorei	Two planted specimens (~3 m in height) recorded adjacent entry track to Lot 1 on DP208249. Both plants to be retained.	Vulnerable			
Lepiderema pulchella	11-13 individuals (<3 m in height) and 2 mature flowering individuals (<10 m) recorded within Communities 1 and 4. As these communities will be retained and buffered in association with the development no impact is expected.	Vulnerable			



**Figure 3-2 Threatened flora locations** 



**Figure 3-3 Threatened fauna recordings** 

### 4. Environmental control measures

Environmental requirements and control measures are identified in the CoA and the EIS. Specific measures and requirements to address biodiversity and rehabilitation are outlined in Table 4-1.

#### Table 4-1 Environmental controls and mitigation measures

Ref.	Environmental Management Measure	Timing	Responsibility
B1	<ul> <li>All employees and subcontractors will undergo site induction training relating to flora and fauna management issues, including:</li> <li>Pre-clearing requirements</li> <li>Fauna rescue requirements</li> <li>Weed control measures</li> <li>No-go areas</li> <li>Unexpected finds procedure</li> <li>Legal consequences of unauthorised clearing</li> </ul>	Pre-operation and operation	Quarry Manager
B2	Engage a registered surveyor to mark out the boundaries of the approved limits of extraction and submit a copy of the survey plan to the Secretary. ensure that these boundaries are clearly marked at all times in a permanent manner that allows operating staff and inspecting officers to clearly identify those limits.	Pre-operation and operation	Quarry Manager
B3	<ul> <li>Prior to starting quarrying operations on the site, Holcim will lodge a rehabilitation bond for the project with the Secretary. The sum of the bond must be calculated at:</li> <li>(a) \$2.50/m<sup>2</sup> for the total area to be disturbed and/or revegetated in each 5 year review period; and</li> <li>(b) \$1.50/m<sup>2</sup> for the total area of land previously disturbed and/or rehabilitated by the project, to the satisfaction of the Secretary.</li> <li>The rehabilitation bond is to be reviewed within 6 months of each Independent Environmental Audit, excluding the inaugural audit, to the satisfaction of the Secretary – refer to Schedule 3, Condition 31.</li> </ul>	Operation	Quarry Manager
B4	Implement the RRMP (Appendix A)	Pre-operation and operation	Quarry Manager
B5	Implement the Koala Management Plan (Appendix D)	Operation	Quarry Manager
B6	Within 12 months of the commencement of quarrying operations, the proponent will make suitable arrangements to provide appropriate long term security for the revegetation area to ensure it is managed for conservation purposes	Operation	Quarry Manager
B7	<ul> <li>Implement the Long-term Management Plan (Appendix C) so at the completion of extraction, the infrastructure is removed and the site is rehabilitated to the satisfaction of the Secretary to ensure it is:</li> <li>Safe</li> <li>Hydraulically and geotechnically stable</li> <li>Non-polluting</li> <li>Fit for the intended land use and integrated in to the surround landform</li> </ul>	Post operation	Quarry Manager

Ref.	Environmental Management Measure	Timing	Responsibility
B8	Install and monitor (refer Section 5) fauna boxes within the Rehabilitation Areas, including:	Operation	Quarry Manager
	2 x sugar glider boxes		
	2 x cockatoo boxes		
	2 x possum boxes		
	1 x rosella/lorikeet box		
	1 x microbat box		
	• 1 x owl box		
	• 1 x kingfisher box		

## 5. Monitoring and reporting

#### 5.1 Environmental inspections

Routine weekly inspections by the Quarry Manager (or delegate) will occur throughout the operational lifetime of the quarry to identify any ad-hoc issues such as weeds and pest animals using the *Environmental Inspection Checklist* in the Environmental Monitoring and Management Plan.

#### 5.2 Monitoring

In relation to the LMP, routine weekly monitoring will be recorded on the *Environmental Inspection Checklist* in the Environmental Monitoring and Management Plan.

Specific monitoring of Rehabilitation Areas will utilise systematic visual monitoring to determine the success of the measures implemented and establishment of the rehabilitation areas. This data will be used to provide comparisons between monitoring efforts.

Monitoring requirements are adapted from the RRMP (Appendix A) and detailed in Table 5-1. The plot based monitoring forms (i.e., Forms C and D, Appendix E) are to be undertaken at all 13 permanent monitoring locations, as shown on Figure 5-1. The plot-less monitoring forms (i.e., Forms A and B, Appendix E) are to be undertaken within the whole area of each of the three Rehabilitation Areas or the 13 permanent monitoring locations.

Monitoring Requirement	Frequency	Details
Routine Rehabilitation Monitoring	Quarterly	See Form A (Appendix E)
Site Condition	Six Monthly	See Form B (Appendix E)
Revegetation / Forest Structure	Annually (end-calendar-year)	See Form C (Appendix E)
Floristic Composition	Annually (end-calendar-year)	See Form D (Appendix E)
Photographs at established photo points	Quarterly	See Figure 5-1
Fauna Box Monitoring	Six Monthly	See Fauna Box Monitoring Form (Appendix E)
Koala monitoring	On going	See Appendix D.

#### Table 5-1 Rehabilitation area monitoring

Thirteen permanent photo points have been established (Figure 5-1) where photographs will be taken at regular intervals to provide a visual indication of plant growth (height and extent) and weed presence. Photographs shall be taken at the SW, SE, NW, NE corners of each monitoring site.



Figure 5-1 Permanent photo points and monitoring locations

#### 5.3 Contingency plan

Contingency plans are provided in the RRMP (Appendix A).

#### 5.4 Reporting

The general reporting requirements are described in the EMS.

Annual reports summarising the findings of monitoring detailed in Section 5.2 will be prepared in Quarter 1 each calendar year for the previous calendar year. This summary report will be presented in the Annual Report (refer to the EMS).

All records will be:

- Maintained in a legible form.
- Kept for at least 4 years.
- Produced to any authorised officer of the EPA upon request.

## 6. Review and improvement

Continuous improvement of this LMP will be achieved in accordance with the EMS, through the ongoing evaluation of environmental management performance against environmental policies, objectives and targets.

The continuous improvement process is designed to:

- Identify areas of opportunity for improvement of environmental management and performance.
- Determine the cause or causes of non-conformances and deficiencies.
- Develop and implement a plan of corrective and preventative action to address any nonconformances and deficiencies.
- Verify the effectiveness of the corrective and preventative actions.
- Document any changes in procedures resulting from process improvement; and make comparisons with objectives and targets.

## **Appendices**

# **Appendix A** – Rehabilitation and Revegetation Management Plan



## REHABILITATION & REVEGETATION MANAGEMENT PLAN

DUNLOE PARK SAND QUARRY

KELLEHERS ROAD, POTTSVILLE

Prepared for RAMTECH Pty Ltd



Planit Consulting Pty Ltd June 2016



This report has been prepared by:



Planit Consulting Pty Ltd

ABN 20 099 261 711

Level 1, 2247 Gold Coast Highway

Nobby Beach QLD 4218

PO Box 206

Nobby Beach QLD 4218

Telephone: (07) 5526 1500

Facsimile: (07) 5526 1502

Email: admin@planitconsulting.com.au

Web: www.planitconsulting.com.au

#### **Document Control**

Issue	Date	Description	Prepared By	Checked By
А	30/6/2016	Draft Issue	TR	AS
В	01/07/16	Final Issue	TR	AS

#### TABLE OF CONTENTS

SECTION	TITLE	PAGE NO.
1.0	INTRODUCTION	1
2.0	SITE DESCRIPTION	3
2.1	SITE LOCATION AND CURRENT USE	3
2.2	APPROVED PROJECT DESCRIPTION	4
2.3	GEOLOGY, LANDFORM AND TOPOGRAPHY	12
2.4	EXISTING VEGETATION COMMUNITIES	15
2.5	THREATENED FLORA SPECIES	15
2.6	EXISTING FAUNA	16
2.7	HABITATS AND COMMUNITIES OF SIGNIFICANCE	16
3.0	SITE REHABILITATION	22
4.0	REHABILITATION STRATEGY	26
4.1	WEED MANAGEMENT STRATEGY	30
4.2	REVEGETATION STRATEGY	37
4.3	REVEGETATION TYPES	37
4.3.1	ASSISTED NATURAL REGENERATION (ANR)	38
4.3.1.1	DESCRIPTION OF ANR ZONES	40
4.3.2	RECONSTRUCTION/BUFFER PLANTING	64
4.3.3	<b>REVEGETATION TECHNIQUES</b>	69
4.4	MAINTENANCE, MONITORING & REPORTING	69
4.5	ENVIRONMENTAL PROTECTION ZONE PERFORMANCE CRITERIA	79
4.6	CORRECTIVE ACTIONS	81
5.0	FAUNA HABITAT ENHANCEMENT	83
5.1	FAUNA MANAGEMENT DURING CLEARING	92
5.1.1	GENERAL FAUNA MANAGEMENT	92
5.1.2	SUCCESSIONAL CLEARING PROTOCOL	93
5.1.3	ONGOING FAUNA MANAGEMENT DURING QUARRYING	94
5.2	FAUNA NESTBOXES	95
5.3	ANNUAL FAUNA SURVEY	98
6.0	LIST OF ATTACHMENTS	98

FIGURE NO	TITLE	PAGE NO
FIGURE 1	AERIAL PHOTOGRAPH	6
FIGURE 2	APPROVED DEVELOPMENT CONCEPT	8
FIGURE 3	TOPOGRAPHY	12
FIGURE 4	REGIONAL GEOLOGY	14
FIGURE 5	VEGETATION COMMUNITY MAP	17
FIGURE 6	VEGETATION COMMUNITY MAP: NORTHERN	18
FIGURE 7	VEGETATION COMMUNITY MAP: SOUTHERN	19
FIGURE 8	THREATENED FLORA RECORDS	20
FIGURE 9	THREATENED FAUNA RECORDS	21
FIGURE 10	APPROVED REVEGETATION CONCEPT PLAN	24
FIGURE 11	DETAILED REHABILITATION/REVEGETATION PLAN	25
FIGURE 12	EXAMPLE OF NATURAL REGENERATION TRIALS	28
FIGURE 13	WEED MANAGEMENT COMPARISON	31
FIGURE 14	<b>REHABILITATION/REVEGETATION ZONE 1</b>	40
FIGURE 15	BEFORE & AFTER REHABILITATION IMAGES OF REHABILITATION/REVEGETATION ZONE 1	41
FIGURE 16	STAGING OF ZONE 1A	42
FIGURE 17	STAGING OF ZONE 1B	42
FIGURE 18	STAGING OF ZONE 1C	43
FIGURE 19	FRESHWATER MODULE PIONEER PLANTING	45
FIGURE 20	FRESHWATER MODULE DIVERSITY PLANTING	46
FIGURE 21	SWAMP SCLEROPHYLL MODULE PIONEER PLANTING	47
FIGURE 22	SWAMP SCLEROPHYLL MODULE DIVERSITY PLANTING	48
FIGURE 23	SWAMP SCLEROPHYLL GROUNDCOVER DIVERSITY PLANTING	48
FIGURE 24	<b>REHABILITATION/REVEGETATION ZONE 2</b>	49
FIGURE 25	BEFORE & AFTER IMAGES OF REHABILITATION/REVEGETATION ZONE 2	46
FIGURE 26	STAGING OF ZONE 2A	51
FIGURE 27	STAGING OF ZONE 2B	52
FIGURE 28	STAGING OF ZONE 2C	53
FIGURE 29	WETLAND COMMUNITY MODULE	54
FIGURE 30	FRESHWATER MODULE PIONEER PLANTING	54
FIGURE 31	SWAMP SCLEROPHYLL, SHE-OAK & BANKSIA MODULE PIONEER PLANTING	56
FIGURE 32	SWAMP SCLEROPHYLL, SHE-OAK & BANKSIA MODULE DIVERSITY PLANTING	56

FIGURE NO	TITLE	PAGE NO
FIGURE 33	SWAMP SCLEROPHYLL, SHE-OAK & BANKSIA GROUNDCOVER PLANTING	57
FIGURE 34	<b>REHABILITATION/REVEGETATION ZONE 3</b>	57
FIGURE 35	JRE 35 IMAGES OF REHABILITATION/REVEGETATION ZONE 3 - 2009	
FIGURE 36	STAGING OF ZONE 3	58
FIGURE 37	SWAMP SCLEROPHYLL & SHE-OAK PLANTING	59
FIGURE 38	SWAMP SCLEROPHYLL & SHE-OAK PIONEER PLANTING MODULE	60
FIGURE 39	SWAMP SCLEROPHYLL & SHE-OAK DIVERSITY PLANTING MODULE	60
FIGURE 40	SWAMP SCLEROPHYLL & SHE-OAK GROUNDCOVER PLANTING MODULE	61
FIGURE 41	STAGING OF ZONE/AREA 4	61
FIGURE 42	REHABILITATION AREA 4 (WET SCHLEROPHYLL FOREST) MODULE	62
FIGURE 43	AREA 4 PLANTING MODULE	63
FIGURE 44	10M WESTERN VISUAL BUFFER LOCATION	64
FIGURE 45	EXTRACTION LAKE BUFFER ZONE LOCATION	65
FIGURE 46	BUFFER ZONE PLANTING LIST	66
FIGURE 47	BUFFER ZONE PLANTING MODULE	67
FIGURE 48	BUFFER ZONE PLANTING PROFILE	67
FIGURE 49	AREAS PROPOSED FOR AGRICULTURAL USE	68
FIGURE 50	CLEARING ZONES	92



### 1.0 INTRODUCTION

This Amended Rehabilitation and Revegetation Management Plan has been prepared in response to the NSW Department of Planning conditions of Development Consent relating to an approved Sand Quarry Project located at Pottsville (refer Attachment 1). Specifically, this report addresses the requirements of Conditions 26, 27a and 29 of the Section 75J approval issued for the quarry. Also relevant to the preparation of the report are the previous findings and commitments of the previously prepared and submitted Flora and Fauna Assessment (Planit Consulting, 2006) and Rehabilitation Concept Plan (Planit Consulting, 2007) and the previously approved Rehabilitation & Revegetation Management Plan (Planit Consulting, 2009).

The progressive rehabilitation of the wider site throughout the lifespan of the sand quarry operations is a commitment made by Ramtech with the aim of providing for a well established and long term environmentally sound landscape at the completion of operational works of the quarry and visual buffering to adjoining properties throughout the lifespan of the project.

This Rehabilitation and Revegetation Management Plan outlines a program of staged rehabilitation of various areas of the site. These areas to be rehabilitated were previously approved within the development consent 06\_0030. This plan shows a detailed commitment to revegetating previously disturbed areas to create and enhance wildlife corridors, protect riparian areas and ensure the stability of ground and surface water quality within the catchment of Mooball Creek. It is envisaged that the revegetation of the site can, where possible, through future evolving legislation, be implemented within the scope of a carbon offset program. It is noted that revegetation has already commenced in numerous zones in accordance with the approved Rehabilitation & Revegetation Plan (Planit Consulting, 2009). The purpose of this Amended Rehabilitation & Revegetation Plan is to incorporate minor changes regarding areas proposed to be rehabilitated and monitoring procedures, as well as future uses of the site after the project has ceased operation.

This document will outline the procedures for management and monitoring to achieve the objectives beyond the lifespan of the quarry operations.

Objectives of this Plan

- To ensure that rehabilitation works undertaken during the lifespan of the quarry continue to thrive following the completion of operational works of the quarry.
- > To implement management and monitoring procedures throughout the completion of quarry operations to provide for a permanent, healthy, local ecosystem that successfully functions within the natural parameters of the existing, localised, vegetative communities.
- Upon closure of the quarry operations the rehabilitated and revegetated areas are considered well established and capable of thriving without the need for continuing works and management.

Potential future uses of the site following a lengthy lifespan project such as this are can difficult to accurately predict. Many variables could occur that may dictate end uses such as State, Regional and Local Government, strategies, policies and legislation



changes and directions over time. The certainty is that the mitigation and remedial measures implemented during the operations and at the cessation of operations, as presented throughout the Environmental Management Plan (EMP) and this Plan, will ensure that the site will be remediated to the satisfaction of the company, the relevant Government agencies and the community, ensuring environmental quality in the locality.

Much of the existing areas of the site that are presently under environmental protection through zoning within the Tweed Shire Council LEP may remain under various legislated forms of environmental protection. Those areas of rehabilitated land (particularly those linking and expanding wildlife corridors) not within these protected zones, may overtime be included.

Similarly, agriculturally zoned lands within the site that will not be impacted by the sand quarry operations may remain under such zonings into the future.

The proposed use for the land surrounding the lakes is for agricultural purposes, in particularly the plantation of fruiting trees (i.e. avocado trees). More details regarding the proposed agricultural use of these areas are provided later on in this report.



#### 2.0 SITE DESCRIPTION

#### 2.1 LOCATION AND CURRENT USE

The following table identifies the allotments which comprise the total site area of the property. As described within table 1 below, the subject lots are considered to constitute 'the site' within this report.

Real Property Description	Lot Number	Plan Number	
	Lot 1	Deposited Plan 208249	
	Lot 182	Deposited Plan 755721	
	Lot 183	Deposited Plan 755721	
	Lot 44	Deposited Plan 755721	
	Lot 81	Deposited Plan 755721	
	Lot 162	Deposited Plan 755721	
	Lot 2	Deposited Plan 780199	
	Lot 1	Deposited Plan 780199	
	Lot 1	Deposited Plan 780200	
	Lot 2	Deposited Plan 785895	
	Lot 13	Deposited Plan 73451	
	Lot 1	Deposited Plan 785895	
	Lot 1	Deposited Plan 208248	
Address	Pottsville/Mooball Road, Pottsville		
Total Site Area	629.74 hectares		
Total Extraction Area	56.7 hectares		

The property is informally known and referred to as, 'Dunloe Park'.

The site is currently used for agricultural purposes (grazing of livestock) and Stage 1 of the Sand Quarry. All existing site improvements are associated with these agricultural uses, with the exception of existing dwelling houses as depicted in the aerial photograph and facilities/infrastructure associated with Stage 1 of the sand quarry.

In addition, the site currently contains the following:

- Maintained and intact boundary and paddock fencing
- Agricultural drainage lines running predominantly east west
- Small live stock water dams
- Internal access tracks for farm machinery movement.

Large agricultural sheds are also present which are utilised for the purpose of storage and ancillary maintenance of machinery and plant equipment associated with the quarry.

Access to the site is primarily provided from Pottsville Mooball Road and Kelleher's Road. Various unconstructed road reserves and dirt/gravel tracks provide access within the property. A dedicated ingress and egress facility occurs on Pottsville Mooball Road through to the quarry site via the haul road.

The site is large in extent (~630ha) and exhibits evidence of having been previously cleared in association with the historical farming uses (i.e. even aged/height canopies of existing vegetation communities). Evidence of logging on the upper slopes and ridges is also apparent with cut-stumps recorded and existing trees with bifurcated trunks regrown from older, larger stumps also noted. It is considered likely that a



previous selective logging cycle removing Blackbutt, Mahogany, Tallowwood and Turpentine has occurred with Brushbox left as a result of its lower timber production value.

The majority of the site is utilized for agricultural purposes (with the exception of the sand quarry). This historical and ongoing use has resulted in the construction of numerous paddocks (partitioned by barbed wire fencing, cattle grates, constructed drains etc) and rotational agriculture plots which are dominated by pasture grasses and associated herbaceous weed species (i.e. Fireweed, Blue Billygoat, Blady Grass etc). Remnant trees do occur within the paddock zones (Figs, Eucalypts) with regrowth Casuarina also common adjacent the constructed channels which drain the grazing/agriculture areas. One large dam is also present proximate to the farmhouse on the northern lot 2 with several small dams other dams scattered throughout the property.

Whilst the rural uses dominate the site, native vegetation communities occur in the form of regrowth Wet Sclerophyll Forest on the western upper slopes and a combination of Swamp Sclerophyll (Melaleuca, Casuarina) and Coastal Forests on sand (Swampbox, Littoral Rainforest, Banksia Forest) in the eastern areas adjacent Mooball Creek. An examination of the structural diversity of the lower strata of the recorded forests and review of the grazing use indicates that cattle have been largely excluded from the eastern coastal forests for the duration of their growth. The western sclerophyll forest patches, however, exhibit impacts of ongoing cattle grazing within the lower strata which have reduced typical structural and floristic diversity.

Portions of the eastern areas are nominated as being contained within the SEPP 26 (Littoral Rainforest) and SEPP 14 (Coastal Wetlands) designations. Such areas reflect the mosaics of Swampbox//Littoral Rainforest communities on sand and Melalecua/Casuarina Forests/Swamps respectively.

#### 2.2 APPROVED PROJECT DESCRIPTION

A section 75J approval has been issued under the Environmental Planning and Assessment Act 1979 to allow the sand quarry development. The relevant conditions of approval are contained within Attachment 2. A brief discussion of the major issues considered within the Environmental Assessment contributing to the approval of the development in 2008 are provided below:

#### EXTRACTION PROPOSAL

The development involves the staged extraction of sand from two proposed pond areas comprising at total of 56.7 hectares. The northern extraction pond (31.7ha) will form Stage 1 of the proposal; and Stage 2 will involve the extraction of sand from the southern pond (25ha). It is noted that Stage 1 works has commenced operations.

Mineral Resources within the two approved extraction pits total 6.88Mm<sup>3</sup>, including approximately 0.22 Mm<sup>3</sup> of overburden. Computer modelling of a proposed pit design indicated that up to 6 Mm<sup>3</sup> of sand products could be extracted from the proposed pits; processing of the sand would result in <100 000 m<sup>3</sup> of fines requiring re-internment in the pits.



The approved development will ultimately yield approximately 230,000 cubic metres or 300,000 tonnes of sand per annum, with an anticipated lifespan of 26 years.





#### FIGURE 1: AERIAL PHOTOGRAPH



A relatively small works area has been established adjacent to the south western corner of extraction pond stage one (1), which contains the work plant and machinery, stockpiled material and a small building housing workers amenities and an administration area.

Results of soil testing have indicated that the sand resource is of a quality that is in line with Australian Standard 2758.1-1998 (Aggregates and Rock for Engineering Purposes) and is suitable for use in the manufacture of concrete. Calculations indicate that the *in situ* sand resource volume present within the proposed extraction areas, assuming 35% batters is approximately 6,000,000m<sup>3</sup>. The volume of overburden has been calculated at approximately 220,000m<sup>3</sup>. Of the estimated 6,000,000m<sup>3</sup> of extractable material, 90,000 tonnes has been identified as being suitable for brickies loam. However, the latter was estimated from an area of only 4ha subject to investigations in this regard.

The approved development layout and extraction ponds are confined to areas containing mainly pastoral grasses, thereby avoiding disturbance to surrounding areas of wetlands and littoral rainforest. Approximately 17 hectares of the site will undergo revegetation to increase the flora and fauna links between the existing areas of vegetation on and adjacent to the site.

The approved development will see unwanted material and potential acid sulfate soils strategically re-inturned within the extraction ponds to limit chances for oxidation (below the watertable).

#### LAND USE AND CAPABILITY

The site is currently used for agriculture, primarily focused on livestock grazing, and the existing sand quarry (Stage 1). The site contains scattered and fringe areas of native vegetation bordering the site to the east and north and sugar cane fields to the south and west. The site contains a majority of Class 3 & 4 agricultural lands with small areas of class 5 agricultural lands. The class 3 and 4 lands have been identified as generally suitable for grazing and pasture improvement only.

#### HYDROGEOLOGY AND GROUND WATER QUALITY

Existing and proposed ground water levels and quality have been modelled within the evolution of the development proposal. The modelling undertaken has predicted that groundwater quality or levels will not be affected by the proposal.

On going groundwater depth and quality monitoring will continue throughout and beyond the life of the operation.

#### SURFACE WATER

The site is located within the Mooball Creek catchment and Sheens Creek subcatchment areas. Detailed flood modelling confirms that the proposal will have no significant impact upon existing drainage regimes within the catchment.





FIGURE 2: PREVIOUSLY APPROVED DEVELOPMENT CONCEPT (Total Rehabilitation Area 15Ha, exact location and extent determined in OPW Rehabilitation Plans)

## Amended Rehabilitation & Revegetation Management Plan Dunloe Park Sand Quarry Ramtech Pty Ltd



Extraction operations have been designed in conformity with best practice environmental management procedures, including the use of appropriate sediment and water quality devices and the retention of ground cover in areas outside of the extraction ponds.

Sub stage top soil stripping is used to ensure the potential for erosion and sediment problems are limited. On going surface water monitoring will provide a safety net which will be present throughout and beyond the life of the operation.

#### NOISE

To mitigate noise levels, the dredge contains acoustical treatment when operating within the southern extraction pond. In addition, strategically placed earth mounds are constructed, all trucks on and off the site are to be fitted with residential mufflers and on site speed limits are restricted to 25km/h.

#### AIR QUALITY

Airborne particulate matter concentrations and dust deposition from the sand quarry have been predicted to exceed the relevant requirements prescribed by the Department of Environment and Conservation at three of the eight monitoring locations. In particular, exceedances are expected as a result of dust generated from the use of unsealed access roads by haul vehicles.

In order to meet the prescribed requirements, dust controls included sealing of the entire internal roadway length and planting of vegetated buffers. These have already been implemented.

#### FLORA AND FAUNA

A number of threatened species have been identified within the surrounding vegetated areas of the site with none being found, or expected to occur, within the previously disturbed areas of the site (including current/proposed extraction areas).

Rehabilitation measures implemented/proposed will provide increased flora and fauna habitat, additional food resources for identified threatened species, improved opportunities for breeding and dispersal and other benefits associated with returning ~17 hectares of cleared pasture to native vegetation.

No clearing of remnant vegetation is required in respect of the proposal, inclusive of haulage routes and operational areas. Removal of small areas of swamp oak and expansive areas of pasture will be necessary to facilitate the quarry use.



#### ARCHAEOLOGY

A heritage assessment focusing on both Aboriginal and non-Aboriginal heritage has been previously carried out, with no areas of concern identified. Recommendations stemming from the investigations undertaken include that in the unlikely event that any cultural material is exposed during quarrying operations, works that disturb soil or subsoils will cease immediately and government representatives from the Department of Environment and Conservation (now OEH) are to be invited to the site.

#### TRAFFIC AND ACCESS

Entry and exit from the site occurs via a single asphalted route. Entrance to the extraction areas are gained via the dedicated 'haulage track', before turning to the south along an unnamed road. The proposed exit route follows the same path. The haulage track connects with Pottsville Mooball Road for travel to and from the Pacific Highway. Connection to Pottsville Mooball Road is towards the northern extent of the site.

Traffic travelling to and from the site utilises the Pacific Highway, via the Cudgera Creek interchange.

The quarry relies on the use of Council road reserves adjacent to the extraction areas. All roads utilised within the quarry site is sealed.

#### VISUAL

A Visual Impact Assessment has been carried out in relation to the existing visual catchment. The site is located on low lying lands at the coastal end of the Mooball Creek catchment and Sheens Creek sub-catchment areas. Accordingly, on face value that appears to be potential for the site to be visually intrusive.

In particular, properties fronting the northern side of Warwick Park Road and the eastern side of Pottsville Mooball Road have been identified as being within a visual catchment that may be impacted upon by the quarry.

The closest dwelling house to the extraction areas is located at a distance of approximately 900m. Due to the undulating topography and the presence of existing vegetation surrounding Warwick Park Road, very few direct or unimpeded view lines are available to the proposed extraction area. Where direct views are available, it is considered that the ameliorating benefits of significant spatial separation and the inclusion of extensive screen and buffer plantings are sufficient to ensure that no unacceptable visual impacts are created.

#### WASTE

Solid waste generated on site is disposed of by licensed contractor and disposed of or recycled accordingly. Effluent generated by staff employed upon the site is pumped off site for treatment at a Council facility.

Fuel for machinery is currently stored off site. If in the future fuel is to be stored on site, bunding around the fuel storage will occur in accordance with conditions of consent.



#### SOCIO-ECONOMIC ASPECTS

A Socio Economic Impact Assessment has been prepared for the Sand Quarry. The sand quarry will produce quality sand over a 26 year period which is suitable for use in the manufacture of concrete, filling and brickies loam. The size and location of the sand quarry will enable the quarry to service the growth areas of not only the Tweed Shire, but also the Gold Coast Local Government Area to the north and Byron, Ballina and Lismore City Local Government areas to the south. It is anticipated that the population of the region will continue to grow over the life of the quarry.

It has been found that the Sand Quarry would have a positive economic impact upon the local economy extending over a twenty-six-year period. Total revenues have been estimated at \$117 million based on an assumed sales price of \$15 per tonne for processed sand. The proposal will have a measurable contribution to the local economy of \$47 million with an economy multiplier effect (2.7 X) of \$126 million on direct costs over the life of the quarry. The Sand quarry will also provide healthy competition and enable concrete batchers, landscapers and builders to source material from a second quarry within Tweed Shire.

As the quarry is located to the south of the Pottsville township, and as suitable mitigation measures are proposed to manage potential environmental impacts, no adverse impacts are anticipated upon surrounding residential areas.

Haulage of material is to be directed westwards to the Pacific Highway for destinations to the north and south of the quarry, and no traffic will be directed through the Pottsville Village.

In summary, the potential socio economic impacts arising from the proposal are of a positive nature and provide strong justification for the proposed development to proceed.

#### ESD AND CUMULATIVE ASPECTS

The principles of Ecologically Sustainable Development have been considered and underpin the evolution of the sand quarry. Ramtech Pty Ltd strongly support the requirements to limit cumulative impacts, support the precautionary principle, support social and intergenerational equity and the conservation of biological diversity and ecological integrity.

The sand quarry demonstrates adherence to these principles via the retention and regeneration of extensive, previously disturbed areas, the assurance of no impacts upon adjacent sensitive areas, the maintenance of existing groundwater quality and the adoption of management measures relating to air quality, acoustic management and traffic efficiency.

MITIGATION MEASURES AND ENVIRONMENTAL MANAGEMENT



An environmental management plan has been prepared and implemented for the development in accordance with the conditions of Development Consent and reflective of the numerous technical reports undertaken by various specialist consultants.

#### 2.3 <u>GEOLOGY, LANDFORM & TOPOGRAPHY</u>

#### LANDFORM & TOPOGRAPHY

The majority of the development site is generally low lying with elevations ranging from <1.3m AHD towards the south western boundary to only a few metres above sea level in the south eastern corner of the proposed Extraction Area No.1. The north western corner of the site has levels in the vicinity of 3.3m AHD



FIGURE 3: TOPOGRAPHY (SOURCE: GILBERT & SUTHERLAND, 2007)



The extraction areas proper are generally limited in elevation to between 1.3m AHD and 2.0m AHD. The subject site abuts elevated areas to the north (Kelleher's Road) and also to the south (Warwick Park Road). The Sheens Creek Catchment continues at a similar grade to the excavation area back to the Pacific Highway.

#### SITE GEOLOGY

As indicated by the previous soil assessments undertaken, the dominate soil order within the site is Podzols. The site is generally underlain by a uniform soil profile consisting of a small amount of topsoil (a silty sand of less than 0.3m depth), overlaying the resource of fine to medium grained sand of average depth 12m. Marine clay of variable thickness (0.5m to 5m) generally underlies the sand resource located at an average depth of 13m.

Generally, the topsoil materials are greyish brown to brownish black silty sands which grade gradually to very dark grey to yellowish grey, fine to medium sands. Top soil materials occasionally contained trace dark fine silt. Areas of brown/black indurated sands were encountered at depths between approximately 6m to 15m below natural ground level in sporadic location.

#### REGIONAL GEOLOGY

The Tweed 1:250,000 Geological Series Sheet identifies the topographically elevated ridgeline to the west and north of the site as outcrops of greywacke, slate, phyllite and quartzite (part of the regionally extensive Neranleigh Fernvale Group). The low-lying floodplain sections of the site generally consisted of unconsolidated deposits of river gravels, alluvium, sand and clay of Holocene age.

#### SOIL LANDSCAPE

As classified under the Great Soil Group Classifications (GSGC), the soil landscapes in the local area are defined as Kingscliff (variant) and Pottsville soil landscapes. Descriptions of soil landscapes in the local area were conducted by Morand (1996), the following general descriptions were noted.

- Kingscliff soil landscape variant (Kib) These soil landscapes were identified throughout most of Lot 1 in DP755721, in the western half of Lot 1 in DP780199 and throughout most of Lot 2 in DP780199. This soil landscape typically consists of extremely low, level to gentle undulating Pleistocene sand sheet overlaying peat and alluvium. Soils are described as Podzols, are non-cohesive, highly erodible, permeable, have low fertility and are prone to water logging with high water tables.
- Pottsville soil landscape (po) These soil landscapes were identified in the eastern portions of Lot 1 in DP755721 and Lot 2 in DP780199, adjacent to Mooball Creek. This soil landscape typically consists of poorly drained depressions within Pleistocene sand sheets and dunes. Ponded surface water is common, as is shallow (0.1m below NSL) water tables. Soils are described as poorly drained Podzols, Humus Podzols, poorly drained Humic Gleys and Acid Peats. Soils are non-cohesive, highly erodible and permeable, have low fertility and are prone to water logging and high water tables.




FIGURE 4: REGIONAL GEOLOGY (SOURCE GILBERT & SOUTHERLAND, 2008)



The underlying geological and upper soil layer units (above) are utilised in association with the existing vegetation of the site to determine regional ecosystem types (vegetation communities in a bioregion that are consistently associated with a particular combination of geology, landform and soil [Sattler & Williams, 1999]). Analysis of the regional ecosystem types occurring within bushland on similar geology proximate to the rehabilitation areas can provide insight on what pre-clearing vegetation communities previously occurred and are likely to be successful if utilised within a revegetation project.

## 2.4 EXISTING VEGETATION COMMUNITIES

A detailed Flora and Fauna Assessment was undertaken over the Dunloe Park lands by Planit Consulting in 2006. This assessment described and mapped (refer Figures 5-7) eleven Vegetation Communities:

Community 1:	Tall Closed Forest (Littoral Rainforest) [T7D]
Community 2:	Very Tall Open Forest/Woodland (Blackbutt) [T8M/S]
Community 3:	Tall/Very Tall Closed Forest (Brushbox) [T7D]
Community 4:	Tall Closed Forest (Coastal Swamp Box with Littoral Rainforest understorey) [T7D]
Community 5:	Mid-High Woodland (Banksia) [T6S]
Community 6:	Tall/Very Tall Closed Forest (Paperbark) [T8D]
Community 7:	Mid-high/Tall Closed/Open Forest (Paperbark/Swamp Oak) [T6D]
Community 8:	Mid-high/Tall Closed/Open Forest (Swamp Oak) [T6D]
Community 9:	Low/Tall Open Forest (Mangrove) [T5M]
Community 10:	Low/Mid-high Closed Grassland (Pasture)/Open Paddock with scattered trees [G2D]
Community 11:	Very Tall Closed Grassland (Sugarcane) [G4D]

Of these communities, numbers 1, 6, 7, 8 and 9 are reflective of Endangered Ecological Communities all of which will be retained in association with the approved quarry development.

## 2.5 THREATENED FLORA SPECIES

Three threatened flora species were recorded onsite during the previous survey works (refer Figure 8):

#### Stinking Cryptocarya

8-10 individuals (<5m in height) were recorded within Communities 1 and 6. As these communities will be retained and buffered in association with the development no impact is expected.

#### Durrobby

Two planted specimens (~3m in height) were recorded adjacent entry track to Lot 1 on DP208249. Both plants to be retained in association with the sand quarry.



## Fine-leaved Tuckeroo

11-13 individuals (<3m in height) and 2 mature flowering individuals (<10m) were recorded within Communities 1 and 4. As these communities will be retained and buffered in association with the development no impact is expected.

## 2.6 EXISTING FAUNA

The fauna survey of the site (and immediately adjacent areas) resulted in the recording of 71 species of bird, 8 reptiles, 7 amphibians and 17 mammals (or evidence of their presence).

Of these species, three (Koala, Grey Headed Flying-fox, Little Bentwing Bat) are listed as vulnerable within the *Threatened Species Conservation Act 1995.* 

A Section 5A of the *Environmental Planning and Assessment Act 1979* (the '7-Part Test of Significance') was conducted for the three recorded species plus an additional three species which are considered possible occurrences on site <u>and</u> may have the potential to be impacted as a result of the proposal. Section 5A was also conducted for the recorded Endangered Ecological Communities. The assessment concluded that the impacts of the development are unlikely to threaten the viability of any local populations of the nominated species/communities. A species impact was therefore not required and the development was subsequently approved subject to conditions.

## 2.7 HABITATS AND COMMUNITIES OF SIGNIFICANCE

The areas of the site to be protected from development impacts contain several vegetation communities/ecosystems of significance which mostly occur in the eastern portions of the site adjacent Mooball Creek and the Wooyung Nature Reserve. These communities (interconnected eastern Communities 1 & 4-9) are considered to be of ecological significance as a result of one or more of the following:

- Being a wetland environment (freshwater or marine)
- Representing riparian communities fringing Mooball Creek
- Being representative of a rare, vulnerable or endangered forest ecosystem (Upper Northeast Bioregion) or being a regionally significant vegetation community (Tweed Shire)
- Being representative of an Endangered Ecological Community
- Containing vulnerable or endangered plant species

As such, the development does not encroach into these areas nominated as being significant which are largely contained within the existing 7a zone and/or SEPP14/26 designations. Recommendations for management of weeds and revegetation to promote the continued viability of these ecosystems are incorporated within this report.

























# 3.0 SITE REHABILITATION

Rehabilitation has been conditioned as part of the development to plant a partially vegetated buffer around the two extraction ponds and to also rehabilitate part of the site with native vegetation.

This landscaping has been carried out for the following reasons:

- To provide an effective, aesthetically appealing and practical vegetation buffer to the perimeter of the development; and
- To Rehabilitate selected areas within the overall Dunloe Park Land holding to consolidate and enhance the existing flora and fauna corridors to the north and east of the extraction areas.

A conceptual Revegetation/Rehabilitation Plan has been prepared by Planit Consulting which was approved as part of the Sand Quarry approval (refer Figure 10) and refined and implemented within the approved Rehabilitation & Revegetation Management Plan (Planit Consulting, 2009) which identified 15 hectares of land to be rehabilitated in accordance with Condition 27 of the Consent issued. In addition, a further 2ha will be provided along watercourses and a road within the site to provide wildlife linkage corridors between habitats of the site and locality.

## REHABILITATION PROCESS

A long-term staged rehabilitation/revegetation process has also been derived and implemented for the 15ha required by the Development Consent (additional 2ha now provided) and associated approved concept plans. This strategy has chronologically focused upon assisting natural regeneration, plantings of pioneer species (to establish cover and suppress pasture weeds) and increasing floristic diversity. Additional actions to exclude cattle and increase foraging potential for threatened fauna has also be performed.

Planting initially begun on the western side of the development site and concentrated around the stockpiling locations and haulage routes and has now included Rehabilitation Zones 1-3 (as described within this report). Planting surrounding the extraction areas (proposed for agricultural use) will take place once extraction has ceased. This is due to the continuous need for accessibility (i.e. removing the dredge from the lake for servicing) which would be difficult with vegetated buffers surrounding the extraction lake during operation.

## REHABILITATION SPECIES

A wide range of native species endemic to the area has been selected for the rehabilitation and buffer areas. In this regard Rehabilitation Zone No.1 has been revegetated (combination of 'assisted natural regeneration' and revegetation/landscaping) with endangered ecological community (EEC) Swamp Sclerophyll plus Eucalypt Open Forest species and EEC Coastal Wetland within the localised soaks.





Rehabilitation Zone No. 2 has been revegetated with a combination of EEC Swamp Sclerophyll, EEC Swamp Oak and Banksia open forest species with EEC Saltmarsh within a minor tidal extension zone of the existing paddock.



Rehabilitation Zone No. 3 has been revegetated with a combination of EEC Swamp Sclerophyll and EEC Swamp Oak open forest.

Rehabilitation Zone No.4 is proposed to be revegetated reflective of a Wet Sclerophyll Forest.

The shrub and small tree layer of all rehabilitated forest communities has also incorporated (via regeneration or revegetation) a reasonable diversity of EEC Littoral Rainforest species reflective of existing remnant communities and to expand foraging habitat for threatened flying foxes and rainforest pigeons.

The rehabilitation program over the  $\sim$ 17 hectares will be staged over a 10-15 year timeframe (commenced  $\sim$  6 years ago for Zones 1-3) and will ensure that regular monitoring and maintenance will continue for the life of the extraction project.

An amended Rehabilitation and Revegetation Management Plan has been prepared within this document which expands upon the previously approved conceptual plans / Rehabilitation & Revegetation Management Plan and satisfies the relevant conditions of the approval issued relating to site rehabilitation. The main purpose of the amended



Rehabilitation and Revegetation Management Plan is to incorporate additional revegetation areas proposed (Zone No. 4) and the proposed future agricultural use surrounding the two lakes (i.e. avocado tree plantations).



#### FIGURE 10: APPROVED REVEGETATION CONCEPT PLAN (Total Rehabilitation Area 15Ha, exact location and extent determined in OPW Rehabilitation Plans)





FIGURE 11: DETAILED REVEGETATION/REHABILITATION PLAN (ALSO REFER ATTACHMENT 3)

June 2016



# 4.0 REHABILITATION STRATEGY

This Section outlines the Rehabilitation/Revegetation Strategy for the Environmental Protection Zones (EPZ) which has/will incorporate the following:

- <u>EXISTING VEGETATION COMMUNITIES</u>: Existing vegetation communities located external to the works area (refer Figure 5). The predominate management strategy for these zones will be the continued exclusion of cattle, monitoring and control of environmental weed infestations (to promote ongoing natural regeneration) and maintenance of existing fire trails
- <u>REHABILITATION ZONES</u>: The 17ha rehabilitation zone (Zones 1-4, refer Figure 11). A detailed staged rehabilitation/revegetation strategy for these zones is contained within the following sections of this report and focuses upon removing threatening impacts (i.e. grazing and weed invasion), managing natural regeneration and revegetation works to deliver the proposed vegetation communities. Targeted fauna habitat enhancement works have also been incorporated.

In accordance with best management practice, restoration and rehabilitation works sought to stabilize and reverse the negative effects of habitat fragmentation. Priority has been given to works which protect and expand larger remnants so that they are reintegrated into larger revegetated areas. This is based on the following ecological observations:

- Habitat fragmentation generally reduces the viability of both faunal and floral populations by restricting ranges of fauna below minimum levels and by preventing the natural exchange of genetic material which may ensure genetic vigour.
- Reconnecting fragmented landscapes into larger, (and, where appropriate, more consolidated) units by filling in gaps can increase habitat area and may improve linkages for passage of species. (The configuration of those linkages, however, will determine whether edge/area ratios reduce or increase.)
- Narrow corridor links may themselves be subject to feral predation, disease and species imbalances, therefore, the width of a corridor should exceed the extent of edge effects.
- Biological potential for diverse natural regeneration and expansion frequently exists in and adjacent to remnants. Investments which harness this potential usually provide higher ecological returns than reconstruction elsewhere (Greening Australia, 1999).

As such, the 17 hectares of site rehabilitation/restoration have been sited to close remnant canopy caps, re-connect existing fragmented remnants, increase the area:edge ratio of endangered ecological community remnants and provide/increase fauna habitat corridor widths.

<u>BUFFER ZONES</u>: Visual buffer and extraction lake revegetation works (refer Figure 11). The strategy for these areas is the retention of the adjacent casuarinas buffers and landscaping of the visual buffer to the machinery compound areas. Such



revegetation was established within 12 months in accordance with condition 39 of the consent. On-going weed management and monitoring are routinely performed to ensure the long-term viability of the buffers. Revegetation of riparian plantings to the edge of the lakes would not commence until the cessation of extraction activities cease at each pond. This is necessary as ground anchors for the dredge are required to be set at the edge of the pond during extraction activities and moved around the pond perimeter from time to time as the dredging process dictates.

The below rehabilitation strategy has been prepared, implemented, and aimed to protect and enhance habitats retained within the EPZ.

It often takes a period of several years before the achievement of such aims are realised and as such it is of paramount importance that an appropriate strategy is derived and implemented in the initial phases of rehabilitation.

The primary objectives for the EPZ are quite simple and include:

- Retain significant existing communities (refer Figure 5)
- Retain and enhance existing fauna habitat
- Remove and manage processes potentially threatening the viability of existing remnants (refer Figure 5)
- Increase the extent of vegetation communities and potential fauna habitat over time (refer Figure 11)

EPZ	OVER-RIDING TECHNIQUE
Existing vegetation	Weed removal and ongoing monitoring only (no
communities	revegetation)
Rehabilitation Zones	Assisted Natural Regeneration
Buffers	Reconstruction

Three management techniques will be implemented for the EPZs as follows:

Weed removal and management is discussed within Section 4.1 below and will be undertaken where necessary within the existing vegetation communities in association with routine management.

The 17ha rehabilitation zone has been managed via Assisted Natural Regeneration (GCCC (2007)/Greening Australia (1999)) due to its location within a fragmented landscape and proximity to established native vegetation communities. Greening Australia (1999) notes that 'in or immediately adjacent to remnants, priority is given to facilitation of natural regeneration. In these locations, planting or direct seeding is only carried out where pre-existing species are incapable of colonization – and after a "rest" period sufficient to test natural regeneration. This is based on the following ecological and field observations.

• Natural regeneration potential can be surprisingly persistent in and adjacent to fragmented remnants.



- Natural regeneration maintains natural selection processes, can provide a wider range of site adapted species and genetic stock, demonstrates any capacity for future regeneration, and informs us about a site's regeneration dynamics and any pre-existing species requiring reintroduction.
- Planting can be more expensive, interfere with regeneration and compromise the genetic integrity and scientific value of a site. Planted stock may not regenerate if the species selection or genetic stock is inappropriate to the site.
- The mechanisms of recovery of individual species after natural disturbances (particularly whether they form persistent soil seed banks or not) can provide insight into the restoration approach needed. This determination can be improved by conducting preliminary trials to trigger germination from soil seed banks (e.g. using fire, smoke, tillage or irrigation as appropriate) - Such trials can also help to determine pre-existing plant associations more precisely.'

Cattle were excluded from portions of the 15ha rehabilitation zone (zones 1-3) in late 2008 to allow test trials of natural regeneration to be monitored with germination of pioneer canopy trees (melaleuaca, callistemon, casuarina, lophostemon, banksias etc) noted.



FIGURE 12: EXAMPLE OF NATURAL REGENERATION DURING TRIALS IN 2008

A staged strategy (refer Section 4.0/4.2) which involves a combination of monitoring (and cattle exclusion & weed management) natural regeneration for a minimum period of 12 months, planting pioneer species (if needed) and supplementary planting (groundcovers and diversity plantings-if needed) has therefore been established (and implemented). This approach will ensure that establishment of endemic genetic flora is maximized over the life of the rehabilitation project.

In accordance with GCCC (2007) Assisted Natural Regeneration applies:

- To natural areas where the native plant community is largely healthy and functioning.
- When native plant seed is still stored in the soil or will be able to reach the site from nearby natural areas, by birds or other animals, wind or water.
- Where the natural regeneration processes (seedling germination, rootsuckering, *etc.*) are being inhibited by external factors, such as weed invasion, soil compaction, cattle grazing, mechanical slashing, etc
- When limited human intervention, such as weed removal, minor amelioration of soil conditions, erection of fencing, cessation of slashing, etc. will be enough to trigger the recovery processes through natural regeneration.
- When the major component is weed control.

As discussed above assisted natural regeneration will apply to the 17ha regeneration/restoration zones.

However, the buffer areas associated with the extraction lake and machinery/compound areas (refer Figure 11) are disjunct from native vegetation communities and are located in historical areas of intensive grazing and agriculture. Natural regeneration potential from these areas is considered minimal with growth of pasture species prevalent. As such, Reconstruction/landscaping were required in these areas. Such intensive planting of native species was also considered necessary to comply with Condition 39 of the consent which required the buffer areas to be established within 12 months. In accordance with GCCC (2007) <u>Reconstruction</u> applies:

- Where the site is highly degraded or altered
- When the degree of disturbance has been so great and long-standing that the preexisting native plant community cannot recover by natural means.
- To sites such as areas of fill, sites affected by stormwater flow, and areas that have been drastically cleared, either mechanically or by stock even though there may be a few remaining native trees or shrubs.
- When a greater degree of human intervention is required, such as weed removal, cessation of grazing and/or slashing, amelioration of soil conditions such as importation of soils, drainage works or reshaping of the landscape.



• When a major component is the importation of native species through planting.

## 4.1 WEED MANAGEMENT STRATEGY

The following weed management strategy has been prepared for the EPZ. The intent of the strategy is to progressively remove weed species from these areas. Substantial research regarding weed management activities within areas of native bushland has been undertaken in this regard. The strategy presented has been adapted from the following sources:

- Bradley, J. (1988) *Bringing back the bush: The Bradley method of bush regeneration.* Lansdowne Publishing Pty. Ltd. The Rocks, NSW;
- Buchanan, R. A. (1989) *Bush Regeneration: Recovering Australian Landscapes.* TAFE Student Learning Publications, NSW;
- Robertson, M. (1994) *Stop Bushland Weeds: A guide to successful weeding in South Australia's bushland*. The Nature Conservation Society of South Australia Inc;
- Greening Australia (1995) *Local Greening Plans: A Guide for Vegetation and Biodiversity Management*. Greening Australia, Canberra;
- Department of Planning (1991) Urban Bushland. Department of Planning, Sydney.
- Department of Environment and Heritage (1999) Suggested Conservation Criteria for Development Assessment for use by Local Government Officers. DEH, Brisbane.
- 1. <u>Target areas of least disturbance and weed invasion and work towards the more</u> weed infested areas.

Under these circumstances the indigenous species have the upper hand because their seed or spores are already in the ground and the natural environment favours the plants that have evolved in it (Bradley 1988). This method represents a significant reduction in follow-up time and related costs because the regeneration is more likely to comprise a higher proportion of native species. It is noted that this method is the EPAs preferred method of weed control (DEH, 1999).

#### 2. <u>Minimising the amount of soil disturbance will reduce the potential for a fresh weed</u> invasion.

Weeding will cause some disturbance to the existing soil structure and layers depending on weed methods utilised (i.e. mechanical removal, hand removal). Disturbed soil should therefore always be returned as close as possible to its original layer and firmed down. This includes the mulch which is the first line of defence against a fresh invasion (Bradley 1988).

3. Let native plant regeneration dictate the rate of weed removal



The regeneration of native species is inversely proportional to increasing weed growth. Therefore, weeding should not automatically move into more degraded areas when less infested areas have been initially weeded. It is often better to simply wait for the less infested areas to regenerate before proceeding slowly into the worse areas.

Clearing all infested areas at once causes extensive, additional follow up weeding works (refer Figure 13 below).



FIGURE 13 : Source - Ku-ring-gai Parks Dept. NSW, cited in Buchanan (1989) Comparison of labour hours using traditional weeding and natural regeneration methods.

Implementing these principles into a weed management strategy begun by determining priority species and areas for management. In this regard, numerous inspections of the EPZs note the following with regard to weed presence:



EPZ	OBSERVED WEED PRESENCE
Existing Vegetation	The existing native vegetation communities (refer Figure 5) contain minor elements of weed invasion, mostly on the perimeters or edges. Those recorded are described below:
Communities	<u>Community 3: Tall-Very Tall Closed Forest (Brushbox) on Bedrock</u> : The highly fragmented nature of this community, ongoing grazing and proximity to pasture areas resulted in the recording of numerous weed species on the remnant fringe. Such species included pasture/exotic grasses (Couch, Kikuyu, Guinea Grass, Pigeon Grass, Paspalum), Fireweed, Mistweed, Billygoat Weed, Balloon Cotton, Mile-a- Minute, Camphor Laurel, Bitou Bush, Flatweed, Cobblers Pegs, Thistle, Sirato, Inkweed, Wild Tobacco and White Passionflower.
	<u>Community 4: Tall Closed Forest (Coastal Swamp Box with Littoral Rainforest</u> <u>understorey):</u> Weed species were common in the western areas of the association adjacent to the open paddock/grassland grazing areas and included Lantana, Mile-a- minute, Groundsel and pasture grasses (Whisky Grass, Paspalum, Rhodes Grass etc).
EPZ	OBSERVED WEED PRESENCE
Existing Vegetation Communities	<ul> <li><u>Community 5: Mid-high Woodland (Banksia)</u>: Weed species within this small forest patch included Lantana and pasture grasses (Whisky Grass, Paspalum, Rhodes Grass etc).</li> <li><u>Community 6: Tall/Very Tall Closed Forest (Paperbark)</u>: The most common weed species noted were Camphor Laurel which was a quite common re-growth species within the sub-canopy layer and Bitou Bush and Mickey Mouse Plant in the shrub layer.</li> <li><u>Community 7: Mid-high / Tall Closed-Open Forest (Paperbark / Swamp Oak)</u>: Weeds were present mostly within more recently disturbed areas adjacent to Mooball Creek or on remnant fringes and included Lantana, Groundsel, Camphor Laurel, pasture grasses, Umbrella Tree and Bitou Bush. The small western remnant also included common weeds in the ground-layer (pasture grasses [Couch, Kikuyu, Pigeon Grass, Guinea Grass] Fireweed, Crofton Weed, Mistweed, Billygoat Weed, Balloon Cotton, Mile-a-Minute, Camphor Laurel, Bitou Bush)</li> <li><u>Community 8: Mid-high/Tall Closed-Open Forest (Swamp Oak)</u>: Weeds occupied a number of small patches within this community and mostly included Lantana.</li> </ul>
	Groundsel, Bitou Bush and pasture grasses.
Rehabilitation Zones	All Rehabilitation Zones (with the exception of Zone 4) have been excluded of cattle for a number of years and regeneration is currently in progress. Prior to rehabilitation works, pasture grasses and herbaceous pasture weeds (i.e. siratro, cuphea, fireweed, purple top, bitou bush, flatweed) were observed.
Buffers	These areas contain pasture grasses and occasional herbaceous pasture weeds (i.e. siratro, cuphea, fireweed, purple top, bitou bush, flatweed). These areas have been 'reconstructed' concurrent with adjacent extraction works with continuous weed control occurring



Scientific Name*	Common Name	Recommended Treatment
Panicum maximum Paspalum dilatatum P. conjugatum Setaria sphacelata Andropogon virginicus Chloris gayana Cynodon dactylon Pennisetum clandestinum	Exotic/ pasture grasses	Spot spraying of clumps and hand removal of scattered individuals within rehabilitation zones. Routine spot spraying is to ensure that all foliage is wetted with Roundup Biactive [at a ratio of 1:100 [1 part glyphosate to 100 parts water]].
Lantana camara	Lantana	Spot spraying of Lantana with Roundup Biactive [at a ratio of 1:100] is recommended. Following treatment, Lantana is to be removed via hand and stored until complete browning out occurs. Note that removed biomass must be stored/hung off the ground to prevent re-shooting. Spot spraying is to ensure all foliage is wetted. Addition of surfactant PULSE (200 mL/100 L) to Roundup is recommended to improve Lantana control.
Ageratum houstonianum	Blue Billygoat Weed	Spot spraying with Roundup Biactive [at a ratio of 1:100]. Spot spraying is to ensure all foliage is wetted.
Bidens pilosa	Cobblers Pegs	Spot spraying with Roundup Biactive [at a ratio of 1:100]. Spot spraying is to ensure all foliage is wetted.
Phytolacca octandra	Inkweed	Spot spraying with Roundup Biactive [at a ratio of 1:100]. Spot spraying is to ensure all foliage is wetted.
		Following herbicide treatment, any individuals regenerating from taproots are to be removed by hand. Care to be taken to avoid skin contact with sap.
Baccharis halimifolia	Groundsel	Spot spraying with Roundup Biactive [at a ratio of 1:100] or hand removal is recommended. Spot spraying is to ensure all foliage is wetted.
		NOTE: If groundsel is seeding at any time, individual plants are to be removed via hand and bagged to prevent seed spread via wind.



	Camphor	Saplings of camphor laurel are to be treated via stem injection. 2mL of roundup bioactive is to be injected per cut. Juveniles
Cinnamonum camphora	Laurel	sprayed with Roundup Biactive (at a ratio of 1:100) or hand removal. Spot spraying is to ensure all foliage is wetted.
Ageratina riparia	Mistweed	Spot spraying with Roundup Biactive (at a ratio of 1:100) or hand removal. Spot spraying is to ensure all foliage is wetted.
Ageratina adenophora	Crofton Weed	Spot spraying with Roundup Biactive (at a ratio of 1:100) or hand removal. Spot spraying is to ensure all foliage is wetted.
Conyza spp.	Fleabane	Spot spraying Roundup Biactive [at a ratio of 1:100] (or hand removal for minor infestations). Spot spraying is to ensure all foliage is wetted.
Chrysanthemoides monilifera	Bitou bush	Mature stems to be treated via Cut-stump method. This will involve cutting the trunk at ground level and immediately swabbing the stump surface with Roundup Biactive. Juveniles sprayed with Roundup Biactive (at a ratio of 1:100) or hand removal. Spot spraying is to ensure all foliage is wetted.
Hypochoeris radicata	Flatweed	Spot spraying with Roundup Biactive (at a ratio of 1:100) or hand removal. Spot spraying is to ensure all foliage is wetted.
Senecio madagascariensis	Fireweed	Spot spraying with Roundup Biactive (at a ratio of 1:100) or hand removal. Spot spraying is to ensure all foliage is wetted.
Passiflora subpeltata	Passionflo wer	Hand removal of juvenile stems is recommended with cut-stump for larger specimens.
		Cut stump treatment will involve cutting the stem at ground level and immediately swabbing the stump surface with Roundup Biactive (1 part roundup: 2 parts water).
Solanum mauritianum	Wild Tobacco	Cut-stump treatment is to occur for mature stems. This will involve cutting the trunk at ground level and immediately swabbing the stump surface with Roundup Biactive.
		Spot spraying of juveniles with Roundup Biactive [at a ratio of 1:100]. Spot spraying is to ensure all foliage is wetted.
Verbena bonariensis	Purple Top	Given low leaf surface area of this species, if isolated plants occur (i.e. small numbers, not within an area dominated by other weeds) they are to be removed via hand.
Gomphocarpus physocarpus	Cotton Bush	Spot spraying with Roundup Biactive (at a ratio of 1:100) or hand removal. Spot spraying is to ensure all foliage is wetted.
	1	

\* where herbicide treatment is required the use of Roundup Biactive has been predominately recommended. The active ingredient of this herbicide is Glyphosate isopropylamine which has been found to be non-toxic to frog tadpoles and generally does not require the use of additional surfactants. The use of surfactant is only recommended for *Lantana camara*.



#### RECOMMENDED APPLICATION TECHNIQUE FOR SPRAYING (Knapsack/Handgun Equipment)

The dilution rate is given as a ration (i.e. 1: 100 is 1 part Roundup Biactive: 100 parts water). Adjust equipment to achieve an even spray pattern. Apply to ensure complete and uniform wetting of all foliage. For handgun equipment, a D6 spray tip (Spraying Systems Australia P/L) or equivalent, and an operating pressure of 400-700 kPa is recommended.

#### Additional Recommendations: Surfactant

### GENERAL GUIDELINES

Do not vary from the 200ml of Pulse per 100L of spray solution because as has been shown, this is the optimum rate for Pulse. Do not reduce the rates of Roundup as all trial work has shown that the recommended label rates of Roundup are needed to achieve control. AVOID EXCESSIVE AGITATION BOTH WHEN MIXING AND WHEN SPRAYING, AS FOAMING CAN OCCUR IF SOLUTION IS OVER AGITATED. Wear gloves and a face shield or goggles when handling Pulse undiluted as it is severely irritating to the eyes.

Mixing:

- 1. Half fill tank with water.
- 2. Add the correct amount of Roundup and mix.
- 3. Fill tank until almost full.
- 4. Add Pulse at the rate of 200ml per 100L of spray solution and mix.
- 5. Complete filling tank.
- 6. Mix.

## **RESTRAINTS ON USE**

Pulse should not be added to Roundup as a general-purpose surfactant as some antagonism can occur between Roundup and Pulse on typically easy-tokill grasses such as wild oats and Brome grass. Currently there are no other herbicides recommended for use with Pulse on the Pulse label. Users should check with the manufacturer before using any particular herbicide or other pesticide with Pulse. Pulse I not a general-purpose surfactant but rather a specific spray additive for Roundup herbicide for the improved control of brush and woody weeds.



#### RECOMMENDED APPLICATION TECHNIQUE FOR CUT-STUMP TREATMENT

Cut stump treatment will involve cutting the stem of the plant at ground level and immediately swabbing the stump surface with Roundup Biactive (1 part roundup: 2 parts water).

#### RECOMMENDED APPLICATION TECHNIQUE FOR STEM INJECTION

This shall involve use of an applicator calibrated to deliver 1 or 2 mL of Roundup Biactive per cut. 5cm cuts at 10-15cm centres around the trees circumference are to be made at an oblique angle to ensure penetration of the sapwood beneath the outer bark.

#### Referenced Material:

Mann, M. (2000) Toxicological Impact of Agricultural Surfactants on Australian Frogs (PHD Thesis). Curtin University of Technology, Perth

Nufarm Australia Limited (undated) Roundup Biactive Herbicide by Monsanto NRA Approval No. 48518/1102

Pulse Penetrant online @ http://www.nrrbs.com.au/chemicalspulse.htm



In association with the progressive removal of the nominated weed species from these areas, a revegetation/regeneration strategy should be selected to ensure that the newly weeded areas become established with native species. The regeneration strategies utilised on site are discussed below.

## 4.2 REVEGETATION STRATEGY

Revegetation of disturbed areas have been (or proposed to be) undertaken on site for the following reasons:

- To maintain the existing level of integrity of vegetation communities contained within the EPZ
- To maintain and potentially increase the floristic diversity currently exhibited within the EPZ
- To ensure that degraded and managed areas regenerate and are revegetated with native endemic flora species
- To stabilize areas subjected to weed management and cessation of grazing pursuits
- To restore cover, habitat diversity and dispersal options for the faunal assemblage
- To increase crown cover in open areas such that with the ongoing regrowth succession of the site there is potential for the existing communities to reach remnant status in the future
- To reduce the visual impact of the extraction use via the use of planted buffers.

In addition to the above, revegetation of the ~17 hectares of cleared/pasture areas is necessary to long-term protection to the retained vegetation communities (including riparian and endangered ecological communities) from existing edge effects and ongoing fragmentation including:

- Abiotic effects: those changes in light, temperature, humidity and wind that occur when a remnant edge is formed by the creation of new surrounding land uses, such as clearing land for grazing, agriculture or urban development.
- Direct biological effects: include changes in the number and abundance of species brought about by changed environmental conditions (e.g. the spread of species that adapt well to the altered climatic conditions, and the reduction in recruitment of species that do not prosper).
- Indirect biological effects: changes in the way species interact, particularly modified patterns of competition, pollination, and the dispersal of seeds (Greening Australia, 2000).

#### 4.3 REVEGETATION TYPES

As suggested for the 17ha Rehabilitation Zone and the extraction Buffer Zones, the strategy of Assisted Natural Regeneration (with supplementary plantings where necessary) and Reconstruction (large-scale planting/revegetation) are the methods



that would prove the most feasible and effective. These types of strategies are outlined below in addition to the discussions previously offered in Section 4.0 above.

# 4.3.1 ASSISTED NATURAL REGENERATION (WITH SUPPLEMENTARY PLANTINGS)

Assisted Natural Regeneration (with supplementary plantings where necessary) has been implemented for the Zones 1-3 (and to be implemented within Zone 4). The entire 15ha was subdivided into three smaller zones based upon location (refer Figure 11). These three smaller zones were subdivided further based upon existing levels of natural regeneration (following the initial trials) and adjacent community types and have commenced regeneration procedures. Due to the large area of rehabilitation, each zone has also been broken into manageable units of 0.5-1 hectare. Similar approach is now proposed for Zone 4 which has been divided in three stages. These stages will provide linkages between existing habitats within and immediately external to the site, as well as providing a visual buffer.

The staging process shall/has occurred as follows and has been selected to maximize potential for natural regeneration of endemic flora:

#### NATURAL REGENERATION AND WEED MANAGEMENT

This management stage involved the following:

- Ensuring cattle are removed from the relevant rehabilitation zones/stages. Fencing shall only be necessary in the instance that cattle are grazing adjacent paddocks. If the adjacent paddocks are fallow then separate fencing is unnecessary.
- Monitoring and eradication of weeds as necessary
- Maintenance of fire trails

This process shall occur for a minimum of one year to five years (for later stages).

ASSESSMENT OF REGENERATION EFFORTS AND SUPPLEMENTARY PLANTING (PIONEER SPECIES)

This management stage has involved the following (and to be implemented within upcoming efforts):

- Assessing the success of the preceding natural regeneration to establish cover across the rehabilitation zone/stage.
- Where gaps in native cover occur, planting of pioneer trees and shrubs will be necessary in accordance with selected modules (refer Section 4.3.1.1). Pioneer species have been selected upon previous regeneration trials, adjacent communities intended to be replicated and ability for pioneer species to establish quickly and shade-out weed species. Establishment of quick growing species will bring forward flowering and attraction of fauna to the rehabilitation zone. Birds and bats are likely to then deposit flora seeds from surrounding areas
- Cattle, weed and fire trail monitoring and maintenance will be ongoing

This process shall occur over a period of one year.



ASSESSMENT OF REGENERATION EFFORTS AND SUPPLEMENTARY PLANTING (DIVERSITY)

This management stage has involved the following (and to be implemented within upcoming efforts):

- Assessing the success of the preceding natural regeneration and pioneer plantings to establish cover across the rehabilitation zone/stage.
- Assess the diversity of the flora assemblage to ensure a healthy mix of native species typical to the community being replicated is present (i.e. the community is not mono-specific except in intended areas).
- Where diversity is low, undertake diversity plantings from selected modules (refer Section 4.3.1.1). These species selected are based upon the existing adjacent vegetation communities, endemic littoral rainforest species and species suitable for threatened fauna foraging (in accordance with approved concept plans).
- Cattle, weed and fire trail monitoring and maintenance will be ongoing

This process shall occur over a period of two years.

ASSESSMENT OF REGENERATION EFFORTS AND SUPPLEMENTARY PLANTING (GROUNDCOVERS)

This management stage has involved the following (and to be implemented within upcoming efforts):

- Assessing the success of the preceding natural regeneration, pioneer plantings and diversity plantings to establish cover and floristic diversity across the rehabilitation zone/stage.
- Assess the occurrence of native ground covers within the across the rehabilitation zone/stage. In some instances, native groundcovers may be suppressed in association with early regrowth communities. If ground covers are not sufficiently established than groundcover planting from selected modules shall be performed.
- Cattle, weed and fire trail monitoring and maintenance will be ongoing

This process shall occur over a period of one year.

#### NATURAL REGENERATION AND WEED MANAGEMENT

This management stage has involved the following (and to be implemented within upcoming efforts):

- Assisted regeneration monitoring and maintenance (ongoing)
- Cattle, weed and fire trail monitoring and maintenance (ongoing)

This process shall occur for the balance of the project time-frame. Routine monitoring and maintenance (refer Section 4.4) shall be undertaken with corrective actions (refer Section 4.6) performed in incidences of non-compliance with set rehabilitation performance criteria (refer Section 4.5).



## 4.3.1.1 DESCRIPTION OF ASSISTED REGENERATION ZONES

As discussed zones 1-3 (initial 15ha rehabilitated) was subdivided into three smaller zones based upon site location. An additional area (zone 4) is now proposed to enhance linkages between existing habitats within, and immediately adjacent to the site. Each of these zones is described separately below with discussions provided regarding existing levels of natural regeneration (following trials), intended vegetation communities to be established and planting modules for supplementary planting (only required if regeneration is unsuccessful and to increase floristic diversity). It is noted that rehabilitation has commenced within zones 1-3 (mixture of both supplementary planting and natural regeneration) with additional revegetation areas now proposed for Zone 4 (refer 14). Furthermore, areas surrounding the two extraction lakes are proposed for future agricultural use (i.e. plantation of fruiting trees).

The following has been prepared and implemented within the rehabilitation zones:



REHABILITATION ZONE/AREA 1:

FIGURE 14: REHABILITATION AREA/ZONE 1

This rehabilitation zone covers 8.9 hectares and consolidates a fragmented corridor of swamp sclerophyll endangered ecological communities and wetlands located in the northern areas of the site. Cattle have been excluded from this zone to allow rehabilitation. Species noted include Paperbark, Swamp Oak, Swamp Mahogany, Corkwood, Swampbox, Banskia, Hovea, Hatpins, Pennywort, Blady Grass, Bunchy Sedge and Sundew. Wetland plants are also common with a low-lying soak in the west of the zone managed for Freshwater Wetland Endangered Ecological Community (EEC) regeneration.





## FIGURE 15: BEFORE AND AFTER REHABILITATION IMAGES OF REHABILITATION AREA/ZONE 1

Each zone/area (1A-1C) has been broken down into management stages of approximately 0.5ha to allow staged management as described in Section 4.2.1 above. Zone 1A has been managed to establish a Freshwater Wetland EEC and zones 1B & 1C managed to establish Swamp Sclerophyll EECs. Littoral Rainforest EEC species are utilized in the sub-canopy and shrub layers.





FIGURE 16: STAGING OF ZONE 1A (ALSO REFER ATTACHMENT 3)



# FIGURE 17: STAGING OF ZONE 1B (ALSO REFER ATTACHMENT 3)





FIGURE 18: STAGING OF ZONE 1C (ALSO REFER ATTACHMENT 3)



Revegetation modules for each of the community types have been provided for use in association with 'pioneer', 'diversity' and 'groundcover' supplementary plantings if required (as described in Section 4.2.1 above). A full module and quantities (refer below and Attachment 3) are included in the instances that the rehabilitation project for any reason is not performed or is completely unsuccessful (currently not the case).

However, the modules shall be utilized to select species and densities from to perform 'pioneer', 'diversity' and/or 'groundcover' supplementary plantings in the instances that the performance requirements for the natural regeneration are not achieved or fail (refer Section 4.5).

I.E. Natural regeneration canopy coverage will not be considered achieved for any one area if it does not contain a canopy or small tree typical to the community (refer Section 4.3.1 modules) within an area equal to or greater than 4sqm (i.e. if an area 2m x 2m does not contain any juveniles of required canopy tree/small tree than natural regeneration canopy cover is considered to be unsuccessful at that particular location).

In the event that such a location is deemed to occur then supplementary (pioneer species) planting shall be undertaken within the patch in accordance with Section 4.3.1.

For example, a patch of 10sqm within Rehabilitation Zone 1B is noted to be regenerated with native ground covers and/or shrubs but no tree species typical to the listed 'Swamp Sclerophyll' community occur. In this instance 5 x trees/small trees from the Rehabilitation Zone 1B are to be pocket planted at 75mm pot sizes within the patch.

Further discussions regarding performance requirements and triggers for canopy 'pioneer', 'diversity' and/or 'groundcover' supplementary plantings are provided in Section 4.5.



tal Are	ea: 1.85Ha				
'eaetat	ion Type: Fresh-wate	r Wetland			
ogorai		i i i ondina			
REES - TO	DAM				
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
cas gla	Casuanina glauca	Swamp Oak			
ca sa	Callistemon salignus	Willow Bottlebrush		1404444	21
op sau	Lophostemon sauveolens	Swampbox	У	140/0/0	
mel quí	Melaleuca quinquenervia	Paperbark			
		111 ES)			
	AITON AREA TA (185 MOD	0113)			
EDGES, R	USHES + FERNS				
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
bau jun	Baumea juncea	Bare Twigrush			
bau ter	Baumea teretifoila	Tw/grush			
ble Ind	Blechnum Indicum	Bungwa Fern			
cen asi	Centella asiatica	Pennywort		75MM TUBE	
сур ро	Cyperus polystachyos	Bunchy Sedge			Stage 1 5500
film fer	Fimbristyllis ferruginea	Rusty Fringesedge			
fim pol	Fimbristyllis polytrichoides	Fuzzy Rush			
gah asp	Gahnla aspera	Saw Sedge	100		Stage 2
jun kra	Juncus kraussii	Searush	min 6 species		5500
jun us <b>i</b>	Juncus usitatus	Common Rush	permoa		
phi lan	Phillydrum lanuginosum	Frogsmouth			Stage 3
phr aus	Phragmites australis	Common Reed			7500
sch val	Schoenoplectus validus	Clubrush			
sch it	Schoenoplectus littoralis	Clubrush			
spo v <b>i</b> r	Sporobolus virginicus	Salt Couch			
tri str	Triglochin striatum	Steaked Arrow Grass			







#### REHABILITATION AREAS 1B & 1C TOTAL AREA: 4.2HA VEGETATION TYPE: SWAMP SCLEROPHYLL

TREES - mo	dule 1, 2 and 3			
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE
cal sal	Calistemon salignus	White Bottlebrush		
cor int	Corymbia intermedia	Pink Bloodwood		
euc ter	Eucalyptus tereticornis	Blue Gum	20	75MM TUBE
euc rob	Eucalyptus robusta	Swamp Mahogany	Minimum 3	
mel qui	Melaleuca quinquenervia	Paperbark	selected	
lop sau	Lophostemon sauveolens	Swamp Box	per module	
	53 / SPIKUBS = (110)(1100 I / 0	1101-5		
CODE	PLANT SPECIES		NO PER	SIZE
CODE	PLANT SPECIES		NO PER MODULE	SIZE
CODE all tor	PLANT SPECIES Allocasuarina torulosa	COMMON NAME Forest Oak	NO PER MODULE	SIZE
CODE all tor acr imp	PLANT SPECIES Allocasuarina torulosa Acronychia imperforata	COMMON NAME Forest Oak Beach Acronychia	NO PER MODULE	SIZE
CODE all tor acr imp ban Int	PLANT SPECIES Allocasuarina torulosa Acronychia imperforata Banksia integrifolia	COMMON NAME Forest Oak Beach Acronychia Coastal Banksia	NO PER MODULE	SIZE
CODE all tor acr imp ban Int cup ana	PLANT SPECIES Allocasuarina torulosa Acronychia imperforata Banksia integrifolia Cupaniopsis anarcardiodes	COMMON NAME Forest Oak Beach Acronychia Coastal Banksla Tuckeroo	NO PER MODULE	SIZE
CODE all tor acr imp ban int cup ana dub myo	PLANT SPECIES Allocasuarina torulosa Acronychia imperforata Banksia integrifolia Cupaniopsis anarcardiodes Dubolsia myoporoides	COMMON NAME Forest Oak Beach Acronychia Coastal Banksia Tuckeroo Corkwood	NO PER MODULE	SIZE 75MM TUBE
CODE all tor acr imp ban Int cup ana dub myo hov acu	PLANT SPECIES Allocasuarina torulosa Acronychia imperforata Banksia integrifolia Cupaniopsis anarcardiodes Dubolsia myoporoides Hovea acuttiolia	COMMON NAME Forest Oak Beach Acronychia Coastal Banksla Tuckeroo Corkwood Hovea	NO PER MODULE	SIZE 75MM TUBE
CODE all tor acr imp ban int cup ana dub myo hov acu not lon	PLANT SPECIES Allocasuarina torulosa Acronychia imperforata Banksia integrifolia Cupaniopsis anarcardiodes Duboisia myoporoides Hovea acutifolia Notolaea longifolia	COMMON NAME Forest Oak Beach Acronychia Coastal Banksla Tuckeroo Corkwood Hovea Long-leaved Mock-olive	NO PER MODULE	SIZE 75MM TUBE
CODE all tor acr imp ban int cup ana dub myo hov acu not lon ptt rev	PLANT SPECIES Allocasuarina torulosa Acronychia imperforata Banksia integrifolia Cupaniopsis anarcardiodes Duboisia myoporoides Hovea acutifolia Notolaea longifolia Ptitosporum revolutum	COMMON NAME Forest Oak Beach Acronychia Coastal Banksia Tuckeroo Corkwood Hovea Long-leaved Mock-olive Forest Plttosporum	NO PER MODULE	SIZE 75MM TUBE
CODE all tor acr imp ban Int cup ana dub myo hov acu not Ion ptt rev syz ole	PLANT SPECIES Allocasuarina torulosa Acronychia imperforata Banksia integrifolia Cupaniopsis anarcardiodes Dubolsia myoporoides Hovea acutifolia Notolaea longifolia Pittosporum revolutum Syzygium oleosum	COMMON NAME Forest Oak Beach Acronychia Coastal Banksia Tuckeroo Corkwood Hovea Long-leaved Mock-olive Forest Pittosporum Blue Lillipilli	NO PER MODULE 30 Minimum 3 species selected per module	SIZE 75MM TUBE



GROUNDC	OVERS - module 3 only				
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
aus dul	Austromyrtus dulcis	Midyim			
ble ind	Blechnum indicum	Bungwa <b>l</b> Fern			
cen as <b>i</b>	Centella asiatica	Pennywort			
cyp pol	Cyperus polystachyos	Flat Sedge			
dia cae	Dianella caerulea	Blue Flax Lilly		To be	
gah asp	Gahnia aspera	Saw Sedge	50	75MM TUBE	determined on-site by bush regeneration expert
hib sca	Hibbertia scandens	Snake Vine	Minimum 5		
har vio	Hardenbergia violacea	Native sarsaparilla	species selected		
lom lon	Lomandra longifolia	Matrush	per module		
pte esc	Pteridium esculentum	Braken Fern			
sch val	Schoenoplectus validus	Clubrush	]		
xyr com	Xyris complanata	Yelloweyed Grass	7		









# REHABILITATION ZONE/AREA 2:



## FIGURE 24: REHABILITATION AREA/ZONE 2

This rehabilitation zone covers 5.08 hectares and consolidates a fragmented corridor of swamp sclerophyll endangered ecological communities and wetlands located adjacent to Mooball Creek. Cattle have been excluded from this zone to allow rehabilitation. Species noted prior to regeneration procedures include Banksia, Swamp Oak, Paperbark, Willow Bottlebrush, Corkwood, Swampbox, Geebung, Bracken, Midyim, Blady Grass, Bunchy Sedge and Tuckeroo. Wetland plants are also common with a low-lying soak in the northeast of the zone managed for a combination Freshwater Wetland and Saltmarsh Endangered Ecological Community (EEC) regeneration.




FIGURE 25: BEFORE AND AFTER IMAGES OF REHABILITATION AREA/ZONE 2

Each zone/area (2A-2C) has been broken down into management stages of approximately 0.5-1ha to allow staged management as described in Section 4.2.1 above. Zone 2B is managed to establish a Freshwater Wetland EEC and zones 2A & 2C is managed to establish a combination of Swamp Sclerophyll/Swamp Oak EECs and Banksia Forest. Littoral Rainforest EEC species is utilized in the sub-canopy and shrub layers.





# FIGURE 26: STAGING OF ZONE 2A





# FIGURE 27: STAGING OF ZONE 2B





|--|

Revegetation modules for each of the community types have been provided for use in association with 'pioneer', 'diversity' and 'groundcover' supplementary plantings if required (as described in Section 4.2.1 above). A full module and quantities (refer below and Attachment 3) are included in the instances that the rehabilitation project for any reason is not performed or is completely unsuccessful (currently not the case).

However, the modules shall be utilized to select species and densities from to perform 'pioneer', 'diversity' and/or 'groundcover' supplementary plantings in the instances that the performance requirements for the natural regeneration are not achieved (refer Section 4.5).



Rehabilitation Area 2B - species list							
REHABILITATION AREA 2B TOTAL AREA:0.43Ha (43 MODULES) FIGURE 29: WETLAND COMMUNITY MODULE							
SEDGES, R	USHES + FERNS						
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE			
bau jun	Baumea juncea	Bare Twigrush					
bau ter	Baumea terettiolia	Twigrush	1				
ble Ind	Blechnum Indicum	Bungwall Fern					
cen asi	Centella asiatica	Pennywort	]				
сур ро	Cyperus polystachyos	Bunchy Sedge					
flm fer	Fimbristylis ferruginea	Rusty Fringesedge					
fim pol	Fimbristylis polytrichoides	Fuzzy Rush					
gah asp	Gahn <b>i</b> a aspera	Saw Sedge	100	75MM TURE			
jun kra	Juncus kraussti	Searush	100	7 SMM TODE			
jun usl	Juncus ustitatus	Common Rush					
phi lan	Philydrum lanuginosum	Frogsmouth					
phr aus	Phragmites australis	Common Reed					
sch va	Schoenoplectus validus	Clubrush					
sch llt	Schoenoplectus littoralis	Clubrush					
spo vir	Sporobolus virginicus	Salt Couch					
tn1 str	Triglochin striatum	Steaked Arrow Grass					
xyr com	Xynts complanata	Hatpins					





## REHABILITATION AREAS 2A & 2C TOTAL AREA: 7.1HA VEGETATION TYPE: SWAMP SCLEROPHYLL/SWAMP OAK & BANKSIA

Rehabilitation Area 2A - species list								
REHABILITATION AREA 2A TOTAL AREA:1.40hA (140 MODULES) TREES - module 1, 2 and 3								
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE				
ban int	Banksia integrifolia	Coastal Banksia	20					
cal sal	Calistemon salignus	White Bottlebrush	Minimum 3					
lop sau	Lophostemon sauveolens	Swamp Box	species selected	75MM TOBE				
mel qui	Melaleuca quinquenervia	Paperbark	per module					

SMALL TREES / SHRUBS - module 1, 2 and 3								
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE				
acm hem	Acmena hemilampra	Broad-leaved Lillipilli						
aca sop	Acacla sophorae	Coast Wattle						
ale cor	Alectryon coriaceus	Beach Alectryon						
con bar	Commersonia bartramia	Brown Kurrajong	30	75MM TURE				
cup ana	Cupaniopsis anarcardiodes	Tuckeroo		/ SIMIM TODE				
dub myo	Dubolsia myoporoldes	Corkwood	species					
ela ret	Elaeocarpus reticulatus	Blueberry Ash	per module					
exo lat	Exocarpus latifolius	Broad-leaved Cherry						
per str	Persoonia stradbrokensis	Coast Geebung						

GROUNDCOVERS - module 3 only							
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE			
adi his	Adiantum hispidulum	Rough Maidenhair					
aus dul	Austromyrtus dulcis	Midyim					
ble car	Blechnum cartilagineum	Gristle Fern					
ble ind	Blechnum indicum	Bungwall Fern	50	75MM TUBE			
cri pen	Crinum pendunculatum	Swamp LIIIy	Minimum 5				
gah asp	Gahnia aspera	Saw Sedge	selected				
lom lon	Lomandra longifolia	Matrush	permodule				
•	•	•	•				











REHABILITATION ZONE/AREA 3:



## FIGURE 34: REHABILITATION AREA/ZONE 3

This rehabilitation zone covers 0.6 hectares and will consolidate a fragmented corridor of swamp sclerophyll, swamp she-oak and swamp box/littoral rainforest endangered ecological communities located adjacent to Mooball Creek. Cattle have been excluded from this zone to allow rehabilitation.

Species noted include Banksia, Swamp Oak, Paperbark, Bracken and Blady Grass. Whisky grass is also common.





FIGURE 35: IMAGES OF REHABILITATION AREA/ZONE 3 - 2009

The rehabilitation zone (Zone 3) occurs in one management stage of 0.6ha. The zone is managed to establish a combination of Swamp Sclerophyll and Swamp Oak EECs. Littoral Rainforest EEC species is utilized in the sub-canopy and shrub layers.



## FIGURE 36: STAGING OF ZONE 3

A revegetation module for this community type has been provided for use in association with 'pioneer', 'diversity' and 'groundcover' supplementary plantings if required (as described in Section 4.2.1 above). A full module and quantities (refer below and Attachment 3) are included in the instances that the rehabilitation project for any reason is not performed or is completely unsuccessful (currently not the case).



However, the modules shall be utilized to select species and densities from to perform 'pioneer', 'diversity' and/or 'groundcover' supplementary plantings in the instances that the performance requirements for the natural regeneration are not achieved (refer Section 4.5).

Rehabilitation Area 3 - species list REHABILITATION AREA 3 TOTAL AREA:1.20hA (120 MODULES) TREES - module 1, 2 and 3			FIGURE SCLEROP AK COMM	37: SWAI HYLL/SW UNITY MC	MP AMP DDULE
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
cas gla	Casuarina glauca	Swamp Oak	20		Stage 1
cal sa	Calistemon salignus	White Bottlebrush	Minimum 3		1200
lop sau	Lophostemon sauveolens	Swamp Box	species selected	/ SIMIN TUBE	Stage 2 1200
mel qui	Melaleuca quinquenervia	Paperbark	per module		

SMALL TREES / SHRUBS - module 1, 2 and 3						
CODE	PLANT SPECIES	COMMON NAME NC				
acr İmp	Acronychia imperforata	Beach Acronych <b>i</b> a				
all lit	Allocasuarina littoralis	Black She-oak				
ban int	Banksia integrifolia	Coast Banksia				
com bar	Commersonia bartramia	Brown Kurrajong	30			
dub myo	Duboisia myoporoides	Corkwood	Adiminau uma 2	7 SIMIN TODE		
gou sem	Gouia semiglauca	Wild Quince	species			
m <b>i</b> s pyr	Mischocarpus pyriformis	Yellow Pearfruit	per module			
per str	Persoonia stradbrokensis	Coast Geebung				
pil gla	Pilidiostigma glabrum	Plum Myrtle				
pol ele	Polyscias elegans	Cellerywood				

GROUNDCOVERS - module 3 only							
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE			
ad <b>i</b> his	Adiantum hispidulum	Rough Maldenhair					
aus dul	Austromyrtus dulcis	Midyim					
ble ind	Blechnum indicum	Bungwall Fern					
chr apl	Chrysocephalum apiculatum	Yellow Buttons	50	75MM TUBE			
cri pen	Crinum pendunculatum	Swamp Lilly	Minimum 5				
gah asp	Gahnia aspera	Saw Sedge	selected				
goo rot	Goodenia rotundifolia	Star Goodenia	permodule				
lom lon	Lomandra longifolia	Matrush					
		•	•				











# REHABILITATION ZONE/AREA 4:



## FIGURE 41: STAGING OF ZONE/AREA 4

This rehabilitation zone covers ~2.1ha hectares and is proposed to provide a wildlife corridor (linkage) between existing habitats within and external to the site and a visual buffer between the Sand Quarry and external properties. The 8m wide vegetated corridors are proposed along the northern sides of Sheens Creek, Warwick Park Rd and the unnamed watercourse between Stage 1 and Stage 2 (refer to Figure 41).



Cattle have been excluded from this zone to allow rehabilitation. Species noted include Paperbark, Swamp Oak, Swampbox, Banskia, Pennywort, Blady Grass, Bunchy Sedge and Sundew.

Each zone/area (4A-4C) has been broken down into management stages of approximately 0.5-1ha to allow staged management as described in Section 4.2.1 above. These zones will be managed to establish a Wet Schlerophyll Forest which will act as a wildlife corridor between habitats within and adjacent the site.

REHABILITA Total Area CANOPY T	ATION AREA 4 - STAGE 1 : 9,688M² (9.7 MODULES) REES							
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY			
ban int	Banksia integritolia	Coast Banksia						
cas gla	Casuarina glauca	Swamp Oak	1					
cor int	Corymbia intermedia	Pink Bloodwood	1					
euc gra	Eucalyptus grandis	Flooded Gum	1					
euc ter	Eucalyplus fereficients	Forest Red Gum	1					
euc mic	Eucalyptus microcorys	Tallow Wood	10	45LTR	970			
euc sid	Eucalyptus siderophioia	Ironbark	per mod					
lop sau	Lophostemon sauveolens	Swampbox	1					
lop con	Lophostemon confertus	Brush Box	1					
mel qui	Melaleuca quinquenervia	Paperbark	1					
CANOPY T	REES PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY			
ban int	Banksia Integritolia	Coast Banksla						
cas gla	Casuarina glauca	Swamp Oak	1					
cor Int	Corymbia intermedia	Pink Bloodwood	1					
euc gra	Eucalyplus grandis	Flooded Gum	1					
euc ter	Eucalyplus fereficionis	Forest Red Gum						
euc mic	Eucalyptus microcorys	Tallow Wood	10 min 3 species	45LTR	550			
euc sid	Eucalyptus siderophiola	Ironbark	per mod					
lop sau	Lophostemon sauveolens	Swampbox						
lop con	Lophostemon confertus	Brush Box						
mel qui	Melaleuca quinquenervia	Paperbark						
REHABILITATION AREA 4 - STAGE 3 Total Area: 6,584M <sup>2</sup> (6.6 MODULES) CANOPY TREES								
CODE	PLANT SPECIES	COMMON NAME	MODULE	SIZE	QTY			
ban int	Banksia integrifolia	Coast Banksla	4					
cas gla	Casvarina glavca	Swamp Oak	4					
corint	Corymbia Intermedia	Pink Bloodwood	4					
euc gra	Eucalyplus grandis	Flooded Gum	4					
euc ter	Eucalyptus tereficornis	Forest Red Gum	10	ACITE				
euc mic	Eucalyplus microcorys	Tallow Wood	min 3 species					
euc sid	Eucalyptus siderophiola	Ironbark	per mod					
lop sau	Lophostemon sauveolens	Swampbox	4					
lop con	Lophostemon confertus	Brush Box	4					
mel qui	Melaleuca quinquenervia	Paperbark						

## FIGURE 42: REHABILITATION AREA 4 (WET SCHLEROPHYLL FOREST) MODULE









## 4.3.2 RECONSTRUCTION/BUFFER PLANTING

This type of revegetation has occurred within the western visual buffer to the haul route, processing area, site office and wash plant. In accordance with condition 39 of the Development Consent these buffer areas were established within 12 months. Following planting of the buffers (with endemic native species) the areas are monitored and maintained as per the Rehabilitation Zones and in accordance with Section 4.4 below.



## FIGURE 44: 10M WESTERN VISUAL BUFFER LOCATION







#### BUFFER PLANTING AREA 1 Total Area: 0.34Ha (34 MODULES) CANOPY TREES

CANOLLI	REED				
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SZE	QTY
ban int	Banksia integrifolija	Coast Banksia			
cas gla	Casuarina glauca	Swamp Oak			
cor int	Corymbia intermedia	Pink Bloodwood	]		
euc gra	Eucallyptus grandiis	Flooded Gum	]		
euc ter	Eucallyptus teretticornis	Forest Red Gum	]		
euc milc	Eucallyptus microcorys	Tallow Wood	10 min 3 species	45LTR	340
euc sild	Eucallyptus siderophioia	Ironbark	per mod		
lop sau	Lophostemon sauveolens	Swampbox	]		
lop con	Lophostemon confertus	Brush Box	1		
mel qui	Melaleuca quinquenervia	Paperbark	1		
syn glo	Syzyglum moorel	Watermelon Tree	1		

#### SUBCANOPY TREES

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY	
alp exc	Alphitonia excelsa	Red Ash				
acr ob	Acronychia oblongifolia	Yellow Wood				
aca mel	Acacia melanoxylon	Blackwood				
cup ana	Cupaniopsis anacardiodes	Tuckeroo				
cry foe	Cryptocarla foetida	Cryptocarla				
cal sal	Callistemon salignus	Willow Bottlebrush		1		
dub myo	Duboisia myoporoides	Corkwood	20	300MM	680	
ela obo	Elaeocarpus obovatus	Hard Quandong	per mod			
ela obo	Elaeocarpus reticulatus	Blueberry Ash				
flc obl	Ficus obliqua	SmallHeaved Fig				
hib ti	Hib)scus till)aceus	Cotton Tree				
t aus	Litsea Australia	Laurel				
pol ele	Polyscías elegans	Cellerywood				
plitund	Pittosporum undulatum	Sweet Pittosporum				

# EARTH SCREENING MOUND Total Area: 0.14Ha

#### **CANOPY TREES** COMMON NAME SPACING SIZE CODE PLANT SPECIES ban int Citrus sinensis Orange Tree cas gla Davldsonla jerseyana Davidson Plum 3 METER CENTRES 45LTR cor int Magnifera Inidica Mango Tree euc ter Persea americano Avocado Tree

## FIGURE 46: BUFFER ZONE PLANTING LIST





# FIGURE 48: BUFFER ZONE PLANTING PROFILE

## FUTURE AGRICULTURAL USE

Vegetation Buffer

As previously mentioned, the proposed future use for the land surrounding the extraction lakes is for agricultural use, in particularly the plantation of fruiting trees (i.e.



avocado trees). The proposed agricultural use will provide additional foraging and refuge for fauna species within the locality.



## FIGURE 49: AREAS PROPOSED FOR FUTURE AGRICULTURAL USE



## 4.3.3 REVEGETATION TECHNIQUES

#### Direct Planting (Landscaping)

Landscaping is the most common method of revegetation, however, it does require establishment, maintenance watering, mulching and weeding and also has the highest cost. The landscaping process within the Buffer Zones should occur as follows:

- Identify areas on site which require planting following completion of initial weed management works;
- Ensure areas identified are appropriately prepared including:
  - Remove all rubbish including any previously dumped vegetative material, building slabs etc. Any existing hollow logs are to be retained in-situ;
  - Scarification of any compacted soil is to be undertaken to a depth of 200mm to promote aeration of the upper soil layer. No ripping is permitted within the drip zone of existing trees or native regeneration with only hand tools to be utilised for the purposes of pocket planting of shrub/groundcover tubestock.
  - Unwanted matter (i.e. stones, concrete, rubbish, rubble, sticks >20mm dia) encountered are to be removed
  - Application of 'osmocote' sustained release fertiliser in granule form or equivalent to topsoil etc).
  - Undertake follow-up weed treatment following a period of two weeks to remove individuals which may have germinated from soil bank resources;
  - Erect temporary fencing/signage to inform that the relevant location is now a rehabilitation area. This will assist in reducing accidental damage to plant stock, introduction of additional weeds and dumping of rubbish. If stock are to be grazed within adjacent paddock areas than an exclusion fence is to be erected to keep cattle out of the rehabilitation zone. Any such fence is to include a minimum 0.5m gap at ground level to allow continued passage of native fauna.
- Plant selected native species at densities documented within Attachment 3 and Figures 45 and 46 above.
- Maintain and monitor area in accordance with section entitled 'Monitoring and Maintenance'.

## 4.4 MAINTENANCE, MONITORING AND REPORTING

#### Maintenance

 Planting areas (determined on a staged basis in accordance with Section 4.3 above) are to be regularly watered for a period of twelve weeks. If during routine inspections of the rehabilitation zones there is evidence of plant poor health due to drought stress than additional watering shall be undertaken;



- Recurrent weeds within rehabilitation zones are to be removed as they occur quarterly for the first five years of assisted regeneration for each rehabilitation stage and six-monthly thereafter for the life of the development;
- Fences (where required due to cattle grazing in adjacent paddocks) are to be inspected quarterly for structural integrity and maintained as necessary;
- Locational survey pegs of the rehabilitation zone boundaries are to be inspected quarterly and maintained/replaced as necessary;
- Replacement planting of stock loss shall occur as required to achieve the performance criteria listed in subsequent section of this report;
- Existing 4wd tracks within the environmental protection zones are to be maintained quarterly to allow continued vehicle access for maintenance and emergency rural fire brigade vehicles;
- A slashed/grazed zone of 25-30m is to be maintained on the development side of the rehabilitation zones to reduce to risk of bushfire spread from the pasture grasslands.

## Monitoring

The success of a regeneration project can be assessed by systematic visual monitoring of rehabilitation areas. This need not be an overly time-consuming process and the data generated can then be used to compare the success of various treatments. The measurements to be monitored are:

- Average height of plants within rehabilitation areas (height in metres for tree, shrub and groundcover species);
- Dominant species (qualitative description of dominant species within tree, shrub and ground layer);
- Canopy cover %,
- Canopy and upper strata species diversity;
- %cover of ground cover and leaf litter;
- Health of vegetation within rehabilitation areas (per Section 4.5 below);
- Area of ground cover covered by weed species (area in square metres);
- Percentage of planted specimens survived;
- Incidence of recruitment, both exotic and native (species and quantity estimates of new species noted [i.e. A = abundant, .R = relatively common, I = isolated/scarce];
- Native fauna presence (native fauna species recorded via observation, track or trace during vegetation inspections are to be noted)

The simplest of all methods of monitoring a site is to establish permanent photo points and take photographs at regular intervals, and to regularly (i.e. six-monthly, annually) traverse the rehabilitation area(s). In this regard, each management stage of each rehabilitation zone shall be photographed in accordance with the below sections (for the life of the project) to provide a visual indication of plant growth (height and extent) and weed presence. Photographs shall be taken at the SW, SE, NW, NE corners of each monitoring site (50m x 20m quadrat or 50m x 8m quadrat (for rehabilitation area 4)).



The below monitoring forms are to be completed on an annual basis (excluding Forms A: Routine Monitoring & B: Condition Monitoring which are to be completed on a six monthly basis for the first 10 years) for each of the rehabilitation zones.

Note that monitoring form C is to commence inspection and completion similar to the staging schedule contained in Attachment 3. These inspections shall commence in the year prior to the first planting module (pioneer species) trigger and continue thereafter (has already commenced). The table below summarises the trigger for relevant inspections and form completion for forms C. Forms A, B and the visual monitoring photos are to be undertaken every six months for every rehabilitation zone. These monitoring efforts will inform maintenance requirements (particularly with regard to weed management). The relevant staging of monitoring (and assessment against performance criteria, trigger for maintenance and/or correction actions) requirements are tabulated below. Note that a less intensive regime is proposed following a five year period for each stage (i.e. when the rehabilitation zones are likely to be relatively stable):

						YE	AR				
<b>REHAB ZONE</b>	STAGE	1	2	3	4	5	6	7	8	9	10+
1A	1	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	2	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	3	R	√R	√R	√R	√R	XS	XS	XS	XS	XS
1B	1	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	2	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	3	R	√R	√R	√R	√R	XS	XS	XS	XS	XS
	4	R	R	√R	√R	√R	√R	XS	XS	XS	XS
	5	R	R	R	√R	√R	√R	√R	XS	XS	XS
1C	1	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	2	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	3	R	√R	√R	√R	√R	XS	XS	XS	XS	XS
2A	1	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	2	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
2B	1	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	2	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
2C	1	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	2	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	3	R	√R	√R	√R	√R	XS	XS	XS	XS	XS
	4	R	R	√R	√R	√R	√R	XS	XS	XS	XS
3	1	R	√R	√R	√R	√R	XS	XS	XS	XS	XS
4	1	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	2	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS
	3	√R	√R	√R	√R	XS	XS	XS	XS	XS	XS

'R' denotes that routine and condition monitoring shall be undertaken six montly during this year. This shall include assessment against performance criteria and implementation of corrective actions (i.e. further weeding, planting) where required

 $\sqrt{}$  denotes that quantitative and qualitative inspections shall be undertaken six monthly including completion of forms C. This shall include assessment against performance criteria and implementation of corrective actions (i.e. further weeding, planting) where required



'S' denotes that routine and condition monitoring shall be undertaken annually during this year. This shall include assessment against performance criteria and implementation of corrective actions (i.e. further weeding, planting) where required

'X' denotes that quantitative and qualitative inspections shall be undertaken annually including completion of forms C. This shall include assessment against performance criteria and implementation of corrective actions (i.e. further weeding, planting) where required

Please note that form content has been adapted from Kanowski, J., Catterall, C. P., Freebody, K. and Harrison, D. A. (2008) *Monitoring Revegetation Projects for Biodiversity in Rainforest Landscapes. Toolkit Version 2.* Report to the Marine and Tropical Sciences Research Facility. Reef and Rainforest Research Centre Limited, Cairns.



FORM A: ROL	<b>JTINE REHABILITATION MONITOF</b>	RING SHEET
General Management	Weeds	Vegetation regeneration
Has there been a fire within the last quarter? Do the bushfire trails or adjacent pasture areas require slashing or maintenance to reduce fire risk?	Have any areas of weeds re- established within the rehabilitation zones during the last quarter?	Natural regeneration is occurring in (record height range estimate): - Tree species - Shrub species
Is there evidence of rubbish dumping within the rehabilitation zones? Is there evidence of plant theft within the	What species? What management was undertaken to eradicate these weeds?	What are the dominant species within each layer? - Tree
rehabilitation zone? Does it appear that the rehabilitation zone has been utilized for stockpiling, vehicle parking, building waste dumping, cattle grazing or person traffic?	If management was undertaken acknowledge that such was performed in accordance with the approved rehabilitation management plan.	<ul> <li>ground covers</li> <li>Have you noticed any new native plant species since the last monthly inspection?</li> <li>If yes name the species or take a photograph</li> </ul>
If yes, acknowledge below what works were undertaken to rectify/restore and the date		Acknowledge that the required routine photographs have been taken within the rehabilitation zones
Biodiversity Have you spotted native fauna within the rehabilitation zone during inspection? If yes, what types? Koala Kangaroo/wallaby Possums/gliders Small mammal (i.e. bandicoot, echidna) Reptiles (i.e.snakes/lizards) Birds of prey	Modifications         Have there been any structural additions (eg. new tracks, buildings) to the rehabilitation zones since the last visit?	Are any of the following performance criteria exceeded (refer Section 4.5 below)? Declared Weeds? Extent of other Weeds? Survival Rate of Plants? Condition of Plants? Canopy Coverage? Tree, Small Tree & Shrub Diversity? Groundcover Coverage? General Coverage/Success?
Large nectar feeding birds (i.e. lorikeets, parrots, cockatoos) Small tree and ground birds (i.e. finches, fairy wrens, treecreepers) Glossy Black Cockatoos Other	Condition of fences <ul> <li>Good</li> <li>Need minor repair</li> <li>Poor (need replacement)</li> </ul>	If yes, what corrective action was performed (i.e. plant showed drought stress and so watering was undertaken, plant was dead so a replacement plant was pocket planted, canopy plant coverage was not achieved so relevant pioneer plants were pocket planted).



#### FORM B: SITE CONDITION

(TO BE UNDERTAKEN WITH ROUTINE PHOTO MONITORING, WEED MANAGEMENT AND REVIEW OF COMPLIANCE WITH PERFORMANCE CRITERIA)

#### Protocol for assessing site condition

The steps in condition assessment are listed below. A pro-forma for assessing site condition, based on this protocol, is provided on the following pages.

1. Obtain any previous condition assessment of the site, and other relevant documentation of the site.

2. Conduct a field inspection of the site. Based on the table below determine whether all or part of the site is:

- o **OK** ('on track' towards target conditions, requires only routine maintenance);
- **Uncertain** (significant problems identified, requires intervention); or
- **Poor** (major problems identified, likely to fail without major intervention).

If outcomes vary across a site, divide the site into zones, and record outcomes for each zone separately.

3. Make overall comments on the condition of the site.

4. Determine whether the condition of the site has changed since last assessment, and comment on any changes.

5. Complete the table describing site condition in detail. Where outcomes vary across the site, divide the site into zones ('A' = OK, 'B' = uncertain; 'C' = poor) and record outcomes for each zone separately. Comment on the attributes of each zone, particularly the factors that appear to be affecting outcomes, such as the canopy species mix and cover, regeneration/stocking rates and heights, weeds, disturbance or maintenance.

6. Make recommendations for maintenance, where relevant. The rating system is closely linked to maintenance requirements:

**Zone A** = routine maintenance only required;

**Zone B** = additional maintenance required, more than routine (need to describe);

**Zone C** = major maintenance effort required (need to describe).

Note that what comprises 'routine' maintenance will often change, e.g. from regular spraying to spot-checking and control of weeds, as sites mature. However, after major disturbance, routine maintenance may revert to regular broad-scale weed control.

7. If desired, calculate an overall 'site condition' score. This score reflects the proportion of the site in good, uncertain or poor condition, and ranges from 0% (when the entire site is in poor condition) to 100% (when the entire site is 'on track' to target conditions).

Various intermediate scores are possible (e.g. a score of 50% could mean 50% of the site is 'OK' and the rest 'poor'; it could also mean that 40% is 'OK', 20% 'uncertain', and the rest 'poor').



To calculate the score, multiply the percentage of the site zoned as A, B or C by a suggested 'condition rating' for each zone: Zone A (OK) = 1; Zone B (uncertain) = 0.5; Zone C (poor) = 0), and add the products.

## CRITERIA FOR ASSESSING CONDITION OF YOUNG REVEGETATED SITES, BEFORE INITIAL CANOPY CLOSURE.

Rating / zone	Status	Canopy cover	Ground cover	Problem weeds	Tree survival	Maintenance requirements
A	<b>OK</b> On track to target conditions	Developing well towards closed canopy	Leaf litter, mulch or soil around trees; grass/ weeds not suppressing tree growth (i.e. sparse around trees)	Not present or minor occurrence	High (at least 90%)	Routine maintenance only
В	Uncertain if will develop towards target conditions. Significant problems.	Not developing well towards closure, or outcomes are patchy	Grass/ weed cover sufficiently dense to suppress the growth of planted trees, at least in places	If present, have the potential to impede site development	Moderate (60-90%), or patchy	Extra effort required to fix problems, additional to routine maintenance
с	<b>Poor</b> Major problems. Likely to fail.	Poorly developed. Unlikely to achieve closure	Dense cover of grass/ weeds which is likely to strongly suppress tree growth	May be common, or likely to impede site development	<b>Poor</b> (less than 60%)	Major effort required to address problems.



## PROFORMA FOR ASSESSING SITE CONDITION

**PROJECT DESCRIPTION** Note: where options are given, put an 'X' next to the appropriate term(s):

Project name:	Project ID:			
Site name:		Site ID:		
Type of on-grounds: Assisted Natural Regeneration	Years since site commenced:	When was this site last assessed?		
Current assessment conducted by:	Date of current assessment:			
Overall comments on site condition:				
Has the condition of the site changed since last assessment? YES or NO If Yes, briefly describe changes in this box, and provide details in table below.				

#### **DETAILED DESCRIPTION OF SITE CONDITION** Complete table as per monitoring schedule, or if conditions have changed since last assessment.

Rating/ zone	Area (ha)	% of site	Location and factors affecting outcomes	Canopy cover (%)	Ground cover	Problem weeds	Tree survival or Recruitment	Other comments	Suggested maintenance
A = OK on track towards target									(should be routine: describe if necessary)
B = Uncertain									
significant problems									(describe)
C = Poor major problems, likely to fail									(describe)
Overall Condition and add the prod	Overall Condition Score (ranges from 0-100%) Multiply percentage of site occupied by each zone (A, B or C), by the condition rating for each zone (A = 1; B = 0.5; C = 0), and add the products: e.g. (70% x 1) + (20% x 0.5) + (10% x 0) = 80%								



## FORM C: REVEGETATION/FOREST STRUCTURE

Monitoring Forest Structure is surveyed on a 50m x 20m plot. Given its narrower width, rehabilitation zone 4 will be surveyed on a 50m x 8m plot. The following form is to be filled on a regular basis in accordance with the monitoring schedule table earlier in Section 4.4.

# **SITE FORM** REHABILITATION TRANSECT FORM (50MX20M)

Location	า					
Site No.	Recorder:		 		Day/Date:	
Purpose			 			
Locality:			 			
GPS coordinates:	Zone	E		N		Datum:

## Vegetation structure

D = dense; M = mid-dense, S = Sparse V = Very Sparse

Stratum	Height interval	Est. cover density (D,M,S,V)	
E			
T1			
T2			
S1			
G			
Structural formation: (including height)			

\_\_\_\_\_

Ecologically dominant layer:

#### Plant species

Record relative (numerical) dominance within each stratum; d - dominant; c - codominant; a - associated; s - suppressed.

Str.	Rel. dom	Scientific and/or Common Name

#### other notes

Geology mapping:	
Geology code and rock types	
Noted habitat elements:	
Noted Environmental Weeds:	



#### Photos:

Insert photo	Insert photo
SW	SE

Insert photo	Insert photo
NW	NE

## Reporting

An annual report is to be prepared documenting all monitoring/review requirements of this rehabilitation strategy including:

- Six monthly/annual visual monitoring results and photographs as outlined above
- Six monthly/annual completion of the required monitoring forms (Forms A-B)
- Six-monthly/annual rehabilitation monitoring and completion of the required forms (Form C) for relevant stages of the four rehabilitation zones.
- Any incidences of non-compliance with the performance criteria (refer Section 4.5 below)
- Corrective actions implemented in response to performance criteria noncompliance
- A work log of all monitoring/maintenance (and corrective action where required) activities performed during the preceding 12 months. This log shall include type of work (i.e. weed inspection, planting quantities, monthly photo-monitoring etc), the work trigger contained within this rehabilitation strategy (i.e. routine quarterly monitoring/maintenance or corrective action as a result of performance criteria non-compliance etc), number of personnel utilized, hours of work performed and any problems/issues identified through the preceding maintenance/monitoring 12-month period.



## 4.5 ENVIRONMENTAL PROTECTION ZONE PERFORMANCE CRITERIA

The following performance criteria are to be achieved within the EPZs:

- Existing native vegetation and areas of natural regeneration to be retained
- All rubbish/vegetation dumping, non-approved structures etc are removed from the EPZs
- No declared weeds are present
- Cattle and domestic animals are excluded
- A significant reduction in the presence of weed species is evident. In practice it is noted that the removal of all individuals of all weed species for 100% of the time is unachievable. Therefore, it is considered appropriate that the following performance criteria be adopted:
  - All large weed/ornamental trees are treated;
  - No weed shrubs/trees older than three months of age are present.
  - Densities of such shrubs/trees is not to exceed 1 per 20m<sup>2</sup>
  - Scattered groundcover weed species may occur but not in any covering an area greater than 5m<sup>2</sup>
- A survival rate of the following minimum standards are to apply for all planted trees, shrubs and groundcovers:
  - One year following planting: 90%
  - Three years following planting: 90%
  - Five years following planting: 85%
- A growth rate of the following minimum standards is to apply for all planted trees, shrubs and groundcovers and for natural regenerated plants within the rehabilitation zones:

GROWTH RATE TARGETS (HEIGHT)						
Strata		Time following initial planting				
	1 yr	3 yr	5yr			
Groundcovers	100mm	100-250mm	250mm+ or mature height (i.e.			
(prostrate)			Xyris at mature height would be			
			below 250mm)			
Groundcovers	100mm	100-300mm	300-1000mm or mature height (i.e.			
(grasses, sedges			Sporobolus at mature height would			
etc)			be below 300mm)			
Shrubs	250mm	450-500mm	1000-2000mm or mature height			
Trees	650mm	2-3m	4m+			

• Planted stock to exhibit fair or healthy conditions:

Condition	Descriptor
Healthy	Leaves green, no abnormal leaf loss
Fair	Leaves green, some yellowing of leaves, but <20% of canopy affected
Poor	Many leaves yellow or brown, substantial reduction in canopy extent
	since last measurement
Dead	Leaves brown or absent, little of the canopy remaining



- Inappropriate public access to the EPZs is to be effectively restricted (through fencing or signage). During routine monitoring (refer section 4.4 above) the EPZs are also to be inspected for the following evidence of access related impacts:
  - o Litter and/or rubbish dumping
  - o Stock theft
  - o Bicycle/pedestrian tracks/trails
  - Soil compaction
  - Fence signage vandalism/removal
  - Cattle access and associated damage (i.e. grazing, trampling etc)

#### ADDITIONAL PERFORMANCE REQUIREMENTS: ASSISTED NATURAL REGENERATION AREAS

The areas to be managed for natural regeneration area are described in Section 4.3.1. Please note that compliance with the below-listed performance requirements (in which non-compliance will trigger a selected type of supplementary planting) is to be assessed at the specified time-frame for each individual management stage within each of the defined three Rehabilitation Zones/Areas (refer Attachment 3 for staging schedules).

## Performance requirements: Canopy Coverage

Natural regeneration canopy coverage will not be considered achieved for any one area if it does not contain a canopy or small tree typical to the community (refer Section 4.3.1 modules) within an area equal to or greater than 4sqm (i.e. if an area 2m x 2m does not contain any juveniles of required canopy tree/small tree than natural regeneration canopy cover is considered to be unsuccessful at that particular location).

In the event that such a location is deemed to occur then supplementary (pioneer species) planting shall be undertaken within the patch in accordance with Section 4.3.1.

For example, a patch of 10sqm within Rehabilitation Zone 1B is noted to be regenerated with native ground covers and/or shrubs but no tree species typical to the listed 'Swamp Sclerophyll' community occur. In this instance 5 x trees/small trees from the Rehabilitation Zone 1B are to be pocket planted at 75mm pot sizes within the patch.

N.B. Please note that the trigger for pioneer plantings for Rehabilitation Zones 1A and 2B shall relate to an area of 4sqm having less than 4 plants of rush/sedge/fern (due to the ecologically dominant layer being rush/sedges rather than trees). In this instance pocket planting of 75mm tubes of the listed rush/sedge/ferns from the 'Wetland' community shall be undertaken at 1 metre centres within the identified gap.

Performance requirements: Tree, Small Tree and Shrub Diversity within Natural Regeneration

Diversity within the upper strata (trees, small trees and shrubs) of natural regeneration shall be required to achieve the following:



Community Type (Forests/Open Forests)	Canopy Tree minimum Diversity per 100sqm	Small Trees/Shrubs minimum Diversity per 100sqm	Applicable Rehabilitation Zones
Swamp Sclerophyll	3	3	1B, 1C, 2C
Swamp Sclerophyll, Swamp She-oak, Banskia	3	3	2A,
Swamp Sclerophyll, Swamp-oak	3	3	3

Community Type	Rushes/sedges/ferns Per 100sqm	Applicable Rehabilitation Zones
Coastal Wetland	Not applicable*	1A, 2B

\*Please note that Rehabilitation Zones 1A and 2B shall be exempt from this diversity performance requirement given that wetland communities naturally exhibit areas of mono-specific aquatic plant growth.

## Performance requirements: Groundcover Coverage

Natural regeneration groundcover coverage will not be considered achieved for any one area if it does not contain native ground cover or leaf litter from the upper strata (grass, sedge, fern, bracken etc) within an area equal to or greater than 2.25sqm (i.e. if an area 1.5m x 1.5m does not contain any ground covers or natural leaf litter layer than natural regeneration of ground covers is considered to be unsuccessful at that particular location). In this instance groundcovers from the relevant Rehabilitation Zone module are to be planted at 75mm pot sizes within the patch.

Please note that this performance requirement does not apply to Rehabilitation Zones 1A and 2B. Rush/sedge/fern coverage within these coastal wetland regeneration zones are addressed through 'canopy coverage' above.

## Performance requirements: General Coverage/Success of Natural Regeneration

Natural regeneration coverage will not be considered achieved for any one area if it contains a bare or denuded area greater than 6.25sqm (i.e. if an area 2.5m x 2.5m is bare than natural regeneration has been unsuccessful at that particular location). In this instance full planting of the relevant module for that particular Rehabilitation Zone shall be required.

## 4.6 CORRECTIVE ACTIONS

The following corrective actions are to be implemented in instances of non-compliance with the performance requirements:

• If rubbish or litter is found within an EPZ it is to be immediately removed



- Where weed re-establishment is noted additional removal/management works are to be instigated in accordance with Section 4.1
- Where planted specimens fail to strike or plantings do not achieve the set survival percentages then supplementary plantings are to be undertaken. Where it is considered that a particular species has failed due to non-suitability or repeated failure within the planting environment a suitable native plant replacement species is to be discussed with and approved by Council's ecologist;
- If regular increases in height and crown cover extend are not recorded within the monitoring period a horticulturalist/arborist is to be employed to identify likely causes and to recommend measures (i.e. fertiliser application, increased watering etc) to encourage increased growth.
- If retained or planted vegetation show signs of ill health (i.e. poor or dead), a horticulturalist/arborist is to be employed to identify likely causes and to recommend mitigation measures to improve regeneration conditions
- If access related impacts are evident (refer 4.5 above) the following actions are to be implemented:
  - Cattle are to be removed
  - Removed/damaged plants is to be via planting the relevant module to the rehabilitation area/zone
  - If damaged to fencing or signage or the viewing platform is recorded the structures are to be repaired to pre-existing condition and Council notified regarding the recorded damage/vandalism.
- In the event that a location within a Rehabilitation Zone is deemed not to comply with the 'canopy coverage' requirements then supplementary (pioneer species) planting shall be undertaken in accordance with Section 4.3.1.
- In the event that a location within a Rehabilitation Zone is deemed not to comply with the 'diversity' requirements then supplementary (tree, small tree and shrub diversity species) planting shall be undertaken in accordance with Section 4.3.1.
- In the event that a location within a Rehabilitation Zone is deemed not to comply with the 'groundcover coverage' requirements then supplementary (groundcover species) planting shall be undertaken in accordance with Section 4.3.1.
- In the event that a location within a Rehabilitation Zone is deemed not to comply with the 'general coverage' requirements then full planting of the relevant module for that particular Rehabilitation Zone shall be required.

Please note that compliance against the relevant 'assisted regeneration' performance requirements (in which non-compliance will trigger a selected type of supplementary planting) is to be assessed at the specified time-frame for each individual management stage within each of the defined three Rehabilitation Zones/Areas (refer Attachment 3 for staging schedules).

N.B. Please note that it is considered likely that unanticipated adaptive management procedures may be necessary throughout the course of the rehabilitation program (i.e. extension of watering regime during times of low rainfall, alteration to weed management measures if unsuccessful for a certain species, etc). Any such amendment to the management regime contained within this document shall be undertaken in consultation with qualified ecologist.



# 5.0 FAUNA HABITAT ENHANCEMENT

Enhancement of fauna habitat values within the EPZ is proposed to increase the potential for native fauna to continue to utilise the area. This will be achieved through the implementation of the following mechanisms:

• <u>Suitable habitat</u>: it is considered that revegetation of Swamp Sclerophyll, Swamp Oak and Banksia Open Forests with Littoral Rainforest elements will (over time) provide potential habitat for a number of threatened fauna species recorded onsite and elsewhere in the region including:

Species	Habitat Requirements
	This species is associated with waterbased habitats including estuaries, coastal wetlands, rivers and streams. The Osprey is predominately a coastal raptor frequenting estuaries, bays, inlets, islands and rocky cliffs within all Australian states except for Tasmania and sporadically within Victoria (DEC, 2005; NPWS, 2002). It is noted however, that the species sometimes inhabits inland islands (Pizzey and Knight, 1997; Readers Digest, 2002). Within suitable environment it usually constructs a nest in an overhanging large tree or upon elevated man made structures such as platforms or telegraph poles.
	The species preys almost exclusively on fish by usually hunting alone and traversing the water's surface for prey which it secures by swooping over the waters surface or plunging below (Readers Digest, 2002; Clancy, 2005). Studies of prey middens on Lizard Island within the Great Barrier Reef also noted that occasional Terns and crustaceans are sourced for food (Smith, 1985).
Osprey (Pandion haliaetus)	The eastern forests (Communities 1 & 4-9) fringing Mooball Creek and the surrounding estuarine/riparian habitats including Wooyung Nature Reserve are considered to provide suitable habitat for the Osprey. The 15ha of rehabilitation proposed is unlikely to significantly benefit the osprey with the exception of expanding the width of existing riparian buffers to Mooball Creek.
	Minor potential occurs for the Glossy Black Cockatoo to frequent the site due to the limited presence of foraging materials ( <i>Allocasuarina</i> spp.) and suitable nesting trees (large hollow within a live or dead Eucalypt: 10-20m, Depth: 40-120cm, Entry: ~21cm: Inside Dia: ~23cm (Forshaw, 1981; Gibbons & Lindenmayer, 2002)) are considered to be largely absent. Suitable habitat is considered to be limited to the NW Brushbox/Blackbutt Forests on Bedrock although Allocasuarina presence is very sparse. Over the balance of the site Allocasuarinas are generally absent although <i>Casuarina</i> <i>glauca</i> is very common.
Glossy Black- Cockatoo (Calyptorhync hus lathami)	Plantings of Allocasuarina will increase foraging resources for the cockatoo onsite. Plantings of eucalypts may also provide potential nest hollows in the 60-100 year timeframe.
	The species is generally associated with wetlands, mudflats, mangroves, swamps and floodplains while it may also sometimes be found in open woodland environs where a grassy understorey is present (NPWS, 2002, Readers Digest, 2002; DEC, 2005). Irrigated lands are also occasionally a foraging resource and it has also been recorded foraging in artificial wetlands of sewerage treatment plants (ERM, 2001).
Black-necked Stork (Ephippiorhync hus asiaticus)	Suitable habitat for the species is considered to occur within the eastern interconnected forest remnants adjacent Mooball Creek (Communities 1 & 4-9) although there is an absence of large areas standing water within these communities. The shallow reaches of the creek itself may function as a foraging locality for the species and overhanging trees may be utilized as nest sites.



	The 15ha of rehabilitation proposed is unlikely to significantly benefit the stork with the exception of expanding the width of existing riparian buffers to Mooball Creek and providing alternate future options for nesting trees. Creation of large lakes may increase potential foraging in the future (following completion of extraction).
	This species primarily occurs within Eucalypt Forest and Woodlands containing a suitable density of favoured food trees within coastal eastern and southeastern Australia. The suitability of forest and woodland communities as habitat for Koalas is influenced by the size and species of trees present, soil nutrients, climate, rainfall and the size and disturbance history of the habitat patches (Reed <i>et al.</i> 1990 in NPWS, 1999).
	Whilst a limited abundance of favoured Koala foraging trees (principally <i>E. robusta, E. tereticornis, E. microcorys</i> ) were recorded on site and favoured habitat (Eucalypt Forest) was restricted to the fragmented patches of Brushbox Forest on Bedrock (Community 3) and small copses of Blackbutt Forest (Community 2) the Koala was recorded during survey. Two individuals were recorded during spotlighting within the largest of the western Brushbox Forest remnants with an additional individual recorded during diurnal survey.
Koala (Phascolarctos cinereus)	Surveying of the eastern forest remnants (interconnected Communities 1 & 4-9) did not locate any Koalas with a general absence of suitable foraging resources and preferred habitat type noted. Extension of forest remnants through revegetation and incorporation of favoured foraging trees will increase potential habitat for the koala.
Little Bentwing	This species utilises well-timbered habitats including rainforest, <i>Melaleuca</i> swamps and dry sclerophyll forests where it It feeds on insects within the canopy and requires caves, mines, stormwater drains and/or tree hollows to roost (Strahan eds, 2002). Potential habitat for the species occurs in association with the remnant forest patches on site (Communities 1-3 & 4-6).
Bai (Miniopterus australis)	15ha of potential habitat for the bentwing bat will occur in association with revegetation works.



Species	Habitat Requirements
	This raptor utilises coastal-subcoastal tall forests/woodlands, savanna traversed by forested rivers and rainforest fringes (Marchant & Higgins, 1993; NPWS, 2002; NPWS, 1999). In NSW, the goshawk is associated with frequent mixed subtropical rainforest, <i>Melaleuca</i> swamp forest and open eucalypt forest along coastal rivers (Debus 1993 in NPWS, 2002). Nesting is restricted to tall trees within proximity of a creek, river or wetland (NPWS, 1999; NT Parks & Wildlife Commission, 2002). Hunting occurs for medium-large birds within open forests and riparian/gallery forests over a very large home range of up to 200km2 (Blakers <i>et al.</i> , 1984, Aumann and Baker-Gabb, 1991, Czechura and Hobson, 2000; NPWS, 2002).
	It is considered that potential habitat for this species occurs in association with the eastern connected forests fringing Mooball Creek. It is noted that these remnants (~27ha) are highly unlikely to sustain an individual goshawk which would occupy a much larger home range. It is noted that previous records of Goshawks from the region (NPWS database) are limited to a 1957-87 record within Billinudgel Reserve to the south. If this was an accurate record and Red Goshawks persist it could be considered reasonably likely that they would hunt within the Mooball Creek environments to the east of the site.
Red Goshawk (Erythrotriorchis radiatus)	The 15ha of rehabilitation proposed is unlikely to significantly benefit the goshawk with the exception of expanding the width of existing riparian buffers to Mooball Creek. However, provision of an additional 15ha of forest is, over time, likely to result in an increased abundance of forest birds which provide potential prey for the goshawk.
	Long-nosed Potoroos are generally restricted to areas with an annual rainfall greater than 760 mm where they inhabit dry and wet sclerophyll forests and woodland with a heathy understorey (Johnson in Strahan, 2002; DEC, 2005). The preferred habitat in north eastern NSW is dry and wet open shrubland (Mason 1997, DEC, 2005, Johnston in Strahan, 2002). In all habitats the species requires relatively thick groundcover growing on friable soils (Bennett, 1993). Within these areas the Potoroo digs for its food the main component of which is hypogeal fungi with other important items including hard-bodied arthropods, vascular plant tissues, seeds and fleshy fruits (Bennett & Baxter, 1989; Claridge et al, 1993) Within the site the eastern forests occur on sandy, friable soils although a thick, heathy groundcover is mostly absent. Some areas of Casuarina Forest (Community 8) and Swampbox Forest on sand (Community 4) do have small patches of dense grass, bracken and lantana growth which may provide potential habitat. Trapping, spetiabling and regular
	traverse of such areas failed to record the species. One noted recorded occurs from the region in association with Billinudgel Reserve further south which is mapped as containing more suitable shrubland/heathland complexes on sand (NPWS, 2001).
Long-nosed Potoroo ( <i>Potorous</i> <i>tidactylus</i> )	The 15ha of revegetation is considered unlikely to be beneficial for potoroos given their absence from the site and absence of terrestrial corridors of favoured habitat between the site and existing known population locations.
	This species forages on a variety of fruits, flowers and pollen. It occurs in subtropical and temperate rainforests, tall sclerophyll forests and woodlands, heaths and swamps (Eby 1995). It additionally utilises cultivated fruit crops and urban gardens.
	This species was recorded mostly within the interconnected eastern forests (Communities 1 & 4-9) on flowering Melaleucas and Banksias and fruiting Figs and Acronychias. It was also recorded within the western forests on bedrock (Communities 2 & 3) and the open grassland/pasture with scattered trees (Community 10) on flowering Eucalypts and fruiting Figs.
Grey-headed Flying-fox ( <i>Pteropus</i> poliocephalus)	15ha of potential habitat for the flying fox will occur in association with revegetation works. Favoured heavily flowering trees and fruiting species will provide an additional source for foraging.


Species	Habitat Requirements		
	This macropod species is known to inhabit dense forest vegetation (usually rainforest) but is also known from wet sclerophyll forest and occasionally in dry vine-thickets (DEC, 2005; Vernes et al, 1995, Johnson & Vernes, 1994). A dense understorey is required for shelter and refuge from predators although it will forage in pastures adjacent the forested edges as well as browse grasses, fruits, fungi and shrubs within the forest interior (DEC, 2005). The home range individual is 1-4ha (Johnson & Vernes in Strahan, 2002) with a study of fragmented rainforest in north Qld noting the range to be larger during diurnal periods than nocturnal periods.		
	Within the site potential habitat is considered to be limited to the eastern patches of Littoral Rainforest although the understorey whilst structurally diverse in the mid and upper strata is not as dense within the ground layer which is dominated by leaf litter and woody debris. Thickets of grass, shrub/heath, bracken/fern and/or rushes/sedges are isolated. Spotlighting and regular diurnal traverses (particularly proximate to mammal trapping lines) failed to detect the species presence.		
Red-legged Pademelon ( <i>Thylogale</i> stigmata)	The 15ha of revegetation is considered unlikely to be beneficial for pademelons given their absence from the site and absence of terrestrial corridors of favoured habitat between the site and existing known population locations.		
	The Flying-fox forages on the nectar and pollen of native trees and fruits of native rainforest trees and vines as well as ornamental and orchard trees/crops. The preferred food includes the blossoms of Eucalypts, paperbarks and turpentines. Other native and introduced blossoms and fruits are also eaten (Strahan eds, 2002; Nicola and Hall, 2004).		
Black Flying-fox ( <i>Pteropus alecto</i> )	15ha of potential habitat for the flying fox will occur in association with revegetation works. Favoured heavily flowering trees and fruiting species will provide an additional source for foraging.		
	This species typically prefers the coastal forested and wooded lands of tropical and temperate Australia where it appears to occupy large hunting ranges of more than 100km2 (Marchant & Higgins 1993; NPWS, 1999; DEC, 2005). A common feature of the kite's habitat is the presence of profuse eucalypt blossom and attendant nectarivorous/passerine birds which are the favoured prey of the kite (Readers Digest, 2002, NPWS, 1999).		
	It is considered unlikely that the species would inhabit the site due to the absence of large tracts of eucalypt forest/woodlands. Some potential occurs in association with the eastern connected forests fringing Mooball Creek which at its upper extent is proximate to the previous record of the kite in the region at Billinudgel Reserve (NPWS database). If this was an accurate record and the Kite persists in this habitat it could be considered likely that they would hunt within the Mooball Creek environments to the east of the site.		
Square-tailed Kite (Lophoictinia isura)	The 15ha of rehabilitation proposed is unlikely to significantly benefit the kite with the exception of expanding the width of existing riparian buffers to Mooball Creek. However, provision of an additional 15ha of forest is, over time, likely to result in an increased abundance of nectarivorous/passerine birds which provide potential prey.		



Species	Habitat Requirements				
	This species generally occurs within sub-tropical rainforest, camphor laurel and occasionally wet sclerophyll and swamp forests which contain suitable fruiting species for foraging (DEC, 2005; Recher et al, 1995). As an obligate frugivore a high proportion of fruiting species (figs, lillipillis, laurels etc) is necessary and as such rainforest habitats are favoured. The species is considered a partial migrant and moves north in autumn/winter and returning in spring/summer to breed (Recher et al, 1995).				
Rose-crowned	Within the site it is considered that preferred habitat is limited to the Littoral Rainforest (Community 1) in the eastern areas of the site which contain the highest density of fruiting species. Avifauna survey including call playback failed to detect the species although the survey was limited by the time of year performed.				
Fruit Dove ( <i>Ptilinopus</i> <i>regina</i> )	Incorporation of littoral rainforest type fruiting species will provide an additional source for foraging within the rehabilitation zone.				
	This species is known to inhabitat a broad range of habitats incorporating a dense ground cover layer including rainforest, eucalypt forest, heathland, marshland, grassland and rocky areas (Redhead in Strahan, 2002; Lewis, 2005).				
	Potential habitat is considered to occur within the interconnected eastern forests on sand (Communities 1 & 4-9) which incorporate suitable understorey cover and diversity. Pitfall trapping within these areas failed to record the species.				
Common Planigale ( <i>Planigale maculata</i> )	The 15ha of revegetation is considered unlikely to be beneficial for planigales given their absence from the site and absence of terrestrial corridors of favoured habitat between the site and existing known population locations. However, if planigales occur (but were unrecorded during previous survey) the rehabilitation of 15ha of native vegetation on sandy, friable soils will provide increase potential habitats and dispersal zones for the species.				
Common	This species is one of the smallest members of the flying fox family (Pteropodidae) and is considered to be a specialist pollen feeder favouring Banksia, Melaleuca, Callistemon and certain species of Eucalypt (Strahan eds, 2002). Required habitats include Coastal rainforest, heathlands and Melaleuca swamps. The presence of Melaleuca Forest (Community 6) on site and extensive areas of Melaleuca Forest and Eucalypt Forest within the adjacent reserve indicates the species may be a potential occurrence.				
Blossom Bat (Syconycteris australis)	15ha of potential habitat for the blossom bat will occur in association with revegetation works. Favoured heavily flowering trees (particularly banksias) will provide an additional source for foraging.				
Eastern Bentwing ( <i>Miniopterus</i>	This species usually forages on insects within intact, well timbered forest complexes and have been found to roost within caves, tunnels, stormwater culverts or disused mining areas (Strahan eds, 2002; DEH, 2005). They utilise a broad range of habits including wet and dry sclerophyll forest, open woodland, paperbark forests, rainforests and open grasslands (North & Pasic, 2006).				
schreibersii oceanensis)	15ha of potential habitat for the bentwing bat will occur in association with revegetation works.				
	In NSW this species primary habitat is mangrove forest where a few colonies exist at scattered localities, including the Tweed, Richmond and Clarence River estuaries and Stuarts Point south of Macksville (DEC, 2005). The honeyeater has also been recorded from other coastal forest types including casuarinas and paperbark forest (DEC, 2005).				
Mangrove Honeyeater ( <i>Lichenostomu</i> s fasciogularis)	Within the site the eastern Mangrove areas (Community 9) fringing Mooball Creek is considered to provide suitable habitat for the species. The 15ha of rehabilitation proposed is unlikely to significantly benefit the honeyeater with the exception of expanding the width of existing riparian buffers to Mooball Creek				



Species	Habitat Requirements				
	The Brolga inhabits the large open swamplands/wetlands of coastal and subtropical coastal Australia where it may form flocks of several hundred individuals during the breeding season (Readers Digest, 2002). Studies conducted in southern NSW and Northern Victoria (Charles Sturt University, 2000) indicates that most Brolga breeding sites were large (>50 ha) remnant wetlands with extensive areas of water around 30 cm deep. More than 90% of breeding sites were dominated by Canegrass ( <i>Eragrostis australasica, E. infecunda</i> ) or Spike-rushes ( <i>Eleocharis</i> species), with emergent vegetation cover usually around 25% and 90 cm in height. DEC (2005) notes that the species may also forage within grassed paddocks or ploughed fields.				
Brolga ( <i>Grus</i> <i>rubicund</i> a)	exception of expanding the width of existing riparian buffers to Mooball Creek. Creation of large lakes may increase potential foraging in the future (following completion of extraction).				
	This species favors coastal rivers and inlets from the Clarence River, north. It prefers densely overgrown margins of permanent terrestrial freshwater wetlands such as creeks and rivers, billabongs, ponds, swamps, waterholes, dams, lakes and roadside ditches (Muranyi and Baverstock, 1996).				
Bush-hen ( <i>Amaurornis</i> <i>olivaceus</i> )	Potential habitat is considered to occur within the area in association with Mooball Creek and adjacent forest remnants (Communities 1 and 4-9). The 15ha of rehabilitation proposed is unlikely to significantly benefit the hen with the exception of expanding the width of existing riparian buffers to Mooball Creek. Creation of large lakes may increase potential foraging in the future (following completion of extraction).				
	This species is widespread throughout predominately coastal Australia where its preferred habitat consists of open forest-woodlands containing a grassy understorey with fallen timber and leaf litter (Readers Digest, 2002; NPWS, 2006). Foraging however, has been noted to occur over a broader spectrum of habitats including paddocks, grasslands, domentic areas (gardens, sports fields, [golf courses, residential areas pers. obs] etc), estuarine areas (mudflats, saltmarsh, mangrove forest, swamp oak, melaleuca forest) (NPWS, 1999; 2006).				
Bush Stone- curlew ( <i>Burhinus</i>	Potential habitat is present on the site in association with the eastern and western forested remnants (Communities 1-8). It is also considered that the open paddock/pasture areas (Community 10) provide potential foraging habitat as the species is known to utilize domestic areas and be tolerant of human presence/disturbance.				
grallarius)	This species is distributed throughout coastal western, northern and eastern Australia from Norwest Cape to the Manning River (Readers Digest, 2002). Within this area it utilised open beaches, islands, reefs and sand/mudflats (NPWS, 2005; 1999; 2002) where it forages on crabs and other hard shelled marine invertebrates (Readers Digest, 2002).				
Beach Stone- curlew ( <i>Esacus</i> neglectus)	Mudflat and sandy beach areas are present adjacent the site to the east in association with Moball Creek and Wooyung Nature Reserve. The 15ha of rehabilitation proposed is unlikely to significantly benefit the curlew with the exception of expanding the width of existing riparian buffers to Mooball Creek.				



Species	Habitat Requirements				
	This species is generally recorded within tussock-grasslands but has also been noted to occur within heathland, swamps, coastal dunes, tree-lined creeks, treeless plains, mangrove fringes, grassy gaps between trees and crops and sugar cane plantation (Garnett and Crowley 2000; Pizzey and Knight, 1997). Within these habitats it sources a wide range of prey including birds, insects and terrestrial mammals. However, it feeds predominately on rodents and its population numbers can fluctuate wildly with the rise and fall of prey populations (Olsend and Doran, 2002). The fall of primary prey species following plague events (during which owl breeding increases) can result in widespread dispersal by the Owls with starvation also noted as the forage base reduces (Debus et al, 1998).				
Grass Owl ( <i>Tyto</i> capensis)	The 15ha of rehabilitation proposed is unlikely to significantly benefit the grass owl with the exception of expanding the width of existing riparian buffers to Mooball Creek which contains marginal areas of habitat.				
	The Myotis roosts within caves, tunnels, hollow-bearing trees, bridges, buildings and dense tree foliage always in close proximity to permanent water (NPWS, 2002; Richards, 2002). It forages over waterbodies where it scoops insects and small fish from the water surface or catches insects aerially (DEH, 2005; Menkhorst, 1996; Richards, 2002). It has been recorded foraging over small creeks, coastal rivers, estuaries, lakes and inland rivers (Law & Anderson, 1999) and other smaller waterbodies including farm dams (Law et al, 1998).				
Large-footed Myotis ( <i>Myotis</i> <i>adversus</i> )	Within the site potential foraging habitat is associated with the eastern forests fringing Moball Creek (Communities 1 & 4-9) and the creekline itself. The 15ha of rehabilitation proposed is unlikely to significantly benefit the myotis with the exception of expanding the width of existing riparian buffers to Mooball Creek. Creation of large lakes may increase potential foraging in the future (following completion of extraction).				
Wallum	This species is generally associated with Melaleuca Forest, Heathland or sedgelands containing acidic tannin-stained water in the coastal zone of SE Qld and NE NSW (Robinson, 1993; Qld Frog Society, 1999). Potential habitat is considered to occur in the eastern portions of the site in association with Paperbark and Paperbark/Swamp Oak Forests (Communities 6 and 7) which contain occasional freshwater soaks with sedge/rushes (Baumea, Juncus, Phragmites etc). Significant areas of Paperbark Forest are also present to the north of the site in association with an extensive SEPP 14 wetland area.				
Froglet ( <i>Crinia</i> <i>tinnula</i> )	Expansion of swamp sclerophyll habitats on coastal sands may increase the extent of potential habitat in the local area for the froglet.				
	This species is known from ephemeral wetlands and acid swamps containing sedgeland, banksias (wallum) and melaleuca forest/woodland within the coastal sandy zones of NE NSW and SE QLD (DEH, 2005; NPWS, 2002). During wet periods the frog can be found on emergent vegetation (rushes, sedges, ferns) whilst during drier periods it may be found at the base of such vegetation (DEH, 2005).				
Wallum Sedge-frog	Potential habitat is considered to occur in the eastern portions of the site in association with Paperbark and Paperbark/Swamp Oak Forests (Communities 6 and 7) which contain occasional freshwater soaks with sedge/rushes (Baumea, Juncus, Phragmites etc). Significant areas of Paperbark Forest are also present to the north of the site in association with an extensive SEPP 14 wetland area.				
(Litoria olongburensis)	Expansion of swamp sclerophyll habitats on coastal sands may increase the extent of potential habitat in the local area for the sedge frog.				



Species	Habitat Requirements				
	This species of Glider is associated with dry sclerophyll forest and woodlands although in northern NSW and Qld it has been recorded from wet sclerophyll environments (Suckling in Strahan eds, 2002; Lindenmayer 2002). It is considered to be most abundant in associations containing winter flowering Eucalypts and/or environments with a high abundance of Acacia, Banksia species in the lower layers (Smith & Murray, 2003; Menkhorst et al, 1998; Quinn, 1995).				
	Within the canopy of the preferred habitat numerous trees bearing hollows are critical habitat values required to support populations of the species (Quinn, 1995; Smith & Murray, 2003; Lindenmayer, 2002). Gliders are known to regularly swap den trees and utilise a number of such dens (between 6 and 19 den trees per Glider) within their home range (van der Ree, 2000). These results are supported by survey work undertaken by Southern Cross University (June/July 2002) which indicated that 12 radio tracked gliders utilised 37 den trees incorporating live hollow bearing trees and stags (Cited in Warren, 2004).				
Squirrel Glider ( <i>Petaurus</i> <i>norfolcensis</i> )	Within the site, favoured habitats are considered to be restricted to the eastern forests (some minor presence of bloodwood and ironbark with <i>Banksia integrifolia</i> relatively common) although the low abundance of suitable hollowbearing trees and general dry sclerophyll forest may limit the potential of this area to be of significance for the species. Rehabilitation/planting of eucalypts and banksias will increase foraging resources for the glider onsite. Eucalypts may also provide potential nest hollows in the 60-100 year timeframe.				
	This species of bat inhabits lowland subtropical rainforest and wet and swamp eucalypt forest, extending into adjacent moist eucalypt forest with coastal rainforest and patches of coastal scrub particularly favoured (DEC, 2005; NPWS, 2002). Roosting occurs within tree-hollows, under bark and/or palm fronds and within dense foliage with a seasonal shift in roost sites from rainforest edges (summer) to the rainforest interior (winter) (NPWS, 2002; Parnaby in Strahan, 2002; Lunney et al, 1995).				
Eastern Long- eared Bat ( <i>Nyctophilus bifax</i> )	It is considered that the potential habitat for the species on site occurs within the swamp and coastal forests in the eastern areas of the site (Communities 1 & 4-9) which will be retained in association with the proposal. 15ha of potential habitat for the long-eared bat will occur in association with revegetation works				
Barred Cuckoo-shrike	This species has been recorded from a variety of habitats including rainforest, eucalypt forests and woodlands, clearings in secondary growth, swamp woodlands and timber along watercourses within Coastal NSW (NPWS, 2002). Foraging requirements include fruiting tree species within in rainforest, wet sclerophyll forest, vegetation remnants or isolated trees (DEC, 2005) and insects captured among foliage (NPWS, 2002).				
(Coracina lineata)	Incorporation of littoral rainforest type fruiting species will provide an additional source for foraging within the rehabilitation zone.				
	This species favours coastal wetlands and swamps with prolific reed/sedge growth mostly within northern Australia (NPWS, 2002; Tulloch et al, 1981). Breeding is confined to the northern areas in association with large floodplains of creeks/rivers generally within 80km of the coast (Frith and Davies, 1961). Dense sedge/rush growth within shallow waters in these locations is favoured for nest formation (Tulloch et al, 1981; Bayliss and Yeoman, 1990).				
Magpie Goose (Anseranas semipalmata)	The 15ha of rehabilitation proposed is unlikely to significantly benefit the goose with the exception of expanding the width of existing riparian buffers to Mooball Creek. Creation of large lakes may increase potential foraging in the future (following completion of extraction).				



Species	Habitat Requirements			
	The species is widely distributed throughout the coastal regions of Australia but is more common in the northern extent of the country. Within its distribution, the species shows a preference for densely vegetated areas within terrestrial and aquatic wetlands. It has been recorded from a variety of vegetation types (including grassland, mangroves, wet sclerophyll forest, rainforest) where permanent water is present (Marchant & Higgins, 1990; Simpson & Day, 1996; NPWS, 2001).			
Black Bittern ( <i>Ixobrychus</i> <i>flavicollis</i> )	Suitable habitat is considered to occur in proximity to the site in association with Mooball Creek and the adjacent estuarine/freshwater communities (Mangroves, Paperbark Forest). The 15ha of rehabilitation proposed is unlikely to significantly benefit the bittern with the exception of expanding the width of existing riparian buffers to Mooball Creek.			
	This species is recorded in coastal Australia from Shark Bay to the Clarence River where it is almost exclusively associated with mangrove and estuarine areas (NPWS, 2005; Readers Digest, 2002). The species is considered possible to occur given the presence of mangrove forest and estuarine communities both on and adjacent the eastern areas of the site.			
Collared Kingfisher ( <i>Todiramphus</i> <i>chloris</i> )	The 15ha of rehabilitation proposed is unlikely to significantly benefit the honeyeater with the exception of expanding the width of existing riparian buffers to Mooball Creek			
	This species is confined to mature rainforest and adjacent wet sclerophyll environments in eastern Australia from Cape York to around Coffs Harbour. As an obligate fruigivore it requires a high availability of fruiting materials which it generally feeds on in the high canopy (Recher et al, 1995).			
Wompoo Fruit Dove ( <i>Ptilinopus</i> <i>magnificus</i> )	Suitable habitat for this species is considered to be largely absent from the site although Community 1: Littoral Rainforest contains species typically favoured by the species (Figs, palms, laurels, lillipillis, barbed-wire vine). Previously clearing of the site in the 1950s limits the height of the canopy (15-20m) of this association which is considered to be lower than typically favoured by the species. Incorporation of littoral rainforest type fruiting species will provide an additional source for foraging within the rehabilitation zone.			

Reviewing the above habitat requirements, it is considered that rehabilitation and restoration proposed likely to increase the potential habitat and refuge available for the listed species within the sub-region through the provision of:

- <u>Foraging resources</u>: Identified food resources for various significant fauna species will be incorporated within revegetated/rehabilitated areas (i.e. *Allocasuarina* species for glossy black cockatoos, nectar and fruit producing flora species for birds and bats, eucalypts for koalas etc);
- <u>Cover</u>: Native understorey regeneration and additional supplementary plantings will provide cover for small native species from predators whilst moving throughout the rehabilitation zones. The locations of the Rehabilitation Zones which aim to close canopy gaps, increase corridor widths, close the gaps between currently isolated remnants and rejoin the riparian habitats of Mooball Creek to the western eucalypt forests on bedrock, will significantly increase dispersal options for native fauna;



# 5.1 FAUNA MANAGEMENT DURING VEGETATION CLEARING

The approved quarry development requires only the minor removal of semi-mature Swamp Oaks and isolated blue gums and areas of pasture grassland in association with preparation and ongoing sand extraction procedures (refer Figure 50 below).



FIGURE 50: CLEARING ZONES

Notwithstanding the minor extent of native vegetation removal required, best management practice requires that wildlife whose habitat has been destroyed by human activity, must be considered, through the formation of a fauna management plan (FMP) to address the issue. The following actions will effectively reduce potential fauna mortality due to removal of habitat as part of the proposed vegetation clearing works and through the ongoing operation of the quarry.

# 5.1.1 GENERAL FAUNA MANAGEMENT

It is intended that, in regard to the clearing process and associated staff, a wildlife spotter-catcher shall confirm that the tree felling operation shall occur in a manner set out below that allows safe dispersal or capture of fauna:



- Clearing of trees, will occur after inspection to confirm absence of current or anticipated fauna activity (e.g. active bird nests, arboreal mammals).
- All static fauna valued trees will be clearly identified with high visibility tape or marker spray paint.
- The static fauna values, specifically hollow-bearing trees, termite mounds or birds nests, will be isolated for a period of 24 hrs to encourage dispersal prior to being removed. Techniques applicable to this stage of spotter-catcher duties vary due to the site specifics regarding topography, structure and stature of trees and OHS limitations.
- Hollow-bearing trees will be accessed and examined with torch, chainsaw, buffer rags with all fauna located during spotter-catcher duties to be assessed for species, injury and maturity prior to immediate release or being placed in a cotton capture bag

Captured fauna will be held in suitable ambient conditions prior to release within normal activity times for the animal concerned. All fauna captured will be released in adjacent bushland, containing suitable habitat, outside and within 500 m of the proposed clearing zone.

Insect bat species located during clearing will be identified prior to release at an appropriate crepuscular time with Anabat recording of calls.

• Any diseased, injured or juvenile fauna located that is incapable of dispersal or release will be taken to an experienced wildlife vet, Wildcare or Currumbin Wildlife Sanctuary for treatment or fostering prior to release.

# 5.1.2 SUCCESSIONAL CLEARING PROTOCOL

As koalas have been previously recorded on the site sequential clearing is recommended despite the unlikelihood of any koalas being present within the clearing zones. In this regard clearing of trees is carried out in a way that ensures koalas (and other animals) living in or near the area being cleared (the clearing site) have enough time to move out of the clearing site without human intervention by:

Clearing of trees is carried out in a way that ensures, while the clearing is being carried out, appropriate habitat links are maintained within the clearing site and between the site and its adjacent areas, to allow koalas/fauna living on the site to move out of the site

In this regard it is recommended that clearing be undertaken generally in a south-north (northern sand extraction area) & west-east (southern sand extraction area) direction to enable dispersal to retained bushland. Importantly clearing must not disperse fauna to the south or the west where they may become trapped or disperse into adjacent construction sites and roadways]

No tree in which fauna is present, and no tree with a crown overlapping a tree in which fauna is present, is cleared. A fauna spotter means a person who has demonstrated experience in locating koalas in koala habitats or conducting fauna surveys. Prior to the commencement of, and during felling operations, it is the responsibility of the spotter to identify trees in which a koala or other fauna are present and any trees where



their crown overlaps trees in which a koala or other fauna is present and convey this information to the person(s) conducting the clearing.

During this identification process, there will a differentiation between *Complete cover trees* and *See-through trees* where *Complete cover trees* will be retained until the spotter-catcher responsible is certain of absence of fauna prior to the tree being felled.

A *Complete cover tree* is defined as a tree with abundant foliage that does not allow confirmation of absence of fauna without a full 360 degree viewing and if required extended viewing during peripheral clearing operations to detect movement.

A *See-through tree* is defined as sparsely foliated tree where a 360 degree viewing confirms the absence of fauna.

Any tree in which fauna is located (and any adjacent overlapping tree) is to be tagged and excluded from clearing until the fauna has safely dispersed (over the period, to be a minimum of 12 hours, between cessation of works on that day and commencement of works on the subsequent day)

When located, clearing exclusion zones (20m minimum) around the active fauna tree will be set out where no activity can occur for the day's duration to confirm animal safety and allow dispersal. Secondly, fauna response to peripheral human activity will be monitored by spotter-catcher to confirm acceptable disturbance and if required cessation of clearing process within a larger radius (i.e. 50m) to reduce stress to the observed fauna.

The site contractor/foreman is to be clearly shown the identification of all trees and exclusions zones which are not permitted for any day's clearing prior to starting the first piece of construction equipment on that day of clearing.

Opportunistic capture/release is only permitted where an *Un-located Koala/Fauna* interaction occurs. If or when this occurs, all clearing will cease until the spotter-catcher responsible has determined the severity of incident. In circumstances where the animal has had a major stress reaction from close proximity to the clearing process, clearing will be directed away from proximity and the animal's reaction monitored until stress reaction is deemed acceptable. In circumstances where the tree containing the animal has been felled, it will be examined for general alertness, potential injuries to paws, limbs, body, vision, wings, claws, etc prior to the decision to vet check the animal or hold with observations repeated prior to release and post release monitoring to confirm successful dispersal out of the clearing zone.

An *Un-located Koala/Fauna* is defined as a Koala or other animal that has eluded detection, despite the defined actions of the spotter-catcher responsible occurring (as outlined within the preceding points) and is generally associated with dense vegetation areas or canopies exhibiting dense coppicing.

# 5.1.3 ONGOING FAUNA MANAGEMENT DURING QUARRY OPERATIONS

Although the quarrying use is well separated from retained fauna habitat areas potential does existing for fauna injury/mortality during ongoing operation of the use as follows:



• Potential occurs for vehicle strike along the haulage road, particularly in the northern areas proximate to koala habitat.

The following actions are recommended to minimize the above listed potential impacts:

- In accordance with the Development Consent issued, trucks and machinery are to be confined to defined haulage routes and operate during daylight hours only (7am-5pm weekdays, 7am-12pm weekends) at a maximum speed of 25km/hr (as signed within the site).
- A registered wildlife spotter catcher is to be contacted regarding the presence of any fauna trapped or injured within the quarry zone (including snakes). Any such fauna is to be inspected for species, injury and maturity prior to immediate release in adjacent bushland (depending on time of day) or being placed in a cotton capture bag.

Captured fauna will be held in suitable ambient conditions prior to release within normal activity times for the animal concerned. All fauna captured will be released in adjacent bushland, containing suitable habitat, outside but within 500m of the quarry zone.

 Any diseased, injured or juvenile fauna capture that is incapable of dispersal or release will be taken to an experienced wildlife vet, Wildcare or Currumbin Wildlife Sanctuary for treatment or fostering prior to release.

# 5.2 FAUNA NESTBOXES

In accordance with the Development Consent issued fauna boxes are to be installed onsite to increase potential nesting options for the existing assemblage. To potentially allow for a wider range of species to utilize the fauna boxes to be provided it is recommended that boxes be designed in accordance with a variety of specifications. In particular, the following is to be provided targeting a range of arboreal mammals and avifauna:

- 2 x squirrel/sugar glider boxes
- 2 x cockatoo/parrot boxes
- 2 brushtail/ringtail possum boxes
- 1 rosella/lorikeet boxes
- 1 microbat boxes
- 1 owl box
- 1 kingfisher boxes

The purchase, installation and monitoring of the above fauna boxes is to comply with the below requirements:

- Fauna boxes are only to be sourced from a reputable supplier
- Boxes are to be well-insulated, rainproof and facing away from prevailing winds and direct midday summer sunlight



- Boxes are to be located where casual access is difficult to reduce potential vandalism
- Boxes are to be maintained and inspected on an annually basis

# Fauna Box Reporting

Within six months of the commencement of the use a brief report is to be prepared containing the following information:

- An accurate graphical representation of each fauna box within the retained vegetation communities and its associated GPS coordinates
- Detail of the type, number and location of each of the 10 fauna boxes
- The installed height of the box from the ground

Every twelve months an inspection and a brief report shall be prepared/undertaken for each of the boxes including:

- A discussion of the maintenance of the fauna boxes over the preceding quarterly period and any associated problems encountered (i.e. vandalism, presence of feral insects, birds or mammals etc) and any resultant corrective actions implemented
- Documentation of any changes to the location of fauna boxes as a result of corrective actions implemented
- Completion of the below monitoring form



ROUTINE FAUNA BOX MONITORING FORM			
Location/Number of Fauna Box			
Description of Fauna Box			
Inspected by	Name	Name: Signature:	
Inspection date		/ /	
Element	N/Y?	Comments/description	Action Required
Empty Box			
Native Fauna Present			
Eggs Present			
Number of eggs			
Colour/description of eggs			
Nest present			
Partial nest present			
Hatchlings/fledglings present			
Box empty but scats/trace present			
Box occupied by pest species (i.e. bees, myna, black rat etc)			
Roof, hinges and/or supports broken or in need of repair/replacement			
Evidence of warping			
Evidence of vandalism			
Other comments/maintenance perform	ed?:		



# 5.3 ANNUAL FAUNA SURVEY

In accordance with the previous commitments of the Planit (2006) Flora and Fauna Assessment an annual fauna survey (between November and March) of the Rehabilitation Zones (and immediately adjacent areas) is to be undertaken in accordance with Section 4 of the 2006 report. Results of the survey are to be recorded including new species (to that originally recorded) and discussions of any threatened fauna recorded. Particular mention is to be made of any threatened fauna utilizing the Rehabilitation Zones or installed fauna boxes.

# 6.0 ATTACHMENTS

ATTACHMENT 1:APPROVED PLAN OF DEVELOPMENTATTACHMENT 2:DEVELOPMENT CONSENT CONDITIONSATTACHMENT 3:LANDSCAPING PLANSATTACHMENT 4:VEGETATION MANAGEMENT PLAN



# ATTACHMENT 1 – APPROVED PLAN OF DEVELOPMENT







# ATTACHMENT 2 – DEVELOPMENT CONSENT OF CONDITIONS

# Project Approval

Section 75J of the Environmental Planning and Assessment Act 1979

I approve the project application referred to in schedule 1, subject to the conditions in schedules 2 to 5.

The reason for these conditions is to:

- prevent, minimise, and/or offset adverse environmental impacts;
- set standards and performance measures for acceptable environmental performance;
- require regular monitoring and reporting; and

2 4 NOV 2009

provide for the on-going environmental management of the project.

The Hon Kristina Keneally MP Minister for Planning

Sydney 2008 SCHEDULE 1 **Project Application:** 06\_0030 Proponent: Ramtech Pty Ltd **Approval Authority:** Minister for Planning Land: Lot 1 DP208249; Lot 182 DP755721; Lot 183 DP755721; Lot 44 DP755721; Lot 81 DP755721; Lot 162 DP755721; Lot 2 DP 780199; Lot 1 DP780199; Lot 1 DP780200; Lot 2 DP785895 Project: **Dunice Park Sand Project** 

TABLE	OF	CONT	ENTS
	•••		

DEFINITIONS	3
SCHEDULE 2 ADMINISTRATIVE CONDITIONS	4
Obligation to Minimise Harm to the Environment Terms of Approval Limits on Approval Management Plans/Monitoring Programs Structural Adequacy Demolition Operation of Plant and Equipment Section 94 Contributions	4 4 4 4 5 5
SCHEDULE 3 ENVIRONMENTAL PERFORMANCE CONDITIONS	6
General Extraction and Processing Provisions Noise Air Quality Soil and Water Rehabilitation and Landscaping Aboriginal Cultural Heritage Traffic and Transportation Visual Impact Waste Management Emergency and Hazards Management Production Data	6 6 7 8 10 12 12 12 13 13 13
SCHEDULE 4 ADDITIONAL PROCEDURES	14
Notification of Landowners Independent Review	14 14
SCHEDULE 5 ENVIRONMENTAL MANAGEMENT AND MONITORING CONDITIONS	15
Environmental Management Strategy Environmental Monitoring Program Incident Reporting Annual Report Independent Environmental Audit Community Consultative Committee Access to Information	15 15 15 16 16 16
APPENDIX 1: PROJECT SITE APPENDIX 2: REHABILITATION STAGING PLAN APPENDIX 3: STATEMENT OF COMMITMENTS APPENDIX 4: INDEPENDENT DISPUTE RESOLUTION PROCESS	17 18 19 23

AEMR BCA CCC Council Day

DECC Department Director-General DPI DWE EA

EP&A Act EP&A Regulation EPL

Land

Minister Privately-owned land Project Proponent RTA SEPP Statement of Commitments Site

#### DEFINITIONS

Annual Environmental Management Report Building Code of Australia **Community Consultative Committee** Tweed Shire Council The period from 7.00am to 5.00pm, Monday to Friday, and 8.00am to 1.00pm on Saturdays Department of Environment and Climate Change Department of Planning Director-General of the Department of Planning, or delegate Department of Primary Industries Department of Water and Energy Environmental Assessment of the project titled Environmental Assessment Part 3A – Environmental Planning and Assessment Act, 1979, Dunloe Sands - Ramtech P/L, prepared by PLANIT Consulting dated 17 September 2007, including the response to issues raised in submissions, dated March 2008 Environmental Planning and Assessment Act 1979 Environmental Planning and Assessment Regulation 2000 Environment Protection Licence under the Protection of the Environment Operations Act 1997 (POEO Act) Land means the whole of a lot, or contiguous lots owned by the same landowner, in a current plan registered at the Land Titles Office at the date of this approval Minister for Planning, or delegate Land that is not owned by a public agency or a quarrying company The development as described in the EA Ramtech Pty Ltd, or its successors in title Roads and Traffic Authority State Environmental Planning Policy The Proponent's commitments in Appendix 3 Land to which the Project Approval applies

#### SCHEDULE 2 ADMINISTRATIVE CONDITIONS

#### **Obligation to Minimise Harm to the Environment**

1. The Proponent shall implement all practicable measures to prevent and/or minimise any harm to the environment that may result from the construction, operation, or rehabilitation of the project.

#### **Terms of Approval**

2.

- The Proponent shall carry out the project generally in accordance with the:
  - (a) EA;
  - (b) statement of commitments; and
  - (c) conditions of this approval.

Note: The general layout of the project is shown in the figure in Appendix 1.

- 3. If there is any inconsistency between the documents referred to above, the conditions of this approval shall prevail to the extent of the inconsistency.
- 4. The Proponent shall comply with any reasonable requirement/s of the Director-General arising from the Department's assessment of:
  - (a) any reports, plans, programs or correspondence that are submitted in accordance with this approval; and
  - (b) the implementation of any actions or measures contained in these reports, plans, programs or correspondence.

#### Limits on Approval

5. Sand extraction operations may take place until 1 January 2035.

Note: Under this Approval, the Proponent is required to rehabilitate and revegetate the site to the satisfaction of the Director-General. Consequently this approval will continue to apply in all other respects other than the right to conduct quarrying operations until the site has been rehabilitated and revegetated to a satisfactory standard.

6. The Proponent shall not extract to a depth of more than 12 m below the natural ground surface.

Notes: The Department acknowledges that this limit may have a variance of +/- 1m.

- 7. The Proponent shall not transport more than 300,000 tonnes of sand material a year from the site.
- 8. The Proponent shall ensure that heavy vehicle movements (in and out) associated with the project do not exceed 8 per hour.

#### Management Plans/Monitoring Programs

9. With the approval of the Director-General, the Proponent may submit any management plan or monitoring program required by this approval on a progressive basis.

#### **Structural Adequacy**

 The Proponent shall ensure that any new buildings and structures, and any alterations or additions to existing buildings and structures, are constructed in accordance with the relevant requirements of the BCA.

Notes:

- Under Part 4A of the EP&A Act, the Proponent is required to obtain construction and occupation certificates for the proposed building works.
- Part 8 of the EP&A Regulation sets out the requirements for the certification of the project.

#### Demolition

11. The Proponent shall ensure that all demolition work is carried out in accordance with AS 2601-2001: *The Demolition of Structures*, or its latest version.

#### **Operation of Plant and Equipment**

- 12. The Proponent shall ensure that all plant and equipment used at the site is:
  - (a) maintained in a proper and efficient condition; and
  - (b) operated in a proper and efficient manner.

#### **Section 94 Contributions**

13. Prior to carrying out any development, or as otherwise agreed by Council, the Proponent shall pay Council \$47,250 in accordance with Council's Tweed Road Contribution Plan and \$399.40 in accordance with Tweed Council Section 94 Plan No.18.

#### SCHEDULE 3 ENVIRONMENTAL PERFORMANCE CONDITIONS

#### GENERAL EXTRACTION AND PROCESSING PROVISIONS

#### Identification of Boundaries

- 1. Within 1 month of the date of approval of the Landscape Management Plan (see condition 27 below), the Proponent shall:
  - (a) engage a registered surveyor to mark out the boundaries of the approved limits of extraction;
  - (b) submit a survey plan of these boundaries to the Director-General; and
  - (c) ensure that these boundaries are clearly marked at all times in a permanent manner that allows operating staff and inspecting officers to clearly identify those limits.

Note: The limit of extraction includes the area described in the EA and shown conceptually on the plan in Appendix 1.

#### NOISE

#### Impact Assessment Criteria

2. The Proponent shall ensure that the noise generated by the project does not exceed the noise impact assessment criteria in Table 1.

Receiver Location	Day L <sub>Aeq (15 min)</sub> dB(A)
Residences on privately-owned land	48

Table 1: Noise Impact Assessment Criteria

Notes:

- Noise from the site is to be measured at the most affected point within the residential boundary, or at the most
  affected point within 30 metres of the dwelling where the dwelling is more than 30 metres from the boundary,
  to determine compliance with the identified noise limits, except where otherwise specified below.
- Where it can be demonstrated that direct measurement of noise from the project is impractical, alternative means of determining compliance may be acceptable (see Chapter 11 of the NSW Industrial Noise Policy).
- The modification factors presented in Section 4 of the NSW Industrial Noise Policy shall also be applied to the measured noise level where applicable.
- The identified noise emission limits apply under meteorological conditions of wind speed up to 3m/s at 10 metres above ground level, and temperature inversion conditions.

#### **Hours of Operation**

3. The Proponent shall comply with the operating hours in Table 2.

Activity	Day	Time
Sand extraction and	Monday – Friday	7:00am to 5:00pm
processing, delivery and distribution, and other quarry	Saturday	7:00am to 12:00pm
related activities	Sunday and Public Holidays	Nil
Maintenance (if inaudible at neighbouring residences)	Any day	Any time

Table 2: Operating Hours

#### **Continuous Improvement**

- 4. The Proponent shall:
  - (a) implement all reasonable and feasible best practice noise mitigation measures;
  - (b) investigate ways to reduce the noise generated by the project; and

(c) report on these investigations and the implementation and effectiveness of these measures in the AEMR,

to the satisfaction of the Director-General.

#### **Noise Monitoring Program**

- 5. The Proponent shall prepare and implement a Noise Monitoring Program for the project to the satisfaction of the Director-General. This program must:
  - (a) be submitted to the Director-General prior to carrying out any development on the site;
  - (b) be prepared in consultation with the DECC;
  - (c) include details of how the noise performance of the project would be monitored, and include a noise monitoring protocol for evaluating compliance with the relevant noise limits in this approval; and
  - (d) include a protocol of monitoring haulage truck noise on Pottsville Road.

#### **AIR QUALITY**

#### Impact Assessment Criteria

6. The Proponent shall ensure that dust generated by the project does not cause additional exceedances of the criteria listed in Tables 3 to 5 at any privately owned land.

Pollutant	Averaging period	Criterion
Particulate matter < 10 μm (PM <sub>10</sub> )	24 hour	50 μg/m <sup>3</sup>

#### Table 3: Short Term Impact Assessment Criteria for Particulate Matter

Pollutant	Averaging period	Criterion
Total suspended particulate (TSP) matter	Annual	90 μg/m <sup>3</sup>
Particulate matter < 10 μm (PM <sub>10</sub> )	Annual	30 μg/m <sup>3</sup>

Table 4: Long Term Impact Assessment Criteria for Particulate Matter

Pollutant	Averaging period	Maximum increase in deposited dust level	Maximum total deposited dust level
Deposited dust	Annual	2 g/m <sup>2</sup> /month	4 g/m <sup>2</sup> /month

Table 5: Long Term Impact Assessment Criteria for Deposited Dust

Note: Deposited dust is assessed as insoluble solids as defined by Standards Australia, 1991, AS/NZS 3580.10.1-2003: Methods for Sampling and Analysis of Ambient Air - Determination of Particulates - Deposited Matter - Gravimetric Method.

#### Dust Monitoring Program

- 7. The Proponent shall prepare and implement a Dust Monitoring Program for the project to the satisfaction of the Director-General. This program must:
  - (a) be submitted to the Director-General prior to carrying out any development on the site;
  - (b) be prepared in consultation with DECC; and

(c) include details of how the air quality performance of the project would be monitored, and include a protocol for evaluating compliance with the relevant air quality criteria in this approval.

Note: Initially, this program may concentrate on monitoring the dust deposition impacts of the project. However, in time, it may be expanded to include other pollutants.

#### SOIL AND WATER

#### Discharges

8. Except as may be expressly provided for by an EPL, the Proponent shall not discharge any water from the project or ancillary operational areas. The Proponent shall ensure that the extraction pit subject to dredging is maintained and operated to prevent discharges of any surface water from these ponds.

#### Water Quality Objectives

9. The Proponent shall aim to meet the water quality objectives in Table 6 for water in the dredge ponds and in groundwater adjacent the dredge ponds, unless otherwise approved by the Director-General.

Pollutant	Unit of Measure	Water Quality Objectives
Turbidity	NTU	5 - 20
рН	рН	6.5 – 8.5
Oil and Grease	mg/L	10
Salinity	μS/cm	<3,000
Dissolved oxygen	mg/L	>6
Chorophyll-a	μg/L	2-10
Faecal coliforms	Median No./100mL	<1000
Enterococci	Median No./100mL	<230
Alace and blue groop alace	No.cells/mL (M.aeruginosa)	<50,000
Aigae and blue-green aigae	mm <sup>3</sup> /L (total biovolume)	<4
Sodium	mg/L	500
Potassium ion	mg/L	40
Magnesium ion	mg/L	100
Chloride ion	mg/L	1000
Sulphate ion	mg/L	800
Bicarbonate ion	mg/L	400
Soluble ion	mg/L	20
Soluable aluminium ion	mg/L	0.5
Ammonium ion	mg/L	20

#### Table 6: Water Quality Objectives

Notes:

- The objectives for dissolved oxygen, turbidity and algae are relevant to surface water only.
- The Department acknowledges that short term exceedances of these objectives may occur during natural events such as flooding.
- The Department acknowledges that pre-existing water quality may not meet the objectives for some analytes, including salinity. The proponent shall strive to meet the water quality objectives through implementation of the Soil and Water Management Plan (see condition 18 below), as far as is reasonable and feasible and within the Proponent's control, to the satisfaction of the Director-General.

#### **Fines Management**

- 10. The Proponent shall ensure that all excavated potential acid sulfate soil fines material is returned back to below the watertable as soon as possible to prevent oxidation. No potential acid sulfate soil shall be removed from the site, unless adequately neutralised in accordance with methods approved under the Soil and Water Management Plan.
- 11. The Proponent shall ensure that all potential acid sulfate soil fines material is discharged into the pond at a depth of no less than 3 metres from the water surface, and that all fines are deposited to a final depth of at least 8 metres from the water surface, unless an alternative method(s) is approved by DWE and the Director-General.

#### Wastewater Treatment

12. The Proponent shall manage on-site sewage to the satisfaction of Council and DECC. The facility must comply with the requirements of the *Environment and Health Protection Guidelines – On-site Sewage Management for Single Households (1998).* 

#### **Flood Management**

- 13. The Proponent shall ensure that flood bunding around the Stage 1 and Stage 2 works does not exceed 300 mm in height above natural surface level, to a maximum height of 2.0 m AHD, unless otherwise approved by the Director-General.
- 14. The Proponent shall ensure that perimeter drainage must be installed and operational prior to the construction of bunding or the placement of fill on site.
- 15. All earthworks, including flood and acoustic bunding works, shall be contained wholly within the site.
- 16. The Proponent shall cease dredging and processing activities not less than 24 hours prior to the commencement of overflow from any dredge pond. No dredging or processing shall occur when the dredge ponds are overflowing.
- 17. The Proponent shall ensure that the flood storage capacity of the site is no less than the pre-existing flood storage capacity at all stages of the project. Details of the available flood storage capacity shall be reported in the AEMR.

#### Management and Monitoring

- 18. The Proponent shall prepare and implement a Soil and Water Management Plan for the project to the satisfaction of the Director-General. This plan must:
  - (a) be prepared in consultation with DWE and DECC;
  - (b) include a:
    - Water Balance;
      - Erosion and Sediment Control Plan;
      - Acid Sulfate Soil Management Plan;
      - Blue-Green Algae Management Plan;
      - Surface Water Monitoring Program; and
      - Groundwater Monitoring Program; and
  - (c) be submitted to the Director-General prior to starting quarrying operations, and prior to carrying out any development site in the case of the Erosion and Sediment Control Plan.
- 19. The Water Balance shall include:
  - (a) details of all water extracted, transferred, used and/or discharged by the quarry;
  - (b) the source of all water collected or stored on the site, including rainfall, stormwater and groundwater; and
  - (c) measures to minimise water use by the project.
- 20. The Erosion and Sediment Control Plan shall:
  - (a) be consistent with the requirements of *Managing Urban Stormwater: Soils and Construction, Volume 1, 4<sup>th</sup> Edition, 2004* (Landcom), and Council's codes including its *Code of Practice for*

Soil and Water Management on Construction Sites, Development Design Specification D7 – Stormwater Quality and Tweed Urban Stormwater Quality Management Plan;

- (b) identify activities that could cause soil erosion and generate sediment;
- (c) describe measures to minimise soil erosion and the potential for the transport of sediment to downstream waters;
- (d) describe the location, function, and capacity of erosion and sediment control structures; and
- (e) describe what measures would be implemented to maintain these structures over time.
- 21. The Acid Sulfate Soil Management Plan shall:
  - (a) be consistent with the NSW Acid Sulphate Soil Advisory Committee's Acid Sulfate Soil Manual; and
  - (b) define procedures for managing the potential acid sulfate soils on the site, including sample testing and procedures.
- 22. The Blue-Green Algae Management Plan shall:
  - (a) be prepared by a suitably qualified blue-green algae expert, whose appointment has been approved by the Director-General;
  - (b) be consistent with extant guidelines for blue-green algae management including the NHMRC's *Guidelines for Managing Risks in Recreational Water*,
  - (c) describe the measures that would be implemented to prevent and control the sources of algal blooms over the short, medium and long term; and
  - (d) define procedures for the management and notification of identified algal blooms.
- 23. The Surface Water Monitoring Program shall include:
  - (a) detailed baseline data on surface water quality;
  - (b) surface water impact assessment criteria;
  - (c) a program to monitor surface water flows and quality;
  - (d) a program to manage water releases from the site;
  - (e) a program to monitor bank and bed stability; and
  - (f) a protocol for the investigation, notification and mitigation of identified exceedances of the surface water impact assessment criteria.
- 24. The Ground Water Monitoring Program shall include:
  - (a) detailed baseline data on groundwater levels and quality, based on statistical analysis;
  - (b) groundwater impact assessment criteria;
  - (c) a program to monitor ground water levels and quality;
  - (d) a program to monitor ground water level effects on vegetation, and on ground water supply to adjoining properties; and
  - (e) a protocol for the investigation, notification and mitigation of identified exceedances of the groundwater impact assessment criteria.

#### **REHABILITATION AND LANDSCAPING**

#### Rehabilitation and Revegetation

- 25. The Proponent shall rehabilitate the site to the satisfaction of the Director-General.
- 26. The Proponent shall:
  - (a) rehabilitate and revegetate the 15 ha hectares of land identified in the EA (see the revegetation plan in Appendix 2); and
  - (b) within 12 months of the commencement of quarrying operations, make suitable arrangements to provide appropriate long term security for the revegetation area to ensure it is managed for conservation purposes,
  - to the satisfaction of the Director-General.

#### Landscape Management Plan

- 27. The Proponent shall prepare and implement a Landscape Management Plan for the project to the satisfaction of the Director-General. This plan must:
  - (a) be prepared:
    - by suitably qualified consultants, including a specialist hydrologist, coastal engineer, wetlands ecologist and landscape architect;

- in consultation with Council, DWE, DECC, DPI-Fisheries and the CCC; and
- in accordance with extant guidelines including the DWE's *Constructed Wetlands Manual*, *Volumes 1 and 2* and the DPI's *Policy and Guidelines: Aquatic Habitat Management*, 1999;
- (b) be submitted to the Director-General prior to starting quarrying operations on the site; and
   (c) include a:
  - Rehabilitation and Revegetation Management Plan; and
    - Long Term Management Strategy.

Note: The Department accepts that the initial Landscape Management Plan may not include the detailed Long Term Management Strategy. However, a conceptual strategy must be included in the initial plan, along with a timetable for augmentation of the strategy with each subsequent review of the plan.

- 28. The Rehabilitation and Revegetation Management Plan must include:
  - (a) the rehabilitation objectives for the site and revegetation areas;
  - (b) a description of the short, medium, and long term measures that would be implemented to:
    - rehabilitate and stabilise the site;
    - implement the revegetation strategy; and
    - manage the remnant vegetation and habitat on the site and in the revegetation areas;
  - (c) detailed performance and completion criteria for the rehabilitation and stabilisation of the site and implementation of the revegetation strategy;
  - (d) a detailed description of how the performance of the rehabilitation of the quarry and the revegetation areas would be monitored over time to achieve the stated objectives;
  - (e) a detailed description of what measures would be implemented over the next 5 years to rehabilitate and manage the landscape of the site and revegetation areas including the procedures to be implemented for:
    - progressively rehabilitating and stabilising areas disturbed by quarrying;
    - implementing revegetation and regeneration within the disturbance areas and revegetation areas;
    - protecting areas outside the disturbance areas, including SEPP 14 wetlands and SEPP 26 littoral rainforests;
    - vegetation clearing protocols;
    - managing impacts on fauna;
    - controlling terrestrial and aquatic weeds and pests;
    - controlling access;
    - bushfire management; and
    - reducing the visual impacts of the project;
  - a description of the potential risks to successful rehabilitation and/or revegetation, and a description of the contingency measures that would be implemented to mitigate these risks; and
  - (g) details of who is responsible for monitoring, reviewing, and implementing the plan.
- 29. The Long Term Management Strategy must:
  - (a) define the objectives and criteria for quarry closure and post-extraction management;
  - (b) investigate options for the future use of the site;
  - (c) describe the measures that would be implemented to minimise or manage the ongoing environmental effects of the project; and
  - (d) describe how the performance of these measures would be monitored over time.

#### **Rehabilitation Bond**

- 30. Prior to starting quarrying operations on the site, the Proponent shall lodge a rehabilitation bond for the project with the Director-General. The sum of the bond shall be calculated at:
  - (a) \$2.50/m<sup>2</sup> for the total area to be disturbed and/or revegetated in each 5 year review period (see condition 31 below); and
  - (b) \$1.50/m<sup>2</sup> for the total area of land previously disturbed and/or rehabilitated by the project,
  - to the satisfaction of the Director-General.

Notes:

• If the rehabilitation and revegetation works are completed to the satisfaction of the Director-General, the Director-General will release the rehabilitation bond.

- If the rehabilitation and revegetation works are not completed to the satisfaction of the Director-General, the Director-General will call in all or part of the rehabilitation bond, and arrange for the satisfactory completion of the relevant works.
- 31. Within 6 months of each Independent Environmental Audit (see condition 6 of schedule 5) excluding the inaugural audit, unless the Director-General directs otherwise, the Proponent shall review, and if necessary revise, the sum of the rehabilitation bond to the satisfaction of the Director-General. This review must consider:
  - (a) the effects of inflation;
  - (b) any changes to the total area of disturbance; and
  - (c) the performance of the rehabilitation and revegetation to date.

#### **ABORIGINAL CULTURAL HERITAGE**

#### Aboriginal Cultural Heritage Management Plan

- 32. The Proponent shall prepare and implement an Aboriginal Cultural Heritage Management Plan to the satisfaction of the Director-General. This plan must:
  - (a) be prepared in consultation with DECC and all relevant Aboriginal communities;
  - (b) be submitted to the Director-General for approval prior to commencement of construction; and
  - (c) include a:
    - program for additional archaeological survey/s of the disturbance area;
    - description of the measures that would be implemented to salvage any identified Aboriginal sites within the disturbance area;
    - description of the measures that would be implemented to protect any Aboriginal sites outside the disturbance area; and
    - description of the measures that would be implemented if any new Aboriginal objects or skeletal remains are discovered during the project.

#### TRAFFIC AND TRANSPORTATION

#### **Road Haulage**

- 33. Prior to commencement of operations the Proponent shall:
  - (a) design and construct the haul road and its intersection with Pottsville-Mooball Road; and
     (b) install advanced truck turning warning signage along Pottsville-Mooball Road, to the satisfaction of Council.
- 34. The Proponent shall ensure that all loaded vehicles entering or leaving the site have their loads covered.
- 35. The Proponent shall ensure all loaded vehicles leaving the site are cleaned of materials that may fall on the road before they are allowed to leave the site.

#### Parking

36. The Proponent shall provide sufficient parking on-site for all project-related traffic and visitors, in accordance with Council's parking codes and to the satisfaction of the Director-General. No on street parking shall be undertaken.

#### **VISUAL IMPACT**

#### Visual Amenity

- 37. The Proponent shall minimise the visual impacts of the project to the satisfaction of the Director-General.
- 38. The Proponent shall establish and subsequently maintain the vegetated buffer around the extraction area within 12 months of the date of this approval.

Note: The vegetation buffer shall be detailed in the Landscape Management Plan.

#### Lighting Emissions

#### 39. The Proponent shall:

- (a) take all practicable measures to mitigate off-site lighting impacts from the project; and
- (b) ensure that all external lighting associated with the project complies with Australian Standard AS4282 (INT) 1995 Control of Obtrusive Effects of Outdoor Lighting,
- to the satisfaction of the Director-General.

#### Advertising

40. The Proponent shall not erect or display any advertising structure(s) or signs on the site without the written approval of the Director-General.

Note: This does not include business identification, traffic management and safety or environmental signs.

#### WASTE MANAGEMENT

41. The Proponent shall minimise the amount of waste generated by the project to the satisfaction of the Director-General.

#### EMERGENCY AND HAZARDS MANAGEMENT

#### **Dangerous Goods**

42. The Proponent shall ensure that the storage, handling, and transport of dangerous goods are conducted in accordance with the relevant *Australian Standards*, particularly AS1940 and AS1596, and the *Dangerous Goods Code*.

#### Safety

43. The Proponent shall secure the project to ensure public safety to the satisfaction of the Director-General.

#### **Bushfire Management**

- 44. The Proponent shall:
  - (a) ensure that the project is suitably equipped to respond to any fires on-site; and
  - (b) assist the rural fire service and emergency services as much as possible if there is a fire onsite.

#### **PRODUCTION DATA**

- 45. The Proponent shall:
  - (a) provide annual production date to the DPI using the standard form for that purpose; and
  - (b) include a copy of this data in the AEMR.

#### SCHEDULE 4 ADDITIONAL PROCEDURES

#### NOTIFICATION OF LANDOWNERS

1. If the results of monitoring required in schedule 3 identify that impacts generated by the project are greater than the relevant impact assessment criteria, then the Proponent shall notify the Director-General, affected landowners, and/or existing or future tenants accordingly, and provide quarterly monitoring results to each of these parties until the results show that the project is complying with the relevant criteria.

#### INDEPENDENT REVIEW

2. If a landowner considers that the project is exceeding the impact assessment criteria in schedule 3, then he/she may ask the Director-General in writing for an independent review of the impacts of the project on his/her land.

If the Director-General is satisfied that an independent review is warranted, the Proponent shall within 3 months of the Director-General advising that an independent review is warranted:

- (a) consult with the landowner to determine his/her concerns;
- (b) commission a suitably qualified, experienced and independent person, whose appointment has been approved by the Director-General, to conduct monitoring on the land, to determine whether the project is complying with the relevant criteria in schedule 3, and identify the source(s) and scale of any impact on the land, and the project's contribution to this impact; and
- (c) give the Director-General and landowner a copy of the independent review.
- 3. If the independent review determines that the project is complying with the relevant criteria in schedule 3, then the Proponent may discontinue the independent review with the approval of the Director-General.
- 4. If the independent review determines that the project is not complying with the relevant criteria in schedule 3, and that the project is primarily responsible for this non-compliance, then the Proponent shall:
  - (a) implement all reasonable and feasible measures, in consultation with the landowner, to ensure that the project complies with the relevant criteria; and
  - (b) conduct further monitoring to determine whether these measures ensure compliance; or
  - (c) secure a written agreement with the landowner to allow exceedances of the relevant criteria in schedule 3,

to the satisfaction of the Director-General.

If the additional monitoring referred to above subsequently determines that the project is complying with the relevant criteria in schedule 3, or the Proponent and landowner enter into a negotiated agreement to allow these exceedances, then the Proponent may discontinue the independent review with the approval of the Director-General.

5. If the landowner disputes the results of the independent review, either the Proponent or the landowner may refer the matter to the Director-General for resolution.

If the matter cannot be resolved within 21 days, the Director-General shall refer the matter to an Independent Dispute Resolution Process (see Appendix 4).

#### SCHEDULE 5 ENVIRONMENTAL MANAGEMENT AND MONITORING CONDITIONS

#### ENVIRONMENTAL MANAGEMENT STRATEGY

- 1. The Proponent shall prepare and implement an Environmental Management Strategy for the project to the satisfaction of the Director-General. This strategy must:
  - (a) be submitted to the Director-General prior to starting quarrying operations on the site;
  - (b) be prepared in consultation with the relevant agencies;
  - (c) provide the strategic context for environmental management of the project;
  - (d) identify the statutory requirements that apply to the project;
  - (e) describe in general how the environmental performance of the project would be monitored and managed;
  - (f) describe the procedures that would be implemented to:
    - keep the local community and relevant agencies informed about the construction, operation and environmental performance of the project;
    - receive, handle, respond to, and record complaints;
    - resolve any disputes that may arise during the life of the project;
    - respond to any non-compliance;
    - manage cumulative impacts; and
    - respond to emergencies; and
  - (e) describe the role, responsibility, authority, and accountability of the key personnel involved in the environmental management of the project.

#### ENVIRONMENTAL MONITORING PROGRAM

2. The Proponent shall prepare an Environmental Monitoring Program for the project to the satisfaction of the Director-General. This program must be submitted to the Director-General prior to starting quarrying operations on the site, and consolidate the various monitoring requirements in schedule 3 of this approval into a single document.

#### **INCIDENT REPORTING**

- 3. Within 24 hours of detecting an exceedance of the limits/performance criteria in this approval or the occurrence of an incident that causes (or may cause) material harm to the environment, the Proponent shall notify the Department and other relevant agencies of the exceedance/incident.
- 4. Within 6 days of notifying the Department and other relevant agencies of an exceedance/incident, the Proponent shall provide the Department and these agencies with a written report that:
  - (a) describes the date, time, and nature of the exceedance/incident;
  - (b) identifies the cause (or likely cause ) of the exceedance/incident;
  - (c) describes what action has been taken to date; and
  - (d) describes the proposed measures to address the exceedance/incident.

#### ANNUAL REPORTING

- 5. Within 12 months of the date of this approval, and annually thereafter, the Proponent shall submit an AEMR to the Director-General and relevant agencies. This report must:
  - (a) identify the standards and performance measures that apply to the project;
  - (b) describe the works carried out in the last 12 months;
  - (c) describe the works that will be carried out in the next 12 months;
  - (d) include a summary of the complaints received during the past year, and compare this to the complaints received in previous years;
  - (e) include a summary of the monitoring results for the project during the past year;
  - (f) include an analysis of these monitoring results against the relevant:
    - impact assessment criteria/limits;
    - monitoring results from previous years; and
    - predictions in the EA;
  - (g) identify any trends in the monitoring results over the life of the project;
  - (h) identify any non-compliance during the previous year; and
  - (i) describe what actions were, or are being, taken to ensure compliance.

#### INDEPENDENT ENVIRONMENTAL AUDIT

- 6. Within 2 years of the start of quarrying operations on site, and every 5 years thereafter, unless the Director-General directs otherwise, the Proponent shall commission and pay the full cost of an Independent Environmental Audit of the project. This audit must:
  - (a) be conducted by a suitably qualified, experienced, and independent person(s) whose appointment has been approved by the Director-General;
  - (b) include consultation with the relevant agencies;
  - (c) assess the environmental performance of the project, and its effects on the surrounding environment;
  - (d) assess whether the project is complying with the relevant standards, performance measures and statutory requirements;
  - (e) review the adequacy of any strategy/plan/program required under this approval; and, if necessary,
  - (f) recommend measures or actions to improve the environmental performance of the project, and/or any strategy/plan/program required under this approval.
- 7. Within 1 month of completion of each Independent Environmental Audit, the Proponent shall submit a copy of the audit report to the Director-General and relevant agencies, with a response to any of the recommendations in the audit report.
- 8. Following each Independent Environmental Audit, the Proponent shall review and if necessary revise each of the environmental management and monitoring strategies/plans/programs in schedules 3 and 5, to the satisfaction of the Director-General. The revised strategies/plans/programs shall be submitted to the Director-General within 6 months of completing the audit.

#### COMMUNITY CONSULTATIVE COMMITTEE

9. Prior to starting quarrying operations on the site, the Proponent shall establish a CCC for the project. This CCC must be established and operated in accordance with the *Guideline for Establishing and Operating Community Consultative Committees for Mining Developments*, and to the satisfaction of the Director-General.

#### ACCESS TO INFORMATION

- 10. Within 1 month of the approval of any plan/strategy/program required under this approval (or any subsequent revision of these plans/strategies/programs), or the completion of any independent environmental audit or AEMR, the Proponent shall:
  - (a) provide a copy of the relevant document/s to Tweed Shire Council and relevant agencies; and
  - (b) ensure that a copy of the relevant document/s is made publicly available on site and/or at the Proponent's regional office,

to the satisfaction of the Director-General.

- 11. During the project, the Proponent shall:
  - (a) make a summary of monitoring results required under this approval publicly available at the Proponent's regional office; and
  - (b) update these results regularly (at least every 3 months),

to the satisfaction of the Director-General.





APPENDIX 2 REHABILITATION STAGING PLAN

#### APPENDIX 3 STATEMENT OF COMMITMENTS

#### Amended Statement of Commitments

The following sections summaries the commitments by Ramtech Pty Ltd regarding mitigations and control measures to be implemented for the proposal:

#### Sand Extraction

- Additional resource assessment works are to be undertaken in respect of two (2) key elements, being the extent of resource available for brickles loam and the extent of resource available with respect of mineral sands. Such assessments are to be undertaken in accord with the matters raised by the Department of Primary Industries in correspondence dated 21<sup>o</sup> December 2007.
- Annual production data will be made available in accord with DPI requirements.
- Extraction of marine clay will not be undertaken
- Sand extraction below the watertable will be by suction dredge only. No dry extraction of sand
  will occur on site, with the exception of the initial overburden and brickles loam.
- Prior to commencement of extraction, the extent of the approved extraction areas shall be clearly and permanently marked by a licensed surveyor with survey posts.
- Fines will be re-interred approximately below the watertable at approximately 10m below NSL.

#### Sediment and Erosion Control

- A perimeter bund and catch drain shall be constructed around each dredge pond and processing areas. The bund is to be vegetated.
- Installation of sediment control fences at the downslope perimeter of cleared or disturbed land. These are to be functional before clearing commences.
- A negative grade will be maintained around the dredge ponds within the bunded perimeter.
- Additional Erosion and sediment control devices shall be installed on an 'as required' basis. Such measures will be installed in accordance with the "Soils and Construction Guidelines – Managing Urban Stormwater".
- Where practical, surface waters from undisturbed areas shall be diverted away from extraction/works areas.
- All processing areas will drain towards the onsite water bodies. No discharge of processing waters from the site shall occur under normal conditions (ie. Not flooded).
- Topsoil stripping will be undertaken in sub-stages of 1 hectare or less
- All existing ground cover around the site is to remain and be maintained to limit sediment and erosion.
- Any on-site stockpiles of commercial sand shall remain damp and will have appropriate sediment and erosion control devices installed at all times.
- Stockpiled topsoil shall be seeded so as to achieve adequate vegetation cover and sediment and erosion devices will be installed around all stockpiles at all times.
- No discharge of processing water from the site shall occur under normal conditions (ie. not flooded)

#### Surface Water Control & Quality

- Installation of surface and ground water monitoring devices as located on figure GJ0400.9.2 (Appendix Q)
- Surface water monitoring shall be undertaken in accordance with requirements as outlined with the draft EMP under Appendix G.

- All processing areas will drain towards the onsite water bodies. No discharge of processing waters from the site shall occur under normal conditions (ie. not flooded).
- Provision of reliable in situ monitoring equipment at all times for use by Quarry staff. This
  equipment will be calibrated at least monthly.
- All effluent generated will be pumped off site for treatment at Council facility.

#### Groundwater Movement and Quality

- Installation of ground water monitoring devices as located on figure GJ0400.8.1 (Appendix Q)
- Ground water monitoring shall be undertaken in accordance with the requirements as outlined within the draft EMP under Appendix G.
- Provision of reliable in situ monitoring equipment at all times for use by Quarry staff. This
  equipment will be calibrated at least monthly.
- All groundwater bores will be licensed by DIPNR.
- Dewatering from on site water bodies will not be undertaken
- Contour profiling of groundwater head data will be undertaken as part of site monitoring and reporting procedures.

#### Fuel Management and Land Contamination

- Fuel storage is to be contained within a bund area, and protected form the elements. Bunding
  will be sufficient to contain 110% of the volume of fuel storage.
- Operating procedures for containing and cleaning up oil spills on water to be established and
  implemented on site, with all staff to be trained in these procedures. Such measures are to
  form part of the site EMP.
- Products designed to contain and absorb oil spills on water will be available on site. Quantity
  and type of product will be monitored and will be available in sufficient quantities to deal with
  any potential spill on site.
- Materials stored on site will be limited to
- One (1) month supply of diesel
  - Machine and equipment oils and grease
  - limited quantities of petrol

#### Air Quality

- The full length of internal haulage roadways will be sealed.
- A vegetation barrier for dust control along the southern boundary adjoining Warwick Park Road will be established (species and planting in accordance with rehabilitation plan, Appendix H).
- Topsoil stripping will be undertaken in sub-stages of 1 hectare of less.
- Topsoil stripping will not be undertaken on day with excess winds.
- All trucks entering/leaving the site shall be covered.
- Stockpiled topsoil shall be seeded so as to achieve adequate vegetation cover and sediment and erosion devices will be installed around all stockpiles.
- Any disturbed or unsealed movement areas will be watered by an onsite cart to ensure that such areas remain damp. Watering rates shall not be less than 2.5!/m<sup>2</sup>/hour.

#### Noise

Construction of a 4m high earth mound shielding the operational areas. This mound shall form
part of the stockpile and be lightly top solled, seeded and maintained with native grasses.
- The earth mound is to be constructed between 7.30am and 5pm. The occupants of the nearest
  dwellings to the site be notified as to the hours of operation for the mound construction, and be
  provided with a contact telephone number should they have concerns regarding noise from this
  stage of the operation;
- The buildozer used to construct the earth barriers should be well maintained, and fitted with residential mufflers;
- Upon plant dredge pump selection, noise levels should be less than or equal to 88dB(A) at 1m from the plant. If this level is not achievable, further acoustic treatment in the form of a semi enclosure will be required to reduce source noise levels to within the acceptable range;
- Upon sand screening plant selection, noise levels should be less than or equal to 30dB(A) at 1m from the plant. If this level is not achievable, further acoustic treatment in the form of a semi enclosure will be required to reduce source noise levels to within the acceptable range;
- Haulage trucks and the wheeled loader should be well maintained, and fitted with residential mufflers;
- Internal haulage route and the crossover to Pottsville Road be as smooth as possible, and well maintained;
- Prior to commencement of operations, an acoustic test be conducted to ensure compliance with the noise limit criteria;
- All operations to be limited to 7am to 6pm, Monday to Friday, and 7am to midday, Saturdays.

#### Flora & Fauna

- A detailed regeneration and rehabilitation plan is to be prepared and approved by the DECCC prior to the issue of a construction certificate.
- Removal of existing weed species (particularly infestations of Bitou Bush and Lantana).
- Revegetation of disturbed areas within existing native vegetation communities with plants endemic to the locality. Revegetation areas will total approximately 15ha in area. No clearing of vegetation is required with the exception of approximately 20 Casuarina trees lining minor drainage channels in the extraction areas.
- Revegetation of nominated areas within the rehabilitation plan as attached under Appendix H to strengthen potential habitat comidors, extend the distribution of Swamp Solerophyll and Littoral Rainforest environments and to extend the perimeter of the narrowest portions of the eastern forest to provide long-term mitigation against potential edge effects.
- A monitoring program will be implemented in respect of the rehabilitation areas external to the
  extraction areas. Such program will be subject to final approval by the DECCC and will contain
  consideration of additional rehabilitation / regeneration areas at the expense of implementation
  of the nest boxes. Consideration will also be given to strengthening corridor linkages where
  possible.
- Incorporation of potential foraging resources for threatened fauna species within landscaping/revegetation areas including;
  - Allocasuarina littoralis, A. torulosa for Glossy Black Cockatoos
  - Flowering species (Eucalypts, Corymbians, Melaleuca, Banksia) for flying fox/bat species
  - Littoral Rainforest fruiting species for Fruit-doves/Rainforest Pigeons
  - Favoured Eucalypts for Koalas
  - Installation of nest boxes within the existing vegetation communities.

#### Waste

- All soil waste will be disposed of by licensed contractor in accordance with the Protection of Environmental Operations Act 1997 (POEO Act 1997) & Waste Minimisation Act 1995.
- All solid wastes are to be managed in accord with the attached HMC Solid Waste Management Plan.
- Effluent will be treated on site in accord with the attached HMC report.

#### Flooding

Adoption of a maximum 300mm bund height to all levels below 2m AHD.

#### Cultural Heritage

- The proponent will undertake additional archaeological surveys of the proposed extraction areas prior to works commencing to provide assurance that no aboriginal sites exist within the extraction areas
- If aboriginal sites are found within the extraction areas the proponent must undertake surveys /
  assessment to determine the outural heritage values of the site and report how those values
  might be impacted upon and provide appropriate avoidance, mitigation or compensatory
  measures. Such additional assessments must be in accord with guidelines and requirements of
  the DECCC.

#### Views and Landscapes

- Sm to 10m vegetated buffers will be established around the extraction areas in accordance with the rehabilitation plan attached under Appendix H. The objectives of these areas are for visual protection primarily.
- A detailed rehabilitation plan will be prepared for those areas external to the extraction areas.

#### Monitoring, Reporting and Consultation

 Reporting and consultation measures will be implemented as outlined with the draft EMP under Appendix G.

#### Licensing

The proponent will commit to gaining an Environmental Protection Licence from DECCC.

APPENDIX 4 INDEPENDENT DISPUTE RESOLUTION PROCESS

# Independent Dispute Resolution Process (Indicative only)



NSW Government Department of Planning



# ATTACHMENT 3 – LANDSCAPING PLANS



BASE PROVIDED BY	

OPERATIONAL WORKS LANDSCAPE PLAN - KEY PLAN

DRAWING TITLE



# Drawing Schedule

Sheet 1	Drawing Schedule + Key Plan	845-0
Sheet 2	Buffer Planting - Stage 1 Haul Route	845-02
Sheet 3	Buffer Planting - Stage 1 Sand Extraction	845-03
Sheet 4	Rehabilitation Area 1A	845-04
Sheet 5	Rehabilitation Area 1B	845-05
Sheet 6	Rehabilitation Area 1C	845-00
Sheet 7	Rehabilitation Area 2A	845-07
Sheet 8	Rehabilitation Area 2B	845-08
Sheet 9	Rehabilitation Area 2C	845-09
Sheet 10	Rehabilitation Area 3	845-10
Sheet 11	Rehabilitation Area 4 - Stage 1-3	845-1
Sheet 12	Agricultural Area	845-12
Sheet 13	Specifications	845-13

## Dunloe Park Sand Extraction Scale: 1:5000 at full size A1

	<b>scale:</b> AS SHOWN	DATE: JUNE 2016
_	<b>design:</b> JB/GD	CHECKED: AS
	<b>drawn:</b> JB	DRAWING NO: DS_OPW_845_01

CLIENT RAMTECH PTY LTD



Scale: 1:5000 at full size A1

O Ō ()

 $\mathbf{\nabla}$ 

Telephone: 07 5526 1500 Fax: 07 5526 1502









DRAWN: DRAWING NO: DS\_OPW\_845\_02 JB





BASE PROVIDED BY

SCALE: AS SHOWN	DATE: JUNE 2016	CLIENT RAMTECH PTY LTD	Level 1 : Nobby I
<b>design:</b> JB/GD	CHECKED: AS		PO Box
drawn: JB	<b>DRAWING NO:</b> DS_OPW_845_03		

Stage	0 - 6 months	6 - 12 months	Year 1+
<b>er Planting</b> Area: 3.10Ha	PLANTING MODULE 1         Canopy/Subcanopy Trees         10 Canopy trees per module         20 Subcanopy trees per module         20 Subcanopy trees per module         refer to planting list this sheet         Potential for salvaged plant stock (refer         above plan) to be translocated to achieve         planting densities	PLANTING MODULE 2 Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
ban int	Banksia integrifolia	Coast Banksia			
cas gla	Casuarina glauca	Swamp Oak			
cor int	Corymbia intermedia	Pink Bloodwood			
euc gra	Eucalyptus grandis	Flooded Gum			
euc ter	Eucalyptus tereticornis	Forest Red Gum			
euc mic	Eucalyptus microcorys	Tallow Wood	10 min 3 species	45LTR	340
euc sid	Eucalyptus siderophloia	Ironbark	per mod		
lop sau	Lophostemon sauveolens	Swampbox			
lop con	Lophostemon confertus	Brush Box			
mel qui	Melaleuca quinquenervia	Paperbark			
syn glo	Syzygium moorei	Watermelon Tree			



# **Rehabilitation Area 1A**

Total Area: 1.85Ha

Vegetation Type: Fresh-water Wetland

# TREES - TO DAM

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
cas gla	Casuarina glauca	Swamp Oak			
cal sal	Callistemon salignus	Willow Bottlebrush	0	140MM	21
lop sau	Lophostemon sauveolens	Swampbox	7		
mel qui	Melaleuca quinquenervia	Paperbark			

# **REHABILITATION AREA 1A (185 MODULES)**

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
bau jun	Baumea juncea	Bare Twigrush			
bau ter	Baumea teretifolia	Twigrush			
ble ind	Blechnum indicum	Bungwall Fern			
cen asi	Centella asiatica	Pennywort			
cyp pol	Cyperus polystachyos	Bunchy Sedge			Stage 1
fim fer	Fimbristylis ferruginea	Rusty Fringesedge			5500
fim pol	Fimbristylis polytrichoides	Fuzzy Rush		75MM TUBE	
gah asp	Gahnia aspera	Saw Sedge	100		Stage 2
jun kra	Juncus kraussii	Searush	min 6 species		5500
jun usi	Juncus usitatus	Common Rush	per mod		
phi lan	Philydrum lanuginosum	Frogsmouth			Stage 3
phr aus	Phragmites australis	Common Reed			7500
sch val	Schoenoplectus validus	Clubrush			
sch lit	Schoenoplectus littoralis	Clubrush			
spo vir	Sporobolus virginicus	Salt Couch			
tri str	Triglochin striatum	Steaked Arrow Grass			
xyr com	Xyris complanata	Hatpins			



Planting Module 1 - Freshwater Wetland PIONEER SPECIES PLANTING 10 x 10 meter module

> NO DATE REVISION BY 01 06/16 OPW REVISED MB

NTS

Dunloe Park Sand Extraction Scale: 1:1000 at full size A1

DUNLOE PARK - REHABILITATION PLAN

OPERATIONAL WORKS LANDSCAPE PLAN - AREA 1A

BASE PROVIDED BY

DRAWING TITLE

**PROJECT TITLE** 



**Rehabilitation Area 1A** - location plan Dunloe Park Sand Extraction



Boundary of rehabilitation areas to be pegged out by surveyor prior to commencement of works. Pegs to be star pickets / galv pipe to minimum 1 meter above ground painted yellow with notation to match as noted on these plans. Existing vegetation line to form all other boundaries as indicated this sheet. FENCES

Boundary of rehabilitation areas to be fenced if cattle are to be introduced to adjacent grazing areas.



To be planted as part of Stage 1 works Vegetation Type: Fresh-water Wetland

# Rehabilitation Schedule

Stage	Year 1	Year 2	Year 3	Year 4	Year 5+
Stage 1Total Area: 0.55Ha		PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplimentary planting to increase diversity.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
	On-going monitoring and mainter	ance as per Rehabilitation Plan 🕨 🕨 🕨			
<b>Stage 2</b> Total Area: 0.55Ha	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09 Existing Dam to be planted as per Callout this sheet	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 1A species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplimentary planting to increase diversity.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
		<ul> <li>On-going monitoring and mainter</li> </ul>	nance as per Rehabilitation Plan 🕨 🕨 🕨		
<b>Stage 3</b> Total Area: 0.75Ha	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 1A species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplimentary planting to increase diversity.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
			On-going monitoring and mainter	ance as per Rehabilitation Plan	

 										+ + + + + + + + + + + + + + + + + + +			Sedges, Rushes and Ferns Supplementary planting of species taken from Planit List 1A this sheet to increase diversity and maintain densities
												· · · · · · · · · · · · · · · · · · ·	<b>Existing Trees</b> Existing trees to remain
Planting Module 2 - Freshwater Wetland       NTS         SUPPLEMENTARY SPECIES PLANTING 10 x 10 meter module       NTS													

scale: AS SHOWN	DATE: JUNE 2016
<b>design:</b> JB/GD	CHECKED: AS
drawn: JB	DRAWING NO: DS_OPW_845_04

# CLIENT RAMTECH PTY LTD



Level 1 2247 Gold Coast HwyTelephone: 07 5526 1500Nobby BeachFax: 07 5526 1502PO Box 206 Nobby Beach QLD 4218admin@planitconsulfing.com.au







OPERATIONAL WORKS LANDSCAPE PLAN - AREA 1B

BASE PROVIDED BY

DRAWING TITLE

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY	
cal sal	Calistemon salignus	White Bottlebrush			Stage 1	
cor int	Corymbia intermedia	Pink Bloodwood			2600 <b>Stage 2</b>	
euc ter	Eucalyptus tereticornis	Blue Gum	20	75MM TUBE	2000 <b>Stage 3</b>	
euc rob	Eucalyptus robusta	Swamp Mahogany	Minimum 3		2000 Stage 4	
mel qui	Melaleuca quinquenervia	Paperbark	selected		2400 Stage 5	
lop sau	Lophostemon sauveolens	Swamp Box	per module		1500	

SMALL TREES / SHRUBS - module 1, 2 and 3	MAI	EES / SHRUBS - r	module 1, 2 c	and 3
--	-----	------------------	---------------	-------

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
all tor	Allocasuarina torulosa	Forest Oak			Stage 1
acr imp	Acronychia imperforata	Beach Acronychia	]		3900
ban int	Banksia integrifolia	Coastal Banksia			Stage 2
cup ana	Cupaniopsis anarcardiodes	Tuckeroo			3000
dub myo	Duboisia myoporoides	Corkwood	30	75MM TUBE	Stage 3
hov acu	Hovea acutifolia	Hovea	Minimum 3		3000
not lon	Notolaea longifolia	Long-leaved Mock-olive	species		<b>Stage 4</b> 3000
pit rev	Pittosporum revolutum	Forest Pittosporum	per module		
syz ole	Syzygium oleosum	Blue Lillipilli	1		2250 Stage 5
tro lau	Trococarpa laurina	Tree Heath	1		

Stage	Year 1	Year 2	Year 3	Year 4	Year 5	Year 6	Year 7	Year 8	Year 9
<b>Stage 1</b> Total Area: 1.3Ha	PLANTING MODULE 1 PIONER SPECIES PLANTING Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - "Trees" and "Small Trees" selected from Rehabilitation Area 18 species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
	Natural Regeneration	PLANTING MODULE 1		PLANTING MODULE 2		Natural Regeneration	Natural Regeneration	Natural Regeneration	Natural Regeneration
<b>Stage 2</b> Total Area: 1.0Ha	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 1B species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determined onsite by Regeneration expert)	SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet.	SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet.	SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
	Natural Regeneration	Natural Regeneration					Natural Regeneration	Natural Regeneration	Natural Regeneration
<b>Stage 3</b> Total Area: 1.0Ha	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determined onsite by Regeneration expert)	SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet.	SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet.	SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. Trees' and "Small Trees' selected from Rehabilitation Area 18 species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
			• •	On-going monitoring and maintenance	as per Rehabilitation Plan				
<b>Stage 4</b> Total Area: 1.0Ha	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
				• •	On-going monitoring and maintenance	as per Rehabilitation Plan 🕨 🕨 🕨			
<b>Stage 5</b> Total Area: 0.75Ha	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 1B species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 1B species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. Trees' and 'Small Trees' selected from Rehabilitation Area 18 species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09

SCALE: AS SHOWN	DATE: JUNE 2016	CLIENT RAMTECH PTY LTD	Level 1 22 Nobby Be
<b>design:</b> JB/GD	CHECKED: AS		PO Box 20
<b>drawn:</b> JB	<b>drawing no:</b> DS_OPW_845_05		

CODE     PLANT SPECIES     COMMON NAME     NO PER MODULE     SIZE       aus dul     Austromyrtus dulcis     Midyim       ble ind     Blechnum indicum     Swamp Water Fern       cen asi     Centella asiatica     Pennywort       cyp pol     Cyperus polystachyos     Flat Sedge	QTY
aus dul     Austromyrtus dulcis     Midyim       ble ind     Blechnum indicum     Swamp Water Fern       cen asi     Centella asiatica     Pennywort       cyp pol     Cyperus polystachyos     Flat Sedge	
ble ind     Blechnum indicum     Swamp Water Fern       cen asi     Centella asiatica     Pennywort       cyp pol     Cyperus polystachyos     Flat Sedge	
cen asi     Centella asiatica     Pennywort       cyp pol     Cyperus polystachyos     Flat Sedge	1
cyp pol Cyperus polystachyos Flat Sedge	
dia cae Dianella caerulea Blue Flax Lilly	To be
gah asp Gahnia aspera Saw Sedge 50 75MM TUBE	determined on-site by bush
hib sca Hibbertia scandens Snake Vine Minimum 5	regeneration expert
har vio Hardenbergia violacea Native sarsaparilla species selected	
lom Ion Lomandra longifolia Matrush	
pte esc Pteridium esculentum Braken Fern	
sch val Schoenoplectus validus Clubrush	
xyr com Xyris complanata Yelloweyed Grass	











# **Rehabilitation Area 1C**

# Total Area: 2.0Ha

Vegetation Type: Swamp Sclerophyll + Littoral Rainforest Understorey

# **REHABILITATION AREA 1C**

TOTAL AREA: 2.0Ha (200 MODULES) TREES - module 1, 2 and 3

CODE	PLANT SPECIES	COMMON NAME		SIZE	QTY				
cal sal	Calistemon salignus	White Bottlebrush			Stage 1				
cor int	Corymbia intermedia	Pink Bloodwood			Stage 2				
euc ter	Eucalyptus tereticornis	Blue Gum	20	75MM TUBE	1300				
euc rob	Eucalyptus robusta	Swamp Mahogany	Minimum 3		<b>Stage 3</b>				
mel qui	Melaleuca quinquenervia	Paperbark	species		1400				
lop sau	Lophostemon sauveolens	Swamp Box	per module						

# SMALL TREES / SHRUBS - module 1, 2 and 3

_						
	CODE	PLANT SPECIES COMMON NAME		NO PER MODULE	SIZE	QTY
Γ	all tor	Allocasuarina torulosa	Forest Oak			
	acr imp	Acronychia imperforata	Beach Acronychia			Stage 1
	ban int	Banksia integrifolia	inksia integrifolia Coastal Banksia upaniopsis anarcardiodes Tuckeroo			1950
	cup ana	Cupaniopsis anarcardiodes				Stage 2
	dub myo	Duboisia myoporoides     Corkwood       Hovea acutifolia     Hovea		30	75MM TUBE	1950
	hov acu			Minimum 3		Stage 3
	not lon	Notolaea longifolia	Hovea acutifolia     Hovea       Notolaea longifolia     Long-leaved Mock-olive			2100
	pit rev	Pittosporum revolutum	Forest Pittosporum	per module		
	syz ole	Syzygium oleosum	Blue Lillipilli			
Γ	tro lau	Trococarpa laurina	Tree Heath			

# GROUNDCOVERS - module 3 only

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
aus dul	Austromyrtus dulcis	Midyim		R LE SIZE 75MM TUBE of ref so ad Jule	
ble ind	Blechnum indicum	COMMON NAME     NO PER MODULE       Midyim			
cen asi	Centella asiatica				
cyp pol	Cyperus polystachyos				
dia cae	Dianella caerulea			To be	
gah asp	Gahnia aspera	Saw Sedge	50	75MM TUBE	determined on-site by bush
hib sca	Hibbertia scandens	Saw Sedge Snake Vine	Minimum 5		expert
har vio	Hardenbergia violacea	Native sarsaparilla	species selected		
lom lon	Lomandra longifolia	Matrush	per module		
pte esc	Pteridium esculentum	Braken Fern			
sch val	Schoenoplectus validus	Clubrush	]		
xyr com	Xyris complanata	Yelloweyed Grass			

Scale: 1:1000 at full size A1

Stage 1 Total Area: 0.65

Stage

Stage 2 Total Area: 0.65

Stage 3

Total Area: 0.7H

PROJECT TITLE	ROJECT TITLE					
	DUNLOE PARK - REHABILITATION PLAN		NO	DATE	REVISION	BY
			01	06/16	OPW REVISED	MB
DRAWING TITLE						
OPERATIONAL WORKS LANDSCAPE PLAN - AREA 1C						
BASE PROVIDED BY	BASE PROVIDED BY			1	I	

BASE

Planting Module 1 - Swamp Sclerophyll PIONEER SPECIES PLANTING 10 x 10 meter module

**Rehabilitation Area 1C - species list** 

	Year I	Year 2	Year 3	Year 4	Year 5	Year 6	Year /+
5Ha	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	<b>PLANTING MODULE 2</b> SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
	• •	On-going monitoring and maintenance of the second secon	as per Rehabilitation Plan				
5Ha	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	<b>PLANTING MODULE 2</b> SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet.	<b>PLANTING MODULE 2</b> SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
		▶ ▶	On-going monitoring and maintenance of	as per Rehabilitation Plan 🕨 🕨 🕨			
На	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	<b>PLANTING MODULE 2</b> SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet.	<b>PLANTING MODULE 2</b> <b>SUPPLEMENTARY SPECIES PLANTING</b> Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. 'Trees' and 'Small Trees' selected from Rehabilitation Area 1C species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
			▶ ▶	On-going monitoring and maintenance of	is per Rehabilitation Plan 🕨 🕨 🕨		

scale: AS SHOWN	DATE: JUNE 2016	CLIENT RAMTECH PTY LTD	Level 1 22 Nobby Be
<b>design:</b> JB/GD	CHECKED: AS		PO Box 20
<b>drawn:</b> JB	DS_OPW_845_06		

Dunloe Park Sand Extraction

# **Dunloe Park** - Rehabilitation Plan



Small Trees / Shrubs

Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

# Existing Trees

Existing trees to remain

Planting Module 2 - Swamp Sclerophyll SUPPLEMENTARY SPECIES PLANTING 10 x 10 meter module

Groundcovers

Introduction of Groundcover species, exact numbers dependent on natural regeneration.

NTS

- Small Trees / Shrubs Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined

onsite by bush regeneration expert

Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

 Existing Trees Existing trees to remain

NTS

5m

Planting Module 3 - Swamp Sclerophyll NTS SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS 10 x 10 meter module

Gold Coast Hwy Telephone: 07 5526 1500 n Fax: 07 5526 1502 obby Beach QLD 4218 admin@planitconsulfing.com.au





Ο

 $\bigcirc$ 

O

 $\bigcirc$ 

СГ

ehabilitation









# **Dunloe Park** - Rehabilitation Plan

**OPW Landscape Plans** 



PROJECT TITLE DUNLOE PARK - REHABILITATION PLAN	NO         DATE           01         06/16	REVISION     BY       OPW REVISED     MB	SCALE: AS SHOWN	<b>DATE:</b> JUNE 2016	CLIENT RAMTECH PTY LTD	Level 1 22 Nobby Be
DRAWING TITLE			<b>design:</b> JB/GD	CHECKED: AS		PO Box 20
OPERATIONAL WORKS LANDSCAPE PLAN - AREA 2A BASE PROVIDED BY			drawn: JB	DRAWING NO: DS_OPW_845_07		

ľ	U	N	A	KE	F	٩	ZA		
:	1	.4	0h	Α	(	1	40	мо	DU

dule 1, 2 and 3				
PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
Banksia integrifolia	Coastal Banksia	20		Stage 1
Calistemon salignus	White Bottlebrush	Minimum 3		Stage 2
Lophostemon sauveolens	Swamp Box	species selected	7 SIWIWI TOBE	900
Melaleuca quinquenervia	Paperbark	per module		Stage 3 900



Planting Module 1 - Swamp Sclerophyll, She-oak + Banksia NTS PIONEER SPECIES PLANTING 10 x 10 meter module

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
acm hem	Acmena hemilampra	Broad-leaved Lillipilli			
aca sop	Acacia sophorae	Coast Wattle		75MM TUBE	Stage 1
ale cor	Alectryon coriaceus	Beach Alectryon			1500
con bar	Commersonia bartramia	Brown Kurrajong	30		Stage 2 1350 Stage 3 1350
cup ana	Cupaniopsis anarcardiodes	Tuckeroo			
dub myo	Duboisia myoporoides	Corkwood	Minimum 3 species		
ela ret	Elaeocarpus reticulatus	Blueberry Ash	selected per module		
exo lat	Exocarpus latifolius	Broad-leaved Cherry			
per str	Persoonia stradbrokensis	Coast Geebung			

Year 1 PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by	Year 2 PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A	Year 3 PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase	Year 4 PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO	Year 5 Natural Regeneration	Year 6 Natural Regeneration	Year 7+
PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO	Natural Regeneration	Natural Regeneration	Natural Regeneration
Regeneration expert)	species list this sheet.	aversity. Proneer species - Trees and Small Trees' selected from Rehabilitation Area 2A species list this sheet.	INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. 'Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
	natural regeneration (determeined onsite by Regeneration expert) Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09 Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Instructive densities as stated dependent of natural regeneration (determeined onsite by Regeneration expert)       Interest selected from Rehabilitation Area 2A species list this sheet.         Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       PLANTING MODULE 1         Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       Planting Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09         Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Indexe defaultes studied dependent of natural regeneration (determeined onsite by Regeneration expert)       Index selected from Reindbillication Area 2A species list this sheet.       Index selected from Reindbillication Area 2A species list this sheet.         Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       PLANTING MODULE 1       PLANTING MODULE 1         Ploneer species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)       PLANTING Regeneration Area 2A species list this sheet.       Supplementary planiting to increase diversity. Ploneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet.       Supplementary planiting to increase diversity. Ploneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet.       Supplementary planiting to increase diversity. Ploneer species - Trees' and 'Small Trees' selected from Rehabilitation Area 2A species list this sheet.         Natural Regeneration (determeined onsite by Regeneration gand maintenance as per Rehabilitation Plan - prepared by Planit January 09       On-going monitoring and maintenance as per Rehabilitation Area 2A species list this sheet.         Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       Natural Regeneration (determeined onsite by Regeneration Queed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       Planting Trees' selected from Rehabilitation Area 2A species list this sheet. Exact numbers to achieve densitis as stated dependent on natural regeneration expert) </td <td>Indurative definition using of determined on site by Regeneration expert)       Interest specified from Reindomitation Ared 2A species list this sheet.       Site of determined on site by species list this sheet.       Site of determined on site by species list this sheet.       Site of determined on site by species list this sheet.       Site of determined on site by species list this sheet.       Site of determined on site of Groundcover species, dependent on natural regeneration.         Natural Regeneration Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       PLANTING MODULE 1 PIONEER SPECIES PLANING Treest selected from Rehabilitation Area 2A species list this sheet.       PLANTING MODULE 2 Supplementary planing to increase diversity. Ploneer species - Treest and Small Treest selected from Rehabilitation Area 2A species list this sheet.       Supplementary planing to increase diversity. Ploneer species - Treest and Small Treest selected from Rehabilitation Area 2A species list this sheet.       Supplementary planing to increase diversity. Ploneer species - Treest and Small Treest selected from Rehabilitation Area 2A species list this sheet.         Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       PLANTING MODULE 1 Plant Rest Selected from Rehabilitation Area 2A species list this sheet.       Supplementary planing to increase diversity. Ploneer species - Treest and Small Treest selected from Rehabilitation Area 2A species list this sheet.         Notificiting and maintenance as per Rehabilitation Plan - prepared by Planit January 09</td> <td>Control defendemendent       Index statistical control (Refrequencies on control (Refrequencies on control (Refrequencies on control (Refrequencies on control (Refrequencies on control (Refrequencies))       Index statistical control (Refrequencies)       Index statistical contron Refrequencical control (Refrequencies)       <td< td=""><td>Land write mains as a land means and appendent of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the reduction of the steel.          Let by specified in the reduction of the reducti</td></td<></td>	Indurative definition using of determined on site by Regeneration expert)       Interest specified from Reindomitation Ared 2A species list this sheet.       Site of determined on site by species list this sheet.       Site of determined on site by species list this sheet.       Site of determined on site by species list this sheet.       Site of determined on site by species list this sheet.       Site of determined on site of Groundcover species, dependent on natural regeneration.         Natural Regeneration Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       PLANTING MODULE 1 PIONEER SPECIES PLANING Treest selected from Rehabilitation Area 2A species list this sheet.       PLANTING MODULE 2 Supplementary planing to increase diversity. Ploneer species - Treest and Small Treest selected from Rehabilitation Area 2A species list this sheet.       Supplementary planing to increase diversity. Ploneer species - Treest and Small Treest selected from Rehabilitation Area 2A species list this sheet.       Supplementary planing to increase diversity. Ploneer species - Treest and Small Treest selected from Rehabilitation Area 2A species list this sheet.         Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09       PLANTING MODULE 1 Plant Rest Selected from Rehabilitation Area 2A species list this sheet.       Supplementary planing to increase diversity. Ploneer species - Treest and Small Treest selected from Rehabilitation Area 2A species list this sheet.         Notificiting and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Control defendemendent       Index statistical control (Refrequencies on control (Refrequencies on control (Refrequencies on control (Refrequencies on control (Refrequencies on control (Refrequencies))       Index statistical control (Refrequencies)       Index statistical contron Refrequencical control (Refrequencies) <td< td=""><td>Land write mains as a land means and appendent of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the reduction of the steel.          Let by specified in the reduction of the reducti</td></td<>	Land write mains as a land means and appendent of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the steel.          Let by specified in the reduction of the reduction of the steel.          Let by specified in the reduction of the reducti



# Small Trees / Shrubs

Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

## Trees

Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

**Existing Trees** Existing trees to remain

Planting Module 2 - Swamp Sclerophyll, She-oak + Banksia SUPPLEMENTARY SPECIES PLANTING 10 x 10 meter module



Groundcovers

Introduction of Groundcover species, exact numbers dependent on natural regeneration.

# Small Trees / Shrubs

Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

# Trees

Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

- Existing Trees Existing trees to remain

Planting Module 3 - Swamp Sclerophyll, She-oak + Banksia SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS 10 x 10 meter module

GROUNDC	OVERS - module 3 only				
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
adi his	Adiantum hispidulum	Rough Maidenhair			To be determined on-site by bush regeneration expert
aus dul	Austromyrtus dulcis	Midyim		75MM TUBE	
ble car	Blechnum cartilagineum	Gristle Fern			
ble ind	Blechnum indicum	Bungwall Fern	50		
cri pen	Crinum pendunculatum	Swamp Lilly	Minimum 5		
gah asp	Gahnia aspera	Saw Sedge	selected		
lom lon	Lomandra longifolia	Matrush	per module		



Gold Coast Hwy Telephone: 07 5526 1500 h Fax: 07 5526 1502 lobby Beach QLD 4218 admin@planitconsulfing.com.au









PROJECT TITLE							
	DUNLOE PARK - REHABILITATION PLAN		NO	DATE	REVISION	BY	
			01	06/16	OPW REVISED	MB	
		{					
DRAWING TITLE							
	OPERATIONAL WORKS LANDSCAPE PLAN - AREA 2B						
BASE PROVIDED B	1						-



Year 3

Natural Regeneration

# spo tri str

Stage

Stage 1

Total Area: 0.43Ha

SCALE:

DESIGN:

DRAWN:

JB/GD

JB

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY		
bau jun	Baumea juncea	Bare Twigrush					
bau ter	Baumea teretifolia	Twigrush					
ble ind	Blechnum indicum	Bungwall Fern					
cen asi	Centella asiatica	Pennywort					
cyp pol	Cyperus polystachyos	Bunchy Sedge	]				
fim fer	Fimbristylis ferruginea	Rusty Fringesedge					
fim pol	Fimbristylis polytrichoides	Fuzzy Rush					
gah asp	Gahnia aspera	Saw Sedge	100		Stage 1		
jun kra	Juncus kraussii	Searush	100	7 SINIM TOBE	4300		
jun usi	Juncus usitatus	Common Rush					
phi lan	Philydrum lanuginosum	Frogsmouth					
phr aus	Phragmites australis	Common Reed	]				
sch val	Schoenoplectus validus	Clubrush	1				
sch lit	Schoenoplectus littoralis	Clubrush					
spo vir	Sporobolus virginicus	Salt Couch	]				
tri str	Triglochin striatum	Steaked Arrow Grass					
xyr com	Xyris complanata	Hatpins					

Year 1

**PLANTING PHASE 1** 

Year 2

PLANTING PHASE 2

# Rehabilitation Area 2B - species list

**REHABILITATION AREA 2B** TOTAL AREA:0.43Ha (43 MODULES)

ES, R	USHES + FERNS				
DE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
jun	Baumea juncea	Bare Twigrush			
ter	Baumea teretifolia	Twigrush			<b>Stage 1</b> 4300
nd	Blechnum indicum	Bungwall Fern			
asi	Centella asiatica	Pennywort			
pol	Cyperus polystachyos	Bunchy Sedge			
er	Fimbristylis ferruginea	Rusty Fringesedge			
ol	Fimbristylis polytrichoides	Fuzzy Rush			
asp	Gahnia aspera	Saw Sedge	100		
ara	Juncus kraussii	Searush	100	75MM TUBE	
ısi	Juncus usitatus	Common Rush			
an	Philydrum lanuginosum	Frogsmouth			
aus	Phragmites australis	Common Reed			
val	Schoenoplectus validus	Clubrush			
it	Schoenoplectus littoralis	Clubrush	1		
vir	Sporobolus virginicus	Salt Couch	1		

Year 4

Natural Regeneration

FENCES Boundary of rehabilitation areas to be fenced if cattle are to be introduced to adjacent grazing areas.

• REHAB 1.0 **BOUNDARY PEGS** Boundary of rehabilitation areas to be pegged out by surveyor prior to commencement of works. Pegs to be star pickets / galv pipe to minimum 1 meter above ground painted yellow with notation to match as noted on these plans. Existing vegetation line to form all other boundaries as indicated this sheet.



 $\mathsf{Q}$ 2 O  $\mathbb{O}$ СГ habilitation



ſ



Level 1 2247 Gold Coast HwyTelephone: 07 5526 1500Nobby BeachFax: 07 5526 1502PO Box 206 Nobby Beach QLD 4218admin@planitconsulfing.com.au





# **Rehabilitation Area 2C**

# Total Area: 3.67Ha

Vegetation Type: Swamp Sclerophyll with Littoral Rainforest Understorey

PROJECT TITLE		$\neg$				
	DUNLOE PARK - REHABILITATION PLAN		NO	DATE	REVISION	BY
			01	06/16	OPW REVISED	MB
		-				
DRAWING TITLE	G TITLE OPERATIONAL WORKS LANDSCAPE PLAN - AREA 2C					
BASE PROVIDED BY						





Planting Module 1 - Swamp Sclerophyll, She-oak + Banksia **PIONEER SPECIES PLANTING** 10 x 10 meter module

# Rehabilitation Area 2C - species list

mel qui Melaleuca quinquenervia Paperbark

**REHABILITATION AREA 2C** TOTAL AREA:3.67Ha (367 MODULES)

TREES - mo	dule 1, 2 and 3			
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE
ban int	Banksia integrifolia	Coastal Banksia	20	
cal sal	Calistemon salignus	White Bottlebrush	Minimum 3	754444 11
lop sau	Lophostemon sauveolens	Swamp Box	species	75/00010

	SMALL TRE	ES / SHRUBS - module 1, 2 c	ind 3		
QTY	CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE
<b>Stage 1</b> 2000	acm hem	Acmena hemilampra	Broad-leaved Lillipilli		
	aca sop	Acacia sophorae	Coast Wattle		
	ale cor	Alectryon coriaceus	Beach Alectryon		
2000 <b>Stage 2</b>	con bar	Commersonia bartramia	Brown Kurrajong	30	754444 TUB
	cup ana	Cupaniopsis anarcardiodes	Tuckeroo	t dimines and 2	/ 5/4/14/100
Stage 3	dub myo	Duboisia myoporoides	Corkwood	species	

Blueberry Ash

Coast Geebung

Broad-leaved Cherry

# **Rehabilitation Schedule Area 2C**

Stage	Year 1	Year 2	Year 3	Year 4	Year 5
<b>Stage 1</b> Total Area: 1.0Ha	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet. Introduction of Groundcover species, dependent on natural regeneration.	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
<b>Stage 2</b> Total Area: 1.0Ha	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet.	PLANTING MODULE 3 SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS Supplementary planting to increase diversity. 'Trees' and 'Small Trees' selected from Rehabilitation Are 2C species list this sheet. Introduction of Groundcover species, dependent on natur regeneration.
<b>Stage 3</b> Total Area: 1.0Ha	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet.
<b>Stage 4</b> Total Area: 0.67Ha	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 2C species list this sheet.
				▶ <b>▶</b>	On-going monitoring and maintena

Stage 2000

**Stage 4** 1340

ela ret

exo lat

per str

Elaeocarpus reticulatus

Persoonia stradbrokensis

Exocarpus latifolius

selected per module

SCALE: DATE: CLIENT RAMTECH PTY LTD as shown JUNE 2016 DESIGN: CHECKED: JB/GD AS DRAWN: DRAWING NO: DS\_OPW\_845\_09 JB

Dunloe Park Sand Extraction Scale: 1:2000 at full size A1



Nobby Beach Fax: 07 5526 1502 PO Box 206 Nobby Beach QLD 4218 admin@planitconsulting.com.au





Φ



# Total Area: 0.6Ha

Vegetation Type: Swamp Sclerophyll, She-oak +Littoral Rainforest Understorey

PROJECT TITLE						
	DUNLOE PARK - REHABILITATION PLAN		NO	DATE	REVISION	BY
			01	06/16	OPW REVISED	MB
DRAWING TITLE						
	OPERATIONAL WORKS LANDSCAPE PLAN - AREA 3					
BASE PROVIDED B	3Y					



Rehabilitation Area 3 - location plan



Planting Module 1 - Swamp Sclerophyll, She-oak +Littoral Rainforest NTS PIONEER SPECIES PLANTING 10 x 10 meter module

# Rehabilitation Area 3 - species list

**REHABILITATION AREA 3** 

TOTAL AREA:0.60hA (60 MODULES) TREES - module 1, 2 and 3

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY		
cas gla	Casuarina glauca	Swamp Oak	20		Stage 1		
cal sal	Calistemon salignus	White Bottlebrush	Minimum 3				
lop sau	Lophostemon sauveolens	Swamp Box	species selected	7 SIMIM TOBE	1200		
mel qui	Melaleuca quinquenervia	Paperbark	per module				

SMALL TREES / SHRUBS - module 1, 2 and 3								
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY			
acr imp	Acronychia imperforata	Beach Acronychia		<b>75MM TUBE</b>				
all lit	Allocasuarina littoralis	Black She-oak			<b>Stage 1</b> 1800			
ban int	Banksia integrifolia	Coast Banksia						
com bar	Commersonia bartramia	Brown Kurrajong	20					
dub myo	Duboisia myoporoides	Corkwood						
gou sem	Gouia semiglauca	Wild Quince	species					
mis pyr	Mischocarpus pyriformis	Yellow Pearfruit	per module					
per str	Persoonia stradbrokensis	Coast Geebung						
pil gla	Pilidiostigma glabrum	Plum Myrtle						
pol ele	Polyscias elegans	Celerywood						

# **Rehabilitation Schedule Area 3**

Stage	Year 1	Year 2	Year 3	Year 4
<b>Stage 1</b> Total Area: 0.60Ha	PLANTING MODULE 1 PIONEER SPECIES PLANTING 'Trees' and 'Small Trees' selected from Rehabilitation Area 3 species list this sheet. Exact numbers to achieve densities as stated dependent on natural regeneration (determeined onsite by Regeneration expert)	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLATING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 3 species list this sheet.	PLANTING MODULE 2 SUPPLEMENTARY SPECIES PLANTING Supplementary planting to increase diversity. Pioneer species - 'Trees' and 'Small Trees' selected from Rehabilitation Area 3 species list this sheet.	PLANTING MODULE SUPPLEMENTARY SPECIES PLANTIN INCLUDE GROUNDCOVERS Supple planting to increase diversity. 'Tree 'Small Trees' selected from Rehab 3 species list this sheet. Introducti Groundcover species, dependen regeneration.

NTS

-	SCALE: AS SHOWN DESIGN: JB/GD	DATE: JUNE 2016 CHECKED: AS	CLIENT RAMTECH PTY LTD	Level 1 22 Nobby Be PO Box 20
	<b>drawn:</b> JB	<b>drawing no:</b> DS_OPW_845_10		



# Small Trees / Shrubs

Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

## Tree

Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

- Existing Trees Existing trees to remain

Planting Module 2 - Swamp Sclerophyll, She-oak +Littoral Rainforest **SUPPLEMENTARY SPECIES PLANTING** 10 x 10 meter module



Groundcovers

Introduction of Groundcover species, exact numbers dependent on natural regeneration

- Small Trees / Shrubs Supplementary planting to increase diversity and maintain densities.

(exact numbers to be determined onsite by bush regeneration expert)

Supplementary planting to increase

diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)

 Existing Trees Existing trees to remain

Planting Module 3 - Swamp Sclerophyll, She-oak +Littoral Rainforest SUPPLEMENTARY SPECIES PLANTING TO INCLUDE GROUNDCOVERS 10 x 10 meter module

# GROUNDCOVERS - module 3 only

Skoundeovers - module 5 only							
CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY		
adi his	Adiantum hispidulum	Rough Maidenhair					
aus dul	Austromyrtus dulcis	Midyim					
ble ind	Blechnum indicum	Bungwall Fern	50	75MM TUBE	To be determined on-site by bush regeneration expert		
chr api	Chrysocephalum apiculatum	Yellow Buttons					
cri pen	Crinum pendunculatum	Swamp Lilly	Minimum 5				
gah asp	Gahnia aspera	Saw Sedge	selected				
goo rot	Goodenia rotundifolia	Star Goodenia	permodule				
lom lon	Lomandra longifolia	Matrush					

# 3

NG TO lementary rees' and bilitation Area tion of ent on natural

# Year 5

Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09

# Year 6+

Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09

Gold Coast Hwy Telephone: 07 5526 1500 h Fax: 07 5526 1502 lobby Beach QLD 4218 admin@planitconsulfing.com.au







Rehabilitation Planting Area 4 | STAGE 1 Total Area: 9,688M<sup>2</sup>



# Rehabilitation Planting Area 4 | STAGE 2 Total Area: 5,432M<sup>2</sup>

# **Rehabilitation Planting Area 4**

Total Area: 21,705M<sup>2</sup>

EXTENT OF REHABILITATION PLANTING WORKS

Rehabilitation Works | 8 meter wide buffer to northern sides of creek

Stage 1: 9,688M<sup>2</sup> Stage 2: 5,432M<sup>2</sup> Stage 3: 6,585M<sup>2</sup>

Boundary of Buffer Planting Area to be pegged out by surveyor prior to commencement of works. Pegs to be star pickets / galv pipe to minimum 1 meter above ground painted yellow



Canopy Trees 10 Canopy trees per module, refer to planting list this sheet





Canopy Trees

NO DATE REVISION

01 06/16 OPW REVISED

Supplementary planting to increase

diversity and maintain densities.



Planting Module 1

10 x 10 meter module shown - refer also typical section



Planti



BY

MB

PROJECT TITLE

DUNLOE PARK - REHABILITATION PLAN

DRAWING TITLE OPERATIONAL WORKS LANDSCAPE PLAN - REHABILITATION PLANTING AREA 4 STAGE 1-3

BASE PROVIDED BY

Rehabilitation Planting Area 4 | STAGE 3 Total Area: 6,585M<sup>2</sup>

# Works Schedule Rehabilitation Planting

Stage	0 - 6 months	6 - 12 months	Year 1+
I <b>bilitation</b> l <b>ing</b> Area: 21,705M²	<b>PLANTING MODULE 1</b> <b>Canopy/Subcanopy Trees</b> 10 Canopy trees per module 20 Subcanopy trees per module refer to planting list this sheet Potential for salvaged plant stock (refer above plan) to be translocated to achieve planting densities	<b>PLANTING MODULE 2</b> Supplementary planting to increase diversity and maintain densities. (exact numbers to be determined onsite by bush regeneration expert)	Natural Regeneration Weed Management Monitoring and maintenance as per Rehabilitation Plan - prepared by Planit January 09
	On-going monitoring and mainter	enance as per Rehabilitation Plan 🕨 🕨 🕨	

Site subjet to periods of frost All esisting tracks and trails onsite to remain

Typical Buffer Profile

Dunloe Park Sand Extraction Scale: 1:2000 at full size A1

SCALE: AS SHOWN	DATE: JUNE 2016	CLIENT RAMTECH PTY LTD	Level 1 22 Nobby Be
<b>design:</b> JB/GD	CHECKED: AS		PO Box 20
drawn: JB	DRAWING NO: DS_OPW_845_11		

# **Dunloe Park** - Rehabilitation Plan

**OPW Landscape Plans** 

# REHABILITATION AREA 4 - STAGE 1 Total Area: 9,688M<sup>2</sup> (9.7 MODULES)

ioiai	AIC	ч.	,000
CAN	OPY	TR	EES

CODE	PLANT SPECIES	COMMON NAME	NO PER MODULE	SIZE	QTY
ban int	Banksia integrifolia	Coast Banksia			
cas gla	Casuarina glauca	Swamp Oak			
cor int	Corymbia intermedia	Pink Bloodwood		45LTR	970
euc gra	Eucalyptus grandis	Flooded Gum			
euc ter	Eucalyptus tereticornis	Forest Red Gum			
euc mic	Eucalyptus microcorys	Tallow Wood	10		
euc sid	Eucalyptus siderophloia	Ironbark	per mod		
lop sau	Lophostemon sauveolens	Swampbox			
lop con	Lophostemon confertus	Brush Box	]		
mel qui	Melaleuca quinquenervia	Paperbark			

# **REHABILITATION AREA 4 - STAGE 2** Total Area: 5,432M<sup>2</sup> (5.5 MODULES) CANOPY TREES

		· · · · · · · · · · · · · · · · · · ·			
CODE	PLANT SPECIES		NO PER MODULE	SIZE	QTY
ban int	Banksia integrifolia	Coast Banksia			550
cas gla	Casuarina glauca	Swamp Oak		45LTR	
cor int	Corymbia intermedia	Pink Bloodwood			
euc gra	Eucalyptus grandis	Flooded Gum			
euc ter	Eucalyptus tereticornis	Forest Red Gum	]		
euc mic	Eucalyptus microcorys	Tallow Wood	10		
euc sid	Eucalyptus siderophloia	Ironbark	per mod		
lop sau	Lophostemon sauveolens	Swampbox			
lop con	Lophostemon confertus	Brush Box			
mel qui	Melaleuca quinquenervia	Paperbark			

CODE	PLANT SPECIES		NO PER MODULE	SIZE	QTY
ban int	Banksia integrifolia	Coast Banksia			
cas gla	Casuarina glauca	Swamp Oak	1		
cor int	Corymbia intermedia	Pink Bloodwood			
euc gra	Eucalyptus grandis	Flooded Gum			
euc ter	Eucalyptus tereticornis	Forest Red Gum	1	45LTR	660
euc mic	Eucalyptus microcorys	Tallow Wood	10		
euc sid	Eucalyptus siderophloia	Ironbark	per mod		
lop sau	Lophostemon sauveolens	Swampbox			
lop con	Lophostemon confertus	Brush Box			
mel qui	Melaleuca quinquenervia	Paperbark			



# Refer to Plant Modules and Species Listings this sheet for details

Telephone: 07 5526 1500 Fax: 07 5526 1502



**Canopy Trees** 

and rainforest re-growth

Mid - High woodland - Banksia, Casuarina

# 1:100 at full size A1

NORTH

 $\mathfrak{S}$ **(**) 0 D St planting ation bilit O

ſΘ







Total Area: 396,975M<sup>2</sup>

PROJECT TITLE DUNLOE PARK - REHABILITATION PLAN	<b>NO DATE</b> 01 06/16	REVISION OPW REVISED	BY MB	<b>SCALE:</b> AS SHOWN	<b>DATE:</b> JUNE 2016	CLIENT RAMTECH PTY LTD
DRAWING TITLE OPERATIONAL WORKS LANDSCAPE PLAN - AGRICULTURAL PLANTING AREA				<b>design:</b> JB/GD	<b>Checked</b> : AS	
BASE PROVIDED BY				drawn: JB	<b>drawing no:</b> DS_OPW_845_12	

200m

0m 50m 100m



# Dunloe Park - Agricultural Planting OPW Landscape Plans



# **EXTENT OF REHABILITATION PLANTING WORKS** Rehabilitation Works | Fruiting trees (eg. Hass Avocado Trees) Total Area: 396,975M<sup>2</sup>

# Table 1. Row and Tree Spacing Options

N	ROW SPACING (M)	TREE SPACING (M)	NO. TREES PER HECARE
	6 - 8	4 - 6	208 - 417
	8 - 9	5 - 7	159 - 250
	10 - 12	10 - 12	69 - 100

Density information source: Australian Avocado Growers' Federation Section 6







evel 1 2247 Gold Coast Hwy Telephone: 07 5526 1500 obby Beach Fax: 07 5526 1502 O Box 206 Nobby Beach QLD 4218 admin@planitconsulting.com.au



# **Dunloe Park** - Rehabilitation Plan

**OPW Landscape Plans** 

# **REVEGETATION SPECIFICATION**

# **GENERAL NOTES:**

ALL WORKS TO BE CARRIED OUT BY A QUALIFIED BUSH REGENERATION EXPERT WITH AS A MINIMUM, A DIPLOMA OF HORTICULTURE AND 5 YEARS DEMONSTRATED EXPERIENCE IN BUSH REGENERATION. EVIDENCE OF QUALFICATIONS WILL BE REQUIRED TO BE SUBMITTED TO THE SUPERINTENDANT FOR APPROVAL PRIOR TO THE COMMENCEMENT OF THE CONTRACT PERIOD.

ALL WORKS TO BE CARRIED OUT IN ACCORDANCE WITH:

(1) NOXIOUS WEEDS ACT 1993

- (2) THREATENED SPECIES ACT 1995
- (3) PESTICIDES ACT 1999
- (4) NATIVE VEGETATION CONSERVATION ACT 1997

(5) GOLD COAST CITY COUNCIL STANDARDS AND RELEVANT AUSTRALIAN STANDARDS

TO BE READ INCONJUNCTION WITH REHABILITATION AND REVEGETATION MANAGEMENT PLAN PREPARED BY PLANIT CONSULTING FEBRUARY 2009

# SITE PREPARATION

# 1.1 VERIFY BOUNDARIES OF REVEGETATION AREAS

THE CONTRACTOR IS TO ASSESS THE SITE REVEGETATION AREAS IN ORDER TO ASCERTAIN A CLEAR CONSTRUCTION BOUNDARY. ALL BOUNDARIES WILL BE MARKED OUT BY QUALIFIED SURVEYOR PRIOR TO WORK COMMENCEMENT, MARKER PEGS TO HAVE NOTATION AS PER THESE PLANS.

# **1.2 WEED REMOVAL**

# **REFER TO REHABILITATION MANAGEMENT PLAN PREPARED BY PLANIT CONSULTING SECTION 4.1**

# **1.3 SOIL CULTIVATION**

# - AS REQUIRED WITHIN BUFFER PLANTING TO SAND EXTRACTION STAGE 1

CULTIVATE BY RIPPING TO THE DEPTHS SPECIFIED BELOW TO LOOSEN AND AERATE GROUND. DO NOT DISTURB EXISTING NATIVE TREE AND PLANT ROOTS. REMOVE UNWANTED MATTER INCLUDING STONES, RUBBISH AND RUBBLE EXCEEDING 20MM IN DIAMETER AND STICKS AND WEEDS BROUGHT TO THE SURFACE DURING CULTIVATION. FINELY CULTIVATE THE SURFACE AND RAKE FREE OF CLODS. DO NOT REMOVE LARGE LOGS UP TURNED STUMPS ETC AS THIS NATURAL DEBRI FORMS FAUNA HABITAT STRUCTURE

# 2.00 PLANTING

# 2.1 THE WORKS

PLANTING WORK COMPRISES:

 SUPPLY AND INSTALLATION OF ALL GROUND COVER SHRUBS AND TREES AS SPECIFIED ON PLAN AS DETAILED IN 'REHABILITATION SCHEDULE'. EXACT NUMBERS OF PIONEER PLANTS AND SUPPLIMENTARY PLANTS TO BE DETERMINED ON SITE BY BUSH REGENERATION EXPERT - IN STRICT ACCORDANCE WITH SECTION 4 OF THE REHABILITATION/REVEGETATION MAMAGEMENT PLAN PREPARED BY PLANIT CONSULTING FEBRUARY 2009

# 2.2 PLANTS

PLANTS SHALL BE VIGOROUS, WELL ESTABLISHED, OF GOOD FORM, NOT SOFT OR FORCED, HARDENED OFF, FREE FROM DISEASE AND PESTS WITH LARGE HEALTHY ROOT SYSTEMS AND NOT POT BOUND. THE ROOT SYSTEM SHALL BE WELL BALANCED IN RELATION TO THE SIZE OF THE PLANT.

PLANT CONTAINERS SHALL BE OF AN APPROPRIATE SIZE AND FREE FROM WEEDS. PLANTS SHALL NOT EXHIBIT SIGNS OF BEING STRESSED AT ANY STAGE DURING THEIR DEVELOPMENT DUE TO INADEQUATE WATERING, EXCESSIVE SUNLIGHT, PHYSICAL DAMAGE OR HAVE RESTRICTED GROWTH DUE TO NURSERY ROWS. NO SUBSTITUTIONS SHALL BE MADE UNLESS APPROVED BY THE LANDSCAPE DESIGNER.

KEEP PLANTS IN GOOD CONDITION DURING STORAGE. PREVENT DRYING OUT OR DAMAGE FROM ANY CAUSE INCLUDING FROST, WIND, SUN, THEFT, VERMIN ETC.

# 2.3 PLANTING

ENSURE PLANTS CAN BE WATERED IN AT TIME OF PLANTING. DO NOT ALLOW PLANTS TO BE INSTALLED ON A DAY THEY CAN'T BE WATERED.

PLANTING TO BE EXECUTED IN ACCORDANCE WITH THE RELEVANT PLANTING DETAIL. DO NOT PLANT INTO DRY OR MUDDY SOIL OR IN EXTREME WEATHER CONDITIONS. ENSURE PLANT ROOT SYSTEMS ARE MOIST WHEN REMOVED FROM CONTAINER AND NOT ALLOWED TO DRY OUT AND PLANTING AREA HAS BEEN THOROUGHLY WATERED. PLANT OUT WITH A MINIMUM DISTURBANCE TO ROOT BALL ENSURING FINISHED COMPACTED SOIL LEVELS COINCIDE WITH THE NATURAL SOIL LEVEL OF THE PLANT. PLANT, STAKE, TIE AND MULCH ACCORDING TO DETAIL DRAWINGS AND THEN DEEP SOAK THE WHOLE OF THE PLANTING AREA.

2.4 DEPTH OF PLANTING

WHEN THE PLANT IS IN ITS FINAL POSITION IN ITS HOLE THE TOP SOIL LEVEL OF THE PLANT ROOT BALL SHALL BE LEVEL WITH THE FINISHED SURFACE OF THE SOIL SURROUNDING THE HOLE.

# 2.5 PLACING

AND PLUMB.

2.6 BACKFILLING PLANTING HOLE.

# 2.7 PLANTING FERTILIZER

'OSMOCOTE' SUSTAINED RELEASE FERTILISER IN GRANULE FORM OR APPROVED EQUIVALENT SUITABLE FOR NATIVE PLANTS SHALL BE ADDED TO THE SITE TOPSOIL AS PER MANUFACTURERS DIRECTION.

# 3.00 MAINTENANCE - 52 WEEKS

3. 1 GENERAL THE LANDSCAPE CONTRACTOR SHALL BE RESPONSIBLE FOR ESTABLISHMENT MAINTENANCE AND DEFECTS LIABILITY CLAUSES FOR 52 WEEKS FOLLOWING COMPLETION OF THE LANDSCAPE CONSTRUCTION.

# 3.2 MIN. MAINTENANCE REQUIREMENTS:

WATER

# WEEDING

REPLACEMENTS

SOIL SUBSIDENCE

# PROTECTION

PROTECT PLANTED AREAS FROM DAMAGE, EITHER MALICIOUS, IRRESPONSIBLE OR ACCIDENTAL. AREAS OF REHABILITATION TO BE FENCED IF CATTLE ARE TO BE GRAZING ON ADJACENT PASTURE LANDS.

PROJECT TITLE					
	DUNLOE PARK - REHABILITATION PLAN	NO	DATE	REVISION	BY
		01	06/16	OPW REVISED	MB
DRAWING TITLE					
	OPERATIONAL WORKS LANDSCAPE PLAN - SPECIFICATIONS				
BASE PROVIDED BY			•		

WHEN THE HOLE APPEARS TO BE THE CORRECT SIZE, AND NOT BEFORE REMOVE THE PLANT FROM THE CONTAINER WITH MINIMUM DISTURBANCE TO THE ROOT BALL AND PLACE IN ITS FINAL POSITION, IN THE CENTRE OF THE HOLE

BACKFILL WITH SITE TOPSOIL. LIGHTLY TAMP DOWN THE MIXTURE TO ENSURE NO AIR POCKETS ARE LEFT IN

REGULARLY WATER THE PLANTS TO ENSURE ESTABLISHMENT WITH WATER TRUCK.

# REFER TO SECTION 4 OF THE REHABILITATION PLAN PREPARED BY PLANIT CONSULTING FEBRUARY 2009

REPLACE ALL PLANTS THAT ARE UNHEALTHY OR DEAD. REPLACEMENTS SHALL BE IDENTICAL IN SIZE AND SPECIES. REFER CORRESPONDING PLANT LISTS FOR SUBSTITUTIONS WHERE REQUIRED.

MAKE GOOD IF DUE TO WORKMANSHIP OF LANDSCAPE CONTRACTOR.

SCALE

DRAWN:







111



SHOWN	<b>DATE:</b> JUNE 2016	CLIENT RAMTECH PTY LTD	Level 1 2 Nobby B
GD	CHECKED: AS		PO Box 2
	drawing no: DS_OPW_845_13		



CONSULTING



Gold Coast Hwy

ephone: 07 5526 1500 obby Beach QLD 4218 admin@planitconsulting.com.au

JUVENILE CANOPY SPECIES

COMMUNITIES

+ SOME DYING BACK

TYPICAL TO INTENDED VEGETATION

GRASSES / FERNS + VINE GROWTH ESTABLISHED TO COVER

PIONEER SPECIES REACHING MATURITY EMERGENT UNDERSTORY ESTABLISHED EXHIBITING SUITABLE DIVERSITY

S

 $\bigcirc$ 

PIONEER GRASS SPECIES ESTABLISHED TO FULL COVER

GRASSES AND WEEDS AS REQUIRED



# ATTACHMENT 4 – VEGETATION MANAGEMENT PLAN



This Vegetation Management Plan (VMP) has been prepared for both the clearing of all vegetation situated within approved sand quarrying areas and retention and protection of all other native vegetation on the Dunloe Park site.



# APPROVED QUARRYING EXTRACTION AND COMPOUND AREAS

The VMP is to be used as a tool during the construction and operational phases of the development, identifying tasks to be undertaken, the timing of such works and responsible parties for supervision/implementation of vegetation removal/retention on the site.

This plan identifies appropriate vegetation protection methods and site rehabilitation strategies to retain and enhance wildlife habitat and also addresses the necessary removal of vegetation as described above. Strict implementation of the following methodologies will ensure that no retained vegetation will pose any detrimental impacts to future built components on the site.



Element	Vegetation Management
Objectives	To remove native and exotic vegetation located within approved quarrying and compound areas
	• To retain and protect all other native vegetation communities and trees within the Dunloe Park site
	To minimise tree-clearing impacts on native fauna.
	To revegetate rehabilitation and buffer areas
Action	• Tree-clearing zones are to be clearly delineated on-site to ensure that all areas affected by this VMP are readily identifiable.
	• Vegetation to be retained within 20 metres of the approved clearing/works zones are to be tagged/marked/delineated for easy identification (do not use permanent paints or similar) i.e. trees and/or areas of vegetation to be retained are to be clearly fenced similar to below:
	• The contractor/developer must adequately protect from damage any vegetation on private and/or public property which is not designated for removal in association with this development.
	• Tree-clearing, fauna spotter-catcher and environmental consultants working in conjunction with Civil Works Contractors on this project are to be informed of all provisions specified under this VMP.
	• Cleared vegetation is to be disposed of in accordance with accepted measures. This includes mulching for future site-landscaping purposes and/or removal of millable timber where appropriate.
	• Remaining debris not disposed of in either of the above methods is to be removed off-site by the owner to an approved green-waste disposal facility.
	<ul> <li>The following activities are not permitted within the drip zones of vegetation to be retained on or offsite:</li> </ul>



Action	- Sto - Veh - Liqu - Mac - Site - Ligh - Rub - Exc	rage and m nicle parking uids disposa chinery repa office and/ nting of fires oble, soil or avation.	ixing of mater g; al; airs or refuelli or shed erect s; debris stockp	rials; ng; ion; viling; and	
	<ul> <li>No tree obtaining</li> </ul>	es on Coun ng necessar	cil property a y Council cor	are to be removed, pruned or injur nsent.	ed prior to
	<ul> <li>If root/c areas c site, tre thinning</li> </ul>	rown dama occurs/is rec eatment by ) is to be in	age (or other quired during a suitably nplemented.	significant disturbance) to retained approved earthworks/quarrying act qualified Arborist (ie root truncation	vegetation ivity on the ons, crown
	<ul> <li>A recog areas o 5.1 of th</li> </ul>	nised faun f retained v ne attached	a spotter-cat vegetation du report	cher is to ensure safe dispersal of ring clearing works in accordance w	fauna into /ith Section
	<ul> <li>Effectiv provide Environ</li> </ul>	e sedimen d at eart mental Mar	t and erosic hworks bou nagement Pla	on control devices are to be iden ndaries in accordance with the In.	ntified and prepared
	<ul> <li>Areas</li> <li>stabilise</li> </ul>	disturbed a ed. Stabilisa	as a result o ation is to occ	of tree-clearing and/or earthworks ur within ten days of completion of w	are to be orks.
	Revege     Rehabil	tation and itation and	rehabilitation Revegetation	is to occur in accordance with th Management Plan.	e attached
Performance Indicators	Tree-cle	earing activi	ities are restri	icted to identified areas.	
	Native	plants withir	n the identifie	d retention zones are protected.	
	<ul> <li>Sedime accorda</li> </ul>	nt and er ance with th	osion contro e EMP.	ol devices are installed and mai	intained in
	Retaine	d vegetatio	n on and/or c	ffsite is to demonstrate healthy cond	litions:
		Grade	Condition	Descriptor	
		1	Healthy	Leaves green, no abnormal leaf	
		2	Fair	loss	
		2		leaves, but <20% of canopy affected	
		3	Poor	Many leaves yellow or brown, substantial reduction in canopy extent since last measurement	
		4	Dead	Leaves brown or absent, little of the canopy remaining	
	<ul> <li>Rehabil</li> <li>Rehabil</li> </ul>	itation and itation and	buffer zones Revegetation	are managed in accordance with th Management Plan.	ne attached
	No faur	a injury or i	mortality occu	irs during the clearing/construction p	hase



Frequency/	Identification of retained vegetation prior to commencement of clearing works.
Deadline	<ul> <li>Implement tree clearing works upon receipt of tree clearing approval - completion within 12 months.</li> </ul>
	<ul> <li>Recognised fauna spotter-catcher to be present on-site prior to and during all vegetation-clearing works.</li> </ul>
	• All revegetation, weed management and rehabilitation works occur at the timeframes stipulated in the attached Rehabilitation and Revegetation Management Plan.
Person Responsible	• The owner/developer/site manager is responsible for informing all contractors, sub-contractors, consultants and government authorities working on the site of the provisions of this VMP.
	<ul> <li>A suitably qualified consultant is responsible for the supervision and implementation of clearing works.</li> </ul>
	<ul> <li>A recognised fauna spotter-catcher is to be contracted for fauna capture/relocation as necessary.</li> </ul>
	<ul> <li>A suitably qualified consultant is responsible for the supervision and implementation of all rehabilitation works.</li> </ul>
	• A suitably qualified Arborist is responsible for assessing and implementing any remediation works to damaged vegetation retained within protection zones areas if/where required.
	<ul> <li>A suitably qualified consultant is responsible for installing and monitoring erosion and sediment control devices.</li> </ul>
Reporting and	• A suitably qualified consultant is responsible for reporting to the developer where actions specified in this VMP are not undertaken and/or compromised.
Reviewing	<ul> <li>The developer is responsible for reporting to Council where actions specified in this VMP are not implemented</li> </ul>
	• Routine and specified monitoring of fauna, weeds, revegetation and rehabilitation zones is to occur as stipulated in the attached Rehabilitation and Revegetation Management Plan.
Corrective Action	• If vegetation not identified for removal is disturbed during the operational works phase, the need for supplementary rehabilitation works is to be negotiated between the developer and Council.
	• If retained trees show signs of ill health (i.e. poor or dead), likely causes are to be determined, methods of mitigating such effects are to be identified in consultation with a suitably qualified Arborist and Council officers, and mitigation measures to improve growth conditions are to be put in place.
	• Where a tree shows signs of any loss in structural integrity or a potentially unsafe condition, then in the opinion of a suitably qualified Arborist and Council officers the tree shall be either stabilised or removed to avoid any future risk.
	• Where sediment and erosion control structures fail, likely causes are to be identified and additional measures installed.

Appendix B – Agency consultation



Our Ref: DOC19/676169 Your Ref: 06\_0030

> GHD 230 Harbour Drive Coffs Harbour NSW 2450

Attention: Mr Ben Luffman

## Dear Mr Luffman

# Subject: Dunloe Quarry MOD 2 - Landscape and Aboriginal Cultural Heritage Management Plans

Thank you for your letter dated 8 August 2019 about the amended management plans for the Dunloe Quarry MOD 2 seeking comments from the Biodiversity and Conservation Division (BCD) of the NSW Department of Planning, Industry and Environment. I appreciate the opportunity to provide input.

The Biodiversity and Conservation Division was formerly part of the Office of Environment and Heritage (OEH) but now forms part of the new Environment, Energy and Science Group in the Department of Planning, Industry and Environment (see https://www.dpie.nsw.gov.au).

We have reviewed the documents supplied and advise there are several issues apparent with the Landscape Management Plan (LMP), particularly the koala management plan, which lacks adequate detail. The main issues include:

- a) no description of the author qualifications in the Landscape Management Plan (LMP);
- b) insufficient or incorrect references to approval conditions and document sections;
- c) lack of supporting information in the Koala Management Plan (KMP);
- d) the need to refine koala management, monitoring and contingency measures.

These issues are discussed in detail in Attachment 1 to this letter.

Prior to finalising the landscape and Aboriginal cultural heritage management plans we recommend that GHD:

- 1. Revises the Landscape Management Plan to:
  - a) include the relevant qualifications and experience of the contributors to demonstrate compliance with Project Approval 27(a); and
  - b) indicate in Table 2.1 that Project Approval Condition 28 clauses (h) and (i) are addressed in Appendix C.

Locked Bag 914 Coffs Harbour NSW 2450 Federation House, Level 8, 24 Moonee Street Coffs Harbour NSW 2450 Tel: (02) 6659 8200 Fax: (02) 6659 8281 ABN 20 770 707 468 www.dpie.nsw.gov.au

- 2. Revises the Rehabilitation and Revegetation Management Plan to include Project Condition 28 in the list of conditions addressed.
- 3. Revises the Koala Management Plan to:
  - a) replace references to sections and tables of other management plans with the relevant content being referred to in order to minimise potential errors resulting from subsequent management plan revisions or amendments;
  - b) include mapping of koala habitat, koala records and potential koala movement corridors (i.e. habitat links) within and adjacent to the subject land and along the haul road between the quarry site and the Pottsville Road intersection;
  - c) acknowledge the possibility of infrequent koala movements during hours of quarry operation;
  - d) identify the most likely areas of interaction between koalas and quarry vehicles (e.g. koala habitat links);
  - e) include a proposed amendment to the quarry induction process to include an explanation of the legal consequences of unauthorised clearing of native vegetation on the quarry site;
  - f) include provision of compensatory koala food tree plantings as a contingency measure in the event of unauthorised clearing taking place;
  - g) ensure the proposed monitoring methodology focuses on identifying areas of koala activity susceptible to road strike rather than attempting to identify temporal changes in koala densities.
  - h) reduce the proposed koala road-strike threshold for management action from three koalas for the year to any koala at any time
- 4. Revise the Aboriginal Cultural Heritage Assessment Report to:
  - a) include a statement confirming the report was prepared in accordance with the Code of Practice for Archaeological Investigation of Aboriginal Objects in NSW 2010;
  - b) remove references to a requirement for an Aboriginal Heritage Impact Permit; and
  - c) ensure the recommendations are consistent with the requirement for a Care Agreement to remove Aboriginal objects from the approved project boundary for long-term management if required.
- 5. Revise the Aboriginal Cultural Heritage Management Plan to replace references to the Office of Environment and Heritage (OEH) with the Biodiversity and Conservation Division (BCD).

If you have any further questions about this issue, Mr Don Owner, Senior Conservation Planning Officer, Biodiversity and Conservation, can be contacted on 6659 8233 or at don.owner@environment.nsw.gov.au.

Yours sincerely

Vinition Juny 4 September 2019

DIMITRI YOUNG Senior Team Leader Planning, North East Branch <u>Biodiversity and Conservation</u>

Contact officer: DON OWNER 6659 8233

Enclosure: Attachment 1: Detailed BCD Comments - Dunloe Quarry MOD 2 - Landscape and ACH Management Plans

Attachment 1: Detailed Biodiversity and Conservation Comments – Dunloe Quarry MOD 2 – Landscape and Aboriginal Cultural Heritage Management Plans

## Landscape Management Plan

## Author Qualifications

Project Approval 27(a) requires the plan to be prepared by suitably qualified consultants, including specialist hydrologist, coastal engineer, wetlands ecologist and landscape architect. The Landscape Management Plan (LMP) does not contain the qualifications of the contributing authors, which is necessary to demonstrate this condition has been met.

## BCD Recommendation:

1. Revise the LMP to include the relevant qualifications and experience of the contributors to demonstrate compliance with Project Approval 27(a).

## Project Approval Condition 28

The Rehabilitation and Revegetation Management Plan (RRMP) contained in Appendix A of the LMP claims to address the requirements of Conditions 26, 27a and 29 of the Project Approval. However, Project Approval Condition 28, which specifies what must be included in the RRMP, has not been referred to in the RRMP.

## BCD Recommendation:

2. Revise the RRMP to include Project Condition 28 in the list of conditions addressed.

### Koala Management Plan

Table 2.1 of the LMP indicates Project Approval Condition 28 clauses (h) and (i), which relate to koala management, are addressed in the RRMP provided in Appendix A. However, these conditions have not been addressed in the RRMP. Instead, they have been addressed in the Koala Management Plan (KMP) contained in Appendix C of the LMP.

## BCD Recommendation:

3. Revise Table 2.1 of the LMP to indicate Project Approval Condition 28 clauses (h) and (i) are addressed in Appendix C.

Section 1.3 of the KMP refers to Section 5.4.1 of the LMP for details of the KMP reporting provisions. However, the LMP does not contain a Section 5.4.1. It is also indicated in this section that the presence of koalas will be noted during the monitoring program set out in Table 5.1 of the LMP. However, Table 5.1 provides a very brief summary of the monitoring program for the rehabilitation area, which would not include monitoring of the haul road.

### BCD Recommendation:

4. Revise the KMP to replace references to sections and tables of other management plans with the relevant content being referred to in order to minimise potential errors resulting from subsequent management plan revisions or amendments.

The KMP does not provide any supporting information to inform or provide context to the proposed koala management or monitoring provisions.

## BCD Recommendation:

5. Revise the KMP to include mapping of koala habitat, koala records and potential koala movement corridors (i.e. habitat links) within and adjacent to the subject land and along the haul road between the quarry site and the Pottsville Road intersection.

Section 1.2.1 of the KMP indicates the risk of koala mortalities due to quarry-related vehicle strike will be low due to the quarry operating hours not coinciding with the main periods of koala movement. This assumption is generally accurate. However, although infrequent, some koala movements may still occur during the quarry operating hours (e.g. movements resulting from territorial disputes or during dispersal of young from natal areas).

## BCD Recommendations:

- 6. Revise the KMP to acknowledge the possibility of infrequent koala movements during hours of quarry operation.
- 7. Revise the KMP to identify the most likely areas of interaction between koalas and quarry vehicles (e.g. koala habitat links).

Section 1.2.2 of the KMP briefly discusses the potential impacts of unauthorised clearing of koala habitat. However, no preventative or contingency measures other than clearly demarcating the limit of authorised clearing have been proposed.

## BCD Recommendation:

- Revise the KMP to include a proposed amendment to the quarry induction process to include an explanation of the legal consequences of unauthorised clearing of native vegetation on the quarry site.
- 9. Revise the KMP to include provision of compensatory koala food tree plantings as a contingency measure in the event of unauthorised clearing taking place.

Proposed monitoring of the frequency of koala sightings would be based on incidental records rather than application of a systematic repeatable sampling method. The resulting dataset will not provide a reliable measure for determining temporal changes in koala occupancy levels or local population size, as intended in the KMP.

Nevertheless, such information may have some use in identifying areas of important koala occupancy, which could then be used to identify potential road strike '*black-spots*' and to formulate mitigation measures (e.g. speed limits, warning signage, traffic calming devices etc.).

### BCD Recommendation:

10. Revise the KMP to ensure the proposed monitoring methodology focuses on identifying areas of koala activity susceptible to road strike rather than attempting to identify temporal changes in koala densities.

Table 1 of the KMP lists one of the proposed management triggers as being '*quarry-related vehicle koala strikes reach or exceed three for the year*'. This trigger threshold is too high given that three koalas are likely to represent a significant proportion of the koala population utilising the subject land and adjoining areas in any given year.

## BCD Recommendation:

11. Revise the KMP to reduce the proposed koala road-strike threshold for management action from three koalas for the year to any koala at any time.

## Aboriginal Cultural Heritage Management

RPS prepared an Aboriginal cultural heritage assessment report (ACHAR) to investigate the potential for harm to Aboriginal cultural heritage resulting from proposed expansion of the extraction boundaries into currently fenced off sand ridge areas. However, the ACHAR was not undertaken under the defence of being in accordance with the *Code of Practice for Archaeological Investigation of Aboriginal Objects in NSW 2010*.

References in the ACHAR to the need for an Aboriginal Heritage Impact Permit (AHIP) are incorrect given any proposed harm to Aboriginal objects from the approved project would be regulated by a management plan approved by the Secretary of the Department rather than an AHIP.

However, any Aboriginal objects identified whilst undertaking the approved project works could be lawfully removed from within the approved project boundary if the removal is consistent with the will of the Registered Aboriginal Parties and undertaken in accordance with a Care Agreement issued by the Department of Planning, Industry and Environment under section 85A(1)(c) of the National Parks and Wildlife Act 1974.

We note that one of the three recommendations provided in the ACHAR for inclusion in the updated Aboriginal Cultural Heritage Management Plan (ACHMP) appears inconsistent with the requirement for a Care Agreement to remove Aboriginal objects from the approved project boundary for long term management if required.

The relevant state government point of contact provided in the ACHMP for various aspects of Aboriginal cultural heritage recording and reporting (i.e. OEH) has recently changed and this should be amended to the Biodiversity and Conservation Division (BCD).

## BCD Recommendations:

- 12. Revise the ACHAR to:
  - a) include a statement confirming the ACHAR was prepared in accordance with the Code of Practice for Archaeological Investigation of Aboriginal Objects in NSW 2010;
  - b) remove references to a requirement for an AHIP.
  - c) ensure the recommendations are consistent with the requirement for a Care Agreement to remove Aboriginal objects from the approved project boundary for long term management if required.
- 13. Revise the ACHMP to replace references to the Office of Environment and Heritage (OEH) with the Biodiversity and Conservation Division (BCD).



OUR REF: C19/467

13 September 2019

Mr Ben Luffman Technical Director – Environment GHD 230 Harbour Drive COFFS HARBOUR NSW 2450

Dear Mr Luffman

## Re: C19/467 Dunloe Sand Quarry (MP 06\_0031) Landscape Management Plan consultation

I refer to your email dated 8 August 2019 requesting consultation with DPI Fisheries for the purpose of preparing a Landscape Management Plan for the Dunloe Sand Quarry Project (MP 06\_0031). It is noted that consultation with DPI Fisheries for the development of the Landscape Management Plan is a specific requirement within Condition 27 of the Minster's Conditions of Approval (CoA).

DPI Fisheries is responsible for ensuring that fish stocks are conserved and that there is "no net loss" of key fish habitats upon which they depend. To achieve this, the Coastal Systems Unit assesses activities under Part 4 and Part 5 of the *Environmental Planning and Assessment Act 1979* in accordance with the objectives of the FM Act, the aquatic habitat protection and threatened species conservation provisions in Parts 7 and 7A of the FM Act, and the associated and *Policy and Guidelines for Fish Habitat Conservation and Management (2013 Update)* (DPI Fisheries P&G). This document is available online at:

<u>https://www.dpi.nsw.gov.au/fishing/habitat/protecting-habitats/toolkit</u>. In addition, DPI Fisheries is responsible for ensuring the sustainable management of viable commercial fishing and aquaculture; quality recreational fishing; and to promote the continuation of Aboriginal cultural fishing within NSW.

It is noted that the submitted draft Landscape Management Plan, titled *Holcim Australia Dunloe Sand Quarry Landscape Management Plan* and dated August 2019, contains the relevant requirements as stipulated by Condition 27 of the CoA and, if implemented in full, would contribute to the protection of adjacent key fish habitats from potential impacts of the sand quarry. In summary, DPI Fisheries has no objection to the draft Landscape Management Plan.



If you have any queries, please contact me on 02 6626 1375 or jonathan.yantsch@dpi.nsw.gov.au.

Yours sincerely

Manter

Jonathan Yantsch Senior Fisheries Manager, Coastal Systems (North Coast) Authorised delegate of the Minister for Primary Industries

# **Ben Luffman**

From:	Ben Luffman
Sent:	Thursday, 8 August 2019 12:50 PM
To:	'landuse.enquiries@dpi.nsw.gov.au'
Cc:	'Victoria Musgrove'
Subject:	Dunloe Quarry Management Plan consultation
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
<b>КероТуре:</b>	JOD

Hi,

We have updated the management plans for Dunloe Quarry following the recent approval of MOD2. The conditions of the Project Approval – SSD 06\_0030 require a number of the plans to be prepared in consultation with the DoI. We have therefore provided a link below to the relevant plans for review.

https://ghd.sendthisfile.com/M3RFj9HigPcjj1ATu8LUMpEj

The updates have mainly been a reformatting to remove duplication and inclusion of additional information to address the new requirements of the conditions.

We would appreciate your comments by 23 August 2019.

Please contact me if you have any questions.

### Regards

# Ben Luffman | A GHD Associate

B.App.Sc. (Hons) | Grad.Dip. Urban and Regional Planning | Environmental Auditor **Technical Director - Environment** 

### GHD

#### **Proudly employee owned** T: +61 2 6650 5613 | M: +61 415 271 319 | E: ben.luffman@ghd.com

230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.ghd.com</u>



WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION



GHD acknowledges the Traditional Owners of Country throughout Australia. We pay respect to their continuing culture and Elders past, present and emerging. <u>Click here</u> to learn about our Reconciliation Action Plan.

Please consider the environment before printing this email

1 ream of paper = 6% of a tree / 5.4kg CO2 in the atmosphere | 3 sheets of A4 paper = 1 litre of water

# **Ben Luffman**

From:	cassandra.mcnamara@dpi.nsw.gov.au on behalf of DPI Cabinet <dpi.cabinet@dpi.nsw.gov.au></dpi.cabinet@dpi.nsw.gov.au>
Sent:	Friday, 23 August 2019 10:34 AM
То:	Ben Luffman
Cc:	Brendan Stone
Subject:	DPI Advice - Local Project - Dunloe Quarry Management Plan Consultation
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
RepoType:	Job

Hi Ben

# Subject: Request for input Dunloe Quarry Management Plan Consultation

I refer to your email of 8 August 2019 to the Department of Planning, Industry and Environment (DPIE) - Department of Primary Industries (DPI) regarding the above matter.

The Department of Primary Industries has reviewed the various management plans and has no objection or comment.

Any further referrals to DPIE - DPI can be sent by email to dpi.cabinet@dpi.nsw.gov.au

Kind regards, Cass

DPI Coordination Team: Cass McNamara, Manager - 0404 087 481 Jane Bak, A/Manager - 0438 458 914 (27 Aug - 20 Sept) Sophia Stanley, Policy & Project Officer - 0427 326 931

eCabinet: https://ecab.nsw.gov.au/ecabinet-prod/login?0

NSW Department of Primary Industries Lvl 49 MLC Centre | 19 Martin Place | Sydney NSW 2000 E: <u>dpi.cabinet@dpi.nsw.gov.au</u>



REMINDER: The eCabinet system is a secure system and all documents are physically and electronically marked with the specific details of the system user. Alerts are created when these actions are made. Do not print or save any document from the eCabinet system.

------ Forwarded message ------From: **Ben Luffman** <<u>Ben.Luffman@ghd.com</u>> Date: Thu, 8 Aug 2019 at 12:50 Subject: Dunloe Quarry Management Plan consultation To: <u>landuse.enquiries@dpi.nsw.gov.au</u> <<u>landuse.enquiries@dpi.nsw.gov.au</u>> Cc: Victoria Musgrove <<u>victoria.musgrove@lafargeholcim.com</u>>

Hi,

We have updated the management plans for Dunloe Quarry following the recent approval of MOD2. The conditions of the Project Approval – SSD 06\_0030 require a number of the plans to be prepared in consultation with the DoI. We have therefore provided a link below to the relevant plans for review.

 $\underline{https://ghd.sendthisfile.com/M3RFj9HigPcjj1ATu8LUMpEj}$ 

The updates have mainly been a reformatting to remove duplication and inclusion of additional information to address the new requirements of the conditions.

We would appreciate your comments by 23 August 2019.

Please contact me if you have any questions.

Regards

# Ben Luffman | A GHD Associate

<u>B.App.Sc</u>. (Hons) | Grad.Dip. Urban and Regional Planning | Environmental Auditor **Technical Director - Environment** 

## GHD

**Proudly employee owned** T: +61 2 6650 5613 | M: +61 415 271 319 | E: <u>ben.luffman@ghd.com</u> 230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.ghd.com</u>



# **Ben Luffman**

From:	Colleen Forbes <cforbes@tweed.nsw.gov.au></cforbes@tweed.nsw.gov.au>
Sent:	Wednesday, 21 August 2019 4:08 PM
То:	Ben Luffman
Cc:	Victoria Musgrove; Scott Hetherington; Vince Connell (InTouch); Tracey Stinson
Subject:	RE: Dunloe Quarry - Landscape Management Plan consultation
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
RepoType:	Job

## Dear Ben,

As requested and as part of your consultation process, Council has reviewed the *Dunloe Sand Quarry Landscape Management Plan* August 2019, primarily in relation to **Appendix E - Koala Management Plan** which Council was concerned about in Mod 2.

General comment:

• The bulk of the overall management plan comprises material from between 2006 and 2016 and it is difficult to interpret which previous study or work the published content refer to.

The Koala Management Plan is not supported in its current form due to the following:

- Hours of operation are proposed as the key threat mitigating factor in relation to koalas and vehicle strike. This does not satisfactorily account for winter, when the acknowledged high risk times of dawn and dusk occur during these hours of operation.
- The plan should note that koalas can be on the ground at any time of day or night.
- The plan should also identify the times of the year when risk is higher due to seasonal movements of young males, presence of females with back young and the concurrence of haulage times with dawn and dusk during winter as per above.
- All actions in relation to koala sightings and quarry related vehicle strike are assigned responsibility to the 'Planning and Environment Manager – NSW'. The document needs specify what organisation this relates to.
- No specific actions are proposed in response to quarry related vehicle koala strikes.
- There is no obligation or accountability for quarry staff to record koala sightings.
- 3 koala strikes per year is too many for the endangered population to sustain.
- 'Increased presence of koalas on haulage routes' is intimated as the likely cause for vehicle strikes. This is considered to be a premature and unfounded assumption.

Please contact me if you require any clarification on the matters raised above.

Regards, Colleen

# **Colleen Forbes**

Team Leader Development Assessment



Your actions matter: print less to save more

From: Ben Luffman [mailto:Ben.Luffman@ghd.com]
Sent: Thursday, 8 August 2019 12:52 PM
To: Colleen Forbes
Cc: Victoria Musgrove
Subject: Dunloe Quarry Management Plan consultation

## Hi Colleen,

As discussed, we have updated the management plans for Dunloe Quarry following the recent approval of MOD2. The conditions of the Project Approval – SSD 06\_0030 require the Landscape Management Plan to be prepared in consultation with Council. We have therefore attached the relevant plan for review.

The updates have mainly been a reformatting to remove duplication and inclusion of additional information to address the new requirements of the conditions.

We would appreciate your comments by 23 August 2019.

Please contact me if you have any questions.

### Regards

### Ben Luffman | A GHD Associate

B.App.Sc. (Hons) | Grad.Dip. Urban and Regional Planning | Environmental Auditor **Technical Director - Environment** 

### GHD

#### Proudly employee owned

T: +61 2 6650 5613 | M: +61 415 271 319 | E: <u>ben.luffman@qhd.com</u> 230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.ghd.com</u>



WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION



GHD acknowledges the Traditional Owners of Country throughout Australia. We pay respect to their continuing culture and Elders past, present and emerging. <u>Click here</u> to learn about our Reconciliation Action Plan.

#### Please consider the environment before printing this email

1 ream of paper = 6% of a tree / 5.4kg CO2 in the atmosphere | 3 sheets of A4 paper = 1 litre of water

Appendix C – Long Term Management Strategy
# Long Term Management Strategy

### Introduction

The progressive rehabilitation of the wider site throughout the operational life of the Dunloe Sand Quarry is a commitment made by Holcim with the aim of providing for a well-established and long term environmentally sound landscape at the completion of operational works of the quarry. The following outlines the performance criteria to which Holcim or any future owners of the quarry will be required to meet, with an indication as to possible end uses of the land area.

End of operations management and monitoring commitments, to ensure adherence to the performance criteria are also described.

### **Background - Staged Rehabilitation**

The Rehabilitation and Revegetation Management Plan (RRMP) presented in Appendix A, outlines a program of staged rehabilitation of various areas of the site. These rehabilitation areas were previously approved under Development Consent 06\_0030. The RRMP, as amended from time to time, shows a detailed commitment to revegetating previously disturbed areas to create and enhance wildlife corridors, protect riparian areas and ensure the stability of ground and surface water quality within the catchment of Mooball Creek.

The detail outlined in the RRMP has been and continues to be implemented These works will ensure that by the completion of sand extraction, the required rehabilitated areas will be well established with vegetation to a point of consolidation of continued growth. Upon completion of extraction operations, site conditions will be such that plantings can also be pursued to enhance the end of life appearance and function of the land.

The remainder of this document will outline the procedures for management and monitoring to achieve the objectives beyond the life of the quarry operations.

### **Objectives**

- To ensure that rehabilitation works undertaken during the life of the quarry continue to thrive following the completion of operational works of the quarry.
- To implement management and monitoring procedures following the completion of quarry operations to provide for a permanent, healthy, local ecosystem that successfully functions within the natural parameters of the existing, localised, vegetative communities.
- To allow for and facilitate agricultural use around the perimeter of the extraction areas.
- Upon closure of the quarry operations the rehabilitated and revegetated areas are considered well established and capable of thriving without the need for continuing works and management.

### **Performance Criteria (Quarry Closure and Post-Extraction)**

The following details the end of quarry operational life performance criteria:

- Healthy, thriving vegetation shall be well established upon closure of the quarry and be in a state considered to be capable of thriving into the future with minimal requirements for ongoing management.
- No visible weed infestations in those areas that have been rehabilitated or revegetated or the riparian zones associated with Mooball Creek within the site of the quarry.
- No visible erosion or sediment accumulation within rehabilitated areas and all erosion and sediment control preventative measures are in place prior to closure of the quarry.

- In regard to groundwater quality, target criteria to be used throughout the life of the quarry shall be required to be met upon closure of the quarry (refer to the Soil and Water Management Plan).
- In regard to surface water quality, target criteria to be used throughout the life of the quarry shall be required to be met upon closure of the quarry (refer to the Soil and Water Management Plan).
- Blue-green algae water quality targets presented in the following table and prescribed by Condition 9 of Schedule 3 of the Development Consent shall be met upon closure of the quarry:

Algae and Blue-	No. cells/mL (M.aeruginosa)	<50,000
green algae	mm <sup>3</sup> /L (total biovolume)	<4

### Implementation

1. Surface water monitoring will continue to be undertaken in accordance with the approved schedule at nominated monitoring locations (including monitoring for blue-green algae within the lakes) (refer to the Soil and Water Management Plan) for a period of 12 months after completion of extractive operations or as long as is necessary to indicate that the extraction area will have no impact upon the environment.

2. Groundwater monitoring will be undertaken in accordance with an approved schedule at nominated environmental monitoring locations (refer to the Soil and Water Management Plan) for a period of no greater than 12 months after completion of extractive operations should the monitoring continue to show compliance with set requirements of the Development Consent.

3. All bunding and catch drain lines shall have been removed around each on-site excavation area to allow natural surface water flow to the remaining lakes.

4. Inspection of all rehabilitated areas in accordance with an approved schedule to ensure plantings have been successful and to ensure no weed infestation is present at any of these localities (covered by bond) (refer to the RRMP).

### Monitoring

Quarterly surface water monitoring in accordance with the approved schedule (refer to the Soil and Water Management Plan) will continue following the cessation of extraction. This would continue for a period of 12 months only, if water quality objectives can be demonstrated.

If there is visual evidence of an algal bloom, samples will be collected immediately for algal cell count/identification analysis.

Quarterly groundwater monitoring in accordance with the approved schedule (refer to the Soil and Water Management Plan) will continue following the cessation of extraction. This would continue for 12 months only, if groundwater quality objectives can be demonstrated.

As the revegetated areas will essentially be stable forests upon cessation of quarry operations, monitoring of these areas will continue to be undertaken 12 months from the cessation of operations to ensure quality of the vegetation is maintained. The monitoring is to include weeds, erosion and plant health, density and diversity.

After 12 months, monitoring of the rehabilitation areas will cease should all areas be considered stable and effectively free of weeds and approved for no further management by the DPI&E or until such time that it can be demonstrated that no adverse impacts associated with the quarry exist.

### Auditing

Groundwater, surface water, blue-green algae and revegetation area monitoring results will be reviewed quarterly for the first year following cessation of operations by the respective contracted environmental consultant/s. Results shall be compared with those found within previous AEMR documents to ensure compliance with criteria.

### **Corrective Action**

Should final monitoring of revegetated areas show the presence of significant areas of weed infestation at cessation of the operations, a program will be implemented to remedy this situation to the satisfaction of the DPI&E.

In regard to groundwater, surface water and blue-green algae, the recommended corrective actions presented within the Soil and Water Management Plan will be implemented.

### **Potential Future Uses**

Potential future uses of the site following a project with a lengthy operational life such as this are difficult to accurately predict. Many variables could occur that may dictate end uses such as State, Regional and Local Government, strategies, policies and legislation changes and directions over time. The certainty is that the mitigation and remedial measures implemented during the operations and at the cessation of operations, as presented throughout the EMP, will ensure that the site will be remediated to the satisfaction of the company, the relevant Government agencies and the community, ensuring environmental quality in the locality.

The proposed rehabilitation and revegetation of the site will allow future agricultural activities to occur within the surrounds of the lakes

Much of the existing areas of the site that are presently under environmental protection through zoning within the Tweed Shire Council LEP may remain under various legislated forms of environmental protection. Those areas of rehabilitated land (particularly those linking and expanding wildlife corridors) not within these protected zones may overtime be included.

Similarly, agriculturally zoned lands within the site that will not be impacted by the sand quarry operations may remain under such zonings into the future.

The environmental measures presented throughout the EMP allow the site following cessation of operations to be suitable and capable of the potential end land uses.

Appendix D – Koala Management Plan

# **Koala Management Plan**

### **D1.1** Introduction

Koala habitat has been identified in areas surrounding the Dunloe Sand Quarry and access road, as well as the Pottsville Mooball Road which is used to access the quarry.

Schedule 3, Condition 28 (h) and (i) require the Dunloe Sand Quarry Rehabilitation and Revegetation Management Plan (RRMP) to include:

- A monitoring and reporting program of the project's impacts on Koalas, including road strike, to the satisfaction of the Secretary.
- Adaptive management options for managing impacts on Koalas, including specific impact triggers, developed in consultation with Council.

This Koala Management Plan (KMP) forms part of and should be read in conjunction with the RRMP, Landscape Management Plan and Environmental Management Strategy for the Dunloe Sand Quarry.

Koalas are mostly nocturnal animals and they are most active during the night and at dawn and dusk. However, it is noted that koalas can be on the ground at any time of day or night. Koalas are more active during the spring and summer breeding period. The birth of young Koalas generally takes place between November and February, with the young remaining in the pouch for six months. The joey will then ride on the mother's back for between six and 12 months.

### **D1.2** Potential Impacts

### D1.2.2 Quarry-related vehicle strike

The quarry access road runs parallel with the northern Brushbox Forest (Community 3), which provides koala habitat.

Potential impacts from the project on koalas in the area may arise from vehicle strike where haulage/access routes are located close to koala habitat or areas used by koalas. Under the Development Consent 06\_0030, haulage vehicle movements associated with delivery and distribution of quarry materials are restricted to the following operational hours:

- 7:00 am to 5:00 pm Monday to Friday; and
- 7:00 am to 12 pm Saturday.

The risk of vehicle strike is likely to be higher during the winter months when dawn and dusk fall within the approved operational hours for the quarry. Similarly, the risk would be higher during the spring and summer breeding period.

### **D1.2 Unauthorised clearing of koala habitat**

Clearing of koala habitat will not be undertaken as part of the ongoing quarry operations, however unauthorised clearing in these habitat areas would have the potential to impact koalas.

As outlined in the Landscape Management Plan, approved disturbance areas will be demarcated and access restricted to areas outside these areas, except for monitoring and management purposes.

### **D1-3** Monitoring

In order to monitor these potential impacts, Holcim will record any unauthorised clearing, koala sightings or vehicle strikes involving quarry-related vehicles in its online incident reporting system. This will allow such occurrences to be tracked and areas of koala activity and potential road strike identified. Reporting provisions are detailed in Section 5.4 of the Landscape Management Plan and the EMS.

The presence of koalas and any unauthorised clearing of koala habitat will also be noted during the monitoring program set out in Table 5-1 of the Landscape Management Plan.

### **D1-4 Reporting**

Records of any unauthorised habitat clearing, koala sightings and vehicle strikes involving quarry-related vehicles and koalas will be reported in the Annual Report, along with a discussion of any identified cause, trends such as specific area, time of day, time of year, climatic conditions, etc. associated with the occurrences.

### **D1-5 Adaptive Management**

Impacts to koalas as a result of quarry activities, specifically haulage, or unauthorised habitat clearing, will be reviewed annually as part of the review of monitoring results, which forms part of the Annual Report, to assess any adverse impacts to koalas. The monitoring results will also be used to identify any trends in occurrences of koala sightings or vehicle strikes.

Responses to the results of this monitoring will be required where certain triggers are met or exceeded. Details of the triggers, actions and responses for koala management are provided in Table 1.

It should be noted that there have been no koala vehicle strikes as a result of quarry vehicles during the site's operation.

Trigger	Action	Responsibility
Approval of KMP (2021)	Incorporate koala impact management materials into site induction, identifying risk areas (i.e. haulage route particularly in proximity to the northern Brushbox Forest (Community 3)), seasonal considerations (i.e. breeding season November to April, potential presence of young / joeys throughout the year) and the requirement to report all koala sightings, near misses and vehicle strikes.	Quarry Manager
Approval of KMP (2021)	Install koala warning signage along internal haul route	Quarry Manager

#### Table 1 Koala management triggers

Trigger	Action	Responsibility
Unauthorised clearing of koala habitat	Investigate cause of unauthorised clearing	Quarry Manager
	Implement measures to prevent recurrence of unauthorised clearing (will depend on cause identified)	Quarry Manager
	Compensatory planting at a ratio of 10 to 1.	Quarry Manager
Koala sightings along haul routes increases year on year	Investigate if any trends are determinable from these occurrences, including whether increased sightings correlates with increased quarry-related vehicle strikes	Planning and Environment Manager – NSW
	Continue to monitor koala sightings in subsequent years to determine if occurrences continue to increase	Planning and Environment Manager – NSW
Quarry-related vehicle koala strike is reported	Review all occurrences and investigate if any trends are determinable from these occurrences	Planning and Environment Manager – NSW
Quarry-related vehicle koala strike recorded for any two	Investigate if any trends are determinable from these occurrences	Planning and Environment Manager – NSW
(2) years in a rolling five (5) year period	Consult with local koala experts and Tweed Shire Council to understand potential causes for increased presence of koalas on or nearby haulage routes and/or incidence of koala vehicle strikes associated with quarry operations and to identify potential measures to be implemented to minimise impacts to koalas as a result of quarry-related vehicle strike.	Planning and Environment Manager – NSW

Potential management measures that could be implemented where impacts to koalas as a result of vehicle strike are identified are provided in Section 1.6 of the Koala Management Plan. Any measures to be implemented will be selected in response to the findings of the investigation into the incident (e.g. location, contributing factors such as non-compliance with site rules, etc.) and in consultation with local koala experts and Tweed Shire Council.

### **D1-6 Potential Management Measures**

Measures that may be implemented to manage the risk of koala vehicle strike include:

- Fauna exclusion fencing
- Traffic calming measures
- Lighting
- Road verge maintenance
- Speed reduction signage.

The success of such measures will be monitored in line with Section 2 above and any specific monitoring program recommended by the appointed ecologist, reported in line with Section 3 above and further adaptive management measures progressed where impacts to koalas as a result of quarry-related vehicle strike continues to occur.

# Appendix E – Monitoring forms

ROUTINE FA	AUNA B	OX MONITORING FORM	
Location/Number of Fauna Box			
Description of Fauna Box			
Inspected by	Name	Sign	ature:
		Sign	
Inspection date			1
Element	N/Y?	Comments/description	Action Required
Empty Box			
Native Fauna Present			
Eggs Present			
Number of eggs			
Colour/description of eggs			
Nest present			
Partial nest present			
Hatchlings/fledglings present			
Box empty but scats/trace present			
Box occupied by pest species (i.e.			
bees, myna, black rat etc)			
Roof, hinges and/or supports broken			
Evidence of warping			
Evidence of vendeliers			
Evidence of vandalism			
Other comments/maintenance performe	ed?:		

FORM A: ROUTINE QUARTERLY REHABILITATION MONITORING SHEET					
General Management	Weeds	Vegetation regeneration			
Has there been a fire within the last quarter? Do the bushfire trails or adjacent	Have any areas of weeds re- established within the rehabilitation zones during the last quarter?	Natural regeneration is occurring in (record height range estimate): - Tree species			
maintenance to reduce fire risk?		- ground covers			
Is there evidence of rubbish dumping within the rehabilitation	What species?	each layer?			
Is there evidence of plant theft within	What management was undertaken to eradicate these weeds?	- Shrub			
the rehabilitation zone?		- ground covers			
Does it appear that the rehabilitation zone has been utilized for stockpiling, vehicle parking, building waste dumping,	If management was undertaken acknowledge that such was performed in accordance with	Have you noticed any new native plant species since the last monthly inspection?			
cattle grazing or person traffic?	the approved rehabilitation management plan.	If yes name the species or take a photograph			
If yes, acknowledge below what works were undertaken to rectify/restore and the date		Acknowledge that the required routine photographs have been taken within the rehabilitation zones			
Piodiversity	Modifications				
Have you spotted native fauna within the rehabilitation zone during	Have there been any structural additions (eg. new tracks, buildings) to the rehabilitation	criteria exceeded (refer Section 4.5 below)?			
inspection? If yes, what types?	zones since the last visit?	Declared Weeds? Extent of other Weeds? Survival Rate of Plants?			
Koala Kangaroo/wallaby Possums/gliders		Canopy Coverage? Tree, Small Tree & Shrub			
Small mammal (i.e. bandicoot, echidna)	remove any illegal modifications?	Groundcover Coverage? General Coverage/Success?			
Birds of prey Large nectar feeding birds (i.e. lorikeets, parrots, cockatoos) Small tree and ground birds (i.e. finches, fairy wrens, treecreepers) Glossy Black Cockatoos Other	Condition of fences - Good - Need minor repair - Poor (need replacement)	If yes, what corrective action was performed (i.e. plant showed drought stress and so watering was undertaken, plant was dead so a replacement plant was pocket planted, canopy plant coverage was not achieved so relevant pioneer plants were pocket planted).			

### FORM B - ASSESSING SITE CONDITION

PROJECT DESCRIPTION Note: where options are given, put an 'X' next to the appropriate term(s):

Project name:		Project ID:		
Site name: Si		Site ID:		
Type of on-grounds: Assisted Natural Regeneration	Years since site commenced:	When was this site last assessed?		
Current assessment conducted by:	Date of current assessment:			
Overall comments on site condition:				
Has the condition of the site changed since last assessment? YES or NO If Yes, briefly describe changes in this box, and provide details in table below.				

#### DETAILED DESCRIPTION OF SITE CONDITION Complete table quarterly, or if conditions have changed since last assessment. Also draw map.

Rating/ zone	Area (ha)	% of site	Location and factors affecting outcomes	Canopy cover (%)	Ground cover	Problem weeds	Tree survival or Recruitment	Other comments	Suggest	ed maintenance
A = OK on track towards target									(should	be routine: describe if necessary)
B = Uncertain										
significant problems									(describe)	
C = Poor major problems, likely to fail	= Poor major oblems, likely fail				be)					
Overall Condition Score (ranges from 0-100%) Multiply percentage of site occupied by each zone (A, B or C), by the condition rating for each zone (A = 1; B = 0.5; C = 0), and add the products: e.g. (70% x 1) + (20% x 0.5) + (10% x 0) = 80%					%					

MAP OF SITE CONDITION Note: also describe the condition of the site (previous page). Draw a map of the site, showing variation in outcomes as zones (Zone A = OK, Zone B = Uncertain, Zone C = Poor). Include an approximate scale (e.g. 0-100 m) and North arrow.

#### PROFORMA FOR MONITORING FOREST STRUCTURE

Project name:	Project ID:
Site name:	Site ID:
Assessed by:	Date:

#### LOCATION OF MONITORING PLOTS

Provide details and also mark on the map of the site	Plot
Location at 0 m point of plot (grid / GPS coordinates):	
Datum:	
Compass bearing / direction of transect (from 0 m point)	
Landform (e.g. plateau, crest, upper slope, mid-slope, lower slope, stream bank, floodplain)	
Slope (: e.g. flat/steep)	
Aspect (compass bearing / direction of fall of slope)	

#### MAP OF MONITORING PLOTS

In the box, insert a map of the site showing the location of monitoring plots (mark 0 m point) in relation to notable features of the site (e.g. property boundaries, roads, waterways). Also show notable features of the monitoring plots (e.g. non-standard layout, presence of remnant trees) and location of any landscape photopoints. Include a scale bar (e.g. 0-100 m) and North arrow.

Site name: Date:

GROUND COVER, CANOPY COVER and CANOPY HEIGHT For each survey plot, lay out a 50 m transect. Then survey quadrats centred on the 5 m, 25 m and 45 m points

Ground cover = proportion of ground covered by (a) vegetation within 1 m of ground (categorised by life form), (b) leaf litter and fine woody debris, (c) coarse woody debris, d) rock, (e) soil, or (f) other. At the 5 m, 25 m and 45 m points, define a 1 m x 1 m quadrat, using four 1 m sticks. Looking down at the quadrat from 1 m, estimate the % of ground covered by each type (as would be seen on a photo: total must add to 100%).

Ground Cover			Plot
Location of quadrat:	5 m	25 m	45 m

a) Ve get atio n wit hin 1 m of the gro

gro un d

5			
Grass (and sedges)	%	%	%
Herbs (soft-stemmed plants)	%	%	%
Ferns	%	%	%
Vines and scramblers	%	%	%
Tree seedlings and shrubs	%	%	%
Moss (and liverworts and lichens)	%	%	%
b) Leaf litter and fine woody debris <10 cm diameter	%	%	%
c) Coarse woody debris >10 cm diameter	%	%	%
d) Bare rock	%	%	%
e) Bare soil	%	%	%
f) Other (including tree trunks, roots, etc.)	%	%	%
TOTAL (must add up to 100%)	100%	100%	100%

Canopy (foliage) cover = projective cover of ecologically dominant layer above ground level (shade cast by foliage and stems, if the sun was overhead, assessed (approximately) above the entire 10 m x 10 m quadrat around each point. It can be estimated by eye (although this can be very subjective) or from a photo. 1. Estimate foliage cover visually, e.g. by comparison with reference photos. 2. Take a wide-angled digital photo looking up from the

centre of each 10 x 10 m quadrat, and use to calculate foliage cover). Record the number of each photo for later reference.

Canopy (foliage) cover	Plot			
Location of quadrat:	5 m	25 m	45 m	
Visual estimate of canopy (foliage) cover	%	%	%	
Canopy (foliage) cover calculated from photo	%	%	%	
Record number of canopy photo for reference				



CANOPY COVER PHOTOGRAPHS PER WALKER AND HOPKINS (1990)

Canopy height The height of the tallest tree in the canopy of each 10 m x 10 m quadrat (the canopy is the layer of foliage forming the 'roof' of the forest: it may be broken by gaps or incomplete). In some sites, it may be necessary to distinguish canopy trees from emergents: i.e. trees projecting well above the canopy with crowns exposed on all sides Note: Estimating height is difficult. Use a clinometer & tape measure, or range finder, or other measure. Alternatively, place a 2.5 m pole against a tree, & standing at a distance, estimate height in multiples of 2.5 m.

Canopy height	Plot		
Location of quadrat:		25 m	45 m
Canopy height (tallest trees in canopy)			
Height of emergent trees (if present)			

Site name:	Date:

SPECIAL LIFE FORMS: Record presence of life forms in each 10 m x 10 m quadrat centred on the 5 m, 25 m and 45 m points. If life forms are present on site, but not in quadrats, record in last column. Do not count no. of individuals.

Special Life Forms		Plot			On site?
Location of quadrat:		5 m	25 m	45 m	
Strangler figs Figs with network of roots around stem of host tree, rooted in ground					
Hemi-epiphytes Climbing plants adhering to e.g. Pothos, climbing pandanus	tree trunks, rooted in ground,				
Vines Climbing woody-stemmed plants dependent on trees for support, and rooted	Slender (stem <5 cm diam.)				
in the ground	Robust (stem >5 cm diam.)				
Vine towers Dense columns of vines growing crowns and stems	over and smothering tree				
Vine tangles Dense masses of interwoven vine midstorey	e stems in understorey or				
Thorny scramblers Thicket-forming vines or shrubs, often spiny, e.g. Calamus,	Individual plants present				
lantana, cockspur, raspberry, other vines (e.g. Eleagnus, Maesa)	Thickets present				
Palm trees Palms with stems >2 m high					
Understorey palms with stems <2 m high, e.g. walking stick palms (also includes juvenile palm trees)					
Tree ferns Ferns with stems usually >0.5 m h	igh				
Ground ferns Ferns or fern-like plants withou ground	ut stems, growing on the				
Clumping epiphytic ferns e.g. staghorns, bas	sket ferns				
Other epiphytes Growing on trees, e.g. traili on ground	ng ferns, orchids, not rooted				
Cordylines 'Palm-lilies': shrubs to 5 m high, o long leaves	occasionally branched, with				
Herbs with long wide leaves e.g. gingers, cu	njevoi, bananas				
Herbs with long strap-like leaves e.g. lilies,	mat-rush				
Cycads Plants with leathery palm-like	Stout stems, e.g. Lepidozamia				
ground (subterranean stems)	Ground cycads, e.g. Bowenia				
Pandanus Shrub / small tree with serrated st	rap-like leaves				
Other life forms: describe					

Woody debris = fallen logs and branches lying on or within 1 m of the ground.									
Tally the number of times logs are intercepted by the 50 m transect, by diameter class at the point of intersection. If a log is intercepted by the transect more than once, it is tallied each time, by diameter at each of the points of intersection									
Tally intercepts with fallen logs by diameter	Fine woody debris <10 cm dia		Coarse woody debris (CWD) > 10 cm diameter						
class on each transect	2.5-5 cm	5-10 cm	10-20	20-30	30-40	40-50	50-75	75- 100	>10 0
50 m transect									

### FORM D: PROFORMA FOR MONITORING FLORISTIC COMPOSITION

Project name:	Project ID:
Site name:	Site ID:
Assessed by:	Date:

#### LOCATION OF MONITORING PLOTS

Provide details and also mark on the map of the site	Plot
Location at 0 m point of plot (grid / GPS coordinates):	
Datum:	
Compass bearing / direction of transect (from 0 m point)	
Landform (e.g. plateau, crest, upper slope, mid-slope, lower slope, stream bank, floodplain)	
Slope (: e.g. flat/steep)	
Aspect (compass bearing / direction of fall of slope)	

#### MAP OF MONITORING PLOTS

In the box, insert a map of the site showing the location of monitoring plots (mark 0 m point) in relation to notable features of the site (e.g. property boundaries, roads, waterways). Also show notable features of the monitoring plots (e.g. non-standard layout, presence of remnant trees) and location of any landscape photopoints. Include a scale bar (e.g. 0-100 m) and North arrow.

Site name:	Date:

GENERAL COMMENTS on the composition of vegetation at the site (e.g. dominant or notable species, variation across the site): record by strata as follows:

Canopy/ Ecologically Dominant Layer:

Midstorey:

Understorey/ Ground cover:

RECRUITMENT: What species are common recruits to the site? Any other comments about recruitment?

Does this site have any WEED or MAINTENANCE ISSUES that need attention?

Any other comments on the site? Mark an 'X' here \_\_\_\_\_and add extra page(s) as required.

### GHD

230 Harbour Drive Coffs Harbour NSW 2450 T: 61 2 6650 5600 F: 61 2 6650 5601 E: cfsmail@ghd.com

© GHD 2021

This document is and shall remain the property of GHD. The document may only be used for the purpose for which it was commissioned and in accordance with the Terms of Engagement for the commission. Unauthorised use of this document in any form whatsoever is prohibited.

22-SO-976847825-

9/https://projects.ghd.com/oc/Newcastle3/holcimdunloesandquar/Delivery/Documents/2220056\_RPT\_ Dunloe Landscape Management Plan.docx

Document	Status
----------	--------

Revision	Author	Reviewer		Approved fo	Approved for Issue		
		Name	Signature	Name	Signature	Date	
0	Planit					January 2009	
1	Planit					February 2009	
2	Holcim					October 2016	
3	Holcim					July 2018	
4	B Luffman	S Lawer		S Lawer		23/09/2019	
5	B Luffman	S Lawer		S Lawer		04/06/2021	
6	B Luffman	S Lawer	fan	S Lawer	fan	02/07/2021	

# www.ghd.com



**Appendix F** – Noise management plan



# Holcim (Australia) Pty Ltd

Dunloe Sand Quarry Noise Management Plan

July 2020

This report: has been prepared by GHD for Holcim (Australia) Pty Ltd and may only be used and relied on by Holcim (Australia) Pty Ltd for the purpose agreed between GHD and the Holcim (Australia) Pty Ltd as set out in Section 1.1 of this report.

GHD otherwise disclaims responsibility to any person other than Holcim (Australia) Pty Ltd arising in connection with this report. GHD also excludes implied warranties and conditions, to the extent legally permissible.

The services undertaken by GHD in connection with preparing this report were limited to those specifically detailed in the report and are subject to the scope limitations set out in the report.

The opinions, conclusions and any recommendations in this report are based on conditions encountered and information reviewed at the date of preparation of the report. GHD has no responsibility or obligation to update this report to account for events or changes occurring subsequent to the date that the report was prepared.

The opinions, conclusions and any recommendations in this report are based on assumptions made by GHD described in this report. GHD disclaims liability arising from any of the assumptions being incorrect.

GHD has prepared this report on the basis of information provided by Holcim (Australia) Pty Ltd and others who provided information to GHD (including Government authorities)], which GHD has not independently verified or checked beyond the agreed scope of work. GHD does not accept liability in connection with such unverified information, including errors and omissions in the report which were caused by errors or omissions in that information.

# **Table of contents**

1.	Introd	duction	1
	1.1	Objectives	1
	1.2	Targets	1
	1.3	Consultation	1
2.	Envir	ronmental requirements	2
	2.1	Legislation	2
	2.2	Guidelines	2
	2.3	Conditions of approval	2
	2.4	Environment protection licence	4
3.	Existi	ing environment and impacts	5
	3.1	The site and sensitive receivers	5
	3.2	Operational noise impact	9
	3.3	Traffic noise impacts	10
4.	Oper	ating criteria	11
	4.1	Noise criteria	11
	4.2	Hours of operation	11
5.	Envir	ronmental control measures	12
6.	Monit	toring and reporting	14
	6.1	Environmental inspections	14
	6.2	Noise monitoring	14
	6.3	Contingency plan	14
	6.4	Reporting	1
7.	Revie	ew and improvement	2

# **Table index**

Table 2-1	Consent conditions relevant to the NMP	2
Table 2-2	EPL conditions relevant to the NMP	4
Table 3-1	Identified noise sensitive receivers (from site operations)	5
Table 3-2	Identified noise sensitive receivers (along Pottsville Road)	5
Table 3-3	Noise levels at sensitive receivers from onsite operation	9
Table 3-4	Noise levels at sensitive receivers from Quarry Road operation	9
Table 3-5	Noise levels at sensitive receivers from truck operation along Pottsville Road	10
Table 4-1	Noise criteria	11
Table 4-2	Operating hours	11
Table 5-1	Environmental controls and mitigation measures	12

Table 6-1	Contingency plan	.14
-----------	------------------	-----

# **Figure index**

Figure 3-1	Location of site, sensitive receivers and predicted impact	6
Figure 3-2	Sensitive receivers and predicted impact from trucks on quarry access road	7
Figure 3-3	Sensitive receivers and predicted impact along Pottsville Road	8

# **Appendices**

Appendix A – Agency consultation

# 1. Introduction

This Noise Management Plan (NMP) forms part of the Environmental Management Strategy (EMS) for Dunloe Sand Quarry. This NMP has been prepared to meet the requirements of the Minister's Conditions of Approval (CoA) outlined in Development Consent No. 06\_0030, the mitigation measures outlined in MOD2 (GHD 2017), the Environmental Impact Statement (EIS) (Planit 2007), the Environment Protection Licence 13077 (EPL) and relevant legislation.

### 1.1 **Objectives**

The key objective of the NMP is to ensure appropriate controls and procedures are implemented in order to minimise the noise impacts to the local community and the built environment.

To achieve this objective, Holcim will undertake the following:

- Identify sensitive receivers and ensure appropriate environmental controls and procedures are implemented during operational activities.
- Minimise potential adverse noise impacts to the environment and community.
- Manage impacts if they occur through a systematic analysis of mitigation strategies.
- Ensure appropriate measures are implemented to address the relevant CoA outlined in Table 2-1.
- Ensure appropriate measures are implemented to comply with all relevant legislation and other requirements as described in Section 2.1 of this NMP.

### **1.2 Targets**

The following targets have been established for the management of noise during the operational lifetime of Dunloe Sand Quarry:

- Ensure full compliance with the relevant legislative requirements and CoA.
- No exceedance of the operational noise limits.
- No justified complaints from adjacent residents in relation to noise generation.
- No out of hours work.

### **1.3** Consultation

Extensive consultation was undertaken with the local community during preparation of the EIS and MOD2. Any concerns identified by relevant stakeholders were addressed in the EIS and MOD2 mitigation measures which have been incorporated into this NMP.

As per CoA 5(b), Schedule 3, the Environment Protection Authority (EPA) were consulted in relation to the NMP. Evidence of the consultation is provided in Appendix A.

# 2. Environmental requirements

### 2.1 Legislation

Legislation relevant to noise management includes:

- Protection of the Environment Operations Act 1997
- Protection of the Environment Operations (Noise Control) Regulation 2017

Further discussion of the above legislation is provided in the EMS, as well as the MOD2 and the EIS.

### 2.2 Guidelines

The following guidelines have been consulted during development of this NMP:

• Noise Policy for Industry (EPA 2017)

### 2.3 Conditions of approval

The consent conditions relevant to this NMP are listed in Table 2-1. A cross reference is also included to indicate where the condition is addressed in this NMP or other environmental management documents.

Condition No.	Requirement			Reference
Schedule 3, Condition 2	The Proponent must ensure that the noise generated by the project does not exceed the criteria in Table 1 at any residence on privately-owned land.			Section 5
	Table 1: Noise Impact Assessmen	t Criteria	5(4)	
	Receiver Location	Day LAeq (15 min)	Day LAeq (15 min) dB(A)	
	R6 and R7	42		
	R8	48		
	All other residences	41		
	Noise generated by the project must be monitored and measured in accordance with the relevant procedures and exemptions (including certain meteorological conditions) of the <i>NSW Noise Policy for Industry</i> (EPA, 2017).			
	The noise criteria in Table 1 do agreement with the owner/s of exceed the noise criteria, and t Department in writing of the ter			
Schedule 3, Condition 3	The Proponent must comply with the operating hours in Table 2.			Section 5
	Activity	Day	Time	
	Sand extraction and processing, delivery and distribution, and other quarry related activities	Monday – Friday Saturday Sunday and Public Holidays	7:00 am to 5:00 pm 7:00 am to 12:00 pm Nil	
	Maintenance (if inaudible at neighbouring residences)	Any day	Any time	
Schedule 3, Condition 4	The Proponent must: (a) implement best practice management to minimise the construction, operational and road transportation noise of the development;		Section 5	
	(b) minimise the noise impacts of the project during meteorological conditions when the noise criteria in this approval do not apply;			Section 5

#### Table 2-1 Consent conditions relevant to the NMP

Condition No.	Requirement	Reference
	(c) carry out attended noise monitoring (at least every 3 months or as otherwise agreed by the Secretary) to determine whether the project is complying with the relevant conditions of this approval; and	Section 6
	<ul> <li>(d) regularly assess noise monitoring data and modify and/or stop operations on site to ensure compliance with the relevant conditions of this approval to the satisfaction of the Secretary.</li> <li>Note: Monitoring under this approval is not required at all residences and the use of representative monitoring locations can be used to demonstrate compliance with criteria, if agreed to by the Secretary.</li> </ul>	Section 5
Schedule 3, Condition 5	Within three months of the approval of Modification 2, the Proponent must prepare a Noise Management Plan for the project to the satisfaction of the Secretary. This plan must:	This plan
	<ul> <li>(a) be prepared by a suitably qualified and experienced person/s whose appointment has been endorsed by the Secretary;</li> </ul>	
	(b) be prepared in consultation with the EPA;	Appendix A
	<ul> <li>(c) describe the measures to be implemented to ensure:</li> <li>(i) compliance with the noise criteria and operating conditions in this approval;</li> <li>(ii) best practice management is being employed;</li> <li>(iii) noise impacts of the project are minimised during noise-enhancing meteorological conditions;</li> </ul>	Section 5
	(d) describe the noise management system; and	Section 5
	<ul> <li>(e) include a monitoring program to be implemented to measure noise from the project against the noise criteria in Table 1, and which evaluates and reports on the effectiveness of the noise management system on site.</li> <li>The Proponent must implement the Noise Management Plan as approved by the Secretary.</li> </ul>	Section 6
Schedule 5, Condition 1A	The Proponent must ensure that the management plans required under this approval are prepared in accordance with any relevant guidelines, and include:	
	(a) a summary relevant background or baseline data;	Section 3
	<ul> <li>(b) a description of:</li> <li>the relevant statutory requirements (including any relevant approval, licence or lease conditions);</li> <li>any relevant limits or performance measures/criteria; and the specific performance indicators that are proposed to be used to judge the performance of, or guide the implementation of, the project or any management measures;</li> </ul>	Section 1.2 and Section 2
	(c) a description of the measures that to be implemented to comply with the relevant statutory requirements, limits, or performance measures/criteria;	Section 5
	<ul> <li>(d) a program to monitor and report on the:</li> <li>impacts and environmental performance of the project; and effectiveness of any management measures (see (c) above);</li> </ul>	Section 6
	(e) a contingency plan to manage any unpredicted impacts and their consequences and to ensure that ongoing impacts reduce to levels below relevant impact assessment criteria as quickly as possible;	Section 6.3
	(f) a program to investigate and implement ways to improve the environmental performance of the project over time	Section 7
	<ul> <li>(g) a protocol for managing and reporting any:</li> <li>incidents;</li> <li>complaints;</li> <li>non-compliances with statutory requirements; and exceedances of the impact assessment criteria and/or performance criteria; and</li> </ul>	Refer to the EMS
	(h) a protocol for periodic review of the plan.	Section 7

### 2.4 Environment protection licence

The EPL conditions, relevant to this NMP, are listed in Table 2-2. A cross reference is also included to indicate where the condition is addressed in this NMP or other environmental management documents.

Condition No.	Requirement	Reference
L4.1	Noise from the premises must not exceed an LAeq (15 minute) noise emission criterion of 48 dB(A), except as expressly provided by this licence.	Section 4
L4.2	Noise from the premises is to be measured at the most affected point within the residential boundary, or at the most affected point within 30 metres of the dwelling where the dwelling is more than 30 metres from the boundary, to determine compliance with this condition.	Section 6
L4.3	Where it can be demonstrated that direct measurement of noise from the premises is impractical, the EPA may accept alternative means of determining compliance. See Chapter 11 of the NSW Industrial Noise Policy January 2000 for general guidance on determining compliance.	Section 6
L4.4	The noise emission limits identified in this section apply under calm meteorological conditions of wind speeds u pto 3m/s at 10 metres above ground level, and temperature inversion conditions.	Section 5
L5.1	Activities covered by this licence must only be carried out between the hours of 0700 hrs to 1700 hrs Monday to Friday, and 0700 hrs to 1200 hrs Saturday, and at no time on Sundays and Public Holidays.	Section 4.2
L5.2	<ul> <li>Exemptions to standard construction hours</li> <li>a) Any maintenance work that is inaudible from neighbouring residences;</li> <li>b) Emergency work to avoid the loss of lives or property, and/or to prevent environmental harm.</li> </ul>	Section 4.2

### Table 2-2 EPL conditions relevant to the NMP

# **Existing environment and impacts**

### 3.1 The site and sensitive receivers

The site is located within the Tweed Local Government Area, approximately 25 kilometres south of the Tweed Heads town centre. Road access to the site is via Pottsville Road from the north.

The nearest identified sensitive receivers located in the vicinity of the site are detailed in Table 3-1 and shown in Figure 3-1 below. Distances are stated from the receiver to the nearest point at the site boundary.

Receiver	Receiver type	Address	Distance from site activity (metres)	Direction from site
R1	Residential	265 Warwick Park Road	1,030	South
R2	Residential	265 Warwick Park Road	1,060	South
R3	Residential	200 Warwick Park Road	1,070	Southwest
R4	Residential	200 Warwick Park Road	1,060	Southwest
R5	Residential	175 Warwick Park Road	960	Southwest
R6	Residential	157 Warwick Park Road	970	Southwest
R7	Residential	129 Warwick Park Road	1,090	West
R8	Residential	679 Pottsville Road	1,720	Northwest

### Table 3-1 Identified noise sensitive receivers (from site operations)

Sensitive receivers were also identified along Pottsville Road, which will be used to access the quarry. These are detailed in Table 3-2 and Figure 3-2 and Figure 3-3 below.

Receiver	Receiver type	Address	Distance from Pottsville Road (metres)
R9	Residential	765 Pottsville Road	96
R10	Residential	771 Pottsville Road	14
R11	Residential	834 Pottsville Road	56
R12	Residential	854 Pottsville Road	46
R13	Residential	866 Pottsville Road	10
R14	Residential	883 Pottsville Road	14
R15	Residential	883 Pottsville Road	16
R16	Residential	940 Pottsville Road	79
R17	Residential	943 Pottsville Road	104
R18	Residential	56 Hazelwood Drive	164
R19	Residential	943 Pottsville Road	69
R20	Residential	940 Pottsville Road	51
R21	Residential	3 Hazelwood Drive	27

#### Table 3-2 Identified noise sensitive receivers (along Pottsville Road)



N:\AU\Sydney\Projects\22118823\GIS\Maps\Deliverables\22\_18823\_Z003\_NoiseFigures.mxd © 2017. Whilst every care has been taken to prepare this map. GHD (and Sixmaps, NSW Land and Property Information) make no represen and responsibility of any kind (whether in contract, tort or otherwise) for any expenses, losses, damages and/or costs (including indirect or cor in any way and for any reason. y, completeness or suitability for any particular purpose and cannot accept liability ed by any party as a result of the map being inaccurate, incomplete or unsuitable ne) which are incurr

Data source: Aerial Imagery: Sixmaps (2017 - NSW LPI), General Topo: NSW DTDB 2012. Created by:mking3



N:\AU\Sydney\Projects\22\18823\GIS\Maps\Deliverables\22\_18823\_Z003\_NoiseFigures.mxd © 2017. Whilst every care has been taken to prepare this map. GHD (and Sixmaps, NSW Land and Property Information) make no represer and responsibility of any kind (whether in contract, tort or otherwise) for any expenses, losses, damages and/or costs (including indirect or co in any way and for any reason. ties about its accuracy, reliability, completeness or suitability for any particular purpose and cannot accept liability ige) which are or may be incurred by any party as a result of the map being inaccurate, incomplete or unsuitable ential dama

Data source: Aerial Imagery: Sixmaps (2017 - NSW LPI), General Topo: NSW DTDB 2012. Created by:mking3



N:\AU\Sydney\Projects\22\18823\GIS\Maps\Deliverables\22\_18823\_Z003\_NoiseFigures.mxd © 2017. Whilst every care has been taken to prepare this map. GHD (and Sixmaps, NSW Land and Property Information) make no represer and responsibility of any kind (whether in contract, tort or otherwise) for any expenses, losses, damages and/or costs (including indirect or co in any way and for any reason. ies about its accuracy, reliability, completeness or suitability for any particular purpose and cannot accept liability ge) which are or may be incurred by any party as a result of the map being inaccurate, incomplete or unsuitable

Data source: Aerial Imagery: Sixmaps (2017 - NSW LPI), General Topo: NSW DTDB 2012. Created by:mking3

## 3.2 Operational noise impact

### 3.2.1 Site operation

The site operation, consisting of plant and equipment within the quarry, was modelled and the resulting noise levels at sensitive receivers is shown in Table 3-3 below. Results show that the proposed scenario of twelve trucks per hour will comply with the noise criteria of 48 dBA at all sensitive receivers. A noise emission map for the site has also been provided in Figure 3-1 below.

Receiver	Sound Pressure Level LAeq(15min) dB(A)	Compliance
R1	40	Yes
R2	38	Yes
R3	36	Yes
R4	36	Yes
R5	41	Yes
R6	42	Yes
R7	42	Yes
R8	31	Yes

### Table 3-3 Noise levels at sensitive receivers from onsite operation

### 3.2.2 Quarry road operation

Operation of the quarry access road was assessed against the noise limits provided in the original project approval. Truck volumes along the quarry access road are expected to increase from the current scenario of four trucks per hour (8 movements) to twelve trucks per hour (24 movements). The expected truck volume of twelve trucks per hour (24 movements) has been used to assess noise from operation of the quarry access road. The resulting noise levels at each receiver are shown below in Table 3-4.

Results show that the proposed scenario of twelve trucks per hour on the quarry access road will comply with the noise criteria of 48 dBA at all sensitive receivers. Noise emission maps for the current and proposed cases have been provided for the Quarry Road in Table 3-4 and Figure 3-2.

Receiver	Proposed development twelve trucks per hour L <sub>Aeq(15min)</sub> dB(A)	Criteria, dBA	Compliance
R1	26	48	Yes
R2	27	48	Yes
R3	24	48	Yes
R4	24	48	Yes
R5	29	48	Yes
R6	31	48	Yes
R7	31	48	Yes
R8	46	48	Yes

#### Table 3-4 Noise levels at sensitive receivers from Quarry Road operation

### 3.3 Traffic noise impacts

The impact on traffic levels and resulting noise at sensitive receivers along Pottsville Road was modelled using SoundPLAN as outlined above. The noise impact was assessed against the noise criteria for a sub-arterial road of 60 dBA  $L_{Aeq(15hr)}$ . The resulting noise levels at sensitive receivers are shown in Table 3-5. A noise emission map for Pottsville Road is also provided in Figure 3-3.

Receiver	Proposed development LAeq(15hr) dBA	Criteria, dBA	Compliance
R9	52	60	Yes
R10	57	60	Yes
R11	45	60	Yes
R12	49	60	Yes
R13	51	60	Yes
R14	56	60	Yes
R15	48	60	Yes
R16	44	60	Yes
R17	50	60	Yes
R18	42	60	Yes
R19	54	60	Yes
R20	48	60	Yes
R21	38	60	Yes

# Table 3-5 Noise levels at sensitive receivers from truck operation along Pottsville Road
### 4. Operating criteria

#### 4.1 Noise criteria

Development Consent No. 06\_0030 originally had a noise criteria of 48 dBA at all sensitive receivers. EPL 13077 also has a noise criteria of 48 dBA. On the basis of the Noise Impact Assessment (GHD 2017) issued to support MOD2, the noise criteria were updated, as provided in Table 4-1.

#### Table 4-1 Noise criteria

Receiver Location	Day LAeq (15 min) dB(A)
R6 and R7	42
R8	48
All other residences	41

Noise generated by the project must be monitored and measured in accordance with the relevant procedures and exemptions (including certain meteorological conditions) of the *NSW Noise Policy for Industry* (EPA, 2017).

The noise criteria in Table 4-1 do not apply if Holcim has an agreement with the owner/s of the relevant residence or land to exceed the noise criteria, and Holcim has advised the DPI&E in writing of the terms of this agreement.

#### 4.2 Hours of operation

Holcim are required to comply with the operating hours in Table 4-2.

#### **Table 4-2 Operating hours**

Activity	Day	Time
Sand extraction and	Monday-Friday	7:00 am to 5:00 pm
distribution, and other quarry	Saturday	7:00 am to 12:00 pm
related activities	Sunday and Public Holidays	Nil
Maintenance (if inaudible at neighbouring residences)	Any day	Any time

### 5. Environmental control measures

Environmental requirements and control measures are identified in the CoA, EIS, MOD2 and EPL. Specific measures and requirements to address noise impacts are outlined in Table 5-1.

#### Ref **Environmental Management Measure** Responsibility Timing N1 During the induction, all personnel on site should Operation Quarry be made aware of the potential for noise impacts Manager and should aim to minimise impact or elevated noise levels, where possible. N2 The operating hours specified in the Operation Quarry development consent and in Table 4-2 must be Manager complied with. Noise generated by the development must not N3 Operation Quarry exceed the criteria in Table 4-1. Manager N4 Excessively noisy machinery or activities must Operation Quarry cease during meteorological conditions when the Manager noise criteria in the consent do not apply e.g. high winds (refer NSW Noise Policy for Industry (EPA, 2017)). N5 Prescribed buffer zones around the extraction Operation Quarry ponds to be planted and maintained. Manager N6 Noise monitoring must be carried out at least Operation Quarry every 3 months or as otherwise agreed with the Manager Secretary to determine whether the development is complying with the relevant conditions of the consent and the adopted noise criteria. N7 Noise monitoring will also be conducted following Operation Quarry any change in operating conditions that are likely Manager to increase noise emissions from the site (such as a sudden increase in production rate or heavy vehicle movements) or move noise sources significantly closer to noise sensitive receivers. N8 Compliance noise monitoring will also be Operation Quarrv undertaken following receipt of a justified Manager complaint relating to noise emissions from the site. N9 The noise monitoring data will be regularly Operation Quarry assessed and on site operations will be modified Manager or stopped to ensure compliance with the relevant conditions of the consent. For example, dredge to be fitted with suitable mufflers, truck speed reduced. Where practical, machines will be operated at N10 Operation Quarry low speed or power and switched off when not Manager being used rather than left idling for prolonged periods. N11 Keep truck drivers informed of designated Operation Quarry vehicle routes, parking locations and delivery Manager hours. N12 Avoid dropping materials from height and avoid Operation Quarry metal to metal contact on material. Manager N13 All engine covers would be kept closed while Operation Quarry equipment is operating. Manager

#### **Table 5-1 Environmental controls and mitigation measures**

Ref	Environmental Management Measure	Timing	Responsibility
N14	Vehicles will be kept properly serviced and fitted with appropriate mufflers. The use of exhaust brakes will be eliminated, where practicable.	Operation	Quarry Manager
N15	Machines found to produce excessive noise compared to industry best practice will be removed from the site or stood down until repairs or modifications can be made.	Operation	Quarry Manager
N16	The Quarry Manager is to erect a sign at the entrance of the quarry with a phone number and permanent site contact so that noise complaints can be received and addressed in a timely manner.	Operation	Quarry Manager

### 6. Monitoring and reporting

#### 6.1 Environmental inspections

Routine weekly inspections by the Quarry Manager (or delegate) will occur throughout the operational lifetime of the quarry to identify any ad-hoc noise issues such as faulty equipment, noisy works, using the *Environmental Inspection Checklist* in the Environmental Monitoring and Management Plan.

#### 6.2 Noise monitoring

Attended noise compliance monitoring will occur at R6, R7 and R8 (refer to Figure 3-1) quarterly for the first two years following approval of this NMP. If this monitoring indicates compliance with the criteria in Table 4-1 and Department of Planning, Industry and Environment (DPI&E) approve, the noise monitoring will cease unless there is:

- A justifiable noise complaint in relation to Dunloe Sand Quarry operations
- A change in operating conditions that are likely to increase noise emissions from the site

The assessment must be conducted by a suitably qualified and experienced acoustic consultant in accordance with the *Noise Policy for Industry* (EPA 2017).

#### 6.3 Contingency plan

If the above monitoring detects an impact or there is a justified community, noise related, complaint in relation to Dunloe Sand Quarry operations, a contingency plan or trigger and response plan is to be implemented, as described in Table 6-1.

In general, all issues will be investigated and corrective actions determined within 24 hours. The timeframe to implement the corrective actions will depend on the risk and consequence of the issue. The nature of the corrective action will also influence the implementation timeframe.

#### Table 6-1 Contingency plan

Trigger	Response
Excessively noisy equipment identified	<ul> <li>Quarry Manager to commission testing of equipment for faults to confirm excessive noise generation.</li> <li>Quarry Manager to take plant or equipment out of service and maintain or modify (if practicable).</li> <li>Quarry Manager to replace faulty plant or equipment with plant or equipment that is in good working order and recommence activities.</li> </ul>
Non-compliance with noise criteria	<ul> <li>Identify the noise source that has caused the exceedance.</li> <li>Reassess the mitigation measures employed at the site to reduce the impact of the noise source.</li> <li>Following the adoption of noise mitigation, conduct further noise monitoring to ensure the success of the mitigation measure.</li> </ul>
Justified noise related complaint	<ul> <li>Consult complainant to determine time and source of noise.</li> <li>Offer noise monitoring.</li> <li>If monitoring indicates the complaint is justified and relates to Dunloe Sand Quarry operations, reassess the mitigation measures employed at the site to reduce the impact of the noise source.</li> <li>Following the adoption of noise mitigation, conduct further noise monitoring to ensure the success of the mitigation measure.</li> <li>Provide details of the response and noise monitoring to the complainant.</li> </ul>

#### 6.4 Reporting

The general reporting requirements are described in the EMS. In relation to the noise monitoring, the routine noise monitoring will be recorded on the *Environmental Inspection Checklist* in the Environmental Monitoring and Management Plan.

A report will be prepared by the acoustic consultant following the compliance monitoring. This is to include, as a minimum:

- The date(s) of the monitoring
- The time(s) of the monitoring
- The location of the monitoring
- The activities occurring during the monitoring
- A comparison of the results with the adopted noise criteria

If an exceedance of the criteria is recorded, the affected resident and Department of Planning, Industry and Environment (DPI&E) will be notified in writing and provided with regular monitoring results until the results show that the project is complying with the relevant criteria.

A summary of these results will be presented in the Annual Report (refer to the EMS). All records will be:

- Maintained in a legible form
- Kept for at least 4 years
- Produced to any authorised officer of the EPA and/or DPI&E upon request

### 7. Review and improvement

Continuous improvement of this NMP will be achieved by reviewing the plan in accordance with the EMS and through the ongoing evaluation of environmental management performance against environmental policies, objectives and targets.

The continuous improvement process is designed to:

- Identify areas of opportunity for improvement of environmental management and performance.
- Determine the cause or causes of non-conformances and deficiencies.
- Develop and implement a plan of corrective and preventative action to address any nonconformances and deficiencies.
- Verify the effectiveness of the corrective and preventative actions.
- Document any changes in procedures resulting from process improvement; and make comparisons with objectives and targets.

### Appendices

Appendix A – Agency consultation

#### **Ben Luffman**

From: Sent: To: Subject:	Geff Cramb <geff.cramb@epa.nsw.gov.au> Wednesday, 21 August 2019 11:20 AM Ben Luffman RE: Dunloe Quarry Management Plan consultation</geff.cramb@epa.nsw.gov.au>
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
<b>RepoType:</b>	Job

Dear Ben

The EPA do not intend to review and provide comment upon the management plans. The EPA are content with the scope. However, it is understood that EPA will undertake compliance reviews against the requirements of the Environment Protection Licence issued and the implementation of the management plan at our discretion.

Regards Geff

#### **Geff Cramb**

Operations Officer – Environment Management Unit North Coast, NSW Environment Protection Authority

+61 2 6640 2510

geff.cramb@epa.nsw.gov.au www.epa.nsw.gov.au ♥@EPA NSW Report pollution and environmental incidents 131 555 (NSW only) or +61 2 9995 5555



I work flexibly. I'm sending this message now because it's a good time for me, but I don't expect that you will read, respond to or action it outside of your own regular hours.

From: Ben Luffman <<u>Ben.Luffman@ghd.com</u>>
Sent: Thursday, 8 August 2019 12:51 PM
To: Peter Lynch <<u>Peter.Lynch@epa.nsw.gov.au</u>>
Cc: Janelle Bancroft <<u>Janelle.Bancroft@epa.nsw.gov.au</u>>; Victoria Musgrove
<<u>victoria.musgrove@lafargeholcim.com</u>>
Subject: Dunloe Quarry Management Plan consultation

#### Hi Peter,

Not sure if you are the correct person to contact but we have updated the management plans for Dunloe Quarry following the recent approval of MOD2. The conditions of the Project Approval – SSD 06\_0030 require a number of the plans to be prepared in consultation with the EPA. We have therefore attached the relevant plans for review.

The updates have mainly been a reformatting to remove duplication and inclusion of additional information to address the new requirements of the conditions.

We would appreciate your comments by 23 August 2019.

Please contact me if you have any questions.

#### Regards

#### Ben Luffman | A GHD Associate

B.App.Sc. (Hons) | Grad.Dip. Urban and Regional Planning | Environmental Auditor **Technical Director - Environment** 

#### GHD

#### Proudly employee owned

T: +61 2 6650 5613 | M: +61 415 271 319 | E: <u>ben.luffman@ghd.com</u> 230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.qhd.com</u>



WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION

GHD acknowledges the Traditional Owners of Country throughout Australia. We pay respect to their continuing culture and Elders past, present and emerging. <u>Click here</u> to learn about our Reconciliation Action Plan.

Please consider the environment before printing this email 1 ream of paper = 6% of a tree / 5.4kg CO2 in the atmosphere | 3 sheets of A4 paper = 1 litre of water

CONFIDENTIALITY NOTICE: This email, including any attachments, is confidential and may be privileged. If you are not the intended recipient please notify the sender immediately, and please delete it; you should not copy it or use it for any purpose or disclose its contents to any other person. GHD and its affiliates reserve the right to monitor and modify all email communications through their networks.

\_\_\_\_\_

This email is intended for the addressee(s) named and may contain confidential and/or privileged information.

If you are not the intended recipient, please notify the sender and then delete it immediately.

Any views expressed in this email are those of the individual sender except where the sender expressly and with authority states them to be the views of the Environment Protection Authority.

PLEASE CONSIDER THE ENVIRONMENT BEFORE PRINTING THIS EMAIL

This e-mail has been scanned for viruses

GHD

230 Harbour Drive Coffs Harbour NSW 2450 T: 61 2 6650 5600 F: 61 2 6650 5601 E: cfsmail@ghd.com

#### © GHD 2020

This document is and shall remain the property of GHD. The document may only be used for the purpose for which it was commissioned and in accordance with the Terms of Engagement for the commission. Unauthorised use of this document in any form whatsoever is prohibited. 22-SO-976847825-

3/https://projects.ghd.com/oc/Newcastle3/holcimdunloesandquar/Delivery/Documents/2220056\_RP T\_Dunloe Noise Management Plan.docx

**Document Status** 

Revision	Author	Reviewer Approved		Approved for	r Issue	
		Name	Signature	Name	Signature	Date
0	B Luffman	S Lawer	0	S Lawer	$\rho$	23/09/2019
1	B Luffman	S Lawer	Ja-	S Lawer	ta	17/07/2020

# www.ghd.com



**Appendix G** – Air quality management plan



## Holcim (Australia) Pty Ltd Dunloe EMP

Air Quality Management Plan

February 2020

This report: has been prepared by GHD for Holcim (Australia) Pty Ltd and may only be used and relied on by Holcim (Australia) Pty Ltd for the purpose agreed between GHD and the Holcim (Australia) Pty Ltd as set out in this report.

GHD otherwise disclaims responsibility to any person other than Holcim (Australia) Pty Ltd arising in connection with this report. GHD also excludes implied warranties and conditions, to the extent legally permissible.

The services undertaken by GHD in connection with preparing this report were limited to those specifically detailed in the report and are subject to the scope limitations set out in the report.

The opinions, conclusions and any recommendations in this report are based on conditions encountered and information reviewed at the date of preparation of the report. GHD has no responsibility or obligation to update this report to account for events or changes occurring subsequent to the date that the report was prepared.

The opinions, conclusions and any recommendations in this report are based on assumptions made by GHD described in this report. GHD disclaims liability arising from any of the assumptions being incorrect.

GHD has prepared this report on the basis of information provided by Holcim (Australia) Pty Ltd and others who provided information to GHD (including Government authorities)], which GHD has not independently verified or checked beyond the agreed scope of work. GHD does not accept liability in connection with such unverified information, including errors and omissions in the report which were caused by errors or omissions in that information.

### **Table of contents**

1.	Intro	duction1
	1.1	Objectives1
	1.2	Targets1
	1.3	Consultation1
2.	Envir	onmental requirements
	2.1	Legislation2
	2.2	Guidelines2
	2.3	Conditions of approval2
	2.4	Environment protection licence4
3.	Exist	ing environment and impacts5
	3.1	Existing environment5
	3.2	Predicted impacts7
4.	Oper	ating criteria9
	4.1	Air quality criteria9
5.	Envir	onmental control measures10
6.	Moni	toring and reporting12
	6.1	Environmental inspections12
	6.2	Dust monitoring12
	6.3	Contingency plan12
	6.4	Reporting13
7.	Revie	ew and improvement14

### **Table index**

Table 2-1	Consent conditions relevant to the AQMP	2
Table 2-2	EPL conditions relevant to the NMP	4
Table 3-1	Identified air quality sensitive receivers	5
Table 3-2	Dust deposition sampling results for 2016	5
Table 3-3	Predicted maximum 24 hour PM <sub>10</sub> concentration from trucks µg/m <sup>3</sup>	7
Table 4-1	Air quality criteria	9
Table 5-1	Environmental controls and mitigation measures	10
Table 6-1	Contingency plan	12

### **Figure index**

Figure 3-1	Site location and sensitive receivers	.6
Figure 3-2	Wind rose for Coolangatta	.7

### **Appendices**

Appendix A – Agency consultation

### 1. Introduction

This Air Quality Management Plan (AQMP) forms part of the Environmental Management Strategy (EMS) for Dunloe Sand Quarry. This AQMP has been prepared to meet the requirements of the Minister's Conditions of Approval (CoA) outlined in Development Consent No. 06\_0030, the mitigation measures outlined in MOD2 (GHD 2017), the Environmental Impact Statement (EIS) (Planit 2007), the Environment Protection Licence 13077 (EPL) and relevant legislation.

#### 1.1 Objectives

The key objective of the AQMP is to ensure appropriate controls and procedures are implemented in order to minimise the air quality impacts to the local community and the built environment.

To achieve this objective, Holcim will undertake the following:

- Ensure appropriate controls and procedures are implemented during the operation of the quarry to avoid or minimise dust generation, air quality impacts and potential adverse impacts to sensitive receivers.
- Ensure appropriate measures are implemented to address the relevant CoA outlined in Table 2-1.
- Ensure appropriate measures are implemented to comply with all relevant legislation and other requirements as described in Section 2.1.

### **1.2 Targets**

The following targets have been established for the management of air quality during the operational lifetime of Dunloe Sand Quarry:

- Minimise and manage potential air quality/dust impacts from the development in accordance with relevant legislative requirements and CoA.
- Control dust and exhaust emissions of plant and equipment from quarrying activities.
- Achieve particulate matter and dust concentrations that meet the approved air quality criteria.
- No visible offsite dust emissions as a result of site operations.
- No justifiable complaints related to air quality attributable to site operations.

#### **1.3** Consultation

Extensive consultation was undertaken with the local community during preparation of the EIS and MOD2. Any concerns identified by relevant stakeholders were addressed in the EIS and MOD2 mitigation measures, which have been incorporated into this AQMP.

As per CoA 7A(b), Schedule 3, the Environment Protection Authority (EPA) were consulted in relation to the AQMP. Evidence of the consultation is provided in Appendix A.

### 2. Environmental requirements

#### 2.1 Legislation

Legislation relevant to air quality management includes:

- Protection of the Environment Operations Act 1997
- Protection of the Environment Operations (Clean Air) Regulation 2010

Further discussion of the above legislation is provided in the EMS, as well as the MOD2 and the EIS.

#### 2.2 Guidelines

The following guidelines have been consulted during development of this AQMP:

- National Environment Protection Council (NEPC) National Environment Protection Measure (NEPM) for Ambient Air Quality
- AS 3580.1.1:2007 Methods for sampling and analysis of ambient air: Part 1.1: Guide to siting air monitoring equipment
- Approved Methods for the Modelling and Assessment of Air Pollutants in NSW (Department of Environment and Conservation NSW (DEC), 2005)
- Approved Methods for the Sampling and Analysis of Air Pollutants in New South Wales (DEC, 2007)

#### 2.3 Conditions of approval

The consent conditions relevant to this AQMP are listed in Table 2-1. A cross reference is also included to indicate where the condition is addressed in this AQMP or other environmental management documents.

#### Table 2-1 Consent conditions relevant to the AQMP

Condition No.	Requirement			Reference
Schedule 3, Condition 6	The Proponent must ensure that particulate matter emissions generated by the project do not cause exceedances of the criteria in Table 3 at any residence on privately-owned land.			Section 4
	Pollutant	Averaging Period	Criterion	
	Particulate matter < 10 µm (PM10)	Annual	<sup>a,c</sup> 30 µg/m <sup>3</sup>	
		24 hour	<sup>ь</sup> 50 μg/m³	
	Total suspended particulates (TSP)	Annual	<sup>a,c</sup> 90 μg/m <sup>3</sup>	
	Deposited dust	Annual	<sup>b</sup> 2 g/m <sup>2</sup> /month	
			<sup>a</sup> 4 g/m <sup>2</sup> /month	
	Notes:			
	<ul><li>(a) Total impact (i.e. incremental increase in concentrations due to the development plus background concentrations due to all other sources).</li><li>(b) Incremental impact (i.e. incremental increase in</li></ul>			
	(b) Incremental imp concentrations due	act (i.e. incremental to the project on its c	increase in own).	

Condition No.	Requirement	Reference
	<ul> <li>(c) Excludes extraordinary events such as bushfires, prescribed burning, dust storms, fire incidents or any other activity agreed by the Secretary.</li> <li>(d) Deposited dust is to be assessed as insoluble solids as defined by Standards Australia, AS/NZS 3580.10.1:2003: Methods for Sampling and Analysis of Ambient Air - Determination of Particulate Matter - Deposited Matter - Gravimetric Method.</li> </ul>	
Schedule 3, Condition 7	The Proponent must: (a) implement best management practice to minimise the dust emissions of the project, including routinely watering haul roads being used by heavy vehicles and equipment;	Section 5
	(b) regularly assess meteorological and air quality monitoring data to guide the day-to-day planning of operations and implementation of air quality mitigation measures to ensure compliance with the relevant conditions of this approval;	Section 5
	<ul> <li>(c) minimise the air quality impacts of the project during adverse meteorological conditions and extraordinary events (see Note c to Table 3 above);</li> </ul>	Section 5
	(d) monitor and report on compliance with the relevant air quality conditions in this approval; and	Section 6
	(e) minimise surface disturbance of the site, other than as permitted under this approval, to the satisfaction of the Secretary.	Section 5
Schedule 3, Condition 7A	Within three months of the approval of Modification 2, the Proponent must prepare an Air Quality Management Plan for the project to the satisfaction of the Secretary. This plan must:	
	person/s whose appointment has been endorsed by the Secretary;	
	(b) be prepared in consultation with the EPA;	Appendix A
	<ul> <li>(c) describe the measures to be implemented to ensure:</li> <li>(i) compliance with the air quality criteria and operating conditions in this approval;</li> <li>(ii) best practice management is being employed; and</li> </ul>	Section 5
	(iii) air quality impacts of the project are minimised during adverse meteorological conditions and extraordinary events;	
	(d) describe the air quality management system; and	Section 5
	<ul> <li>(e) include an air quality monitoring program that:</li> <li>(i) is capable of evaluating the performance of the project against the air quality criteria;</li> <li>(ii) adapted by approximate the air quality monogram ant</li> </ul>	Section 6
	system; and	
	exceedance, incident or noncompliance and for notifying the Department and relevant stakeholders of these events.	
	The Project must implement the Air Quality Management Plan as approved by the Secretary.	

Condition No.	Requirement	Reference
Schedule 5, Condition 1A	The Proponent must ensure that the management plans required under this approval are prepared in accordance with any relevant guidelines, and include:	
	(a) a summary relevant background or baseline data;	Section 3
	<ul> <li>(b) a description of:</li> <li>the relevant statutory requirements (including any relevant approval, licence or lease conditions);</li> <li>any relevant limits or performance measures/criteria; and</li> <li>the specific performance indicators that are proposed to be used to judge the performance of, or guide the implementation of, the project or any management measures;</li> </ul>	Section 1.2 and Section 2
	(c) a description of the measures that to be implemented to comply with the relevant statutory requirements, limits, or performance measures/criteria;	Section 5
	<ul> <li>(d) a program to monitor and report on the:</li> <li>impacts and environmental performance of the project; and</li> <li>effectiveness of any management measures (see (c) above);</li> </ul>	Section 6
	(e) a contingency plan to manage any unpredicted impacts and their consequences and to ensure that ongoing impacts reduce to levels below relevant impact assessment criteria as quickly as possible;	Section 6.3
	(f) a program to investigate and implement ways to improve the environmental performance of the project over time;	Section 7
	<ul> <li>(g) a protocol for managing and reporting any:</li> <li>incidents;</li> <li>complaints;</li> <li>non-compliances with statutory requirements; and</li> <li>exceedances of the impact assessment criteria and/or performance criteria; and</li> </ul>	Refer to the EMS
	(h) a protocol for periodic review of the plan.	Section 7

#### 2.4 Environment protection licence

The EPL conditions, relevant to this AQMP, are listed in Table 2-2. A cross reference is also included to indicate where the condition is addressed in this NMP or other environmental management documents.

#### Table 2-2 EPL conditions relevant to the NMP

Condition No.	Requirement	Reference
O3.1	The premises must be maintained in a condition which minimises or prevents the emission of dust from the premises.	Section 5

### **Existing environment and impacts**

#### 3.1 Existing environment

The nearest identified sensitive receivers located in the vicinity of the site are detailed in Table 3-1 and shown in Figure 3-1 below. Distances are stated from the receiver to the nearest point at the site boundary.

Receiver	Receiver type	Address	Distance from site activity (metres)	Direction from site
R1	Residential	265 Warwick Park Road	1030	South
R2	Residential	265 Warwick Park Road	1060	South
R3	Residential	200 Warwick Park Road	1070	Southwest
R4	Residential	200 Warwick Park Road	1060	Southwest
R5	Residential	175 Warwick Park Road	960	Southwest
R6	Residential	157 Warwick Park Road	970	Southwest
R7	Residential	129 Warwick Park Road	1090	West

#### Table 3-1 Identified air quality sensitive receivers

Holcim conducts monthly dust deposition monitoring at four locations surrounding the site, as shown on Figure 3-1, as required by Project Approval 06/-0030. A summary of the results provided by Holcim for 2016 is provided in Table 3-2. The annual average dust deposition for all four sites is well below the criteria of 4g/m<sup>2</sup>/year. The highest annual average dust levels are at site DDG2, and represent only 30 per cent of the allowable dust levels, however these results are skewed from one month with elevated levels. The results show that dust impacts from the site are minimal and the site is readily complying with the criteria.

Month / Site	DDG 1	DDG 2	DDG 3	DDG 4
January	0.3	0.4	0.5	0.6
February	0.4	0.6	0.5	0.5
March	0.2	4.7	0.3	0.5
April	0.2	1.6	0.2	0.8
May	0.3	1.2	0.3	1.6
June	0.3	1.1	1.6	0.5
July	0.1	0.5	0.4	0.4
August	0.6	0.5	0.3	0.4
September	0.8	0.5	0.4	0.3
October	0.8	0.5	0.4	0.3
November	0.4	1.9	0.3	0.4
December	0.5	1.7	0.6	0.5
Annual Average	0.41	1.23	0.48	0.57

#### Table 3-2 Dust deposition sampling results for 2016



N:AU(Sydney)Projects/2218823/GIS/Maps\Deliverables/22\_18823\_Z003\_NoiseFigures.mxd @ 2019. Whilst every care has been taken to prepare this map, GHD (and Sixmaps, NSW Land and Property Information) make no representations or warranties about its accuracy, reliability, completeness or suitability for any particular purpose and cannot accept liability and responsibility of any kind (whether in contract, tort or otherwise) for any expenses, losses, damages and/or costs (including indirect or consequential damage) which are or may be incurred by any party as a result of the map being inaccurate, incomplete or unsuitable in any way and for any reason.

Data source: Aerial Imagery: Sixmaps (2017 - NSW LPI), General Topo: NSW DTDB 2012. Created by:apmiller

The transport and dispersion of the air emissions from the quarry is influenced by prevailing meteorology including vertical temperature profiles that will alter both diurnally and with wind direction.

Meteorological data from the Bureau of Meteorology's Coolangatta Automatic Weather Station (AWS) indicates that the wind speeds, which are of particular importance when determining the potential for dust impacts, are typically greater in spring and summer. As shown in Figure 3-2, the five-year wind rose shows that calm, light and gentle winds occur for nearly 70 per cent of the time, with roughly 30 per cent of wind above 19.8 kilometres per hour. This is a level that could cause nuisance dust. Most high winds occur from the north, meaning that dust impacts would be more likely to occur opposite to this direction, southwards.



#### Figure 3-2 Wind rose for Coolangatta

#### **3.2 Predicted impacts**

A summary of the predicted impacts from the quarry based on truck volumes and access road conditions is presented below in Table 3-3.

### Table 3-3Predicted maximum 24 hour PM10 concentration from trucks<br/>µg/m³

Sensitive receptor	Existing scenario one (four trucks)	Proposed scenario two (twelve trucks) gravel road	Proposed scenario three (twelve trucks) sandy road
R1	0.6	1.8	0
R2	0.7	1.5	0
R3	0.7	1.7	0
R4	0.9	1.8	0
R5	0.9	2.0	0
R6	0.8	1.6	0
R7	0.2	0.7	0

The results in Table 3-3 show that dust impacts from the additional trucks do not have any significant impact at nearby sensitive receptors. The maximum predicted increment of 2  $\mu$ g/m<sup>3</sup> is well below the criteria of 50  $\mu$ g/m<sup>3</sup>. This was predicted from a worst case gravel road, which does not currently or would likely exist and has been presented as a comparison only.

Due to the quarry extracting sand which has a low dust potential, cumulative impacts are not expected, which is demonstrated by the low levels of dust in the monthly sampling at all sites.

Dust monitoring data provided in Section 3.1, shows that the annual and monthly deposition results to be consistently below the criteria and importantly no dust complaints have been made at the site.

Results show that worst case dust impacts from increasing the number of trucks to twelve each hour are minimal and adverse dust impacts are not expected.

### 4. Operating criteria

#### 4.1 Air quality criteria

Development Consent No. 06\_0030 air quality criteria are provided in Table 4-1.

#### Table 4-1 Air quality criteria

Pollutant	Averaging Period	Criterion
Particulate matter < 10 µm (PM10)	Annual	<sup>a,c</sup> 30 µg/m³
	24 hour	<sup>b</sup> 50 µg/m³
Total suspended particulates (TSP)	Annual	<sup>a,c</sup> 90 µg/m³
d Deposited dust	Annual	<sup>b</sup> 2 g/m <sup>2</sup> /month
		<sup>a</sup> 4 g/m <sup>2</sup> /month

Notes:

a Total impact (i.e. incremental increase in concentrations due to the development plus background concentrations due to all other sources).

b Incremental impact (i.e. incremental increase in concentrations due to the project on its own).

c Excludes extraordinary events such as bushfires, prescribed burning, dust storms, fire incidents or any other activity agreed by the Secretary.

d Deposited dust is to be assessed as insoluble solids as defined by Standards Australia, AS/NZS 3580.10.1:2003: Methods for Sampling and Analysis of Ambient Air - Determination of Particulate Matter - Deposited Matter - Gravimetric Method.

### 5. Environmental control measures

Environmental requirements and control measures are identified in the CoA, EIS and MOD2. Specific measures and requirements to address air quality impacts are outlined in Table 5-1.

Ref.	Environmental Management Measure	Timing	Responsibility
AQ01	The air quality criteria specified in the consent conditions and outlined in Table 4-1 are not to be exceeded.	At all times during operation	Quarry Manager
AQ02	Monitor meteorological conditions daily and if adverse conditions or extraordinary events (i.e. bushfires, prescribed burning, dust storms, sea fog, fire incidents) exist, quarry operations will reduce and potentially cease accordingly.	Daily during operation	Quarry Manager
AQ03	Monitoring and reporting on compliance with the relevant air quality conditions is required, refer to Section 6.	As per triggers in Table 6-1	Quarry Manager
AQ04	The full length of internal haulage roadways will be sealed and maintained.	At all times	Quarry Manager
AQ05	A vegetation barrier for dust control along the southern boundary adjoining Warwick Park Road has been established and is maintained in accordance with Section 4.4, Appendix A of the Landscape Management Plan).	At all times	Quarry Manager
AQ06	The area of surface disturbance is to be minimised and progressive rehabilitation must be undertaken at the site, as per Section 4.0, Appendix A of the Landscape Management Plan.	At all times during operation	Quarry Manager
AQ07	Topsoil stripping will be undertaken in sub- stages of 1 hectare or less and not on days with excessive winds.	At all times during operation	Quarry Manager
AQ08	Water sprays are required on screening plant, when dust is visible.	During operation when dust is visible	Quarry Manager
AQ09	A rumble grid is installed at the weighbridge to minimise mud tracking onto the road and material collected by the rumble grid removed monthly.	Monthly	Quarry Manager
AQ10	When mud tracking on to Pottsville Road is observed, a street sweeper will be engaged to clean the road.	When observed	Quarry Manager
AQ11	Haul road routes will be watered as required, particularly during peak periods of high frequency vehicle movements and extended dry spells.	When dust is observed	Quarry Manager
AQ12	Loaded trucks leaving the site will be covered to minimise the transport of dust off site.	At all times during operation	Quarry Manager

#### Table 5-1 Environmental controls and mitigation measures

Ref.	Environmental Management Measure	Timing	Responsibility
AQ13	Stockpiled topsoil (and stockpiles to remain undisturbed for 3 months or longer) will be seeded to stabilise.	As soon as possible following establishment of stockpiles	Quarry Manager
AQ14	Dust monitoring, as per Section 6.2), will be undertaken.	Monthly – refer to Section 6.2	Quarry Manager
AQ15	<ul> <li>Mitigation measures to reduce greenhouse gas emissions are:</li> <li>Opportunities for the use of biodiesel will be investigated.</li> <li>Efficient plant and vehicles will be used where reasonable and feasible to do so.</li> <li>Turn off engines when not in use.</li> <li>All machinery and vehicles will be maintained in good working order and made to comply with relevant exhaust standards.</li> </ul>	At all times during operation	Quarry Manager

### 6. Monitoring and reporting

#### 6.1 Environmental inspections

Routine weekly inspections by the Quarry Manager (or delegate) will occur throughout the operational lifetime of the quarry to identify any ad-hoc air quality issues such as dust emissions, using the *Environmental Inspection Checklist* in the Environmental Monitoring and Management Plan.

#### 6.2 Dust monitoring

Deposited dust is an indicator of the effectiveness of site dust management practices and the potential for off-site dust nuisance. Deposited dust has been monitored at four locations, surrounding the quarry (as shown on Figure 3-1), in the past. Monitoring is conducted with dust deposition gauges that are located both upwind and downwind of the activity area to reflect the impact of the quarry operations during the most predominant wind directions.

Based on the separation distance between the nearest sensitive receptors and the site operations, it is considered that monitoring of deposited dust will provide the best indicator for site impacts on local amenity. Adverse health impacts due to fine particulate matter from an operation of this size usually have an impact zone measured in tens of metres rather than hundreds of metres.

Airborne particulate monitoring of PM<sub>10</sub> and TSP is only required to be undertaken if annual production rates increase to 200,000 tonnes or above, or in the event of a valid complaint relating to Dunloe Sand Quarry operations.

Monitoring results will be reviewed regularly, as discussed in Section 6.3 and 6.4 below. After two years of dust deposition monitoring or three months of PM<sub>10</sub> and TSP monitoring, the results will be reviewed by a suitably qualified and experienced air quality consultant. The requirement for further dust monitoring will be determined based on this assessment. If monitoring results clearly indicate that no dust impacts have been recorded, dust monitoring will cease. Dust monitoring will then only occur following:

- A justifiable dust complaint in relation to Dunloe Sand Quarry operations, or
- A change in operating conditions that are likely to increase dust emissions from the site.

#### 6.3 Contingency plan

If the above monitoring detects an impact or there is a justified community, dust related, complaint in relation to Dunloe Sand Quarry operations, a contingency plan or trigger and response plan is to be implemented, as described in Table 6-1.

Trigger	Response
Excessively dusty conditions or dust	<ul> <li>Quarry Manager to stop work and implement additional controls e.g. watering, cover stockpiles</li> </ul>
blown offsite	<ul> <li>Revegetate any unused, unvegetated areas within the Dunloe Sand Quarry operational area.</li> </ul>
Non-compliance with dust criteria	<ul> <li>Identifying the dust source that has caused the exceedance.</li> <li>Reassess the mitigation measures employed at the site to reduce the impact of the dust source.</li> </ul>
	<ul> <li>Following the adoption of additional dust mitigation, conduct further dust monitoring to ensure the success of the mitigation measure</li> </ul>

#### Table 6-1 Contingency plan

Trigger	Response
Justified dust related complaint	<ul> <li>Consult complainant to determine time and source of dust.</li> <li>Offer dust monitoring.</li> <li>If monitoring indicates the complaint is justified and relates to Dunloe Sand Quarry operations, reassess the mitigation measures employed at the site to reduce the impact of the dust source.</li> <li>Following the adoption of dust mitigation, conduct further dust monitoring to ensure the success of the mitigation measure.</li> <li>Provide details of the response and dust monitoring to the complainant.</li> </ul>

#### 6.4 Reporting

The general reporting requirements are described in Section 8.4 of the EMS. In relation to the dust monitoring, the routine dust monitoring will be recorded on the *Environmental Inspection Checklist* in the Environmental Monitoring and Management Plan.

A report will be prepared by the Quarry Manager following the 12 months of compliance monitoring. This is to include, as a minimum:

- The date(s) of the monitoring
- The time(s) of the monitoring
- The location of the monitoring
- The activities occurring during the monitoring
- A comparison of the results with the adopted dust criteria

If an exceedance of the criteria is recorded, the affected landowners and Department of Planning, Industry and Environment (DPIE) will be notified in writing and provided with quarterly monitoring results until the results show that the project is complying with the relevant criteria.

A summary of air quality results will be presented in the Annual Report (refer to Section 8.4.1 of the EMS). All records will be:

- Maintained in a legible form
- Kept for at least 4 years
- Produced to any authorised officer of the EPA and/or DPIE upon request

### 7. Review and improvement

Continuous improvement of this AQMP will be achieved in accordance with Section 9 of the EMS, through the ongoing evaluation of environmental management performance against environmental policies, objectives and targets.

The continuous improvement process is designed to:

- Identify areas of opportunity for improvement of environmental management and performance.
- Determine the cause or causes of non-conformances and deficiencies.
- Develop and implement a plan of corrective and preventative action to address any nonconformances and deficiencies.
- Verify the effectiveness of the corrective and preventative actions.
- Document any changes in procedures resulting from process improvement; and make comparisons with objectives and targets.

Appendices

**Appendix A** – Agency consultation

#### **Ben Luffman**

From: Sent: To: Subject:	Geff Cramb <geff.cramb@epa.nsw.gov.au> Wednesday, 21 August 2019 11:20 AM Ben Luffman RE: Dunloe Quarry Management Plan consultation</geff.cramb@epa.nsw.gov.au>
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
<b>RepoType:</b>	Job

Dear Ben

The EPA do not intend to review and provide comment upon the management plans. The EPA are content with the scope. However, it is understood that EPA will undertake compliance reviews against the requirements of the Environment Protection Licence issued and the implementation of the management plan at our discretion.

Regards Geff

#### **Geff Cramb**

Operations Officer – Environment Management Unit North Coast, NSW Environment Protection Authority

+61 2 6640 2510

geff.cramb@epa.nsw.gov.au www.epa.nsw.gov.au ♥@EPA NSW Report pollution and environmental incidents 131 555 (NSW only) or +61 2 9995 5555



I work flexibly. I'm sending this message now because it's a good time for me, but I don't expect that you will read, respond to or action it outside of your own regular hours.

From: Ben Luffman <<u>Ben.Luffman@ghd.com</u>>
Sent: Thursday, 8 August 2019 12:51 PM
To: Peter Lynch <<u>Peter.Lynch@epa.nsw.gov.au</u>>
Cc: Janelle Bancroft <<u>Janelle.Bancroft@epa.nsw.gov.au</u>>; Victoria Musgrove
<<u>victoria.musgrove@lafargeholcim.com</u>>
Subject: Dunloe Quarry Management Plan consultation

#### Hi Peter,

Not sure if you are the correct person to contact but we have updated the management plans for Dunloe Quarry following the recent approval of MOD2. The conditions of the Project Approval – SSD 06\_0030 require a number of the plans to be prepared in consultation with the EPA. We have therefore attached the relevant plans for review.

The updates have mainly been a reformatting to remove duplication and inclusion of additional information to address the new requirements of the conditions.

We would appreciate your comments by 23 August 2019.

Please contact me if you have any questions.

#### Regards

#### Ben Luffman | A GHD Associate

B.App.Sc. (Hons) | Grad.Dip. Urban and Regional Planning | Environmental Auditor **Technical Director - Environment** 

#### GHD

#### Proudly employee owned

T: +61 2 6650 5613 | M: +61 415 271 319 | E: <u>ben.luffman@ghd.com</u> 230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.qhd.com</u>



WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION

GHD acknowledges the Traditional Owners of Country throughout Australia. We pay respect to their continuing culture and Elders past, present and emerging. <u>Click here</u> to learn about our Reconciliation Action Plan.

Please consider the environment before printing this email 1 ream of paper = 6% of a tree / 5.4kg CO2 in the atmosphere | 3 sheets of A4 paper = 1 litre of water

CONFIDENTIALITY NOTICE: This email, including any attachments, is confidential and may be privileged. If you are not the intended recipient please notify the sender immediately, and please delete it; you should not copy it or use it for any purpose or disclose its contents to any other person. GHD and its affiliates reserve the right to monitor and modify all email communications through their networks.

\_\_\_\_\_

This email is intended for the addressee(s) named and may contain confidential and/or privileged information.

If you are not the intended recipient, please notify the sender and then delete it immediately.

Any views expressed in this email are those of the individual sender except where the sender expressly and with authority states them to be the views of the Environment Protection Authority.

PLEASE CONSIDER THE ENVIRONMENT BEFORE PRINTING THIS EMAIL

This e-mail has been scanned for viruses

GHD

230 Harbour Drive Coffs Harbour NSW 2450 T: 61 2 6650 5600 F: 61 2 6650 5601 E: cfsmail@ghd.com

#### © GHD 2020

This document is and shall remain the property of GHD. The document may only be used for the purpose for which it was commissioned and in accordance with the Terms of Engagement for the commission. Unauthorised use of this document in any form whatsoever is prohibited.

#### 22-SO-976847825-

5/https://projects.ghd.com/oc/Newcastle3/holcimdunloesandquar/Delivery/Documents/2220056\_RP T\_Dunloe Air Management Plan.docx

Revision	Author	Reviewer		Approved for Issue		
		Name	Signature	Name	Signature	Date
0	B Luffman	S Lawer	S Lawer	S Lawer	S Lawer	23/09/2019
1	B Luffman	S Lawer	S Lawer	S Lawer	S Lawer	21/11/2019
2	B Luffman	S Lawer	Ja-	S Lawer	fan	04/02/2020

#### **Document Status**
# www.ghd.com



**Appendix H** – Traffic management plan



## Holcim (Australia) Pty Ltd

Dunloe Sand Quarry Traffic Management Plan

May 2019

This report: has been prepared by GHD for Holcim (Australia) Pty Ltd and may only be used and relied on by Holcim (Australia) Pty Ltd for the purpose agreed between GHD and the Holcim (Australia) Pty Ltd as set out in section 1.1 of this report.

GHD otherwise disclaims responsibility to any person other than Holcim (Australia) Pty Ltd arising in connection with this report. GHD also excludes implied warranties and conditions, to the extent legally permissible.

The services undertaken by GHD in connection with preparing this report were limited to those specifically detailed in the report and are subject to the scope limitations set out in the report.

The opinions, conclusions and any recommendations in this report are based on conditions encountered and information reviewed at the date of preparation of the report. GHD has no responsibility or obligation to update this report to account for events or changes occurring subsequent to the date that the report was prepared.

The opinions, conclusions and any recommendations in this report are based on assumptions made by GHD described in this report. GHD disclaims liability arising from any of the assumptions being incorrect.

GHD has prepared this report on the basis of information provided by Holcim (Australia) Pty Ltd and others who provided information to GHD (including Government authorities)], which GHD has not independently verified or checked beyond the agreed scope of work. GHD does not accept liability in connection with such unverified information, including errors and omissions in the report which were caused by errors or omissions in that information.

## **Table of contents**

1. Introduction		duction	1
	1.1	Objectives	1
	1.2	Targets	1
	1.3	Consultation	1
2.	Envi	onmental requirements	3
	2.1	Regulatory requirements	3
	2.2	Conditions of approval	3
3.	Exist	ing environment	6
	3.1	Existing environment	6
4.	Risk	assessment	11
5.	Environmental control measures		12
6. Compliance management		pliance management	13
	6.1	Environmental inspections and monitoring	13
	6.2	Contingency plan	13
	6.3	Reporting	13
7.	Revi	ew and improvement	14

## **Table index**

Table 1-1	Consultation summary1
Table 2-1	Consent conditions relevant to the TMP3
Table 5-1	Environmental controls and mitigation measures12
Table 6-1	Contingency plan

## **Figure index**

Figure 3-1	Pottsville Road in proximity to the site access intersection	6
Figure 3-2	Site location and layout	7
Figure 3-3	Pottsville Road/Cudgera Creek Road intersection	8
Figure 3-4	Dunloe Sands Quarry access road	9
Figure 3-5	Daily traffic volumes on Pottsville Road	9
Figure 3-6	Hourly traffic profile for Pottsville Road (Friday 10 March 2017)	10

## **Appendices**

Appendix A – Consultation

Appendix B – Driver's Code of Conduct

Appendix C – Pre-start check

## 1. Introduction

This Traffic Management Plan (TMP) forms part of the Environmental Management Strategy (EMS) for Dunloe Sand Quarry. This TMP has been prepared to meet the requirements of the Minister's Conditions of Approval (CoA) outlined in Development Consent No. 06\_0030, the mitigation measures outlined in MOD2 (GHD 2017), the Environmental Impact Statement (EIS) (Planit 2007), the Environment Protection Licence 13077 (EPL) and relevant legislation.

## 1.1 **Objectives**

The key objective of the TMP is to ensure appropriate controls and procedures are implemented in order to minimise the impacts to the environment and local community from traffic associated with the operation of Dunloe Sand Quarry.

## **1.2 Targets**

The following targets have been established for the management of traffic impacts during the operational lifetime of Dunloe Sand Quarry:

- Ensure full compliance with the relevant legislative requirements and CoA.
- No justified complaints related to site traffic.
- No road damage from quarry vehicle movements beyond normal wear and tear.

## **1.3** Consultation

Extensive consultation was undertaken with the local community during preparation of the EIS. Concerns identified by relevant stakeholders were addressed in the EIS via mitigation measures and incorporated into this TMP.

As per CoA 35B, Schedule 3, the Roads and Maritime Service (RMS) and Tweed Shire Council (TSC) were consulted during the preparation of the TMP. Evidence of the consultation is provided in Appendix A. A summary of the RMS and TSC responses are provided in Table 1-1, along with how their comments were addressed.

Agency	Comment	Response
RMS	RMS has been using the Operational TMP for Blakebrook Quarry as an example for other quarries. The only thing missing are the Traffic Control Plans (TCPs) that would be used for specific incidents/contingencies.	The Operational TMP for Blakebrook was reviewed and discussed with RMS. In response a Risk Assessment chapter and additional controls in the Drivers Code of Conduct were included. To adopt the layout of the Blakebrook example would not be consistent with the other subplans of the EMS. In regards to the TCP, it was considered more appropriate for these to be prepared by the contractor undertaking the work, once the scope of works are known.

#### **Table 1-1 Consultation summary**

Agency	Comment	Response
TSC	The Traffic Management Plan appears to address the condition. Please note that the speed limit on Pottsville Road is 80 km/h and not 100 km/h. This needs to be changed in your plan.	Reference to 100 km/h on Pottsville Road was revised to 80 km/h.

## 2. Environmental requirements

### 2.1 Regulatory requirements

Regulatory requirements relevant to traffic management includes:

- Roads Act 1993
- Australian Standard 1742.3 2009 Traffic control for works on roads
- NSW Road Rules

Further discussion of the above legislation is covered in Section 2 of the EMS and in the EIS.

### 2.2 Conditions of approval

The conditions from Development Consent No. 06\_0030 relevant to this TMP are listed in Table 2-1. A cross reference is also included to indicate where the condition is addressed in this TMP or other environmental management documents.

#### Table 2-1 Consent conditions relevant to the TMP

Condition No.	Requirement	Reference
Schedule 3, Condition 33	<ul> <li>Prior to commencement of operations the Proponent must:</li> <li>(a) design and construct the haul road and its intersection with Pottsville-Mooball Road; and</li> <li>(b) install advanced truck turning warning signage along Pottsville-Mooball Road, to the satisfaction of Council.</li> </ul>	Section 5
Schedule 3, Condition 34	The Proponent must ensure that all loaded vehicles entering or leaving the site have their loads covered.	Section 5 and Appendix B
Schedule 3, Condition 35	The Proponent must ensure all loaded vehicles leaving the site are cleaned of materials that may fall on the road before they are allowed to leave the site.	Section 5 and Appendix B
Schedule 3, Condition 35A	Within 12 months of approval of Modification 2, unless otherwise agreed by the Secretary, the Proponent must implement and pay the full cost of implementing the recommended treatments listed in Table 3 of the <i>Independent Road Safety Audit</i> prepared by Bitzios Consulting and dated 16 October 2017, and outlined in the Proponent's <i>Response to Road Safety Audit</i> dated March 2018, to the satisfaction of Council, including any additional line marking and signage requested by Council. <i>Note: The Proponent must obtain all necessary approvals under section 138 of the Roads Act 1993 from Council before implementing the recommended treatments.</i>	Council has been consulted to confirm the works required. The consultation is provided in Appendix A and the agreed scope is included in Section 5.

Condition No.	Requirement	Reference
Schedule 3, Condition 35B	<ul> <li>The Proponent must prepare a Traffic Management Plan for the project to the satisfaction of the Secretary. This plan must: <ul> <li>(a) be prepared in consultation with RMS and Council;</li> <li>(b) be submitted to the Secretary for approval prior to the commencement of operations under Modification 2, unless otherwise agreed by the Secretary;</li> <li>(c) describe the processes in place for the management of trucks entering and exiting the site;</li> <li>(d) include a Drivers' Code of Conduct that details:</li> <li>safe and quiet driving practices that must be used by drivers travelling to and from the quarry;</li> <li>a map of the primary haulage routes highlighting critical locations for application of safe and quiet driving practices, including residential areas and school bus routes;</li> <li>an induction process for drivers and regular toolbox meetings;</li> <li>complaint resolution procedures;</li> <li>any community consultation measures in respect of peak haulage periods; and</li> <li>consideration of seasonal traffic and events;</li> <li>(e) describe the measures to be put in place to ensure compliance with the Drivers' Code of Conduct; and</li> <li>(f) propose measures to minimise the transmission of dust and tracking of material onto public roads from vehicles leaving the site.</li> </ul> </li> </ul>	This Plan and Appendix B. The induction and toolbox process and complaint resolution procedures are in the EMS. The community consultation measures during peak haulage periods is in Section 5
Schedule 3, Condition 36	The Proponent must provide sufficient parking on-site for all project-related traffic and visitors, in accordance with Council's parking codes and to the satisfaction of the Secretary. No on street parking must be undertaken.	Section 5
Schedule 5, Condition 1A	The Proponent must ensure that the management plans required under this approval are prepared in accordance with any relevant guidelines, and include:	This plan
	(a) a summary relevant background or baseline data;	Section 3
	<ul> <li>(b) a description of:</li> <li>the relevant statutory requirements (including any relevant approval, licence or lease conditions);</li> <li>any relevant limits or performance measures/criteria; and</li> <li>the specific performance indicators that are proposed to be used to judge the performance of, or guide the implementation of, the project or any management measures:</li> </ul>	Section 2.1 NA Section 1.2
	<ul> <li>(c) a description of the measures that to be implemented to comply with the relevant statutory requirements, limits, or performance measures/criteria;</li> </ul>	Section 5
	<ul> <li>(d) a program to monitor and report on the:</li> <li>impacts and environmental performance of the project; and</li> <li>effectiveness of any management measures (see (c) above);</li> </ul>	Section 6.1
	(e) a contingency plan to manage any unpredicted impacts and their consequences and to ensure that ongoing impacts reduce to levels below relevant impact assessment criteria as quickly as possible;	Section 6.2

Condition No.	Requirement	Reference
	(f) a program to investigate and implement ways to improve the environmental performance of the project over time	Section 7
	<ul> <li>(g) a protocol for managing and reporting any:</li> <li>incidents;</li> <li>complaints;</li> <li>non-compliances with statutory requirements; and</li> <li>exceedances of the impact assessment criteria and/or performance criteria; and</li> </ul>	Section 6
	(h) a protocol for periodic review of the plan.	Section 7

## 3. Existing environment

## 3.1 Existing environment

The site is located on an access road that intersects Pottsville Road at a priority controlled intersection, located approximately 2.3 kilometres to the south of Cudgera Creek Road (Figure 3-2).

### 3.1.1 Pottsville Road

Pottsville Road functions as a collector road that provides connectivity between Mooball (at Tweed Valley Way) and Pottsville. Pottsville Road (in addition to the sand quarry) typically provides access to low density rural dwellings, however, to the north of the site it also provides access to Pottsville golf course.

In the vicinity of the site, Pottsville Road has the following characteristics:

- A two way sealed undivided carriageway of approximately 7.5 metres
- Marked double barrier lines
- A sign posted speed limit of 80 kilometres per hour



#### Figure 3-1 Pottsville Road in proximity to the site access intersection

To the north of the site, Pottsville Road intersects Cudgera Creek Road at a priority (stop) controlled T-junction (Figure 3-3).

Outputs from Google Traffic indicate that during peak periods of road network operation, the intersection of Pottsville Road and Cudgera Creek Road operates efficiently with only minor delays.



© 2017. Whilst every care has been taken to prepare this map, GHD (and Sixmaps, LPI) make no representations or warranties about its accuracy, reliability, completeness or suitability for any particular purpose and cannot accept liability and responsibility of any kind (whether in contract, fort or otherwise) for any expenses, losses, damages and/or costs (including indirect or consequential damage) which are or may be incurred by any party as a result of the map being inaccurate, incomplete or unsuitable in any way and for any reason. Data source: Aerial Imagery: Sixmaps (2017 - NSW LPI), LPI DCDB: Cadastre, 2012; LPI DTDB: Topo base data, 2012. Created by:mking3



Figure 3-3 Pottsville Road/Cudgera Creek Road intersection

Source: Google Maps

#### 3.1.2 Cudgera Creek Road

To the east of the Pacific Motorway, Cudgera Creek Road operates as a collector road. However, to the west of the motorway, Cudgera Creek Road operates as a local road that provides access to rural properties.

Cudgera Creek Road intersects the Pacific Motorway at an interchange with grade separated on-ramps and off-ramps.

East of the Pacific Motorway, Cudgera Creek Road has the following characteristics:

- A two way sealed undivided carriageway of approximately 12 metres wide (including shoulders).
- Shoulders that are approximately two metres wide.
- Marked double barrier lines.
- Turning lanes (left and right) at Pottsville Road.
- A sign posted speed limit of 60 kilometres per hour.

#### 3.1.3 Dunloe Sands Quarry access road

Access to the site is provided via a two-way sealed road, with a single travel lane in each direction and a carriageway width of approximately seven metres.



#### Figure 3-4 Dunloe Sands Quarry access road

Heavy vehicles access/egress the site utilising the grade separated Pacific Motorway/Cudgera Creek Road Interchange, Cudgera Creek Road and Pottsville Road.

#### Existing traffic volumes

The identified traffic volumes (per day and direction) on Pottsville Road south of Cudgera Creek Road are displayed in Figure 3-5.



#### Figure 3-5 Daily traffic volumes on Pottsville Road

The data in Figure 3-5 indicates that daily (two-way) traffic volumes on Pottsville Road are currently in the order of 1,400 – 1,800 vehicles, with peak demand recorded on Friday 10 March 2017.

The hourly (two-way) traffic volumes recorded on Pottsville Road on Friday 10 March 2017 are displayed in Figure 3-6.



Figure 3-6 Hourly traffic profile for Pottsville Road (Friday 10 March 2017)

The data in Figure 3-6 indicates that peak hour (two-way) traffic activity on Pottsville Road was:

AM Peak (9:00 am - 10:00 am) -149 vehicles per hour

PM Peak (4:00 pm - 5:00 pm) - 159 vehicles per hour

The survey data indicates that approximately seven per cent of vehicles on Pottsville Road consist of heavy vehicles.

## 4. Risk assessment

During the development application process, a Road Safety Audit (Bitzios 2017) was completed which identified a few road safety issues. GHD provided a response with options on how to address the significant safety issues highlighted in the Road Safety Audit (Bitzios 2017):

- Item 1.4 and Item 1.5 Safe Intersection Sight Distance and Stopping Sight Distance.
- Item 1.16 to 1.19 and Item 1.22 Unprotected hazards in clearzone (embankment, trees) on inside of curve (eastern and western side of Pottsville Road).

Council reviewed the options presented and provided advice in a letter dated 30 April 2018. Council's advice was clarified in an email dated 19 February 2019 (see Appendix A).

Based on the advice from Council, the mitigation measures required to address the safety issues identified, include:

- Install traffic signs referred to in Councils email dated 19 February 2019.
- Line marking and installation of road edge guide posts.

The above mitigation measures have been included in Section 4.

5. Environmental control measures

Environmental requirements and control measures are identified in the CoA of Development Consent No. 06\_0030 and the EIS. Specific measures and requirements to address traffic impacts are outlined in Table 5-1.

#### Ref. **Environmental Management Measures** Timing Responsibility T01 Implement a site Induction Procedure (Refer to Operation Quarry Manager EMS). T02 All drivers will be required to agree to and sign Operation Quarry Manager the Drivers Code of Conduct (Appendix B) for and Drivers the transport of materials on public roads. T04 Haulage of guarry materials from the site will be Quarry Manager Operation limited to the approved hours of operation and and Drivers EPL. T05 Design and construct the haul road and its Pre operation Quarry Manager intersection with Pottsville-Mooball Road. T06 Install advanced truck turning warning signage Pre operation Quarry Manager along Pottsville-Mooball Road, to the satisfaction of Council. T07 All loaded vehicles entering or leaving the site Operation Quarry Manager will have their loads covered. and Drivers T08 All loaded vehicles leaving the site will be **Quarry Manager** Operation cleaned of materials that may fall on the road and Drivers before they are allowed to leave the site. Within 12 months of approval of Modification 2, T09 Operation **Quarry Manager** Holcim will: Install traffic signs and line marking, including road edge guide posts to the satisfaction of Council. T10 Install a rumble grid at the weighbridge. Operation **Quarry Manager** T11 When mud tracking on to Pottsville Road is Operation Quarry Manager observed a street sweeper will be engaged to clean the road. Plant and equipment are to be operated and T12 Operation Quarry Manager maintained in a proper and efficient manner. and Drivers Provide sufficient parking on-site for all project-T13 Operation Quarry Manager related traffic and visitors, in accordance with Council's parking codes and to the satisfaction of the Secretary. No on street parking must be undertaken. T14 Truck speeds on internal roads is to be limited Operation Quarry Manager to a maximum of 40km/h. and Drivers T15 Accurate records of truck numbers and the Operation Quarry Manager amount of material transported will be maintained. T16 At each Community Consultative Committee Operation Quarry Manager meeting, Holcim will report on the number of peak haulage periods (i.e., greater than 20 trucks per hour) since the last meeting and any forecast peak haulage periods in the future.

#### Table 5-1 Environmental controls and mitigation measures

GHD | Report for Holcim (Australia) Pty Ltd - Dunloe Sand Quarry, 2220056 | 12

## 6.1 Environmental inspections and monitoring

Routine weekly inspections by the Quarry Manager (or delegate) will occur throughout the operational lifetime of the quarry to identify any ad-hoc traffic issues such as speeding, tracking of mud and silt and uncovered loads.

Truck movements to and from the site (including time of arrival and dispatch) will be monitored and recorded using the Dunloe Daily Sales register.

Holcim trucks and machinery are to be checked daily before use, using the pre-start check in Appendix C.

## 6.2 Contingency plan

If the above monitoring detects an impact or there is a justified community or traffic related complaint, a contingency plan or trigger and response plan is to be implemented, as shown below.

#### Table 6-1 Contingency plan

Trigger	Response
Traffic incident on a public road	<ul> <li>Contact emergency services, if required.</li> <li>Investigate the cause of the incident.</li> <li>Reassess the mitigation measures employed at the site to avoid the issue reoccurring, including consultation with drivers.</li> <li>Consult RMS, Council and DPE, if required.</li> <li>Conduct toolbox talks/meetings with drivers on the outcomes of the investigation.</li> <li>Monitor the issue to ensure the mitigation measures are effective.</li> </ul>
Mud/debris deposited on Pottsville Road	<ul> <li>Section of road to be swept.</li> <li>Check rumble grid and maintain, if necessary.</li> <li>Toolbox talks/meetings with drivers on the expectations regarding tracking.</li> </ul>
Traffic related complaint	<ul> <li>Investigate the issue in accordance with the EMS.</li> <li>Reassess the mitigation measures employed at the site to avoid the issue reoccurring, including consultation with drivers.</li> <li>Consult RMS, Council and DPE, if required.</li> <li>Conduct toolbox talks/meetings with drivers on the outcomes of the investigation.</li> <li>Monitor the issue to ensure the mitigation measures are effective.</li> </ul>

### 6.3 Reporting

The general reporting requirements are described in the EMS. The routine traffic monitoring will be recorded on the *Environmental Inspection Checklist*, in the *Environmental Management and Monitoring Plan* and Dunloe Daily Sales register.

A summary of these results will be presented in the *Annual Report* (refer to the EMS). All records will be:

- Maintained in a legible form.
- Kept for at least 4 years.
- Produced to any authorised officer of the EPA and/or DPE upon request.

## 7. Review and improvement

Continuous improvement of this TMP will be achieved in accordance with the EMS, through the ongoing evaluation of environmental management performance against environmental policies, objectives and targets.

The continuous improvement process is designed to:

- Identify areas of opportunity for improvement of environmental management and performance.
- Determine the cause or causes of non-conformances and deficiencies.
- Develop and implement a plan of corrective and preventative action to address any nonconformances and deficiencies.
- Verify the effectiveness of the corrective and preventative actions.
- Document any changes in procedures resulting from process improvement; and make comparisons with objectives and targets.

## Appendices

## **Appendix A** – Consultation

## **Ben Luffman**

From:	SCIEFER Grea < Grea SCIEFER@rms nsw aoy au>
Sont:	Monday & April 2019 3:18 PM
To:	Ben Luffman
Subject:	TMP for Dunloe Sand Quarry
Attachments:	Operational Traffic Management Plan - Blakebrook Quarry.pdf
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
<b>RepoType:</b>	Job

Ben

Reference is made to discussions on 8 April 2019 concerning the required consultation for the required Traffic Management Plan (TMP) for Dunloe Sand Quarry. Attached is a copy of the Operational TMP for Blakebrook Quarry for your information. RMS has been using this as an example for other quarries. The only thing missing are the Traffic Control Plans (TCPs) that would be used for specific incidents/contingencies. Typical plans can be acquired from RTAs "Traffic Control at Worksites Manual" which is available online. They can be modified to suit the situation.

RMS would be happy to provide further comments on your TMP before submitting and appreciate a final copy for future reference as these plans evolve with experience..

Thanks Greg Sciffer Development Assessment Officer Northern



#### Before printing, please consider the environment

IMPORTANT NOTICE: This email and any attachment to it are intended only to be read or used by the named addressee. It is confidential and may contain legally privileged information. No confidentiality or privilege is waived or lost by any mistaken transmission to you. Roads and Maritime Services is not responsible for any unauthorised alterations to this email or attachment to it. Views expressed in this message are those of the individual sender, and are not necessarily the views of Roads and Maritime Services. If you receive this email in error, please immediately delete it from your system and notify the sender. You must not disclose, copy or use any part of this email if you are not the intended recipient.

This e-mail has been scanned for viruses

From: Ray Clark <<u>rayc@tweed.nsw.gov.au</u>> Sent: Tuesday, 19 February 2019 9:47 AM To: Bernie Samson Cc: Danny Conaghan Subject: RE: Holcim Dunloe Sands Quarry Access/ Pottsville Road - Response to Road Safety Audit Report

Hi Bernie Sorry for tardiness in reply. I have included comments against each item in red below. Regards

Ray Clark Traffic Engineer Roads and Stormwater



p (02) 6670 2578 m 0439 223 094

contact us | website | your say tweed | our values



Your actions matter: print less to save more

From: Bernie Samson [mailto:Bernie.Samson@ghd.com] Sent: Monday, 18 February 2019 12:15 PM To: Ray Clark Subject: Re: Holcim Dunloe Sands Quarry Access/ Pottsville Road - Response to Road Safety Audit Report

Hi Ray,

As per our previous discussion, you would provide a response on 15/02/2019. May I request some updates regarding this request (as per email below)?

Regards,

Bernie

From: Bernie Samson
Sent: Friday, 1 February 2019 5:48 PM
To: Ray Clark
Cc: Ashleigh Douglas; Sharon Denlay
Subject: RE: Holcim Dunloe Sands Quarry Access/ Pottsville Road - Response to Road Safety Audit Report

#### Hi Ray,

As per our previous discussion, below is the summary of GHD's understanding of the attached letter from Council and the resulting scope of works (for Pottsville Road improvement works) after the completion of Road Safety Audit (RSA), GHD's assessment and recommendations and Council's review of RSA and GHD's report (as per attached letter). GHD would like to request Tweed Shire Council's review of this understanding and the proposed scope of works.

1. The main reference for the preparation of scope of works is the following clause from the attached letter from Council:

The Proponent must implement all recommended road improvement works listed in Option 1 of Section 3.4 Clear Zone Assessment of the (Draft) Response to Road Safety Audit prepared by GHD, dated March 2018, in consultation with Council. Additional signage (as required by Council) shall also be installed. The works must be implemented prior to any increase in truck movements associated with Mod 2.

#### correct

2. GHD's understanding of this clause is that Option 1 of Section 2.4 (sight distance assessment) in GHD's report is not supported by Council

#### correct

3. GHD's understanding of this clause is that Option 2 of Section 2.4 (sight distance assessment) in GHD's report is not supported by Council and Council would instead recommend the installation of the following signs. The installation of these signs is to be included in the scope of works.

#### correct

S/B approach	-	N/B approach	
W2-4 (L)	W5-22	W2-9(R)	W5-22
	<b></b> 5	~	

4. Council to confirm Items 2 and 3. Note that the recommendations in Option 2 of Section 2.4 (sight distance assessment) in GHD's report also include the review of existing auxiliary lane (left turn lane) at the southbound direction. With the current understanding of GHD, the only works required for the existing auxiliary lane is to provide the appropriate AUL linemarkings and pavement arrows which will form part of the following Item no. 5.

- 5. GHD's understanding of this clause is that Option 1 of Section 3.4 (Clear Zone assessment) in GHD's report is supported by Council. The following recommendations of GHD in Option 1 of Section 3.4 are to be included in the scope of works.
  - Apply new and clear pavement markings (i.e., no overtaking barrier lines, edge lines, auxiliary lane continuity line)

already in place, needs to be confirmed on site, left turn arrows required

Consider audio-tactile edge lines

Consider against RMS guidelines (further guidance from our road safety officer Alana Brooks on 0266702586)

• Apply new road edge guide posts (REGP)

Review against RMS delineation guidelines Part 16

• • • Apply retroreflective raised pavement markers (RRPM)

Review against RMS delineation guidelines Part 16

 Install relevant signages such as W1-3 on approaches to bend and D4-6 (chevron alignment markers)



#### As per AS1742.2

- 6. The clear zone assessment in Section 3.4 of GHD's report did not consider the installation of safety barriers for the following reason:
  - The RSA identified hazards (i.e. trees, embankment) are mostly located on the inside of the road horizontal curve. Installation of safety barriers at this location would cause further stopping sight distance issues (i.e. safety barrier would obstruct SSD sight line). Refer Section 3.4 of GHD's report.

#### correct

7. GHD will include the pavement mill and resheet in the scope of works. New road surfacing will provide a complete visibility of the proposed linemarkings and RRPM.

Pottsville road has been recently reline marked. Milling and resheet is not required or supported.

8. In summary, GHD is proposing the following scope of works:

• Signs and linemarking design works, including the installation of REGP correct Limited new linemarking expected. A review should be carried out against RMS delineation guidelines and submitted before s138 application is made

 Pavement mill and resheet along Pottsville Road. Please note that this scope is a preparation work for linemarkings and RRPM installation only (i.e. pavement rehabilitation works not included).
 Not supported

Furthermore, GHD would like to request the following from Council in relation to the proposed Pottsville Road improvement works

1. lidar survey (if detailed survey is not available) sorry not available

2. a documented process of S138 application for the proposed road improvement works. Documents attached

Kind Regards, Bernie Samson CPEng, MIEAust, RPEQ, NER, BSCEng Technical Director - Transport

#### GHD

Proudly employee owned T: +61 7 3316 3253 | M: +61 451 059 406 | E: bernie.samson@ghd.com 145 Ann Street Brisbane QLD 4000 Australia | www.ghd.com



WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY& BUILDINGS | TRANSPORTATION

Please consider our environment before printing this email

## **Ben Luffman**

From:	Ray Clark <rayc@tweed.nsw.gov.au></rayc@tweed.nsw.gov.au>
Sent:	Thursday, 18 April 2019 3:17 PM
То:	Ben Luffman
Cc:	Colleen Forbes
Subject:	FW: Dunloe Sand Quarry Traffic Management Plan
Attachments:	2220056_RPT-B_Dunloe Traffic Management Plan.pdf
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
<b>RepoType:</b>	Job

#### Ben

The Traffic Management Plan appears to address the condition. Please note that the speed limit on Pottsville Road is 80km/h and not 100km/h. This needs to be changed in your plan. Regards

#### **Ray Clark**

Traffic Engineer Roads and Stormwater



contact us | website | your say tweed | our values



Your actions matter: print less to save more

From: Ben Luffman [mailto:Ben.Luffman@ghd.com]
Sent: Thursday, 4 April 2019 10:44 AM
To: development.northern@rms.nsw.gov.au; Ray Clark
Cc: Victoria Musgrove; luke.edminson@lafargeholcim.com
Subject: Dunloe Sand Quarry Traffic Management Plan

Hi,

Please find attached the Draft Dunloe Sand Quarry Traffic Management Plan. Condition 35B, Schedule 3 of Project Application 06\_0030 (as reproduced below), requires the Traffic Management Plan to be prepared in consultation with Council and RMS. To satisfy this requirement, could you please review the attached and provide any comments by 19 April 2019.

The Proponent must prepare a Traffic Management Plan for the project to the satisfaction of the Secretary. This plan must:

(a) be prepared in consultation with RMS and Council;
 (b) be submitted to the Secretary for approval prior to the commencement of operations under Modification 2, unless otherwise agreed by the Secretary;

(c) describe the processes in place for the management of trucks entering and exiting the site;

(d) include a Drivers' Code of Conduct that details:

- safe and quiet driving practices that must be used by drivers travelling to and from the quarry;
- a map of the primary haulage routes highlighting critical locations for application of safe and quiet driving practices, including residential areas and school bus routes;
- an induction process for drivers and regular toolbox meetings;
- complaint resolution procedures;
- any community consultation measures in respect of peak haulage periods; and
- consideration of seasonal traffic and events;

 (e) describe the measures to be put in place to ensure compliance with the Drivers' Code of Conduct; and
 (f) propose measures to minimise the transmission of dust and tracking of material onto public roads from vehicles leaving the site.

The Proponent must implement the plan as approved by the Secretary.

### Regards

Ben Luffman | A GHD Associate B App Sc (Hons) | Grad Dip Urban and Regional Planning Technical Director - Environment

### GHD

Proudly employee owned

T: +61 2 6650 5613 | M: +61 415 271 319 | E: <u>ben.luffman@ghd.com</u> 230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.ghd.com</u>



WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION

**Appendix B** – Driver's Code of Conduct

## **Driver's Code of Conduct**

## 1. Objective

The objectives of the Driver's Code of Conduct are:

• To assist in the management of traffic to ensure the safety and health of work personnel, contractors, operators of mobile plant, heavy vehicles, light vehicles and the public who will be impacted by traffic activities.

## 2. General

Drivers are to:

- Complete the site induction (including this Driver's Code of Conduct) the first time they enter site.
- Be fit, not suffering from fatigue, nor under the influence of drugs or alcohol and will cooperate in the undertaking of random drug and alcohol testing, if required.
- Monitor their own performance and take regular breaks and rest times as required by road safety legislation and requirements.
- Inspect their vehicle prior to start up and report any issues before proceeding with work.
- Ensure the heavy vehicle is appropriate for the load to be carried in it.
- Ensure that all safety equipment fitted to the vehicle works properly.
- Not operate the vehicle unless the UHF radio attached to the vehicle is working properly.
- Listen carefully and abide by instructions on the UHF radio, particularly advice from the Quarry Manager.
- Use good road manners and operate the heavy vehicle safely and responsibly while considering the safety of himself/herself and the general public, in particular school children on the designated route.
- Complete all work diaries and timesheets properly and accurately and supply all necessary and appropriate paper work to the operator of the quarry if required.
- Ensure they are wearing the appropriate personal and protective equipment, including steel capped safety boots, high visibility clothing/vests and hearing protection.

## Non-compliance with this code of conduct will result in a review by the Quarry Manager and may result in a refusal to load in future.

## 3. Operating hours

Heavy vehicles are not to arrive or leave the quarry outside of the proposed operating hours of:

- Weekdays 7.00 am 5.00 pm
- Saturdays -7.00 am 12.00 pm

No heavy vehicles are to enter or leave the quarry on a Sunday or a public holiday.

## 4. Public roads

- The designated haulage routes are shown on Figure 1.
- The current intersection arrangement does not support heavy vehicles accessing the site south of Pottsville Road and undertaking a right turn into the access road. All heavy vehicles are to access and egress the site to/from Cudgera Creek Road.
- Slow to 40km/h when bus lights flash. Drivers should look out for children who are crossing the road or waiting to cross. Drive to the conditions but do not exceed the posted speed limit of 80km/h.
- Understand and abide by all road rules, including speed limits, road signs, use of seatbelts, avoiding taking unnecessary risks, avoiding overhead obstructions, not driving in a convoy and always parking well off the road.

## 5. Seasonal traffic

During seasonal traffic (e.g. cane harvest) or events, drivers must:

- Proactively communicate, via UHF, with other quarry truck drivers and other heavy vehicles regarding location and hazards.
- Stagger arrival and departure times to avoid congestion.
- Drive to the road conditions, which may mean slowing down.
- Be courteous to other road users.

### 6. Entry to the site

- All vehicles with UHF capabilities are to communicate entry to the site via UHF channel 39 upon entry from Pottsville Road.
- All drivers must call up through the zones until reaching the quarry entrance i.e 1 to 2, 2 to 3, 3 to 4, as signposted on route.
- All speed limits are to be obeyed on the access road.
- No overtaking any vehicle at any stage on access road.
- Give way to vehicles exiting the site on two culvert bridges both signposted.
- No texting or mobile phone use unless mobile hands free kit available.



© 2019. Whilst every care has been taken to prepare this map, GHD (and Sixmaps, LPI) make no representations or warranties about its accuracy, reliability, completeness or suitability for any particular purpose and cannot accept liability and responsibility of any kind (whether in contract, tort or otherwise) for any expenses, losses, damages and/or costs (including indirect or consequential damage) which are or may be incurred by any party as a result of the map being inaccurate, incomplete or unsuitable in any way and for any reason Data source: Aerial Imagery: Sixmaps (2017 - NSW LPI), LPI DCDB: Cadastre, 2012; LPI DTDB: Topo base data, 2012. Created by:apmiller



### Figure 2 Zone signs on entry and exit

### 7. On site

- Communicate position at all times within the site and receive acknowledgement.
- Before passing any mobile plant, drivers must inform operators of their intention and receive acknowledgement before passing and communicate once passed.
- Do not enter any area of the quarry where you are not required or is not your usual place of work. If you do have to be in a different area drivers must call up and wait for acknowledgement before proceeding.
- Drivers must obey all traffic signs within the site.
- Use of mobile phones outside of a vehicle is prohibited unless in a designated area i.e. office.
- Use of mobile phones while driving within the site is prohibited unless using a hands free kit.
- If waiting to be loaded, or being loaded, drivers must wait in their vehicle.
- To receive a docket, drivers must wait in their vehicle until loader driver informs them it's safe to walk over to the loader. The loader must be stationary and bucket on the ground.
- Park vehicles in designated parking areas only.



### Figure 3 Example of on site sign

### 8. Tarp station

- Appropriately cover/secure loads before leaving the quarry site.
- Ensure drawbars, tailgates, side combing rails and duals are clear of rocks before leaving the quarry site.
- Only tarp trucks at the designated tarping station on the exit road.
- No tarping within the sand plant load out area.

### 9. Livestock

- Quarry Manager to communicate to all vehicles via UHF Channel 39 of livestock on roadway.
- All drivers to reduce speed and be prepared to stop until livestock cleared.
- Never sound any horns, rev engines or try to drive through livestock (unless stockman has indicated it is safe to do so).

### 10. Exiting the site

- Only exit quarry over the sand grate bars unless given permission by Quarry Manager to use an alternative exit.
- All vehicles with UHF capabilities are to communicate exit of the site via UHF channel 39 to Pottsville Road. Those without must have either site office staff or loader driver communicate on UHF that a vehicle is leaving site and heading for the exit.

- All must call up through the zones until reaching the sand plant exit on Pottsville Road i.e 4 to 3, 3 to 2, 2 to 1, as signposted on route.
- All speed limits are to be obeyed on the road.
- No overtaking any vehicle at any stage.
- No texting or mobile phone use unless mobile hands free kit available.
- **STOP** at the Pottsville Road intersection and only exit when safe.

## **11. Incident response**

In the event of an emergency:

- Stop your vehicle immediately and secure it.
- Ensure the safety of those around you and yourself.
- Dial 000 and seek support from Police, Ambulance or Fire Brigade as required.
- Contact the Quarry Manager and advise of the emergency.
- Await further instructions from the Quarry Manager or your supervisor.

If a heavy vehicle driver becomes aware of a hazard (e.g. mud tracking) or a circumstance that the driver considers dangerous (e.g. stock on the road) the driver must immediately inform the Quarry Manager and if necessary must deal with the hazard.

Report all hazards and incidents to the Quarry Manager and, if relevant, Council and RMS.

## 12. Employee acknowledgement

I hereby acknowledge receipt of and training in the Driver's Code of Conduct for Dunloe Sand Quarry and commit to implementing the requirements.

Name	Signature
### Appendix C – Pre-start check

# Holcim

### FA5.4.001

Week Start Date:		Machine Location:							
SMU Reading Start of the Week: SMU Reading End of the Week: Note: Please fill in each space Wednesday Thursday			Machine Model:						
			Unit Number: Day - A/No				Day - A/Noon	on - Night	
			Friday	Saturday	Sunday	Monday	Tuesday	Total =	
Requires Attention	1U Hrs								Total:
Tick if Ok	ift Hrs								Total:
N/A Not Applicable	ilisation	%	%	%	%	%	%	%	Total: %
	Lizer	70	Jure:	Hire:	Hrs:	Hrs:	Hrs:	Hrs:	Total:
n Shift Planned Downtime	Hrs:		Hrs.	11.5.	Heat	Liro:	Hre:	Hrs	Total
n Shift Unplanned Downtime	Hrs:	11000	Hrs:	Hrs:			115.	1113.	
Walk Around Inspecti	ons	( And the second	T T	15 273, AV					Last Service
. Ground Engaging Tools		_							Hrs
2 lyres / Rims / wheel Nuts									250 1000
<ol> <li>Fluid Leaks(engine oil, coolar</li> </ol>	it, hyd, etc.)				1				□500□2000
<ol> <li>Steps / Ladders / Handrail:</li> </ol>	5								2750 3 Month
5. Mirrors and Windows									Next Service
3. Clean Pre-Cleaner									Hrs
7. Air filter Restriction Indicat	ior						-		250 1000
<ol><li>Air Leaks / Drain Air Tanks</li></ol>				1					500 2000
<ol> <li>Loose Bolts</li> </ol>									0750 3 Month
10. Grease Machine		Container.	and the second second second	A CONTRACTOR OF STREET	Contractor States	And the second second	A COLORED	COVO 1999 9	ACCURE OF
Check Fluids	Contraction of the local		C THERE IS NOT STOLEN				Ok / Amount	Ok / Amount	1
	Ok / A	lmount d	Ok / Amount Added	Ok / Amount Added	Added	Added	Added	Added	
11 Fuel	71000								
17. Engine Oils									
12. Engine Coolant									
14. Transmission Oils			1						
16. Hydraulic Oile			-						
16 Bump Drive Oils (Excavat	orl								
10. Fump Drive Oils (Excavat					CASES TO LEASE	addition of the			
17 EMS / Gauges / Horp	1	and the second second	1	1	1				Mark E.
18 Lights (eg. brake, reverse	etc.)								
10. Bockup Alarm / Camera /	Proximity								Conversion of the
20 Minors / Masher									
21 Air Conditioner									
22 Supplemental Steering S	vstem								
23. Steering System									
24. Braking System and Park	Brake								- El Calor
25 Secondary / Emergency	Brake								
26 Seat and Seat Belts									1111020
27 Two Way Radio									ALL AND DE
28 Body/Hoist Up Warning D	Devices								14-4-6
Operator's Name:									
Signature:									
General Cleanliness	100					Contract of State			
29. Clean Cab, Windows, Mir	rors, Lights								
30. No Loose Items in Cab									
Weekly Operator Ch	eck			ALC: NO		Ale the second		and the second	
31. Clean Battery Box / Term	inals		32. Check Be	ly Guards		37. Chec	k Auto Lube		
33. Check Tyre Pressure			34.Clean Mad	hine / Washdow	wn	38. Chec	k Battery Fluid	Levels	
35. Check Fire Extinguishers	s Charge		36.Tyre Chair	ins 39. Fire Suppression System			stem		
Operator's Signature: Supervisor's		Signature: Workshop Supervisor's Signature:				Signature			
Date			Date:	Date:					
Comments & Reasons for I	Downtime & Rep	airs:							

COPIES: (White -- Maintenance Supervisor) (Pink -- Remains in log book) IOTE: Add Operator comments on back of White Copy if required. Maintenance comments to be entered on back of Pink Copy. OTE: "SMU" stands for –Service Meter Units. "Shift Hrs" is regular site hours each day (normally 8 to 10) "Utilisation" SMU Hrs / Shift Hrs

GHD 230 Harbour Drive Coffs Harbour NSW 2450 T: 61 2 6650 5600 F: 61 2 6650 5601 E: cfsmail@ghd.com

#### © GHD 2019

This document is and shall remain the property of GHD. The document may only be used for the purpose for which it was commissioned and in accordance with the Terms of Engagement for the commission. Unauthorised use of this document in any form whatsoever is prohibited. 22-SO-976847825-

1/https://projects.ghd.com/oc/Newcastle3/holcimdunloesandquar/Delivery/Documents/2220056\_RP T\_Dunloe Traffic Management Plan.docx

#### **Document Status**

Revision	Author	Reviewer		Approved for Issue		
		Name	Signature	Name	Signature	Date
0	B Luffman	S Lawer		S Lawer		20/05/2019
1	B Luffman	S Lawer	fan	S Lawer	fa	29/05/2019

## www.ghd.com



# **Appendix I** – Aboriginal cultural heritage management plan



### Holcim (Australia) Pty Ltd

Dunloe Sand Quarry Aboriginal Cultural Heritage Management Plan

December 2019

This report: has been prepared by GHD for Holcim (Australia) Pty Ltd and may only be used and relied on by Holcim (Australia) Pty Ltd for the purpose agreed between GHD and the Holcim (Australia) Pty Ltd as set out in this report.

GHD otherwise disclaims responsibility to any person other than Holcim (Australia) Pty Ltd arising in connection with this report. GHD also excludes implied warranties and conditions, to the extent legally permissible.

The services undertaken by GHD in connection with preparing this report were limited to those specifically detailed in the report and are subject to the scope limitations set out in the report.

The opinions, conclusions and any recommendations in this report are based on conditions encountered and information reviewed at the date of preparation of the report. GHD has no responsibility or obligation to update this report to account for events or changes occurring subsequent to the date that the report was prepared.

The opinions, conclusions and any recommendations in this report are based on assumptions made by GHD described in this report. GHD disclaims liability arising from any of the assumptions being incorrect.

GHD has prepared this report on the basis of information provided by Holcim (Australia) Pty Ltd and others who provided information to GHD (including Government authorities), which GHD has not independently verified or checked beyond the agreed scope of work. GHD does not accept liability in connection with such unverified information, including errors and omissions in the report which were caused by errors or omissions in that information.

### **Table of contents**

1.	Introc	luction	1
	1.1	Objectives	1
	1.2	Targets	1
	1.3	Consultation	1
2.	Envir	onmental requirements	2
	2.1	Legislation	2
	2.2	Guidelines	2
	2.3	Conditions of approval	2
3.	Existi	ng environment	4
4.	Envir	onmental control measures	6
5.	Comp	pliance management	7
	5.1	Environmental Inspections and monitoring	7
	5.2	Contingency plan	7
	5.3	Reporting	7
6.	Revie	w and improvement	8

### **Table index**

Table 2-1	Consent conditions relevant to ACHMP	.3
Table 4-1	Environmental controls and mitigation measures	6

### **Figure index**

Figure 3-1	Sand Ridge Areas	5
------------	------------------	---

### **Appendices**

Appendix A – Agency consultation

### 1. Introduction

This Aboriginal Cultural Heritage Management Plan (ACHMP) forms part of the Environmental Management Strategy (EMS) for Dunloe Sand Quarry. This ACHMP has been prepared to meet the requirements of the Minister's Conditions of Approval (CoA) outlined in Development Consent No. 06\_0030, the mitigation measures outlined in MOD2 (GHD 2017), the Environmental Impact Statement (EIS) (Planit 2007) and relevant legislation.

#### 1.1 **Objectives**

The key objective of the ACHMP is to ensure appropriate controls and procedures are implemented in order to avoid damage or disturbance of heritage items.

#### **1.2 Targets**

The following targets have been established for the management of both historical and Aboriginal cultural heritage during the operational lifetime of Dunloe Sand Quarry:

- Ensure full compliance with the relevant legislative requirements and CoA
- No damage to heritage items
- All site staff and contractors trained on unexpected finds protocol

#### **1.3** Consultation

Extensive consultation was undertaken with the local community during preparation of the EIS and MOD2. Any concerns identified by relevant stakeholders were addressed in the EIS and MOD2 mitigation measures which have been incorporated into this ACHMP.

As per CoA 32(a), Schedule 3, the Office of Environment and Heritage (OEH) (now Biodiversity and Conservation Division (BCD)) and relevant Aboriginal communities were consulted in relation to the ACHMP. Evidence of the consultation is provided in Appendix A. No response has been received from the relevant Aboriginal communities to date.

Agency	Response
Biodiversity and Conservation Division	
Revise the Aboriginal Cultural Heritage Assess	ment report to:
Include a statement confirming the report was prepared in accordance with the <i>Code of</i> <i>Practice for Archaeological Investigation of</i> <i>Aboriginal Objects in NSW</i> 2010	Updated but not part of the ACHMP
Remove references to a requirement for an Aboriginal Heritage Impact Permit	Updated but not part of the ACHMP
Ensure the recommendations are consistent with the requirement for a Care Agreement to remove Aboriginal objects from the approved project boundary for long-term management if required	Updated and recommendations incorporated into Table 4-1 of the ACHMP
Revise the Aboriginal Cultural Heritage Assessment to replace references to the Office of Environment and Heritage (OEH) with the Biodiversity and Conservation Division (BCD)	Updated

### 2. Environmental requirements

#### 2.1 Legislation

Legislation relevant to heritage management includes:

- Environmental Planning and Assessment Act 1979
- National Parks and Wildlife Act 1974
- National Parks and Wildlife Amendment Regulation 2010
- Native Title Act 1993 (Commonwealth)
- Heritage Act 1977 (Heritage Act)
- Aboriginal Land Rights Act 1983
- Native Title Act 1993 (Commonwealth)
- Environment Protection Biodiversity Conservation Act 1999 (EPBC Act) (Commonwealth)

Further discussion of the above legislation is provided in the EMS, as well as the EIS and MOD2.

#### 2.2 Guidelines

The following guidelines have been reviewed during development of this ACHMP:

- Aboriginal cultural heritage consultation requirements for proponents 2010 (DECCW, 2010)
- Due Diligence Code of Practice for the Protection of Aboriginal Objects in NSW (DECCW, 2010)
- Guide to Investigating, Assessing and Reporting on Aboriginal Cultural Heritage in NSW (OEH, 2011)
- NSW Heritage Manual Investigating Heritage Significance: Draft Guideline (NSW Heritage Office, 2004)
- NSW Heritage Manual Assessing Heritage Significance (NSW Heritage Office, 2001)

#### 2.3 Conditions of approval

The consent conditions relevant to this ACHMP are listed in Table 2-1. A cross reference is also included to indicate where the condition is addressed in this ACHMP or other environmental management documents.

#### Table 2-1 Consent conditions relevant to ACHMP

	Reference
The Proponent must prepare an Aboriginal Cultural Heritage Management Plan to the satisfaction of the Secretary. This plan must:	Section 4
(a) be prepared in consultation with OEH and all relevant Aboriginal communities;	Section 1.3
(b) be submitted to the Secretary for approval prior to commencement of construction; and	
<ul> <li>(c) include a:</li> <li>program for additional archaeological survey/s of the disturbance area;</li> <li>description of the measures that would be implemented to salvage any identified Aboriginal sites within the disturbance area;</li> <li>description of the measures that would be implemented to protect any Aboriginal sites outside the disturbance area; and</li> <li>description of the measures that would be implemented if any new Aboriginal objects or skeletal remains are discovered during the project.</li> <li>The Proponent must implement the plan as approved by the Secretary.</li> </ul>	Section 4
The Proponent must ensure that the management plans required under this approval are prepared in accordance with any relevant guidelines, and include:	
(a) a summary relevant background or baseline data;	Section 3
<ul> <li>(b) a description of:</li> <li>the relevant statutory requirements (including any relevant approval, licence or lease conditions);</li> <li>any relevant limits or performance measures/criteria; and</li> <li>the specific performance indicators that are proposed to be used to judge the performance of, or guide the implementation of, the project or any management measures;</li> </ul>	Section 1.2 and Section 2
(c) a description of the measures that to be implemented to comply with the relevant statutory requirements, limits, or performance measures/criteria;	Section 4
<ul> <li>(d) a program to monitor and report on the:</li> <li>impacts and environmental performance of the project; and</li> <li>effectiveness of any management measures (see (c) above);</li> </ul>	Section 5.1
(e) a contingency plan to manage any unpredicted impacts and their consequences and to ensure that ongoing impacts reduce to levels below relevant impact assessment criteria as quickly as possible;	Section 5.2
(f) a program to investigate and implement ways to improve the environmental performance of the project over time	Section 6
<ul> <li>(g) a protocol for managing and reporting any:</li> <li>incidents;</li> <li>complaints;</li> <li>non-compliances with statutory requirements; and</li> <li>exceedances of the impact assessment criteria and/or performance criteria; and</li> <li>(h) a protocol for periodic review of the plan.</li> </ul>	Refer to the EMS Section 6
	The Proponent must prepare an Aboriginal Cultural Heritage Management Plan to the satisfaction of the Secretary. This plan must: (a) be prepared in consultation with OEH and all relevant Aboriginal communities; (b) be submitted to the Secretary for approval prior to commencement of construction; and (c) include a: • program for additional archaeological survey/s of the disturbance area; • description of the measures that would be implemented to salvage any identified Aboriginal sites within the disturbance area; • description of the measures that would be implemented to protect any Aboriginal sites outside the disturbance area; and • description of the measures that would be implemented if any new Aboriginal objects or skeletal remains are discovered during the project. The Proponent must ensure that the management plans required under this approval are prepared in accordance with any relevant guidelines, and include: (a) a summary relevant background or baseline data; (b) a description of: • the relevant statutory requirements (including any relevant approval, licence or lease conditions); • any relevant limits or performance measures/criteria; and • the specific performance indicators that are proposed to be used to judge the perjoert or any management measures; (c) a description of the measures that to be implemented to comply with the relevant statutory requirements, limits, or performance measures/criteria; (d) a program to monitor and report on the: • impacts and environmental performance of the project; and • effectiveness of any management measures (see (c) above); (e) a contingency plan to manage any unpredicted impacts and their consequences and to ensure that ongoing impacts reduce to levels below relevant impact assessment criteria as quickly as possible; (f) a program to investigate and implement ways to improve the environmental performance of the project over time (g) a protocol for managing and reporting any: • incidents; • complaints; • non-compliances with statutory requirements; and • exceeda

### 3. Existing environment

A search of the Aboriginal Heritage Information Management System (AHIMS) database, consultation and site survey by Everick Heritage Consultants and the LALC, revealed no areas within Lot 162 in DP 755721 or Lots 1 and 2 in DP 780199 were identified as containing any Aboriginal places of cultural heritage which would be impacted upon by the proposal.

However, an area of sand ridges to the east of the extraction area (Figure 3-1) has been identified as being of potential cultural sensitivity. This area was identified as potentially containing subsurface deposits of Aboriginal cultural material, and there is a low possibility that this area may contain Aboriginal burials.

The approved excavation strategy was completed in late 2018 and early 2019 in accordance with Condition 32(c) of Schedule 3 of the development consent. The implementation of the excavation strategy identified no Aboriginal objects or places as a result of the pedestrian survey and subsequent archaeological test excavations. It was determined that the project area possesses nil archaeological significance. The outcomes of the excavation strategy is reported in the *Aboriginal Cultural Heritage Assessment Report – Dunloe Sand Quarry, Pottsville, NSW* (RPS 2019).

The area outside of the sand ridges has a history of high disturbance, and it is considered unlikely that any cultural material would remain within this area.

In conjunction to the above surveys, no evidence in the form of buildings, equipment, 'springboard trees', artefacts or any other material which might be considered to be of European heritage value were identified.



Figure 3-1 Sand Ridge Areas

### 4. Environmental control measures

Environmental requirements and control measures are identified in the CoA and the EIS. Specific measures and requirements to address potential heritage impacts are outlined in Table 4-1.

Ref.	Environmental Management Measure	Timing	Responsibility
H1	All Holcim (Australia) Pty Ltd personnel and subcontractors must be advised of the requirements of the NPWS Act 1974 that it is an offence for any person to knowingly destroy, deface, damage or permit destruction, or defacement to an Aboriginal relic or place without the consent of the Director General of BCD.	Operation	Quarry Manager
H2	In the event that skeletal remains are identified, work must cease immediately in the vicinity of the remains and the area must be cordoned off. The proponent must contact the local NSW Police who will make an initial assessment as to whether the remains are part of a crime scene or possible Aboriginal remains. If the remains are thought to be Aboriginal, BCD must be contacted on Enviroline 131 555. A BCD officer will determine if the remains are Aboriginal or not; and a management plan must be developed in consultation with the relevant Aboriginal stakeholders before works recommence.	Operation	All employees and contractors
H3	If any Aboriginal archaeological deposits and/or objects are found during operation, then all work is to cease in the immediate vicinity of the deposits and/or objects, the area is to be demarcated and BCD, on Enviroline 131 555, the participating Aboriginal stakeholders and a qualified archaeologist are to be notified.	Operation	Quarry Manager

#### Table 4-1 Environmental controls and mitigation measures

#### 5.1 Environmental Inspections and monitoring

Routine weekly inspections by the Quarry Manager (or delegate) will occur throughout the operational lifetime of the quarry using the *Environmental Inspection Checklist* in Appendix A of the Environmental Monitoring Program. This will ensure the actions in Table 4-1 are implemented, monitored, maintained and reported.

#### 5.2 Contingency plan

In the unlikely event that Aboriginal cultural materials or skeletal remains are uncovered, the actions outlined in Table 4-1 are to be followed.

#### 5.3 Reporting

The general reporting requirements are described in Section 8.4 of the EMS. In relation to the monitoring, the routine monitoring will be recorded on the *Environmental Inspection Checklist* in the Environmental Monitoring and Management Plan.

If Aboriginal cultural materials or skeletal remains are uncovered, this will need to be documented in accordance with Table 4-1 and BCD requirements.

A summary of any finds of Aboriginal cultural materials or skeletal remains will be presented in the Annual Report (refer to Section 8.4.1 of the EMS).

### 6. **Review and improvement**

Continuous improvement of this ACHMP will be achieved in accordance with Section 9 of the EMS, through the ongoing evaluation of environmental management performance against environmental policies, objectives and targets.

The continuous improvement process is designed to:

- Identify areas of opportunity for improvement of environmental management and performance.
- Determine the cause or causes of non-conformances and deficiencies.
- Develop and implement a plan of corrective and preventative action to address any nonconformances and deficiencies.
- Verify the effectiveness of the corrective and preventative actions.
- Document any changes in procedures resulting from process improvement.

### Appendices

**Appendix A** – Agency consultation



Our Ref: DOC19/676169 Your Ref: 06\_0030

> GHD 230 Harbour Drive Coffs Harbour NSW 2450

Attention: Mr Ben Luffman

#### Dear Mr Luffman

### Subject: Dunloe Quarry MOD 2 - Landscape and Aboriginal Cultural Heritage Management Plans

Thank you for your letter dated 8 August 2019 about the amended management plans for the Dunloe Quarry MOD 2 seeking comments from the Biodiversity and Conservation Division (BCD) of the NSW Department of Planning, Industry and Environment. I appreciate the opportunity to provide input.

The Biodiversity and Conservation Division was formerly part of the Office of Environment and Heritage (OEH) but now forms part of the new Environment, Energy and Science Group in the Department of Planning, Industry and Environment (see https://www.dpie.nsw.gov.au).

We have reviewed the documents supplied and advise there are several issues apparent with the Landscape Management Plan (LMP), particularly the koala management plan, which lacks adequate detail. The main issues include:

- a) no description of the author qualifications in the Landscape Management Plan (LMP);
- b) insufficient or incorrect references to approval conditions and document sections;
- c) lack of supporting information in the Koala Management Plan (KMP);
- d) the need to refine koala management, monitoring and contingency measures.

These issues are discussed in detail in Attachment 1 to this letter.

Prior to finalising the landscape and Aboriginal cultural heritage management plans we recommend that GHD:

- 1. Revises the Landscape Management Plan to:
  - a) include the relevant qualifications and experience of the contributors to demonstrate compliance with Project Approval 27(a); and
  - b) indicate in Table 2.1 that Project Approval Condition 28 clauses (h) and (i) are addressed in Appendix C.

Locked Bag 914 Coffs Harbour NSW 2450 Federation House, Level 8, 24 Moonee Street Coffs Harbour NSW 2450 Tel: (02) 6659 8200 Fax: (02) 6659 8281 ABN 20 770 707 468 www.dpie.nsw.gov.au

- 2. Revises the Rehabilitation and Revegetation Management Plan to include Project Condition 28 in the list of conditions addressed.
- 3. Revises the Koala Management Plan to:
  - a) replace references to sections and tables of other management plans with the relevant content being referred to in order to minimise potential errors resulting from subsequent management plan revisions or amendments;
  - b) include mapping of koala habitat, koala records and potential koala movement corridors (i.e. habitat links) within and adjacent to the subject land and along the haul road between the quarry site and the Pottsville Road intersection;
  - c) acknowledge the possibility of infrequent koala movements during hours of quarry operation;
  - d) identify the most likely areas of interaction between koalas and quarry vehicles (e.g. koala habitat links);
  - e) include a proposed amendment to the quarry induction process to include an explanation of the legal consequences of unauthorised clearing of native vegetation on the quarry site;
  - f) include provision of compensatory koala food tree plantings as a contingency measure in the event of unauthorised clearing taking place;
  - g) ensure the proposed monitoring methodology focuses on identifying areas of koala activity susceptible to road strike rather than attempting to identify temporal changes in koala densities.
  - h) reduce the proposed koala road-strike threshold for management action from three koalas for the year to any koala at any time
- 4. Revise the Aboriginal Cultural Heritage Assessment Report to:
  - a) include a statement confirming the report was prepared in accordance with the Code of Practice for Archaeological Investigation of Aboriginal Objects in NSW 2010;
  - b) remove references to a requirement for an Aboriginal Heritage Impact Permit; and
  - c) ensure the recommendations are consistent with the requirement for a Care Agreement to remove Aboriginal objects from the approved project boundary for long-term management if required.
- 5. Revise the Aboriginal Cultural Heritage Management Plan to replace references to the Office of Environment and Heritage (OEH) with the Biodiversity and Conservation Division (BCD).

If you have any further questions about this issue, Mr Don Owner, Senior Conservation Planning Officer, Biodiversity and Conservation, can be contacted on 6659 8233 or at don.owner@environment.nsw.gov.au.

Yours sincerely

Vinition Juny 4 September 2019

DIMITRI YOUNG Senior Team Leader Planning, North East Branch <u>Biodiversity and Conservation</u>

Contact officer: DON OWNER 6659 8233

Enclosure: Attachment 1: Detailed BCD Comments - Dunloe Quarry MOD 2 - Landscape and ACH Management Plans

Attachment 1: Detailed Biodiversity and Conservation Comments – Dunloe Quarry MOD 2 – Landscape and Aboriginal Cultural Heritage Management Plans

#### Landscape Management Plan

#### Author Qualifications

Project Approval 27(a) requires the plan to be prepared by suitably qualified consultants, including specialist hydrologist, coastal engineer, wetlands ecologist and landscape architect. The Landscape Management Plan (LMP) does not contain the qualifications of the contributing authors, which is necessary to demonstrate this condition has been met.

#### BCD Recommendation:

1. Revise the LMP to include the relevant qualifications and experience of the contributors to demonstrate compliance with Project Approval 27(a).

#### Project Approval Condition 28

The Rehabilitation and Revegetation Management Plan (RRMP) contained in Appendix A of the LMP claims to address the requirements of Conditions 26, 27a and 29 of the Project Approval. However, Project Approval Condition 28, which specifies what must be included in the RRMP, has not been referred to in the RRMP.

#### BCD Recommendation:

2. Revise the RRMP to include Project Condition 28 in the list of conditions addressed.

#### Koala Management Plan

Table 2.1 of the LMP indicates Project Approval Condition 28 clauses (h) and (i), which relate to koala management, are addressed in the RRMP provided in Appendix A. However, these conditions have not been addressed in the RRMP. Instead, they have been addressed in the Koala Management Plan (KMP) contained in Appendix C of the LMP.

#### BCD Recommendation:

3. Revise Table 2.1 of the LMP to indicate Project Approval Condition 28 clauses (h) and (i) are addressed in Appendix C.

Section 1.3 of the KMP refers to Section 5.4.1 of the LMP for details of the KMP reporting provisions. However, the LMP does not contain a Section 5.4.1. It is also indicated in this section that the presence of koalas will be noted during the monitoring program set out in Table 5.1 of the LMP. However, Table 5.1 provides a very brief summary of the monitoring program for the rehabilitation area, which would not include monitoring of the haul road.

#### BCD Recommendation:

4. Revise the KMP to replace references to sections and tables of other management plans with the relevant content being referred to in order to minimise potential errors resulting from subsequent management plan revisions or amendments.

The KMP does not provide any supporting information to inform or provide context to the proposed koala management or monitoring provisions.

#### BCD Recommendation:

5. Revise the KMP to include mapping of koala habitat, koala records and potential koala movement corridors (i.e. habitat links) within and adjacent to the subject land and along the haul road between the quarry site and the Pottsville Road intersection.

Section 1.2.1 of the KMP indicates the risk of koala mortalities due to quarry-related vehicle strike will be low due to the quarry operating hours not coinciding with the main periods of koala movement. This assumption is generally accurate. However, although infrequent, some koala movements may still occur during the quarry operating hours (e.g. movements resulting from territorial disputes or during dispersal of young from natal areas).

#### BCD Recommendations:

- 6. Revise the KMP to acknowledge the possibility of infrequent koala movements during hours of quarry operation.
- 7. Revise the KMP to identify the most likely areas of interaction between koalas and quarry vehicles (e.g. koala habitat links).

Section 1.2.2 of the KMP briefly discusses the potential impacts of unauthorised clearing of koala habitat. However, no preventative or contingency measures other than clearly demarcating the limit of authorised clearing have been proposed.

#### BCD Recommendation:

- Revise the KMP to include a proposed amendment to the quarry induction process to include an explanation of the legal consequences of unauthorised clearing of native vegetation on the quarry site.
- 9. Revise the KMP to include provision of compensatory koala food tree plantings as a contingency measure in the event of unauthorised clearing taking place.

Proposed monitoring of the frequency of koala sightings would be based on incidental records rather than application of a systematic repeatable sampling method. The resulting dataset will not provide a reliable measure for determining temporal changes in koala occupancy levels or local population size, as intended in the KMP.

Nevertheless, such information may have some use in identifying areas of important koala occupancy, which could then be used to identify potential road strike '*black-spots*' and to formulate mitigation measures (e.g. speed limits, warning signage, traffic calming devices etc.).

#### BCD Recommendation:

10. Revise the KMP to ensure the proposed monitoring methodology focuses on identifying areas of koala activity susceptible to road strike rather than attempting to identify temporal changes in koala densities.

Table 1 of the KMP lists one of the proposed management triggers as being '*quarry-related vehicle koala strikes reach or exceed three for the year*'. This trigger threshold is too high given that three koalas are likely to represent a significant proportion of the koala population utilising the subject land and adjoining areas in any given year.

#### BCD Recommendation:

11. Revise the KMP to reduce the proposed koala road-strike threshold for management action from three koalas for the year to any koala at any time.

#### Aboriginal Cultural Heritage Management

RPS prepared an Aboriginal cultural heritage assessment report (ACHAR) to investigate the potential for harm to Aboriginal cultural heritage resulting from proposed expansion of the extraction boundaries into currently fenced off sand ridge areas. However, the ACHAR was not undertaken under the defence of being in accordance with the *Code of Practice for Archaeological Investigation of Aboriginal Objects in NSW 2010*.

References in the ACHAR to the need for an Aboriginal Heritage Impact Permit (AHIP) are incorrect given any proposed harm to Aboriginal objects from the approved project would be regulated by a management plan approved by the Secretary of the Department rather than an AHIP.

However, any Aboriginal objects identified whilst undertaking the approved project works could be lawfully removed from within the approved project boundary if the removal is consistent with the will of the Registered Aboriginal Parties and undertaken in accordance with a Care Agreement issued by the Department of Planning, Industry and Environment under section 85A(1)(c) of the National Parks and Wildlife Act 1974.

We note that one of the three recommendations provided in the ACHAR for inclusion in the updated Aboriginal Cultural Heritage Management Plan (ACHMP) appears inconsistent with the requirement for a Care Agreement to remove Aboriginal objects from the approved project boundary for long term management if required.

The relevant state government point of contact provided in the ACHMP for various aspects of Aboriginal cultural heritage recording and reporting (i.e. OEH) has recently changed and this should be amended to the Biodiversity and Conservation Division (BCD).

#### BCD Recommendations:

- 12. Revise the ACHAR to:
  - a) include a statement confirming the ACHAR was prepared in accordance with the Code of Practice for Archaeological Investigation of Aboriginal Objects in NSW 2010;
  - b) remove references to a requirement for an AHIP.
  - c) ensure the recommendations are consistent with the requirement for a Care Agreement to remove Aboriginal objects from the approved project boundary for long term management if required.
- 13. Revise the ACHMP to replace references to the Office of Environment and Heritage (OEH) with the Biodiversity and Conservation Division (BCD).

#### **Ben Luffman**

From:	Jackie McDonald <mctogo2@gmail.com></mctogo2@gmail.com>
Sent:	Monday, 14 October 2019 4:34 PM
То:	Ben Luffman
Cc:	Paul J Buxton; Victoria Musgrove (InTouch)
Subject:	Re: Dunloe Quarry - Heritage Management Plan
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
RepoType:	Job

Hello Ben Sorry for our delay in responding. I have spoken to both Paul and Jason and I am satisfied with the updated Aboriginal Cultural Heritage Management Plan for Dunloe Sand Quarry. Regards Jackie McDonald

Sent from my iPhone

On 4 Oct 2019, at 3:53 pm, Ben Luffman <<u>Ben.Luffman@ghd.com</u>> wrote:

Thanks Jackie

#### Regards

#### Ben Luffman | A GHD Associate

B.App.Sc. (Hons) | Grad.Dip. Urban and Regional Planning | Environmental Auditor **Technical Director - Environment** 

#### GHD

#### Proudly employee owned

T: +61 2 6650 5613 | M: +61 415 271 319 | E: <u>ben.luffman@qhd.com</u> 230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.ghd.com</u>

#### Connect

<image001.png> <image002.png> <image003.png> <image004.png>

WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION

GHD acknowledges the Traditional Owners of Country throughout Australia. <image005.png> We pay respect to their continuing culture and Elders past, present and emerging. <u>Click here</u> to learn about our Reconciliation Action Plan.

Please consider the environment before printing this email 1 ream of paper = 6% of a tree / 5.4kg CO2 in the atmosphere | 3 sheets of A4 paper = 1 litre of water

From: Jackie McDonald <<u>mctogo2@gmail.com</u>> Sent: Friday, 4 October 2019 2:22 PM To: Ben Luffman <<u>Ben.Luffman@ghd.com</u>>

#### **Ben Luffman**

From: Sent: To: Subject:	Paul Buxton <paul.j.buxton@gmail.com> Monday, 14 October 2019 4:42 PM Ben Luffman Dunloe Sand Quarry updated ACHMP</paul.j.buxton@gmail.com>
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
<b>RepoType:</b>	Job

Hello Ben,

I have spoken to my sister Jackie and I concur with her and am satisfied with the updated ACHMP for the Dunloe Sand Quarry. Sorry for the delay.

Regards

Paul Buxton

### **Buxtons** Work Town n Country Wear Shop 4/15 William Street Beaudesert Qld 4285 Home/Office: (07) 55454307 Shop: (07) 55411373

Email: paul.j.buxton@gmail.com

This e-mail has been scanned for viruses

#### **Ben Luffman**

From: Sent:	Sites <sites@tblalc.com> Thursday, 26 September 2019 2:11 PM</sites@tblalc.com>
To:	Ben Luffman; Admin; paul.j.buxton@gmail.com; hesion@live.com.au; mctogo2 @gmail.com
Cc:	Victoria Musgrove; Cultural Heritage
Subject:	RE: Dunloe Quarry - Heritage Management Plan
CompleteRepository:	2220056
Description:	Dunloe EMP
JobNo:	20056
OperatingCentre:	22
RepoEmail:	2220056@ghd.com
RepoType:	Job

#### Hi Ben

TBLALC concurs with the two reports:

- Holcim (Australia) Pty Ltd Dunloe Sand Quarry Aboriginal Cultural Heritage Management Plan August 2019, and
- RPS Aboriginal Cultural Heritage Assessment Report Dunloe Sand Quarry, Pottsville NSW 1.2 September 2019 which, I note, includes the previous comments on the draft report by the Registered Aboriginal Parties.

My only comment, *which is a very general one*, is that references to the previous 'disturbance' of an area as a qualifying factor in the conduct of ACH assessments is a recurring dilemma. I am aware that disturbance is mentioned in this way in various government publications. The physical destruction or removal of evidence doesn't obliterate history (in fact, often, the act of obliteration is a major part of the historical record). There are many other cultures that make this same point – often with religious fervour. Aboriginal culture, more than most others, is *totally* connected to country and can't be removed. Logically, 'disturbance' is only a relevant consideration where it is *likely* that the *physical* record of Aboriginal culture has actually been destroyed.

For clarity, I am <u>not</u> suggesting that this is relevant to the Dunloe Sand Quarry. Quite the opposite, we have been most appreciative of the detailed evaluation that has been undertaken.

#### Kind Regards

#### Mauríce Gannon

Maurice Gannon Conservation Planning Officer **Tweed Byron Local Aboriginal Land Council** PO Box 6967, Tweed Heads South NSW 2486 Ph: 07 5536 1926 Mb: 0407 643 349 www.tblalc.com



The information in this message is intended only for the recipient named in this email. If you are not that recipient, you may not read, copy, distribute or act upon this message as the information it contains may be privileged and confidential. If you have received this message in error, please notify the sender. Thank you for your cooperation.

From: Ben Luffman [mailto:Ben.Luffman@ghd.com]
Sent: Wednesday, 18 September 2019 2:23 PM
To: Admin <admin@tblalc.com>; Sites <sites@tblalc.com>; paul.j.buxton@gmail.com; hesion@live.com.au; mctogo2@gmail.com
Cc: Victoria Musgrove <victoria.musgrove@lafargeholcim.com>
Subject: Dunloe Quarry - Heritage Management Plan

Hi,

We have updated the management plans for Dunloe Quarry following the recent approval of MOD2. The conditions of the Project Approval – SSD 06\_0030 require the Heritage Management Plan to be prepared in consultation with relevant Aboriginal communities. We have therefore attached the relevant plan for review.

The updates have mainly been a reformatting to remove duplication, inclusion of additional information to address the new requirements of the conditions and removing actions no longer relevant because further investigations have found no evidence of Aboriginal heritage at the site. The report prepared following the further investigations is also attached.

We would appreciate your comments by 27 September 2019.

Please contact me if you have any questions.

#### Regards

Ben Luffman | A GHD Associate B.App.Sc. (Hons) | Grad.Dip. Urban and Regional Planning | Environmental Auditor Technical Director - Environment

GHD *Proudly employee owned* T: +61 2 6650 5613 | M: +61 415 271 319 | E: <u>ben.luffman@qhd.com</u> 230 Harbour Drive, Coffs Harbour, NSW, 2450 | <u>www.qhd.com</u>



WATER | ENERGY & RESOURCES | ENVIRONMENT | PROPERTY & BUILDINGS | TRANSPORTATION



GHD acknowledges the Traditional Owners of Country throughout Australia. We pay respect to their continuing culture and Elders past, present and emerging. <u>Click here</u> to learn about our Reconciliation Action Plan.

GHD

230 Harbour Drive Coffs Harbour NSW 2450 T: 61 2 6650 5600 F: 61 2 6650 5601 E: cfsmail@ghd.com

#### © GHD 2019

This document is and shall remain the property of GHD. The document may only be used for the purpose for which it was commissioned and in accordance with the Terms of Engagement for the commission. Unauthorised use of this document in any form whatsoever is prohibited. 22-SO-976847825-

6/https://projects.ghd.com/oc/Newcastle3/holcimdunloesandquar/Delivery/Documents/2220056\_RP T\_Dunloe Heritage Management Plan.docx

Document	Status
----------	--------

Revision	Author	Reviewer		Approved for Issue		
		Name	Signature	Name	Signature	Date
0	B Luffman	S Lawer	S Lawer	S Lawer	S Lawer	19/09/2019
1	B Luffman	S Lawer	ta	S Lawer	ta	18/12/2019

## www.ghd.com



**Appendix J** – Waste management plan



### Holcim (Australia) Pty Ltd

Dunloe Sand Quarry Waste Management Plan

July 2020

This report has been prepared by GHD for Holcim (Australia) Pty Ltd and may only be used and relied on by Holcim (Australia) Pty Ltd for the purpose agreed between GHD and the Holcim (Australia) Pty Ltd as set out in Section 1.1 of this report.

GHD otherwise disclaims responsibility to any person other than Holcim (Australia) Pty Ltd arising in connection with this report. GHD also excludes implied warranties and conditions, to the extent legally permissible.

The services undertaken by GHD in connection with preparing this report were limited to those specifically detailed in the report and are subject to the scope limitations set out in the report.

The opinions, conclusions and any recommendations in this report are based on conditions encountered and information reviewed at the date of preparation of the report. GHD has no responsibility or obligation to update this report to account for events or changes occurring subsequent to the date that the report was prepared.

The opinions, conclusions and any recommendations in this report are based on assumptions made by GHD described in this report. GHD disclaims liability arising from any of the assumptions being incorrect.

GHD has prepared this report on the basis of information provided by Holcim (Australia) Pty Ltd and others who provided information to GHD (including Government authorities), which GHD has not independently verified or checked beyond the agreed scope of work. GHD does not accept liability in connection with such unverified information, including errors and omissions in the report which were caused by errors or omissions in that information.

### **Table of contents**

1.	Introduction		
	1.1	Objectives	1
	1.2	Targets	1
2.	Envir	onmental requirements	2
	2.1	Legislation	2
	2.2	Guidelines	2
	2.3	Conditions of approval	2
	2.4	Environment protection licence	3
3.	Existi	ng environment and impacts	4
	3.1	Existing environment	4
	3.2	Impacts	4
4.	Envir	onmental control measures	5
5.	Monit	oring and reporting	7
	5.1	Environmental inspections and monitoring	7
	5.2	Contingency plan	7
	5.3	Reporting	7
6.	Revie	w and improvement	8

### **Table index**

Table 2-1	Consent conditions relevant to the WMP	2
Table 2-2	EPL conditions relevant to the WMP	3
Table 4-1	Waste management measures	5
Table 5-1	Contingency plan	7

### **Appendices**

Appendix A – On-site sewage management system approval

### 1. Introduction

This Waste Management Plan (WMP) forms part of the Environmental Management Strategy (EMS) for Dunloe Sand Quarry. This WMP has been prepared to meet the requirements of the Minister's Conditions of Approval (CoA) outlined in Development Consent No. 06\_0030, the mitigation measures outlined in MOD2 (GHD 2017), the Environmental Impact Statement (EIS) (Planit 2007), the Environment Protection Licence 13077 (EPL) and relevant legislation.

#### 1.1 **Objectives**

The key objective of the WMP is to ensure appropriate controls and procedures are implemented in order to minimise the impacts to the local environment and community from waste.

#### **1.2 Targets**

The following targets have been established for the management of waste during the operational lifetime of Dunloe Sand Quarry:

- Ensure full compliance with the relevant legislative requirements and CoA
- Waste generation minimised through the hierarchy of waste management priorities
- All waste is managed in accordance with the Waste Classification Guidelines (EPA, 2014)

### 2. Environmental requirements

#### 2.1 Legislation

Legislation relevant to waste management includes:

- Protection of the Environment Operations Act 1997
- Protection of the Environment Operations (Waste) Regulation 2014
- Waste Avoidance and Resource Recovery Act 2001

Further discussion of the above legislation is provided in the EMS, as well as the EIS and MOD2.

#### 2.2 Guidelines

The following guidelines have been reviewed during development of this WMP:

- Waste Classification Guidelines (EPA, 2014)
- Australian Standard AS 1940-2004: The Storage and Handling of Flammable and Combustible Liquids

#### **2.3 Conditions of approval**

The CoA relevant to this WMP are listed in Table 2-1. A cross reference is also included to indicate where the condition is addressed in this WMP or other environmental management documents.

Condition No.	Requirement	Reference
Schedule 3, Condition 12	The Proponent must manage on-site sewage to the satisfaction of Council and EPA. The facility must comply with the requirements of the Environment and Health Protection Guidelines – On-site Sewage Management for Single Households (1998).	Section 4 and Appendix A
Schedule 3, Condition 41	The Proponent must minimise the amount of waste generated by the project to the satisfaction of the Secretary.	Section 4
Schedule 5, Condition 1A	The Proponent must ensure that the management plans required under this approval are prepared in accordance with any relevant guidelines, and include:	
	(a) a summary relevant background or baseline data;	Section 3
	<ul> <li>(b) a description of:</li> <li>the relevant statutory requirements (including any relevant approval, licence or lease conditions);</li> <li>any relevant limits or performance measures/criteria; and</li> <li>the specific performance indicators that are proposed to be used to judge the performance of, or guide the implementation of, the project or any management measures;</li> </ul>	Section 1.2 and Section 2.1
	(c) a description of the measures that to be implemented to comply with the relevant statutory requirements, limits, or performance measures/criteria;	Section 4

#### Table 2-1 Consent conditions relevant to the WMP

Condition No.	Requirement	Reference
	<ul> <li>(d) a program to monitor and report on the:</li> <li>impacts and environmental performance of the project; and</li> <li>effectiveness of any management measures (see (c) above);</li> </ul>	Section 5
	(e) a contingency plan to manage any unpredicted impacts and their consequences and to ensure that ongoing impacts reduce to levels below relevant impact assessment criteria as quickly as possible;	Section 5.2
	(f) a program to investigate and implement ways to improve the environmental performance of the project over time.	Section 6
	<ul> <li>(g) a protocol for managing and reporting any:</li> <li>incidents;</li> <li>complaints;</li> <li>non-compliances with statutory requirements; and</li> <li>exceedances of the impact assessment criteria and/or performance criteria; and</li> </ul>	Refer to the EMS
	(h) a protocol for periodic review of the plan.	Section 6

#### 2.4 Environment protection licence

The EPL conditions, relevant to this WMP, are listed in Table 2-2. A cross reference is also included to indicate where the condition is addressed in this WMP or other environmental management documents.

#### Table 2-2 EPL conditions relevant to the WMP

Condition No.	Requirement	Reference
L3.1	The licensee must not cause, permit or allow any waste generated outside the premises to be received at the premises for storage, treatment, processing, reprocessing or disposal or any waste generated at the premises to be disposed of at the premises, except as expressly permitted by the licence.	Section 4
# 3.1 Existing environment

The existing quarry would generate various waste streams including construction and excavation waste, vegetation waste, packaging materials and liquid wastes as well as officebased (domestic) waste. The volumes of solid wastes are expected to be relatively small as most waste would be reused or recycled on site e.g. excavation waste and vegetation waste. General construction waste is likely to be the most significant and this is currently managed, where possible, in accordance with the waste management hierarchy of avoid, reuse, recycle and dispose.

Liquid wastes consist of oil, paint, lubricants, glue, toilets and stormwater. The oil, paint, lubricants and glue are minor sources of waste. Significant volumes of stormwater would be generated from the site and is addressed in the Soil and Water Management Plan.

# 3.2 Impacts

Quarrying involves the stripping and emplacement of topsoil and overburden, extraction, screening and stockpiling of the raw materials and product loading and distribution. The operation of the quarry would generate the following waste types:

- Excavated material (topsoil and overburden not suitable for sale)
- Domestic waste
- Green waste
- Construction waste
- Effluent from toilet facilities
- Used lubricants and oils
- Contaminated soil
- Stormwater runoff from disturbed areas and the processing plant

Potential impacts from the production and inappropriate disposal of waste generated from the proposal includes:

- Contamination of land
- Pollution of waterways
- Air pollution
- Overuse of scarce resources
- Human and animal health impacts

# 4. Environmental control measures

Environmental requirements and control measures are identified in the Conditions of Approval, EIS, MOD2 and EPL. As specified by the EPL, Holcim must not cause, permit or allow any waste generated outside the site to be received for storage, treatment, processing, reprocessing or disposal or any waste generated at the premises to be disposed of at the site. Other specific measures and requirements to address waste impacts are outlined in Table 4-1.

Waste Type	Waste Management Hierarchy				
	Avoid	Reuse/ Recycle/Recover	Dispose		
Excavated material	Avoid excess excavation	Use excess material on site as fill and/or in rehabilitation works	Minimise excess excavated material to be disposed offsite		
Green waste	Avoid clearing vegetation where not required for each stage of operation	Mulch cleared vegetation and use on site e.g. in soil stabilisation, planted areas	Minimise green waste to be disposed offsite		
General construction waste	Materials to be sourced and ordered in appropriate quantities / materials to be pre- fabricated where possible	<ul> <li>Reuse excess material on-site e.g. in maintenance works or other projects. If not reused, recycle as follows:</li> <li>Metals – to be removed by contractor for recycling</li> <li>Paper &amp; cardboard, glass, recyclable plastic – to be removed by contractor for recycling</li> <li>Batteries – stored on battery pallet in maintenance shed for removal by contractor for recycling</li> </ul>	Non recyclable material is to be removed by contractor, for disposal in accordance with the Waste Classification Guidelines		
Contaminated soil	Proper storage of all chemicals and fuels (e.g. bunded areas with 110% capacity)	Utilise bioremediation for large quantities of fuel- impacted soil. Tracking during transportation to be carried out where required under legislation	Place small contaminated materials in the Contaminated Waste Bin to be removed by contractor. Dispose contaminated waste in accordance with the Waste Classification Guidelines		
Liquid waste	Materials to be sourced and ordered in appropriate quantities	Reuse excess material on-site wherever possible (e.g. in other machinery or future maintenance work). If not reused, place in oil tank which is removed by contractor for recycling	Excess liquid waste that cannot be reused on site or recycled by a contractor will be disposed of offsite, in accordance with the Waste Classification Guidelines, by a contractor		

### **Table 4-1 Waste management measures**

Waste Type		Waste Management Hierarchy					
	Avoid	Reuse/ Recycle/Recover	Dispose				
Wastewater	Divert clean water from the site through stormwater management e.g. diversion drains	Wastewater to be pumped to a holding pond and used on-site e.g. for dust suppression/ plant watering etc.	Discharge wastewater, in accordance with EPL requirements				
Biological (sewage) waste	Minimise use of site facilities e.g. toilets	Consider using composting toilet	Sewage waste to be disposed via the onsite treatment system, see approval in Appendix A				
Domestic waste	Materials to be sourced and ordered in appropriate quantities	Reuse excess material on-site wherever possible. If not reused, place in industrial recycling bulk bin to be removed by contractor for recycling	Non recyclable material is to be removed by contractor				

# 5. Monitoring and reporting

## 5.1 Environmental inspections and monitoring

Routine weekly inspections by the Quarry Manager (or delegate) will occur throughout the operational lifetime of the quarry, using the *Environmental Inspection Checklist* in the Environmental Monitoring and Management Plan, to ensure and record the site is left clean and tidy.

# 5.2 Contingency plan

If the above monitoring detects an impact or there is a justified, waste related, community complaint, a contingency plan or trigger and response plan is to be implemented, as shown below.

In general, all issues will be investigated and corrective actions determined within 24 hours. The timeframe to implement the corrective actions will depend on the risk and consequence of the issue. The nature of the corrective action will also influence the implementation timeframe.

### Table 5-1 Contingency plan

Trigger	Response
Waste and/or hazardous materials issue identified	<ul> <li>Undertake management and mitigation measures in accordance with Table 4-1</li> <li>Review management measures and update, if necessary</li> </ul>
Complaint received by a member of the community regarding waste and hazardous materials management	<ul> <li>Quarry Manager to confirm the cause of the management issue and agree on required corrective actions.</li> <li>Quarry Manager to implement corrective actions to reduce/ improve management of waste.</li> <li>Quarry Manager to initiate complaint response process as described in the EMS.</li> </ul>

## 5.3 Reporting

The general reporting requirements are described in the EMS. Routine waste monitoring will be recorded on the *Environmental Inspection Checklist* in the Environmental Monitoring and Management Plan.

Waste removed from site will be recorded in a waste register along with the relevant dockets from the contractors removing the material.

A summary of the waste monitoring and management will be presented in the Annual Report (refer to the EMS).

# 6. Review and improvement

Continuous improvement of this WMP will be achieved by reviewing the plan in accordance with the EMS and the ongoing evaluation of environmental management performance against environmental policies, objectives and targets.

The continuous improvement process is designed to:

- Identify areas of opportunity for improvement of environmental management and performance.
- Determine the cause or causes of non-conformances and deficiencies.
- Develop and implement a plan of corrective and preventative action to address any nonconformances and deficiencies.
- Verify the effectiveness of the corrective and preventative actions.
- Document any changes in procedures resulting from process improvement; and make comparisons with objectives and targets.

# Appendices

 $\ensuremath{\textbf{GHD}}\xspace$  | Report for Holcim ( Australia ) Pty Ltd - Dunloe Sand Quarry , 2220056 | 9

**Appendix A** – On-site sewage management system approval

Council Reference: OSSM02484 DA08/1247 LN: 10173 MB Your Reference:



12 July 2017

Ramtech Pty Ltd 30-32 Lundberg Drive SOUTH MURWILLUMBAH NSW 2484 Customer Service | 1300 292 872 | (02) 6670 2400

tsc@tweed.nsw.gov.au www.tweed.nsw.gov.au

Fax (02) 6670 2429 PO Box 816 Murwillumbah NSW 2484

Please address all communications to the General Manager

ABN: 90 178 732 496

Dear Sir/Madam

## Approval Notice - Approval to Operate On-Site Sewage Management System OSSM02484 at Lot 162 DP 755721; Warwick Park Road WOOYUNG

Further to Council's inspection carried out 11 July 2017 please find enclosed your Certificate of Approval to Operate.

The aerated wastewater treatment system and pressurised effluent irrigation system shall be serviced every three months by a qualified and approved service technician. A record of each service shall be provided by the service agent to Council after each service. Council maintains a database of service records for each system. Servicing the system is a legal requirement and penalties apply for non-compliance.

The electricity supply to the AWTS shall not be switched off and the system shall remain active at all times.

The pressurised sub subsurface effluent irrigation system shall be checked and flushed at each service and shall not be disconnected or altered without the approval of Council.

Yours faithfully

Ian Grimshaw ENVIRONMENTAL COMPLIANCE OFFICER



# CERTIFICATE OF APPROVAL TO OPERATE a System of Sewage Management Local Government Act 1993

Ramtech Pty Ltd 30-32 Lundberg Drive SOUTH MURWILLUMBAH NSW 2484

Being the applicant in respect of Sewage Management Application No: OSSM02484 for:

- Econocycle AWTS with 200m2 subsurface pressurised effluent irrigation area for Office on Lot 162//755721
- Risk assessment: Low

Pursuant to Chapter 7 of the Local Government Act 1993, notice is hereby given of the determination by the Council, as Approval Authority, of the Sewage Management System Application relating to the land described as:

## • Lot 162 DP 755721; Warwick Park Road WOOYUNG

The Application to Operate a System of Sewage Management has been determined by granting of Approval subject to compliance with conditions.

# This approval operates from 12 July 2017 and expires on

## **12 July 2023** or upon transfer of this property to another party.

**NOTE:** This approval relates to the operation of a system of sewage management on land and the granting of this approval in no way authorises any ancillary activity on the land that may require approval under the Local Government Act 1993 and /or other legislation.

# IMPORTANT NOTICE

THIS IS A LEGAL DOCUMENT. Please ensure that you read the document carefully, and particularly note all attendant instructions and conditions.



# **CONDITIONS**

1. The owner/occupier shall maintain the system in accordance with the approved plans, specifications and conditions of approval.

[OSO0225]

2. The sewage management system shall be maintained in accordance with the attached management procedures.

**NOTE:** It is the owner's responsibility to ensure the sewage management system is operated and maintained at all times. The management procedures are relevant to the sewage management system on site and are required to be carried out to prevent detrimental environmental and public health risks.

If foul odours or ponding of effluent on ground surface occurs this indicates a malfunctioning system and requires immediate attention. Contact your local plumber and or Tweed Shire Council for advice.

[OSO0226]

3. This approval shall expire on the expiration date specified above, unless the approval is renewed or extended by Council.

NOTE: The Council may extend the period of an approval following an inspection or at any other time. The operator may apply for renewal at any time.

[OSO0227]

4. The person operating the system of sewage management shall provide details of operation and maintenance, and evidence of compliance with the conditions of this approval, to the Council whenever reasonably required to do so.

[OSO0228]

5. The Council may carry out an audit inspection of the sewage management system to determine compliance with conditions of approval and may charge the approved fee specified for the service in the Council's fees and charges. An approved fee may also be charged for follow up compliance inspections of rectification work.

[OSO0229]

6. Effluent arising from an on-site sewage management system shall not be permitted to discharge into any natural waterway or storm water drain.

[OSO0235]

7. The on-site sewage management system shall be operated and maintained in a sanitary condition and in accordance with the relevant requirements of the Local Government (General) Regulation 2005, and other relevant operating specifications.

[OSO0245]

- 8. Except in circumstances beyond the control of the operator, the system of sewage management shall be operated in a manner that achieves the following performance standards:
  - (a) Prevention of the spread of disease by micro-organisms,
  - (b) Prevention of the spread of foul odours,
  - (c) Prevention of contamination of water,
  - (d) Prevention of degradation of soil and vegetation,



- (e) Discouragement of insects and vermin,
- (f) Ensuring that persons do not come into contact with untreated sewage or effluent (whether treated or not) in their ordinary activities on the premises concerned,
- (g) Minimisation of any adverse impacts on the amenity of the premises and surrounding lands.

**NOTE:** Disease may be spread if there is any human contact with sewage wastewater. Water pollution may occur as a result of surface run off from failing land application areas, from percolation of effluent into ground water and by drainage to waterways. Components should be well maintained and monitored to eliminate disease risks and water pollution.

[OSO0255]

- 9. Maintenance and operating conditions of the system of sewage management specified in any certificate of accreditation issued by the Director-General of NSW Health, in relation to the Aerated Wastewater Treatment System (AWTS) shall be complied with, including:
  - (a) The owner of the AWTS shall enter into an annual contract with the manufacturer or another suitably qualified or experienced person requiring the system to be **serviced every quarter** in accordance with the Department's requirements,
  - (b) The effluent shall at all times comply with the following standard;
    - (i) Biological Oxygen Demand (BOD<sup>5</sup>) less than 20 mg/L,
    - (ii) Suspended Solids less than 30 mg/L,
    - (iii) Free Residual Chlorine 0.5mg/L (min) to 2.0 mg/L (max)
    - (iv) Thermotolerant Coliforms less than 30 cfu/100ml
  - (c) A Telephone number for emergency service shall be fixed and located in or near the alarm control panel so as to be easily visible.

[OSO0265]

10. The effluent land application area shall not be used for the production of low growing crops that are eaten without cooking for human consumption.

[OSO0625]

11. Soaker hoses and standard household sprinklers and attachments shall not be used for the irrigation of effluent from any on-site sewage management system.

[OSO0645]

12. The irrigation system shall be operated in such a way as to prevent any run-off of effluent from the land application area.

[OSO0655]

13. All effluent land application areas are to be kept clear of weeds and the plants trimmed so as to prevent the area from becoming overgrown.

[OSO0665]

14. Vehicle traffic and livestock shall be excluded from the effluent land application area and this may involve fencing of the area.

[OSO0675]

15. The effluent land application area shall not be used for active recreational purposes.



16. Any alteration to existing sewage management facilities shall be approved by Council.

[OSO0686]

# Failure to comply with these conditions may result in:

- Issuing Orders under Section 124 of the Local Government Act 1993; or
- Prosecution under Section 627 of the Local Government Act for failing to comply with conditions for Approval with a maximum penalty of \$2200; or
- Issuing of a Penalty Notice (on the spot fine) of \$330.

**NOTE 1:** If an owner or occupier of land is the holder of an <u>approval to operate</u> a system of sewage management on the land (being an approval that is in force), any other owner of occupier of that land may operate the system of sewage management (without obtaining a further approval) in accordance with the conditions of the approval.

**NOTE 2:** A person who purchases (or otherwise acquires) land on which any sewage management facilities are installed or constructed may operate a system of sewage management without the approval required under Section 68 of the Local Government Act for the period of 3 months after the date on which the land is transferred or otherwise conveyed to the person (whether or not an approval to operate a system of sewage management on that land is in force at that date). If an application for <u>approval to operate a system of sewage management</u> is submitted within 2 months of the transfer of the land to the new owner, any existing sewage management facilities may continue to be operated until the application is determined by Council.

# **REVIEW OF DETERMINATION**

Under the provisions of Section 100 of the Local Government Act, 1993, an applicant may request the Council to review the determination of an application for approval. The request for a review must be made within twenty-eight (28) days after the date of the determination.

## **DISSATISFACTION WITH DETERMINATION-RIGHT OF APPEAL**

Under the provisions of section 176 of the Local Government Act, 1993, an applicant who is dissatisfied with the determination of Council with respect to the application for an approval may appeal to the Land and Environment Court. The appeal must be made within 12 months after the date from which the approval operates.

## FOR FURTHER INFORMATION

If you require further information in relation to this approval please contact Council's Building and Environmental Health Unit on the above number between the hours of 8.30am and 10.00am, Monday to Friday.

## Signed on behalf of the Tweed Shire Council

lan Grimshaw, Environmental Compliance Officer

GHD

230 Harbour Drive Coffs Harbour NSW 2450 T: 61 2 6650 5600 F: 61 2 6650 5601 E: cfsmail@ghd.com

### © GHD 2020

This document is and shall remain the property of GHD. The document may only be used for the purpose for which it was commissioned and in accordance with the Terms of Engagement for the commission. Unauthorised use of this document in any form whatsoever is prohibited.

### 22-SO-976847825-

7/https://projects.ghd.com/oc/Newcastle3/holcimdunloesandquar/Delivery/Documents/2220056\_RP T\_Dunloe Waste Management Plan.docx

Document	Status
----------	--------

Revision	Author	Reviewer		Approved for	Issue	
		Name	Signature	Name	Signature	Date
0	B Luffman	S Lawer		S Lawer		02/08/2019
1	B Luffman	S Lawer	0	S Lawer	0	23/09/2019
2	B Luffman	S Lawer	Ja-	S Lawer	tan	17/07/2020
		(	$\bigcirc$	(		

# www.ghd.com



# **Appendix K** – Pollution incident response management plan



Strength, Performance, Passion

# Pollution Incident Response Management Plan Dunloe

1

# **Dunloe Sands Pollution Incident Response Management Plan (PIRMP)**

Revision/Checking History

Revision Number	Date	Checked	Issued By
1	2016	Daniel Lidbetter – NSW/ACT Planning & Environment Coordinator Garth Stacey – Quarry Manager Daniel Dwyer – Production Suoervisor	Daniel Lidbetter
2	26/10/17	Amy Nelson - NSW/ACT Planning & Environment Coordinator Garth Stacey – Quarry Manager Daniel Dwyer – Production Supervisor	Amy Nelson
3	10/08/18	Alana White – Senior Environment and Community Liaison Garth Stacey – Quarry Manager Daniel Dwyer Production Supervisor	Alana White

# **Table of Contents**

Pollu	tion Incident Response Management plan Dunloe Sands	1
1.	Purpose	4
2.	Scope	7
3.	Definitions	7
4.	Associated Documentation	7
5.	Responsibility	7
6.	Record Retention	8
7.	Procedure	9
7.1	Environmental Impact and Hazard Register	10
7.2	Pollutant and Safety Equipment Information	17
7.3	Emergency Response Map	26
7.4	Pollution Incident Response Management Plan	27
7.5	Communication Strategy	32
7.6	Staff Training	33
7.7	Continual Improvement	33
7.8	Review	33

# **Appendices**

- A Emergency Contact Details
- B Pollution Incident Response Test Checklist
- C Community Notification Strategy

# **Glossary of Acronyms**

- PIDS- Pollution Information Data Sheet
- PPE- Personnel Protective Equipment
- MSDS- Material Safety Data Sheets
- PEOA- Protection of the Environment Operations Act 1997

### 1. Purpose

The purpose of this document is to detail the pollution incident response management plan for the Dunloe Sands, to comply with Section 5.7A of the Protection of the Environment Operations (POEO) Act:

### Protection of the Environment Operations Act 1997 No 156

# Part 5.7A Duty to prepare and implement pollution incident response management plans

# 153A Duty of licence holder to prepare pollution incident response management plan

The holder of an environment protection licence must prepare a pollution incident response management plan that complies with this Part in relation to the activity to which the licence relates.

Maximum penalty:

- (a) in the case of a corporation—\$1,000,000 and, in the case of a continuing offence, a further penalty of \$120,000 for each day the offence continues, or
- (b) in the case of an individual—\$250,000 and, in the case of a continuing offence, a further penalty of \$60,000 for each day the offence continues.

# 153B EPA may direct other persons to prepare pollution incident response management plan

- (1) The EPA may, in accordance with the regulations, require the occupier of premises at which industry is carried out to prepare a pollution incident response management plan that complies with this Part in relation to activities at the premises.
- (2) A person must not fail to comply with such a requirement.

Maximum penalty:

- (a) in the case of a corporation—\$1,000,000 and, in the case of a continuing offence, a further penalty of \$120,000 for each day the offence continues, or
- (b) in the case of an individual—\$250,000 and, in the case of a continuing offence, a further penalty of \$60,000 for each day the offence continues.
- (3) The regulations may make provision for or with respect to:
  - (a) the class or classes of premises, or industries carried out at premises, that may be the subject of a requirement to prepare a pollution incident response management plan, and
  - (b) the circumstances in which some or all premises within those classes may be the subject of a requirement to prepare a pollution incident response management plan.

### 153C Information to be included in plan

A pollution incident response management plan must be in the form required by the regulations and must include the following:

(a) the procedures to be followed by the holder of the relevant environment protection licence, or the occupier of the relevant premises, in notifying a pollution incident to:

- (i) the owners or occupiers of premises in the vicinity of the premises to which the environment protection licence or the direction under section 153B relates, and
- (ii) the local authority for the area in which the premises to which the environment protection licence or the direction under section 153B relates are located and any area affected, or potentially affected, by the pollution, and

(iii) any persons or authorities required to be notified by Part 5.7,

- (b) a detailed description of the action to be taken, immediately after a pollution incident, by the holder of the relevant environment protection licence, or the occupier of the relevant premises, to reduce or control any pollution
- (c) the procedures to be followed for co-ordinating, with the authorities or persons that have been notified, any action taken in combating the pollution caused by the incident and, in particular, the persons through whom all communications are to be made,
- (d) any other matter required by the regulations.

### 153D Keeping of plan

A person who is required to prepare a pollution incident response management plan under this Part must ensure that it is kept at the premises to which the relevant environment protection licence relates, or where the relevant activity takes place, and is made available in accordance with the regulations.

#### Maximum penalty:

- (a) in the case of a corporation—\$1,000,000 and, in the case of a continuing offence, a further penalty of \$120,000 for each day the offence continues, or
- (b) in the case of an individual—\$250,000 and, in the case of a continuing offence, a further penalty of \$60,000 for each day the offence continues.

#### 153E Testing of plan

A person who is required to prepare a pollution incident response management plan under this Part must ensure that it is tested in accordance with the regulations.

#### Maximum penalty:

- (a) in the case of a corporation—\$1,000,000 and, in the case of a continuing offence, a further penalty of \$120,000 for each day the offence continues, or
- (b) in the case of an individual—\$250,000 and, in the case of a continuing offence, a further penalty of \$60,000 for each day the offence continues.

### 153F Implementation of plan

If a pollution incident occurs in the course of an activity so that material harm to the environment (within the meaning of section 147) is caused or threatened, the person carrying on the activity must immediately implement any pollution incident response management plan in relation to the activity required by this Part.

#### Maximum penalty:

- (a) in the case of a corporation—\$2,000,000 and, in the case of a continuing offence, a further penalty of \$240,000 for each day the offence continues, or
- (b) in the case of an individual—\$500,000 and, in the case of a continuing offence, a further penalty of \$120,000 for each day the offence continues.

## 1.Scope

The scope of this management plan includes:

Pollution Incident Response Management Plan (PIRMP) for environmental pollution generated at the Dunloe;

# 2.Definitions

Pollution Incident -	An incident or set of circumstances during or as a consequence of which there is or is likely to be a leak, spill or other escape or deposit of a substance, as a result of which pollution has occurred, is occurring or is likely to occur. It includes an incident or set of circumstances in which a substance has been placed or disposed of on premises, but it does not include an incident or set of circumstances involving only the emission of any noise.
Material Harm -	<ul><li>(i) it involves actual or potential harm to the health or safety of human beings or to ecosystems that is not trivial, or</li></ul>
	(ii) it results in actual or potential loss or property damage of an amount, or amounts in aggregate, exceeding \$10,000 (or such other amount as is prescribed by the regulations), It does not matter that harm to the environment is caused only in the premises where the pollution incident occurs, and
Loss -	the reasonable costs and expenses that would be incurred in taking all reasonable and practicable measures to prevent, mitigate or make good harm to the environment.

### 3.Associated Documentation

- Protection of the Environment Operations Act 1997
- Protection of the Environment (General) Amendment (Pollution Incident Response Management Plans) Regulation 2012
- Appendix A: Emergency Contact Details
- Appendix B: Pollution Incident Response Test Checklist
- Appendix C: Community Notification Strategy

### 4.Responsibility

The following personnel are responsible for the PIRMP;

- 1) Activating the plans and managing the response: Garth Stacey Quarry Manager
- 2) Notifying and coordinating relevant authorities: Garth Stacey Quarry Manager
- 3) Implementation and management of this document: Garth Stacey Quarry Manager
- 4) Annual review and test of PIRMP Hema Vignaraja SHE Reporting Analyst

# 5.Record Retention

A copy of all Quarry pollution incident response records will be retained on site in accordance with SHE guideline 1.4 – Administrative and Legal Requirements. A copy will also be saved electronically on google drive in the 'Site PIRMPs 2017 Final' folder that will be shared with the site.

Records must be made available to EPA officers and any person responsible for the PIRMP.

## 6.Procedure

The following section outlines the management procedures for pollution incident response management. The protocol is split into three sections:

- 1) Eliminating and/or avoiding Generation of Pollution
- 2) Management of pollution incidents
- 3) Improving and reviewing Pollution Incident Response Management Plan

# 7.1 Environmental Impact and Hazard Register

In order to effectively plan for a potential pollution event, a register of environmental hazards has been created. Each hazard has been assessed in accordance with the Holcim SHE Risk Assessment tool (see Table 1 below).

The hazards have been grouped according to the area of environmental impact. By identifying these hazards ahead of time, mitigation measures can be identified and implemented through site procedures to minimise the risk of a pollution event occurring (table 2 below).

Table 1: Holcim SHE Risk Score Matrix

### Step 1 - Consider the Consequence

What are the consequences of the most reasonable worst case scenario considering a credible failure of existing controls?

Consequence	Disaster	Severe	Serious	Significant	Minor		
Environment On Site & Off Site	Major event, unconfined impact, severe permanent damage with low likelihood of recovery.	Significant permanent damage; reversible damage with recovery time of years; high potential for prosecution	Minor permanent damage; temporary damage that is widespread or that has moderate impact	Damage that is near source confined, temporary and minor	No measurable damage to environment		
Compliance With Legal and Other Requirements	Blatant or serious breech of legal requirement, leading to operation being suspended or severely reduced. Prosecution expected.	Breach of external requirement (license, legislation, regulation, contract etc) with high potential for prosecution and/or high impact.	Non-compliance with external requirement with moderate potential for impact.	Repeated non-compliance with internal procedure, non- compliance with external requirement with low potential impact	Minor non-compliance with internal procedures.		
Community Perception and Reputation	Significant adverse media attention (state or national level), loss of reputation or work nationally or across product groups.	Prosecution, significant impacts on social license to operate, loss of reputation or ability to secure work across product groups.	Local adverse media attention, loss of reputation or ability to secure work in local area, complaints that result in changes to external requirements.	Multiple community complaints or complaints that require changes to internal operating procedures.	Community complaint resolved with no changes to existing operating procedures.		

Note: Temporary environmental damage has a duration of up to approximately one week to rectify

	Step 2 - Consider the Likelihood								
	What is the likelihood that the proposed consequence will occur with a credible failure of existing controls?								
Likelihood	Certain	Likely	Possible	Unlikely	Rare				
Description	Event that is expected to occur on multiple occasions	Event that is likely to occur at least once	Event that may occur	Event that is unlikely to occur	Event that may occur only in exceptional circumstances				
Frequency	Event is likely to occur more than twice a year.	Event is likely to occur once or twice a year.	Event is likely to occur more than once or twice in a 10 year period	Event is likely to occur once or twice in a 10 year period	Event is likely to occur once or twice in a 100 year period				

Step 3 - Determine Risk Rating from the Risk Matrix								
Consequences								
Liklihood	Disaster	Severe	Serious	Significant	Minor			
Certain	High	High	High	Medium	Medium			
Likely	High	High	Medium	Medium	Low			
Possible	High	Medium	Medium	Low	Low			
Unlikely	Medium	Medium	Low	Low	Low			
Rare	Medium	Low	Low	Low	Low			

Table 2: Holcim Quarry Environmental Impact and Hazard Register

	Key Environmental Hazards		Risk				vised k	l
Ke	y Environmental Hazards	L	с	R	miligation measures		с	R
Α	ir Quality							
1	Excessive dust emissions	Possible	Serious	Medium	<ul> <li>Complete monitoring &amp; assess results quarterly</li> <li>Review results &amp; monitoring program quarterly</li> <li>Water carts/spraying</li> <li>Minimise disturbed areas</li> <li>Stop dust generating activities as necessary</li> <li>Progressively rehabilitate disturbed areas</li> <li>Restrict works during periods of high wind</li> <li>Dust minimisation training</li> <li>Maintenance of dust control equipment</li> </ul>	Unlikely	Significant	Low
2	Health issues off site	Rare	Severe	Low	<ul> <li>As per (1) Excessive Dust Emissions</li> <li>Complaints hot line</li> <li>Issue monitoring results</li> <li>Communicate construction activities to neighbours plus potential for dust</li> </ul>	Rare	Serious	Low
3	Equipment exhaust emissions exceed limits	Unlikely	Significant	Low	<ul> <li>Inspect equipment engine emissions regularly</li> <li>All equipment is serviced and maintained to OEM requirements</li> <li>Excessive equipment emissions to trigger out of service procedures</li> </ul>	Rare	Significant	Low
G	roundwater							

Kay Environmental Hazarda		Risk			Mitigation Measures		Revised Risk	
re	y Environmental Hazards	L C R			mitigation measures	L	с	R
1	Groundwater contamination	Unlikely	Serious	Low	<ul> <li>Implement Monitoring and response plan</li> <li>Review monitoring results quarterly &amp; action as necessary</li> <li>Ensure storage, handling and transport of dangerous goods are conducted in accordance with Australian Standards</li> <li>Identify, classify, quantify &amp; appropriately store hazardous waste</li> <li>Develop &amp; implement oil &amp; fuel spillage controls</li> <li>Ensure hazardous waste is minimised</li> <li>Licenced contractors to remove hazardous waste from site</li> <li>Keep records of all hazardous waste movements</li> <li>Develop &amp; implement oil &amp; fuel spillage controls</li> <li>Implement bunding to appropriate areas</li> <li>Ensure adequate spill kits are available on site including adequate training</li> <li>Minimise hazardous waste storage quantities on site</li> </ul>	Rare	Serious	Low
2	Lowering of groundwater table	Rare	Serious	Low	<ul> <li>Monitor &amp; report on ground water levels</li> <li>Comply with Water Management Plan water balance</li> </ul>	Rare	Significant	Low
3	Acid-sulphate soils	Likely	Serious	Medium	<ul> <li>Acid sulphate status is known</li> <li>Implement acid-sulphate management plan</li> <li>Regular review of acid-sulphate management plan outcomes</li> </ul>		Serious	Low
<b>S</b> 1	urface Water Discharge of sediment	Possible	Serious	Medium	<ul> <li>Develop &amp; implement Water Management Plan</li> <li>Implement Monitoring Program</li> <li>Review monitoring results quarterly &amp; action as necessary</li> <li>Develop &amp; implement Surface &amp; Groundwater Response Plan</li> <li>Develop &amp; implement Erosion &amp; Sediment Control Plan</li> <li>Implement dust control procedures as per AIR</li> </ul>	Unlikely	Serious	Low

2 Discharge of hazardous materia	L	C		Mitigation Measures			
2 Discharge of hazardous materia		L C R		Mitigation measures	L	С	R
	ls Rare	Severe	Low	<ul> <li>As per i</li> <li>Ensure storage, handling and transport of dangerous goods are conducted in accordance with relevant Australian Standard</li> <li>Review monitoring results quarterly &amp; action as necessary</li> <li>Identify classify, quantify &amp; appropriately store hazardous waste</li> <li>Develop &amp; implement oil &amp; fuel spillage controls</li> <li>Implement bunding to appropriate areas</li> <li>Ensure adequate spill kits are available on site including adequate training for effective use</li> <li>Minimise hazardous waste storage quantities on site</li> <li>Appropriate location of hazardous materials storage areas to prevent off-site discharges</li> </ul>	Rare	Serious	Low
Ecology							
1 Damage to local flora	Possible	Serious	Medium	<ul> <li>Develop &amp; implement Biodiversity Action Plan</li> <li>Put in adequate physical protection measures including signage</li> <li>Monitor &amp; report on site flora health regularly</li> <li>Suitable training re flora protection</li> <li>Removal of stock from sensitive areas</li> <li>Implement bushfire hazard reduction tasks</li> <li>Removal of feral animals from sensitive areas</li> <li>Noxious weed control in sensitive areas</li> </ul>	Unlikely	Significant	Low
2 Damage to site fauna	Unlikely	Serious	Rare	<ul> <li>As per (1)</li> <li>Information re local WIRES for distressed or injured fauna</li> </ul>	Rare	Serious	Low
3 Dust pollution onto site sensitive ecological areas	Unlikely	Severe	Medium	<ul> <li>As per (1)</li> <li>Comply with site Management Plans</li> <li>Regular review of riparian areas (as per Management Plans)</li> </ul>	Unlikely	Significant	Low

		Risk					vised k	
Ke	Rey Environmental Hazards			R	Mitigation Measures	L	с	R
1	Spill of liquid fuel whilst in storage	Possible	Severe	Medium	<ul> <li>Fuels stored according to Holcim's bunding requirements.</li> <li>Measures in place to ensure spills do not leave site boundaries ie diverting flow away from boundaries, stormwater drains.</li> <li>Bunding subject to regular inspection and maintenance</li> </ul>	Significant	Unlikely	Low
2	Spill during delivery of fuel to mobile equipment	Possible	Severe	Medium	<ul> <li>Breakaway couplings installed on mobile fuel delivery vehicles.</li> <li>Drivers stay with vehicle during refuelling</li> <li>Emergency spill kits located on fuel delivery vehicles.</li> <li>Spill response equipment is regularly inspected and maintained</li> <li>Mobile refuelling takes place in the pit</li> <li>Drivers trained in spill response procedures.</li> <li>Refuelling takes place in designated refuelling areas.</li> </ul>	Unlikely	Significant t	Low
3	Spill during delivery of fuel to storage tank	Possible	Severe	Medium	<ul> <li>Supplier's fuel transfer procedure is known</li> <li>Fuel transfer is supervised against suppliers procedure</li> </ul>		Significant	Low
4	Land contamination	Likely	Significant	Medium	<ul> <li>Holcim land contamination strategy is known and applied</li> </ul>	Unlikely	Significant	Low

# 7.2 Pollutant and Safety Equipment Information

Legislative requirements under the Protection of the Environment Operations (POEO) Act dictate that the site is to provide information for all pollutants that are used and stored on the site. This information is required as it assists personnel responsible for coordinating spill responses to more effectively manage spills.

This information must be presented as a manifest detailing the pollutants stored at the site, the location of these storage areas, and the safety equipment to be made available at these areas. A Pollution Information Data Sheet (PIDS) has been prepared that includes the following information for each pollutant. Refer to table 3 below

- The intended use for the pollutant
- How the pollutant is stored
- MSDS information
- Safety equipment or other devices that are used to minimise the risks to human health or the environment and to contain or control a pollution incident
- PPE needed to safely manage a spill of the pollutant
- Procedure for cleaning up a spill of the pollutant.

In order to ensure the currency and reliability of the information in the PIDS, the information should be reviewed and updated on a monthly basis.

 Table 3: Pollutant Information Data Sheet

Pollutant	Storage Location	Current MSDS held Yes/No	Emission control equipment <sub>(1)</sub>	PPE (1)	Spill Clean Up Method (1)	Quantity Stored on Site
Fuel	Designated Workshop Area	Yes- held in office	Sand, earth, vermiculite	PVC gloves, safety glasses, goggles	<ul> <li>Large Spill</li> <li>1) In the case of large spills contact relevant personnel</li> <li>2) Stop leak without risk.</li> <li>3) Move containers from spill area.</li> <li>4) Approach the release from upwind</li> <li>5) Prevent entry into sewer, water courses, basements or confined areas.</li> <li>6) Wash spillages into an effluent treatment plant or proceed as follows.</li> <li>7) Contain and collect spillage with non-combustible, absorbent material e.g. sand, earth, vermiculite or diatomaceous earth and place into a container according to local legislation.</li> <li>8) Use spark-proof tools and explosive proof equipment. Dispose of via a licensed waste disposal contractor</li> <li>9) Contaminated absorbent material may pose the same hazard as the spilt product</li> <li>10) In the case of spillage on water, prevent the spread of product by the use of suitable barrier equipment.</li> <li>11) Recover product from the</li> </ul>	4000lts Total

Pollutant	Storage	Current	Emission	PPE (1)		Quantity Stored on Site
	Location	MSDS held Yes/No	control equipment <sub>(1)</sub>		Spill Clean Up Method (1)	
Vehicle fluids & Lubricants	Designated workshop area	Yes – held in office	Sand, earth, vermiculite	PVC Gloves, safety glasses	Surface         Small Spill         1)       Stop leak without risk.         2)       Move containers from spill area         3)       Absorb with an inert material and place in appropriate waste disposal container.         4)       Use spark-proof tools and explosion-proof equipment.         5)       Dispose of via a licensed waste disposal contractor.         Large Spill       1)         1)       In the case of large spills contact relevant personnel         2)       Stop leak without risk.         3)       Move containers from spill area.         4)       Approach the release from upwind         5)       Prevent entry into sewer, water courses, basements or confined areas.         6)       Wash spillages into an effluent treatment plant or proceed as follows.         7)       Contain and collect spillage with non-combustible, absorbent material e.g. sand, earth, vermiculite or diatomaceous earth and place into a container according to local legislation.	200lts oil Total 1 -200lt oil drum Contractors bring in most lubricants when servicing very little stored on site

Pollutant	Storage Location	Current MSDS held Yes/No	Emission control equipment <sub>(1)</sub>	PPE <sub>(1)</sub>	Spill Clean Up Method (1)	Quantity Stored on Site
Truck wash	Adjacent workship containers	Yes – held in office	Sand, earth, vermiculite	PVC Gloves, safety glasses	<ul> <li>8) Contaminated absorbent material may pose the same hazard as the spilt product</li> <li><u>Small Spill</u> <ol> <li>Stop leak without risk.</li> <li>Move containers from spill area</li> <li>Dilute with water and mop up, or absorb with an inert dry material and place in appropriate waste disposal container</li> <li>Dispose of via a licensed waste disposal contractor.</li> </ol> </li> <li><u>Large Spill</u> <ol> <li>In the case of large spills contact relevant personnel</li> <li>Stop leak without risk.</li> <li>Move containers from spill area.</li> <li>Approach the release from upwind</li> </ol> </li> <li>Prevent entry into sewer, water courses, basements or confined areas.</li> <li>Wash spillages into an effluent treatment plant or proceed as follows.</li> <li>Contain and collect spillage with non-combustible, absorbent material e.g. sand, earth, vermiculite or diatomaceous earth and place into a container according to local legislation.</li> </ul>	3000lts est Separate pit Reviewing and possible move to a new location

Pollutant	Storage Location	Current MSDS held Yes/No	Emission control equipment <sub>(1)</sub>	PPE (1)	Spill Clean Up Method (1)	Quantity Stored on Site
					<ul> <li>material may pose the same hazard as the spilt product</li> <li><u>Small Spill</u></li> <li>1) Stop leak without risk.</li> <li>2) Move containers from spill area</li> <li>3) Dilute with water and mop up, or absorb with an inert dry material and place in appropriate waste disposal container</li> <li>4) Dispose of via a licensed waste disposal contractor.</li> </ul>	
Liquid Nitrogen	Workshop designated cage area	Yes-held in office	Nil	Goggles, Safety boots and insulated or leather gloves, air-line respirator (if inhalation risk exists)	<ol> <li>Spillage         <ol> <li>Release of liquid to atmosphere will generate vapour fog clouds which can travel considerable distances and affect visibility.</li> <li>These clouds should be treated as asphyxiating atmospheres as the evaporated liquid will have displaced air</li> <li>Refer to vessel operating instructions</li> <li>In an emergency allow liquid and gas to escape to atmosphere</li> <li>Monitor oxygen concentration in confined spaces</li> <li>Contact relevant authorities for guidance</li> <li>Leak checking may be done by pressure drop test or soapy water at joints and outlets</li> <li>Shut liquid and gas valves to</li> </ol> </li> </ol>	None held on site
Pollutant	Storage	Current	Emission	PPE (1)		Quantity Stored on Site
----------------	--------------------------------	----------------------------	--	--------------------------------	---	---
	Location	MSDS held Yes/No	control equipment <sub>(1)</sub>		Spill Clean Up Method <sub>(1)</sub>	
					stop leak if possible and safe to do so.	
Surfactant	Workshop designated area	Yes – held in office	Soil, sand, vermiculite	Safety glasses, PVC gloves	<ul> <li><u>Small spills</u></li> <li>1) Contain using sand or diatomaceous earth</li> <li>2) Collect and seal in properly labelled drums</li> <li>3) Wash residue with water</li> <li><u>Large Spills</u></li> <li>1) Restrict access to area</li> <li>2) Provide PPE</li> <li>3) Remove chemicals which react with spill of material</li> <li>4) Spills are slippery</li> <li>5) Contain spill or leak</li> <li>6) Do not allow entry to drains or water ways</li> <li>7) Spilled material should be contained by dyking with inert material, sand, soil etc.</li> <li>8) Solutions can be recovered or carefully diluted with water</li> </ul>	80 litres total 4 – 20 litre containers
Dry Powders	Designated workshop area	Yes – held in office	Access to council sweeper, soil, sand, vermiculite	Safety glasses, PVC Gloves,	Accidental Release Measures Emergency procedures: Prevent entry to area by unprotected personnel. Methods and material for containments and clean up 1) Vacuum or wet sweep material avoiding generation of dusts. 2) A fine water spray should be	<u>180kgs_total</u> <u>4 – 20kg cement bags</u> <u>100kgs dehydrated lime for ASS</u>

Pollutant	Storage Location	Current MSDS held Yes/No	Emission control equipment <sub>(1)</sub>	PPE <sub>(1)</sub>	Spill Clean Up Method (1)	Quantity Stored on Site
					<ul> <li>used to suppress dust when sweeping.</li> <li>3) Product dampened with water may be collected with a clean shovel.</li> <li>4) Seal all spilled product and wastes in vapour tight labelled plastic containers for reuse/recycle where possible or eventual disposal.</li> </ul>	
Welding gas	Designated workshop area	Yes – held in office	Ventilation	Respirator	<ul> <li><u>Occupational Release:</u></li> <li>1) Avoid heat, flames, sparks and other sources of ignition.</li> <li>2) Stop leak if possible without personal risk.</li> <li>3) Reduce vapours with water spray</li> <li>4) Keep unnecessary people away, isolate hazard area and deny entry.</li> <li>5) Remove sources of ignition.</li> <li>6) Ventilate closed spaces before entering.</li> </ul>	<u>No storage of cylinders on site</u> <u>Contractors supply with jobs</u> <u>undertaken</u>
Effluent	Tanks by office	Yes – held in office	Access to council commercial vacuum/ pump truck,, soil, sand, bleach, hydrated lime	PVC Gloves, goggles, overalls	<ul> <li><u>Accidental Release Measures</u></li> <li>1) Contaminated area must be clearly marked or cordoned off to restrict access.</li> <li>2) Protective clothing should be worn when cleaning up a sewage spill.</li> <li>3) If the spilled material can't be recovered using hand tools, a</li> </ul>	3000 litre septic tank

Pollutant	Storage Location	Current MSDS	Emission control	PPE (1)		Quantity Stored on Site
		held Yes/No	equipment <sub>(1)</sub>		Spill Clean Up Method (1)	
					<ul> <li>commercial vacuum / pump truck should be called to remove all visible liquid and solid material.</li> <li>4) When the area is visibly clean, either a chlorine / water solution or hydrated lime should be applied to the spill area to disinfect.</li> <li>5) If a major spill has occurred hydrated lime should be applied to the area in place of chlorine bleach</li> <li>6) .Enough hydrated lime should be applied to raise the pH to at least 12. By raising the pH to 12 for at least 1 hour, the area will be disinfected.</li> <li>7) Because lime is a caustic material, access to the area treated with lime must be restricted during the disinfection period</li> </ul>	
Untreated water from sediment Dams*	Site dams	Not applicable	Freeboard	Not applicable	Discharge Measures           1) Assess the water for TSS, pH and oil and grease           2) If within the consent and licence parameters, no further action           3) If outside of the consent and licence parameters, enact the PIRMP and assess for likelihood or actual material harm.           4) If any are available, implement measures available for reducing flow and enacting clean up. le	Dunloe has 2 main dams One for sediment One for operational dredging Est 3/400 ML in both

Pollutant	Storage Location	Current MSDS held Yes/No	Emission control equipment <sub>(1)</sub>	PPE (1)	Spill Clean Up Method <sub>(1)</sub>	Quantity Stored on Site
					lowering water levels in dam, putting in place coir logs, use of flocculant or coagulant, etc.	

(1) This information is drawn from a review of the MSDS or manufacturer/supplier technical information

\*Note: this is only a potential pollutant and must be tested on discharge into the receiving environment to determine whether the contents within the water are at levels considered to be pollutant

## 7.3 Emergency Response Map

In addition to the PIDS the site needs to prepare an emergency response map that provides the following information;

- address of site
- location of pollutant storage
- location of safety equipment
- emergency evacuation / muster points
- stormwater drains / flow paths
- sensitive receivers
- sediment dam overflow locations
- location of MSDS
- surrounding area that is likely to be affected by a pollution incident
- discharge location of stormwater drains to nearest water coarse or water body

Existing site maps that have been developed to comply with Holcim SHE system requirement 1.84 may be used if all the required items have been included. If an existing map is not available it should be created.

It is important to clearly identify these items so as to be able to respond in an emergency situation.



26

## 7.4 Pollution Incident Response Management Plan

If it is suspected that an incident may cause material environmental harm the Pollution Incident Management Response Plan will be executed. This plan is based on seven phases:

1) Assess

2) Stop

3) Notify

4) Contain

5) Mitigate

6) Clean up

7) Review

Details of the requirements and responsibilities for each phase are explained below.





#### Contact key individuals

- Individuals responsible for activating and managing plans (nominated site representatives)
- Individuals authorised to notify and coordinate relevant authorities (nominated site representatives)

#### **Contact Relevant Authorities**

- Firstly, call 000 if the incident presents an immediate threat to human health or property.
- If the incident does not require an initial combat agency, or once the 000 call has been made, notify the relevant authorities in the following order. The 24-hour hotline for each authority is given when available:
- the Appropriate Regulatory Authority
- EPA
- Ministry of Health via the local Public Health Unit
- WorkCover Authority
- the local authority if this is not the ARA
- Fire and Rescue NSW



- Utilise barriers (absorbent booms, banks of soil or any other safe objects) or spill absorbent to prevent the emission from spreading.
- When an emission is on a hard surface use appropriate absorbent materials ie absorbent granules or sand
- The main priority is to prevent the emitted material from discharging off site





- Conduct an investigation into the event and assist the EPA and investigators with external enquiries
- Complete internal reporting;
  - As per Holcim SHE requirement 5.1
- Test the effectiveness of Pollution Incident Response
   Management Plan annually and one month after the incident to ensure controls are replenished.

Testing protocol is provided in appendix B

## 7.5 Communication Strategy

It is a legal requirement of the Protection of the Environment Operations (POEO) Act, to notify key stakeholders in neighbouring properties that may been affected by an incident.

Communicating with neighbours and the community in the event of an environmental incident I is vital as they have a right to know about any spill that can potentially lead to material harm to their properties or themselves. The communication strategy in the PIRMP provides sites with a method of communicating with key stakeholders.

Key stakeholders include neighbouring residential and/or commercial properties, sensitive receivers ie farms, hospitals schools within the impact area. Consideration must be given to sensitive receivers that may be affected if the emission reaches a water body. For example a farmer that is cultivating crops down river from your site will need to be informed of a spill to prevent him spraying his crops with polluted irrigation water.

The PIRMP must include details of the mechanisms that will be used for providing early warnings and regular updates to the owners and occupiers of premises who may be affected by an incident occurring at the premises.

The communication strategy should also make reference to any actions or arrangements that will be in place to minimise the risk of harm to any persons who will be on the premises or who are likely to be on the premises at the time of an incident. This is a legislative requirement that needs to be included in the PIRMP.

For a table detailing the communication strategy for this site:

#### Refer to Appendix C – Community Notification Strategy



## 7.6 Staff Training

Sites need to develop a toolbox talk based on the PIRMP. This training should be delivered to all appropriate personnel on site and be conducted to include potential scenarios that may require implementation of the plan.

#### Frequency of training

Training for site staff should be repeated annually, and after each update to the plan. In the event of an incident requiring the PIRMP to be activated a training drill should be carried within one month of the incident occurring.

#### How Records of training are kept

Training records should be stored on site and in the Chris 21 data base. This data base is the primary online tool for tracking individual staff training records and frequency for training and refresh courses. These records are to be made available to relevant authorities on request.

### 7.7 Continual Improvement

It is a legislative requirement for this plan to be tested and updated on an annual basis and within one month of an incident.

To complete this requirement a Pollution Incident Response Test Checklist has been prepared and provided as Appendix B. The checklist includes the major elements of the plan that require testing:

- Contact numbers
- Evacuation drills
- Desktop assessment
- Staff training and awareness
- Environmental controls & PPE

Desktop assessments require site personnel, responsible for testing the plan, to select a scenario from the hazard and impact register (table 2) and ensure that all the required controls for the scenario are in place. During the desktop assessment environmental control and PPE equipment supplies should be inspected to ensure that they are functional and that there are enough materials to ensure that emissions relating to the scenario can be controlled effectively and safely.

### 7.8 Review

This PIRMP is to be reviewed and updated at least annually to ensure that incident response systems are fully functioning and are ready to be implemented if an incident occurs. This requirement shall be listed as an action item and scheduled on the environmental compliance planner.



Strength, Performance, Passion

# Appendix A - Emergency Contact Details



Contacts	Phone Number
Individuals responsible for activating the	Production Supervisor
plans and managing the response	Daniel Dwyer – 0411 795 060
Individuals Authorised to Notify and	Quarry Manager
Coordinate Relevant Authorities	Garth Stacey – 0429 790 217
Emergency Services	Fire
	(02) 6592 6999
	Police
	(02) 6552 0399
	Ambulance
	131 233
Fire & Rescue NSW	(02) 6551 5246
EPA	131 555
The Ministry of Health via the local Public	The Tweed Hospital
Health Unit	(07) 5536 1133
WorkCover Authority	13 10 50
Local Council	Tweed Shire Council
	(02) 6592 5399
Fire and Rescue NSW	000



Strength, Performance, Passion

# Appendix B - Pollution Incident Response Test Checklist



#### Strength. Performance. Passion

Date:	
Site:	 
Address:	 

Pollution Incident Scenario: .....

#### Instructions

1. Select an Environmental Incident applicable to the site to test in a Pollution Scenario (this may include a major spill, equipment failure or breaches of license consent that may cause impacts onsite and to the surrounding community);

2. Using the scenario conduct a desktop review using the Test Checklist as a prompt to ensure that each component of the PIRMP is up to date;

3. Sign off the checklist, scan and send to the NSW Planning & Environmental Coordinator;

4. Planning & Environmental Coordinator will make amendments to the plans and submit these to the site managers for review and approval;

5. Site Managers to hold a tool box talk with staff on the details of the PIRMP and keep a copy of the PIRMP onsite for future reference.

	Phone Numbers	
Are all contact details within the plan current and up to date?	Currency	Updated Number
Individuals responsible for activating the plans and managing the response		
Individuals Authorised to Notify and Coordinate Relevant Authorities		
Emergency Services		
EPA		
The Ministry of Health via the local Public Health Unit		
WorkCover Authority		
Local Council		
Fire and Rescue NSW		
Additional Contacts relevant to the licensee's premises		
Other Organisations or agencies that need to be advised of the incident		



Г

Environmental Hazards and Control Standards	Yes/ No	Actions
Are the descriptions of environmental hazards up to date?		
Are the potential and likelihood of incidents that could occur still correct and relevant to the site operations?		
Are the pre-emptive actions for risk management of the relevant activity correct and relevant to the site?		
Is there an inventory of pollutants (including quantities of pollutants onsite)?		
Is the listed safety equipment & PPE correct and up to date?		
Is there a map/s located onsite detailing the following;		
- The site and the surrounding area likely to be affected in the event of an incident		
- The Locations of storage/ holding points of pollutants		
- Stormwater drains and discharge points offsite		
Are the nature and objectives of staff training set out in the plan?		
Are there details of mechanisms for providing early warnings and regular updates to the owners and occupiers?		
Is there a copy of the plan onsite and up to date?		

1

Has there been an evacuation drill in the last 12 months?
Date:
Notes:



## Improvements to the Pollution Incident Response Management Plan:

1).	
2).	
3).	
ý 4).	
5)	
6).	
υ,.	

## **Comments / Recommendations / Review**

1)
2)
3)
4)
5)
6)
7)
8)
9)
10)

Pollution Incident Response Test Checklist Assessor:
Signed:



Strength, Performance, Passion

# Appendix C – Community Notification Strategy



#### Strength. Performance. Passion

 If there is an Environmental Incident that has the potential to cause harm to the following stakeholders they will be contacted by TELEPHONE

Stakeholder Component	Name	Contact Information
Quarry Neighbours	Sasha Peterson	Ph: 0401 805 446
Quarry Neighbours	Noni Woodward	Ph: 0407 776 173

 $\label{eq:product} \textbf{Appendix} \ \textbf{L} - \text{Environmental monitoring program}$ 



# Holcim (Australia) Pty Ltd

Dunloe Sand Quarry Environmental Monitoring Program

July 2021

This report: has been prepared by GHD for Holcim (Australia) Pty Ltd and may only be used and relied on by Holcim (Australia) Pty Ltd for the purpose agreed between GHD and the Holcim (Australia) Pty Ltd as set out in section 1 of this report.

GHD otherwise disclaims responsibility to any person other than Holcim (Australia) Pty Ltd arising in connection with this report. GHD also excludes implied warranties and conditions, to the extent legally permissible.

The services undertaken by GHD in connection with preparing this report were limited to those specifically detailed in the report and are subject to the scope limitations set out in the report.

The opinions, conclusions and any recommendations in this report are based on conditions encountered and information reviewed at the date of preparation of the report. GHD has no responsibility or obligation to update this report to account for events or changes occurring subsequent to the date that the report was prepared.

The opinions, conclusions and any recommendations in this report are based on assumptions made by GHD described in this report. GHD disclaims liability arising from any of the assumptions being incorrect.

GHD has prepared this report on the basis of information provided by Holcim (Australia) Pty Ltd and others who provided information to GHD (including Government authorities), which GHD has not independently verified or checked beyond the agreed scope of work. GHD does not accept liability in connection with such unverified information, including errors and omissions in the report which were caused by errors or omissions in that information.

# **Table of contents**

1.	Introduction	1
2.	Monitoring summary	2

## **Table index**

Table 2-1	Monitoring summary	 .2
Table 2-1	wonitoring summary	

## **Appendices**

Appendix A - Environmental Inspection Checklist

## 1. Introduction

The Environmental Monitoring Program provides a summary of the monitoring required at Dunloe Quarry and addresses Condition 2, Schedule 5 of Development Consent 06\_0030. Table 2-1 summarises the monitoring requirements from the individual sub plans in the Environmental Management Strategy (EMS).

The intent of the Environmental Monitoring Program is to consolidate the environmental monitoring requirements for Dunloe Sand Quarry in a single document. If further information is required regarding the monitoring, objectives, or contingencies if the monitoring identifies an impact, the individual sub plan will be referred to.

# 2. Monitoring summary

### Table 2-1 Monitoring summary

Aspect	Frequency	Details Analysis and interim trigger values/Criteria		Responsibility	
General					
Routine inspections	Weekly	Routine weekly inspections by the Quarry Manager (or delegate) will occur throughout the operational lifetime of the quarry, using the <i>Environmental Inspection Checklist</i> in Appendix A.	-	Quarry Manager	
Soil and Water Manag	ement Plan (SWMP)				
Weather	Daily	Weather forecasts will be monitored to inform quarry operations, for example:	-	Quarry Manager	
		<ul> <li>If fair is forecast, sediment and erosion controls will be checked and maintained.</li> <li>If dry weather and winds are forecast, dust controls will be implemented.</li> </ul>			
Rainfall	Daily	Rainfall at the premises must be measured and recorded in millimetres per 24 hour period, at the same time each day.	-	Quarry Manager	
Erosion and sediment controls	diment Weekly and following Erosion and sediment controls are to be monitored and - rain (>10 mm in 24 hr) maintained.		Quarry Manager		
Blue Green Algae monitoring within the	October to April – fortnightly	A sample will be collected from the extraction ponds and analysed for blue green algae.	<50,000 cells/mL ( <i>M.aeruginosa</i> ) <4 mm 3/L (total biovolume)	Quarry Manager	
extraction area	May to September - monthly	If results indicate cell growth to a level exceeding 500 cells/mL, weekly sampling will be implemented until such time as the results of the testing indicate <500 cells/mL for a period of greater than 3 months			
	Quarterly	Chlorophyll a	2-10 μg/L	Quarry Manager	
Surface water monitoring within the extraction area	Monthly	Water quality monitoring at Pond 1 and Pond 2 locations on Figure 6-1 in the SWMP.	pH - 5.0 – 8.5 Electrical conductivity - <5.50 mS/cm Dissolved oxygen - >4.00 mg/L Turbidity - <20 NTU Oil and grease - <10 mg/L	Quarry Manager	

Aspect	Frequency	Details	Analysis and interim trigger values/Criteria	Responsibility
	Quarterly	Water quality monitoring at Pond 1 and Pond 2 locations on Figure 6-1 in the SWMP.	As above monthly monitoring, plus: Manganese - $0.15 \text{ mg/L}$ Magnesium - $40 \text{ mg/L}$ Sodium - $280 \text{ mg/L}$ Potassium - $17.5 \text{ mg/L}$ Bicarbonate - $400 \text{ mg/CaCO3}$ Chloride - $285 \text{ mg/L}$ Sulfate - $175 \text{ mg/L}$ Aluminium - $0.75 \text{ mg/L}$ Arsenic - $<0.005 \text{ mg/L}$ Iron - $<7.5 \text{ µg/L}$ Chlorophyll a - $2-10 \text{ µg/L}$	Quarry Manager
	Quarterly	Vertical profile monitoring at one-metre intervals will be undertaken in the active extraction area.	pH - 5.0 – 8.5 Electrical conductivity - <5.50 mS/cm Dissolved oxygen - >4.00 mg/L Turbidity - <20 NTU Oil and grease - <10 mg/L	Quarry Manager
	When discharging either naturally (i.e. solely as a result of rainfall at the premises less than 82.5 millimetres over any consecutive five day period) or manually (i.e. pumped) but not if reused on site (e.g. dust suppression, wash plant)	Sampling at EPA Point 1 and EPA Point 2, as shown in Figure 6-1 in the SWMP. Sampling is to be done once <24 hours prior to; and, sampling the discharge daily during, each discharge event arising from rainfall of <b>less than</b> 82.5 mm in total over a period of up to five days duration. If the concentration limits are not achieved, the water will need to be treated and resampled. If the results do not achieve the criteria, treatment of the water may be required.	Oil and grease – Nil visible pH – 6.5-8.5 TSS – 50 mg/L	Quarry Manager

Aspect	Frequency	Details	Analysis and interim trigger values/Criteria	Responsibility
Surface water	Rainfall/event based and quarterly	Water quality monitoring at the locations SW3, SW4, SW9 and SW10 on Figure 6-1 in the SWMP would be conducted monthly and during large rainfall events. Department of Planning, Industry and Environment (DPIE) acknowledges that short term exceedances of these objectives may occur during natural events such as flooding. DPIE acknowledges that pre-existing water quality may not meet the objectives for some analytes, including salinity. Holcim must strive to meet the water quality objectives through implementation of the SWMP, as far as is reasonable and feasible and within the Proponent's control, to the satisfaction of the Secretary.	pH – 5.5-7.5 EC – 1800-2400 µS/cm Suspended solids - <25 mg/L Dissolved oxygen - >6 mg/L Total nitrogen - <1 mg/L Total phosphorus - <0.08 mg/L	Quarry Manager
Streambank and bed profile	Rainfall/event based and quarterly	Streambank and bed profile and conditions at the locations SW3, SW4, SW9 and SW10 on Figure 6-1 in the SWMP will be visually inspected during surface water sampling events.	Any changes due to site operations are identified and repaired.	Quarry Manager
Spill kit	Monthly and following use	The spill kit is to be checked and any missing materials to be replaced.		Quarry Manager
Groundwater	Monthly	The groundwater monitoring well locations are shown on Figure 6-1 in the SWMP. DLP1, DLP3, DLP5, DLP6 and DLP7 are to be monitored during Stage 1 of the extraction. DLP5, DLP6 DLP7, DLP8, and DLP10 are to be monitored during Stage 2 of the extraction.	pH – 4.2 – 7.0 EC – <2.0 mS/cm Dissolved oxygen - >1.50 mg/L Level - <20% change from historical levels	Quarry Manager
	Quarterly	objectives may occur during natural events such as flooding. DPIE acknowledges that pre-existing water quality may not meet the objectives for some analytes, including salinity. Holcim must strive to meet the water quality objectives through implementation of the Soil and Water Management Plan, as far as is reasonable and feasible and within the Proponent's control, to the satisfaction of the Secretary.	As monthly monitoring, plus: Ammonia - NA Calcium – 55 mg/L Magnesium – 0.40 mg/L Sodium - 280 mg/L Potassium – 17.5 mg/L Bicarbonate - 400 mg/L Sulfate - 175 mg/L Chloride - 285 mg/L Dissolved iron – 7.5 mg/L Dissolved aluminium – 0.75 mg/L Dissolved arsenic – 0.005 mg/L Oil and grease – NA	Quarry Manager

Aspect	Frequency	Details Analysis and interim trigger values/Criteria		Responsibility		
Noise Management Pla	an (NMP)					
Noise	Quarterly       Attended noise compliance monitoring will occur at R6, R7       R6 and R7 – 42 dB(A)         and R8 (refer to Figure 3-1 in the NMP) quarterly for the first two years following approval of the NMP. If this monitoring indicates compliance with the criteria and DPIE approve, the noise monitoring will cease unless there is:       R8 – 48 dB(A)         All other residences - 41 dB(A)         Quarry operations       A change in operating conditions that are likely to increase noise emissions from the site         The assessment must be conducted by a suitably qualified and experienced acoustic consultant in accordance with the Noise Policy for Inductry (EPA 2017)		Quarry Manager			
Air Quality Manageme	Air Quality Management Plan (AQMP)					
Dust	Monthly	Deposited dust is to be monitored at four locations, surrounding the quarry (as shown on Figure 3-1 in the AQMP). Airborne particulate monitoring of PM10 and TSP is only required to be undertaken if annual production rates increase to 200,000 tonnes or above, or in the event of a valid complaint relating to Dunloe Sand Quarry operations. Monitoring may cease after 2 years for deposited dust and three months for PM10 and TSP – refer to AQMP for details.	PM10 (annual) - 30 μg/m <sup>3</sup> PM10 (24 hour) - 50 μg/m <sup>3</sup> TSP (annual) – 90 μg/m <sup>3</sup> Deposited dust - 4 g/m <sup>2</sup> /month	Quarry Manager		
Landscape Manageme	ent Plan (LMP)					
Routine Rehabilitation Monitoring	Quarterly	Monitor the whole area of each of the three Rehabilitation Areas or the 13 permanent monitoring locations shown on Figure 5-1 in the LMP using Form A (Appendix F of the LMP)	-	Quarry Manager		
Site Condition	Six Monthly	Monitor the whole area of each of the three Rehabilitation Areas or the 13 permanent monitoring locations shown on Figure 5-1 in the LMP using Form B (Appendix F of the LMP)	-	Quarry Manager		
Revegetation / Forest Structure	Annually (end-calendar- year)	Monitor the 13 permanent monitoring locations, as shown on Figure 5-1 in the LMP using Form C (Appendix F of the LMP)	-	Quarry Manager		

Aspect	Frequency	Details Analysis and interim trigger values/Criteria		Responsibility
Floristic Composition	Annually (end-calendar- year)	Monitor the 13 permanent monitoring locations, as shown on Figure 5-1 in the LMP using Form D (Appendix F of the LMP)	-	Quarry Manager
Photographs at established photo points	Quarterly	Thirteen permanent photo points have been established (Figure 5-1 in the LMP) where photographs will be taken at regular intervals to provide a visual indication of plant growth (height and extent) and weed presence. Photographs shall be taken at the SW, SE, NW, NE corners of each monitoring site.	-	Quarry Manager
Fauna Box Monitoring	Six Monthly	Refer to Fauna Box Monitoring Form (Appendix F of the LMP)	-	Quarry Manager
Koala Monitoring	On-going	Refer to Section 1.3 Appendix E of the LMP	-	Quarry Manager
Transport Managemer	nt Plan (TMP)			
Truck movements	Truck movements Daily Truck movements to and a arrival and dispatch) will b the Dunloe Daily Sales reading to the Dunloe Daily Sales reading t			Quarry Manager
Routine check	Daily	Holcim trucks and machinery are to be checked daily before - use, using the pre-start check in Appendix C of the TMP.		Quarry Manager
Waste Management Pl	an (WMP)			
Waste	Weekly	Waste removed from site will be recorded in a waste register along with the relevant dockets from the contractors removing the material.	-	Quarry Manager

# **Appendices**

GHD | Report for Holcim (Australia) Pty Ltd - Dunloe Sand Quarry, 2220056

Appendix A – Environmental Inspection Checklist

# **Environmental Inspection Checklist**

Inspection Date :	Inspection Conducted By:
Weather Conditions: Dry Slight Wind Calm Rain	Strong Wind
Rainfall (Past 5 Days):	

ltem	Observation	Action required & location	Responsibility	Closed
Soil and V	Vater			
	Erosion and Sediment controls installed & in good working order?			
	Are all disturbed areas and stockpiles draining back to the ponds?			
	Are basins full and requiring treatment and discharge? If yes, check water quality			
	Are monitoring points discharging? If yes, check water quality			
	Are clean and dirty water separated?			
	Are extracted materials hydraulically separated with pyretic fines being returned at depth no less than 3 m from the water surface?			
	Are blue green algae signs in place?			
	Are survey pegs in place?			
	Are site access locations free from mud or excessive dirt on the local roads? Do any rubble grids installed require maintenance?			
Air Quality	1		·	
	Is dust from quarry operation visible during inspection?			
	Are dust suppression measures in place? And effective?			
	Are non-active areas and topsoil stockpiles stabilised?			
	Is any machinery emitting smoky exhaust?			
	Are truck loads leaving site covered?			
Flora and	Fauna			
	Is vegetation protected with fencing? Is fencing in good repair?			
	Are hygiene measures being implemented?			
	Are there any weeds?			
	Have any unexpected species of fauna (eg koala) been identified / relocated in the past week?			
Noise & V	ibration		_	
	Are noise controls installed and effective?			
	Have all activities occurred within permitted hours in the last week?			
	Are any works outside the approved operation hours proposed?			
	Do plant/equipment have appropriate mitigation measures installed and been serviced?			

ltem	Observation	Action required & location	Responsibility	Closed
Traffic			•	
	Are trucks leaving site clean?			
	Are trucks travelling at 40km/h or less?			
Heritage			•	
	Have any unexpected finds of heritage items or human remains occurred in the past week?			
Hazardou	s Substances & Dangerous Goods			
	Are spill kits available in all work areas and in stock?			
	Is there evidence of hydraulic/vehicle oil spills/leaks on site?			
	Are stored chemicals in bunded areas with at least 110% storage capacity of the largest container? Is the bunded area free from accumulated storm water or spilled chemicals?			
Waste Ma	nagement		+	
	Is the site free from litter and waste bins available?			
	Are appropriate and segregated waste bins available?			
Complain	is	_		
	Have any complaints relating to environmental issues been received in the past week? If so, please detail.			
Additonal Opportuni improvem	Items/ ties for ent/ Innovations			
Signature				
Signature				

GHD

230 Harbour Drive Coffs Harbour NSW 2450 T: 61 2 6650 5600 F: 61 2 9475 0725 E: cfsmail@ghd.com

© GHD 2021

This document is and shall remain the property of GHD. The document may only be used for the purpose for which it was commissioned and in accordance with the Terms of Engagement for the commission. Unauthorised use of this document in any form whatsoever is prohibited.

22-SO-976847825-35/https://projects.ghd.com/oc/Newcastle3/holcimdunloesandquar/Delivery/Documents/2220056\_R PT-A\_Dunloe Environmental Monitoring Plan.pdf.docx

**Document Status** 

Revision	Author	Reviewer		Approved for Issue		
		Name	Signature	Name	Signature	Date
0	B Luffman	S Lawer		S Lawer		17/10/2019
1	B Luffman	S Lawer		S Lawer	0	18/12/2019
2	B Luffman	S Lawer	Jan )	S Lawer	tan	15/07/2021

# www.ghd.com


GHD 230 Harbour Drive Coffs Harbour NSW 2450 T: 61 2 6650 5600 F: 61 2 6650 5601 E: cfsmail@ghd.com

## © GHD 2021

This document is and shall remain the property of GHD. The document may only be used for the purpose for which it was commissioned and in accordance with the Terms of Engagement for the commission. Unauthorised use of this document in any form whatsoever is prohibited. 22-SO-976847825-

11/https://projects.ghd.com/oc/Newcastle3/holcimdunloesandquar/Delivery/Documents/2220056\_R PT\_Dunloe Environmental Management Strategy.docx

## **Document Status**

Revision	Author	Reviewer		Approved for Issue		
		Name	Signature	Name	Signature	Date
0	B Luffman	S Lawer		S Lawer		17/10/2019
1	B Luffman	S Lawer		S Lawer		13/01/2020
2	B Luffman	S Lawer		S Lawer	_	17/07/2020
3	B Luffman	S Lawer	fan	S Lawer	fan	15/07/2021

## www.ghd.com

